K.M.J. Menon Aaron C. Goldstrohm *Editors*

Post-transcriptional Mechanisms in Endocrine Regulation



Post-transcriptional Mechanisms in Endocrine Regulation

K.M.J. Menon • Aaron C. Goldstrohm Editors

Post-transcriptional Mechanisms in Endocrine Regulation



Editors K.M.J. Menon Department of Biological Chemistry University of Michigan Medical School Ann Arbor, MI, USA

Department of Obstetrics and Gynecology University of Michigan Medical School Ann Arbor, MI, USA Aaron C. Goldstrohm Department of Biological Chemistry University of Michigan Medical School Ann Arbor, MI, USA

ISBN 978-3-319-25122-6 DOI 10.1007/978-3-319-25124-0 ISBN 978-3-319-25124-0 (eBook)

Library of Congress Control Number: 2015960457

Springer Cham Heidelberg New York Dordrecht London

© Springer International Publishing Switzerland 2016, Corrected Publication 2020

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Printed on acid-free paper

Springer International Publishing AG Switzerland is part of Springer Science+Business Media (www.springer.com)

Preface

The endocrine system is both diverse and complex, controlling a wide array of physiological processes and cellular responses. Hormones induce a variety of effects on cells, including regulation of metabolic processes, by increasing the levels or activity of enzymes and proteins in the target cells. Hormone responses can be amplified through cascades of signal transduction pathways leading to changes in gene expression including synthesis, processing, stability, and translation of mRNAs. Moreover, hormone responses can change enzyme activity through covalent modifications. While activation of gene expression and covalent modification of proteins have been the intense focus of studies for over the past several decades, recent advances in the RNA biology have provided new insights into the role of post-transcriptional mechanisms involved in cellular responses to hormones and other biomolecules.

Throughout biology, post-transcriptional regulatory mechanisms play an essential role in controlling dynamic gene expression. The outcome of this regulation includes control of the amount, timing, and location of protein expression. Regulation is mediated by cis-acting RNA sequences and structures and transacting RNA-binding proteins and noncoding RNAs, including microRNAs. Recent advances in the characterization of these regulatory factors have revealed enormous regulatory potential; thousands of new RNA-binding proteins and microRNAs, which control protein expression in fascinating ways, have been identified in mammals. The goal of this book is to highlight the advances made in the understanding of the regulatory mechanisms by which hormones control these processes.

The first chapter of this volume provides an overview of our current understanding of the various components of post-transcriptional mechanisms including regulatory factors, elements, and general mechanisms that control protein expression. The intent is to provide early stage investigators with an introduction, emphasizing key concepts underlying post-transcriptional regulatory mechanisms.

In subsequent chapters, specific system-based studies on post-transcriptional regulation are described. These chapters were composed by leading experts who summarize current state of knowledge, remaining key questions, and provide perspective for future research goals.

Chapter 2 describes the intricate regulation of expression of insulin and insulin-like growth factors (IGF). The mRNAs encoding these peptide hormones are subject to multiple regulatory mechanisms including alternative RNA processing, regulated mRNA stability, and modulation of translation efficiency. RNA-binding proteins, RNA degrading and modifying enzymes, and noncoding regulatory RNAs (i.e., microRNAs) control expression of insulin and IGF. This work has important relevance to the physiology of energy homeostasis, diabetes, and cancer.

In Chap. 3, mechanisms of regulation of cytokine mRNA during inflammatory response are presented. Emphasis is placed on the cis-acting elements of cytokine mRNAs including the now classic adenosine-uridine rich elements (AREs), which act as bifunctional switches to modulate cytokine mRNA stability. The authors introduce the concepts of combinatorial regulatory control and competitive binding, which form regulatory networks to control cytokine production and modulate innate immune responses. Defects in these control mechanisms contribute to inflammatory autoimmune diseases and cancer.

Chapters 4–7 deal with post-transcriptional regulation of target cell function by reproductive hormones. These processes have important roles in reproductive processes and dysfunction contributes to diseases such as infertility and cancer. In Chap. 4, a novel mechanism of regulation of luteinizing hormone receptor (LHR) mRNA expression by an mRNA-binding protein is presented. In response to physiological changes in the secretion of LH, the expression of LHR mRNA in the ovary undergoes rapid changes, the most striking change occurring after LH surge to induce ovulation. During this period the LH receptor mRNA expression is transiently downregulated by increasing LHR mRNA degradation that is mediated by an unanticipated LHR mRNA-binding protein-a steroid metabolism protein that "moonlights" as an RNA-binding regulator. This chapter highlights important concepts including regulation though mRNA localization, mRNA stability, and translational control. In Chap. 5, the regulation of estrogen receptor, a member of the nuclear receptor family, by its own ligand is described. Evidence is presented showing that estrogen stabilizes its receptor by increasing the binding of a specific protein to the receptor's mRNA. Moreover, microRNAs contribute to translational control of estrogen receptor.

Chapter 6 focuses on the regulation of ovarian function by microRNAs. This chapter provides information on the synthesis and processing of microRNAs. The mechanisms of translational control and mRNA degradation caused by microRNAs are presented. The regulation of microRNAs by differential processing and competitive inhibition by endogenous competitor RNAs is also considered. The authors then describe the important roles of microRNAs and their processing enzymes in controlling ovarian gene expression during reproductive cycles.

Chapter 7 provides a detailed discussion of post-transcriptional control in the germline. Regulation of translation, RNA localization, and decay pervades the germline, gametes, and early embryos in part due to the quiescence of transcription during early developmental stages. Maternal mRNAs are deposited in developing oocytes, stored in a repressed state within RNA-protein granules, and subsequently activated to drive development in response to hormonal cues and fertilization.

Moreover, post-transcriptional control specifies body pattern formation and designates the primordial germline. This chapter highlights multiple key regulators that control these crucial events.

Chapter 8 focuses on VEGF-A, a hormone that controls angiogenesis during development and wound healing, among other processes. Proper control of VEGF-A is crucial and dysregulation contributes to cancer and other disorders. Control of VEGF-A is intricate, as the authors guide the reader through the diverse mechanisms that control VEGF-A expression and function including alternative mRNA processing, mechanisms of destabilization and stabilization of the mRNA, and translational control, mediated by cis-acting RNA elements, RNA-binding proteins, and microRNAs.

Prostaglandins are lipid hormones that control reproductive physiology and inflammation. Chapter 9 explores the pathway of prostanoid biosynthesis and the post-transcriptional mechanisms that control these enzymes. Multiple RNA-binding proteins and microRNAs control translation, stability, and localization of mRNAs encoding prostanoid biosynthesis enzymes. This regulation involves regulated and coordinated mRNA degradation pathways and enzymes. Additionally, mRNA localization to specialized intracellular granules contributes to the control of prostaglandins, which is crucial since unregulated expression contributes to cancer and inflammatory diseases.

Chapter 10 explores post-transcriptional regulation of production of the secreted peptide hormone leptin. Leptin is produced by fat cells and contributes to the control of appetite and energy storage and homeostasis. This hormone has a major impact on obesity and dieting. Leptin production is controlled at the post-transcriptional level by insulin signaling through the important mammalian target of rapamycin pathway (mTOR), which controls translation efficiency of mRNAs.

Post-transcriptional mechanisms that control parathyroid hormone gene expression are explored in Chap. 11. Parathyroid hormone has crucial role in the control of mineral metabolism and bone strength. Regulation of this secreted peptide hormone is achieved by RNA-binding proteins that interact with the hormone's mRNA to stabilize or degrade the mRNA in response to serum calcium, phosphate, and vitamin D.

Chapter 12 illuminates the pathways of steroid hormone biosynthesis, which broadly influence metabolism, physiology, reproduction, and immune function. The authors explain the roles of post-transcriptional and post-translational mechanisms that operate to control steroid production, emphasizing the roles of microRNAs in translational control and mRNA degradation of the biosynthetic enzymes.

Glucocorticoids are steroid hormones that control immune function and stress response. Chapter 13 examines the post-transcriptional regulation by glucocorticoid including modulation of cytokine translation and mRNA stability. Multiple mechanisms appear to contribute to regulation. Glucocorticoid receptor, a ligand-activated transcription factor, regulates RNA synthesis to modulate signal transduction and the expression and activity of RNA-binding proteins and microRNAs. These effects drive changes in mRNA stability and translation. Surprisingly, recent evidence

indicates that glucocorticoid receptor can bind to mRNA directly to control degradation of specific mRNAs in the cytoplasm.

Chapter 14 surveys post-transcriptional regulation by brain-derived neurotrophic factor (BDNF) in the nervous system. Protein expression in neurons is controlled by intricate mechanisms including RNA localization to synapses and repression and activation of specific mRNAs in response to signals. These mechanisms contribute to the control of neuronal activity, synaptic plasticity, and longer-term memory formation. BDNF plays an important role in controlling activity-dependent gene expression in neurons by globally enhancing translation through mTOR pathway. Certain mRNAs are specifically affected by BDNF signaling, and this selectivity is determined by the activities of specific RNA-binding factors that control translation and mRNA stability.

We thank the authors for their time and effort to contribute to this book and for their cooperation for timely submission. We hope the information presented in this book will be a valuable source of current state of knowledge for experts as well as beginners who wish to pursue future research in this exciting area.

Finally we thank Springer for the opportunity to edit this volume dealing with post-transcriptional mechanism in endocrine regulation. Special thanks to former Editor, Meredith Clinton, of Endocrinology at Springer for her support in the initial stages and Kelly Wilson, Associate Editor, Endocrinology, Springer Science and Business Media, for her support in making our effort a reality. We appreciate the assistance of Kelly Studer, Administrative Assistant at the University of Michigan, who cheerfully provided us administrative support during the course of this project.

Ann Arbor, MI, USA

K.M.J. Menon Aaron C. Goldstrohm

Contents

1	Mechanisms of Post-transcriptional Gene Regulation René Arvola, Elizabeth Abshire, Jennifer Bohn, and Aaron C. Goldstrohm	. 1
2	Post-transcriptional Regulation of Insulin and Insulin Like Growth Factors Eun Kyung Lee and Wook Kim	37
3	Post-transcriptional Regulation of Cytokine Signaling During Inflammatory Responses Irina Vlasova-St. Louis and Paul R. Bohjanen	55
4	Post-transcriptional Regulation of Luteinizing Hormone Receptor mRNA Expression in the Ovary K.M.J. Menon, Bindu Menon, and Thippeswamy Gulappa	71
5	Post-transcriptional Regulation of Steroid Hormone Receptors	91
6	MicroRNA Regulation of Endocrine Functions in the Ovary Pavla Brachova, Wei-Ting Hung, Lynda K. McGinnis, and Lane K. Christenson	109
7	Translational Control in the Germ Line Marco Conti, João Pedro Sousa Martins, Seung Jin Han, and Federica Franciosi	129
8	Post-transcriptional Regulation of VEGF-A	157
9	Post-transcriptional Regulation of Prostaglandin Biosynthesis Fernando F. Blanco, Noémie Legrand, Cyril Sobolewski, and Dan A. Dixon	181
10	Translational and Post-translational Control of Leptin Production by Fat Cells Konstantin V. Kandror	221

11	Post-transcriptional Regulation of Parathyroid Hormone Gene Expression in Health and Disease Tally Naveh-Many	235
12	Post-transcriptional and Post-translational Regulation of Steroidogenesis Wen-Jun Shen, Zhigang Hu, Jie Hu, Fredric B. Kraemer, and Salman Azhar	253
13	Post-transcriptional Regulation of Glucocorticoid Function Faoud T. Ishmael and Cristiana Stellato	277
14	Post-transcriptional Regulation by Brain-Derived Neurotrophic Factor in the Nervous System Alexandra M. Amen, Daniel L. Pham, and Mollie K. Meffert	315
Cor Sign	rection to: Post-transcriptional Regulation of Cytokine naling During Inflammatory Responses	C 1
Ind	Index	

Contributors

Elizabeth Abshire Department of Biological Chemistry, University of Michigan Medical School, Ann Arbor, MI, USA

Alexandra M. Amen Solomon H. Snyder Department of Neuroscience, The Johns Hopkins University School of Medicine, Baltimore, MD, USA

René Arvola Department of Biological Chemistry, University of Michigan Medical School, Ann Arbor, MI, USA

Salman Azhar Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, Palo Alto, CA, USA

Division of Gastroenterology and Hepatology, Stanford University, Stanford, CA, USA

Fernando F. Blanco Department of Cancer Biology, University of Kansas Medical Center, Kansas City, KS, USA

Department of Pharmacology and Experimental Therapeutics, Thomas Jefferson University, Philadelphia, PA, USA

Paul R. Bohjanen, MD, PhD Department of Medicine, Center for Infectious Diseases and Microbiology Translational Research, University of Minnesota, McGuire Translational Research Facility, Minneapolis, MN, USA

Department of Microbiology, University of Minnesota, Minneapolis, MN, USA

Jennifer Bohn Department of Biological Chemistry, University of Michigan Medical School, Ann Arbor, MI, USA

Pavla Brachova Department Molecular and Integrative Physiology, University of Kansas Medical Center, Kansas City, KS, USA

Lane K. Christenson Department Molecular and Integrative Physiology, University of Kansas Medical Center, Kansas City, KS, USA

Marco Conti Center for Reproductive Sciences, University of California, San Francisco, CA, USA

Eli and Edythe Broad Center of Regeneration Medicine and Stem Cell Research, University of California, San Francisco, CA, USA

Department of Obstetrics and Gynecology and Reproductive Sciences, University of California, San Francisco, CA, USA

Dan A. Dixon Department of Cancer Biology, University of Kansas Medical Center, Kansas City, KS, USA

Federica Franciosi Center for Reproductive Sciences, University of California, San Francisco, CA, USA

Eli and Edythe Broad Center of Regeneration Medicine and Stem Cell Research, University of California, San Francisco, CA, USA

Department of Obstetrics and Gynecology and Reproductive Sciences, University of California, San Francisco, CA, USA

Aaron C. Goldstrohm Department of Biological Chemistry, University of Michigan Medical School, Ann Arbor, MI, USA

Thippeswamy Gulappa Department of Obstetrics and Gynecology, The University of Michigan Medical School, Ann Arbor, MI, USA

Department of Biological Chemistry, The University of Michigan Medical School, Ann Arbor, MI, USA

Seung Jin Han Department of Biological Sciences, Inje University, Gimhae, Republic of Korea

Zhigang Hu Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, Palo Alto, CA, USA

Division of Endocrinology, Stanford University, Stanford, CA, USA

Jiangsu Key Laboratory for Molecular and Medical Biotechnology and College of Life Science, Nanjing Normal University, Nanjing, China

Jie Hu Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, Palo Alto, CA, USA

Wei-Ting Hung Department Molecular and Integrative Physiology, University of Kansas Medical Center, Kansas City, KS, USA

Nancy H. Ing Department of Animal Science, Texas A&M University, College Station, TX, USA

Faoud T. Ishmael Department of Medicine, Division of Pulmonary, Allergy, and Critical Care Medicine, Section of Allergy and Immunology, The Pennsylvania State University Milton S. Hershey Medical Center, Hershey, PA, USA Konstantin V. Kandror Boston University School of Medicine, Boston, MA, USA

Wook Kim, PhD Department of Molecular Science and Technology, Ajou University, Suwon, South Korea

Fredric B. Kraemer Division of Endocrinology, Stanford University, Stanford, CA, USA

Eun Kyung Lee, PhD Department of Biochemistry, College of Medicine, The Catholic University of Korea, Seoul, South Korea

Noémie Legrand Department of Microbiology and Molecular Medicine, Geneva Medical Faculty, Geneva, Switzerland

João Pedro Sousa Martins Center for Reproductive Sciences, University of California, San Francisco, CA, USA

Eli and Edythe Broad Center of Regeneration Medicine and Stem Cell Research, University of California, San Francisco, CA, USA

Department of Obstetrics and Gynecology and Reproductive Sciences, University of California, San Francisco, CA, USA

Lynda K. McGinnis Department Molecular and Integrative Physiology, University of Kansas Medical Center, Kansas City, KS, USA

Mollie K. Meffert Solomon H. Snyder Department of Neuroscience, The Johns Hopkins University School of Medicine, Baltimore, MD, USA

Department of Biological Chemistry, The Johns Hopkins University School of Medicine, Baltimore, MD, USA

K.M.J. Menon Department of Biological Chemistry, The University of Michigan Medical School, Ann Arbor, MI, USA

Department of Obstetrics and Gynecology, The University of Michigan Medical School, Ann Arbor, MI, USA

Bindu Menon Department of Obstetrics and Gynecology, The University of Michigan Medical School, Ann Arbor, MI, USA

Department of Biological Chemistry, The University of Michigan Medical School, Ann Arbor, MI, USA

Tally Naveh-Many, PhD Minerva Center for Calcium and Bone Metabolism, Department of Nephrology, Hadassah Hospital, Hadassah Hebrew University Medical Center, Jerusalem, Israel

Daniel L. Pham Solomon H. Snyder Department of Neuroscience, The Johns Hopkins University School of Medicine, Baltimore, MD, USA

Hervé Prats Inserm UMR1037, Centre de Recherches en Cancérologie de Toulouse, ERL5294 CNRS, Toulouse Cedex 1, France

Université Toulouse III Paul-Sabatier, Toulouse, France

Wen-Jun Shen Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, Palo Alto, CA, USA

Division of Endocrinology, Stanford University, Stanford, CA, USA

Cyril Sobolewski Department of Cancer Biology, University of Kansas Medical Center, Kansas City, KS, USA

Department of Cellular Physiology and Metabolism, Geneva Medical Faculty, Geneva, Switzerland

Cristiana Stellato, MD, PhD Department of Medicine and Surgery, University of Salerno, Baronissi, SA, Italy

Department of Medicine, Division of Allergy and Clinical Immunology, Johns Hopkins University, Baltimore, MD, USA

Christian Touriol Inserm UMR1037, Centre de Recherches en Cancérologie de Toulouse, ERL5294 CNRS, Toulouse Cedex 1, France

Université Toulouse III Paul-Sabatier, Toulouse, France

Irina Vlasova-St. Louis, MD, PhD Department of Medicine, Center for Infectious Diseases and Microbiology Translational Research, University of Minnesota, Minneapolis, MN, USA

Mechanisms of Post-transcriptional Gene Regulation

René Arvola, Elizabeth Abshire, Jennifer Bohn, and Aaron C. Goldstrohm

1 Introduction

Gene expression involves multiple sequential steps which are highly regulated and coordinated. Regulation of gene expression is crucial for proper function of the endocrine system. Over the past several decades, an abundance of data has accumulated which demonstrates the importance of post-transcriptional mechanisms in control of the endocrine function, which is the subject of this book. These mechanisms can control the amount, timing, and location of protein expression. Moreover, post-transcriptional mechanisms of RNA processing and editing can change the properties of expressed proteins. The purpose of this chapter is to provide an overview of post-transcriptional regulation, focusing on translation and mRNA degradation pathways. We also explore our current knowledge of the mechanisms that regulate protein synthesis and mRNA stability. We refer readers to recent comprehensive reviews on additional post-transcriptional mechanisms including RNA processing (Elkon et al. 2013; Fu and Ares 2014) and modifications (Wang and He 2014), nucleo-cytoplasmic transport (Wickramasinghe and Laskey 2015) and localization (Buxbaum et al. 2015).

Opportunities for post-transcriptional control are numerous. Messenger RNAs (mRNAs) contain extensive cis-acting sequences that can control RNA processing, translation efficiency, and mRNA stability. Furthermore, eukaryotic genomes encode a large repertoire of trans-acting regulators, including RNA binding proteins

1

Note: Due to the space limitations and the broad nature of this chapter, we were unable to cite many important contributions, thus we apologize to colleagues whose work could not be highlighted.

R. Arvola • E. Abshire • J. Bohn • A.C. Goldstrohm (🖂)

Department of Biological Chemistry, University of Michigan Medical School,

¹¹⁵⁰ West Medical Center Drive, Ann Arbor, MI 48109, USA e-mail: acgold@med.umich.edu

[©] Springer International Publishing Switzerland 2016

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_1

and non-coding RNAs, with potential regulatory functions. Current estimates indicate that the human genome encodes some 1500 RNA binding proteins (Gerstberger et al. 2014) and thousands of non-coding RNAs, such as the small regulatory microRNAs (miRNAs) (Bartel 2004; Geisler and Coller 2013). Yet the function of the vast majority of this regulatory potential remains to be explored, especially in physiological systems. The chapters of this volume highlight elegant cases wherein intricate post-transcriptional regulatory mechanisms control responses to and responses from endocrine pathways.

2 Translation Regulation

Translation efficiency plays an important role in determining the level of protein expression. Globally, protein levels correlate poorly with mRNA levels, and translation efficiencies of mRNAs vary widely (Schwanhausser et al. 2011). Translation of mRNAs can be regulated on a global scale, for instance in response to stress or infection (Liu and Qian 2014). Moreover, translation of specific mRNAs or groups of mRNAs can be regulated to control specific pathways, environmental responses, and developmental programs (Curtis et al. 1995; Kong and Lasko 2012; Micklem 1995). In this section, we will provide an overview of the process of translation followed by an exploration of the major paradigms of translational control. For more detailed insight, we refer readers to several excellent reviews on this subject (Aitken and Lorsch 2012; Hinnebusch 2014; Jackson et al. 2010; Kong and Lasko 2012; Sonenberg and Hinnebusch 2009). To provide context for discussion of translational control and its role in endocrine function, we will review the mechanism of translation, and then present principles and examples of regulatory mechanisms that control translation.

2.1 Mechanism of Translation

The process of translation can be conceptualized by four steps: (1) mRNA activation, (2) initiation, (3) elongation, and (4) termination, each having unique mechanisms of control. (1) An mRNA is activated for translation when it is associated with the trans-acting factors necessary to begin recruitment of ribosomes. (2) Translation initiation involves assembly of the ribosome on the activated mRNA, which is facilitated by translation initiation factors (Aitken and Lorsch 2012). During initiation, the ribosome scans the mRNA to identify the translation start site (i.e. the initiation codon), which is recognized by base-pairs formed between the mRNA start codon and the initiator tRNA (tRNAi) anticodon (Hinnebusch 2014). As we will see in the following sections, initiation involves a series of carefully orchestrated events involving multiple translation factors, and therefore is a ratedetermining step of protein synthesis that is subject to multiple mechanisms of regulation (Jackson et al. 2010; Sonenberg and Hinnebusch 2009). (3) Once the start site is identified, the ribosome catalyzes elongation of the nascent polypeptide chain, (4) which is terminated upon encountering a stop codon (Aitken and Lorsch 2012).

2.1.1 The Ribosome and Translation Factors

The process of translation is catalyzed by the ribosome, a large multisubunit ribonucleoprotein complex. Translation is facilitated by an assortment of eukaryotic initiation factors (eIFs), elongation factors (eEFs) and termination factors (Aitken and Lorsch 2012; Inge-Vechtomov et al. 2003; Riis et al. 1990).

The eukaryotic ribosome is comprised of two subunits: a large 60S subunit and a small 40S subunit comprised of 80 ribosomal proteins and 4 ribosomal RNAs (60S: 5S, 5.8S, and 25S rRNA; 40S: 18S rRNA). During the translation initiation step, the two subunits must join together on the mRNA to form the 80S ribosome, which is capable of catalyzing peptide bond formation. The mRNA is held at the interface of the two subunits, positioned to permit reading of the codons by incoming transfer RNA molecules (tRNAs) (Aitken and Lorsch 2012). For detailed information on the fascinating structure and function of the ribosome components, readers are directed to recent reviews (Korobeinikova et al. 2012; Wilson and Doudna Cate 2012; Yusupova and Yusupov 2014).

Transfer RNAs that are charged with amino acids are essential ingredients for translation. Each amino acid is covalently appended to the appropriate cognate tRNA by an amino-acyl tRNA synthetase. These charged tRNAs are then delivered to the ribosome as RNA-protein complexes with special translation factors. The tRNA involved in translation initiation, tRNAi, is charged with methionine (Met-tRNAi). Met-tRNAi forms a ternary complex with the GTP-bound form of translation initiation factor eIF2, and together they associate with the 40S ribosomal subunit to function during initiation of protein synthesis. During initiation, the Met-tRNAi is positioned in the Peptidyl-site (P-site) of the ribosome. The other charged tRNAs associate with the GTP-bound factor eEF1A, which delivers them to the Amino-acyl site (A-site) of the ribosome as specified by the mRNA's codons (Dever and Green 2012; Ibba and Soll 2000).

2.1.2 Key mRNA Features Necessary for Translation

Messenger RNAs possess features that influence their translational efficiency, including two important non-templated modifications. The first is the 7-methyl-guanosine nucleotide cap at the 5' end of the transcript, which is added during transcription in the nucleus. In the cytoplasm, this cap facilitates translation by interacting with the translation initiation factor, eIF4F, which is composed of subunits eIF4E, eIF4G, and eIF4A. All three subunits bind to the RNA, with eIF4E directly contacting the 5' cap (Aitken and Lorsch 2012). The 3' ends of all mRNAs (with the exception of replication-dependent histone mRNAs) are also enzymatically modified by addition of a polyadenosine tail. Addition of this poly(A) tail by poly(A) polymerase enzyme is coupled to 3' end processing of the nascent transcript, which occurs through a sequence-specific endonucleolytic cleavage event (Elkon et al. 2013). The poly(A) tail is bound by poly(A) binding proteins (Kuhn and Wahle 2004). In the cytoplasm, the PABPC1 protein coats the poly(A) tail and enhances the efficiency of translation (Kuhn and Wahle 2004).

The protein coding capacity of the mRNA is specified by several features that determine where translation will begin and end, defining the open reading frame (ORF).

The translation initiation site is typically the first AUG codon, from the 5' end of the mRNA, with the proper surrounding sequence context (Jackson et al. 2010). This context, originally characterized by Marilyn Kozak, surrounds the initiation site AUG (Kozak 1987). In vertebrates, the general "Kozak" consensus sequence is gccRccAUGG, where the underlined AUG initiation codon is flanked by uppercase nucleotides, denoting strong influence on initiation, and lowercase nucleotides denoting lesser importance for initiation. Note that the "R" indicates a purine nucleotide base. Using transcriptome-wide datasets, AUG codon contexts have now been defined for multiple species (Nakagawa et al. 2008). The 3' end of the protein coding region is specified by an in-frame stop codon, either UAA, UGA, or UAG (Dever and Green 2012).

Messenger RNAs can contain other features that affect translation. Only a portion of the mRNA sequence encodes protein, while the remaining sequences are 5' and 3' Untranslated Regions (UTR). These UTRs can range from tens of nucleotides to thousands of nucleotides and—as we shall see throughout this chapter and book—play important regulatory roles to control and alter translation (Mignone et al. 2002).

2.1.3 Activated mRNAs and the Closed-Loop Conformation

Before engaging ribosomes, mRNAs must be activated. The 5' and 3' ends of the mRNA are brought together through interactions between cap-bound eIF4F and poly(A)-bound PABPC1. This "looping" is mediated by the eIF4G component of eIF4F, which bridges the cap binding protein eIF4E and PABPC1 via protein-protein interactions. Circular RNA-protein complexes have been observed by atomic force and electron microscopy (Afonina et al. 2014; Wells et al. 1998). Moreover, evidence in several systems demonstrates synergistic stimulation of translation mediated by the 5' cap and poly(A) tail (Borman et al. 2000; Gallie 1991; Michel et al. 2000). Thus, mRNAs complexed with eIF4F and PABPC1 can be considered to be in an activated state that is potentiated for subsequent loading of ribosomes. It is interesting to note that replication-dependent histone mRNAs maintain a cap-totail closed loop, though they do not possess a poly(A) tail; instead, the histone mRNA closed loop is formed by specialized RNA binding proteins that recognize a unique 3' end RNA structure, thereby promoting histone protein synthesis (Marzluff et al. 2008). Thus, closed loop formation is thought to be a generalized feature of activated mRNAs (Fig. 1.1).

2.1.4 Initiation: Assembly of the Pre-initiation Complex

Translation initiation requires at least 12 different initiation factors that act to bring together the ribosome subunits on the mRNA and, as the most complex stage of translation, this process is highly regulated. First, the small ribosomal subunit must locate the initiation codon, facilitated by base-pairing with the anticodon of the tRNAi. Initiation begins with the formation of a pre-initiation complex (Fig. 1.1a). First, the 40S small ribosomal subunit associates with initiation factors eIF1, eIF1A, eIF5, eIF3 and the eIF2 ternary complex (composed of GTP-bound eIF2 and mettRNAi) to form the 43S pre-initiation complex (PIC). The 43S PIC is then joins an activated mRNA to form the 48S PIC, mediated by many protein and RNA contacts



Fig. 1.1 Translation initiation and common mechanisms of regulation. The steps of translation initiation are outlined in (**a**). The ternary complex is comprised of eIFs 1, 1A, 3 and 5 joined to the 40S small ribosomal subunit. eIF2 bound to GTP and the initiator tRNA joins the ternary complex to form the 43S pre-initiation complex (43S PIC). The 43S PIC is then bound to an activated mRNA to form the 48S PIC. The 48S PIC performs scanning until the Kozak sequence is found. The 48S PIC is then joined by the 60S large ribosomal subunit, signaling the end of initiation and the start of elongation. (**b**) Demonstrates how translational efficiency is regulated negatively through phosphorylation of eIF2. eIF2 is bound to GTP, which is hydrolyzed during initiation. The eIF2 α subunit of eIF2 can be phosphorylated at Serine 51 (S51) to negatively regulate its role in translation. Unphosphorylated eIF2 can initiate translation normally, whereas phosphorylated eIF2 α prevents exchange of GDP for GTP, inhibiting initiation. (**c**) shows how members of the eIF4F complex are regulated by mTOR. Phosphorylation by various kinases of the mTOR signaling cascade enhances translational efficiency through inhibition of 4E-BP, as well as enhancing association of eIF4B with eIF4F

including those formed between eIF4F, PABP and mRNA with the multisubunit initiation factor, eIF3. Once these translation factors have assembled on the mRNA, the next step is to locate the proper initiation codon (Aitken and Lorsch 2012; Jackson et al. 2010).

2.1.5 Initiation: Scanning for the Initiation Codon

Once the 48S PIC is assembled, the ribosome must locate the translation start site (typically the first AUG codon) to initiate protein synthesis. In order for the 48S PIC to search for the AUG start codon, it must traverse the 5' UTR in a process known

as ribosome scanning. 5'UTRs frequently contain RNA structures which can impede scanning and thus inhibit translation. In the event that the first AUG codon has a poor context, downstream AUG codons can be utilized to initiate translation, a process referred to as leaky scanning (Hinnebusch 2014).

Scanning through RNA structure by the 48S PIC is promoted by the eIF4A protein, which is an ATPase/helicase that can unwind secondary structure in the 5'UTR. Other helicases may also facilitate scanning. This process requires energy in the form of hydrolysis of adenosine triphosphate (ATP). eIF4B binds single stranded RNA (ssRNA) and also helps in unwinding. eIF4G is involved by facilitating the association of eIF4A (Hinnebusch 2014).

The factors eIF1, eIF1A, and eIF3 aid in scanning by stabilizing the open conformation of the mRNA entry channel of the small ribosome subunit, and also in start codon recognition. The 48S PIC slides along the mRNA, sampling the RNA until the first AUG codon in the proper sequence context is located. Once the AUG start codon successfully base pairs with its anticodon complement on the Met-tRNAi, eIF1 is released and the PIC adopts a more closed conformation. At this point, the Met-tRNAi is positioned in the P-site of the ribosome. The GTP bound to eIF2 is then hydrolyzed and eIF5 and GDP-bound eIF2 are released from the PIC. eIF1A is the only initiation factor from the PIC which remains bound throughout the entire process of initiation. The PIC is now more stably bound to the mRNA and tRNAi and poised for joining of the 60S subunit (Hinnebusch 2014).

2.1.6 Initiation: Formation of the 80S Ribosome

The next phase of initiation is assembly of the 80S ribosome through joining of the 60S subunit to the initiation codon associated 48S PIC. The 60S large ribosomal subunit first assembles with the GTPase protein eIF5B. Upon large and small ribosome subunit joining, eIF5B hydrolyzes its GTP. eIF5B and eIF1A are then released as the ribosome undergoes a conformational change. The resulting 80S ribosome is thereby primed to enter the elongation phase (Aitken and Lorsch 2012; Jackson et al. 2010).

2.1.7 Elongation

Once the 80S ribosome has assembled at the initiation site, protein synthesis can commence through ribosome-catalyzed peptide bond formation between the MettRNAi located in the P-site and the incoming amino-acylated tRNA in the A-site. The nascent polypeptide chain is extended through sequential rounds of peptide bond formation and translocation of the ribosome along the mRNA. Subsequent amino acid additions are specified through complementary base-pairing between tRNA anti-codons and the triplet codons of the mRNA. Elongation in eukaryotes is mediated by two elongation factor proteins: eEF1 and eEF2 (Dever and Green 2012). eEF1 is a multisubunit complex that delivers the amino-acylated tRNA to the ribosome (Sasikumar et al. 2012). Upon proper positioning of the tRNA in the A-site, the eEF1A subunit hydrolyzes GTP and eEF1 dissociates from the ribosome. The eEF2 factor facilitates the translocation of the ribosome and hydrolysis of GTP (Dever and Green 2012).

Typically, multiple ribosomes are sequentially assembled on and traverse the mRNA simultaneously. As the first ribosome elongates away from the initiation site,

new ribosomes can initiate and follow. The resulting mRNA with multiple associated ribosomes is referred to as a poly-ribosome or polysome (Slayter et al. 1963; Warner et al. 1963). The density of ribosomes on an mRNA is proportional to the length of the open reading frame and the rates of initiation and elongation (Ingolia 2014).

2.1.8 Termination

Translating ribosomes traverse the mRNA until they encounter a stop codon (UAA, UAG, and UGA) within the A-site, which signals the termination of polypeptide chain elongation. Since there is no tRNA anticodon complementary to the stop codon, no amino acid can be added to the end of the peptide chain. Instead, a release factor (eRF) binds the stop codon and triggers the release of the complete polypeptide from the ribosome. Eukaryotes have two release factors: eRF1, which is involved in stop codon recognition and hydrolysis of the nascent protein from the P-site bound tRNA, and eRF3, a GTPase which promotes polypeptide release. Upon termination of translation, the ribosome is disassembled into its large and small subunits, assisted by additional translation factors (Dever and Green 2012; Jackson et al. 2012).

2.2 Regulation of Translation

Translation can be regulated in multiple steps to control the amount, timing and location of protein synthesis. Cis-acting sequence elements and trans-acting factors can either activate or repress translation. Translation can be regulated globally, affecting protein synthesis from all mRNAs (as is the case with the mTOR pathway), or specifically from certain mRNAs (as with sequence-specific RNA-binding proteins and microRNAs). Groups of mRNAs can be translationally regulated in a coordinated fashion by common cis-elements and trans-factors (Abaza and Gebauer 2008; Jackson et al. 2010). One classic example of this type of post-transcriptional regulation is the 5' terminal oligo-pyrimidine (TOP) mRNAs that encode multiple components of the translation apparatus and are coordinately regulated in response to stress (Meyuhas and Kahan 2015). Here, we will discuss control of translation and explore some of its general mechanisms (Fig. 1.2).



Fig. 1.2 mRNAs contain cis-acting regulatory information that controls translation efficiency and stability. Eukaryotic mRNAs have a 5' 7-methylguanosine (7mG) cap that promotes translation and stability. The poly(A)denosine (poly(A)) tail is, recognized by poly(A) binding proteins PABP, promotes translation and stability. Removal of the cap and poly(A) tail result in subsequent mRNA degradation. Cis regulatory elements can be contained in the 5' and 3' untranslated regions (UTRs) of an mRNA. These can include binding sides of RNA binding proteins and non-coding RNAs that modulate stability and translation. The open reading frame (ORF) can also contribute to regulation through the Kozak sequence and codon optimality (how commonly the codons it contains are found in the cell)

2.2.1 Regulation of Translation Initiation Through the Closed Loop

The closed loop conformation of mRNAs promotes translation initiation through the cap-to-poly(A) tail interactions mediated by eIF4F and PABP. Formation of the closed loop represents an important regulatory stage. Trans-acting factors can disrupt the closed loop to inhibit translation by displacing eIF4F or PABP from the mRNA or by disrupting their protein–protein interactions (Kawahara et al. 2008; Weidmann et al. 2014; Zekri et al. 2013). Enzymatic removal of the 5' cap structure or poly(A) tail can disrupt closed loop formation, thereby silencing translation and leaving the mRNA vulnerable to degradation, as described below. The length of the poly(A) tail can modulate translation like a rheostat, with longer poly(A) tails promoting protein expression whereas shortening of the tail reduces it. As such, factors that stabilize or shorten the tail can control translation efficiency (Goldstrohm and Wickens 2008).

Translation efficiency can be controlled by proteins that interact with PABP. PABP-interacting proteins (PAIP1 and PAIP2) bind to PABP to either stimulate or inhibit translation, respectively (Khaleghpour et al. 2001; Roy et al. 2002). PAIP1 shares homology with eIF4G and forms a complex with initiation factors eIF4A and eIF3 to enhance translation by bridging PABP's interaction with eIF4F and stabilizing the closed loop (Craig et al. 1998; Martineau et al. 2008). PAIP2 competes with eIF4G and PAIP1 to bind PABP, reducing its affinity for the poly(A) tail and reducing translation efficiency (Khaleghpour et al. 2001).

The 5' cap and associated eIF4F complex are major targets for translational control mechanisms. Proteins that compete with eIF4F for binding to the 5' cap can inhibit translation (Cho et al. 2005, 2006). A second, widely utilized control mechanism is mediated by proteins that directly bind to cap-binding eIF4F subunit, eIF4E. These eIF4E Binding Proteins (4E-BPs) competitively bind to the same region of eIF4E as the eIF4G subunit. In doing so, 4E-BPs disrupt eIF4F and the closed loop, thereby inhibiting formation of the PIC. In addition to repressing translation, several 4E-BPs have been shown to promote degradation of mRNAs (Andrei et al. 2005; Blewett and Goldstrohm 2012a, b; Igreja and Izaurralde 2011; Rendl et al. 2012).

Multiple signaling pathways intersect on 4E-BPs, providing a nexus for controlling translation. Unphosphorylated 4E-BPs have a high affinity for eIF4E whereas phosphorylation of 4E-BPs prevents their interaction with eIF4E (Fig. 1.1c). The Target of Rapamycin (TOR) pathway is a major regulator of translation that responds to the availability of nutrients and amino acids. TOR is an important regulator of cell growth and proliferation and is inhibited in response to stress conditions and starvation. TOR also integrates signals from hormones such as Insulin and Brain-Derived Neurotrophic Factor. In turn, TOR pathway regulates translation of peptide hormones such as Leptin. TOR promotes translation in several ways, including phosphorylation of 4E-BPs and activation of S6 kinase, which phosphorylates the small ribosomal subunit 6 and eIF4B, among other targets, to promote translation (Dennis et al. 2012; Ma and Blenis 2009; Tavares et al. 2015). This cascade of TOR signaling controls translation initiation on a broad level.

2.2.2 Regulation of Initiation Through Initiation Factor eIF2

Translation initiation depends on delivery of Met-tRNAi to the 40S subunit by GTPbound eIF2 and subsequent PIC formation. Thus, eIF2 represents an important regulatory target. eIF2 is inhibited by phosphorylation at Serine 51 (S51) on the α subunit by various kinases in response to diverse signals (Baird and Wek 2012) (Fig. 1.1b). Kinases that phosphorylate eIF2 include: (1) PKR-like endoplasmic reticulum kinase (PERK), which is activated by the unfolded protein response; (2) General Control Nonderepressible 2 (GCN2), which is activated by diverse stressors, such glucose and amino acid starvation; (3) Protein Kinase R (PKR), which is activated by dsRNAs greater than 30 bp in length and plays an important role in anti-viral response; and (4) Heme-regulated inhibitor kinase (HRI) in erythroid cells, which is activated in response to heme deficiency (Baird and Wek 2012; Lemaire et al. 2008). Phosphorylation at S51 prevents exchange of GDP for GTP, thus the phosphorylated eIF2 cannot enter new rounds of translation. As a result, translation initiation is inhibited globally (Baird and Wek 2012).

2.2.3 RNA Binding Proteins Regulate Translation

The untranslated regions (UTRs) of many mRNAs can contain important regulatory information that controls translation (Mignone et al. 2002). RNA binding proteins (RBPs) often recognize these regions to regulate the translational efficiency and stability of target mRNAs. RBPs serve many important biological roles where gene expression needs to be quantitatively, temporally and/or spatially controlled, such as in response to hormone mediated signaling. RBPs can bind to specific RNA structures, (e.g. stem-loop structures), or they can bind to specific single-stranded sequence motifs. Upon binding to a transcript, RBPs can use diverse mechanisms to modulate translation by either repressing or activating protein synthesis (Abaza and Gebauer 2008). Here, we will explore some specific mechanisms of RBP translational repressors and activators.

RBP repressors can inhibit initiation by binding to the 5' UTR of a target mRNA and blocking assembly of the PIC. A classic example of this mechanism is the Iron Response Protein (IRP), which, in response to low intracellular iron, binds to specific RNA stem-loop structure in the 5'UTR of ferritin mRNA, the Iron Response Element, to impede 43S joining and thus represses translation of ferritin (Muckenthaler et al. 1998).

RBP repressors can also bind to the 3'UTR of transcripts to control translation. One mechanism is to prevent assembly of the 80S ribosome. For instance, the RBPs hnRNP-K and hnRNP-E1 repress lipoxygenase mRNA by preventing 60S subunit joining to the 48S PIC (Ostareck et al. 2001). Other 3'UTR-bound RBPs can recruit 4E-BPs to a specific message to disrupt the closed loop and repress translation; examples of such interactions are numerous and include Bruno and Cup (Nakamura et al. 2004), Smaug and Cup (Nelson et al. 2004), and Puf5 and Eap1 (Blewett and Goldstrohm 2012a). This mechanism is illustrated by the RBP called Cytoplasmic Polyadenylation Element Binding Protein (CPEB), which binds to U-rich sequences known as Cytoplasmic Polyadenylation Elements (CPEs) in the 3'UTR of certain

mRNAs. One mechanism of CPEB repression is recruitment of the 4E-BP Maskin to inhibit translation of specific mRNAs, such as Cyclin B, during oogenesis (Groisman et al. 2000; Stebbins-Boaz et al. 1999).

Translation initiation can be inhibited by RBP-mediated recruitment of an eIF4E Homologous Protein (4EHP) that competes with eIF4E for binding to the mRNAs 5' cap (Rom et al. 1998). However, unlike eIF4E, 4EHP does not interact with eIF4G and thus prevents translation initiation. The 3'UTR binding protein Bicoid recruits 4EHP to repress translation of specific mRNAs during Drosophila embry-onic development (Cho et al. 2005).

RBPs can repress translation of specific mRNAs by causing shortening of the mRNAs poly(A) tail-a process referred to as deadenylation-thereby reducing or eliminating the occupancy of PABP to diminish translation initiation. One of the first examples of deadenylation mediated silencing was the finding that cytokine and growth factor mRNAs contained Adenine-Uridine Rich Elements (AREs) in their 3'UTRs which accelerated deadenylation and mRNA degradation, limiting protein expression (Wilson and Treisman 1988). These AREs can be bound by several RBPs, including the repressive Tristetraprolin (TTP) protein which binds and recruits a multisubunit complex of poly(A) degrading enzymes that shorten the poly(A) tail of TTP bound mRNAs (Lykke-Andersen and Wagner 2005). Likewise, members of the Pumilio and Fem3 Binding (PUF) family of sequence-specific RBPs bind to 3'UTRs and recruit specialized poly(A) degrading enzymes that remove the poly(A) tail to repress protein expression (Goldstrohm et al. 2006; Van Etten et al. 2012). CPEB, as mentioned earlier, also promotes deadenylation of the mRNAs to which it binds by recruiting the poly(A) specific ribonuclease (PARN), contributing to translational repression (Kim and Richter 2006).

3'UTR-bound RBPs can also repress translation by promoting removal of the message's 5' cap structure. The TTP protein interacts with and recruits decapping enzymes to specific transcripts that contain ARE sequences in their 3'UTRs (Fenger-Gron et al. 2005; Lykke-Andersen and Wagner 2005). One PUF protein can promote decapping of mRNAs by using a 4E-BP to disrupt eIF4F and to recruit decapping factors to the message, resulting in translational repression and mRNA degradation (Blewett and Goldstrohm 2012a, b). Through these mechanisms, RBP mediated translational repression and mRNA degradation are directly interrelated, a subject that we shall revisit in subsequent discussion of mRNA decay pathways in post-transcriptional control.

2.2.4 RBP Activators

Translation can also be activated by cis- and trans-acting factors, which can boost the amount of protein produced by an mRNA. They can also reanimate mRNAs that have been stored in a quiescent status, a common event in developmental contexts (Gray and Wickens 1998; Ivshina et al. 2014). Just as the poly(A) tail is a target for repressive mechanisms, it can also be employed to activate mRNAs. Polyadenylation (that is, lengthening of the poly(A) tail) and the resulting increased recruitment of PABP can promote translational activation. Thus, dormant, deadenylated mRNAs can be activated by polyadenylation in the cytoplasm via recruitment of poly(A) polymerase enzymes, such as GLD2 (Ivshina et al. 2014). Perhaps the best characterized example of polyadenylation mediated activation is the sequence-specific RBP CPEB (Charlesworth et al. 2013). As described earlier, CPEB represses mRNAs via deadenylation and Maskin-mediated inhibition of eIF4F. CPEB acts as a bifunctional regulator during oogenesis, switching from repression to activation in response to signal the steroid hormone progesterone (Groisman et al. 2002; Ivshina et al. 2014; Sarkissian et al. 2004). Aurora A kinase phosphorylates CPEB, thereby switching it to an activation mechanism wherein CPEB interacts with and recruits GLD2 poly(A) polymerase. CPEB-Gld2 mediated polyadenylation requires that the mRNA contain both a CPE and the polyadenylation element (AAUAAA) at the 3' end of the mRNA. The polyadenylation element is recognized by a cytoplasmic version of the Cleavage and Polyadenylation Specificity Factor (CPSF). The CPEB-GLD2 complex extends the poly(A) tails of the target mRNAs and increases the occupancy of poly(A) binding proteins. Thus, CPEB activation includes derepression and polyadenylation resulting in increased efficiency of translation (Ivshina et al. 2014).

2.2.5 Regulation of Translation Elongation

The process of elongation is iterated with an average rate of 6 amino acid additions per second (Ingolia et al. 2011). Global analyses suggest that protein synthesis rates vary over a wide range (Schwanhausser et al. 2011). Various factors impinge on the elongating ribosome to influence its speed and the location and quality of protein expression. For instance, elongation rate can be influenced by synonymous codon usage and the availability of the necessary amino-acylated tRNAs (Pechmann and Frydman 2013; Presnyak et al. 2015; Quax et al. 2015; Tarrant and von der Haar 2014). For membrane bound and secreted proteins such as hormones, a signal peptide in the nascent polypeptide is recognized by the Signal Recognition Particle to direct the translating mRNA to the proper intracellular location. Chaperones (for example, heat shock proteins) associate with and fold the nascent peptide cotranslationally (Jha and Komar 2011). In several examples, signal transduction pathways have been shown to target elongation factors to influence the rate of protein synthesis (Dever and Green 2012; Sasikumar et al. 2012).

2.2.6 Alternative Mechanisms of Initiation: Cap-Independent Translation Initiation

Translation generally requires the 5' cap; however, in specialized instances, translation can initiate in a cap-independent manner, mediated by internal ribosome entry sites (IRES). IRES are highly structured elements present in the 5'UTR of specific mRNAs which can allow translational initiation on that mRNA without the requirement of the 5' cap and certain initiation factors through complex interactions with the ribosome, circumventing the process of scanning. IRES were first discovered in viral mRNAs, but examples of cellular IRES-containing mRNAs have emerged. These alternative initiation mechanisms permit translation of specific proteins when cap-dependent translation is turned off by the cell in response to viral infection, or other cellular stresses (Hellen and Sarnow 2001).

One well known example of a viral IRES-containing mRNA is that of the Hepatitis C Virus, which contains both 5' and 3' IRES elements (Fraser and Doudna 2007).

An extreme example of a viral 5' IRES is that the Cricket Paralysis Virus, which is able to bypass the requirement for all translation initiation factors (including eIF2-tRNAi) by mimicking the initiator tRNA in the P site of the ribosome (Fernandez et al. 2014; Hellen and Sarnow 2001). A well-studied example of a cellular IRES-containing mRNA which will be discussed in Chap. 8 is Vascular Endothelial Growth Factor A (VEGF-A), a mitogen and important stimulator of angiogenesis (Akiri et al. 1998; Huez et al. 1998; Miller et al. 1998).

3 Regulation of Gene Expression by mRNA Degradation

All mRNAs undergo decay as a part of normal gene expression. Decay of individual mRNAs can be highly regulated to control proper levels of protein expression and to spatially and temporally restrict protein production. Further, intrinsic or extrinsic signals can alter the decay rates of specific transcripts, including endocrine signals that alter gene expression by effecting mRNA decay. In this section, we provide an overview of mRNA decay pathways in mammals. We then discuss mechanisms of post-transcriptional regulation through RNA decay, including important gaps in current knowledge.

RNA decay pathways are initiated from either the 3' or 5' ends of the transcript, or by endonucleolytic cleavage. Multiple pathways can overlap in order to efficiently degrade an mRNA. First, we consider a major pathway of transcript decay that proceeds through the processes of deadenylation, decapping and exonucleolytic decay (Fig. 1.3).

3.1 Deadenylases Remove the Poly(A) Tail

Decay of cytoplasmic mRNAs typically initiates by the progressive shortening of the 3' poly(A) tail, a process referred to as deadenylation (Fig. 1.3). Deadenylation is frequently the rate-limiting step of mRNA decay and is an important node for control of translation and mRNA decay. Multiple enzymes catalyze deadenylation (Goldstrohm and Wickens 2008). These deadenylases are magnesium-dependent exoribonucleases that degrade mRNAs from the 3' to 5' end of the poly(A) tail, releasing 5' AMP as a product. General mRNA deadenylation occurs in two phases. Once the poly(A) tail is shorted to a critical threshold, the mRNA is rapidly destroyed.

The first phase is comprised of an initial shortening of the poly(A) tail by the Pan2-Pan3 deadenylase complex (Boeck et al. 1996; Brown et al. 1996; Yamashita et al. 2005). The Pan2 subunit contains a DEDD type nuclease domain that catalyzes deadenylation. The activity of Pan2-Pan3 is stimulated by Poly(A) Binding Protein (PABP) (Uchida et al. 2004). This interaction occurs through a specific protein interaction motif located on Pan3 (Siddiqui et al. 2007). PABP therefore serves a dual role: during translation, PABP stimulates translation initiation, and it also participates in the first phase of mRNA decay. Pan2-Pan3 also has a WD40



Fig. 1.3 Eukaryotic mRNA decay pathways. Decay of mRNA in eukaryotes begins with deadenylation by Pan2-Pan3 (**a**), which shortens the poly(A) tail and may dissociate PABP. The remainder of the poly(A) tail is degraded by the Ccr4-Not complex (**b**). Both deadenylase complexes release AMP molecules as the product of the deadenylation reaction. After deadenylation, mRNAs go through one of two pathways to decay the remainder of the mRNA. The 5' to 3' decay pathway starts with decapping by Dcp1/Dcp2 to remove the 7-methylguanosine cap (**c**) and is concluded by degradation of the remainder of the message by Xrn1 (**d**). The 3' to 5' decay pathway comprises of degradation of the RNA down to the cap by the Exosome (**e**) and subsequent scavenging of the cap by DcpS (**f**)

protein–protein interaction domain, suggesting that other protein partners may contact and regulate Pan2 activity. In support of this idea, Pan2 is known to associate with at least one RNA binding regulatory complex (Christie et al. 2013; Huntzinger et al. 2013; Kuzuoglu-Ozturk et al. 2012). Additionally, Pan2 has a catalytically dead ubiquitin hydrolase domain, though the function of this domain remains enigmatic. The Pan3 subunit contains a pseudokinase domain and zinc finger domain, both of which have been hypothesized to contribute to RNA binding (Jonas et al. 2014; Schafer et al. 2014; Wolf et al. 2014).

The second phase of deadenylation is catalyzed by the multisubunit CCR4-NOT deadenylase complex (CNOT) (Yamashita et al. 2005). Originally discovered in

yeast, the CNOT complex contains two active deadenylase subunits, orthologs of the Ccr4 and Pop/Caf1 proteins. In mammals, the Ccr4 orthologs include CNOT6 and CNOT6L proteins while the Pop2 orthologs include CNOT7 and CNOT8 proteins. The CNOT complex includes at least seven other components with various functions associated with regulation and coordination of deadenylase activity, in addition to functions beyond deadenylation (Collart 2003; Tucker et al. 2001). In contrast to Pan2-Pan3, which is stimulated by PABP, addition of PABP in in vitro experiments inhibits the activity of CNOT (Tucker et al. 2002). CNOT deadenylates the poly(A) tail down to a short oligo(A) that is incapable of binding PABP to facilitate translation initiation. Thus, PABP and CNOT have opposing stimulatory and inhibitory activities with respect to translation, placing poly(A) at the nexus of control of translation and mRNA decay.

The CNOT complex is regulated by protein interactions with multiple partners (Goldstrohm and Wickens 2008). For example, multiple members of the PUF family of sequence specific RNA binding proteins have been shown to bind the Pop2 subunit and recruit the CNOT complex to specific mRNAs (Goldstrohm et al. 2006, 2007; Weidmann et al. 2014). As a consequence, the poly(A) tails of the mRNAs are removed more quickly, the resulting deadenylated mRNA is translationally repressed, and its stability is decreased. A variety of RBPs have now been shown to recruit the CNOT complex to achieve repression, including the ARE binding protein TTP (Lykke-Andersen and Wagner 2005; Sandler et al. 2011) and the RBP Roquin (Leppek et al. 2013), both of which cause repression and degradation of cytokine mRNAs. RBPs can recruit the complex via interactions with specific subunits. The CNOT1 subunit is important for deadenylation, though it has no catalytic function itself. Instead, CNOT1 serves as a molecular scaffold and a target for RBP mediated recruitment to specific substrate mRNAs. For instance, TTP binds directly to CNOT1. Other CNOT complex subunits play additional functional roles, as recently reviewed by Shirai et al. (2014).

Other deadenylases have also been identified in mammalian systems. The presence of multiple deadenylases could lead to redundancy in their function, yet certain deadenylases have been shown to have specific biological functions such as control of development, anti-viral response, and regulation of metabolism (Goldstrohm and Wickens 2008). Also, some deadenylases have unique activities and features. The PARN deadenylase, expressed in many higher eukaryotes, is unique in that it binds to the 5' cap of mRNAs, resulting in enhancement of PARN's ability to deadenylate the mRNAs 3'end (Gao et al. 2000; Martinez et al. 2000). This property indicates that PARN likely acts on translationally inactive mRNAs. PARN has been implicated in developmental processes in plants and Xenopus laevis and is also expressed in mammals. Of particular interest in the field of endocrinology, the deadenylase Nocturnin is controlled by the circadian clock (Green and Besharse 1996), and genetic analysis indicates that Nocturnin's main function is to regulate fat metabolism (Green et al. 2007). Still, much remains to be elucidated with regard to the functions of deadenylase family members and their roles in post-transcriptional regulation.

3.2 Decapping Enzymes Catalyze Removal of the 5' Cap

In addition to its role in translation initiation, the 5' 7-methylguanosine cap protects the 5' end of the mRNA from exonucleases. These features put the 5' cap at the nexus of regulation of translation and mRNA decay. The 5' cap can be removed by hydrolysis of the 5' to 5' triphosphate linkage that covalently joins the 7-methyl guanosine to the first nucleotide of the mRNA. This process is referred to as decapping and is catalyzed by specialized decapping enzymes. Decapping typically follows deadenylation, coordinated by protein interactions between the deadenylase complex and the decapping machinery. However, deadenylation independent decapping has also been observed for some mRNAs (Badis et al. 2004; Fromont-Racine et al. 1993).

Multiple decapping enzymes have been identified (Li and Kiledjian 2010; Song et al. 2013). One of the best-characterized decapping enzymes is the highly conserved Dcp2 enzyme, which releases ⁷mGDP from the mRNA 5' end through activity of its Nudix domain (Wang and Kiledjian 2002). Dcp2 also possesses a Box A domain that functions in RNA binding.

Decapping enzymes are highly regulated by protein partners (Jonas and Izaurralde 2013). Dcp2 activity is stimulated by its protein partner Dcp1, which forms a stable complex with Dcp2 through the Dcp2 Box A domain (Piccirillo et al. 2003; She et al. 2006). Dcp2 is additionally activated by Dhh1, an RNA helicase, and Pat1, which recruits the Lsm complex. (Ling et al. 2011). The Lsm protein complex (Lsm 1-7) associates with the 3' deadenylated end of mRNA and increases Dcp2 decapping efficiency (Tharun and Parker 2001). An additional decapping factor, Ge-1, is found in this complex in higher eukaryotes, further activating decapping activity (She et al. 2008; Yu et al. 2005).

RNA binding proteins greatly influence decapping rates of specific mRNAs, as explored below and in several recent reviews (Arribas-Layton et al. 2013; Li and Kiledjian 2010). For instance, the ARE-binding protein TTP recruits decapping factors to accelerate decapping of specific transcripts (Gao et al. 2001; Lykke-Andersen and Wagner 2005). Decapping of mRNAs is frequently the fate-determining step that targets mRNAs for destruction; however, recent evidence indicates that some transcripts may be recapped, suggesting that decapping and recapping could serve as a regulatory mechanism to repress and then activate mRNAs (Mukherjee et al. 2012).

3.3 Exoribonucleolytic Decay Can Initiate from the 5' End

Following decapping, the exposed 5' end of the mRNA can be attacked by exoribonucleases, degrading the mRNA in a 5' to 3' direction while releasing nucleotide monophosphate products. Several 5' exoribonucleases have been identified in eukaryotes including XRN1 and XRN2 (Nagarajan et al. 2013). Both enzymes act on multiple types of substrate RNAs. Here we consider several important roles.

The exoribonuclease XRN1 is primarily responsible for 5' mRNA decay in the cytoplasm. XRN1 has specificity for RNAs with a 5' monophosphate, coinciding

with the product of DCP2-mediated decapping. This enzyme is fast and highly processive; decay intermediates are rarely detected (Nagarajan et al. 2013).

XRN2, a 5' to 3' exoribonuclease with homology to XRN1, is primarily located in the nucleus and is conserved across a range of species (Miki and Grosshans 2013). Functions of XRN2 include maturation of rRNA and snoRNAs, as well as transcription termination (Boisvert et al. 2007; Luo et al. 2006; Wang and Pestov 2011). XRN2 is also involved in RNA quality control pathways, decaying aberrant RNAs in the nucleus, as well as unspliced pre-mRNAs to control mRNA levels (Das et al. 2003).

Decapping and 5' decay of mRNAs are coordinated via physical interaction of the XRN1 C-terminus with decapping factors (Fischer and Weis 2002; Sinturel et al. 2012). XRN1 is further regulated by interactions with RNA binding proteins such as the ARE-binding protein TTP.

3.4 Exoribonucleolytic Decay from the 3' End

Messenger RNAs can also be degraded from the 3' end by the exosome, a large multisubunit complex (10–12 subunits) which acts in a 3' to 5' direction. Two variations of the exosome complex have been characterized: one nuclear exosome and one cytoplasmic exosome. The exosome acts to process and/or degrade multiple types of RNA including mRNAs (Liu et al. 2006). The cytoplasmic exosome acts on mRNAs that have been deadenylated, degrading them from the 3' end to produce nucleotide monophosphates. The Dis3 subunit is responsible for this activity (Reis et al. 2013). Once the mRNA is degraded to a capped m⁷GpppN (where N is fewer than 10 nucleotides) product, a specialized scavenger decapping enzyme, DcpS, hydrolyzes the cap structure to produce 7-methylguanosine monophosphate and nucleotide diphosphate (Chen et al. 2005; Liu and Kiledjian 2005). DcpS action prevents the potentially toxic effects of accumulated capped mRNA fragments, which could competitively inhibit eIF4F function. Interestingly, defects in this final mRNA decay step cause a form of intellectual disability and neuromuscular disease (Ng et al. 2015).

As we have learned, mRNAs can be degraded from the 5' end, the 3' end, or both. The 5' and 3' decay pathways appear to compete to degrade certain mRNAs, whereas other mRNAs appear to be mainly degraded by one or the other pathway. The determinants and factors that affect these destructive choices remain incompletely understood.

3.5 Endonucleases Cut mRNAs to Initiate Decay

As an alternative pathway to exoribonucleolytic decay, select messages undergo endonucleolytic cleavage (Fig. 1.4). The resulting 5' and 3' fragments are subsequently degraded by exoribonucleases including XRN1 and the exosome. In fact, the Dis3 subunit of the exosome is also an endonuclease, in addition to its



Fig. 1.4 Endonucleolytic decay pathways. Representative examples of eukaryotic endonucleolytic decay pathways are illustrated here. Components of the exosome, such as Dis3, have endonuclease activity (**a**). Under ER stress conditions, Ire-1 decays ER-associated mRNAs to reduce translation (**b**). Pro-inflammatory mRNAs are degraded by Regnase-1 to negatively regulate cytokine expression (**c**). In nonsense mediated decay, mRNAs with premature stop codons are cleaved by SMG6 before being degraded by exonucleases (**d**). In microRNA-mediated decay pathways and RNAi pathways, precursor miRNAs are processed by endonucleases such as Dicer and Drosha to make mature miRNAs (**e**). Endonucleolytic decay products typically serve as substrates for further decay by exonucleases (**f**)

exoribonuclease activity (Arraiano et al. 2010). Other examples of endonucleases highlight the diversity within this class of enzymes. One of the first characterized mammalian endoribonucleases was the PMR1 enzyme, which was found to be an estrogen induced factor that initiates mRNA decay (Pastori et al. 1991a, b). PMR1 associates with polysomes to efficiently degrade specific mRNAs. Recent evidence indicates that PMR1 has an important role in reducing parathyroid hormone mRNA levels, discussed further in the chapter by Naveh-Many (Nechama et al. 2009). The Regnase-1 endonuclease regulates immune function by degrading pro-inflammatory cytokine mRNAs (Matsushita et al. 2009) (Fig. 1.4). Regnase-1 is rapidly degraded upon stimulation of immune responses, and in mouse models, Regnase-1 is implicated in autoimmune disorders (Liang et al. 2010).

Endonucleolytic mRNA decay is also an important for stress response. The IRE1 protein is a transmembrane endoribonuclease located on the endoplasmic reticulum. During stress conditions, IRE1 decays mRNAs through the regulated IRE1 dependent decay (RIDD) pathway (Hollien et al. 2009; Maurel et al. 2014) (Fig. 1.4). IRE1 substrates are specifically associated with the ER and decay faster during stress conditions to reduce translation (Gaddam et al. 2013). Additionally, these substrate mRNAs are highly enriched for transcripts involved in secretory pathways, including the hormone insulin and multiple cell surface receptors (Han et al. 2009a).

Nonsense mediated decay (NMD) of mRNAs is a vital pathway for clearing mRNA transcripts with premature termination codons. These aberrant transcripts are targeted by NMD machinery for degradation, as reviewed by Popp and Maquat (2013). One NMD component, SMG6, is an endonuclease that cleaves the mRNA, resulting in subsequent exoribonucleolytic decay of the fragments (Franks et al. 2010; Schmid and Jensen 2008) (Fig. 1.4). Endonucleases function in a variety of other RNA decay and processing pathways in the cell, including the Argonaute, Dicer, and Drosha endonucleases that participate in RNA interference pathways, addressed in a later section of this chapter.

3.6 Regulation of mRNA Decay

Cis-acting RNA sequences, either linear motifs or secondary structures, can control decay of the transcript. These elements are recognized by sequence specific RNA binding regulatory factors. In turn, the regulators recruit the mRNA decay machinery to facilitate removal of the poly(A) tail, the 5' cap, or to promote exonucleolytic or endonucleolytic decay. There are many hundreds of RNA binding proteins encoded in mammalian cells (Gerstberger et al. 2014). Moreover, mRNA decay can be controlled by small non-coding RNAs that form complexes with regulatory proteins (Jonas and Izaurralde 2015), as described in subsequent sections. Here, we a provide a few examples or regulators that control mRNA stability and refer readers to recent in-depth reviews (Garneau et al. 2007; Goldstrohm and Wickens 2008; Li and Kiledjian 2010; Schoenberg and Maquat 2012).

Two well-characterized instability elements are the Adenine and Uridine Rich Elements (AREs) and Guanine and Uridine Rich Elements (GREs), which are frequently found in 3' UTR of mRNAs such as cytokine mRNAs (Bakheet et al. 2006; Vlasova-St Louis and Bohjanen 2011). Multiple RNA binding proteins recognize these sequences to control mRNA stability. Certain ARE binding proteins can recruit mRNA decay machinery to increase decay. For instance, TTP recruits the CNOT deadenylase complex, the DCP2 decapping complex, and exonucleases to mRNAs to promote their destruction (Chen et al. 2001; Fenger-Gron et al. 2005; Lykke-Andersen and Wagner 2005; Sandler et al. 2011).

AREs can act as a bifunctional switch, causing mRNA decay in one state while stabilizing mRNAs in other conditions. This is achieved by at least two mechanisms. First, stabilizing ARE binding proteins, such as HuR, can compete with destabilizing factors such as TTP. Second, post-translational modifications can alter the activity of the ARE binding proteins (Garneau et al. 2007). Phosphorylation of TTP can alter its RNA binding, protein interactions, and protein stability (Brooks and Blackshear 2013). ARE mediated regulation will be further discussed in Chaps. 3, 5, 8, 9, 11, and 13.

GRE elements are bound by CUG-Binding protein, also referred to as CELF1 (Vlasova-St Louis and Bohjanen 2011) (see Chap. 3 of this volume). CELF1 plays multiple roles in mRNA processing, translation and stability. This evolutionarily conserved repressor represses protein expression by causing deadenylation and mRNA decay. The mechanism is incompletely understood, but evidence indicates that CELF1 interacts with the deadenylase PARN to promote deadenylation (Moraes et al. 2006).

3.7 Nonsense-Mediated, Non-stop, and No-Go mRNA Decay Pathways

Decay of mRNA is a vital point of normal gene regulation, but also plays an important role in quality control of gene expression (Ghosh and Jacobson 2010). Aberrant, defective mRNAs are degraded to ensure fidelity. These mRNAs can arise due to mutation, misprocessing, or breakdown in the complex processes necessary to decode them (i.e. translation). Nonsense mediated decay, discussed above, is responsible for decaying transcripts with premature termination codons, protecting cells from the potentially deleterious effects of producing truncated proteins with abnormal function (Popp and Maquat 2014). NMD destroys these aberrant transcripts via deadenylation dependent decay and endonucleolytic decay (Schoenberg and Maquat 2012). Messenger RNAs that lack a stop codon, as the result of mutation or misprocessing, are targeted by Non-stop decay pathway, wherein the exosome destroys the transcript. An additional quality control pathway, so-called No-Go Decay, clears faulty mRNAs stalled ribosomes stuck in the act of translation. This decay pathway releases the ribosome and then degrades the abnormal transcript via endonucleolytic cleavage. For more information on quality control mechanisms, we refer readers to recent reviews (Harigaya et al. 2010; Popp and Maquat 2013).

4 Post-transcriptional Regulation by Non-coding RNAs

The protein-coding sequence in the human genome only accounts for 1 % of the entire genome meaning that the majority of the genome is noncoding DNA (Mattick 2004). Much of this noncoding DNA is actually transcribed into noncoding RNAs (Bertone et al. 2004; Cheng et al. 2005; Kampa et al. 2004). This finding suggests that higher eukaryotes have evolved new and complex regulatory mechanisms, both structural and functional, that involve not just proteins but also noncoding RNAs. Many of these noncoding transcripts are synthesized similar to mRNAs including being capped at the 5' end, often spliced, and polyadenylated at the 3' end. Further

processing may take place once these RNAs are in processing complexes to generate functional noncoding RNAs. The regulatory potential of RNAs comes from the ability for it to interact with other nucleic acids and proteins allowing for intricate formation of regulatory RNA-protein (RNP) complexes. Such regulatory RNAs include small noncoding RNAs such as microRNAs (miRNAs) and short interfering RNAs (siRNAs) (Carthew and Sontheimer 2009). Long noncoding RNAs and circular RNAs are additional forms of noncoding RNA that have more recently been shown to contribute to gene regulation (Geisler and Coller 2013). These RNAs can serve many critical functional roles in the cell including control of transcription, post-transcriptional regulation of mRNA decay and translational control.

4.1 Small Non-coding RNAs

4.1.1 MicroRNAs

MicroRNAs are 22–25 nt small noncoding RNAs that participate in RNA-base pairing interactions with specific mRNAs in order to repress expression of target mRNAs (Bartel 2004). They do so by inhibiting translation and activating mRNA degradation (Jonas and Izaurralde 2015). Post-transcriptional gene regulation by microRNAs is widespread (Bartel 2004, 2009). Currently, over 1500 microRNAs have been identified in the human genome, each of which can regulate expression of multiple genes (Freedman and Tanriverdi 2013; Lewis et al. 2005). Some of these are conserved across mammals and even to lower eukaryotes, while others are endogenous to the human genome only. Many are expressed in a tissue specific manner, (Londin et al. 2015) while others are expressed in stage-specific manners during development (Krichevsky et al. 2003; Pasquinelli et al. 2000). Because of this, microRNAs are critical regulators of many tissue specific functions. For example: one pancreatic specific microRNA, miR-375, regulates several mRNA targets that are critical for the secretion of insulin from the islets of Langerhans (Poy et al. 2004). If miR-375 is lost, insulin secretion is upregulated and when miR-375 is abundant, insulin secretion is downregulated. Another microRNA, miR-143, was identified through a microarray analysis to play an important regulatory role in adipocyte differentiation (Esau et al. 2004). These are just a few examples of the numerous biological roles of microRNAs. This book will emphasize the regulatory roles of microRNAs in endocrine function, as discussed in Chaps. 2, 5, 6, 9, 12, 13 and 14.

Biogenesis of MicroRNAs

MicroRNAs are processed from precursor transcripts through several steps before they become functional. First, microRNAs are transcribed by RNA Polymerase II; these transcripts are usually capped and poly(A)denylated. Typically, these primary transcripts can encode for one or more microRNAs and can even code for clusters of microRNAs (Lagos-Quintana et al. 2001; Lau et al. 2001). Alternatively, one transcript can encode for both a microRNA and a protein simultaneously. In these cases, the microRNA sequence is many times encoded in the intronic sequence, so-called mirtrons (Berezikov et al. 2007; Ha and Kim 2014; Okamura et al. 2007).

The initial primary microRNA (pri-miRNA) has extra sequence extending past the microRNA itself (22–24 core nucleotides) both on the 5'- and 3'-ends of the transcript. Next, the pri-miRNA is folded into a stem-loop structure and is excised from the primary transcript while still in the cell's nucleus. A complex including an RNase III endonuclease, Drosha, carries out this cleavage step. Drosha is accompanied in this complex by DGCR8, a protein that serves to recognize the primiRNA. The Drosha-DGCR8 complex cleaves the pri-miRNA such that a 60–70 nucleotide stem-loop intermediate with a mature 5'-phosphate and a 3'-nucleotide overhang remains. This product of Drosha cleavage is referred to as a pre-miRNA and can be exported to the cytoplasm for further processing and maturation (Lee et al. 2003; Zeng and Cullen 2003).

In the cytoplasm, the pre-miRNA becomes a substrate for Dicer, an RNase III endonuclease which catalyzes a second cleavage event. Dicer specifically recognizes the 3'-nucleotide overhang that was previously generated during Drosha processing. This endonuclease event is responsible for removal of the terminal loop and consequently the release of a small double stranded RNA duplex. This Dicer cleavage event additionally dictates the length of the duplex RNA for the mature miRNA of 22–25 nucleotides. Once the mature miRNA is formed, it can enter into a functional regulatory complex referred to as the RNA-induced silencing complex (RISC) (Ha and Kim 2014; MacRae et al. 2007).

Regulation of MicroRNA Biogenesis

MicroRNA biogenesis can be regulated at each processing step, thereby altering the regulatory response controlled by the resulting microRNA (Ha and Kim 2014). Drosha processing activity can be modulated by DGCR8 autoregulation in the first step of microRNA processing (Han et al. 2009b). The DGCR8 mRNA contains conserved stem loop structures that closely resemble the structure formed by the primiRNA and therefore the Drosha/DGCR8 complex is able to cleave the DGCR8 mRNA, controlling the expression of DGCR8 at the post-transcriptional level. Furthermore, the DGCR8 component of the complex forms protein–protein interactions with Drosha in order to further stabilize Drosha creating a feedback mechanism to control the levels of this microprocessor complex (Han et al. 2009a, b). Importantly, this mechanism is highly conserved from humans to zebrafish to drosophila.

Another example of the regulation of microRNA biogenesis occurs at the stage of Dicer-catalyzed microRNA processing (Ha and Kim 2014). The best example is the let-7 microRNA and one of its canonical targets, Lin28. Lin28 is typically translationally repressed by the let-7 microRNA; however, when Lin28 protein is present, it binds to the terminal loop of the let-7 pre-miRNA leading. This interaction impedes the binding and cleavage by Dicer and thus the pre-let-7 miRNA maturation is inhibited (Carthew and Sontheimer 2009; Ha and Kim 2014). Instead, Lin28 recruits Terminal Uridyl Transferases (TUTs) to cause oligouridylation, or the addition of uridylate residues, to the 3' end of the pre-let-7 miRNA and consequently degradation of the microRNA intermediate (Heo et al. 2012; Thornton et al. 2012). These types of mechanisms allow for tight control of microRNA production and processing and by controlling the levels of mature microRNAs, the cell is able to modulate regulation by microRNAs.

4.1.2 Short Interfering RNAs

Short interfering RNAs (siRNAs) are 21–23 nucleotide RNAs processed from exogenous double-stranded RNAs (dsRNAs). SiRNAs are fully complementary to their target mRNA sequences and work primarily by Watson-Crick base-pairing (Carthew and Sontheimer 2009). Targeting of an siRNA to an mRNA results in an endonucleolytic cleavage and subsequent exonucleolytic decay of the mRNA fragments. In research, siRNAs are often administered to cell culture for programmed knockdown of a particular gene via RNA interference (RNAi) or post-transcriptional gene silencing (PTGS). In cells, Dicer can also process double-stranded RNAs into functional siRNAs.

Another class of siRNAs, endo-siRNAs, are produced from endogenous cellular dsRNA precursors (Ghildiyal et al. 2008; Yang and Kazazian 2006). Endo-siRNAs differ from microRNAs in the ways in which they are generated and processed. These RNAs are synthesized and processed in a variety of different ways but typically require Dicer for cleavage of a double stranded RNA (dsRNA) precursor and/ or RNA-dependent RNA polymerases in order to be synthesized and processed. RNA-dependent RNA polymerases can use mature mRNAs as templates for synthesis, and the products are cleaved by Dicer. The resulting endo-siRNAs can act as silencers of both transcription and translation (Ghildiyal and Zamore 2009). EndosiRNAs produced from viral RNAs can protect the cell from viral infections (Li et al. 2013; Wang et al. 2006). Historically, there is strong evidence for this type of mechanism in plants and flies; however, more recently, these anti-viral RNAi response pathways have been observed in multiple mammalian cell lines, suggesting evolutionary conservation (Claycomb 2014; Li et al. 2013).

4.2 Assembly of the RNA Induced Silencing Complex (RISC)

To control gene expression, microRNAs and siRNAs must be loaded into protein complexes termed RNA induced silencing complexes (RISC) (Jonas and Izaurralde 2015) (Fig. 1.5). A core component of RISC is the Argonaute protein family (Meister 2013). When assembled, RISC contains one of several Argonaute (Ago) proteins that bind directly to the small RNAs. Eukaryotes often possess more than one Argonaute protein that share similar architecture including a domain with homology to endonucleases (Meister 2013). There are four human Argonautes (AGO1-4), each of which can bind to small RNAs; however, only AGO2, is enzymatically capable of endonucleolytic cleavage of bound mRNA targets.

Argonaute associates with one of the two strands of the non-coding RNA, referred to as the guide strand. First, the double stranded RNA duplex, produced by Dicer, is loaded onto Argonaute. Next, selection of the guide strand is determined by thermodynamics of base pairing at the 5'-ends of each of the two strands (Khvorova et al. 2003). The strand that is less tightly base paired will allow for easier loading and further strand separation (Khvorova et al. 2003). Removal of the RNA strand that is selected against is an intricate process that may involve an RNA helicase activity to aid in the unwinding of the small RNA duplex (Kawamata et al. 2009; Yoda et al.


Fig. 1.5 MicroRNA mediated post-transcriptional regulation of translation and mRNA decay. (a) Assembly of the microRNA-induced silencing complex (miRISC) begins with the microRNA partially base-pairing with the 3'UTR of the mRNA. Argonaute proteins associate with the miRNA-mRNA duplex and recruit GW182, a glycine and tryptophan rich protein that mediates protein–protein interactions that underlie translational repression and mRNA decay. (b) Translational inhibition is mediated by displacement of PABP from the poly(A) tail, and disruption of the eIF4F complex bound to the cap. miRNA RISC thereby disrupts the translationally active closed loop structure of the mRNA, resulting in mRNA destruction. As a consequence of PABP displacement, the poly(A) tail of the mRNA is left unprotected and susceptible to deadenylases such as PAN2/3 and the CCR4-NOT complex, both of which are recruited by the GW182 component of RISC. The miRISC also recruits the DCP1/2 decapping complex which triggers $5' \rightarrow 3'$ decay of the mRNA as shown in Fig. 1.3

2010). The unloaded strand, or the passenger strand, dissociates and is enzymatically destroyed (Peters and Meister 2007). This preferential loading of the guide strand is the first step to RISC assembly and occurs similarly for RISC complexes containing either microRNAs or siRNAs. The Argonaute protein maintains tight interactions with the 3'-end of the guide strand allowing the 5'-end of the guide strand, including the seed site, to engage in Watson-crick base pairing for target scanning and recognition (Carthew and Sontheimer 2009; Ha and Kim 2014)

4.3 Targeting Mechanism of microRNAs

Both microRNAs and siRNAs function by base-pairing with target mRNAs but their degree of complementarity to target mRNAs influences the outcome of regulation. Complete complementarity of the small RNA to its target leads to endonucleolytic cleavage of the mRNA, catalyzed by the Argonaute protein. This mechanism predominates for silencing by siRNAs. Likewise, if a microRNA-mRNA interaction is fully base paired, mRNA cleavage can occur. However, most miRNAs are not fully complementary to their targets. Instead, miRNA recognition occurs by more limited Watson-Crick base pairing, with strong preference for pairing of the second through eighth nucleotide of the microRNA, referred to as the seed site. Functional microRNA targeting tends to exhibit perfect seed site complementarity and weaker, imperfect pairing for the remainder of the RNA. Functional miRNAs sites tend to be located in the 3'UTR of messages, though not exclusively. Rather than using endonuclease activity to cleave mRNA targets, microRNA RISC causes translational inhibition or canonical mRNA decay processes, as described in the following section.

4.4 Mechanism of microRNA Mediated Repression

MicroRNA-bound Argonaute associates with multiple protein partners to repress protein expression from target mRNAs (Fig. 1.5). One protein partner, GW182, is crucial for microRNA mediated repression. Argonaute proteins all have a conserved site on the surface of the protein that can tightly interact with conserved tryptophan residues that are characteristic of GW182 proteins (Jonas and Izaurralde 2015; Pfaff et al. 2013). GW182 functions as a critical adapter that mediates protein interactions necessary for downstream effects of microRNA regulation (Carthew and Sontheimer 2009; Jonas and Izaurralde 2015; Pfaff et al. 2013). This microRNA RISC complex regulates mRNA stability and translation by multiple mechanisms. RISC displaces PABP from the mRNA to disrupt closed loop conformation of translation (Zekri et al. 2013) (Fig. 1.5b). RISC also recruits both PAN2 and CCR4-NOT deadenylase complexes to accelerate removal of the mRNAs poly(A) tail. Additionally, RISC recruits DCP2 decapping complex to facilitate removal of the 5' cap, thereby accelerating 5' to 3' mRNA degradation (Fig. 1.5c). Other mechanisms of microRNA mediated repression have been described including inhibition of translation initiation and elongation (Fukao et al. 2014; Fukaya et al. 2014; Zekri et al. 2013); this area of research remains highly active (Jonas and Izaurralde 2015). This combination of translational inhibition and mRNA degradation facilitate the effectiveness of microRNA repression. Moreover, repression also occurs in combination with other miRNAs and RBPs to ensure proper regulation of gene expression in a wide array of contexts, including response to endocrine signals.

4.5 Long Noncoding RNAs

Noncoding RNAs have long been known to play a variety of functions including catalysis of translation and splicing, scaffolding of protein complexes, chromatin/ chromosome modifiers, and mediators of intracellular localization and sequence specific targeting (Cech and Steitz 2014; Geisler and Coller 2013). More recently, a large number of new long noncoding RNAs (lncRNAs) have been identified and the association of specific lncRNAs with disease states has primed interest in their molecular functions.

Long noncoding RNAs are generally classified as transcripts that are longer than 200 nucleotides that do not contain a high confidence open reading frame above a specific threshold length—typically 100 amino acids. The defining features of long noncoding RNAs are rapidly developing. The vast majority of newly reported long noncoding RNAs do not have a known function, but intense research efforts have revealed that some lncRNAs are key regulatory players in chromatin structure, transcriptional control or post-transcriptional control. Here, we focus on several examples of lncRNAs that act post-transcriptionally to control protein expression.

Many of these long noncoding RNAs look like typical mRNAs; they are transcribed by RNA Pol II and possess a 5' cap and 3' poly(A) tail. Long noncoding RNAs can originate from near classified genes (lncRNAs) or intergenic regions (lincRNAs). To date, thousands of these RNAs have been reported, facilitated by new detection methods such as next-generation sequencing. The abundance of lncRNAs varies widely, with some being among the most abundant in cells, whereas others are low abundance (Cabili et al. 2015). Furthermore, many lncRNAs have specific expression patterns related to developmental stage, tissue and cell type, hinting that these molecules have important yet undiscovered roles (Batista and Chang 2013). Evolutionary conservation of some lncRNAs also indicates that they have important biological functions (Ulitsky and Bartel 2013). Yet other lncRNAs appear to be recent additions to the genome. Given the propensity of RNA to adopt secondary and tertiary structures that are important for function, conservation of primary nucleotide sequence may be compensated by conserved structures of some parts of the RNA molecule (Diederichs 2014). Pseudogenes, gene duplicates that have lost the ability to produce functional protein, sometimes produce lncRNAs. These pseudogenes retain the ability to be transcribed, whether it is through its own promoter or from transcription read through at the neighboring gene locus.

4.5.1 LncRNAs Play a Role in mRNA Processing

Nascent transcripts are processed extensively by capping, splicing, editing and 3' end cleavage and polyadenylation. Each of these steps can be regulated to control and diversify mRNAs. A prime example is splicing, which can be regulated to produce multiple mRNAs, and therefore multiple protein isoforms, from the same gene. Long noncoding RNAs are emerging as important regulators of alternative splicing events (Tripathi et al. 2010). For example, one class of lncRNAs, known as natural antisense transcripts (NATs), are antisense transcripts of protein-coding genes that hybridize to the mRNA and prevent alternative splicing. The highly abundant and stable lncRNA MALAT1 localizes to nuclear regions associated with

mRNA splicing factors, nuclear speckles, and evidence indicates that MALAT1 can modulate alternative splicing. Yet the molecular mechanism of how MALAT1 affects splicing remains unknown.

4.5.2 LncRNAs Affect Translation and mRNA Stability

LncRNAs can regulate the fate of mRNAs in the cytoplasm by affecting their translation efficiency and mRNA stability. Multiple mechanisms are emerging. For instance, lncRNAs can alter protein expression by competing with microRNAs for cis-acting sites on the mRNA, thereby reducing miRNA mediated repression (Ebert and Sharp 2010; Tay et al. 2014). Alternatively, lncRNAs can act as competing endogenous RNAs (ceRNAs) to sequester miRNAs away from their target mRNAs. This so called sponging or decoy action alleviates repression caused by specific microRNAs, thereby stabilizing the target mRNAs and increasing protein expression. An example of this competing endogenous RNA function is the PTENP1 pseudogene lncRNA (Poliseno and Pandolfi 2015; Poliseno et al. 2010). PTENP1 affects its protein-coding counterpart gene, PTEN, a well-documented tumor suppressor. PTEN is extensively regulated at the post-transcriptional level by microRNAs (He 2010). PTENP1 competitively binds and sequesters the microRNAs that repress PTEN, thus allowing for increased expression of PTEN at the protein level (Poliseno and Pandolfi 2015; Poliseno et al. 2010). Other lncRNAs may function in a similar manner to competitively disrupt regulatory switches, whether microRNAs or RNA binding proteins, that control expression of protein coding genes.

LncRNAs can anneal to specific mRNAs and mediate their degradation. Long noncoding RNAs with Alu repeats have been reported to form dsRNA duplexes with Alu-like elements in the 3'UTRs of targets, creating a high affinity binding site for the dsRNA binding protein, Staufen (Gong and Maquat 2011; Park and Maquat 2013). Following duplex formation and Staufen (STAU1) binding, the transcript is then degraded via Staufen mediated decay (SMD) (Park and Maquat 2013).

Translational control can also be affected by lncRNAs that interact with specific regions of mRNAs. Control of translation of the UCHL1 mRNA provides the primary example of this mechanism, whereby an antisense lncRNA promotes translation of UCHL1 protein in response to stress (Carrieri et al. 2012). This effect is driven by sequence complementarity between the lncRNA and a short interspersed repeat, SINEB2 elements in the transcript, which coincides with the site of translation initiation. Currently the mechanism of translational enhancement remains unknown.

4.5.3 Long Noncoding RNAs in the Endocrine System

LncRNAs are emerging as important regulators of endocrine function. Here, we discuss one example: lncRNA regulation of the glucocorticoid receptor (GR) protein (Knoll et al. 2015). GR controls gene expression during stress responses, causing growth arrest, dampening immune response, among other effects. The GR protein has a well-documented function as a transcriptional regulator. Glucocorticoid hormone binds to GR, triggering its binding to specific DNA sequence elements known as glucocorticoid response elements (GRE). More recently, GR was shown to also bind RNA and to regulate mRNA turnover in response to stress (Cho et al.

2015; Ishmael et al. 2011; Kino et al. 2010). Interestingly, GR activity is controlled by the lncRNA known as growth arrest specific 5 (Gas5). Gas5 inhibits the activity of GR by competitively binding to the protein, thereby preventing interaction with GRE containing DNA. Thus, Gas5 effectively dampens the cellular response to glucocorticoids and rather than arresting cell growth, these cells are shunted into pro-apoptotic pathways (Hudson et al. 2014; Kino et al. 2010). For more information on the regulation of glucocorticoid function, see Chap. 13 of this volume.

5 Conclusion

This chapter has provided the reader with fundamental knowledge of translation and mRNA degradation mechanisms. Importantly, key paradigms of post-transcriptional regulation were emphasized, supported by well documented examples. The cis and trans-acting factors that control translation and mRNA decay rates of specific mRNAs were discussed. As a result, the reader is now prepared to explore the roles of post-transcriptional regulation in the endocrine system, as illuminated by experts in the subsequent chapters.

Acknowledgments We thank Brian Alzua, who provided proofreading and comments on this chapter. This research was supported by the National Institute of General Medical Sciences of the National Institutes of Health under Award Number R01GM105707. R.A. was supported by the Michigan Predoctoral Training Program in Genetics through NIH National Research Service Award 5T32GM007544. J.B. was supported by the Michigan Predoctoral Training Program in Cellular Biotechnology through NIH National Research Service Award 5T32GM008353. E.A. was supported by the Michigan Predoctoral Training Program in Cellular Biotechnology through NIH National Research Service Award 5T32GM008353. E.A. was supported by the Michigan Predoctoral Training Program in Chemistry and Biology Interface through NIH National Research Service Award 2T32GM008597. A.C.G. was supported by a Research Scholar Grant, RSG-13-080-01-RMC, from the American Cancer Society.

References

- Abaza I, Gebauer F (2008) Trading translation with RNA-binding proteins. RNA 14:404-409
- Afonina ZA, Myasnikov AG, Shirokov VA, Klaholz BP, Spirin AS (2014) Formation of circular polyribosomes on eukaryotic mRNA without cap-structure and poly(A)-tail: a cryo electron tomography study. Nucleic Acids Res 42:9461–9469
- Aitken CE, Lorsch JR (2012) A mechanistic overview of translation initiation in eukaryotes. Nat Struct Mol Biol 19:568–576
- Akiri G, Nahari D, Finkelstein Y, Le SY, Elroy-Stein O, Levi BZ (1998) Regulation of vascular endothelial growth factor (VEGF) expression is mediated by internal initiation of translation and alternative initiation of transcription. Oncogene 17:227–236
- Andrei MA, Ingelfinger D, Heintzmann R, Achsel T, Rivera-Pomar R, Luhrmann R (2005) A role for eIF4E and eIF4E-transporter in targeting mRNPs to mammalian processing bodies. RNA 11:717–727
- Arraiano CM, Andrade JM, Domingues S, Guinote IB, Malecki M, Matos RG, Moreira RN, Pobre V, Reis FP, Saramago M et al (2010) The critical role of RNA processing and degradation in the control of gene expression. FEMS Microbiol Rev 34:883–923
- Arribas-Layton M, Wu D, Lykke-Andersen J, Song H (2013) Structural and functional control of the eukaryotic mRNA decapping machinery. Biochim Biophys Acta 1829:580–589

- Badis G, Saveanu C, Fromont-Racine M, Jacquier A (2004) Targeted mRNA degradation by deadenylation-independent decapping. Mol Cell 15:5–15
- Baird TD, Wek RC (2012) Eukaryotic initiation factor 2 phosphorylation and translational control in metabolism. Adv Nutr 3:307–321
- Bakheet T, Williams BR, Khabar KS (2006) ARED 3.0: the large and diverse AU-rich transcriptome. Nucleic Acids Res 34:D111–D114

Bartel DP (2004) MicroRNAs: genomics, biogenesis, mechanism, and function. Cell 116:281-297

Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136:215-233

- Batista PJ, Chang HY (2013) Long noncoding RNAs: cellular address codes in development and disease. Cell 152:1298–1307
- Berezikov E, Chung WJ, Willis J, Cuppen E, Lai EC (2007) Mammalian mirtron genes. Mol Cell 28:328–336
- Bertone P, Stolc V, Royce TE, Rozowsky JS, Urban AE, Zhu X, Rinn JL, Tongprasit W, Samanta M, Weissman S et al (2004) Global identification of human transcribed sequences with genome tiling arrays. Science 306:2242–2246
- Blewett NH, Goldstrohm AC (2012a) An eIF4E-binding protein promotes mRNA decapping and is required for PUF repression. Mol Cell Biol
- Blewett NH, Goldstrohm AC (2012b) A eukaryotic translation initiation factor 4E-binding protein promotes mRNA decapping and is required for PUF repression. Mol Cell Biol 32:4181–4194
- Boeck R, Tarun S Jr, Rieger M, Deardorff JA, Muller-Auer S, Sachs AB (1996) The yeast Pan2 protein is required for poly(A)-binding protein-stimulated poly(A)-nuclease activity. J Biol Chem 271:432–438
- Boisvert FM, van Koningsbruggen S, Navascues J, Lamond AI (2007) The multifunctional nucleolus. Nat Rev Mol Cell Biol 8:574–585
- Borman AM, Michel YM, Kean KM (2000) Biochemical characterisation of cap-poly(A) synergy in rabbit reticulocyte lysates: the eIF4G-PABP interaction increases the functional affinity of eIF4E for the capped mRNA 5'-end. Nucleic Acids Res 28:4068–4075
- Brooks SA, Blackshear PJ (2013) Tristetraprolin (TTP): interactions with mRNA and proteins, and current thoughts on mechanisms of action. Biochim Biophys Acta 1829:666–679
- Brown CE, Tarun SZ Jr, Boeck R, Sachs AB (1996) PAN3 encodes a subunit of the Pab1pdependent poly(A) nuclease in *Saccharomyces cerevisiae*. Mol Cell Biol 16:5744–5753
- Buxbaum AR, Haimovich G, Singer RH (2015) In the right place at the right time: visualizing and understanding mRNA localization. Nat Rev Mol Cell Biol 16:95–109
- Cabili MN, Dunagin MC, McClanahan PD, Biaesch A, Padovan-Merhar O, Regev A, Rinn JL, Raj A (2015) Localization and abundance analysis of human lncRNAs at single-cell and single-molecule resolution. Genome Biol 16:20
- Carrieri C, Cimatti L, Biagioli M, Beugnet A, Zucchelli S, Fedele S, Pesce E, Ferrer I, Collavin L, Santoro C et al (2012) Long non-coding antisense RNA controls Uchl1 translation through an embedded SINEB2 repeat. Nature 491:454–457
- Carthew RW, Sontheimer EJ (2009) Origins and mechanisms of miRNAs and siRNAs. Cell 136:642–655
- Cech TR, Steitz JA (2014) The noncoding RNA revolution-trashing old rules to forge new ones. Cell 157:77–94
- Charlesworth A, Meijer HA, de Moor CH (2013) Specificity factors in cytoplasmic polyadenylation. Wiley Interdiscip Rev RNA 4:437–461
- Chen CY, Gherzi R, Ong SE, Chan EL, Raijmakers R, Pruijn GJ, Stoecklin G, Moroni C, Mann M, Karin M (2001) AU binding proteins recruit the exosome to degrade ARE-containing mRNAs. Cell 107:451–464
- Chen N, Walsh MA, Liu Y, Parker R, Song H (2005) Crystal structures of human DcpS in ligand-free and m7GDP-bound forms suggest a dynamic mechanism for scavenger mRNA decapping. J Mol Biol 347:707–718
- Cheng J, Kapranov P, Drenkow J, Dike S, Brubaker S, Patel S, Long J, Stern D, Tammana H, Helt G et al (2005) Transcriptional maps of 10 human chromosomes at 5-nucleotide resolution. Science 308:1149–1154

- Cho PF, Poulin F, Cho-Park YA, Cho-Park IB, Chicoine JD, Lasko P, Sonenberg N (2005) A new paradigm for translational control: inhibition via 5'-3' mRNA tethering by bicoid and the eIF4E cognate 4EHP. Cell 121:411–423
- Cho PF, Gamberi C, Cho-Park YA, Cho-Park IB, Lasko P, Sonenberg N (2006) Cap-dependent translational inhibition establishes two opposing morphogen gradients in *Drosophila* embryos. Curr Biol 16:2035–2041
- Cho H, Park OH, Park J, Ryu I, Kim J, Ko J, Kim YK (2015) Glucocorticoid receptor interacts with PNRC2 in a ligand-dependent manner to recruit UPF1 for rapid mRNA degradation. Proc Natl Acad Sci U S A 112:E1540–E1549
- Christie M, Boland A, Huntzinger E, Weichenrieder O, Izaurralde E (2013) Structure of the PAN3 pseudokinase reveals the basis for interactions with the PAN2 deadenylase and the GW182 proteins. Mol Cell 51:360–373
- Claycomb JM (2014) Ancient endo-siRNA pathways reveal new tricks. Curr Biol 24:R703-R715
- Collart MA (2003) Global control of gene expression in yeast by the Ccr4-not complex. Gene $313{:}1{-}16$
- Craig AW, Haghighat A, Yu AT, Sonenberg N (1998) Interaction of polyadenylate-binding protein with the eIF4G homologue PAIP enhances translation. Nature 392:520–523
- Curtis D, Lehmann R, Zamore PD (1995) Translational regulation in development. Cell 81:171–178
- Das B, Butler JS, Sherman F (2003) Degradation of normal mRNA in the nucleus of *Saccharomyces* cerevisiae. Mol Cell Biol 23:5502–5515
- Dennis MD, Jefferson LS, Kimball SR (2012) Role of p70S6K1-mediated phosphorylation of eIF4B and PDCD4 proteins in the regulation of protein synthesis. J Biol Chem 287:42890–42899
- Dever TE, Green R (2012) The elongation, termination, and recycling phases of translation in eukaryotes. Cold Spring Harb Perspect Biol 4:a013706
- Diederichs S (2014) The four dimensions of noncoding RNA conservation. Trends Genet 30:121-123
- Ebert MS, Sharp PA (2010) Emerging roles for natural microRNA sponges. Curr Biol 20:R858–R861
- Elkon R, Ugalde AP, Agami R (2013) Alternative cleavage and polyadenylation: extent, regulation and function. Nat Rev Genet 14:496–506
- Esau C, Kang X, Peralta E, Hanson E, Marcusson EG, Ravichandran LV, Sun Y, Koo S, Perera RJ, Jain R et al (2004) MicroRNA-143 regulates adipocyte differentiation. J Biol Chem 279:52361–52365
- Fenger-Gron M, Fillman C, Norrild B, Lykke-Andersen J (2005) Multiple processing body factors and the ARE binding protein TTP activate mRNA decapping. Mol Cell 20:905–915
- Fernandez IS, Bai XC, Murshudov G, Scheres SH, Ramakrishnan V (2014) Initiation of translation by cricket paralysis virus IRES requires its translocation in the ribosome. Cell 157:823–831
- Fischer N, Weis K (2002) The DEAD box protein Dhh1 stimulates the decapping enzyme Dcp1. EMBO J 21:2788–2797
- Franks TM, Singh G, Lykke-Andersen J (2010) Upf1 ATPase-dependent mRNP disassembly is required for completion of nonsense- mediated mRNA decay. Cell 143:938–950
- Fraser CS, Doudna JA (2007) Structural and mechanistic insights into hepatitis C viral translation initiation. Nat Rev Microbiol 5:29–38
- Freedman JE, Tanriverdi K (2013) Defining miRNA targets: balancing simplicity with complexity. Circulation 127:2075–2077
- Fromont-Racine M, Bertrand E, Pictet R, Grange T (1993) A highly sensitive method for mapping the 5' termini of mRNAs. Nucleic Acids Res 21:1683–1684
- Fu XD, Ares M Jr (2014) Context-dependent control of alternative splicing by RNA-binding proteins. Nat Rev Genet 15:689–701
- Fukao A, Mishima Y, Takizawa N, Oka S, Imataka H, Pelletier J, Sonenberg N, Thoma C, Fujiwara T (2014) MicroRNAs trigger dissociation of eIF4AI and eIF4AII from target mRNAs in humans. Mol Cell 56:79–89

- Fukaya T, Iwakawa HO, Tomari Y (2014) MicroRNAs block assembly of eIF4F translation initiation complex in Drosophila. Mol Cell 56:67–78
- Gaddam D, Stevens N, Hollien J (2013) Comparison of mRNA localization and regulation during endoplasmic reticulum stress in Drosophila cells. Mol Biol Cell 24:14–20
- Gallie DR (1991) The cap and poly(A) tail function synergistically to regulate mRNA translational efficiency. Genes Dev 5:2108–2116
- Gao M, Fritz DT, Ford LP, Wilusz J (2000) Interaction between a poly(A)-specific ribonuclease and the 5' cap influences mRNA deadenylation rates in vitro. Mol Cell 5:479–488
- Gao M, Wilusz CJ, Peltz SW, Wilusz J (2001) A novel mRNA-decapping activity in HeLa cytoplasmic extracts is regulated by AU-rich elements. EMBO J 20:1134–1143
- Garneau NL, Wilusz J, Wilusz CJ (2007) The highways and byways of mRNA decay. Nat Rev Mol Cell Biol 8:113–126
- Geisler S, Coller J (2013) RNA in unexpected places: long non-coding RNA functions in diverse cellular contexts. Nat Rev Mol Cell Biol 14:699–712
- Gerstberger S, Hafner M, Tuschl T (2014) A census of human RNA-binding proteins. Nat Rev Genet 15:829–845
- Ghildiyal M, Zamore PD (2009) Small silencing RNAs: an expanding universe. Nat Rev Genet 10:94–108
- Ghildiyal M, Seitz H, Horwich MD, Li C, Du T, Lee S, Xu J, Kittler EL, Zapp ML, Weng Z et al (2008) Endogenous siRNAs derived from transposons and mRNAs in Drosophila somatic cells. Science 320:1077–1081
- Ghosh S, Jacobson A (2010) RNA decay modulates gene expression and controls its fidelity. Wiley Interdiscip Rev RNA 1:351–361
- Goldstrohm AC, Wickens M (2008) Multifunctional deadenylase complexes diversify mRNA control. Nat Rev Mol Cell Biol 9:337–344
- Goldstrohm AC, Hook BA, Seay DJ, Wickens M (2006) PUF proteins bind Pop2p to regulate messenger RNAs. Nat Struct Mol Biol 13:533–539
- Goldstrohm AC, Seay DJ, Hook BA, Wickens M (2007) PUF protein-mediated deadenylation is catalyzed by Ccr4p. J Biol Chem 282:109–114
- Gong C, Maquat LE (2011) lncRNAs transactivate STAU1-mediated mRNA decay by duplexing with 3' UTRs via Alu elements. Nature 470:284–288
- Gray NK, Wickens M (1998) Control of translation initiation in animals. Annu Rev Cell Dev Biol 14:399–458
- Green CB, Besharse JC (1996) Identification of a novel vertebrate circadian clock-regulated gene encoding the protein nocturnin. Proc Natl Acad Sci U S A 93:14884–14888
- Green CB, Douris N, Kojima S, Strayer CA, Fogerty J, Lourim D, Keller SR, Besharse JC (2007) Loss of nocturnin, a circadian deadenylase, confers resistance to hepatic steatosis and dietinduced obesity. Proc Natl Acad Sci U S A 104:9888–9893
- Groisman I, Huang YS, Mendez R, Cao Q, Theurkauf W, Richter JD (2000) CPEB, maskin, and cyclin B1 mRNA at the mitotic apparatus: implications for local translational control of cell division. Cell 103:435–447
- Groisman I, Jung MY, Sarkissian M, Cao Q, Richter JD (2002) Translational control of the embryonic cell cycle. Cell 109:473–483
- Ha M, Kim VN (2014) Regulation of microRNA biogenesis. Nat Rev Mol Cell Biol 15:509-524
- Han D, Lerner AG, Vande Walle L, Upton JP, Xu W, Hagen A, Backes BJ, Oakes SA, Papa FR (2009a) IRE1alpha kinase activation modes control alternate endoribonuclease outputs to determine divergent cell fates. Cell 138:562–575
- Han J, Pedersen JS, Kwon SC, Belair CD, Kim YK, Yeom KH, Yang WY, Haussler D, Blelloch R, Kim VN (2009b) Posttranscriptional crossregulation between Drosha and DGCR8. Cell 136:75–84
- Harigaya Y, Jones BN, Muhlrad D, Gross JD, Parker R (2010) Identification and analysis of the interaction between Edc3 and Dcp2 in Saccharomyces cerevisiae. Mol Cell Biol 30:1446–1456
- He L (2010) Posttranscriptional regulation of PTEN dosage by noncoding RNAs. Sci Signal 3:pe39

- Hellen CU, Sarnow P (2001) Internal ribosome entry sites in eukaryotic mRNA molecules. Genes Dev 15:1593–1612
- Heo I, Ha M, Lim J, Yoon MJ, Park JE, Kwon SC, Chang H, Kim VN (2012) Mono-uridylation of pre-microRNA as a key step in the biogenesis of group II let-7 microRNAs. Cell 151:521–532
- Hinnebusch AG (2014) The scanning mechanism of eukaryotic translation initiation. Annu Rev Biochem 83:779–812
- Hollien J, Lin JH, Li H, Stevens N, Walter P, Weissman JS (2009) Regulated Ire1-dependent decay of messenger RNAs in mammalian cells. J Cell Biol 186:323–331
- Hudson WH, Pickard MR, de Vera IM, Kuiper EG, Mourtada-Maarabouni M, Conn GL, Kojetin DJ, Williams GT, Ortlund EA (2014) Conserved sequence-specific lincRNA-steroid receptor interactions drive transcriptional repression and direct cell fate. Nat Commun 5:5395
- Huez I, Creancier L, Audigier S, Gensac MC, Prats AC, Prats H (1998) Two independent internal ribosome entry sites are involved in translation initiation of vascular endothelial growth factor mRNA. Mol Cell Biol 18:6178–6190
- Huntzinger E, Kuzuoglu-Ozturk D, Braun JE, Eulalio A, Wohlbold L, Izaurralde E (2013) The interactions of GW182 proteins with PABP and deadenylases are required for both translational repression and degradation of miRNA targets. Nucleic Acids Res 41:978–994
- Ibba M, Soll D (2000) Aminoacyl-tRNA synthesis. Annu Rev Biochem 69:617-650
- Igreja C, Izaurralde E (2011) CUP promotes deadenylation and inhibits decapping of mRNA targets. Genes Dev 25:1955–1967
- Inge-Vechtomov S, Zhouravleva G, Philippe M (2003) Eukaryotic release factors (eRFs) history. Biol Cell 95:195–209
- Ingolia NT (2014) Ribosome profiling: new views of translation, from single codons to genome scale. Nat Rev Genet 15:205–213
- Ingolia NT, Lareau LF, Weissman JS (2011) Ribosome profiling of mouse embryonic stem cells reveals the complexity and dynamics of mammalian proteomes. Cell 147:789–802
- Ishmael FT, Fang X, Houser KR, Pearce K, Abdelmohsen K, Zhan M, Gorospe M, Stellato C (2011) The human glucocorticoid receptor as an RNA-binding protein: global analysis of glucocorticoid receptor-associated transcripts and identification of a target RNA motif. J Immunol 186:1189–1198
- Ivshina M, Lasko P, Richter JD (2014) Cytoplasmic polyadenylation element binding proteins in development, health, and disease. Annu Rev Cell Dev Biol 30:393–415
- Jackson RJ, Hellen CU, Pestova TV (2010) The mechanism of eukaryotic translation initiation and principles of its regulation. Nat Rev Mol Cell Biol 11:113–127
- Jackson RJ, Hellen CU, Pestova TV (2012) Termination and post-termination events in eukaryotic translation. Adv Protein Chem Struct Biol 86:45–93
- Jha S, Komar AA (2011) Birth, life and death of nascent polypeptide chains. Biotechnol J 6:623–640
- Jonas S, Izaurralde E (2013) The role of disordered protein regions in the assembly of decapping complexes and RNP granules. Genes Dev 27:2628–2641
- Jonas S, Izaurralde E (2015) Towards a molecular understanding of microRNA-mediated gene silencing. Nat Rev Genet 16:421–433
- Jonas S, Christie M, Peter D, Bhandari D, Loh B, Huntzinger E, Weichenrieder O, Izaurralde E (2014) An asymmetric PAN3 dimer recruits a single PAN2 exonuclease to mediate mRNA deadenylation and decay. Nat Struct Mol Biol 21:599–608
- Kampa D, Cheng J, Kapranov P, Yamanaka M, Brubaker S, Cawley S, Drenkow J, Piccolboni A, Bekiranov S, Helt G et al (2004) Novel RNAs identified from an in-depth analysis of the transcriptome of human chromosomes 21 and 22. Genome Res 14:331–342
- Kawahara H, Imai T, Imataka H, Tsujimoto M, Matsumoto K, Okano H (2008) Neural RNAbinding protein Musashi1 inhibits translation initiation by competing with eIF4G for PABP. J Cell Biol 181:639–653
- Kawamata T, Seitz H, Tomari Y (2009) Structural determinants of miRNAs for RISC loading and slicer-independent unwinding. Nat Struct Mol Biol 16:953–960
- Khaleghpour K, Svitkin YV, Craig AW, DeMaria CT, Deo RC, Burley SK, Sonenberg N (2001) Translational repression by a novel partner of human poly(A) binding protein, Paip2. Mol Cell 7:205–216

- Khvorova A, Reynolds A, Jayasena SD (2003) Functional siRNAs and miRNAs exhibit strand bias. Cell 115:209–216
- Kim JH, Richter JD (2006) Opposing polymerase-deadenylase activities regulate cytoplasmic polyadenylation. Mol Cell 24:173–183
- Kino T, Hurt DE, Ichijo T, Nader N, Chrousos GP (2010) Noncoding RNA gas5 is a growth arrestand starvation-associated repressor of the glucocorticoid receptor. Sci Signal 3:ra8
- Knoll M, Lodish HF, Sun L (2015) Long non-coding RNAs as regulators of the endocrine system. Nat Rev Endocrinol 11:151–160
- Kong J, Lasko P (2012) Translational control in cellular and developmental processes. Nat Rev Genet 13:383–394
- Korobeinikova AV, Garber MB, Gongadze GM (2012) Ribosomal proteins: structure, function, and evolution. Biochemistry (Mosc) 77:562–574
- Kozak M (1987) An analysis of 5'-noncoding sequences from 699 vertebrate messenger RNAs. Nucleic Acids Res 15:8125–8148
- Krichevsky AM, King KS, Donahue CP, Khrapko K, Kosik KS (2003) A microRNA array reveals extensive regulation of microRNAs during brain development. RNA 9:1274–1281
- Kuhn U, Wahle E (2004) Structure and function of poly(A) binding proteins. Biochim Biophys Acta 1678:67–84
- Kuzuoglu-Ozturk D, Huntzinger E, Schmidt S, Izaurralde E (2012) The Caenorhabditis elegans GW182 protein AIN-1 interacts with PAB-1 and subunits of the PAN2-PAN3 and CCR4-NOT deadenylase complexes. Nucleic Acids Res 40:5651–5665
- Lagos-Quintana M, Rauhut R, Lendeckel W, Tuschl T (2001) Identification of novel genes coding for small expressed RNAs. Science 294:853–858
- Lau NC, Lim LP, Weinstein EG, Bartel DP (2001) An abundant class of tiny RNAs with probable regulatory roles in *Caenorhabditis elegans*. Science 294:858–862
- Lee Y, Ahn C, Han J, Choi H, Kim J, Yim J, Lee J, Provost P, Radmark O, Kim S et al (2003) The nuclear RNase III Drosha initiates microRNA processing. Nature 425:415–419
- Lemaire PA, Anderson E, Lary J, Cole JL (2008) Mechanism of PKR activation by dsRNA. J Mol Biol 381:351–360
- Leppek K, Schott J, Reitter S, Poetz F, Hammond MC, Stoecklin G (2013) Roquin promotes constitutive mRNA decay via a conserved class of stem-loop recognition motifs. Cell 153:869–881
- Lewis BP, Burge CB, Bartel DP (2005) Conserved seed pairing, often flanked by adenosines, indicates that thousands of human genes are microRNA targets. Cell 120:15–20
- Li Y, Kiledjian M (2010) Regulation of mRNA decapping. Wiley Interdiscip Rev RNA 1:253–265
- Li Y, Lu J, Han Y, Fan X, Ding SW (2013) RNA interference functions as an antiviral immunity mechanism in mammals. Science 342:231–234
- Liang J, Saad Y, Lei T, Wang J, Qi D, Yang Q, Kolattukudy PE, Fu M (2010) MCP-induced protein 1 deubiquitinates TRAF proteins and negatively regulates JNK and NF-kappaB signaling. J Exp Med 207:2959–2973
- Ling SH, Qamra R, Song H (2011) Structural and functional insights into eukaryotic mRNA decapping. Wiley Interdiscip Rev RNA 2:193–208
- Liu H, Kiledjian M (2005) Scavenger decapping activity facilitates 5' to 3' mRNA decay. Mol Cell Biol 25:9764–9772
- Liu B, Qian SB (2014) Translational reprogramming in cellular stress response. Wiley Interdiscip Rev RNA 5:301–315
- Liu Q, Greimann JC, Lima CD (2006) Reconstitution, activities, and structure of the eukaryotic RNA exosome. Cell 127:1223–1237
- Londin E, Loher P, Telonis AG, Quann K, Clark P, Jing Y, Hatzimichael E, Kirino Y, Honda S, Lally M et al (2015) Analysis of 13 cell types reveals evidence for the expression of numerous novel primate- and tissue-specific microRNAs. Proc Natl Acad Sci U S A 112:E1106–E1115
- Luo W, Johnson AW, Bentley DL (2006) The role of Rat1 in coupling mRNA 3'-end processing to transcription termination: implications for a unified allosteric-torpedo model. Genes Dev 20:954–965

- Lykke-Andersen J, Wagner E (2005) Recruitment and activation of mRNA decay enzymes by two ARE-mediated decay activation domains in the proteins TTP and BRF-1. Genes Dev 19:351–361
- Ma XM, Blenis J (2009) Molecular mechanisms of mTOR-mediated translational control. Nat Rev Mol Cell Biol 10:307–318
- MacRae IJ, Zhou K, Doudna JA (2007) Structural determinants of RNA recognition and cleavage by Dicer. Nat Struct Mol Biol 14:934–940
- Martineau Y, Derry MC, Wang X, Yanagiya A, Berlanga JJ, Shyu AB, Imataka H, Gehring K, Sonenberg N (2008) Poly(A)-binding protein-interacting protein 1 binds to eukaryotic translation initiation factor 3 to stimulate translation. Mol Cell Biol 28:6658–6667
- Martinez J, Ren YG, Thuresson AC, Hellman U, Astrom J, Virtanen A (2000) A 54-kDa fragment of the Poly(A)-specific ribonuclease is an oligomeric, processive, and cap-interacting Poly(A)specific 3' exonuclease. J Biol Chem 275:24222–24230
- Marzluff WF, Wagner EJ, Duronio RJ (2008) Metabolism and regulation of canonical histone mRNAs: life without a poly(A) tail. Nat Rev Genet 9:843–854
- Matsushita K, Takeuchi O, Standley DM, Kumagai Y, Kawagoe T, Miyake T, Satoh T, Kato H, Tsujimura T, Nakamura H et al (2009) Zc3h12a is an RNase essential for controlling immune responses by regulating mRNA decay. Nature 458:1185–1190
- Mattick JS (2004) RNA regulation: a new genetics? Nature reviews. Genetics 5:316-323
- Maurel M, Chevet E, Tavernier J, Gerlo S (2014) Getting RIDD of RNA: IRE1 in cell fate regulation. Trends Biochem Sci 39:245–254
- Meister G (2013) Argonaute proteins: functional insights and emerging roles. Nat Rev Genet 14:447–459
- Meyuhas O, Kahan T (2015) The race to decipher the top secrets of TOP mRNAs. Biochim Biophys Acta 1849:801–811
- Michel YM, Poncet D, Piron M, Kean KM, Borman AM (2000) Cap-Poly(A) synergy in mammalian cell-free extracts. Investigation of the requirements for poly(A)-mediated stimulation of translation initiation. J Biol Chem 275:32268–32276
- Micklem DR (1995) mRNA localisation during development. Dev Biol 172:377-395
- Mignone F, Gissi C, Liuni S, Pesole G (2002) Untranslated regions of mRNAs. Genome Biol 3:REVIEWS0004
- Miki TS, Grosshans H (2013) The multifunctional RNase XRN2. Biochem Soc Trans 41:825-830
- Miller DL, Dibbens JA, Damert A, Risau W, Vadas MA, Goodall GJ (1998) The vascular endothelial growth factor mRNA contains an internal ribosome entry site. FEBS Lett 434:417–420
- Moraes KC, Wilusz CJ, Wilusz J (2006) CUG-BP binds to RNA substrates and recruits PARN deadenylase. RNA 12:1084–1091
- Muckenthaler M, Gray NK, Hentze MW (1998) IRP-1 binding to ferritin mRNA prevents the recruitment of the small ribosomal subunit by the cap-binding complex eIF4F. Mol Cell 2:383–388
- Mukherjee C, Patil DP, Kennedy BA, Bakthavachalu B, Bundschuh R, Schoenberg DR (2012) Identification of cytoplasmic capping targets reveals a role for cap homeostasis in translation and mRNA stability. Cell Rep 2:674–684
- Nagarajan VK, Jones CI, Newbury SF, Green PJ (2013) XRN 5'→3' exoribonucleases: structure, mechanisms and functions. Biochim Biophys Acta 1829:590–603
- Nakagawa S, Niimura Y, Gojobori T, Tanaka H, Miura K (2008) Diversity of preferred nucleotide sequences around the translation initiation codon in eukaryote genomes. Nucleic Acids Res 36:861–871
- Nakamura A, Sato K, Hanyu-Nakamura K (2004) Drosophila cup is an eIF4E binding protein that associates with Bruno and regulates oskar mRNA translation in oogenesis. Dev Cell 6:69–78
- Nechama M, Peng Y, Bell O, Briata P, Gherzi R, Schoenberg DR, Naveh-Many T (2009) KSRP-PMR1-exosome association determines parathyroid hormone mRNA levels and stability in transfected cells. BMC Cell Biol 10:70
- Nelson MR, Leidal AM, Smibert CA (2004) Drosophila cup is an eIF4E-binding protein that functions in Smaug-mediated translational repression. EMBO J 23:150–159

- Ng CK, Shboul M, Taverniti V, Bonnard C, Lee H, Eskin A, Nelson SF, Al-Raqad M, Altawalbeh S, Seraphin B et al (2015) Loss of the scavenger mRNA decapping enzyme DCPS causes syndromic intellectual disability with neuromuscular defects. Hum Mol Genet 24:3163–3171
- Okamura K, Hagen JW, Duan H, Tyler DM, Lai EC (2007) The mirtron pathway generates microRNA-class regulatory RNAs in Drosophila. Cell 130:89–100
- Ostareck DH, Ostareck-Lederer A, Shatsky IN, Hentze MW (2001) Lipoxygenase mRNA silencing in erythroid differentiation: the 3'UTR regulatory complex controls 60S ribosomal subunit joining. Cell 104:281–290
- Park E, Maquat LE (2013) Staufen-mediated mRNA decay. Wiley Interdiscip Rev RNA 4:423-435
- Pasquinelli AE, Reinhart BJ, Slack F, Martindale MQ, Kuroda MI, Maller B, Hayward DC, Ball EE, Degnan B, Muller P et al (2000) Conservation of the sequence and temporal expression of let-7 heterochronic regulatory RNA. Nature 408:86–89
- Pastori RL, Moskaitis JE, Buzek SW, Schoenberg DR (1991a) Coordinate estrogen-regulated instability of serum protein-coding messenger RNAs in *Xenopus laevis*. Mol Endocrinol 5:461–468
- Pastori RL, Moskaitis JE, Schoenberg DR (1991b) Estrogen-induced ribonuclease activity in Xenopus liver. Biochemistry 30:10490–10498
- Pechmann S, Frydman J (2013) Evolutionary conservation of codon optimality reveals hidden signatures of cotranslational folding. Nat Struct Mol Biol 20:237–243
- Peters L, Meister G (2007) Argonaute proteins: mediators of RNA silencing. Mol Cell 26:611-623
- Pfaff J, Hennig J, Herzog F, Aebersold R, Sattler M, Niessing D, Meister G (2013) Structural features of Argonaute-GW182 protein interactions. Proc Natl Acad Sci U S A 110:E3770–E3779
- Piccirillo C, Khanna R, Kiledjian M (2003) Functional characterization of the mammalian mRNA decapping enzyme hDcp2. RNA 9:1138–1147
- Poliseno L, Pandolfi PP (2015) PTEN ceRNA networks in human cancer. Methods 77–78:41–50
- Poliseno L, Salmena L, Zhang J, Carver B, Haveman WJ, Pandolfi PP (2010) A codingindependent function of gene and pseudogene mRNAs regulates tumour biology. Nature 465:1033–1038
- Popp MW, Maquat LE (2013) Organizing principles of mammalian nonsense-mediated mRNA decay. Annu Rev Genet 47:139–165
- Popp MW, Maquat LE (2014) The dharma of nonsense-mediated mRNA decay in mammalian cells. Mol Cells 37:1–8
- Poy MN, Eliasson L, Krutzfeldt J, Kuwajima S, Ma X, Macdonald PE, Pfeffer S, Tuschl T, Rajewsky N, Rorsman P et al (2004) A pancreatic islet-specific microRNA regulates insulin secretion. Nature 432:226–230
- Presnyak V, Alhusaini N, Chen YH, Martin S, Morris N, Kline N, Olson S, Weinberg D, Baker KE, Graveley BR et al (2015) Codon optimality is a major determinant of mRNA stability. Cell 160:1111–1124
- Quax TE, Claassens NJ, Soll D, van der Oost J (2015) Codon bias as a means to fine-tune gene expression. Mol Cell 59:149–161
- Reis FP, Pobre V, Silva IJ, Malecki M, Arraiano CM (2013) The RNase II/RNB family of exoribonucleases: putting the 'Dis' in disease. Wiley Interdiscip Rev RNA 4:607–615
- Rendl LM, Bieman MA, Vari HK, Smibert CA (2012) The eIF4E-binding protein Eap1p functions in Vts1p-mediated transcript decay. PLoS One 7, e47121
- Riis B, Rattan SI, Clark BF, Merrick WC (1990) Eukaryotic protein elongation factors. Trends Biochem Sci 15:420–424
- Rom E, Kim HC, Gingras AC, Marcotrigiano J, Favre D, Olsen H, Burley SK, Sonenberg N (1998) Cloning and characterization of 4EHP, a novel mammalian eIF4E-related cap-binding protein. J Biol Chem 273:13104–13109
- Roy G, De Crescenzo G, Khaleghpour K, Kahvejian A, O'Connor-McCourt M, Sonenberg N (2002) Paip1 interacts with poly(A) binding protein through two independent binding motifs. Mol Cell Biol 22:3769–3782

- Sandler H, Kreth J, Timmers HT, Stoecklin G (2011) Not1 mediates recruitment of the deadenylase Caf1 to mRNAs targeted for degradation by tristetraprolin. Nucleic Acids Res 39:4373–4386
- Sarkissian M, Mendez R, Richter JD (2004) Progesterone and insulin stimulation of CPEBdependent polyadenylation is regulated by Aurora A and glycogen synthase kinase-3. Genes Dev 18:48–61
- Sasikumar AN, Perez WB, Kinzy TG (2012) The many roles of the eukaryotic elongation factor 1 complex. Wiley Interdiscip Rev RNA 3:543–555
- Schafer IB, Rode M, Bonneau F, Schussler S, Conti E (2014) The structure of the Pan2-Pan3 core complex reveals cross-talk between deadenylase and pseudokinase. Nat Struct Mol Biol 21:591–598
- Schmid M, Jensen TH (2008) Quality control of mRNP in the nucleus. Chromosoma 117:419-429
- Schoenberg DR, Maquat LE (2012) Regulation of cytoplasmic mRNA decay. Nat Rev Genet 13:246–259
- Schwanhausser B, Busse D, Li N, Dittmar G, Schuchhardt J, Wolf J, Chen W, Selbach M (2011) Global quantification of mammalian gene expression control. Nature 473:337–342
- She M, Decker CJ, Chen N, Tumati S, Parker R, Song H (2006) Crystal structure and functional analysis of Dcp2p from *Schizosaccharomyces pombe*. Nat Struct Mol Biol 13:63–70
- She M, Decker CJ, Svergun DI, Round A, Chen N, Muhlrad D, Parker R, Song H (2008) Structural basis of dcp2 recognition and activation by dcp1. Mol Cell 29:337–349
- Shirai YT, Suzuki T, Morita M, Takahashi A, Yamamoto T (2014) Multifunctional roles of the mammalian CCR4-NOT complex in physiological phenomena. Front Genet 5:286
- Siddiqui N, Mangus DA, Chang TC, Palermino JM, Shyu AB, Gehring K (2007) Poly(A) nuclease interacts with the C-terminal domain of polyadenylate-binding protein domain from poly(A)binding protein. J Biol Chem 282:25067–25075
- Sinturel F, Brechemier-Baey D, Kiledjian M, Condon C, Benard L (2012) Activation of 5'-3' exoribonuclease Xrn1 by cofactor Dcs1 is essential for mitochondrial function in yeast. Proc Natl Acad Sci U S A 109:8264–8269
- Slayter HS, Warner JR, Rich A, Hall CE (1963) The visualization of polyribosomal structure. J Mol Biol 7:652–657
- Sonenberg N, Hinnebusch AG (2009) Regulation of translation initiation in eukaryotes: mechanisms and biological targets. Cell 136:731–745
- Song MG, Bail S, Kiledjian M (2013) Multiple Nudix family proteins possess mRNA decapping activity. RNA 19:390–399
- Stebbins-Boaz B, Cao Q, de Moor CH, Mendez R, Richter JD (1999) Maskin is a CPEB-associated factor that transiently interacts with eIF-4E. Mol Cell 4:1017–1027
- Tarrant D, von der Haar T (2014) Synonymous codons, ribosome speed, and eukaryotic gene expression regulation. Cell Mol Life Sci 71:4195–4206
- Tavares MR, Pavan IC, Amaral CL, Meneguello L, Luchessi AD, Simabuco FM (2015) The S6K protein family in health and disease. Life Sci 131:1–10
- Tay Y, Rinn J, Pandolfi PP (2014) The multilayered complexity of ceRNA crosstalk and competition. Nature 505:344–352
- Tharun S, Parker R (2001) Targeting an mRNA for decapping: displacement of translation factors and association of the Lsm1p-7p complex on deadenylated yeast mRNAs. Mol Cell 8:1075–1083
- Thornton JE, Chang HM, Piskounova E, Gregory RI (2012) Lin28-mediated control of let-7 microRNA expression by alternative TUTases Zcchc11 (TUT4) and Zcchc6 (TUT7). RNA 18:1875–1885
- Tripathi V, Ellis JD, Shen Z, Song DY, Pan Q, Watt AT, Freier SM, Bennett CF, Sharma A, Bubulya PA et al (2010) The nuclear-retained noncoding RNA MALAT1 regulates alternative splicing by modulating SR splicing factor phosphorylation. Mol Cell 39:925–938
- Tucker M, Valencia-Sanchez MA, Staples RR, Chen J, Denis CL, Parker R (2001) The transcription factor associated Ccr4 and Caf1 proteins are components of the major cytoplasmic mRNA deadenylase in *Saccharomyces cerevisiae*. Cell 104:377–386

- Tucker M, Staples RR, Valencia-Sanchez MA, Muhlrad D, Parker R (2002) Ccr4p is the catalytic subunit of a Ccr4p/Pop2p/Notp mRNA deadenylase complex in *Saccharomyces cerevisiae*. EMBO J 21:1427–1436
- Uchida N, Hoshino S, Katada T (2004) Identification of a human cytoplasmic poly(A) nuclease complex stimulated by poly(A)-binding protein. J Biol Chem 279:1383–1391
- Ulitsky I, Bartel DP (2013) lincRNAs: genomics, evolution, and mechanisms. Cell 154:26-46
- Van Etten J, Schagat TL, Hrit J, Weidmann CA, Brumbaugh J, Coon JJ, Goldstrohm AC (2012) Human Pumilio proteins recruit multiple deadenylases to efficiently repress messenger RNAs. J Biol Chem 287:36370–36383
- Vlasova-St Louis I, Bohjanen PR (2011) Coordinate regulation of mRNA decay networks by GU-rich elements and CELF1. Curr Opin Genet Dev 21:444–451
- Wang, Z, Jiao X, Carr-Schmid A, Kiledjian M (2002) The hDcp2 protein is a mammalian decapping enzyme. Proceedings of the National Academy of Sciences USA 99(20):12663–12668
- Wang X, He C (2014) Dynamic RNA modifications in posttranscriptional regulation. Mol Cell 56:5-12
- Wang M, Pestov DG (2011) 5'-end surveillance by Xrn2 acts as a shared mechanism for mammalian pre-rRNA maturation and decay. Nucleic Acids Res 39:1811–1822
- Wang XH, Aliyari R, Li WX, Li HW, Kim K, Carthew R, Atkinson P, Ding SW (2006) RNA interference directs innate immunity against viruses in adult Drosophila. Science 312:452–454
- Warner JR, Knopf PM, Rich A (1963) A multiple ribosomal structure in protein synthesis. Proc Natl Acad Sci U S A 49:122–129
- Weidmann CA, Raynard NA, Blewett NH, Van Etten J, Goldstrohm AC (2014) The RNA binding domain of Pumilio antagonizes poly-adenosine binding protein and accelerates deadenylation. RNA 20:1298–1319
- Wells SE, Hillner PE, Vale RD, Sachs AB (1998) Circularization of mRNA by eukaryotic translation initiation factors. Mol Cell 2:135–140
- Wickramasinghe VO, Laskey RA (2015) Control of mammalian gene expression by selective mRNA export. Nat Rev Mol Cell Biol 16:431–442
- Wilson DN, Doudna Cate JH (2012) The structure and function of the eukaryotic ribosome. Cold Spring Harb Perspect Biol 4
- Wilson T, Treisman R (1988) Removal of poly(A) and consequent degradation of c-fos mRNA facilitated by 3' AU-rich sequences. Nature 336:396–399
- Wolf J, Valkov E, Allen MD, Meineke B, Gordiyenko Y, McLaughlin SH, Olsen TM, Robinson CV, Bycroft M, Stewart M et al (2014) Structural basis for Pan3 binding to Pan2 and its function in mRNA recruitment and deadenylation. EMBO J 33:1514–1526
- Yamashita A, Chang TC, Yamashita Y, Zhu W, Zhong Z, Chen CY, Shyu AB (2005) Concerted action of poly(A) nucleases and decapping enzyme in mammalian mRNA turnover. Nat Struct Mol Biol 12:1054–1063
- Yang N, Kazazian HH Jr (2006) L1 retrotransposition is suppressed by endogenously encoded small interfering RNAs in human cultured cells. Nat Struct Mol Biol 13:763–771
- Yoda M, Kawamata T, Paroo Z, Ye X, Iwasaki S, Liu Q, Tomari Y (2010) ATP-dependent human RISC assembly pathways. Nat Struct Mol Biol 17:17–23
- Yu JH, Yang WH, Gulick T, Bloch KD, Bloch DB (2005) Ge-1 is a central component of the mammalian cytoplasmic mRNA processing body. RNA 11:1795–1802
- Yusupova G, Yusupov M (2014) High-resolution structure of the eukaryotic 80S ribosome. Annu Rev Biochem 83:467–486
- Zekri L, Kuzuoglu-Ozturk D, Izaurralde E (2013) GW182 proteins cause PABP dissociation from silenced miRNA targets in the absence of deadenylation. EMBO J 32:1052–1065
- Zeng Y, Cullen BR (2003) Sequence requirements for micro RNA processing and function in human cells. RNA 9:112–123

Post-transcriptional Regulation of Insulin and Insulin Like Growth Factors

Eun Kyung Lee and Wook Kim

1 Introduction

1.1 Insulin and Insulin-Like Growth Factors

Insulin and insulin-like growth factors (IGFs) are members of the same family of insulin-like peptides that share significant structural homology, and that engage common targets of downstream signaling pathways (Rajpathak et al. 2009; Cooke et al. 1991). There are, however, some major structural differences and tissue-specific activities attributed to the various family members. Both insulin and IGFs are composed of an alpha and a beta chain connected by disulfide bonds (Rinderknecht and Humbel 1978). Insulin is produced by proteolytic cleavage of a precursor protein, preproinsulin, solely in pancreatic β-cells; in contrast, IGFs are produced mainly by the liver and also in target tissues by a mechanism that does not require proteolytic cleavage of the c-peptide region. Similar to IGFs, insulin plays an important role in the regulation of cellular processes such as growth, proliferation, and survival in nearly all tissues. However, the major role of insulin is to tightly regulate glucose homeostasis in insulin-sensitive tissues, including liver, muscle, and adipose tissue. IGFs also have insulin-like activity as they can regulate glucose and lipid homeostasis by activating specific signal transduction pathways in peripheral tissues such as muscle (Clemmons 2006; Rajpathak et al. 2009). Both insulin and IGF synthesis are coordinately regulated in response to nutrients at the post-transcriptional level by

E.K. Lee, PhD (🖂)

W. Kim, PhD

Department of Biochemistry, College of Medicine, The Catholic University of Korea, 222 Banpodaero, Seocho-gu, Seoul 137-701, South Korea e-mail: leeek@catholic.ac.kr

Department of Molecular Science and Technology, Ajou University, San 5, Woncheon-dong, Yeongtong-gu, Suwon 443-749, South Korea e-mail: wookkim21@ajou.ac.kr

[©] Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms*

in Endocrine Regulation, DOI 10.1007/978-3-319-25124-0_2

trans-acting and mRNA-binding factors. Post-transcriptional mechanisms that regulate the stability and translation of mRNAs encoding insulin and IGFs include those that affect pre-mRNA splicing as well as mRNA stability and translation (Lee and Gorospe 2010; Panda et al. 2013); these regulatory mechanisms are particularly effective in eliciting acute changes in protein abundance in response to stimuli such as altered glucose levels.

In this chapter, we describe mRNA-binding factors involved in post-transcriptional regulation of mRNAs encoding insulin and IGFs and the influence of these factors on alternative splicing, mRNA stability, and translation.

1.2 RNA Binding Proteins and MicroRNAs

Tight regulation of gene expression is essential to maintain cellular homeostasis in response to various cellular signals. Besides transcriptional control, the production of insulin and IGFs are critically regulated at the post-transcriptional level. Post-transcriptional regulation at the RNA level in eukaryotic cells is broadly defined as the control of gene expression via the activity of pre-mRNA splicing factors, modulation of RNA maturation, and control of mRNA transport, storage, decay, and translation (Mitchell and Tollervey 2000; Orphanides and Reinberg 2002; Moore 2005; Lee and Gorospe 2010).

RNA binding proteins (RBPs) are varied in both structure and complexity and influence gene expression by affecting all aspects of RNA metabolism (Lunde et al. 2007). In particular, turnover- and translation-regulatory RBPs (TTR-RBPs) are involved in the regulation of mRNA stability and translation rates (Pullmann et al. 2007). RBPs dynamically interact with both coding and non-coding RNAs and form protein-RNA networks, which enables post-transcriptional regulation of gene expression (Lukong et al. 2008; Lunde et al. 2007; Moore 2005).

MicroRNAs (miRNAs) are a large group of small non-coding RNAs (ncRNAs) (19-23-nucletotide long) that negatively regulate gene expression by triggering mRNA decay and/or translational suppression (Bartel 2009; Mathonnet et al. 2007; Lim et al. 2005). miRNAs are assembled by the Argonaute (Ago) protein family into the miRNA-induced silencing complex and direct the RNA-induced silencing complex (RISC) to target mRNAs; this has a profound influence on various cellular processes, including proliferation, death, differentiation, and development (Kim et al. 2009; Chekulaeva et al. 2006). miRNAs are now recognized as pivotal post-transcriptional regulators of gene expression during physiologic and pathologic processes (Flynt and Lai 2008). Differential expression RBPs of miRNAs is implicated in the pathogenesis of human diseases such as cancer, neurodegenerative diseases and diabetes (Guay and Regazzi 2013; van Kouwenhove et al. 2011; Kim and Lee 2012; Lukong et al. 2008; Cooper et al. 2009).

Recently, the mechanisms by which several post-transcriptional regulators modulate the expression of *INS* and *IGF* mRNAs have been elucidated. In this chapter, the most recent findings regarding post-transcriptional regulation of insulin and IGFs by RBPs and miRNAs will be reviewed.

2 Post-transcriptional Regulation of Insulin

Insulin is synthesized in pancreatic β -cells in order to maintain a very narrow blood glucose range. Insulin biosynthesis is predominantly regulated in response to glucose through stimulation of *INS* transcription, as well as post-transcriptional regulation of *INS* mRNA (Leibiger et al. 1998; Welsh et al. 1985; Itoh and Okamoto 1980; Lee and Gorospe 2010). Post-transcriptional regulation of *INS* mRNA includes control of pre-mRNA splicing and maturation as well as the modulation of mRNA stability and translation (Mitchell and Tollervey 2000; Orphanides and Reinberg 2002; Moore 2005); all of these processes are predominantly governed by RBPs and miRNAs (Valencia-Sanchez et al. 2006; Keene 2007).

2.1 Alternative Splicing of INS mRNA

The *INS* gene contains three exons and two introns (Bell et al. 1980; Steiner et al. 1985). Alternative splicing generates variants of *INS* mRNA that are translated with different efficiencies (Shalev et al. 2002; Minn et al. 2005; Hernandez-Sanchez et al. 2003; Panda et al. 2010). The coding region of the human *INS* gene starts in exon 2, and therefore the 5' UTR includes exon 1 and part of exon 2 (Bell et al. 1980; Steiner et al. 1985). Both introns are flanked by canonical splice sites, and a cryptic 5' splice site exists in intron 1 (Fig. 2.1). Alternative splicing occurs at this splice site, thereby generating an *INS* mRNA splice variant retaining the first 26 bases of intron 1



Fig. 2.1 Schematic representation of the human *insulin* (**a**) and mouse *insulin* 2 (**b**) genes. The both genes consist of three exons (*boxes*), and two introns (*lines*). The coding region shown as *gray boxes* starts in exon 2 and the *arrows* indicate the translation start sites. Intron 1 in both the human *insulin* (**a**) and mouse *insulin* 2 (**b**) genes is flanked by the canonical 5'-splice site sequence AG/GT and the 3'-splice site AG. The *arrowheads* indicate the 5'-alternative splice site within intron 1 in the human *insulin* gene (**a**) and 3'-alternative splice site within exon 2 in the mouse *insulin* 2 gene (**b**). The 5' UTR shown as *white boxes* includes exon 1 and part of exon 2. In humans, usage of the 5'-alternative splice site results in loss of the first 12 bp of exon 2 without altering the coding sequence (**b**) (Modified from Shalev et al. 2002 and Panda et al. 2010)

(Shalev et al. 2002). Through alteration of the 5' UTR secondary structure, which may enhance ribosome binding, this engenders a higher translational efficiency of this splice variant in comparison to *INS* mRNA in vitro and in vivo. The expression of this splice variant is up-regulated in response to glucose and has been found to be significantly higher in diabetic/insulin-resistant mice than in wild-type littermates (Minn et al. 2005). The mouse *INS2* gene also undergoes alternative splicing to generate a variant that contains 5' UTR lacking 12 bases; this does not alter the coding sequence (Panda et al. 2010) (Fig. 2.1). This *INS2* splice variant constitutes about 75 % of total *INS2* mRNA and is more efficiently translated in cells. Thus, alternative splicing of *INS* mRNA is also an important mechanism for post-transcriptional regulation of insulin biosynthesis. However, so far, the precise factors required for the regulation of *INS* mRNA alternative splicing remain to be identified.

2.2 Post-transcriptional Regulation of INS mRNA by miRNAs

The discovery of miRNAs has added a novel regulatory layer to post-transcriptional control of *INS* mRNA in pancreatic β -cells. The major miRNAs reported to play a direct role in this process in pancreatic β -cells are shown in Table 2.1 and have been discussed below.

2.2.1 miR-25 and 92a

In pancreatic β -cells, miRNAs regulate glucose-stimulated insulin secretion and β -cell survival as well as insulin biosynthesis (Kim and Lee 2012). Recently, the Jeyaseelan group reported two miRNAs (miR-25 and miR-92a) as direct regulators of insulin biosynthesis at the mRNA level (Setyowati Karolina et al. 2013). The 3' UTR of *INS* mRNA contains binding sites for miR-25 and 92a that are conserved in the rat and mouse and are partially conserved in humans. Overexpression of miR-25 or miR-92a not only reduced *INS* mRNA but also its biosynthesis; conversely, introduction of anti-miR-25 or miR-92a increases insulin biosynthesis. Interestingly, significant up-regulation of miR-25 and miR92a is observed in the pancreas of diabetic rats, in which *INS* mRNA levels are decreased.

2.2.2 miR-196b

Although miRNAs typically target the 3' UTR of the mRNA for functional inhibition(Pillai et al. 2007; Yekta et al. 2004), they can also target the 5' UTR and coding regions of mRNA as well as increase translation of target mRNAs through the 3' and 5' UTR (Zhou et al. 2009; Orom et al. 2008; Vasudevan et al. 2007). Most recently, the Seshadri group has reported that miR-196b can directly increase *INS* mRNA translation by targeting the 5' UTR (Panda et al. 2014). Mouse miR-196b specifically targets the 5' UTR of *INS* mRNA and the binding site of miR-196b overlaps with that of HuD, a 5' UTR-associated RBP that represses *INS* mRNA translation and this is likely Ago2-dependent process, because miR-196b-mediated activation of insulin expression is abolished when Ago2 levels are depleted (Panda et al. 2014). Interestingly, suppression of miR-196b expression causes increased association of

		INS/IGF		
Target mRNA	Regulators	levels	Function	Reference
INS	HuD	Down	Translational reppression of insulin mRNA	Lee and Kim (2012)
	РТВ	Up	Glucose-induced stabilization of insulin mRNA	Tillmar L (2002)
	PDI and PABP	Up	Translational activation of insulin mRNA	Kulkarni SD (2011)
	miR-196b	Up	Translational activation of insulin mRNA	Panda AC (2014)
	miR-25 and -92	Down	Destabilization of insulin mRNA	Karolina DS (2013)
IGF-II	Zcchc11	Up	up-regulation of IGF-I by uridylation of IGF-I targeting miRNAs	Joens MR (2012)
	Nocturnin	Down	destabilizing of long form of IGF-I mRNA by deadenylation	Kawai M (2010)
	SF2/ASF		Exon 5 inclusion during myogenic differentiation, suclse hypertrophy, and myopathy	Smith PJ (2002)
	miR-1	Down	Translational inhibition of IGF-I in glucose-induced apoptosis	Yu X (2008)
	miR-1	Down	Translational inhibition of IGF-I in cartilage cell proliferation	Hu X (2013)
	miR-1	Down	Translational inhibition of IGF-I in muscle cell and inhibition of IGF signaling	Elia L (2009)
	miR-1/206	Down	Promotion of apoptosis after myocardiac infraction	Shan ZX (2009)
	miR-29	Down	promotion of myoblastic transition of HSC	Kwiecinski M (2012)
	miR-29	Down	inhibition of IGF-I signaling in aged brain	Fenn AM (2013)
	IMPs	Down	translational reppression of IGF-II L3 mRNA during embryonic development	Nielsen J (1999)
	IMP3	Up	translational activation of IGF-II L3 mRNA in K562m glioblastoma	Liao B (2005, 2011)
	IMP2	Up	IRES translation activation of IGF-II L3 mRNA during embryogenesis	Dai N (2011)
	Lin-28	Up	Translational activation of IGF-II and promotion of myogenesis	Polesskaya A (2007)
	miR-125b	Down	Translational inhibition of IGF-II and inhibition of muscle differentiation	Ge Y (2011)
	miR-100	Down	Translational inhibition of IGF-II and inhibition of cell ;proliferation and survival signaing in breast cancer	Gebeshuber CA (2013)

 Table 2.1
 Post-transcriptional regulators of insulin and insulin-like growth factors



Fig. 2.2 A proposed model of HuD and miR-196b action on insulin biosynthesis

HuD with the 5' UTR of *INS* mRNA and, conversely, this association is modestly decreased by overexpression of miR-196b, probably due to altered stem-loop structure of the 5' UTR. HuD and miR-196b might compete for binding to the same site in the 5' UTR of *INS* mRNA, providing an example of coordinated regulation of translation by miRNA and RBP (Fig. 2.2).

2.3 Post-transcriptional Regulation of INS mRNA by RBPs

Despite the short UTRs of *INS* mRNA, post-transcriptional regulation of *INS* mRNA is mediated through the cooperative action of a stem-loop in the 5' UTR and the conserved UUGAA sequence in the 3' UTR (Wicksteed et al. 2001). However, the identities of the specific factor(s) that associate with these elements had remained unclear until recently. Recently, several RBPs in pancreatic β -cells have been identified as pivotal post-transcriptional regulators of *INS* mRNA due to their effects on its stability and translation.

2.3.1 HuD

The Hu/ELAV (human antigen/embryonic lethal abnormal vision-like) family includes the ubiquitously expressed HuR (HuA) and the primarily neuronal HuB, HuC, and HuD. HuD/ELAVL4 is reported to specifically and directly affect posttranscriptional regulation of *INS* mRNA (Kim and Lee 2012). Like HuB and HuC, HuD expression was believed to be restricted primarily to neurons (Hinman and Lou 2008). However, recently it has been shown to be expressed in insulin-producing pancreatic β -cells (Lee et al. 2012; Poy et al. 2009), in which its levels are controlled by the insulin signaling pathway, sequentially implicating Irs2, PI3K, Akt, and FoxO1 (Fig. 2.2). Like other Hu/ELAV family proteins, HuD controls stability and translation of target mRNAs by binding to 5' or 3' UTR bearing AU- and U-rich sequences through three highly conserved RNA recognition motifs (RRMs) (Hinman and Lou 2008; Pascale et al. 2008).

In pancreatic β-cells, HuD, but not HuR and other RBPs, including HuR, TIAR [T-cell-restricted intracellular antigen-1 (TIA-1)-related protein], heterogeneous nuclear ribonucleoprotein (hnRNP) K, and nuclear factor (NF) 90, as well as hnRNP C, Ago, and the fragile X syndrome protein (FMRP), associates with a 22-nucleotide segment in the 5' UTR, but not with the coding region (CR) or 3'UTR of mouse INS2 mRNA (Lee et al. 2012). Binding of HuD to the INS2 5' UTR represses insulin production by reducing the INS2 mRNA translation, and not by altering INS2 mRNA stability. This binding is repressed by glucose treatment, which induces a rapid and robust release of INS2 mRNA from the HuD complex within 30 min of glucose stimulation thereby enables translation of INS2 mRNA (Lee et al. 2012) (Fig. 2.2). Consistently, insulin levels are increased in pancreatic β -cells of HuD-null mice, and conversely, decreased in HuD-transgenic mice. This modulation of insulin by HuD in vivo likely contributes the homeostatic regulation of plasma glucose and insulin concentrations. In support of this is the finding that HuD-transgenic mice display impaired glucose clearance from the blood due to lower plasma insulin levels, thus defining HuD as a pivotal posttranscriptional regulator of insulin.

2.3.2 Polypyrimidine Tract Binding Protein

The polypyrimidine tract-binding protein (PTB), also known as hnRNP I, binds to pyrimidine-rich sequences of single stranded target mRNAs through four RRMs and plays an important role in several cellular processes such as alternative splicing (Garcia-Blanco et al. 1989; Valcarcel and Gebauer 1997; Wagner and Garcia-Blanco 2001), polyadenylation of pre-mRNAs (Lou et al. 1999; Moreira et al. 1998), cytoplasmic RNA localization (Cote et al. 1999), mRNA stability (Tillmar et al. 2002; Knoch et al. 2004) and translation initiation (Hellen et al. 1993). Glucose-induced stabilization of INS mRNA is a key event in the control of insulin biosynthesis and PTB plays an essential role in this process. In pancreatic β -cells, glucose induces the binding of PTB to the polypyrimidine-rich sequence located in the 3' UTR of INS mRNA to stabilize INS mRNA, resulting in a glucose-dependent increase in INS mRNA (Tillmar et al. 2002; Tillmar and Welsh 2002). Glucose stimulation of β -cells enhances the cytoplasmic function of PTB by promoting its accumulation in the cytoplasm, upon which PTB binds and stabilizes INS mRNAs (Knoch et al. 2006). The cytoplasmic accumulation of PTB is regulated by cAMP and protein kinase A (PKA)-dependent phosphorylation (Xie et al. 2003; Knoch et al. 2006). Activation of PKA by elevation of cAMP levels causes the direct phosphorylation and nucleo-cytoplasmic transport of PTB (Fig. 2.3). Consistently, glucagon-like peptide 1 (GLP-1), which activates PKA and potentiates glucosestimulated insulin gene expression and secretion by increasing cAMP levels in β -cells, promotes the phosphorylation of PTB, and conceivably, its cytoplasmic accumulation, which in turn enhances the levels of mRNAs containing PTB binding sites in their 3' UTR (Knoch et al. 2006). However, GLP-1does not stabilize INS mRNA significantly in either rat insulinoma INS-1 cells or freshly isolated islets (Knoch et al. 2006).



Fig. 2.3 A proposed model of PTB action on insulin biosynthesis

2.3.3 PDI and PABP

Poly(A)-binding protein (PABP) is a multifunctional RNA binding protein with an N-terminal RNA binding domain composed of four RRMs, and a C-terminal helical domain (Adam et al. 1986), which is important for its interaction with other proteins and for cooperative binding to poly(A) tails (Melo et al. 2003). Through poly(A) tail binding, PABP modulates the stability and translation of target mRNAs (Sonenberg and Dever 2003). Moreover, PABP can also bind to the A-rich sequence in the 5' UTR to regulate key steps in mRNA translation (de Melo Neto et al. 1995; Oberer et al. 2005). In pancreatic β -cells, PABP binds specifically to the 5' UTR of INS mRNA and increases the rate of ribosome recycling through its interaction with other translation factors; this leads to higher insulin translation during glucose stimulation (Kulkarni et al. 2011). The binding of PABP to the 5' UTR of INS mRNA during glucose stimulation is regulated by protein disulfide isomerase (PDI), which was the first identified ER-resident catalyst of native disulfide bond formation (Anfinsen 1973). Glucose stimulation of β-cells induces PDI activation by promoting its phosphorylation. Activated PDI interacts with PABP and catalyzes the reduction of the PABP disulfide bond resulting in specific binding of PABP to the 5' UTR of INS mRNA and increased insulin translation (Kulkarni et al. 2011).

3 Post-transcriptional Regulation of IGF

Insulin-like growth factor (IGF) is a protein ligand with high sequence similarity to insulin and functions as a key regulator of growth, survival, and differentiation in most cell types (Humbel 1990). There are various mechanisms by which IGF is itself induced and activated, and discrete pathways operate during proliferation,

apoptosis, differentiation, and metabolism (King et al. 1982; Panagakos 1993; Zheng et al. 2000). Production of IGF isoforms is tightly regulated at transcriptional, post-transcriptional, and translational levels (Chew et al. 1995; Adamo et al. 1991; Lowe et al. 1989; Hepler et al. 1990; Zhang et al. 1998; Foyt et al. 1991). Dysregulation of IGF levels is responsible for several pathological conditions such as diabetes, cancer, and aging (Cohen and LeRoith 2012). In this section, we review the post-transcriptional regulators of IGFs and their impact on IGF signaling.

3.1 Alternative Splicing of IGF mRNA

Although various splice variants of IGF-I and IGF-II have been characterized (Goldspink and Yang 2004; Velloso and Harridge 2010; Weller et al. 1993; Yang et al. 1995), the factors that govern the spicing factors are largely unknown. The serine-arginine protein splicing factor-1/alternative splicing factor (SF2/ASF) binds to a purine-rich sequence in exon 5 of *IGF-I* mRNA to promote its inclusion in the mature transcript (Smith et al. 2002). Exon 5 inclusion is responsible for myogenic differentiation, muscle hypertrophy, and myopathy (Barton et al. 2002; Matheny and Nindl 2011). The 5' UTR composition of *IGF-II* mRNA varies throughout development due to alternative splicing (Ohlsson et al. 1994; Ekstrom et al. 1995; Monk et al. 2006), although the mechanisms and regulators required for this currently remain uncharacterized.

3.2 Post-transcriptional Regulation of *IGF* mRNA by miRNAs

3.2.1 Regulation of *IGF-I* mRNA by miR-1, miR-206, miR-29, and miR-320

IGF-I shows high sequence similarity with insulin and regulates various cellular processes such as cell growth, proliferation, survival, and insulin sensitivity in liver, muscle, and kidney (Lee and Gorospe 2010). Several miRNAs are known to modulate IGF-I expression in different model systems. Among them, miR-1 was shown to target the 3' UTR of IGF-I mRNA and down-regulate its expression; in cardiomyocytes, miR-1 levels increased during glucose-induced apoptosis and relieved the anti-apoptotic actions of IGF-I (Yu et al. 2008). miR-1 also down-regulated IGF-1 expression in cardiac and skeletal muscle (Elia et al. 2009) and inhibited cell proliferation by directly targeting IGF-I in Chinese sika deer-derived cartilage cells (Hu et al. 2013). Furthermore, miR-1 and miR-206 levels were up-regulated after myocardial infraction, which resulted in a decrease in IGF-I (Shan et al. 2009). Down-regulation of IGF by miR-320 was also reported in myocardial microvascular endothelial cells isolated from type 2 diabetic Goto-Kakizaki rats (Wang et al. 2009). Recently, miR-29a and miR-29b were shown to increase during the myofibroblastic transition of hepatic stellate cells or in the aged brain, resulting in decreased IGF-1 levels (Kwiecinski et al. 2012; Fenn et al. 2013).

3.2.2 Regulation of IGF-II mRNA by miR-100 and miR-125b

IGF-II functions as an embryonic regulator of myogenesis by initiating myogenic differentiation; it is also essential for metastasis of breast cancers (McCann et al. 1996; Pravtcheva and Wise 1998; Florini et al. 1991). IGF-II expression is modulated by miR-125b and miR-100; miR-125b is involved in the down-regulation of IGF-II via direct targeting of 3' UTR of *IGF-II* mRNA, and its expression decreases during myoblast differentiation (Ge et al. 2011). miR-125b negatively regulates myoblast differentiation by reducing the IGF-II level. miR-100 also down-regulates IGF-II expression, which leads to the inhibition of cell proliferation and reduces survival signaling in breast cancer cells; consistent with this, miR-100 expression has been found to decrease in human breast cancers (Gebeshuber and Martinez 2013).

3.3 Post-transcriptional Regulation of IGF mRNA by RBPs

3.3.1 Zcchc11

Zcchc11 (zinc finger CCHC domain-containing protein 11, terminal uridyltransferase 4, TUTase4) is a uridyltransferase that can suppress miRNA biogenesis by mediating terminal uridylation of miRNAs, including pre-let-7 and miR-26a (Hagan et al. 2009; Heo et al. 2009; Jones et al. 2009). Zcchc11 down-regulates let-7 in embryonic stem cells by uridylation of pre-let-7 via interaction with Lin28, which is essential for maintaining pluripontency (Hagan et al. 2009). Zcchc11 also catalyzes the 3' UTR uridylation of miR-26a, thereby promoting expression of cytokines including interlukin-6 (IL-6) (Jones et al. 2009). In Zcchc11-deficient mice, the length and frequency of terminal uridines for diverse mature miRNAs, including miR-126b and miR-379 that target *IGF-1* mRNA, decreases (Jones et al. 2012). Uridylation of these miRNAs prevents them from suppressing IGF-I, thereby increasing its expression. Zcchc11 overexpression up-regulates IGF-I level by stabilizing *IGF-I* mRNA. This reduction in IGF-I level contributes to the increased mortality and reduced growth of Zcchc11-deficient mice because IGF-I is essential for post-natal growth and survival (Jones et al. 2012) (Fig. 2.4).



Fig. 2.4 A proposed model of Zcchc11 action on IGF-I biosynthesis

3.3.2 Nocturnin

Nocturnin (Noc, Ccrn41) is a peripheral circadian-regulated gene product found in the liver, kidney, and testis where it functions as an mRNA deadenylase mediating mRNA degradation (Wang et al. 2001; Baggs and Green 2003; Douris and Green 2008; Kawai et al. 2010). Noc mediates post-transcriptional control of genes necessary for metabolic function including nutrient absorption, insulin sensitivity, lipid metabolism, adipogenesis, inflammation, and osteogenesis (Stubblefield et al. 2012; Green et al. 2007). The protein binds to the long-form 3' UTR of *IGF-I* mRNA, which contains potential regulatory motifs involved in *IGF-I* mRNA degradation (Kawai et al. 2010). Noc overexpression reduces both mRNA and protein levels of IGF-I indicating that Nocturnin enhances *IGF-I* mRNA degradation, thereby reducing both its mRNA and protein levels. Interestingly, Noc interacts with *IGF-I* mRNA in a strain- and tissue-specific manner in mice (Kawai et al. 2010).

3.3.3 Lin-28

The small RNA-binding proteins Lin-28 was originally identified as a developmental regulator in *Caenorhabidits elegans* and is conserved in multiple species (Moss et al. 1997; Moss and Tang 2003; Thornton et al. 2012). Lin-28 suppresses the biogenesis of let-7 by binding to let-7 precursor and facilitating its terminal uridylation and degradation (Thornton et al. 2012; Hagan et al. 2009; Heo et al. 2009, 2012). Lin-28 is up-regulated during myogenesis and promotes myocyte differentiation. In mouse myoblast C2C12 cells, Lin-28 binds to *IGF-II* mRNA and increases its translation via enhancing its association with RNA-containing translation initiation complexes including eIF3 β (Polesskaya et al. 2007).

3.3.4 IMPs

IGF-II mRNA binding proteins 1, 2, and 3 (IGF2BP1; IMP1, IGF2BP2; IMP2, IGFBP3; IMP3) belong to a conserved RNA binding protein family and function in various cellular processes, such as cell polarization, migration, proliferation, differentiation, embryogenesis and metabolism (Bell et al. 2013). IMPs bind to target mRNAs and regulate their expression post-transcriptionally via modulating mRNA stability, localization, and translation (Nielsen et al. 1999). IMPs associate at multiple sites with in the 5' UTR of IGF-II leader-3 mRNA and repress translation during embryonic development (Nielsen et al. 1999). In contrast, IMP3 functions as a translational activator of IGF-II leader-3 mRNA, thereby promoting cell survival and tumor progression in various cell types, including K562 leukemia cells and glioblastoma cells (Liao et al. 2005, 2011; Suvasini et al. 2011). Together these data suggest that IMPs may influence IGF-II expression in a tissue- or cell-type specific manner. IMP2 binds to an internal ribosome entry site of IGF-II mRNA 5' UTR and the binding of IMP2 is controlled by the mammalian target of rapamycin (mTOR) pathway (Dai et al. 2011). Phosphorylation of IMP2 on Ser162 and Ser164 residues by mTOR enhances the interaction between IMP2 and IGF-II mRNA, promoting translation of the latter (Fig. 2.5). IMP phosphorylation occurs in the mouse embryo as well as in adult tissues including islets of Langerhans, and seems to promote IGF2 expression, thereby regulating fetal growth and glucose homeostasis (Dai et al. 2011).



Fig. 2.5 A proposed model of IMP2 action on IGF-II biosynthesis

4 Conclusion

It is clear that the expression of insulin and IGFs is extensively regulated by post-transcriptional processes, including alternative splicing and mRNA stability and translation. Although the factors required for the regulation of INS and IGF mRNA alternative splicing are still largely unknown, the post-transcriptional regulation of INS and IGF mRNA is a complex process involving an array of different trans-acting factors such as RBPs and miRNAs. Here, we have described nearly all trans-acting factors that are known to directly bind to the UTRs of these mRNAs and modulate their expression. In addition, there are numerous factors, not described here, that indirectly regulate their expression at post-transcriptional levels that do not require direct binding to the UTRs. Through the cooperative regulation of both direct and indirect trans-acting factors, the expression of insulin and IGF expression is coordinately regulated. In response to acute changes in circulating glucose and other metabolic stimuli, the rate of insulin biosynthesis rises more dramatically within minutes by rapid increases in the stability and translation of the encoding mRNAs without de novo synthesis of INS mRNA (Brunstedt and Chan 1982; Itoh and Okamoto 1980). A long-term increase in insulin levels is dictated by both de novo synthesis and post-transcriptional regulation of INS mRNA. As described, although IGFs has insulin-like effects in glucose homeostasis, IGFs play a major role in the regulation of cellular processes such as growth, proliferation, and survival, which occur over longer periods. Thus, the IGF levels in these cases are likely changed by the long-term rather than the short-term regulation of the encoding mRNAs.

In addition to RBPs and miRNAs, long non-coding RNAs (lncRNAs), which are another type of post-transcriptional regulators and are non-protein coding transcripts longer than 200 nucleotides in their mature form, have also been linked to the insulin system. However, it is not known whether lncRNAs participate in the post-transcriptional modulation of *INS* and *IGF* mRNAs by direct binding to their UTRs. Further studies should therefore be performed to investigate whether and how lncRNAs, additional post-transcriptional modulators, or both increase or decrease insulin and IGF biosynthesis.

References

- Adam SA, Nakagawa T, Swanson MS, Woodruff TK, Dreyfuss G (1986) mRNA polyadenylatebinding protein: gene isolation and sequencing and identification of a ribonucleoprotein consensus sequence. Mol Cell Biol 6(8):2932–2943
- Adamo ML, Ben-Hur H, Roberts CT Jr, LeRoith D (1991) Regulation of start site usage in the leader exons of the rat insulin-like growth factor-I gene by development, fasting, and diabetes. Mol Endocrinol 5(11):1677–1686. doi:10.1210/mend-5-11-1677
- Anfinsen CB (1973) Principles that govern the folding of protein chains. Science 181(4096): 223–230
- Baggs JE, Green CB (2003) Nocturnin, a deadenylase in *Xenopus laevis* retina: a mechanism for posttranscriptional control of circadian-related mRNA. Curr Biol 13(3):189–198
- Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136(2):215–233. doi:10.1016/j.cell.2009.01.002
- Barton ER, Morris L, Musaro A, Rosenthal N, Sweeney HL (2002) Muscle-specific expression of insulin-like growth factor I counters muscle decline in mdx mice. J Cell Biol 157(1):137–148. doi:10.1083/jcb.200108071
- Bell GI, Pictet RL, Rutter WJ, Cordell B, Tischer E, Goodman HM (1980) Sequence of the human insulin gene. Nature 284(5751):26–32
- Bell JL, Wachter K, Muhleck B, Pazaitis N, Kohn M, Lederer M, Huttelmaier S (2013) Insulin-like growth factor 2 mRNA-binding proteins (IGF2BPs): post-transcriptional drivers of cancer progression? Cell Mol Life Sci 70(15):2657–2675. doi:10.1007/s00018-012-1186-z
- Brunstedt J, Chan SJ (1982) Direct effect of glucose on the preproinsulin mRNA level in isolated pancreatic islets. Biochem Biophys Res Commun 106(4):1383–1389
- Chekulaeva M, Hentze MW, Ephrussi A (2006) Bruno acts as a dual repressor of oskar translation, promoting mRNA oligomerization and formation of silencing particles. Cell 124(3):521–533. doi:10.1016/j.cell.2006.01.031
- Chew SL, Lavender P, Clark AJ, Ross RJ (1995) An alternatively spliced human insulin-like growth factor-I transcript with hepatic tissue expression that diverts away from the mitogenic IBE1 peptide. Endocrinology 136(5):1939–1944. doi:10.1210/endo.136.5.7720641
- Clemmons DR (2006) Involvement of insulin-like growth factor-I in the control of glucose homeostasis. Curr Opin Pharmacol 6(6):620–625. doi:10.1016/j.coph.2006.08.006
- Cohen DH, LeRoith D (2012) Obesity, type 2 diabetes, and cancer: the insulin and IGF connection. Endocr Relat Cancer 19(5):F27–F45. doi:10.1530/ERC-11-0374
- Cooke RM, Harvey TS, Campbell ID (1991) Solution structure of human insulin-like growth factor 1: a nuclear magnetic resonance and restrained molecular dynamics study. Biochemistry 30(22):5484–5491
- Cooper TA, Wan LL, Dreyfuss G (2009) RNA and disease. Cell 136(4):777–793. doi:10.1016/j. cell.2009.02.011
- Cote CA, Gautreau D, Denegre JM, Kress TL, Terry NA, Mowry KL (1999) A Xenopus protein related to hnRNP I has a role in cytoplasmic RNA localization. Mol Cell 4(3):431–437
- Dai N, Rapley J, Angel M, Yanik MF, Blower MD, Avruch J (2011) mTOR phosphorylates IMP2 to promote IGF2 mRNA translation by internal ribosomal entry. Genes Dev 25(11):1159–1172. doi:10.1101/Gad.2042311
- de Melo Neto OP, Standart N, Martins de Sa C (1995) Autoregulation of poly(A)-binding protein synthesis in vitro. Nucleic Acids Res 23(12):2198–2205
- Douris N, Green CB (2008) NOC out the fat: a short review of the circadian deadenylase nocturnin. Ann Med 40(8):622–626. doi:10.1080/07853890802084878
- Ekstrom TJ, Cui H, Li X, Ohlsson R (1995) Promoter-specific IGF2 imprinting status and its plasticity during human liver development. Development 121(2):309–316
- Elia L, Contu R, Quintavalle M, Varrone F, Chimenti C, Russo MA, Cimino V, De Marinis L, Frustaci A, Catalucci D, Condorelli G (2009) Reciprocal regulation of microRNA-1 and insulin-like growth factor-1 signal transduction cascade in cardiac and skeletal muscle in physiological and pathological conditions. Circulation 120(23):2377–2385. doi:10.1161/ CIRCULATIONAHA.109.879429

- Fenn AM, Smith KM, Lovett-Racke AE, Guerau-de-Arellano M, Whitacre CC, Godbout JP (2013) Increased micro-RNA 29b in the aged brain correlates with the reduction of insulin-like growth factor-1 and fractalkine ligand. Neurobiol Aging 34(12):2748–2758. doi:10.1016/j. neurobiolaging.2013.06.007
- Florini JR, Magri KA, Ewton DZ, James PL, Grindstaff K, Rotwein PS (1991) "Spontaneous" differentiation of skeletal myoblasts is dependent upon autocrine secretion of insulin-like growth factor-II. J Biol Chem 266(24):15917–15923
- Flynt AS, Lai EC (2008) Biological principles of microRNA-mediated regulation: shared themes amid diversity. Nat Rev Genet 9(11):831–842. doi:10.1038/Nrg2455
- Foyt HL, LeRoith D, Roberts CT Jr (1991) Differential association of insulin-like growth factor I mRNA variants with polysomes in vivo. J Biol Chem 266(11):7300–7305
- Garcia-Blanco MA, Jamison SF, Sharp PA (1989) Identification and purification of a 62,000-dalton protein that binds specifically to the polypyrimidine tract of introns. Genes Dev 3(12A):1874–1886
- Ge Y, Sun Y, Chen J (2011) IGF-II is regulated by microRNA-125b in skeletal myogenesis. J Cell Biol 192(1):69–81. doi:10.1083/jcb.201007165
- Gebeshuber CA, Martinez J (2013) miR-100 suppresses IGF2 and inhibits breast tumorigenesis by interfering with proliferation and survival signaling. Oncogene 32(27):3306–3310. doi:10.1038/ onc.2012.372
- Goldspink G, Yang SY (2004) The splicing of the IGF-I gene to yield different muscle growth factors. Adv Genet 52:23–49. doi:10.1016/S0065-2660(04)52002-3
- Green CB, Douris N, Kojima S, Strayer CA, Fogerty J, Lourim D, Keller SR, Besharse JC (2007) Loss of nocturnin, a circadian deadenylase, confers resistance to hepatic steatosis and diet-induced obesity. Proc Natl Acad Sci U S A 104(23):9888–9893. doi:10.1073/ pnas.0702448104
- Guay C, Regazzi R (2013) Circulating microRNAs as novel biomarkers for diabetes mellitus. Nat Rev Endocrinol 9(9):513–521. doi:10.1038/nrendo.2013.86
- Hagan JP, Piskounova E, Gregory RI (2009) Lin28 recruits the TUTase Zcchc11 to inhibit let-7 maturation in mouse embryonic stem cells. Nat Struct Mol Biol 16(10):1021–1025. doi:10.1038/nsmb.1676
- Hellen CU, Witherell GW, Schmid M, Shin SH, Pestova TV, Gil A, Wimmer E (1993) A cytoplasmic 57-kDa protein that is required for translation of picornavirus RNA by internal ribosomal entry is identical to the nuclear pyrimidine tract-binding protein. Proc Natl Acad Sci U S A 90(16):7642–7646
- Heo I, Joo C, Kim YK, Ha M, Yoon MJ, Cho J, Yeom KH, Han J, Kim VN (2009) TUT4 in concert with Lin28 suppresses microRNA biogenesis through pre-microRNA uridylation. Cell 138(4):696–708. doi:10.1016/j.cell.2009.08.002
- Heo I, Ha M, Lim J, Yoon MJ, Park JE, Kwon SC, Chang H, Kim VN (2012) Mono-uridylation of pre-microRNA as a key step in the biogenesis of group II let-7 microRNAs. Cell 151(3):521– 532. doi:10.1016/j.cell.2012.09.022
- Hepler JE, Van Wyk JJ, Lund PK (1990) Different half-lives of insulin-like growth factor I mRNAs that differ in length of 3' untranslated sequence. Endocrinology 127(3):1550–1552. doi:10.1210/endo-127-3-1550
- Hernandez-Sanchez C, Mansilla A, de la Rosa EJ, Pollerberg GE, Martinez-Salas E, de Pablo F (2003) Upstream AUGs in embryonic proinsulin mRNA control its low translation level. EMBO J 22(20):5582–5592. doi:10.1093/emboj/cdg515
- Hinman MN, Lou H (2008) Diverse molecular functions of Hu proteins. Cell Mol Life Sci 65(20):3168–3181. doi:10.1007/s00018-008-8252-6
- Hu W, Meng XY, Lu TC, Wu L, Li T, Li M, Tian YH (2013) MicroRNA-1 inhibits the proliferation of chinese sika deer-derived cartilage cells by binding to the 3'-untranslated region of IGF-1. Mol Med Rep 8(2):523–528. doi:10.3892/mmr.2013.1515
- Humbel RE (1990) Insulin-like growth factors I and II. Eur J Biochem/FEBS 190(3):445-462
- Itoh N, Okamoto H (1980) Translational control of proinsulin synthesis by glucose. Nature 283(5742):100-102

- Jones MR, Quinton LJ, Blahna MT, Neilson JR, Fu S, Ivanov AR, Wolf DA, Mizgerd JP (2009) Zcchc11-dependent uridylation of microRNA directs cytokine expression. Nat Cell Biol 11(9):1157–1163. doi:10.1038/ncb1931
- Jones MR, Blahna MT, Kozlowski E, Matsuura KY, Ferrari JD, Morris SA, Powers JT, Daley GQ, Quinton LJ, Mizgerd JP (2012) Zcchc11 uridylates mature miRNAs to enhance neonatal IGF-1 expression, growth, and survival. PLoS Genet 8(11):e1003105. doi:10.1371/journal. pgen.1003105
- Kawai M, Delany AM, Green CB, Adamo ML, Rosen CJ (2010) Nocturnin suppresses igf1 expression in bone by targeting the 3' untranslated region of igf1 mRNA. Endocrinology 151(10):4861–4870. doi:10.1210/en.2010-0407
- Keene JD (2007) RNA regulons: coordination of post-transcriptional events. Nat Rev Genet 8(7):533–543. doi:10.1038/nrg2111
- Kim W, Lee EK (2012) Post-transcriptional regulation in metabolic diseases. RNA Biol 9(6):772–780. doi:10.4161/rna.20091
- Kim W, Lee EK (2012) Post-transcriptional regulation in metabolic diseases. RNA Biol 9(6):772–780. doi:10.4161/Rna.20019
- Kim VN, Han J, Siomi MC (2009) Biogenesis of small RNAs in animals. Nat Rev Mol Cell Biol 10(2):126–139. doi:10.1038/nrm2632
- King GL, Kahn CR, Samuels B, Danho W, Bullesbach EE, Gattner HG (1982) Synthesis and characterization of molecular hybrids of insulin and insulin-like growth factor I. The role of the A-chain extension peptide. J Biol Chem 257(18):10869–10873
- Knoch KP, Bergert H, Borgonovo B, Saeger HD, Altkruger A, Verkade P, Solimena M (2004) Polypyrimidine tract-binding protein promotes insulin secretory granule biogenesis. Nat Cell Biol 6(3):207–214. doi:10.1038/ncb1099
- Knoch KP, Meisterfeld R, Kersting S, Bergert H, Altkruger A, Wegbrod C, Jager M, Saeger HD, Solimena M (2006) cAMP-dependent phosphorylation of PTB1 promotes the expression of insulin secretory granule proteins in beta cells. Cell Metab 3(2):123–134. doi:10.1016/j. cmet.2005.12.008
- Kulkarni SD, Muralidharan B, Panda AC, Bakthavachalu B, Vindu A, Seshadri V (2011) Glucosestimulated translation regulation of insulin by the 5' UTR-binding proteins. J Biol Chem 286(16):14146–14156. doi:10.1074/jbc.M110.190553
- Kwiecinski M, Elfimova N, Noetel A, Tox U, Steffen HM, Hacker U, Nischt R, Dienes HP, Odenthal M (2012) Expression of platelet-derived growth factor-C and insulin-like growth factor I in hepatic stellate cells is inhibited by miR-29. Lab Investig 92(7):978–987. doi:10.1038/ labinvest.2012.70
- Lee EK, Gorospe M (2010) Minireview: posttranscriptional regulation of the insulin and insulinlike growth factor systems. Endocrinology 151(4):1403–1408. doi:10.1210/en.2009-1123
- Lee EK, Kim W, Tominaga K, Martindale JL, Yang X, Subaran SS, Carlson OD, Mercken EM, Kulkarni RN, Akamatsu W, Okano H, Perrone-Bizzozero NI, de Cabo R, Egan JM, Gorospe M (2012) RNA-binding protein HuD controls insulin translation. Mol Cell 45(6):826–835. doi:10.1016/j.molcel.2012.01.016
- Leibiger B, Moede T, Schwarz T, Brown GR, Kohler M, Leibiger IB, Berggren PO (1998) Shortterm regulation of insulin gene transcription by glucose. Proc Natl Acad Sci U S A 95(16):9307–9312
- Liao B, Hu Y, Herrick DJ, Brewer G (2005) The RNA-binding protein IMP-3 is a translational activator of insulin-like growth factor II leader-3 mRNA during proliferation of human K562 leukemia cells. J Biol Chem 280(18):18517–18524. doi:10.1074/jbc.M500270200
- Liao B, Hu Y, Brewer G (2011) RNA-binding protein insulin-like growth factor mRNA-binding protein 3 (IMP-3) promotes cell survival via insulin-like growth factor II signaling after ionizing radiation. J Biol Chem 286(36):31145–31152. doi:10.1074/jbc.M111.263913
- Lim LP, Lau NC, Garrett-Engele P, Grimson A, Schelter JM, Castle J, Bartel DP, Linsley PS, Johnson JM (2005) Microarray analysis shows that some microRNAs downregulate large numbers of target mRNAs. Nature 433(7027):769–773. doi:10.1038/nature03315
- Lou H, Helfman DM, Gagel RF, Berget SM (1999) Polypyrimidine tract-binding protein positively regulates inclusion of an alternative 3'-terminal exon. Mol Cell Biol 19(1):78–85

- Lowe WL Jr, Adamo M, LeRoith D, Roberts CT Jr (1989) Expression and stability of insulin-like growth factor-I (IGF-I) mRNA splicing variants in the GH3 rat pituitary cell line. Biochem Biophys Res Commun 162(3):1174–1179
- Lukong KE, Chang KW, Khandjian EW, Richard S (2008) RNA-binding proteins in human genetic disease. Trends Genet 24(8):416–425. doi:10.1016/j.tig.2008.05.004
- Lunde BM, Moore C, Varani G (2007) RNA-binding proteins: modular design for efficient function. Nat Rev Mol Cell Biol 8(6):479–490. doi:10.1038/Nrm2178
- Matheny RW Jr, Nindl BC (2011) Loss of IGF-IEa or IGF-IEb impairs myogenic differentiation. Endocrinology 152(5):1923–1934. doi:10.1210/en.2010-1279
- Mathonnet G, Fabian MR, Svitkin YV, Parsyan A, Huck L, Murata T, Biffo S, Merrick WC, Darzynkiewicz E, Pillai RS, Filipowicz W, Duchaine TF, Sonenberg N (2007) MicroRNA inhibition of translation initiation in vitro by targeting the cap-binding complex eIF4F. Science 317(5845):1764–1767. doi:10.1126/science.1146067
- McCann AH, Miller N, O'Meara A, Pedersen I, Keogh K, Gorey T, Dervan PA (1996) Biallelic expression of the IGF2 gene in human breast disease. Hum Mol Genet 5(8):1123–1127
- Melo EO, Dhalia R, Martins de Sa C, Standart N, de Melo Neto OP (2003) Identification of a C-terminal poly(A)-binding protein (PABP)-PABP interaction domain: role in cooperative binding to poly (A) and efficient cap distal translational repression. J Biol Chem 278(47):46357–46368. doi:10.1074/jbc.M307624200
- Minn AH, Lan H, Rabaglia ME, Harlan DM, Peculis BA, Attie AD, Shalev A (2005) Increased insulin translation from an insulin splice-variant overexpressed in diabetes, obesity, and insulin resistance. Mol Endocrinol 19(3):794–803. doi:10.1210/me.2004-0119
- Mitchell P, Tollervey D (2000) mRNA stability in eukaryotes. Curr Opin Genet Dev 10(2):193–198. doi:10.1016/S0959-437x(00)00063-0
- Monk D, Sanches R, Arnaud P, Apostolidou S, Hills FA, Abu-Amero S, Murrell A, Friess H, Reik W, Stanier P, Constancia M, Moore GE (2006) Imprinting of IGF2 P0 transcript and novel alternatively spliced INS-IGF2 isoforms show differences between mouse and human. Hum Mol Genet 15(8):1259–1269. doi:10.1093/hmg/ddl041
- Moore MJ (2005) From birth to death: the complex lives of eukaryotic mRNAs. Science 309(5740):1514–1518. doi:10.1126/science.1111443
- Moreira A, Takagaki Y, Brackenridge S, Wollerton M, Manley JL, Proudfoot NJ (1998) The upstream sequence element of the C2 complement poly(A) signal activates mRNA 3' end formation by two distinct mechanisms. Genes Dev 12(16):2522–2534
- Moss EG, Tang L (2003) Conservation of the heterochronic regulator Lin-28, its developmental expression and microRNA complementary sites. Dev Biol 258(2):432–442
- Moss EG, Lee RC, Ambros V (1997) The cold shock domain protein LIN-28 controls developmental timing in *C. elegans* and is regulated by the lin-4 RNA. Cell 88(5):637–646
- Nielsen J, Christiansen J, Lykke-Andersen J, Johnsen AH, Wewer UM, Nielsen FC (1999) A family of insulin-like growth factor II mRNA-binding proteins represses translation in late development. Mol Cell Biol 19(2):1262–1270
- Oberer M, Marintchev A, Wagner G (2005) Structural basis for the enhancement of eIF4A helicase activity by eIF4G. Genes Dev 19(18):2212–2223. doi:10.1101/gad.1335305
- Ohlsson R, Hedborg F, Holmgren L, Walsh C, Ekstrom TJ (1994) Overlapping patterns of IGF2 and H19 expression during human development: biallelic IGF2 expression correlates with a lack of H19 expression. Development 120(2):361–368
- Orom UA, Nielsen FC, Lund AH (2008) MicroRNA-10a binds the 5'UTR of ribosomal protein mRNAs and enhances their translation. Mol Cell 30(4):460–471. doi:10.1016/j. molcel.2008.05.001
- Orphanides G, Reinberg D (2002) A unified theory of gene expression. Cell 108(4):439–451. doi:10.1016/S0092-8674(02)00655-4
- Panagakos FS (1993) Insulin-like growth factors-I and -II stimulate chemotaxis of osteoblasts isolated from fetal rat calvaria. Biochimie 75(11):991–994
- Panda AC, Kulkarni SD, Muralidharan B, Bakthavachalu B, Seshadri V (2010) Novel splice variant of mouse insulin2 mRNA: implications for insulin expression. FEBS Lett 584(6):1169– 1173. doi:10.1016/j.febslet.2010.02.020

- Panda AC, Grammatikakis I, Yoon JH, Abdelmohsen K (2013) Posttranscriptional regulation of insulin family ligands and receptors. Int J Mol Sci 14(9):19202–19229. doi:10.3390/ ijms140919202
- Panda AC, Sahu I, Kulkarni SD, Martindale JL, Abdelmohsen K, Vindu A, Joseph J, Gorospe M, Seshadri V (2014) miR-196b-mediated translation regulation of mouse insulin2 via the 5'UTR. PLoS One 9(7):e101084. doi:10.1371/journal.pone.0101084
- Pascale A, Amadio M, Quattrone A (2008) Defining a neuron: neuronal ELAV proteins. Cell Mol Life Sci 65(1):128–140. doi:10.1007/s00018-007-7017-y
- Pillai RS, Bhattacharyya SN, Filipowicz W (2007) Repression of protein synthesis by miRNAs: how many mechanisms? Trends Cell Biol 17(3):118–126. doi:10.1016/j.tcb.2006.12.007
- Polesskaya A, Cuvellier S, Naguibneva I, Duquet A, Moss EG, Harel-Bellan A (2007) Lin-28 binds IGF-2 mRNA and participates in skeletal myogenesis by increasing translation efficiency. Genes Dev 21(9):1125–1138. doi:10.1101/gad.415007
- Poy MN, Hausser J, Trajkovski M, Braun M, Collins S, Rorsman P, Zavolan M, Stoffel M (2009) miR-375 maintains normal pancreatic alpha- and beta-cell mass. Proc Natl Acad Sci U S A 106(14):5813–5818. doi:10.1073/pnas.0810550106
- Pravtcheva DD, Wise TL (1998) Metastasizing mammary carcinomas in H19 enhancers-Igf2 transgenic mice. J Exp Zool 281(1):43–57
- Pullmann R Jr, Kim HH, Abdelmohsen K, Lal A, Martindale JL, Yang X, Gorospe M (2007) Analysis of turnover and translation regulatory RNA-binding protein expression through binding to cognate mRNAs. Mol Cell Biol 27(18):6265–6278. doi:10.1128/MCB.00500-07
- Rajpathak SN, Gunter MJ, Wylie-Rosett J, Ho GY, Kaplan RC, Muzumdar R, Rohan TE, Strickler HD (2009) The role of insulin-like growth factor-I and its binding proteins in glucose homeostasis and type 2 diabetes. Diabetes Metab Res Rev 25(1):3–12. doi:10.1002/dmrr.919
- Rinderknecht E, Humbel RE (1978) The amino acid sequence of human insulin-like growth factor I and its structural homology with proinsulin. J Biol Chem 253(8):2769–2776
- Setyowati Karolina D, Sepramaniam S, Tan HZ, Armugam A, Jeyaseelan K (2013) miR-25 and miR-92a regulate insulin I biosynthesis in rats. RNA Biol 10(8):1365–1378. doi:10.4161/ rna.25557
- Shalev A, Blair PJ, Hoffmann SC, Hirshberg B, Peculis BA, Harlan DM (2002) A proinsulin gene splice variant with increased translation efficiency is expressed in human pancreatic islets. Endocrinology 143(7):2541–2547. doi:10.1210/endo.143.7.8920
- Shan ZX, Lin QX, Fu YH, Deng CY, Zhou ZL, Zhu JN, Liu XY, Zhang YY, Li YX, Lin SG, Yu XY (2009) Upregulated expression of miR-1/miR-206 in a rat model of myocardial infarction. Biochem Biophys Res Commun 381(4):597–601. doi:10.1016/j.bbrc.2009.02.097
- Smith PJ, Spurrell EL, Coakley J, Hinds CJ, Ross RJ, Krainer AR, Chew SL (2002) An exonic splicing enhancer in human IGF-I pre-mRNA mediates recognition of alternative exon 5 by the serine-arginine protein splicing factor-2/alternative splicing factor. Endocrinology 143(1):146– 154. doi:10.1210/endo.143.1.8598
- Sonenberg N, Dever TE (2003) Eukaryotic translation initiation factors and regulators. Curr Opin Struct Biol 13(1):56–63
- Steiner DF, Chan SJ, Welsh JM, Kwok SC (1985) Structure and evolution of the insulin gene. Annu Rev Genet 19:463–484. doi:10.1146/annurev.ge.19.120185.002335
- Stubblefield JJ, Terrien J, Green CB (2012) Nocturnin: at the crossroads of clocks and metabolism. Trends Endocrinol Metab 23(7):326–333. doi:10.1016/j.tem.2012.03.007
- Suvasini R, Shruti B, Thota B, Shinde SV, Friedmann-Morvinski D, Nawaz Z, Prasanna KV, Thennarasu K, Hegde AS, Arivazhagan A, Chandramouli BA, Santosh V, Somasundaram K (2011) Insulin growth factor-2 binding protein 3 (IGF2BP3) is a glioblastoma-specific marker that activates phosphatidylinositol 3-kinase/mitogen-activated protein kinase (PI3K/MAPK) pathways by modulating IGF-2. J Biol Chem 286(29):25882–25890. doi:10.1074/jbc. M110.178012
- Thornton JE, Chang HM, Piskounova E, Gregory RI (2012) Lin28-mediated control of let-7 microRNA expression by alternative TUTases Zcchc11 (TUT4) and Zcchc6 (TUT7). RNA 18(10):1875–1885. doi:10.1261/rna.034538.112

- Tillmar L, Welsh N (2002) Hypoxia may increase rat insulin mRNA levels by promoting binding of the polypyrimidine tract-binding protein (PTB) to the pyrimidine-rich insulin mRNA 3'-untranslated region. Mol Med 8(5):263–272
- Tillmar L, Carlsson C, Welsh N (2002) Control of insulin mRNA stability in rat pancreatic islets. Regulatory role of a 3'-untranslated region pyrimidine-rich sequence. J Biol Chem 277(2):1099– 1106. doi:10.1074/jbc.M108340200
- Valcarcel J, Gebauer F (1997) Post-transcriptional regulation: the dawn of PTB. Curr Biol 7(11):R705–R708
- Valencia-Sanchez MA, Liu J, Hannon GJ, Parker R (2006) Control of translation and mRNA degradation by miRNAs and siRNAs. Genes Dev 20(5):515–524. doi:10.1101/gad.1399806
- van Kouwenhove M, Kedde M, Agami R (2011) MicroRNA regulation by RNA-binding proteins and its implications for cancer. Nat Rev Cancer 11(9):644–656. doi:10.1038/nrc3107
- Vasudevan S, Tong Y, Steitz JA (2007) Switching from repression to activation: microRNAs can up-regulate translation. Science 318(5858):1931–1934. doi:10.1126/science.1149460
- Velloso CP, Harridge SD (2010) Insulin-like growth factor-I E peptides: implications for aging skeletal muscle. Scand J Med Sci Sports 20(1):20–27. doi:10.1111/j.1600-0838.2009.00997.x
- Wagner EJ, Garcia-Blanco MA (2001) Polypyrimidine tract binding protein antagonizes exon definition. Mol Cell Biol 21(10):3281–3288. doi:10.1128/Mcb.21.10.3281-3288.2001
- Wang Y, Osterbur DL, Megaw PL, Tosini G, Fukuhara C, Green CB, Besharse JC (2001) Rhythmic expression of nocturnin mRNA in multiple tissues of the mouse. BMC Dev Biol 1:9
- Wang XH, Qian RZ, Zhang W, Chen SF, Jin HM, Hu RM (2009) MicroRNA-320 expression in myocardial microvascular endothelial cells and its relationship with insulin-like growth factor-1 in type 2 diabetic rats. Clin Exp Pharmacol Physiol 36(2):181–188. doi:10.1111/ j.1440-1681.2008.05057.x
- Weller PA, Dickson MC, Huskisson NS, Dauncey MJ, Buttery PJ, Gilmour RS (1993) The porcine insulin-like growth factor-I gene: characterization and expression of alternate transcription sites. J Mol Endocrinol 11(2):201–211
- Welsh M, Nielsen DA, MacKrell AJ, Steiner DF (1985) Control of insulin gene expression in pancreatic beta-cells and in an insulin-producing cell line, RIN-5F cells. II. Regulation of insulin mRNA stability. J Biol Chem 260(25):13590–13594
- Wicksteed B, Herbert TP, Alarcon C, Lingohr MK, Moss LG, Rhodes CJ (2001) Cooperativity between the preproinsulin mRNA untranslated regions is necessary for glucose-stimulated translation. J Biol Chem 276(25):22553–22558. doi:10.1074/jbc.M011214200
- Xie J, Lee JA, Kress TL, Mowry KL, Black DL (2003) Protein kinase A phosphorylation modulates transport of the polypyrimidine tract-binding protein. Proc Natl Acad Sci U S A 100(15):8776–8781. doi:10.1073/pnas.1432696100
- Yang H, Adamo ML, Koval AP, McGuinness MC, Ben-Hur H, Yang Y, LeRoith D, Roberts CT Jr (1995) Alternative leader sequences in insulin-like growth factor I mRNAs modulate translational efficiency and encode multiple signal peptides. Mol Endocrinol 9(10):1380–1395. doi:10.1210/mend.9.10.8544846
- Yekta S, Shih IH, Bartel DP (2004) MicroRNA-directed cleavage of HOXB8 mRNA. Science 304(5670):594–596. doi:10.1126/science.1097434
- Yu XY, Song YH, Geng YJ, Lin QX, Shan ZX, Lin SG, Li Y (2008) Glucose induces apoptosis of cardiomyocytes via microRNA-1 and IGF-1. Biochem Biophys Res Commun 376(3):548–552. doi:10.1016/j.bbrc.2008.09.025
- Zhang J, Chrysis D, Underwood LE (1998) Reduction of hepatic insulin-like growth factor I (IGF-I) messenger ribonucleic acid (mRNA) during fasting is associated with diminished splicing of IGF-I pre-mRNA and decreased stability of cytoplasmic IGF-I mRNA. Endocrinology 139(11):4523–4530. doi:10.1210/endo.139.11.6321
- Zheng WH, Kar S, Dore S, Quirion R (2000) Insulin-like growth factor-1 (IGF-1): a neuroprotective trophic factor acting via the Akt kinase pathway. J Neural Transm Suppl 60:261–272
- Zhou X, Duan X, Qian J, Li F (2009) Abundant conserved microRNA target sites in the 5'-untranslated region and coding sequence. Genetica 137(2):159–164. doi:10.1007/s10709-009-9378-7



3

Post-transcriptional Regulation of Cytokine Signaling During Inflammatory Responses

Irina Vlasova-St. Louis and Paul R. Bohjanen

1 Introduction

Cytokines are important mediators of cell to cell communication, controlling a variety of cellular activities such as inflammation, growth, differentiation, cell proliferation and apoptosis. In the immune system, cytokines play essential roles in native immunity, proper immune effector function and maintenance of peripheral tolerance. In addition, cytokines can act as hormones, regulating proper growth, cellular differentiation, and appropriate programmed cell death to help coordinate the gene expression pathways during development and maturation of multicellular organisms. Given the complexity and crosstalk between pathways involved in cytokine signaling, dysregulated cytokine networks are often associated with human diseases. For example, autoimmune diseases such as diabetes involve abnormal production of inflammatory cytokines, which lead to destruction of beta cells in the pancreas. Consequently, many of the medications used to treat autoimmune and inflammatory diseases target cytokines or kinase signaling downstream of cytokine

P.R. Bohjanen, MD, PhD

The original version of this chapter was revised. The correction to this chapter is available at $https://doi.org/10.1007/978-3-319-25124-0_{15}$

I. Vlasova-St. Louis, MD, PhD ()

Department of Medicine, Center for Infectious Diseases and Microbiology Translational Research, University of Minnesota, McGuire Translational Research Facility, 2001 6th Street SE, Office 3-500B, Minneapolis, MN 55455, USA e-mail: vlaso001@umn.edu

Department of Medicine, Center for Infectious Diseases and Microbiology Translational Research, University of Minnesota, McGuire Translational Research Facility, 2001 6th Street SE, Office 3-500B, Minneapolis, MN 55455, USA

Department of Microbiology, University of Minnesota, Minneapolis, MN 55455, USA e-mail: bohja001@umn.edu

[©] Springer International Publishing Switzerland 2016, Corrected Publication 2020 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_3

receptors (reviewed in Beyaert et al. 2013; Kontzias et al. 2012; Nepom et al. 2013). Cytokine and growth factor dysregulation leading to uncontrolled cell growth is an important mechanism of cancer pathogenesis (Saharinen et al. 2011). Thus, it is crucial that cytokine signaling pathways are tightly regulated through multiple mechanisms, including transcriptional and post-transcriptional mechanisms (reviewed in Duan et al. 2013; Friedel et al. 2009; Ivanov and Anderson 2013; Seko et al. 2006).

Here, we review post-transcriptional regulation of cytokines and cytokine signaling, focusing on the role of AU-rich elements (AREs) and GU-rich elements (GREs) in coordinating signals transmitted by cytokines during inflammatory responses. AREs and GREs are conserved sequences in the 3' untranslated region (UTR) of short-lived transcripts that encode a variety of proteins involved in cellular activation, growth and apoptosis, including components of cytokine signaling pathways. AREs function as post-transcriptional regulators of a variety of cytokines, and the mechanisms by which AREs mediate mRNA degradation have been studied extensively (reviewed in Anderson 2010). Current understanding of AREs has helped to explain the molecular mechanisms of post-transcriptional regulation of cytokine production (Anderson 2008) and has helped us to understand how networks of genes are coordinately regulated by conserved sequences in mRNA. In contrast, our knowledge about the role of GREs in coordinating gene expression is not as well understood. We discuss the role of GREs in regulating cytokine signaling networks, and suggest that AREs and GREs regulate distinct sets of transcripts but function in concert to coordinate cytokine expression with cytokine signaling.

2 Regulation by AREs

During inflammatory responses, cytokines and other early response gene proteins are induced transiently for a defined period of time through a preset program of gene expression events that include transcriptional induction, transcriptional repression, and regulated mRNA stability. For example, T cell receptor stimulation of human T lymphocytes induces ARE-containing cytokine gene expression, such as IL2, IFN-gamma, IL4, TNF-alpha, etc., through transcriptional pulses (Hao and Baltimore 2009; Raghavan et al. 2004). This transcriptional induction is followed by transcript degradation mediated by AREs found in the 3'UTRs of these transcripts (reviewed in Al-Haj and Khabar 2012; Khabar 2007; Schott and Stoecklin 2010). Figure 3.1 shows the intricate network of cytokines that are regulated by AREs in a variety of immune cell types. AREs function as instability elements that mediate cytokine transcript degradation by interacting with ARE-binding proteins that recruit the cellular mRNA degradation machinery to the transcript. Along with regulating mRNA decay, ARE-binding proteins control additional post-transcriptional events such as pre-mRNA processing, transport, intracellular localization, and translation (reviewed in Beisang and Bohjanen 2012; Raghavan and Bohjanen 2004; Stumpo et al. 2010).



Fig. 3.1 The role of cytokines in mediating communication between immune cells. Transcripts in *grey* are cytokine transcripts that contain AREs in their 3'UTRs. Transcripts labeled with an *asterisk* (*) contain GREs in their 3'UTRs. *Arrows* indicate direct interactions and/or activations. *Bluntended lines* indicate inhibitory effects. Reproduced from Vlasova-St. Louis and Bohjanen (2014)

A variety of ARE-containing transcripts interact with many different AREbinding proteins, resulting in complex patterns of post-transcriptional regulation. The destabilizing functions of different AREs, and their ability to interact with different ARE-binding proteins are not equivalent. The sequence characteristics and decay patterns of different AREs allowed them to be categorized into classes based on their sequence features and decay kinetics (Chen and Shyu 1995). A more extensive database of ARE-containing transcripts from mouse, rat and man has been generated using bioinformatics (Bakheet et al. 2003, 2006; Halees et al. 2008). This database groups AREs into five clusters depending on the number of overlapping AUUUA pentamers within the 3'UTR of a transcript with clusters 1–5 having five to one overlapping AUUUA pentamers, respectively. Cluster I AREs are enriched in secreted proteins, such as cytokines, and are involved in the growth of hematopoietic and immune cells. The other ARE clusters are found within a diverse set of transcripts. In total, ARE-containing transcripts compose approximately 5-8 % of the human transcriptome (Bakheet et al. 2001). AREs play decisive roles in regulating the effects of cytokines on inflammatory responses since mutation of the ARE in cytokine genes such as TNF-alpha or IFN-gamma resulted in profound autoimmune-like inflammatory syndrome (Hodge et al. 2014; Jacob et al. 1991).

AREs regulate mRNA decay by interacting with ARE-binding proteins that can either function to stabilize or destabilize the bound transcript. Numerous ARE-binding proteins involved in the regulation of mRNA turnover have been described in a variety of cell types (reviewed in Gratacos and Brewer 2010; Li et al. 2012; Von

Roretz et al. 2011). Typically, mRNA molecules move through different cellular compartments within messenger ribonucleoprotein (mRNP) complexes, dynamically associating with RNA-binding proteins (RBPs) that bind to conserved ciselements found in subsets of transcripts (Turner and Hodson 2012a, b). The association of specific RBPs with subsets of transcripts containing conserved regulatory cis-elements coordinates the fate of these bound transcripts through posttranscriptional processes such as translation, intracellular localization, storage or mRNA decay (reviewed in Blackinton and Keene 2014; Keene 2007). Following cellular activation, the set of transcripts bound by a given RBP can be used to define post-transcriptional regulatory networks. RNA-immunoprecipitation (RNA-IP) experiments, in which RNA-binding proteins are immunoprecipitated from cellular lysates using an antibody against a given RBP, and co-purified mRNAs are identified, provided evidence that such networks of coordinately regulated RNAs exist. Target transcripts of several ARE-binding proteins, including HNRNPD (Wu et al. 2013), ZFP36 (Emmons et al. 2008; Stoecklin et al. 2008), ELAVL1 (Fan et al. 2011; Lopez De Silanes et al. 2004b; Mukherjee et al. 2009), KSRP (Winzen et al. 2007) and TIA-1 (Lopez De Silanes et al. 2005) were identified using RNA-IP techniques, and these targets represent distinct but overlapping sets of transcripts. AREbinding protein targets include transcripts encoding cytokines and cytokine signaling components. Integration of the effects of multiple ARE-binding proteins likely decides the fate of ARE-containing transcripts (Mansfield and Keene 2009).

Several ARE-binding proteins, including ZFP36 (also known as TTP), BRF1, BRF2, KSRP, and HNRNPD (AUF1) destabilize transcripts by recruiting cellular deadenylases and enzymes involved in both $5' \rightarrow 3'$ and $3' \rightarrow 5'$ mRNA decay processes, whereas other ARE-binding proteins, such as ELAVL1 (also known as HuR) and ELAVL4, stabilize target transcripts by preventing deadenylation and decay (Bronicki and Jasmin 2013; Brooks and Blackshear 2013; Khabar 2010; Sarkar et al. 2011; White et al. 2013; Wu et al. 2013). ARE-containing mRNA can also be degraded by ribonucleases, such as ZC3H12A, which initiates decay by cleaving the stem loop mRNA structure in the near proximity of an ARE. Examples of ZC3H12A targets are TNF-alpha, IL1b, IL6 and IL12b, IL2, c-Rel and Ox40 transcripts (Matsushita et al. 2009; Uehata et al. 2013), which are important regulators of lymphocyte activation. These same ARE-containing transcripts are also targets of ELAVL1 (Fan et al. 2011; Lopez De Silanes et al. 2004b; Mukherjee et al. 2009). ELAVL1 has been reported to stabilize transcripts of multiple proinflammatory cytokines including, TNF-alpha, VEGF, IL13, IL17 COX2, GATA3 and a number of chemokines through AREs in their 3'UTRs, often leading to increases in both mRNA and protein levels (Stellato et al. 2011; Winzen et al. 2004). The stabilization of ARE-containing transcripts does not always lead to increased protein expression, however. For example when ELAVL1 is in a complex with TIA-1 protein, TNFalpha and COX2 mRNAs were stabilized but their translation did not increase (Katsanou et al. 2005).

ARE-containing transcripts can interact with different ARE-binding proteins during the course of an inflammatory response through competitive binding. For example, the ARE-binding proteins ELAVL1 and ZFP36 compete with one another
for certain ARE-containing transcripts (Raghavan et al. 2001). More than 50 % of ZFP36 target sites in 3'UTRs also function as ELAVL1 target sites (Mukherjee et al. 2014). This potential for ZFP36 and ELAVL1 to compete for the same binding site on a single mRNA has implications for dynamic changes in binding during the course of adaptive immune responses. For example, shortly following T lymphocyte activation, cytoplasmic ELAVL1 levels increase transiently, stabilizing a network of ARE-containing transcripts and allowing their increased expression. Later in T cell activation, ZFP36 induction promotes the displacement of ELAVL1 from AREcontaining transcripts, allowing ZFP36 to mediate their rapid decay (Raghavan et al. 2001). This process allows the transient expression and subsequent degradation of ARE-containing transcripts during the course of immune responses. Cellular signals can also alter the functions of ARE-binding proteins over the course of innate immune responses. ZFP36 activity is regulated through its phosphorylation by p38 MAPK-activated protein kinase 2 (p38/MK2), following lipopolysaccharide (LPS) stimulation of macrophages. This phosphorylation of ZFP36 promotes its association with 14-3-3 protein and prevents ZFP36 from recruiting 3' deadenylases to the bound transcript (Sun et al. 2007). Thus, through LPS-mediated phosphorylation of ZFP36, ARE-containing transcripts encoding inflammatory mediators such as CCL3, IL1, IL6, IL17, and COX2 are induced by virtue of transcript stabilization and thereby promote inflammation (Kang et al. 2011; Lee et al. 2012; Ronkina et al. 2010). As the immune response resolves, ZFP36 is dephosphorylated by the phosphatase PP2A, allowing ZFP36 to return to its baseline function, promoting the degradation of transcripts encoding these pro-inflammatory mediators (Frasca et al. 2010; Sun et al. 2007).

Understanding the role of AREs in balancing the expression of pro-inflammatory cytokines during inflammatory responses has led to the development of new anti-inflammatory therapies. For example, the novel drug, aganirsen, functions as an anti-inflammatory agent by impairing 14-3-3b-ZFP36 complex formation, prevent-ing the stabilization and upregulation of ARE-containing cytokines, including IL8, TNF-alpha, IL21-beta, IL12, and IL22, during inflammatory responses (Colin et al. 2014). Thus, understanding ARE-mediated regulation of cytokine gene expression may lead to the further development of anti-inflammatory agents. In summary, the coordinate regulation of networks of ARE-containing transcripts, including cytokine networks, is a dynamic process following immune cell activation, which involves activation-induced expression of ARE-binding proteins, competition between ARE-binding proteins with opposing activities, and multiple phosphorylation events mediated through signaling pathways.

3 Regulation by GREs

Studies regarding AREs and ARE-binding proteins have led to identification of other cis-elements and proteins that coordinately regulate mRNA decay (Vlasova and Bohjanen 2008). For example, bioinformatic analysis of short-lived transcripts expressed in primary human T cells led to the identification of the sequence,

UGUUUGUUUGU (known as a GRE), to be highly enriched in the 3'UTR of transcripts that exhibited rapid degradation (Vlasova et al. 2008). Introduction of the GRE into the 3'UTR of a beta-globin reporter transcript induced rapid decay of the otherwise stable beta-globin transcript, demonstrating that the GRE functions as a mediator of mRNA degradation. The GRE binds to the protein CELF1, leading to the decay of GRE-containing transcripts. Knockdown of CELF1 in HeLa cells led to stabilization of GRE-containing reporter transcripts, further implicating CELF1 as a mediator of GRE-dependent mRNA degradation (Rattenbacher et al. 2010; Vlasova et al. 2008). A database of GRE-containing transcripts (Halees et al. 2011) was generated by categorizing GREs into five clusters based on the number of overlapping GUUUG pentamers found in the 3'UTR of a transcript, with clusters 1-5 having five to one overlapping GUUUG pentamers, respectively. These GREcontaining transcripts encode a variety of proteins with diverse biological functions including cellular signaling, growth, development and apoptosis regulation. Our recent analysis of CELF1 revealed that targets contained not only the GRE sequence UGUUUGUUUGU but also a GU-repeat sequence which could bind CELF1 and mediate mRNA decay. Based on these findings, the GRE sequence definition was revised to UGU[G/U]UGU[G/U]UGU (Rattenbacher et al. 2010).

CELF1 binds numerous target transcripts involved in cytokine signaling pathways (Vlasova-St. Louis and Bohjanen 2011, 2014; Vlasova-St. Louis et al. 2013). Genome-wide RNA-IP experiments in HeLa cells (Rattenbacher et al. 2010), primary human T cells (Beisang et al. 2012), and mouse myoblasts (Lee et al. 2010), identified hundreds of CELF1 target transcripts. Comparison of molecular functions of the CELF1 bound transcripts among the different cell types revealed enrichment of mRNAs encoding regulators of transcription, post-transcriptional control and cell cycle. In primary human resting T cells, CELF1 binds to and mediates the degradation of numerous transcripts involved in cellular activation and proliferation, presumably to maintain the cell in a quiescent state. Rapid changes in the expression of these GRE-containing transcripts occur immediately following activation of T lymphocytes. T cell receptor-mediated activation promotes the phosphorylation of CELF1, inhibiting its ability to bind to GRE-containing transcripts (Beisang et al. 2012). The lack of CELF1 binding to mRNA within first 24 h following T cell stimulation correlates with a transient increase in expression of GRE-containing transcripts involved in cellular activation and proliferation. Thus, CELF1 functions in resting T cells to mediate the degradation of GRE-containing transcripts that promote cell proliferation, and subsequent CELF1 phosphorylation following T cell activation prevents CELF1 binding to mRNA, leading to the stabilization and upregulation of GRE-containing transcripts (Vlasova-St. Louis and Bohjanen 2014).

Post-transcriptional networks are instrumental for cytokines to rapidly transmit intercellular signals to maintain homeostasis and direct the normal course of inflammatory responses. Post-transcriptional regulation through AREs impacts cytokine production by affecting the stability of cytokine transcripts. Numerous components of cytokine signaling, however, are regulated through GREs. Figures 3.2, 3.3, and 3.4 depict transcripts encoding signaling components downstream of various



Fig. 3.2 The role of GREs and CELF1 in TNF, TLR or IL1 receptor signaling. Transcripts shown exhibited changes in steady-state levels following T cell receptor stimulation of primary human T cells (Raghavan et al. 2004). Transcripts *in grey* contain GREs. Transcripts with *underlined text* were identified as CELF1 target transcripts in human T cells by RNA-immunoprecipitation (Beisang et al. 2012). This figure is modified from (Vlasova-St Louis and Bohjanen 2014)

cytokine receptors whose steady state mRNA levels changed following immune activation. As can be seen, GRE-containing transcripts (grey transcripts) encode components of signaling pathways downstream from these receptors. Overall, these receptors transduce a broad range of intracellular signals that shift cell activation states, alter the rate of proliferation and production of other cytokines. These pathway analyses indicate that the GRE/CELF1 network plays important roles in regulating the dynamic expression of cytokine signaling components through mRNA degradation and thereby modulates the strength and duration of inflammatory responses.



Fig. 3.3 The role of GREs and CELF1 in interferon and interleukin receptor signaling. Transcripts shown exhibited changes in steady-state levels following T cell receptor stimulation of primary human T cells (Raghavan et al. 2004). Transcripts in grey contain GREs. Transcripts with underlined text were identified as CELF1 target transcripts in human T cells by RNA-immunoprecipitation (Beisang et al. 2012). Transcripts marked with text in italics are also bound to ELAVL1 in activated cells (Mukherjee et al. 2009; Fan et al. 2011). This figure is modified from Vlasova-St. Louis and Bohjanen (2014)

4 Cross-Talk between Networks

The network of ARE-containing transcripts encodes various cytokines (Fig. 3.1) and cytokine signaling molecules that depends on circuits of GRE-containing transcripts to function. In particular, GRE-containing transcripts encode diverse components of cytokine receptor signaling (Figs. 3.2, 3.3, and 3.4). Moreover, numerous GRE-containing CELF1 target transcripts are also targets of the ARE-binding protein ELAVL1, indicating a functional interaction and potential competition between these RNA-binding proteins. RNA recognition sequences for CELF1 require precise GU repeats or GUUU repeats, whereas the recognition sequence for ELAVL1 is less precise, and ELAVL1 binds to a variety of U-rich sequences, including GU-rich sequences or a poly U sequence. Thus, CELF1 and ELAVL1, other ARE-binding proteins that recognize precise AUUU repeats, such as ZFP36, do not bind to GRE sequences and do not compete for binding sites. The comparison of CELF1 targets that were immunoprecipitated from normal T cells (Beisang et al. 2012) with ELAVL1 targets from HeLa (Lopez De Silanes et al. 2004a) or Jurkat cells



Fig. 3.4 The role of GREs in growth factor signaling. Signals transmitted through tumor growth factor receptors (TGFR), vascular endothelial, epidermal, platelet-derived or insulin growth factor receptors (VEGFR, EGFR, IGFR, PTDGFR) are directed through several major pathways, such as SMAD, mTOR, RAS, PLC, or JAK/STAT. Proper expressions of these transcripts cooperatively contribute to the overall cellular signaling outcome (such as cell proliferation, cell survival and cell growth). Transcripts in *grey* contain GREs. Transcripts with *underlined text* were identified as CELF1 target transcripts in resting human T cells by RNA-immunoprecipitation (Beisang et al. 2012). Transcripts marked with *text in italics* are also bind ELAVL1 in activated cells (Mukherjee et al. 2009; Fan et al. 2011). Arrows indicate direct interactions and/or activations. Blunt-ended lines indicate inhibitory effects. This network diagram was built using Ingenuity Pathway Assistant Software

(Mukherjee et al. 2009), showed a subset of transcripts involved in cytokine signaling that are targets of both proteins (Vlasova-St. Louis and Bohjanen 2014). Depending on which protein competitively binds to the GRE in the target transcript, dichotomous biochemical effects may result in either CELF1 mediating degradation or ELAVL1 mediating stabilization. We hypothesize that in resting cells these transcripts are bound to CELF1 and targeted for degradation, but in activated immune cells, CELF1 becomes phosphorylated and is rendered unable to bind to GRE sequences. ELAVL1 then binds to the GRE, displacing CELF1, stabilizing the transcripts, and allowing them to be translated. Later in the immune response, CELF1 becomes dephosphorylated and then displaces ELAVL1 to mediate transcript degradation. Although further evidence is needed, this model could explain the transient expression of certain GRE-containing transcripts following T cell activation.

The dynamic interplay of GRE and ARE networks appears to regulate nearly all known cytokine signaling pathways. For example, signals downstream of TNFalpha receptors interact with signals downstream of FAS (also TNFRSF6) to regulate apoptosis. CELF1 regulates expression of a number of apoptotic proteins downstream of FAS receptor through destabilization of pro-apoptotic mRNAs, simultaneously affecting cell death rate and proliferation. CELF1 has been shown to co-immunoprecipitate with transcripts encoding TNF receptor super family members and other molecules downstream of TNF receptors (underlined transcripts in Fig. 3.2) (Beisang et al. 2012; Lee et al. 2010; Rattenbacher et al. 2010). Interestingly, CELF1 and ELAVL1 both bind to and affect the stability of TNF mRNA (Dean et al. 2001; Zhang et al. 2008) and other regulators including effector caspases and cytochrome C mRNA, which are activated downstream of TNF receptors involved in the regulation of apoptosis. Both CELF1 and ELAVL1 proteins regulate stability of transcripts encoding members of the BCL2 superfamily such as BCL2, BNIP3, BCL2L2, BAD and BAX, which modulate programmed cell death (Abdelmohsen et al. 2007; Beisang et al. 2012; Talwar et al. 2013). Thus, signaling through the TNF receptors is controlled by GRE-containing transcripts (also CELF1 targets), which cooperate with ELAVL1 target transcripts to regulate apoptosis.

As shown in Fig. 3.2, multiple GRE-containing transcripts encode signaling components downstream of the IL1 receptor. Proper signaling through this cytokine receptor requires coordination of ARE and GRE networks. Several IL1 receptor signaling components, including STK4, CDC42, MYD88, TIFA, MEF2, RSK, cyclin D have been shown to be ELAVL1 targets in RNA-IP experiments (Mukherjee et al. 2009). Many target transcripts, shared by CELF1 and ELAVL1, are transcription factors or early response genes such as RUNX1, JUN, myc or CREB that are transiently up-regulated following cellular activation and then down-regulated through rapid mRNA decay. Chronic imbalances in the turnover rates of these transcripts could promote onset and progression in diseases such as autoimmunity or cancer through aberrant immune cell activation, excess inflammation and inappropriate cell death (Candido and Hagemann 2013; Moudgil and Choubey 2011; Vlasova et al. 2005).

Many other interleukin and interferon receptors also utilize signaling proteins encoded by GRE-containing transcripts (Fig. 3.3), and rely on interplay between ARE and GRE networks to maintain homeostasis. Cytokines that bind to receptors containing the common gamma chain, such as IL2, IL4, IL7, IL9, IL15 and IL21, are pivotal for immune responses (Baker et al. 2007; Liao et al. 2011). These receptors activate Janus kinases (JAK1 and JAK2) to phosphorylate members of the STAT (Signal transducers and activators of transcription) family of transcription factors (Jenkins 2014). STAT proteins form hetero- or homodimers, and upon translocation into nucleus they induce the transcription of genes including cytokine, chemokine and adhesion molecule genes. One of the induced gene family, SOCS, plays a role in inhibiting JAK kinase activity and consequently, down-regulates immune

activation (Linossi et al. 2013). Interestingly, SOCS3 mRNA is also a CELF1 and ELAVL1 target, perhaps to allow its expression only at the appropriate time during the resolution phase of the immune response. Engagement of interferons by interferon receptors also activates JAK-STAT, PI3K and MAPK signaling pathways (Khabar and Young 2007). Interferon family members of type I, II, and III contain functional AREs that allow precise post-transcriptional control by ARE-binding proteins. Interferon receptor mRNAs contain GREs and likely function to maintain the balance in cytokine-receptor signaling and ultimately the course of immune responses (Fig. 3.3). Thus, the induction and control of interferon responses requires coordination of ARE and GRE networks.

Growth factor-induced cell proliferation, adhesion, and migration often depend on ARE and GRE post-transcriptional regulatory networks. Transforming growth factor-beta (TGF-beta) signals are transduced by trans-membrane type I and type II serine/threonine kinase receptors (TGFR1/2), which are encoded by GRE- and ARE-containing mRNAs (Fig. 3.4). TGF-beta maintains tissue homeostasis by regulating the cellular proliferation rate, differentiation and survival. The activated TGF receptor complex induces oligomerization of several downstream SMAD protein family members. Activated SMADs regulate transcription of a number of GREcontaining mRNAs (ex. CITED2, CXCL2, NKX, CDKN1A, VEGF, IL3RA, TGFBR1, etc.), which in turn, become a subject of post-transcriptional regulation by CELF1 (or ELAVL1), once they leave the nucleus. In pathological conditions, TGF-beta signaling, by activating RAS oncogenic pathway, upregulates a large set of ARE-containing genes and specifically VEGF mRNA (Kanies et al. 2008), perhaps due in part to dysfunctional mRNA decay pathways. The RAS-PI3K-Akt kinase-mTOR pathway is a well-established signaling axis that modulates the proliferation and survival of many cell types through growth factor receptor signaling (reviewed elsewhere Bitterman and Polunovsky 2012; Cao et al. 2008; Weichhart and Saemann 2008). Growth factors are beneficial for tissue growth and regeneration, but abnormal growth factor signaling can fuel inflammation and metastases (Carmeliet and Jain 2011). Since ARE and GRE-containing transcripts encode important regulators of apoptosis and cell cycle downstream of growth factor receptors, transformed cells may usurp these post-transcriptional networks to their advantage as they become malignant (Fuxe and Karlsson 2012).

Numerous studies describe an important role of ELAVL1 in the pathogenesis of inflammatory diseases that involve over-production of inflammatory cytokines or diseases characterized by cytokine deficiency. Most studies support a role for ELAVL1 malfunction in the promotion of inflammation and proliferation in human diseases such as autoimmune diseases and cancer (Khabar 2010; Srikantan and Gorospe 2012). Furthermore, genetic deletion of ELAVL1 in thymocytes (Papadaki et al. 2009) and myelocytes (Yiakouvaki et al. 2012) predisposed mice to exaggerated inflammatory responses and inflammation-driven oncogenesis. In contrast, CELF1 functions as an inhibitor of a network of transcripts that promote cellular activation and proliferation (Beisang and Bohjanen 2012). The opposing effects of CELF1 and ELAVL1 may have important implications for new therapies for proliferative diseases such as cancer or autoimmunity. Thus, understanding the crosstalk

of CELF1, ELAVL1 and other RBPs may uncover novel therapeutic strategies to fine-tune the balance in proliferative or proinflammatory pathways involved in human diseases. In the future, more work needs to be done to develop interventions targeting post-transcriptional mechanisms involved in inflammatory responses such as the severe reaction that occurs in sepsis, which involves self-amplifying networks of inflammatory mediators often referred to as 'cytokine storm'.

5 Summary and Perspective

Post-transcriptional networks defined by AREs and GREs regulate cytokine production and signaling. The ARE and GRE networks represent distinct subsets of transcripts that work in concert to coordinate the function of cytokines over the course of an immune response. AREs are critical regulators of cytokine production and play important roles in cytokine signaling, but a distinct subset of cytokine signaling components, regulated by GREs, is needed for precise physiological function of cytokines. Thus, an effective immune response requires crosstalk between ARE and GRE pathways to appropriately regulate cytokine responses over time. Perhaps, integrative studies of interactions between GREs, AREs and RBPs as modulators of coordinated gene expression could be translated to novel treatment strategies.

Acknowledgments This work was supported by National Institutes of Health grants A1057484 and A1072068 to P.R.B. We would like to thank Eric Collins, for critical reading and scientific editing of the manuscript. Our acknowledgements to University of Minnesota Supercomputing Institute for providing the access to Ingenuity Pathway Assistant.

References

- Abdelmohsen K, Pullmann R Jr, Lal A et al (2007) Phosphorylation of HuR by Chk2 regulates SIRT1 expression. Mol Cell 25:543–557
- Al-Haj L, Khabar KS (2012) Cloning of cytokine 3' untranslated regions and posttranscriptional assessment using cell-based GFP assay. Methods Mol Biol 820:91–104
- Anderson P (2008) Post-transcriptional control of cytokine production. Nat Immunol 9:353-359
- Anderson P (2010) Post-transcriptional regulons coordinate the initiation and resolution of inflammation. Nat Rev Immunol 10:24–35
- Baker SJ, Rane SG, Reddy EP (2007) Hematopoietic cytokine receptor signaling. Oncogene 26:6724–6737
- Bakheet T, Frevel M, Williams BR et al (2001) ARED: human AU-rich element-containing mRNA database reveals an unexpectedly diverse functional repertoire of encoded proteins. Nucleic Acids Res 29:246–254
- Bakheet T, Williams BR, Khabar KS (2003) ARED 2.0: an update of AU-rich element mRNA database. Nucleic Acids Res 31:421–423
- Bakheet T, Williams BR, Khabar KS (2006) ARED 3.0: the large and diverse AU-rich transcriptome. Nucleic Acids Res 34:D111–D114
- Beisang D, Bohjanen PR (2012) Perspectives on the ARE as it turns 25 years old. Wiley Interdiscip Rev RNA 3:719–731

- Beisang D, Rattenbacher B, Vlasova-St. Louis IA et al (2012) Regulation of CUG-binding protein 1 (CUGBP1) binding to target transcripts upon T cell activation. J Biol Chem 287:950–960
- Beyaert R, Beaugerie L, Van Assche G et al (2013) Cancer risk in immune-mediated inflammatory diseases (IMID). Mol Cancer 12:98
- Bitterman PB, Polunovsky VA (2012) Translational control of cell fate: from integration of environmental signals to breaching anticancer defense. Cell Cycle 11:1097–1107
- Blackinton JG, Keene JD (2014) Post-transcriptional RNA regulons affecting cell cycle and proliferation. Semin Cell Dev Biol 34:44–54
- Bronicki LM, Jasmin BJ (2013) Emerging complexity of the HuD/ELAVI4 gene; implications for neuronal development, function, and dysfunction. RNA 19:1019–1037
- Brooks SA, Blackshear PJ (2013) Tristetraprolin (TTP): interactions with mRNA and proteins, and current thoughts on mechanisms of action. Biochim Biophys Acta 1829:666–679
- C a n d i d o J, Hagemann T (2013) Cancer-related inflammation. J Clin Immunol 33(Suppl 1):S79–S84
- Cao W, Manicassamy S, Tang H et al (2008) Toll-like receptor-mediated induction of type I interferon in plasmacytoid dendritic cells requires the rapamycin-sensitive PI(3)K-mTOR-p70S6K pathway. Nat Immunol 9:1157–1164
- Carmeliet P, Jain RK (2011) Molecular mechanisms and clinical applications of angiogenesis. Nature 473:298–307
- Chen CY, Shyu AB (1995) AU-rich elements: characterization and importance in mRNA degradation. Trends Biochem Sci 20:465–470
- Colin S, Darne B, Kadi A et al (2014) The antiangiogenic insulin receptor substrate-1 antisense oligonucleotide aganirsen impairs AU-rich mRNA stability by reducing 14-3-3betatristetraprolin protein complex, reducing inflammation and psoriatic lesion size in patients. J Pharmacol Exp Ther 349:107–117
- Dean JL, Wait R, Mahtani KR et al (2001) The 3' untranslated region of tumor necrosis factor alpha mRNA is a target of the mRNA-stabilizing factor HuR. Mol Cell Biol 21:721–730
- Duan J, Shi J, Ge X et al (2013) Genome-wide survey of interindividual differences of RNA stability in human lymphoblastoid cell lines. Sci Rep 3:1318
- Emmons J, Townley-Tilson WH, Deleault KM et al (2008) Identification of TTP mRNA targets in human dendritic cells reveals TTP as a critical regulator of dendritic cell maturation. RNA 14:888–902
- Fan J, Ishmael FT, Fang X et al (2011) Chemokine transcripts as targets of the RNA-binding protein HuR in human airway epithelium. J Immunol 186:2482–2494
- Frasca D, Romero M, Landin AM et al (2010) Protein phosphatase 2A (PP2A) is increased in old murine B cells and mediates p38 MAPK/tristetraprolin dephosphorylation and E47 mRNA instability. Mech Ageing Dev 131:306–314
- Friedel CC, Dolken L, Ruzsics Z et al (2009) Conserved principles of mammalian transcriptional regulation revealed by RNA half-life. Nucleic Acids Res 37, e115
- Fuxe J, Karlsson MC (2012) TGF-beta-induced epithelial-mesenchymal transition: a link between cancer and inflammation. Semin Cancer Biol 22:455–461
- Gratacos FM, Brewer G (2010) The role of AUF1 in regulated mRNA decay. Wiley Interdiscip Rev RNA 1:457–473
- Halees AS, El-Badrawi R, Khabar KS (2008) ARED Organism: expansion of ARED reveals AU-rich element cluster variations between human and mouse. Nucleic Acids Res 36:D137–D140
- Halees AS, Hitti E, Al-Saif M et al (2011) Global assessment of GU-rich regulatory content and function in the human transcriptome. RNA Biol 8:681–691
- Hao S, Baltimore D (2009) The stability of mRNA influences the temporal order of the induction of genes encoding inflammatory molecules. Nat Immunol 10:281–288
- Hodge DL, Berthet C, Coppola V et al (2014) IFN-gamma AU-rich element removal promotes chronic IFN-gamma expression and autoimmunity in mice. J Autoimmun 53:33–45

- Ivanov P, Anderson P (2013) Post-transcriptional regulatory networks in immunity. Immunol Rev 253:253–272
- Jacob CO, Hwang F, Lewis GD et al (1991) Tumor necrosis factor alpha in murine systemic lupus erythematosus disease models: implications for genetic predisposition and immune regulation. Cytokine 3:551–561
- Jenkins BJ (2014) Transcriptional regulation of pattern recognition receptors by JAK/STAT signaling, and the implications for disease pathogenesis. J Interferon Cytokine Res 34:750–758
- Kang JG, Amar MJ, Remaley AT et al (2011) Zinc finger protein tristetraprolin interacts with CCL3 mRNA and regulates tissue inflammation. J Immunol 187:2696–2701
- Kanies CL, Smith JJ, Kis C et al (2008) Oncogenic Ras and transforming growth factor-beta synergistically regulate AU-rich element-containing mRNAs during epithelial to mesenchymal transition. Mol Cancer Res 6:1124–1136
- Katsanou V, Papadaki O, Milatos S et al (2005) HuR as a negative posttranscriptional modulator in inflammation. Mol Cell 19:777–789
- Keene JD (2007) RNA regulons: coordination of post-transcriptional events. Nat Rev Genet 8:533–543
- Khabar KS (2007) Rapid transit in the immune cells: the role of mRNA turnover regulation. J Leukoc Biol 81:1335–1344
- Khabar KS (2010) Post-transcriptional control during chronic inflammation and cancer: a focus on AU-rich elements. Cell Mol Life Sci 67:2937–2955
- Khabar KS, Young HA (2007) Post-transcriptional control of the interferon system. Biochimie 89:761–769
- Kontzias A, Kotlyar A, Laurence A et al (2012) Jakinibs: a new class of kinase inhibitors in cancer and autoimmune disease. Curr Opin Pharmacol 12:464–470
- Lee JE, Lee JY, Wilusz J et al (2010) Systematic analysis of cis-elements in unstable mRNAs demonstrates that CUGBP1 is a key regulator of mRNA decay in muscle cells. PLoS One 5, e11201
- Lee HH, Yoon NA, Vo MT et al (2012) Tristetraprolin down-regulates IL-17 through mRNA destabilization. FEBS Lett 586:41–46
- Li X, Lin WJ, Chen CY et al (2012) KSRP: a checkpoint for inflammatory cytokine production in astrocytes. Glia 60:1773–1784
- Liao W, Lin JX, Leonard WJ (2011) IL-2 family cytokines: new insights into the complex roles of IL-2 as a broad regulator of T helper cell differentiation. Curr Opin Immunol 23:598–604
- Linossi EM, Chandrashekaran IR, Kolesnik TB et al (2013) Suppressor of cytokine signaling (SOCS) 5 utilises distinct domains for regulation of JAK1 and interaction with the adaptor protein Shc-1. PLoS One 8, e70536
- Lopez De Silanes I, Fan J, Galban CJ et al (2004a) Global analysis of HuR-regulated gene expression in colon cancer systems of reducing complexity. Gene Expr 12:49–59
- Lopez De Silanes I, Zhan M, Lal A et al (2004b) Identification of a target RNA motif for RNAbinding protein HuR. Proc Natl Acad Sci USA 101:2987–2992
- Lopez De Silanes I, Galban S, Martindale JL et al (2005) Identification and functional outcome of mRNAs associated with RNA-binding protein TIA-1. Mol Cell Biol 25:9520–9531
- Mansfield KD, Keene JD (2009) The ribonome: a dominant force in co-ordinating gene expression. Biol Cell 101:169–181
- Matsushita K, Takeuchi O, Standley DM et al (2009) Zc3h12a is an RNase essential for controlling immune responses by regulating mRNA decay. Nature 458:1185–1190
- Moudgil KD, Choubey D (2011) Cytokines in autoimmunity: role in induction, regulation, and treatment. J Interferon Cytokine Res 31:695–703
- Mukherjee N, Lager PJ, Friedersdorf MB et al (2009) Coordinated posttranscriptional mRNA population dynamics during T-cell activation. Mol Syst Biol 5:288
- Mukherjee N, Jacobs NC, Hafner M et al (2014) Global target mRNA specification and regulation by the RNA-binding protein ZFP36. Genome Biol 15:R12

- Nepom GT, Ehlers M, Mandrup-Poulsen T (2013) Anti-cytokine therapies in T1D: concepts and strategies. Clin Immunol 149:279–285
- Papadaki O, Milatos S, Grammenoudi S et al (2009) Control of thymic T cell maturation, deletion and egress by the RNA-binding protein HuR. J Immunol 182:6779–6788
- Raghavan A, Bohjanen PR (2004) Microarray-based analyses of mRNA decay in the regulation of mammalian gene expression. Brief Funct Genomic Proteomic 3:112–124
- Raghavan A, Robison RL, Mcnabb J et al (2001) HuA and tristetraprolin are induced following T cell activation and display distinct but overlapping RNA binding specificities. J Biol Chem 276:47958–47965
- Raghavan A, Dhalla M, Bakheet T et al (2004) Patterns of coordinate down-regulation of AREcontaining transcripts following immune cell activation. Genomics 84:1002–1013
- Rattenbacher B, Beisang D, Wiesner DL et al (2010) Analysis of CUGBP1 targets identifies GU-repeat sequences that mediate rapid mRNA decay. Mol Cell Biol 30:3970–3980
- Ronkina N, Menon MB, Schwermann J et al (2010) MAPKAP kinases MK2 and MK3 in inflammation: complex regulation of TNF biosynthesis via expression and phosphorylation of tristetraprolin. Biochem Pharmacol 80:1915–1920
- Saharinen P, Eklund L, Pulkki K et al (2011) VEGF and angiopoietin signaling in tumor angiogenesis and metastasis. Trends Mol Med 17:347–362
- Sarkar S, Han J, Sinsimer KS et al (2011) RNA-binding protein AUF1 regulates lipopolysaccharideinduced IL10 expression by activating IkappaB kinase complex in monocytes. Mol Cell Biol 31:602–615
- Schott J, Stoecklin G (2010) Networks controlling mRNA decay in the immune system. Wiley Interdiscip Rev RNA 1:432–456
- Seko Y, Cole S, Kasprzak W et al (2006) The role of cytokine mRNA stability in the pathogenesis of autoimmune disease. Autoimmun Rev 5:299–305
- Srikantan S, Gorospe M (2012) HuR function in disease. Front Biosci J Virtual Libr 17:189-205
- Stellato C, Gubin MM, Magee JD et al (2011) Coordinate regulation of GATA-3 and Th2 cytokine gene expression by the RNA-binding protein HuR. J Immunol 187:441–449
- Stoecklin G, Tenenbaum SA, Mayo T et al (2008) Genome-wide analysis identifies interleukin-10 mRNA as target of tristetraprolin. J Biol Chem 283:11689–11699
- Stumpo DJ, Lai WS, Blackshear PJ (2010) Inflammation: cytokines and RNA-based regulation. Wiley Interdiscip Rev RNA 1:60–80
- Sun L, Stoecklin G, Van Way S et al (2007) Tristetraprolin (TTP)-14-3-3 complex formation protects TTP from dephosphorylation by protein phosphatase 2a and stabilizes tumor necrosis factor-alpha mRNA. J Biol Chem 282:3766–3777
- Talwar S, Balasubramanian S, Sundaramurthy S et al (2013) Overexpression of RNA-binding protein CELF1 prevents apoptosis and destabilizes pro-apoptotic mRNAs in oral cancer cells. RNA Biol 10:277–286
- Turner M, Hodson D (2012a) Regulation of lymphocyte development and function by RNAbinding proteins. Curr Opin Immunol 24:160–165
- Turner M, Hodson DJ (2012b) An emerging role of RNA-binding proteins as multifunctional regulators of lymphocyte development and function. Adv Immunol 115:161–185
- Uehata T, Iwasaki H, Vandenbon A et al (2013) Malt1-induced cleavage of regnase-1 in CD4(+) helper T cells regulates immune activation. Cell 153:1036–1049
- Vlasova IA, Bohjanen PR (2008) Posttranscriptional regulation of gene networks by GU-rich elements and CELF proteins. RNA Biol 5:201–207
- Vlasova IA, Mcnabb J, Raghavan A et al (2005) Coordinate stabilization of growth-regulatory transcripts in T cell malignancies. Genomics 86:159–171
- Vlasova IA, Tahoe NM, Fan D et al (2008) Conserved GU-rich elements mediate mRNA decay by binding to CUG-binding protein 1. Mol Cell 29:263–270
- Vlasova-St. Louis I, Bohjanen PR (2011) Coordinate regulation of mRNA decay networks by GU-rich elements and CELF1. Curr Opin Genet Dev 21:444–451
- Vlasova-St. Louis I, Bohjanen PR (2014) Post-transcriptional regulation of cytokine signaling by AU-rich and GU-rich elements. J Interferon Cytokine Res 34:233–241

- Vlasova-St. Louis I, Dickson AM, Bohjanen PR et al (2013) CELFish ways to modulate mRNA decay. Biochim Biophys Acta 1829:695–707
- Von Roretz C, Di Marco S, Mazroui R et al (2011) Turnover of AU-rich-containing mRNAs during stress: a matter of survival. Wiley Interdiscip Rev RNA 2:336–347
- Weichhart T, Saemann MD (2008) The PI3K/Akt/mTOR pathway in innate immune cells: emerging therapeutic applications. Ann Rheum Dis 67(Suppl 3):iii70–iii74
- White EJ, Brewer G, Wilson GM (2013) Post-transcriptional control of gene expression by AUF1: mechanisms, physiological targets, and regulation. Biochim Biophys Acta 1829: 680–688
- Winzen R, Gowrishankar G, Bollig F et al (2004) Distinct domains of AU-rich elements exert different functions in mRNA destabilization and stabilization by p38 mitogen-activated protein kinase or HuR. Mol Cell Biol 24:4835–4847
- Winzen R, Thakur BK, Dittrich-Breiholz O et al (2007) Functional analysis of KSRP interaction with the AU-rich element of interleukin-8 and identification of inflammatory mRNA targets. Mol Cell Biol 27:8388–8400
- Wu X, Chesoni S, Rondeau G et al (2013) Combinatorial mRNA binding by AUF1 and Argonaute 2 controls decay of selected target mRNAs. Nucleic Acids Res 41:2644–2658
- Yiakouvaki A, Dimitriou M, Karakasiliotis I et al (2012) Myeloid cell expression of the RNAbinding protein HuR protects mice from pathologic inflammation and colorectal carcinogenesis. J Clin Invest 122:48–61
- Zhang L, Lee JE, Wilusz J et al (2008) The RNA-binding protein CUGBP1 regulates stability of tumor necrosis factor mRNA in muscle cells: implications for myotonic dystrophy. J Biol Chem 283:22457–22463

Post-transcriptional Regulation of Luteinizing Hormone Receptor mRNA Expression in the Ovary

K.M.J. Menon, Bindu Menon, and Thippeswamy Gulappa

1 Introduction

In most mammals, the ovary undergoes continuous changes in structure and function during the reproductive period. In the human after menarche, the ovary secretes estrogens and progesterone in a cyclical manner to facilitate ovum production and preparing the endometrium for implantation of the blastocyst if fertilization has occurred. In the absence of blastocyst implantation, corpus luteum, the specialized structure that secretes progesterone, regresses and the cyclical pattern of hormone secretion repeats again. This process repeats continuously until menopause which occurs after the depletion of ovum resulting in the cessation of the cyclical changes in ovarian structure and function. In rodent and human, luteinizing hormone (LH) regulates androgen biosynthesis by the theca-interstitial cells during follicular development and these androgens are converted to estrogens by the granulosa cells (Bjersing 1968). The follicular growth is stimulated mainly by follicle stimulating hormone (FSH) which acts in concert with estradiol and insulin/IGF-1 system (Hirshfield 1991; Zeleznik 2004). More importantly, the growing follicles acquire LH receptors by the combined actions of FSH and estradiol (Zeleznik 2004; Menon et al. 2005). LH is also responsible for ovulation and for the biosynthesis of progesterone, a hormone required for preparing the uterus for the implantation of the blastocyst as well as maintaining the conceptus if implantation has occurred.

The action of LH is mediated by LH receptor which belongs to the glycoprotein subfamily of the large G protein-coupled receptor (GPCR) family with an unusually

K.M.J. Menon (🖂) • B. Menon • T. Gulappa

Department of Biological Chemistry, The University of Michigan Medical School, 6428 Medical Science I, 1150 West Medical Center Drive, Ann Arbor, MI 48109-0617, USA

Supported by NIH grant HD 06656.

Department of Obstetrics and Gynecology, The University of Michigan Medical School, 6428 Medical Science I, 1150 West Medical Center Drive, Ann Arbor, MI 48109-0617, USA e-mail: kmjmenon@umich.edu

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_4

large extracellular domain, seven membrane spanning helices and an intracellular domain with two conserved palmitoylation sites (McFarland et al. 1989; Loosfelt et al. 1989). Its expression increases during follicle development and show a dramatic decline in response to a bolus of LH that occurs prior to ovulation when the oocyte reaches the metaphase of the second meiotic division (Baker 1972). During this period of transition, a refractory period of LH responsiveness occurs that is characterized by a transient loss of LH receptors and consequent loss of the LH responsiveness (Hunzicker-Dunn et al. 1979; Menon et al. 2004). The receptor levels then rises again reaching maximum levels by mid portion of the luteal phase to support progesterone production. The LH receptor expression then falls with the regression of the corpus luteum in the absence of pregnancy. The changes in LHR expression are exquisitely regulated to ensure successful reproduction. While FSH, along with other paracrine factors, is known to regulate the development of the primary follicles to preantral and antral stages, the LH receptor makes its appearance in significant amounts as a result of FSH stimulation (Zeleznik 2004; Channing et al. 1980), Once LHR expression is established in the growing follicle, subsequent changes in its expression during the ovarian cycle is caused by changes in the levels of its own ligand (Menon et al. 2004). We have examined the biochemical mechanism that is responsible for the changes in the levels of LHR expression using both rodent and human models. Since there is striking change in LHR following preovulatory LH surge under physiological conditions, we have used experimental conditions that mimic this situation to examine the biochemical mechanism that regulates LH receptor expression using a rodent model.

2 Post-transcriptional Regulation of LH Receptor mRNA Expression

Administration of hCG, a hormone with significant chemical identity to LH, produces a rapid decline in LHR mRNA and protein in the rat and human ovaries (Hoffman et al. 1991; Lu et al. 1993; LaPolt et al. 1990; Segaloff et al. 1990; Nair et al. 2002). Since G protein coupled receptors are internalized after ligand binding (Menon et al. 2004), the simplest explanation for the disappearance of LH receptor following exposure to high concentration of the ligand is that the ligand-receptor complex undergoes endocytosis and not replaced by newly synthesized receptor causing a reduction in cell surface expression. The steady state level of receptor expression is the balance between the rate at which the receptor is trafficked to the cell surface and the rate of its endocytosis. The loss of steady state levels of receptor expression could also be due to a temporary pause in synthesis of new receptor protein by decreased transcription, or an increase in degradation of mRNA. To examine these possibilities, we used ovaries from super-ovulated rats as a model system. Immature rats were treated with 50 IU PMSG followed by 25 IU hCG 56 h later to induce superovulation. On day 4 of superovulation, one set of animals was treated with 50 IU hCG, and a second control group received normal saline. Examination of LHR expression in the ovaries collected at different time intervals



Fig. 4.1 LHR mRNA levels during hCG-induced downregulation. Northern blot analysis of steady state LHR mRNA levels in total RNA, isolated at the indicated times, from the ovaries of saline-injected (control) (*lanes 1–6*) or hCG-injected (downregulated) (*lanes 7–11*) rats. Blots were probed using a labeled cDNA encoding the LHR carboxyl terminus and a portion of the 3'-UTR (nucleotides 1936–2682). RNA was extracted from control and downregulated ovaries at 0, 6, 12, 24, 48, and 72 h after hCG injection. This research was originally published in J Biol Chem (Kash JC, Menon KMJ. *J Biol Chem.* 1998. 273:10658–64)

up to 72 h (Hoffman et al. 1991; Lu et al. 1993; Peegel et al. 1994) by Northern blot analysis, real time PCR and in situ hybridization analysis in tissue sections showed a steady decline in the expression levels of LH receptor mRNA starting 6 h following hCG treatment reaching undetectable levels by 24 h (Hoffman et al. 1991; Lu et al. 1993; Peegel et al. 1994). The recovery from downregulation was seen after 24 h and reached almost control levels by 72 h, as shown in Fig. 4.1. So, the loss of cell surface receptor via internalization of the ligand-bound receptor does not appear to be the sole mechanism responsible for the disappearance of the receptor during downregulation. The loss of mRNA was not expected, since the conventional thinking was that the rapid internalization of the ligand-receptor complex would be the major cause of the decrease of cell surface receptor expression. The in situ hybridization pattern of the receptor expression in the whole ovary was in full agreement with the Northern blot data which further suggested that the same corpora lutea that were depleted of the receptor mRNA regained the receptor (Peegel et al. 1994). It was interesting to note that the process of LH receptor mRNA loss was very specific since other mRNAs involved in ovarian function such as Cyp11A showed no decline. In fact, the expression of the steroidogenic enzyme mRNA expression showed an increase caused by the initial cellular response to hCG by increasing cyclic AMP production which has been known to activate the expression of key enzymes involved in ovarian steroidogenesis (Menon and Gunaga 1974; Marsh et al. 1966; Miller 2008). The loss of LHR in response to hCG treatment was mediated by increased production of cyclic AMP. This is further supported by the finding

that LHR mRNA downregulation can be achieved by chronic elevation of cyclic AMP levels by repeated administration of inhibitors of type 2 phosphodiesterase (Peegel et al. 2005). The evidence for the decrease in the steady state levels of LHR mRNA expression occurred through post-transcriptional mechanism came from the determination of transcription rates by performing nuclear run-on assays which showed no decrease in transcription rates during the time course of downregulation when compared to the control. In fact, the nuclei isolated from the hCG treated ovaries incorporated [³H] uridine into total RNA at much higher levels compared to the control (Lu et al. 1993). These findings supported the view that the loss of LHR mRNA during ligand-induced downregulation was a result of increased mRNA degradation rather than a decrease in the transcription rate.

3 Accelerated LH Receptor mRNA Degradation as a Regulatory Mechanism

It is now well recognized that the expression of specific, highly regulated mRNAs is regulated, at least in part, at the level of mRNA degradation. There are several examples where the mRNA degradation rate controls the steady state levels of mRNA expression (Port et al. 1992; Bernstein et al. 1992; Sachs 1993; Shyu et al. 1989; Ross 1995; Moghul et al. 1994; Olivas and Parker 2000; Tharun et al. 2000; Pachter et al. 1987; Perez-Ortin et al. 2013; Garneau et al. 2007). In almost all instances where the mRNA expression is regulated by controlling its degradation, the changes in the stability of the mRNA have been shown to result from the binding of specific proteins to specific sequences and/or structures of the mRNA. The specific regions that the mRNA binding proteins interact with may be localized either on the 5' untranslated region, the coding region or the 3' untranslated region of the mRNA (Ross 1995). In general, the steady state levels of the mRNA expression are regulated by either increasing or decreasing the degradation rate resulting from the interaction with specific RNA binding proteins.

The possible existence of a binding protein that specifically recognizes LHR mRNA was examined by incubating [^{32}P] labeled LH receptor mRNA with a cytosolic fraction (100 × g supernatant) prepared from the ovary pretreated with hCG to downregulate LH receptor mRNA expression in the presence of ribonuclease inhibitors (Kash and Menon 1998, 1999). After treating the reaction mixture with ribonuclease T1 to degrade the unreacted RNA probe, and the ribonucleoprotein (RNP) complex was separated by electrophoresis on an 8 % native acrylamide gel and subjected to autoradiography. Two RNP complexes were identified, one prominent band corresponding to 50 kDa and a second less intense band of 45 kDa (Fig. 4.2). A threefold increase in the 50 kDa RNP complex was seen in samples derived from LHR downregulated ovaries but no significant change in the intensity of the 45 kDa band was found. Because of this reason, further studies focused on the larger 50 kDa ribonucleoprotein complex. The protein was named LH receptor mRNA binding protein and abbreviated as LRBP (Kash and Menon 1998, 1999).



The binding of LHR mRNA to LRBP showed that the recognition of the mRNA resided in the amino terminal portion of the LH receptor corresponding to nucleotides 102–282. The RNP complex formation was always higher in the ovarian tissue derived from LH receptor downregulated ovaries with no change in the non-target tissues (Kash and Menon 1998). Further truncation of this region showed that the contact site resided between nucleotides 188–228. RNA hydroxy-radical footprinting revealed a bipartite polypyrimidine rich sequence, (UCUCX7-UCUCCCU)



Fig. 4.3 Diagram of the LRBP RNA binding site. LH/hCG receptor mRNA open reading frame showing the position of the LRBP binding site (LBS). The sequence of the LHR:188–228 is also shown (Modified from Kash, J.C. and Menon, K.M.J., 1999, Fig. 8 with permission from Biochemistry)

corresponding to nucleotides 203–220 of LHR mRNA that specifically interacted with LRBP (Fig. 4.3). Mutation of the C residues within the bipartite sequence revealed that all C residues participated in binding to LRBP (Kash and Menon 1999). The contact site was further confirmed by performing RNA electrophoretic mobility shift assays. The binding was specific, since the RNP complex formation using [³²P] labeled LH receptor mRNA probe (nucleotides 203–220) was not displaced by probes corresponding to other regions of LH receptor mRNA. The interaction of LHR mRNA with LRBP exhibited high affinity with equilibrium dissociation constant (Kd) in the range of 4×10^{-9} M.

After establishing that the ovarian LRBP recognizes LH receptor mRNA with high affinity and specificity, the changes in LRBP activity in relation to tissue levels of LH receptor mRNA were examined. It has been established that LHR mRNA expression increases during maturation of the ovarian follicles in response to FSH. The levels then fall immediately after the preovulatory LH surge and the levels rise again with the growth and development of the corpus luteum. The relationship between LHR mRNA expression and LRBP activity was examined in ovarian tissues after treating immature rats with pregnant mare serum gonadotropin to induce follicle growth and to increase LHR mRNA expression levels. At this time point, the LRBP activity in the S100 fractions prepared from the ovaries showed low levels of activity. Thus a negative correlation between tissue levels of LHR mRNA expression and LRBP activity was observed. However, when LHR expression was suppressed during ligand-induced downregulation by treatment with hCG, the LRBP activity was increased several fold in a time dependent manner. Thus, when the LH receptor mRNA expression level was high, the RNA binding protein activity was at the lowest level. Conversely, the LH receptor mRNA binding activity was high when the mRNA levels began to fall. This inverse relationship suggested that LRBP is an endogenous regulator of LH receptor mRNA expression.

4 Signaling Pathways Involved in Ligand-Mediated Downregulation of LH Receptor mRNA

Since the immediate response of the ovarian tissue to hCG treatment is to increase the production of cyclic AMP (Menon and Gunaga 1974; Marsh et al. 1966), the involvement of cyclic AMP in LHR mRNA downregulation was presumed. In fact, this assumption turned out to be true by the demonstration that chronic elevation in cyclic AMP production in ovaries of superovulated rats with multiple injections of rolipram, an inhibitor of type 2 phosphodiesterase (Peegel et al. 2005), showed downregulation of LH receptor mRNA similar to that produced by the treatment with hCG. The ovaries were then processed for REMSA in order to determine whether there was increased LRBP activity when LH receptor mRNA was downregulated. A substantial increase in the RNA binding activity was seen in ovarian S100 fractions at time intervals when the LH receptor mRNA expression was downregulated. These findings suggested that the downregulation of LH receptor mRNA in response to hCG treatment was caused by an increase in cyclic AMP production. In addition, we were able to demonstrate that cyclic AMP-mediated activation of PKA and ERK1/2 signaling pathways is involved in ligand-mediated LH receptor downregulation in human granulosa cells (Menon et al. 2011). The participation of PKA and ERK1/2 signaling pathways was demonstrated with the use of H89 and U0126, respectively. Furthermore, transfecting cultured human granulosa cells with ERK1/2-specific small interfering RNA abrogated LH receptor downregulation in response to hCG treatment thereby conclusively demonstrating the participation of ERK1/2 signaling pathways in inducing downregulation (Menon et al. 2011).

5 In Vitro mRNA Degradation by the LH Receptor mRNA Binding Protein

After establishing a relationship between the expression of LH receptor mRNA and the RNA binding protein under in vivo conditions, the direct role of the RNA binding protein in LH receptor mRNA degradation was demonstrated by employing a cell-free mRNA decay system developed by Ross and colleagues (Ross 1993) under in vitro conditions. The assay essentially determines the degradation rate of a specific exogenous RNA in the presence and absence of the RNA binding protein under controlled conditions. Although all mRNAs are prone to degradation, the rate of degradation varies depending on the cellular environment. Some mRNAs are more labile than others, a property conferred by the presence of appropriate trans-acting factors that interact with specific structures or sequences present in the mRNA. Initially, we showed that the rate of decay of LH receptor mRNA was very rapid in ribosomes isolated from the ovaries of rats treated with hCG, compared to the degradation of LH receptor mRNA by ribosomes isolated from ovaries of saline treated control animals (Nair et al. 2002), but the rate of decay of exogenously added ovarian RNA by ribosomes isolated from saline treated rats was accelerated by the addition of a partially purified ovarian LH receptor mRNA binding protein

(Nair et al. 2002). The rate of LH receptor mRNA decay was not affected by unrelated proteins added to the reaction mixture. These experiments demonstrated that the LHR mRNA binding protein plays a role in LH receptor mRNA degradation.

6 Characterization of LH Receptor mRNA Binding Protein

Purification of RNA binding proteins from tissue samples presents many challenges including the relatively time consuming assays to follow the RNA binding activity using RNA electrophoretic gel mobility shift assay and its low abundance in the cell. Covalently linked RNA binding sequence to sepharose followed by column chromatography showed that the binding protein was retained on the columns, but the attempts to elute the binding protein retained on the matrix were not successful. Therefore, conventional techniques were used to purify the LH receptor mRNA binding protein from the $100 \times g$ supernatant fraction of the ovarian homogenates. The ovaries were initially downregulated by treatment with the ligand in order to increase the yield of the binding protein. The supernatants were subjected to chromatography on a strong cation exchange resin (Macro-Prep high S support) and eluted with 150 mM KCl. The eluates were concentrated and subjected to SDS-PAGE to separate the proteins. The [32P] LHR mRNA binding activity associated with the protein band on the gel was identified by an overlay assay (Northwestern blot) using [³²P] labeled LH receptor mRNA fragment (203–220) as the probe. After extensive standardization of the assay, the protein band corresponding to the band that showed the RNA binding activity was cut, eluted and renatured. The eluted protein was electrophoresed again to determine the purity of the preparation. The electrophoretically homogeneous protein band was then subjected to amino terminal analysis as well as MS-MALDI analysis to establish its identity. Both analyses revealed the purified protein to be mevalonate kinase (Nair and Menon 2004). The gene encoding rat mevalonate kinase was then cloned and overexpressed in 293 T cells (Nair and Menon 2004). The recombinant protein exhibited a concentration dependent increase in binding LHR mRNA probe (Nair and Menon 2004). The binding exhibited all the characteristics of the expected LHR mRNA binding protein with respect to specificity for binding to the previously identified contact site (nucleotides 203–220), competition by unlabeled LHR mRNA fragment (203–220), dependence on C residues in the ligand binding site and immunoreactivity of the recombinant protein similar to that seen for the electrophoretically purified rat mevalonate kinase by Western blot analysis. Furthermore, since mevalonate kinase is known to have two binding sites, one for ATP and a second site for mevalonate, the involvement of these sites in LH receptor mRNA binding activity was then determined. It was seen that the binding of the [32P] labeled LH receptor mRNA binding sequence to mevalonate kinase was inhibited by ATP and mevalonate. The inhibitory effect was even more pronounced in the presence of both ATP and mevalonate. These results clearly confirm the identity of the RNA binding protein as mevalonate kinase. Mevalonate kinase is a metabolic enzyme involved in cholesterol biosynthesis which catalyzes the conversion of mevalonate to phosphomevalonate, the precursor

of cholesterol and many natural products. In fact, we have shown that mevalonate kinase expression is regulated by LH in the corpus luteum both at the mRNA level as well as at the protein level (Wang and Menon 2005). During LH/hCG-induced downregulation, mevalonate kinase expression was induced in the corpus luteum. The induction of expression of other SRE containing genes such as HMG CoA reductase, farnesyl pyrophosphate synthase and LDL receptor were also seen during this phase (Wang and Menon 2005). These results suggest that mevalonate kinase expression plays a regulatory role in LH receptor mRNA expression in the corpus luteum.

Depletion of mevalonate kinase expression by 25 hydroxycholesterol resulted in the abrogation of LH receptor mRNA downregulation in cultured human granulosa cells (Menon et al. 2011; Wang and Menon 2005). The inhibitory effect of mevalonate kinase on LH receptor expression was documented by Ikeda et al. 2008 who showed that overexpression of mevalonate kinase can abrogate FSH-induced increase in LH receptor mRNA expression in rat ovarian follicles.

The notion that mevalonate kinase, a metabolic enzyme, acts as an mRNA binding protein is consistent with a similar role of several other metabolic enzymes that have been recently described (Menon et al. 2005; Hentze 1994). The iron response element binding protein, which plays a role in iron homeostasis in cells, is an enzyme in the citric acid cycle. Similarly, thymidylate synthase, dihydrofolate reductase, glyceraldehyde-3-phosphate dehydrogenase, glutamate dehydrogenase and lactate dehydrogenase all have been shown to serve as RNA binding proteins that regulate different aspects of RNA metabolism. In the case of mevalonate kinase, structural studies have shown that it belongs to a family of ATP binding proteins containing a conserved glycine motif. Members of this group include galactokinase, homoserine kinase, mevalonate kinase and phosphomevalonate kinase. These proteins share a common fold, a left handed $\beta - \alpha - \beta$ loop known as the ribosomal protein S5 domain 2 like fold. This fold is similar to that found in proteins that interact with nucleic acids including elongation factor G and ribonuclease P (Zhou et al. 2000). Thus from a structural point, it is conceivable that mevalonate kinase can serve as an LH receptor mRNA binding protein to regulate cellular expression of LH receptor under select conditions.

7 Mechanism of LH Receptor mRNA Decay by LRBP

The endogenous association of LRBP with LH receptor mRNA was seen during downregulation supporting the view that LRBP is likely responsible for rapid degradation of LHR (Nair et al. 2002). The evidence comes from studies using ovarian extracts from the downregulated ovaries. The RNP complexes isolated from these ovaries were immunoprecipitated with anti LRBP antibodies and RNA extracted from the immunoprecipitates. RNAs were reverse transcribed and then subjected to PCR amplification using specific probes for LH receptor. Significant amplification of LHR was seen in the downregulated ovarian extracts suggesting that the association of LRBP with LHR mRNA occurs during downregulation (Nair et al. 2002).



Fig. 4.4 Translational suppression of LHR mRNA by LRBP in vitro. (**a**) FLAG-tagged rat LHR mRNA (200 ng) was in vitro translated using a Flexi rabbit reticulocyte lysate system in the presence of [35 S] methionine and partially purified rat LRBP from the ovary or fatty acid-free bovine serum albumin (*BSA*) at different concentrations (2.5 and 5.0 µg). The translated LHR protein was immunoprecipitated with anti-FLAG antibody, separated on 10 % SDS-PAGE, and processed for autoradiography. (**b**) In vitro translation of FLAG-tagged rat LHR mRNA using [35 S]methionine in the presence of increasing concentrations (1, 2, and 3 µg) of purified rat LRBP, followed by immunoprecipitation, SDS-PAGE and autoradiogram. This research was originally published in J Biol Chem. (Nair AK et al. *J Biol Chem.* 2005, 280, 42809–42816)

Since the coding region of LHR mRNA is recognized by LRBP, it is quite likely that LRBP might act as a translational suppressor of LHR mRNA. This was indeed the case. We developed a cell free protein synthesizing system in which FLAG tagged LHR mRNAs were efficiently translated using an in vitro rabbit reticulocyte lysate system to yield 63 kDa protein species corresponding to non-glycosylated LHR. Addition of purified LRBP blocked the translation of LH receptor mRNA while addition of a region of corresponding to LRBP binding sequence (LBS) in which all the cytidine residues were mutated to uridine residues were not affected (Fig. 4.4). The wild type LBS effectively inhibited the translation of LH receptor mRNA. This clearly showed that LRBP acts as a translational suppressor of LH receptor mRNA and the loss of LH receptor mRNA during downregulation suggested that untranslatable LH receptor mRNA might be targeted for degradation. The translation of eukaryotic mRNAs is a complex process. In the cytoplasm, the fully processed mRNA can exist either in a polysome-associated, translationally active form or in an inactive messenger ribonucleoprotein (RNP) complex (Xu et al. 2000; Xu and Grabowski 1999; Mathews et al. 2000; Hershey 1991). The exchange between these two compartments modulates the fraction of mRNA available for translation (Kozak 1989). A clear link between translation and decay are intimately connected since the factors and processes required for mRNA translation and decay are closely connected (Ross 1995; Mathews et al. 2000; Jacobson and Peltz 1996) and translational arrest, in general, alters the degradation rate of most eukaryotic mRNAs (Ross 1995; Mathews et al. 2000).

Since LRBP is a cytosolic protein, we further examined its translocation from cytoplasm to ribosomes during downregulation to cause translational suppression. We were able to show that Western blot analysis of ribosome rich fractions from FPLC-assisted gel filtration of post-mitochondrial fractions showed the presence of LRBP in translating ribosomes from Downregulated ovaries , but not from the control ovaries. Increased association of LH receptor mRNA with LRBP was seen in the polysomes isolated from the downregulated ovaries when compared to the control group. The increased association of LH receptor mRNA with LRBP in polysomes from downregulated ovaries was associated with decreased translational effectiveness of these polysomes again showing the translational suppression by LRBP (Menon et al. 2009). Thus, during downregulation, an increase in the translocation of LRBP to ribosome and its increased association with LHR mRNA leads to its increased degradation.

Increased translocation of LRBP to polysome and increased association in translating ribosomes during downregulation suggested that these processes precede a decrease in LH receptor mRNA levels. The effect of these early events during downregulation showed that the association of LRBP with ribosomes and translational suppression led to its rapid degradation by facilitating decapping. Immunoprecipitation of the complex with 5' cap structure antibody followed by real time PCR analysis using specific LH receptor mRNA probes revealed progressive loss of capped LH receptor mRNA during downregulation (Menon et al. 2013a). RNA immunoprecipitation confirmed the dissociation of eukaryotic initiation factor 4E from the cap structure which is required for decapping. Furthermore, RNA immunoprecipitation with antibody against the P body marker protein, DCP1A revealed that LH receptor mRNA was associated with the P bodies, the cytoplasmic foci that contain RNA degradative enzymes and decapping complex (Menon et al. 2013a). Interestingly, immunohistochemical studies using antibodies against LRBP and DCP1A followed by confocal analysis showed co-localization of LRBP with DCP1A during LHR downregulation. The association of LRBP and LH receptor mRNA in the P bodies during downregulation was also confirmed by examining the association of a second P body marker rck/p54. These results are supportive of the notion that during downregulation, LRBP forms an untranslatable ribonucleoprotein complex which is then translocated to P bodies leading to decapping and degradation by exonucleases (Fig. 4.5).

8 Interaction of LRBP with Other Trans Factors during Downregulation

Since LRBP is devoid of any nuclease activity capable of degrading LH receptor mRNA, the possibility of other proteins that selectively interact with LRBP and if such proteins exist, their functional roles in directing LH receptor mRNA to decay pathways was examined. We used a yeast two hybrid system to screen cDNA library which was constructed from LHR downregulated ovaries (Wang et al. 2010). Using this strategy we identified five potential candidate proteins that interacted with



Fig. 4.5 Proposed mechanism of LRBP mediated degradation of LHR mRNA in the P bodies. LH signaling through its receptor activates cAMP/PKA/ERK pathways, leading to increased levels of LRBP. LRBP binds to the coding region of LHR mRNA in the ribosomes and an untranslatable ribonucleoprotein complex is formed. LRBP also associates with the P body components such as DCP1A and rck/p54 resulting in the P body localization of this mRNP complex. This is followed by the dissociation of eIF4E from the cap structure and the decapping of LHR mRNA in the P bodies, initiated by DCP2 enzyme. Final step is the 5'-3' exonuclease (XRN1) mediated degradation of LHR mRNA

LRBP during downregulation. Among these, UBCE2i which is now recognized as the sumoylating enzyme has been implicated in nuclear-cytoplasmic transport in addition to other cellular functions. SUMOplotT revealed that LRBP has two potential sumoylation sites at K 345 and K 256. Overexpressed LRBP in 293 T cells was assayed for sumoylation (Wang et al. 2010) and revealed that both lysine 256 and 345 were sumoylated whereas mutation of the lysine residues to arginine blocked sumoylation. Although the actual functional significance of sumoylation of LRBP is not understood, it is likely that sumoylation might play a role in the translocation of LRBP-bound LH receptor mRNA to P bodies for degradation.

A second LRBP interacting protein identified was eukaryotic initiation factor 5A (eIF5A). Its functional role in RNA metabolism has been a topic of investigation in several laboratories since it has been shown that it participates in many aspects of RNA metabolism including RNA decay, transport and most recently, translation elongation (Rossi et al. 2014; Zanelli and Valentini 2007; Ruhl et al. 1993; Zuk and Jacobson 1998; Gutierrez et al. 2013; Dever and Green 2012; Saini et al. 2009; Schrader et al. 2006). EIF5A has been shown to undergo hypusination, a unique

post-translational modification which is essential for its function (Park et al. 1993, 2010) including RNA binding. Since eIF5A was found to interact with LRBP in the LHR mRNA-downregulated state, we examined whether eIF5A plays a role in LH receptor downregulation. First we showed that during downregulation, eIF5A undergoes hypusination by transferring 4-aminobutyl group from the polyamine, spermidine to the \in amino group of lysine by following the incorporation of [³H] from [³H] labeled spermidine into eIF5A. After electrophoretic separation by SDS-PAGE, [3H] labeled eIF5A (hypusinated) was detected by fluorography. A time dependent increase in the incorporation of [³H] into eIF5A was detected as early as 4 h after instituting ligand treatment to induce downregulation. The appearance of ³H] labeled eIF5A was blocked by GC7, an inhibitor of hypusination. Pretreatment of rats with GC7 2 h prior to treatment with hCG, showed a marked decrease in the association of eIF5A with LRBP. Even more interesting was the finding that GC7 pretreatment reversed the ligand-induced downregulation of LH receptor mRNA as well as the appearance of functional receptors on the ovarian cell surface. Analysis of the equilibrium dissociation constant for the ligand-receptor interaction showed comparable Kd values for the control group and GC7 treated group, but the number of the binding sites in the GC7 treated group was significantly higher than the group without GC7 pretreatment (Menon et al. 2014a). Recent studies have shown that hypusination of eIF5A also participates in the formation of P bodies, the presumed site of mRNA decay (Gregio et al. 2009) and in the nucleocytoplasmic translocation of HIV-1Rev and human T cell leukemia virus type 1Rex RNA export factors (Elfgang et al. 1999). We have also shown that LH receptor mRNA-LRBP complex is shuttled from ribosomes to P bodies for degradation (Menon et al. 2009). However, since there was no discernible increase in the association of eIF5A with DCP1A, during downregulation the role of hypusinated eIF5A in shuttling the mRNP complex to the degradation machinery was ruled out. It is likely that eIF5A might act by augmenting translational suppression of LH receptor mRNA-LRBP complex which is then targeted for degradation, as summarized in Fig. 4.6.

9 Role of microRNA-122 (miR-122) in LH Receptor Expression

Since LRBP plays a crucial role in LH receptor expression during ligand-induced downregulation, the regulatory factors controlling LRBP expression would be expected to have a major influence on LH receptor mRNA expression. Mevalonate kinase (LRBP) is a member of the family of enzymes encoded by genes containing oxysterol responsive element. Oxysterols act by suppressing the proteolytic cleavage of the precursor of sterol response element binding protein (SREBP) to the active form. The cleavage product, the active form of SREBP binds to sterol response element (SRE) which causes transcriptional activation. Conversely, when cellular concentrations of oxysterols are low, the cleavage of the precursor of SREBP is enhanced to produce active form of SREBP which then accelerates transcription of genes containing SRE elements in the promoter region. Mevalonate

Fig. 4.6 Role of hypusinated eIF5A in LRBP-mediated LHR mRNA downregulation. Binding of LH/hCG to LH receptor stimulates the expression and hypusination of eIF5A, which then forms a complex with LRBP-LHR mRNA leading to the translational suppression and degradation of LHR mRNA



kinase, phosphomevalonate kinase, hydroxymethyl glutaryl CoA reductase are representative of enzymes encoded by SRE containing genes. Our studies have shown that addition of 25 hydroxycholesterol to cultured human granulosa cells suppressed LRBP expression as well as ligand induced downregulation of LH receptor mRNA expression. Krutzfeldt et al. (2005 and Schrader et al. (2006) have identified miR-122 as a positive regulator of MVK expression in the liver. It is therefore possible that miR-122 might regulate LHR mRNA expression during ligandinduced downregulation by regulating LRBP levels. We examined the expression of miR-122 in rat and human ovary by fluorescent in situ hybridization using 5'-fluorescein isothiocyanate labeled miR-122 locked nucleic acid probe, and by real time PCR (Menon et al. 2013b). Our results showed that miR-122 is expressed in the ovary and that the expression levels are increased by treatment with hCG. This suggested that miR-122 might play a regulatory role in ligand-induced downregulation of LH receptor mRNA expression by controlling the expression of LRBP since ovarian tissue endogenously expresses this micro RNA. The expression of miR-122 showed a time dependent increase at earlier time periods after inducing LHR mRNA downregulation and preceded the increases in LRBP mRNA and protein expression. These increases were followed by the downregulation of LHR mRNA expression. Inhibition of protein kinase A and ERK1/2 signaling pathways by H-89 and U0126, respectively blunted ligand-induced increases in miR-122 expression suggesting





that ligand-mediated activation of miR-122 was mediated by cyclic/AMP/ERK1/2 signaling pathway. The involvement of miR-122 was further confirmed by inhibition of LRBP levels by treatment with locked nucleic acid-conjugated antagomir of miR-122 during ligand-induced downregulation. Furthermore, the levels of active forms of SREBP-1a and SREBP-2 were increased during downregulation. The role of miR-122 as an upstream activator of SREBP comes from studies where we showed that SREBP activation was blocked by injecting the miR-122 antagomir into the bursa of the ovary prior to treatment with hCG. Additional support for the role SREBP in LHR mRNA downregulation was obtained by blocking LHR downregulation by pre-treatment with fatostatin, an inhibitor of SREBP processing in the endoplasmic reticulum, prior to treatment with hCG (Menon et al. 2014b). Thus ligand-induced downregulation of LHR mRNA expression is mediated by an increase in miR-122 through cyclic AMP/PKA/ERK signaling pathway leading to increased levels of active forms of SREBPs which, in turn regulate, LRBP expression (Fig. 4.7).

10 Conclusion

The regulation of LHR mRNA expression at the post-transcriptional level in a constantly changing hormonal environment is advantageous to the cell. It provides a means to regulate LHR mRNA levels and consequently the receptor density on the

cell surface by a relatively simple mechanism by changing LHR mRNA stability without having to reprogram the more complex transcriptional machinery. By maintaining appropriate levels of the receptor to meet a number of specific physiological needs of the ovarian cell such as stimulation of the biosynthesis of different classes of steroids with unique functions at different phases of the ovarian cycle, ovulation, differentiation of granulosa cells to luteal cells, stimulation of VEGF production to support angiogenesis and regression of the corpus luteum to end of each cycle. These processes are essential for successful reproduction. The regulatory mechanisms to maintain the appropriate levels of LHR at different phases of the ovarian cycle appear to be programmed exquisitely to meet these challenges. Thus, the post-transcriptional mechanism that we have described here is an efficient means to fine tune the expression levels of LHR mRNA and its translation product, LHR in the ovary to transduce appropriate signals that are necessary to carry out its vital functions crucial for successful reproduction.

References

- Baker TG (1972) Oogenesis and ovulation. Cambridge University Press, Cambridge
- Bernstein PL, Herrick DJ, Prokipcak RD, Ross J (1992) Control of c-myc mRNA half-life in vitro by a protein capable of binding to a coding region stability determinant. Genes Dev 6:642–654
- Bjersing L (1968) On the morphology and endocrine function of granulosa cells in ovarian follicles and corpora lutea. Biochemical, histochemical, and ultrastructural studies on the porcine ovary with special reference to steroid hormone synthesis. Acta Endocrinol (Copenh) Suppl 125:1–23
- Channing CP, Schaerf FW, Anderson LD, Tsafriri A (1980) Ovarian follicular and luteal physiology. Int Rev Physiol 22:117–201
- Dever TE, Green R (2012) The elongation, termination, and recycling phases of translation in eukaryotes. Cold Spring Harb Perspect Biol 4:a013706. doi:10.1101/cshperspect.a013706
- Elfgang C, Rosorius O, Hofer L, Jaksche H, Hauber J, Bevec D (1999) Evidence for specific nucleocytoplasmic transport pathways used by leucine-rich nuclear export signals. Proc Natl Acad Sci U S A 96:6229–6234
- Garneau NL, Wilusz J, Wilusz CJ (2007) The highways and byways of mRNA decay. Nat Rev Mol Cell Biol 8:113–126. doi:10.1038/nrm2104
- Gregio AP, Cano VP, Avaca JS, Valentini SR, Zanelli CF (2009) eIF5A has a function in the elongation step of translation in yeast. Biochem Biophys Res Commun 380:785–790. doi:10.1016/j. bbrc.2009.01.148
- Gutierrez E, Shin BS, Woolstenhulme CJ, Kim JR, Saini P, Buskirk AR, Dever TE (2013) eIF5A promotes translation of polyproline motifs. Mol Cell 51:35–45. doi:10.1016/j. molcel.2013.04.021
- Hentze MW (1994) Enzymes as RNA-binding proteins: a role for (di)nucleotide-binding domains? Trends Biochem Sci 19:101–103
- Hershey JW (1991) Translational control in mammalian cells. Annu Rev Biochem 60:717–755. doi:10.1146/annurev.bi.60.070191.003441
- Hirshfield AN (1991) Development of follicles in the mammalian ovary. Int Rev Cytol 124:43-101
- Hoffman YM, Peegel H, Sprock MJ, Zhang QY, Menon KMJ (1991) Evidence that human chorionic gonadotropin/luteinizing hormone receptor down-regulation involves decreased levels of receptor messenger ribonucleic acid. Endocrinology 128:388–393. doi:10.1210/endo-128-1-388

- Hunzicker-Dunn M, Jungmann R, Derda D, Birnbaumer L (1979) LH-induced desensitization of the adenylyl cyclase system in ovarian follicles. Adv Exp Med Biol 112:27–44
- Ikeda S, Nakamura K, Kogure K, Omori Y, Yamashita S, Kubota K, Mizutani T, Miyamoto K, Minegishi T (2008) Effect of estrogen on the expression of luteinizing hormone-human chorionic gonadotropin receptor messenger ribonucleic acid in cultured rat granulosa cells. Endocrinology 149:1524–1533. doi:10.1210/en.2007-1163
- Jacobson A, Peltz SW (1996) Interrelationships of the pathways of mRNA decay and translation in eukaryotic cells. Annu Rev Biochem 65:693–739. doi:10.1146/annurev. bi.65.070196.003401
- Kash JC, Menon KMJ (1998) Identification of a hormonally regulated luteinizing hormone/human chorionic gonadotropin receptor mRNA binding protein. Increased mRNA binding during receptor down-regulation. J Biol Chem 273:10658–10664
- Kash JC, Menon KMJ (1999) Sequence-specific binding of a hormonally regulated mRNA binding protein to cytidine-rich sequences in the lutropin receptor open reading frame. Biochemistry 38:16889–16897
- Kozak M (1989) The scanning model for translation: an update. J Cell Biol 108:229-241
- Krutzfeldt J, Rajewsky N, Braich R, Rajeev KG, Tuschl T, Manoharan M, Stoffel M (2005) Silencing of microRNAs in vivo with 'antagomirs'. Nature 438:685–689. doi:10.1038/ nature04303
- LaPolt PS, Oikawa M, Jia XC, Dargan C, Hsueh AJ (1990) Gonadotropin-induced up- and downregulation of rat ovarian LH receptor message levels during follicular growth, ovulation and luteinization. Endocrinology 126:3277–3279. doi:10.1210/endo-126-6-3277
- Loosfelt H, Misrahi M, Atger M, Salesse R, Vu Hai-Luu Thi MT, Jolivet A, Guiochon-Mantel A, Sar S, Jallal B, Garnier J et al (1989) Cloning and sequencing of porcine LH-hCG receptor cDNA: variants lacking transmembrane domain. Science 245:525–528
- Lu DL, Peegel H, Mosier SM, Menon KMJ (1993) Loss of lutropin/human choriogonadotropin receptor messenger ribonucleic acid during ligand-induced down-regulation occurs post transcriptionally. Endocrinology 132:235–240. doi:10.1210/endo.132.1.8419125
- Marsh JM, Butcher RW, Savard K, Sutherland EW (1966) The stimulatory effect of luteinizing hormone on adenosine 3',5'-monophosphate accumulation in corpus luteum slices. J Biol Chem 241:5436–5440
- Mathews M, Soenberg N, Hershey J (2000) In: Sonenberg N, Hershey J, Mathews M (eds) Translational control of gene expression. Cold Spring Harbor Laboratory Press, New York, pp 1–32
- McFarland KC, Sprengel R, Phillips HS, Kohler M, Rosemblit N, Nikolics K, Segaloff DL, Seeburg PH (1989) Lutropin-choriogonadotropin receptor: an unusual member of the G protein-coupled receptor family. Science 245:494–499
- Menon KMJ, Gunaga KP (1974) Role of cyclic AMP in reproductive processes. Fertil Steril 25:732–750
- Menon KMJ, Munshi UM, Clouser CL, Nair AK (2004) Regulation of luteinizing hormone/human chorionic gonadotropin receptor expression: a perspective. Biol Reprod 70:861–866. doi:10.1095/biolreprod.103.024471
- Menon KMJ, Clouser CL, Nair AK (2005) Gonadotropin receptors: role of post-translational modifications and post-transcriptional regulation. Endocrine 26:249–257. doi:10.1385/ ENDO:26:3:249
- Menon B, Peegel H, Menon KMJ (2009) Evidence for the association of luteinizing hormone receptor mRNA-binding protein with the translating ribosomes during receptor downregulation. Biochim Biophys Acta 1793:1787–1794. doi:10.1016/j.bbamcr.2009.08.009
- Menon B, Franzo-Romain M, Damanpour S, Menon KMJ (2011) Luteinizing hormone receptor mRNA down-regulation is mediated through ERK-dependent induction of RNA binding protein. Mol Endocrinol 25:282–290. doi:10.1210/me.2010-0366
- Menon B, Sinden J, Menon KMJ (2013a) Association of luteinizing hormone receptor (LHR) mRNA with its binding protein leads to decapping and degradation of the mRNA in the p bodies. Biochim Biophys Acta 1833:1173–1179. doi:10.1016/j.bbamcr.2013.01.024

- Menon B, Sinden J, Franzo-Romain M, Botta RB, Menon KMJ (2013b) Regulation of LH receptor mRNA binding protein by miR-122 in rat ovaries. Endocrinology 154:4826–4834. doi:10.1210/ en.2013-1619
- Menon B, Gulappa T, Menon KMJ (2014a) Eukaryotic initiation factor 5A plays an essential role in luteinizing hormone receptor regulation. Mol Endocrinol 28:1796–1806. doi:10.1210/ me.2014-1132
- Menon B, Gulappa TG, Menon KM (2014) microRNA-122 plays a regulatory role in the expression of luteinizing hormone (LH) receptor in the ovary. In: 47th annual meeting - society for the study of reproduction, Grand Rapids
- Miller WL (2008) Steroidogenic enzymes. Endocr Dev 13:1-18. doi:10.1159/000134751
- Moghul A, Lin L, Beedle A, Kanbour-Shakir A, DeFrances MC, Liu Y, Zarnegar R (1994) Modulation of c-MET proto-oncogene (HGF receptor) mRNA abundance by cytokines and hormones: evidence for rapid decay of the 8 kb c-MET transcript. Oncogene 9:2045–2052
- Nair AK, Menon KMJ (2004) Isolation and characterization of a novel trans-factor for luteinizing hormone receptor mRNA from ovary. J Biol Chem 279:14937–14944. doi:10.1074/jbc. M309484200
- Nair AK, Kash JC, Peegel H, Menon KMJ (2002) Post-transcriptional regulation of luteinizing hormone receptor mRNA in the ovary by a novel mRNA-binding protein. J Biol Chem 277:21468–21473. doi:10.1074/jbc.M111653200
- Olivas W, Parker R (2000) The Puf3 protein is a transcript-specific regulator of mRNA degradation in yeast. EMBO J 19:6602–6611. doi:10.1093/emboj/19.23.6602
- Pachter JS, Yen TJ, Cleveland DW (1987) Autoregulation of tubulin expression is achieved through specific degradation of polysomal tubulin mRNAs. Cell 51:283–292
- Park MH, Wolff EC, Folk JE (1993) Hypusine: its post-translational formation in eukaryotic initiation factor 5A and its potential role in cellular regulation. Biofactors 4:95–104
- Park MH, Nishimura K, Zanelli CF, Valentini SR (2010) Functional significance of eIF5A and its hypusine modification in eukaryotes. Amino Acids 38:491–500. doi:10.1007/ s00726-009-0408-7
- Peegel H, Randolph J Jr, Midgley AR, Menon KMJ (1994) In situ hybridization of luteinizing hormone/human chorionic gonadotropin receptor messenger ribonucleic acid during hormoneinduced down-regulation and the subsequent recovery in rat corpus luteum. Endocrinology 135:1044–1051. doi:10.1210/endo.135.3.8070346
- Peegel H, Towns R, Nair A, Menon KMJ (2005) A novel mechanism for the modulation of luteinizing hormone receptor mRNA expression in the rat ovary. Mol Cell Endocrinol 233:65–72. doi:10.1016/j.mce.2004.12.009
- Perez-Ortin JE, Alepuz P, Chavez S, Choder M (2013) Eukaryotic mRNA decay: methodologies, pathways, and links to other stages of gene expression. J Mol Biol 425:3750–3775. doi:10.1016/j.jmb.2013.02.029
- Port JD, Huang LY, Malbon CC (1992) Beta-adrenergic agonists that down-regulate receptor mRNA up-regulate a M(r) 35,000 protein(s) that selectively binds to beta-adrenergic receptor mRNAs. J Biol Chem 267:24103–24108
- Ross J (1993) RNA-protein interaction protocols. Ed. Haynes S. R. (Humana Press, Totowa, NJ), pp 459–476
- Ross J (1995) mRNA stability in mammalian cells. Microbiol Rev 59:423-450
- Rossi D, Galvao FC, Bellato HM, Boldrin PE, Andrews BJ, Valentini SR, Zanelli CF (2014) eIF5A has a function in the cotranslational translocation of proteins into the ER. Amino Acids 46:645–653. doi:10.1007/s00726-013-1618-6
- Ruhl M, Himmelspach M, Bahr GM, Hammerschmid F, Jaksche H, Wolff B, Aschauer H, Farrington GK, Probst H, Bevec D et al (1993) Eukaryotic initiation factor 5A is a cellular target of the human immunodeficiency virus type 1 Rev activation domain mediating transactivation. J Cell Biol 123:1309–1320
- Sachs AB (1993) Messenger RNA degradation in eukaryotes. Cell 74:413-421
- Saini P, Eyler DE, Green R, Dever TE (2009) Hypusine-containing protein eIF5A promotes translation elongation. Nature 459:118–121. doi:10.1038/nature08034

- Schrader R, Young C, Kozian D, Hoffmann R, Lottspeich F (2006) Temperature-sensitive eIF5A mutant accumulates transcripts targeted to the nonsense-mediated decay pathway. J Biol Chem 281:35336–35346. doi:10.1074/jbc.M601460200
- Segaloff DL, Wang HY, Richards JS (1990) Hormonal regulation of luteinizing hormone/chorionic gonadotropin receptor mRNA in rat ovarian cells during follicular development and luteinization. Mol Endocrinol 4:1856–1865. doi:10.1210/mend-4-12-1856
- Shyu AB, Greenberg ME, Belasco JG (1989) The c-fos transcript is targeted for rapid decay by two distinct mRNA degradation pathways. Genes Dev 3:60–72
- Tharun S, He W, Mayes AE, Lennertz P, Beggs JD, Parker R (2000) Yeast Sm-like proteins function in mRNA decapping and decay. Nature 404:515–518. doi:10.1038/35006676
- Wang L, Menon KMJ (2005) Regulation of luteinizing hormone/chorionic gonadotropin receptor messenger ribonucleic acid expression in the rat ovary: relationship to cholesterol metabolism. Endocrinology 146:423–431. doi:10.1210/en.2004-0805
- Wang L, Gulappa T, Menon KMJ (2010) Identification and characterization of proteins that selectively interact with the LHR mRNA binding protein (LRBP) in rat ovaries. Biochim Biophys Acta 1803:591–597. doi:10.1016/j.bbamcr.2010.02.003
- Xu YH, Grabowski GA (1999) Molecular cloning and characterization of a translational inhibitory protein that binds to coding sequences of human acid beta-glucosidase and other mRNAs. Mol Genet Metab 68:441–454. doi:10.1006/mgme.1999.2934
- Xu YH, Busald C, Grabowski GA (2000) Reconstitution of TCP80/NF90 translation inhibition activity in insect cells. Mol Genet Metab 70:106–115. doi:10.1006/mgme.2000.3010
- Zanelli CF, Valentini SR (2007) Is there a role for eIF5A in translation? Amino Acids 33:351–358. doi:10.1007/s00726-007-0533-0
- Zeleznik AJ (2004) The physiology of follicle selection. Reprod Biol Endocrinol 2:31. doi:10.1186/1477-7827-2-31
- Zhou T, Daugherty M, Grishin NV, Osterman AL, Zhang H (2000) Structure and mechanism of homoserine kinase: prototype for the GHMP kinase superfamily. Structure 8:1247–1257
- Zuk D, Jacobson A (1998) A single amino acid substitution in yeast eIF-5A results in mRNA stabilization. EMBO J 17:2914–2925. doi:10.1093/emboj/17.10.2914

Post-transcriptional Regulation of Steroid Hormone Receptors

5

Nancy H. Ing

1 Introduction

Genome studies of diverse species indicate that mammalian physiology is unique not because of greater numbers of genes, but rather because of greater complexity of gene regulation (Mattick 2001; Maniatis and Reed 2002). Post-transcriptional regulation of gene expression (including mRNA stability and translation) is one of the ways that organisms control and modify the flow of genetic information into the proteome (Keene and Tenenbaum 2002). Within the continuum of gene expression from transcription to protein degradation, regulated mRNA stability is increasingly being recognized as a major effector of gene regulation.

The first discoveries of steroid hormone effects on the stabilities of specific mRNAs were made in highly responsive animal tissues that were initiating large-scale production of new proteins. Early studies determined that estrogen regulated stabilities of mRNAs in the livers of egg-laying animals at the initiation of oogenesis (McKnight and Palmiter 1979; Dodson and Shapiro 2002). In one example, estrogen stabilized the vitellogenin mRNA in frog liver by 40-fold, increasing the half-life of the mRNA from 16 to 600 h. While estrogen stabilized mRNAs encoding egg proteins, it also destabilized mRNAs encoding serum proteins in the frog liver.

The rate of mRNA degradation has been established as equally important as the rate of synthesis in regulating the steady state concentration of the mRNA (Hargrove et al. 1991). The average mRNA half-life in mammalian cells is 24 h, with short-lived messages such as c-fos (*FOS*) mRNA having 20 min long half-lives and long-lived RNAs, such as 28S and 18S rRNAs, having 4 day long half-lives (Hargrove and Schmidt 1989). The half-lives of mRNAs are important in determining how long each mRNA is present to act as a template for translation of proteins.

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_5

N.H. Ing (🖂)

Department of Animal Science, Texas A&M University, 2471 TAMU, College Station, TX 77843-2471, USA e-mail: ning@cvm.tamu.edu

[©] Springer International Publishing Switzerland 2016

In addition, the half-life of the mRNA also controls how rapidly its steady state concentrations can be altered. Expression of genes with short half-lived mRNAs are rapidly regulated by changes in transcription and degradation rates, while long-lived mRNAs may take days to achieve a new steady state level in response to a cell signal such as a steroid hormone. When comparing transcriptional and post-transcriptional up-regulation of gene expression, mRNA stabilization may be advantageous to organisms because it lacks the lag phase and energetic costs that occur for responses that increase the rates of gene transcription.

Current reports elucidate post-transcriptional regulation that finely tunes the expression of steroid hormone receptor genes in animal tissues that are healthy or changing with the progression of a disease (Klinge 2012). This review focuses on post-transcriptional regulation of mRNAs encoding the nuclear steroid hormone receptors.

2 RNA Sequences and Binding Factors that Regulate mRNA Stability

Messenger RNAs are composed of three, functionally diverse regions: the 5' untranslated region (UTR), the coding sequence (cds) and the 3'UTR (Fig. 5.1). The 5'UTR is usually short (100–300 bases long). The function of the 5'UTR is to be scanned by ribosomes, which identify an optimal Kozak sequence for the initiation of translation at the AUG start codon. The length of the cds relates to the length of the protein encoded by the mRNA. Ribosomes initiate translation at the AUG start codon and proceed to translate the cds of the mRNA until they reach a stop codon. The 3'UTRs are highly variable in length between different mRNAs. For example, the length of the 3'UTR of the *ESR2* "variant a" mRNA is 108 bases while the 3'UTR of the *PGR* mRNA is 9492 bases long (Fig. 5.1). The function of the 3'UTR is regulation of mRNA stability. The 3'UTRs of mRNAs are unique in that they maintain stable structures with RNA folding and binding proteins that are not disrupted by the movement of ribosomes.

The 3'UTR sequences of mRNAs are predicted to form stable structures with algorithms such as MFOLD (Zuker 2003). For example, the folded structure of the 4307 base long 3'UTR of *ESR1* mRNA (depicted in Fig. 5.2) has numerous single stranded regions that are susceptible to cleavage by common ribonucleases such as RNase A. Cleavage at any position in the 3'UTR separates the 5'UTR and cds from the polyA tail ("A_n" in Figs. 5.1 and 5.2) and results in rapid degradation of the mRNA (Bevilacqua et al. 2003). In the example of estrogen up-regulation of vitellogenin gene expression in frog liver, estrogen treatment lowers the rate of the cleavage of vitellogenin mRNA are fairly well-conserved across related species, although insertions and deletions of sequence are well tolerated (Mitchell and Ing 2003; Manjithaya and Dighe 2004). The 3'UTR sequences, structures, and binding molecules (proteins and microRNAs) confer its function: regulating the lifespan of the mRNA.

As in the case of mRNA synthesis (transcription) in the nucleus, rates of mRNA degradation in the cytoplasm are regulated by finite sequence elements in the 3'UTR and the proteins that bind them (Guhaniyogi and Brewer 2001). The most well characterized mRNA elements are AU-Rich Elements (AREs). These were first identified as instability elements in the 3'UTRs of oncogene and cytokine mRNAs, e.g. those encoding c-fos (*FOS*) and tumor necrosis factor (*TNF*; Zhang et al. 2002; Bevilacqua et al. 2003). There are four distinct classes of AREs (Stellato 2004). The simplest is individual AUUUA elements scattered in 3'UTR sequences, as in follicle-stimulating hormone β -subunit (*FSHB*) and epidermal growth factor receptor (*EGFR*) mRNAs (Balmer et al. 2001; Shim and Karin 2002; Manjithaya and Dighe 2004). More complex AREs have arrays of several AUUUA elements in nearby regions of the 3'UTR, such as those in many oncogene mRNAs. Other sequence elements involved in stability have been identified, such as C-rich elements on the 3'UTR of the *LHCGR* mRNA encoding the luteinizing hormone receptor (Menon et al. 2004; Ing 2005).

There are multiple families of proteins that bind AREs, including AU-binding factors (e.g. *HNRPD* gene products, also known as AUF1), zinc-finger proteins (e.g. tristetraprolin, the *ZFP36* gene product) and relatives of the Drosophila RNA-binding ELAV protein (e.g. ELAV1 which is also known as HuR; Staton et al. 2000;



Fig. 5.1 A diagram of the human mRNAs encoding steroid hormone receptors. The mRNAs encoding estrogen receptor α (*NR3A1* or *ESR1*), estrogen receptor β (*NR3A2* or *ESR2*), glucocorticoid receptors α and β (*NR3C1* or *GR*), mineralocorticoid receptor (*NR3C2* or *MR*), progesterone receptor (*NR3C3* or *PGR*) and androgen receptor (*NR3C4* or *AR*) are diagrammed. The 5' untranslated regions ("5'UTR") are shown as *narrow black lines*, the coding sequences ("cds") are *solid blue boxes*, the 3'untranslated regions ("3'UTR") are *red patterned boxes*, and the polyA tail is indicated by "A_n". For the different 3'UTRs in the variant ("var.") mRNAs of *ESR2* and *GR*, the different patterns in the 3'UTR boxes indicate unique sequences. The lengths of the 3'UTRs is indicated in bases ("b") above the 3'UTR box. The information is from the reference mRNA sequences from GenBank: NM_000125.3 (*ESR1*); NM_001437.2, NM_001040275.1, NM_001214902.1, NM_001271876.1 (*ESR2*); NM_000176.2, NM_001020825.1 (*GR*); NM_000901 (*MR*); NM_000926.4 (*PGR*); NM_00044.3 (*AR*)



Fig. 5.2 Predicted structure in the 3'UTR of *ESR1* mRNA. The MFOLD algorithm predicted the folded structure shown for the 5' end of the 3'UTR (1330 bases) of the human *ESR1* mRNA. The end at *left* joins the coding sequence ("cds") while the end at *right* joins the rest of the 3'UTR and the polyA tail ("An"). The free energy of the predicted structure is -474 kcal/mol

Shim and Karin 2002). While some ARE-binding proteins (such as tristetraprolin) destabilize the ARE-bearing mRNA, others (such as HuR) have stabilizing effects. The *HNRPD* gene produces four AUF1 proteins AUF1p37, AUF1p40, AUF1p42 and AUF1p45 by alternative splicing of the mRNA. All of the proteins bind to AREs. However, AUF1p37 and AUF1p42 destabilize mRNAs while AUF1p40 and AUF1p45 stabilize the mRNAs they bind (Wagner et al. 1998; Loflin et al. 1999; Tolnay et al. 1999; Chen et al. 2001; Mukherjee et al. 2002; Sarkar et al. 2003; Lal et al. 2006). Most RNA binding proteins are ubiquitously expressed although there is some degree of tissue-specificity in the distribution of some of them (Brewer 2001; Hollams et al. 2002; Zhang et al. 2002; Lu and Schneider 2004). Interestingly, the expression of some genes encoding ARE-binding proteins are regulated by steroid hormones via their receptor proteins (Cuadrado et al. 2003; Arao et al. 2002, 2004; Ing 2010).

While some post-transcriptional regulation of mRNA stability is the result of RNA binding proteins (either stabilizing or destabilizing), other post-transcriptional regulation is performed by non-coding regulatory RNAs such as microRNAs. MicroRNAs are short (about 22 bases) single-stranded, non-coding RNAs that regulate expression of protein encoding genes, usually by destabilizing the mRNA or repressing translation. The microRNAs that effect the changes in gene expression of the greatest magnitude usually do both (Baek et al. 2008). MicroRNAs exist in large complexes with proteins, called RNA-Induced Silencing Complexes or RISC. Within the RISC, microRNAs are very stable even in bodily fluids, such as serum, that contain high levels of RNAses. MicroRNAs act by binding their 5' seed sequence to a complementary sequence within the 3'UTR of the mRNA, which directs the RISC complex. Subsequently, the mRNA is degraded in the exosome. While there are only approximately a thousand microRNA genes in the human genome, each individual microRNA can regulate the expression of hundreds of protein-encoding genes. It is estimated that microRNAs regulate more than a third of human genes (Ing et al. 2009). Therefore, it is not surprising to note that microRNAs regulate the expression of genes encoding the steroid hormone receptors. For discussion of microRNAs regulating steroid hormone action by their effects on nuclear receptor cofactors and kinase signaling cascades, as well as regulation of the expression of microRNA genes by the steroid hormones via the steroid hormone receptor proteins, see current reviews (Ing 2009; Pandey and Picard 2010; Klinge 2012; Manavathi et al. 2013).

3 Autoregulation of the Stabilities of Messenger RNAs Encoding Steroid Hormone Receptors

Estrogen is a steroid hormone that autoregulates the stability of *ESR1* mRNA, which encodes its dominant, active receptor, estrogen receptor α (ER α), in most tissues of the body. Estrogens stabilize *ESR1* mRNA in fish liver during the initiation of oogenesis and in mammalian endometrium during the preovulatory surge of estrogen (Flouriot et al. 1996; Mitchell and Ing 2003; Boyce-Derricott et al. 2010). In both cases, the stabilization and resultant up-regulation of *ESR1* mRNA are dependent upon ER α protein, because ER α antagonists block the effect. The conservation of this estrogen action across these diverse species implies that it is an important mechanism for augmenting further estrogen responses. In sheep endometrium, a single physiological dose of estradiol up-regulates *ESR1* mRNA concentrations fivefold in 24 h, during which time there is no increase in the transcription rate of the *ESR1* gene (Ing et al. 1996). The primary mechanism of *ESR1* mRNA up-regulation, by increasing *ESR1* mRNA stability, was directly demonstrated in vivo with pulse-chase labeling and ex vivo using explants cultured with a transcription inhibitor (Ing and Ott 1999).

The molecular mechanism by which estrogen stabilizes *ESR1* mRNA was discovered using the sheep uterus as a model system. Human and sheep *ESR1* mRNAs carry 14 and 10 ARE elements in their respective 3'UTRs. (There are no ARE elements in the 5'UTRs or cds regions of human or sheep *ESR1* mRNAs.) The AREs are probably responsible for the inherent instability of the *ESR1* mRNA
and of the heterologous mRNAs that have the ESR1 3'UTR sequence transferred to them (Mitchell and Ing 2003; Kenealy et al. 2000). Using an in vitro stability assay developed with cytoplasmic extracts from endometrial samples from control (ovariectomized) and ovariectomized, estradiol-treated ewes, two discrete (82 bases long) Minimal Estradiol-Modulated Stability Sequences (MEMSSs) were identified in the vast (4366 bases long) 3'UTR of the sheep ESR1 mRNA (Mitchell and Ing 2003). These MEMSSs conferred estradiol-enhanced stability to heterologous mRNAs when they were transferred to them. UV-crosslinking identified several endometrial proteins that were induced to bind the MEMSS by estradiol treatment of ewes. One predominant MEMMS binding protein is AUF1p45, one of four protein products of the *HRNPD* gene (Ing et al. 2008). The three other estradiol-induced MEMSS binding proteins of 27, 34 and 70 kDa sizes remain to be identified. When recombinant AUF1p45 fusion protein was produced and purified from bacteria, it stabilized ESR1 mRNA in vitro in the presence of endometrial extracts from ovariectomized ewes. Similar analyses with recombinant HuR, a 34 kDa RNA stabilizing protein that binds AREs, did not stabilize *ESR1* mRNA in vitro (Ing et al. 2008). Intriguingly, estradiol treatment of ewes increased concentrations of AUF1p45 mRNA and protein apparently by stabilizing the AUF1p45 mRNA (Ing 2010). These data lead to the model of the molecular mechanism of estrogen up-regulation of *ESR1* gene expression in Fig. 5.3.

In contrast, estradiol treatment has also been reported to destabilize *ESR1* mRNA in the MCF7 breast cancer cell line (Saceda et al. 1998). The explanation of these contrasting effects of estrogens may become clear when the molecular mechanism(s) that regulate *ESR1* mRNA stability are further investigated.

Estrogens also regulate gene expression by binding estrogen receptor β , which is encoded by the distinct *ESR2* gene. There are numerous variant *ESR2* mRNAs. These



Fig. 5.3 A molecular model for estrogen induction of a stabilizing ribonucleoprotein complex on the 3'UTR of *ESR1* mRNA in the sheep uterus. The cartoon of sheep *ESR1* mRNA shows the cds as a *blue box* that is 5' to the extensive 3'UTR (4366 bases long). Within the 3'UTR are two 82 base long Minimal Estrogen Modulated Stability Sequences ("MEMSSs"), shown as the hairpin structures predicted by MFOLD. Each carries a common 10 base long U-Rich Element ("URE") on the loop of the hairpin. At *left*, in the absence of estrogen, the long 3'UTR is susceptible to cleavage by ribonuclease ("RNase"). However, estrogen treatment induces the formation of a stabilizing protein complex that includes AUF1p45 ("p45") on the MEMSS, which protects the 3'UTR of *ESR1* mRNA from cleavage by RNase (*right panel*)

result from usage of three promoters on the *ESR2* gene, alternative splicing in 5' and 3' UTRs and alternative polyA signal usage. Three protein isoforms (estrogen receptor β -1, -2, and -5) are produced. They vary mainly at a small region of the C-terminus. Representative *ESR2* mRNAs with 3' UTR differences are shown in Fig. 5.1 (from human reference sequences in GenBank). In sharp contrast to the *ESR1* mRNA, the 3' UTRs of *ESR2* mRNAs are very short. It is interesting to note that splicing within the 3'UTR near the stop codon can trigger nonsense mediated mRNA decay (Ni et al. 2007; McGlincy and Smith 2008; Saltzman et al. 2008). Studies of *ESR2* gene expression in normal breast and prostate and cancers derived from those tissues have focused on different 5' UTR sequences and their effects on translational efficiencies (Smith et al. 2010; Lee et al. 2013). However, the dissimilar 3'UTRs interact with 5'UTR sequences to alter translational efficiencies. Estradiol treatment appears to inhibit translational efficiencies of two of 15 human *ESR2* mRNA variants (Smith et al. 2010). However, estradiol treatment of rainbow trout does not alter the stabilities of *ESR2* mRNAs in liver (Boyce-Derricott et al. 2010).

The glucocorticoid receptor proteins GR α and GR β are encoded by the same *NR3C1* gene. The mRNAs that encode GR α and GR β differ by alternative splicing of the last exon, exon 9 (Fig. 5.1). The GR α protein has 50 amino acids encoded by exon 9a prior to the stop codon, while GR β has only 15 amino acids encoded in exon 9b (Schaaf and Cidlowski 2002). The shorter C-terminus of GR^β does not allow glucocorticoid binding, but GR^β regulates transcription of distinct genes from GRa. GR^β also dimerizes with GR α to act as a dominant negative factor (Vandevyver et al. 2014). The transcripts that encode GR α and GR β have unique 3'UTRs that carry 10 and 4 type I AREs (AUUUA sequence elements), respectively (Schaaf and Cidlowski 2002; Stellato 2004). Mutation of all ten AUUUA elements to GUUUA in the mRNA encoding GRa stabilizes the mRNA and increases GRa protein levels in transfected COS-1 cells (Schaaf and Cidlowski 2002). Similar effects occur with mutation of just one AUUUA of the four in the 3'UTR of the mRNA encoding GR β . It is fascinating to note that the mutation occurs naturally in humans, is associated with rheumatoid arthritis, and likely contributes to glucocorticoid resistance (Derijk et al. 2001). While glucocorticoids transcriptionally repress the NR3C1 gene, they also rapidly destabilize NR3C1 mRNA in many cell types (Schaaf and Cidlowski 2003; Ramamoorthy and Cidlowski 2013). Ligand activated GRs are known to act via AREs and kinase pathways to down-regulate a number of genes involved in inflammation (Stellato 2004). There is at least one report that implies that dexamethasone treatment of rats decreases GR mRNA in muscle concurrent with increasing AUF1 mRNA (Sato et al. 2011). However, in that study, $GR\alpha$ and $GR\beta$ and the four AUF1 transcripts (two encode proteins that destabilize mRNAs while the other two encode mRNA stabilizing proteins) were not distinguished from each other.

The *NR3C2* gene encodes the mineralocorticoid receptor (MR). Although the *MR* mRNA has a long 3'UTR (2597 bases long, Fig. 5.1) that carries eight AREs (AUUUA elements determined using the AREsite 1.0 algorithm; Gruber et al. 2011), very little has been published about its regulated stability. There is an indication that restraint stress of rats down-regulates *MR* heteronuclear RNA in nuclei of the hippocampus within 2 h (Herman and Watson 1995). In that study, decreased levels of

MR mRNA in the cytoplasm were not detected. However, the technique used was in situ hybridization, which is considered to be semiquantitative. In addition, the in situ hybridization probes could bind partially degraded *MR* mRNA as well as the full length mRNA, so perhaps degradation was not complete within the 2 h time frame.

The *NR3C3* gene encodes progesterone receptors A and B (PGR-A and PGR-B). Their mRNAs are similar to *ESR1* mRNA in that they carry a very long 3'UTR (9492 bases, Fig. 5.1). The 3'UTR of *PGR* mRNA bears numerous AREs: 28 AUUUA elements, with all but three conserved across mammalian species (AREsite 1.0 algorithm). The mRNAs that encode PGR-A and PGR-B differ in 5' sequence because the transcription of *PGR-B* mRNA initiates at a site upstream of the *PGR-A* mRNA transcription initiation site. The PGR-B protein is longer and has a more efficient transcriptional activation domain than PGR-A. In contrast to the bidirectional autoregulation of *ESR1* mRNA by estrogen, progestins have only been reported to stabilize *PGR* mRNA (Tseng and Zhu 1997). The progestin medroxyprogesterone acetate increased the *PGR* mRNA half-life from 6 to 12 h in primary cultures of stromal cells from human endometrium (Saceda et al. 1998).

The *NR3C4* gene has a single transcript that encodes androgen receptor (AR). Androgens autoregulate the stability of *AR* mRNA in both positive and negative directions (Yeap et al. 1999, 2004). The direction (up or down) of regulation depends upon dose of androgen and the tissue or cell line examined. For instance, androgens stabilize *AR* mRNA in a prostate cancer cell line but destabilize it in the MDA-453 breast cancer cell line. The extensive 3'UTR of *AR* mRNA (6783 bases long, Fig. 5.1) does not carry many AREs (only five AUUUAs, AREsite1.0 algorithm). However, the 3'UTR of *AR* mRNA bears a UC-rich element that is conserved across species. This *AR* mRNA element binds a poly(C)-binding protein as well as HuR (Yeap et al. 1999, 2004). These two proteins cooperate in a cell-type-specific manner to determine whether androgen stimulation stabilizes or destabilizes the *AR* mRNA (Yeap et al. 2004). In prostate cancer cell lines, the ErbB3 binding protein 1 (EBP1) destabilized *AR* mRNA by binding the same UC-rich element in the 3'UTR (Zhou et al. 2010).

4 microRNA Regulation of Messenger RNAs Encoding Steroid Hormone Receptors

The best studied gene encoding a steroid hormone receptor that is regulated by microRNAs is the *ESR1* gene (Ing 2009; Pandey and Picard 2010). One reason for this is that ER α is critical to *PGR* gene expression: so microRNAs that down-regulate the expression of the *ESR1* gene will also down-regulate the *PGR* gene. Other reasons are that *ESR1* gene expression is critical to many aspects of mammalian health and may be dysregulated in disease (Klinge 2012). And finally, the 3'UTR of *ESR1* mRNA is long and is replete with 65 putative binding sites for microRNAs predicted by algorithms such as TargetScan (Pandey and Picard 2010). There are 13 different microRNA binding sites on the 3'UTR of *ESR1* mRNA that have been validated in various model systems (Table 5.1; Pandey and Picard 2010; Manavathi et al. 2013; Klinge 2012; and references therein).

Gene name	3'UTR	miR sites	References		
NR3A1/ESR1	4307 b	Let-7	Manavathi et al. (2013)		
		miR-137	Zhao et al. (2008)		
		miR-145	Zhao et al. (2008)		
		miR-17-5p	Zhao et al. (2008)		
		miR-18a/19b/20b	Manavathi et al. (2013)		
		miR-181	Pandey and Picard (2010)		
		miR-19	Pandey and Picard (2010)		
		miR-193b	Klinge (2012)		
		miR-206	Klinge (2012); Manavathi et al. (2013)		
		miR-221/222	Pandey and Picard (2010); Klinge (2012); Manavathi et al. (2013)		
		miR-22	Pandey and Picard (2010); Klinge (2012); Manavathi et al. (2013)		
		miR 26ab/1297	Pandey and Picard (2010)		
		miR-302	Klinge (2012)		
NR3A2/ESR2	871 b	miR-92	Al-Nakhle et al. (2010)		
NR3C1/GRα	3985 b	miR-101	Riester et al. (2012)		
		miR-130b	Tessel et al. (2011)		
		miR-124/506	Vreugdenhil et al. (2009); Ledderose et al. (2012)		
		miR-142-3p	Riester et al. (2012); Lv et al. (2012)		
		miR-18a	Uchida et al. (2008); Vreugdenhil et al. (2009)		
		miR-433	Riester et al. (2012)		
		miR-96	Riester et al. (2012)		
NR3C2/MR	2597 b	miR-124/506	Sõber et al. (2010)		
		miR-135a	Sõber et al. (2010)		
NR3C3/PGR	9492 b	miR-126-3p	Cui et al. (2011)		
		miR-181	Maillot et al. (2009)		
		miR-200a	Haraguchi et al. (2014)		
		miR-26ab/197	Maillot et al. (2009)		
NR3C4/AR	6783 b	miR-135b	Östling et al. (2011)		
		miR-185	Östling et al. (2011)		
		miR-297	Östling et al. (2011)		
		miR-299-3p	Östling et al. (2011)		
		miR-34 ac	Östling et al. (2011)		
		miR-371	Östling et al. (2011)		
		miR-421	Östling et al. (2011)		
		miR-449ab	Östling et al. (2011)		

 Table 5.1
 Validated miRNA target sites on 3'UTRs of mRNAs encoding human steroid hormone receptors

Length of the 3'UTR in bases ("b") is indicated for the mRNAs of the named genes. All sites are conserved across species except none of the 3'UTRs of the four variant mRNAs encoding ESR2 have any conserved sites (Pandey and Picard 2010; TargetScanHuman v6.2)

Several studies demonstrated that microRNAs regulate expression of the ESR1 gene in breast cancer. One study began with interest in microRNAs that had increased concentrations in ER α -negative breast tumors compared to ER α -positive breast tumors (Iorio et al. 2005). Among those microRNAs, the investigators focused on the few microRNAs that also had predicted binding sites within the 3'UTR of ESR1 mRNA (Adams et al. 2007). This lead to the discovery that experimentally decreasing the concentrations of miR-206 increased ESR1 gene expression in MCF7 cells. The mechanism by which miR-206 decreased ESR1 gene expression was by destabilization of ESR1 mRNA. Further studies demonstrated that estradiol treatment of breast cancer cells decreased concentrations of miR-206, leading to increased ESR1 mRNA stability and ER α protein concentrations. This describes a second mechanism by which estrogens post-transcriptionally up-regulate ESR1 gene expression, in addition to that described in the sheep uterus (Fig. 5.3). Another group searching for microR-NAs that regulate ESR1 gene expression identified 12 microRNAs that were more highly expressed in ERa-negative breast cancer cell lines compared to ERa-positive lines (Zhao et al. 2008). These investigators predicted that miR-221/222 would target the 3'UTR of ESR1 mRNA. By altering levels of the microRNA in the MCF7 and T47D breast cancer cell lines, they demonstrated that miR-221/222 decreased ERα protein concentrations but not ESR1 mRNA concentrations, indicating that this microRNA inhibits translation of ESR1 mRNA. In breast cancer patients who have received chemotherapy, high levels of miR-221 in plasma is predictive of chemoresistance (Zhao et al. 2008). Some have suggested that down-regulation of miR-221 with antagomiR oligonucleotide drugs may be a useful approach to treating breast cancer (Piva et al. 2013).

Studies of a human pancreatic cancer cell line lead to the discovery that treatment with curcumin (a naturally occurring flavonoid that inhibits cancer growth) increased miR-22 and decreased miR-199a-3p concentrations (Sun et al. 2008). Experimental up-regulation of miR-22 suppressed ER α protein concentrations along with those of Sp1, which is a transcription factor that cooperates with ER α to activate transcription on many gene promoters that lack conventional estrogenresponsive elements. Thus, miR-22 appears to inhibit ER α and ER α /Sp1 responsive gene promoters by reducing the concentrations of those transcription factors. These examples show that several different microRNAs regulate the expression of the *ESR1* gene in a variety of tumor cells.

As explained in the preceding section, the *ESR2* gene is distinct from that of *ESR1* and its variant mRNAs have short 3'UTR sequences. Therefore, it is not surprising that only one predicted microRNA site, miR-92, is reported in the literature for "*ESR2* variant a" mRNA (Table 5.1; Al-Nakhle et al. 2010). *ESR2* gene expression is down-regulated in many breast cancer tumors while miR-92 concentrations increased. In the MCF7 breast cancer cell line, anti-miR-92 down-regulated miR-92 while *ESR2* gene expression was upregulated (Al-Nakhle et al. 2010). Similar results occurred when the 3'UTR of *ESR2* mRNA was cloned 3' to a reporter gene.

Within the 3'UTR of the mRNA encoding GR α (3985 bases long; Fig. 5.1), TargetscanHuman v6.2 identified two miR-124/506 binding sites that were conserved across species. Several studies have implicated these and other microRNAs

in down-regulating $GR\alpha$ gene expression (Table 5.1). In the adrenal gland, adrenocorticotrophic hormone increased concentrations of miR-101a, miR-142-3p, miR-433, and miR-96 while $GR\alpha$ gene expression decreased (Riester et al. 2012). Similarly in T-cells from acute lymphoblastic leukemia patients, antagomirs to miR-142-3p up-regulated $GR\alpha$ gene expression (Lv et al. 2012). In T cells from patients with sepsis, glucocorticoid resistance was associated with decreased $GR\alpha$ gene expression and increased levels of miR-124 (Ledderose et al. 2012). In cell lines derived from multiple myelomas, miR-130b down-regulated $GR\alpha$ gene expression (Tessel et al. 2011). In rat brain, increased levels of miR-124/506 and miR-18a were associated with down-regulation of $GR\alpha$ gene expression (Uchida et al. 2008; Vreugdenhil et al. 2009). Although TargetScanHuman v6.2 identified two sites each for miR-181a and miR-4262 within the distinct 3'UTR of the mRNA encoding GR β (1433 bases long; Fig. 5.1), to my knowledge no sites have been validated.

The *MR* mRNA has a 2597 base long 3'UTR that bears several predicted microRNA binding sites that are conserved across vertebrate species (Söber et al. 2010). In HeLa cells transfected with a luciferase reporter carrying the *MR* mRNA 3'UTR, transfection with miR-124 and miR-135a expression plasmids down-regulated luciferase activity (Table 5.1). Intriguingly, the miR-124 site is homologous to that in the 3'UTR of the related *GRa* mRNA. The repression of luciferase activity by miR-124 and miR-135a appeared to be due to a repression of translation, as the luciferase mRNA level was not lowered similarly. However, transfection with the miR-135a expression vector had a slight trend toward down-regulating the concentrations of endogenous *MR* mRNA.

The 3'UTR of the mRNA encoding PGR-B is the longest of any of the mRNAs encoding steroid hormone receptors (9492 bases; Fig. 5.1). However, there are just four reports of regulation by microRNAs (Cochrane et al. 2012). TargetScanHuman v6.2 predicted only six conserved microRNA binding sites within the 3'UTR of PGR mRNA. Two of those, miR-181a and miR-26a, are down-regulated by estradiol treatment in breast cancer cell lines with simultaneous stimulation of proliferation (Table 5.1). When either miR-181a or miR-26a was transfected into MCF7 breast cancer cells, PGR gene expression was down-regulated at the levels of mRNA and protein concentrations with a concurrent decrease in proliferation (Maillot et al. 2009). In mice, it was noted that when proliferation of cervical epithelium was low, Pgr protein levels were low and miR-200a levels were high (Haraguchi et al. 2014). The 3'UTR of mouse Pgr mRNA is much shorter (3473 bases) than the human's. In the same report, the authors cloned the 3'UTR of the mRNA encoding human PGR (with and without the miR-200a binding site mutated) 3' to a luciferase reporter gene. When transfected into COS cells, derived from monkey kidney cells, miR-200a down-regulated reporter gene activity (Haraguchi et al. 2014). In epithelial cells from mouse mammary gland, miR-126-3p binds the 3'UTR of mouse Pgr mRNA and down-regulates both Pgr gene expression and cell proliferation (Cui et al. 2011). These studies pave the way for future research relating microRNAs to PGR gene expression.

Post-transcriptional regulation of the expression of the AR gene was investigated because of its role in prostate cancer (Zhu et al. 2013). The mRNA encoding AR

carries an extensive (6783 bases long) 3'UTR (Table 5.1). Screening of 1179 microRNAs identified 13 microRNAs that down-regulated AR protein levels in a prostate cancer cell line (Östling et al. 2011). All of them bound the 3'UTR of *AR* mRNA. Intriguingly, transfection of expression vectors for each of these 13 microR-NAs also down-regulated the androgen-dependent proliferation of the prostate cancer cell lines.

5 Conclusions and Future Directions

The post-transcriptional regulation of mRNAs that encode steroid hormone receptors is a molecular mechanism with extensive effects on the expression levels of many genes in diverse tissues. The bidirectional control of hormone receptor mRNA stabilities allows positive or negative gene regulation that finely tunes hormone responsiveness. Interactions between the RNA sequence elements and the proteins and microRNAs that bind them are an intriguing area of research (Jing et al. 2005). Mechanistic information about how steroid hormones regulate mRNA stability may provide new targets for therapeutic control of expression of specific genes. Some investigators are already attempting to interrupt crucial mRNA-protein and mRNAmicroRNA interactions for therapeutic interventions (Coulis et al. 2000; DeJong et al. 2002). Alternatively, expression or function of the trans-acting factor, such as a steroid hormone-induced RNA-binding protein, could be individually selected for inhibition or augmentation. To target the proteins that regulate mRNA stability by binding directly to cis-elements on the mRNA, oligonucleotide mimics of the ciselement (also called "RNA decoys") could be used to bind and sequester those RNA-binding proteins and interrupt the proteins' effect on mRNA stability and translation (Makeyev et al. 2002). For example, if uteri were treated with the ESR1 mRNA regions responsible for its stabilization (such as MEMSS), the AUF1p45 and other binding proteins could be sequestered and unable to stabilize ESR1 mRNA. This would block the up-regulation of the expression of the ER gene, as well as the genes it subsequently up-regulates. This approach is unique because it could spare the beneficial influences of basal levels of estrogens on bone, brain and circulatory system health.

Elucidation of the molecular mechanisms by which microRNAs regulate the stability and translation of target mRNAs opens new therapeutic avenues for altering both normal and abnormal physiology of animal tissues. Recent advances in oligonucleotide therapeutics make it feasible to either up- or down-regulate key microR-NAs in cell culture and, subsequently, in vivo (Esau and Monia 2007; Ford and Cheng 2008; Zhao et al. 2008). For example, Krutzfeldt et al. (2005) used antisense microRNAs "antagomirs" successfully in mice to down-regulate an individual microRNA. A particular microRNA might be targeted by an antisense RNA to reverse the actions of the microRNA on expression of critical genes. Increased knowledge about microRNA actions may lead to new diagnostics and therapies that utilize microRNAs to fight diseases such as cancer. In another example, plasma levels of miR-221 increase with breast cancer (Piva et al. 2013). An antagonist to miR-221 lowers miR-221 and proliferation of MCF7 breast cancer cells in routine culture conditions as well as in tumors created in nude mice. Other anti-miR reagents discussed above may themselves be drugs for treating hormone-dependent cancers. Alternatively, therapies might increase concentration of microRNAs that act as tumor suppressors, such as miR-17-5p in breast cancer (Hossain et al. 2006).

As our knowledge of the post-transcriptional regulation of mRNAs encoding steroid hormone receptors grows, so will our ability to alter tissue physiology in order to enhance human and animal health.

References

- Adams BD, Furneaux H, White BA (2007) The micro-ribonucleic acid (miRNA) miR-206 targets the human estrogen receptor-alpha (ERalpha) and represses ERalpha messenger RNA and protein expression in breast cancer cell lines. Mol Endocrinol 21:1132–1147
- Al-Nakhle H, Burns PA, Cummings M, Hanby AM, Hughes TA, Satheesha S, Shaaban AM, Smith L, Speirs V (2010) Estrogen receptor beta 1 expression is regulated by miR-92 in breast cancer. Cancer Res 70:4778–4784
- Arao Y, Kikuchi A, Ikeda K, Nomoto S, Horiguchi H, Kayama F (2002) A + U-rich-element RNA binding factor 1/heterogeneous nuclear ribonucleoprotein D gene expression is regulated by oestrogen in the rat uterus. Biochem J 361:125–132
- Arao Y, Kikuchi A, Kishida M, Yonekura M, Inoue A, Yasuda S, Wada S, Ikeda K, Kayama F (2004) Stability of A+U-rich element binding factor 1 (AUF1)-binding messenger ribonucleic acid correlates with the subcellular relocalization of AUF1 in the rat uterus upon estrogen treatment. Mol Endocrinol 18:2255–2267
- Baek D, Villen J, Shin C, Camargo FD, Gygi SP, Bartel DP (2008) The impact of microRNAs on protein output. Nature 455:64–71
- Balmer LA, Beveridge DJ, Jazayeri JA, Thomson AM, Walker CE, Leedman PJ (2001) Identification of a novel AU-rich element in the 3' untranslated region of epidermal growth factor receptor mRNA that is the target for regulated RNA-binding proteins. Mol Cell Biol 21:2070–2084
- Bevilacqua A, Ceriani MC, Capaccioli S, Nicolin A (2003) Post-transcriptional regulation of gene expression by degradation of messenger RNAs. J Cell Physiol 195:356–372
- Boyce-Derricott J, Nagler JJ, Cloud JG (2010) Variation among rainbow trout (*Oncorhynchus mykiss*) estrogen receptor isoform 3' untranslated regions and the effect of 17beta-estradiol on mRNA stability in hepatocyte culture. DNA Cell Biol 29:229–234
- Brewer G (2001) Messenger RNA, decay during aging and development. Ageing Res Rev 1:607-625
- Chen CY, Gherzi R, Ong SE, Chan EL, Raijmakers R, Pruijn GJ, Stoecklin G, Moroni C, Mann M, Karin M (2001) AU binding proteins recruit the exosome to degrade ARE-containing mRNAs. Cell 107:451–464
- Cochrane DR, Spoelstra NS, Richer JK (2012) The role of miRNAs in progesterone action. Mol Cell Endocrinol 357:50–59
- Coulis CM, Lee C, Nardone V, Prokipcak RD (2000) Inhibition of c-myc expression in cells by targeting an RNA-protein interaction using antisense oligonucleotides. Mol Pharmacol 57:485–494
- Cuadrado A, Navarro-Yubero C, Furneaux H, Munoz A (2003) Neuronal HuD gene encoding a mRNA stability regulator is transcriptionally repressed by thyroid hormone. J Neurochem 86:763–773
- Cui W, Li Q, Feng L, Ding W (2011) MiR-126-3p regulates progesterone receptors and involves development and lactation of mouse mammary gland. Mol Cell Biochem 355:17–25

- DeJong ES, Luy B, Marino JP (2002) RNA and RNA-protein complexes as targets for therapeutic intervention. Curr Top Med Chem 2:289–302
- Derijk RH, Schaaf MJ, Turner G, Datson NA, Vreugdenhil E, Cidlowski J, de Kloet ER, Emery P, Sternberg EM, Detera-Wadleigh SD (2001) A human glucocorticoid receptor gene variant that increases the stability of the glucocorticoid receptor beta-isoform mRNA is associated with rheumatoid arthritis. J Rheumatol 28:2383–2388
- Dodson RE, Shapiro DJ (2002) Regulation of pathways of mRNA destabilization and stabilization. Prog Nucleic Acids Res 72:129–164
- Esau CC, Monia BP (2007) Therapeutic potential for microRNAs. Adv Drug Deliv Rev 59:101-114
- Flouriot G, Pakdel F, Valotaire Y (1996) Transcriptional and post-transcriptional regulation of rainbow trout estrogen receptor and vitellogenin gene expression. Mol Cell Endocrinol 124:173–183
- Ford LP, Cheng A (2008) Using synthetic precursor and inhibitor miRNAs to understand miRNA function. Methods Mol Biol 419:289–301
- Gruber AR, Fallmann J, Kratochvill F, Kovarik P, Hofacker IL (2011) AREsite: a database for the comprehensive investigation of AU-rich elements. Nucleic Acids Res 39:D66–D69
- Guhaniyogi J, Brewer G (2001) Regulation of mRNA stability in mammalian cells. Gene 265:11–23
- Haraguchi H, Saito-Fujita T, Hirota Y, Egashira M, Matsumoto L, Matsuo M, Hiraoka T, Koga K, Yamauchi N, Fukayama M, Bartos A, Cha J, Dey SK, Fujii T, Osuga Y (2014) MicroRNA-200a locally attenuates progesterone signaling in the cervix, preventing embryo implantation. Mol Endocrinol 28:1108–1117
- Hargrove JL, Schmidt FH (1989) The role of mRNA and protein stability in gene expression. FASEB J 3:2360–2370
- Hargrove JL, Hulsey MG, Beale EG (1991) The kinetics of mammalian gene expression. Bioessays 13:667–674
- Herman JP, Watson SJ (1995) Stress regulation of mineralocorticoid receptor heteronuclear RNA in rat hippocampus. Brain Res 677:243–249
- Hollams EM, Giles KM, Thomson AM, Leedman PJ (2002) mRNA stability and the control of gene expression: implications for human disease. Neurochem Res 27:957–980
- Hossain A, Kuo MT, Saunders GF (2006) Mir-17-5p regulates breast cancer cell proliferation by inhibiting translation of AIB1 mRNA. Mol Cell Biol 26:8191–8201
- Ing NH (2005) Steroid hormones regulate gene expression post-transcriptionally by altering the stabilities of messenger RNAs. Biol Reprod 72:1290–1296
- Ing NH (2009) Post-transcriptional effects of estrogens on gene expression: messenger RNA stability and translation regulated by microRNAs and other factors. In: Berhardt LV (ed) Advances in Medicine and Biology. Nova, Hauppauge
- Ing NH (2010) Estradiol up-regulates expression of the A+U-rich binding factor 1 (AUF1) gene in the sheep uterus. J Steroid Biochem Mol Biol 122:172–179
- Ing NH, Massuto DA, Jaeger LA (2008) Estradiol up-regulates AUF1p45 binding to stabilizing regions within the 3'-untranslated region of estrogen receptor alpha mRNA. J Biol Chem 283:1764–1772
- Ing NH, Ott TL (1999) Estradiol up-regulates estrogen receptor-alpha messenger ribonucleic acid in sheep endometrium by increasing its stability. Biol Reprod 60:134–139
- Ing NH, Spencer TE, Bazer FW (1996) Estrogen enhances expression of the estrogen receptor gene in endometrium by a posttranscriptional mechanism. Biol Reprod 54:591–599
- Iorio MV, Ferracin M, Liu CG, Veronese A, Spizzo R, Sabbioni S, Magri E, Pedriali M, Fabbri M, Campiglio M, Menard S, Palazzo JP, Rosenberg A, Musiani P, Volinia S, Nenci I, Calin GA, Querzoli P, Negrini M, Croce CM (2005) MicroRNA gene expression deregulation in human breast cancer. Cancer Res 65:7065–7070
- Jing Q, Huang S, Guth S, Zarubin T, Motoyama A, Chen J, Di Padova F, Lin SC, Gram H, Han J (2005) Involvement of microRNA in AU-rich element-mediated mRNA instability. Cell 120:623–634

- Keene JD, Tenenbaum SA (2002) Eukaryotic mRNPs may represent posttranscriptional operons. Mol Cell 9:1161–1167
- Kenealy M-R, Flouriot G, Sonntag-Buck V, Dandekar T, Brand H, Gannon F (2000) The 3' untranslated region of the human estrogen receptor-α gene mediates rapid messenger ribonucleic acid turnover. Endocrinology 141:2805–2813
- Klinge CM (2012) miRNAs and estrogen action. Trends Endocrinol Metab 23:223-233
- Krutzfeldt J, Rajewsky N, Braich R, Rajeev KG, Tuschl T, Manoharan M, Stoffel M (2005) Silencing of microRNAs in vivo with 'antagomirs'. Nature 438:685–689
- Lal A, Abdelmohsen K, Pullmann R, Kawai T, Galban S, Yang X, Brewer G, Gorospe M (2006) Posttranscriptional derepression of GADD45alpha by genotoxic stress. Mol Cell 22:117–128
- Ledderose C, Mohnle P, Limbeck E, Schutz S, Weis F, Rink J, Briegel J, Kreth S (2012) Corticosteroid resistance in sepsis is influenced by microRNA-124-induced downregulation of glucocorticoid receptor-alpha. Crit Care Med 40:2745–2753
- Lee MT, Ouyang B, Ho SM, Leung YK (2013) Differential expression of estrogen receptor beta isoforms in prostate cancer through interplay between transcriptional and translational regulation. Mol Cell Endocrinol 376:125–135
- Loflin P, Chen CY, Shyu AB (1999) Unraveling a cytoplasmic role for hnRNP D in the in vivo mRNA destabilization directed by the AU-rich element. Genes Dev 13:1884–1897
- Lu J-Y, Schneider RJ (2004) Tissue distribution of AU-rich mRNA-binding proteins involved in regulation of mRNA decay. J Biol Chem 279:12974–12979
- Lv M, Zhang X, Jia H, Li D, Zhang B, Zhang H, Hong M, Jiang T, Jiang Q, Lu J, Huang X, Huang B (2012) An oncogenic role of miR-142-3p in human T-cell acute lymphoblastic leukemia (T-ALL) by targeting glucocorticoid receptor-alpha and cAMP/PKA pathways. Leukemia 26:769–777
- Maillot G, Lacroix-Triki M, Pierredon S, Gratadou L, Schmidt S, Benes V, Roche H, Dalenc F, Auboeuf D, Millevoi S, Vagner S (2009) Widespread estrogen-dependent repression of microRNAs involved in breast tumor cell growth. Cancer Res 69:8332–8340
- Makeyev AV, Eastmond DL, Liebhaber SA (2002) Targeting a KH-domain protein with RNA decoys. RNA 8:1160–1173
- Manavathi B, Dey O, Gajulapalli VN, Bhatia RS, Bugide S, Kumar R (2013) Derailed estrogen signaling and breast cancer: an authentic couple. Endocr Rev 34:1–32
- Maniatis T, Reed R (2002) An extensive network of coupling among gene expression machines. Nature 416:499–506
- Manjithaya RR, Dighe RR (2004) The 3'untranslated region of bovine follicle-stimulating hormone β messenger RNA downregulates reporter expression: involvement of AU-rich elements and transfactors. Biol Reprod 71:1158–1166
- Mattick JS (2001) The evolution of controlled multitasked gene networks: the role of introns and other noncoding RNAs in the development of complex organisms. Mol Biol Evol 18:1611–1630
- McGlincy NJ, Smith CW (2008) Alternative splicing resulting in nonsense-mediated mRNA decay: what is the meaning of nonsense? Trends Biochem Sci 33:385–393
- McKnight GS, Palmiter RD (1979) Transcriptional regulation of the ovalbumin and conalbumin genes by steroid hormones in chick oviduct. J Biol Chem 254:9050–9058
- Menon KMJ, Munshi UM, Clouser CL, Nair AK (2004) Regulation of luteinizing hormone/human chorionic gonadotropin receptor expression: a perspective. Biol Reprod 70:861–866
- Mitchell DC, Ing NH (2003) Estradiol stabilizes estrogen receptor mRNA in sheep endometrium via discrete sequence elements in its 3' untranslated region. Mol Endocrinol 17:562–574
- Mukherjee D, Gao M, O'Connor JP, Raijmakers R, Pruijn G, Lutz CS, Wilusz J (2002) The mammalian exosome mediates the efficient degradation of mRNAs that contain AU-rich elements. EMBO J 21:165–174
- Ni JZ, Grate L, Donohue JP, Preston C, Nobida N, O'Brien G, Shiue L, Clark TA, Blume JE, Ares M Jr (2007) Ultraconserved elements are associated with homeostatic control of splicing regulators by alternative splicing and nonsense-mediated decay. Genes Dev 21:708–718
- Östling P, Leivonen SK, Aakula A, Kohonen P, Makela R, Hagman Z, Edsjo A, Kangaspeska S, Edgren H, Nicorici D, Bjartell A, Ceder Y, Perala M, Kallioniemi O (2011) Systematic analysis

of microRNAs targeting the androgen receptor in prostate cancer cells. Cancer Res 71:1956-1967

- Pandey DP, Picard D (2010) Multidirectional interplay between nuclear receptors and microRNAs. Curr Opin Pharmacol 10:637–642
- Piva R, Spandidos DA, Gambari R (2013) From microRNA functions to microRNA therapeutics: novel targets and novel drugs in breast cancer research and treatment. Int J Oncol 43:985–994
- Ramamoorthy S, Cidlowski JA (2013) Ligand-induced repression of the glucocorticoid receptor gene is mediated by an NCoR1 repression complex formed by long-range chromatin interactions with intragenic glucocorticoid response elements. Mol Cell Biol 33:1711–1722
- Riester A, Issler O, Spyroglou A, Rodrig SH, Chen A, Beuschlein F (2012) ACTH-dependent regulation of microRNA as endogenous modulators of glucocorticoid receptor expression in the adrenal gland. Endocrinology 153:212–222
- Saceda M, Lindsey RK, Solomon H, Angeloni SV, Martin MB (1998) Estradiol regulates estrogen receptor mRNA stability. J Steroid Biochem Mol Biol 66:113–120
- Saltzman AL, Kim YK, Pan Q, Fagnani MM, Maquat LE, Blencowe BJ (2008) Regulation of multiple core spliceosomal proteins by alternative splicing-coupled nonsense-mediated mRNA decay. Mol Cell Biol 28:4320–4330
- Sarkar B, Xi Q, He C, Schneider RJ (2003) Selective degradation of AU-rich mRNAs promoted by the p37 AUF1 protein isoform. Mol Cell Biol 23:6685–6693
- Sato S, Shirato K, Tachiyashiki K, Imaizumi K (2011) Synthesized glucocorticoid, dexamethasone regulates the expressions of beta(2)-adrenoceptor and glucocorticoid receptor mRNAs but not proteins in slow-twitch soleus muscle of rats. J Toxicol Sci 36:479–486
- Schaaf MJ, Cidlowski JA (2002) AUUUA motifs in the 3'UTR of human glucocorticoid receptor alpha and beta mRNA destabilize mRNA and decrease receptor protein expression. Steroids 67:627–636
- Schaaf MJM, Cidlowski JA (2003) Molecular mechanisms of glucocorticoid action and resistance. J Steroid Biochem Mol Biol 83:37–48
- Shim J, Karin M (2002) The control of mRNA stability in response to extracellular stimuli. Mol Cells 14:323–331
- Smith L, Coleman LJ, Cummings M, Satheesha S, Shaw SO, Speirs V, Hughes TA (2010) Expression of oestrogen receptor beta isoforms is regulated by transcriptional and posttranscriptional mechanisms. Biochem J 429:283–290
- Söber S, Laan M, Annilo T (2010) MicroRNAs miR-124 and miR-135a are potential regulators of the mineralocorticoid receptor gene (NR3C2) expression. Biochem Biophys Res Commun 391:727–732
- Staton JM, Thomson AM, Leedman PJ (2000) Hormonal regulation of mRNA stability and RNAprotein interactions in the pituitary. J Mol Endocrinol 25:17–34
- Stellato C (2004) Post-transcriptional and nongenomic effects of glucocorticoids. Proc Am Thorac Soc 1:255–263
- Sun M, Estrov Z, Ji Y, Coombes KR, Harris DH, Kurzrock R (2008) Curcumin (diferuloylmethane) alters the expression profiles of microRNAs in human pancreatic cancer cells. Mol Cancer Ther 7:464–473
- Tessel MA, Benham AL, Krett NL, Rosen ST, Gunaratne PH (2011) Role for microRNAs in regulating glucocorticoid response and resistance in multiple myeloma. Horm Cancer 2:182–189
- Tolnay M, Vereshchagina LA, Tsokos GC (1999) Heterogeneous nuclear ribonucleoprotein D0B is a sequence-specific DNA-binding protein. Biochem J 338:417–425
- Tseng L, Zhu HH (1997) Regulation of progesterone receptor messenger ribonucleic acid by progestin in human endometrial stromal cells. Biol Reprod 57:1360–1366
- Uchida S, Nishida A, Hara K, Kamemoto T, Suetsugi M, Fujimoto M, Watanuki T, Wakabayashi Y, Otsuki K, McEwen BS, Watanabe Y (2008) Characterization of the vulnerability to repeated stress in Fischer 344 rats: possible involvement of microRNA-mediated down-regulation of the glucocorticoid receptor. Eur J Neurosci 27:2250–2261
- Vandevyver S, Dejager L, Libert C (2014) Comprehensive overview of the structure and regulation of the glucocorticoid receptor. Endocr Rev 35:671–693

- Vreugdenhil E, Verissimo CS, Mariman R, Kamphorst JT, Barbosa JS, Zweers T, Champagne DL, Schouten T, Meijer OC, de Kloet ER, Fitzsimons CP (2009) MicroRNA 18 and 124a down-regulate the glucocorticoid receptor: implications for glucocorticoid responsiveness in the brain. Endocrinology 150(5):2220–2228
- Wagner BJ, DeMaria CT, Sun Y, Wilson GM, Brewer G (1998) Structure and genomic organization of the human AUF1 gene: alternative pre-mRNA splicing generates four protein isoforms. Genomics 48:195–202
- Yeap BB, Krueger RG, Leedman PJ (1999) Differential posttranscriptional regulation of androgen receptor gene expression by androgen in prostate and breast cancer cells. Endocrinology 140:3282–3291
- Yeap BB, Wilce JA, Leedman PJ (2004) Novel binding of HuR and poly(C) binding protein to a conserved UC-rich motif within the 3'-untranslated region of the androgen receptor mRNA. Bioessays 26:672–682
- Zhang T, Kruyus V, Huez G, Gueydan C (2002) AU-rich element-mediated translational control: complexity and multiple activities of trans-acting factors. Biochem Soc Trans 30:952–958
- Zhao JJ, Lin J, Yang H, Kong W, He L, Ma X, Coppola D, Cheng JQ (2008) MicroRNA-221/222 negatively regulates estrogen receptor alpha and is associated with tamoxifen resistance in breast cancer. J Biol Chem 283:31079–31086
- Zhou H, Mazan-Mamczarz K, Martindale JL, Barker A, Liu Z, Gorospe M, Leedman PJ, Gartenhaus RB, Hamburger AW, Zhang Y (2010) Post-transcriptional regulation of androgen receptor mRNA by an ErbB3 binding protein 1 in prostate cancer. Nucleic Acids Res 38:3619–3631
- Zhu KC, Lu JJ, Xu XL, Sun JM (2013) MicroRNAs in androgen-dependent PCa. Front Biosci 18:748–755
- Zuker M (2003) Mfold web server for nucleic acid folding and hybridization prediction. Nucleic Acids Res 31:3406–3415

MicroRNA Regulation of Endocrine Functions in the Ovary

6

Pavla Brachova, Wei-Ting Hung, Lynda K. McGinnis, and Lane K. Christenson

1 Introduction

The ovaries have two main roles in maintaining normal reproductive functions: an *exocrine* role of producing the oocyte, and an *endocrine* role of producing sex hormones. The endocrine properties of the ovaries appear in a cyclical manner starting at puberty and continue throughout a female's reproductive lifespan, with each cycle resulting in the maturation of an ovarian follicle or atresia of non-dominant follicles. For successful reproduction to occur, the ovary must complete follicular development, ovulation, formation of a corpus luteum, and luteolysis. Successful networking and cooperation between the somatic cells (granulosa, theca, luteal, endothelial, immune, and epithelial cells) and the germ line (oocyte) is necessary to establish this pattern of regular ovulatory cycles. The gene regulatory networks are temporally and spatially regulated through intricate transcriptional and post-transcriptional regulatory mechanisms. Transcriptional regulation has been well studied and reviewed in detail (Lavoie and King 2009; Richards 1994; Espey and Richards 2002), but much less is known about post-transcriptional regulation in reproductive organs.

Post-transcriptional regulation can involve a multitude of events downstream of transcription, including mRNA splicing, RNA editing, RNA transportation, RNA storage, all the way through translation. This form of regulation is important because it allows for fine-tuning of gene expression, in a tissue specific manner independent of transcription (Carletti and Christenson 2009). These post-transcriptional regulatory mechanisms, however are difficult to systematically study, since they are often gene specific and the cell physiologic context can have dramatic influences on the regulatory mechanisms.

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_6

P. Brachova • W.-T. Hung • L.K. McGinnis • L.K. Christenson (🖂)

Department Molecular and Integrative Physiology, University of Kansas Medical Center, 3901 Rainbow Blvd., Mail Stop 3043, Kansas City, KS 66160, USA e-mail: lchristenson@kumc.edu

[©] Springer International Publishing Switzerland 2016

Thus, much of the evidence for post-transcriptional regulation in ovarian endocrine function has arisen on a gene-by-gene basis typically when differences in mRNA and protein expression are observed to be out of synch with each other. An exception to this limitation is in the study of microRNA (miRNA, miR) mediated post-transcriptional regulation. This is largely due to the use of targeted gene deletion of several key enzymatic steps in the miRNA biosynthetic pathway. MicroRNA signaling is crucial for control of ovarian function by regulating aspects of folliculogenesis, oocyte maturation, ovulation and luteal function. In this review we will focus on miRNA's role in regulation of ovarian endocrine function have been previously reviewed (Christenson 2010; Imbar and Eisenberg 2014; Donadeu et al. 2012; Hossain et al. 2012; Nothnick 2012; McGinnis et al. 2015), so they will not be repeated in-depth in the present review.

2 MicroRNA Biogenesis and Mechanisms of Action

MicroRNA are a class of small, non-coding RNA that are comprised of 21-24 single-stranded nucleotides. These regulatory RNA have key roles in a variety of cellular processes by negatively regulating target messenger RNA (mRNA) through sequence-specific binding that results in inhibition of translation and decreased mRNA stability (Bartel 2009). A few studies have indicated the ability for positive regulation by miRNA, however, this method of miRNA action has been observed only experimentally and more studies are needed to verify this mechanism of action (Place et al. 2008; Orom et al. 2008). MicroRNA are predominately transcribed by RNA polymerase II into primary microRNA (pri-miR) and get processed into shorter (70–100 base pair) hairpin structures called precursor microRNA (pre-miR) by the microprocessor complex, comprised of Drosha and its binding partner, DiGeorge Syndrome Critical Region Gene 8 (DGCR8; Fig. 6.1). In turn, the premiR short hairpin structures associate with the nuclear export receptor, exportin-5 (XPO5), to get translocated into the cytoplasm (Bohnsack et al. 2004). Exportin-5 cooperates with the small GTPase Ran, to mediate the directional transport to the cytoplasm. Following export to the cytoplasm, GTP is hydrolyzed and the pre-miRNA is released into the cytoplasm where it is further processed by the RNAse III enzyme, Dicer, to create a short double stranded miRNA duplex. One of the strands will associate with the RNA induced silencing complex (RISC) and is then referred to as the guide strand or mature miRNA. The opposite strand referred to as the passenger strand then undergoes degradation.

In addition to this canonical pathway of miRNA biogenesis, several non-canonical pathways have been identified. A subset of miRNA called "mirtrons", do not depend on Drosha for their initial processing (Fig. 6.1). These miRNAs are processed from introns of transcripts after splicing (Ruby et al. 2007). The lariat-debranching enzyme involved in splicing resolves the RNA branch site and results in a pre-miRNA hairpin that gets exported to the cytoplasm like other pre-miRNA to be further processed by Dicer (Berezikov et al. 2007). Yet another alternative miRNA biogenesis mechanism can be seen in miR-451, which is processed in a Dicer independent fashion. In the case of miR-451, pre-miR-451 gets loaded into argonaute-2 (Ago2)



Fig. 6.1 Canonical and alternative miRNA biogenesis pathways. Most miRNA are transcribed by RNA Polymerase II to form long primary transcripts (pri-miRNA) which form complex structures that are then recognized and processed into precursor miRNA (pre-miRNA) by the microprocessor complex that includes the DGCR8 and RNAse III enzyme Drosha. These pre-miRNA hairpins are transported into the cytoplasm by exportin-5 and a Ran GTPase. In the cytoplasm, pre-miRNA are further processed into ~22 base pair duplexes by another RNAse III enzyme, Dicer, in association with GW182 and TAR RNA binding protein (TRBP). The mature miRNA then gets incorporated into the RISC complex, where it can then bind to the 3'-untranslated region (UTR) of target mRNA to block translation or degrade the transcript, or both. Also illustrated is the biogenesis of miRNA through the non-canonical pathways, including an intronic splicing mechanism. In these cases, the pre-miRNA is generated directly bypassing the need for Drosha and DGCR8, afterwards the remaining steps of processing are similar to the canonical miRNA maturation pathway. Another alternative process occurs specifically with miR-451, this miRNA utilizes the beginning steps of the canonical pathway however, instead of Dicer the pre-miR is processed by Ago2 in the cytosol

instead of Dicer, and is cleaved by the Ago2 catalytic center where it is then trimmed into a mature miRNA (Cheloufi et al. 2010).

Regardless of the mechanism of biogenesis, the mature miRNA are bound to the RISC to affect their biologic action within the cell. The RISC complex contains numerous proteins, but the primary effector of miRNA-mediated repression reside within a family of proteins known as the argonaute proteins (Gregory et al. 2005). The human genome contains eight argonaute-family proteins, but Ago2 is the only one with RNA cleavage activity and is thought to play the lead in miRNA-mediated silencing. Proteomics approaches have identified dozens of proteins that associate with Ago and RISC, and the combination of protein partners in this complex controls the regulation of RISC activity and the degree of miRNA-mediated repression (Hock et al. 2007). Association of the mature miRNA with RISC and the cognate mRNA at the 3'-UTR can result in repression of translation, or destabilization of the mRNA through decapping and degradation. The mechanistic details about these two methods of post-transcriptional repression remain an area of intense investigation (Wilczynska and Bushell 2015).

Mature miRNA can themselves be subjects to regulation in several ways. The mature miRNA transcript can be bound by other RNA transcripts called competing endogenous RNA (ceRNA) (Tay et al. 2014). Another class of RNA that can affect levels of mature miRNA are called circular RNA (circRNA), which are formed by a covalent link between the 5' and 3' ends of an exon by the spliceosome (Hentze and Preiss 2013). Both the ceRNA and circRNA can function as sponges and titrate mature RNA from the mRNA targets they may bind to. The function of these RNA families in the ovary has not yet been examined, but could provide useful information for understanding the biology of the endocrine functions of the ovaries.

3 Consequences of Global miRNA Depletion in Ovarian Cells

Targeted gene deletion studies in mice suggest a critical role for miRNA regulation throughout the body, including the ovary. Initial studies showed that deletion of the Dicer eliminated almost all miRNA from the developing embryo and caused embryonic lethality at E6.5 due to a loss of pluripotent stem cells and impaired angiogenesis (Bernstein et al. 2003; Yang et al. 2005). Similarly, loss of Ago2 or Drosha was also embryonic lethal by E6.5-E7.5 (Kaneda et al. 2009; Yuan et al. 2014). To overcome these embryonic lethal phenotypes, Dicer conditional knockout (cKO) mouse models were developed and the essential roles of Dicer were demonstrated in a number of critically important tissues and cell types (Luense et al. 2009). In the reproductive organs, Dicer has been deleted using the oocytespecific zona pellucida-3 (Zp3) promoter linked to Cre recombinase (Murchison et al. 2007), another oocyte-specific knockout using the DEAD (Asp-Glu-Ala-Asp) box polypeptide-4 (DDX4 or vasa) Cre and the anti-Mullerian hormone receptor-2 (Amhr2) Cre (Hong et al. 2008; Lei et al. 2010; Nagaraja et al. 2008; Gonzalez and Behringer 2009; Yuan et al. 2014), which is expressed in ovarian granulosa cells and derivatives of the Mullerian duct (oviduct, uterus, and cervix) and pituitary.

The targeted deletion of Dicer in the oocyte using Zp3-Cre resulted in infertility due to disorganized spindles, chromosome alignment defects, and oocyte arrest at metaphase-I of meiosis (Murchison et al. 2007; Tang et al. 2007). The Zp3-cKO of Ago2 resulted in similar chromosome abnormalities although these oocytes are able to mature to metaphase-II (Kaneda et al. 2009). Deleting Dicer at an even earlier stage of oocyte development (E15) caused female infertility due to failure of follicle growth (Yuan et al. 2014). Thus, Dicer, and Ago2 are necessary for normal oocyte meiosis, early follicle development and female fertility. This was originally interpreted to indicate that miRNA had minimal roles in oocyte development or in the crosstalk of the oocyte with the cumulus cells of growing follicles but that miRNA were critical in the oocyte at later stages for the completion of meiotic maturation prior to ovulation. However, more recent studies have discovered that Dicer and Ago2 are also involved in the biogenesis of endogenous small interfering RNA (siRNA) as well as miRNA (Flemr et al. 2013; Yuan et al. 2014). Two studies have helped to differentiate between the siRNA and miRNA pathways in the oocyte (Suh et al. 2010; Yuan et al. 2014). DGCR8 and Drosha are essential for miRNA but not siRNA biogenesis pathways. Therefore, DGCR8 was conditionally deleted from oocytes using the Zp3-Cre and the DDX4-cre, similar to the studies with Dicer and Ago2 (Suh et al. 2010; Yuan et al. 2014). Interestingly, deletion of DGCR8 or Drosha resulted in normal follicular development and oocytes that matured, fertilized and developed to healthy live born offspring indicating a requirement for siRNA but not miRNA in mouse oocyte maturation (Suh et al. 2010; Yuan et al. 2014). Another recent study specific to rats and mice discovered that the Dicer protein in the oocytes (Dicer⁰) of these species is different from the Dicer expressed in their somatic cells (Flemr et al. 2013). Expression of the Dicer^O transcript is driven by an alternative promoter that originates from a retrotransposon. As a result, Dicer^o has a higher affinity for double stranded RNA that have perfect complementarity, thus Dicer^o generates siRNA instead of miRNA (Flemr et al. 2013). These findings explain the severe phenotypes seen in the oocyte-specific knock-outs of Dicer and Ago2, but that are not found in the cKO of DGCR8 and Drosha. Because this oocyte-specific isoform of Dicer is not present in humans or other mammals, the implications only affect studies performed in rodents. It is clear that the function of small RNAs is necessary for correct development and function of the oocyte, however, much remains to be discovered in this young field. The mouse may not be an adequate model for the study of miRNA in the oocyte, but it may be suitable for studies in somatic cells because these cells have the full length Dicer that is required for miRNA processing.

Targeted knockout of Dicer has also been conducted using the *Amhr2*-Cre which deletes Dicer from the mesenchyme of the developing Mullerian ducts and postnatally in ovarian granulosa cells and mesenchyme-derived cells of the female reproductive tract (Gonzalez and Behringer 2009; Hong et al. 2008; Nagaraja et al. 2008; Pastorelli et al. 2009). Female mice with *Amhr2*-cKO of Dicer suffer from complete infertility due to abnormalities in all of these tissues. Progesterone (P4) and estrogen levels tested normal in these mice (Hong et al. 2008). However uteri were short and immature. Oviduct development was severely affected, especially at the isthmus, which showed a lack of smooth muscle leading to bulbous distended sac-like structures and a general disorganization of the oviductal epithelial layer (Nagaraja et al. 2008; Hong et al. 2008; Pastorelli et al. 2009). While the primary cause of infertility in the *Amhr2*-cKO Dicer mice was due to the trapping of oocytes and embryos within the enlarged sacs of the oviduct, these mice also exhibited reduced ovary weight and lower ovulation rates demonstrating a need for Dicer in ovary and follicular development (Hong et al. 2008). In addition, the cKO of Dicer in follicular cells led to luteal insufficiency and failure to maintain pregnancy (Otsuka et al. 2008). Studies such as these alter miRNA in a global manner, and consequently disrupt the more than 2500 miRNAs that have been described (see http://microRNA. sanger.ac.uk). Further research is necessary in order to understand the effect of specific miRNA and their targets on endocrine regulation within the ovary.

4 Ovarian miRNA Expression Profiling

Another mechanism used by researchers to identify whether miRNA might be involved in ovarian function has been through the profiling of miRNA expression in ovarian tissues. Initial studies focused on miRNA gene expression in whole ovarian tissues comparing different ages of animal (i.e., fetal/newborn to adults) (Ahn et al. 2010; Ro et al. 2007; Mishima et al. 2008; Tripurani et al. 2010; Huang et al. 2011) and different physiological states (follicular versus luteal phase) (Hossain et al. 2009). Mouse and cow have been the models most commonly used to identify miRNA expression patterns in the ovaries of adult and fetal animals, but studies on, pigs, goats, and sheep have also been performed (Li et al. 2011; Ling et al. 2014; McBride et al. 2012). While these studies identified ovarian specific miRNA, the advent of more robust techniques such as small RNA sequencing and deeper analysis of more tissues, has left very few miRNA for consideration as ovarian specific. Another major drawback of these studies is that the ovary is comprised of a very heterogeneous population of different cell types and that even within a cell type, the function and therefore gene expression can vary wildly dependent on the stage of the estrous or menstrual cycle at which the cells are evaluated.

Several studies have used more defined populations of ovarian somatic cells or have compared the changes in miRNA expression in extracellular vesicles (i.e., exosomes and microvesicles) isolated from follicular fluid collected at discrete periods of follicle development. Briefly, comparisons of miRNA expression of granulosa cells of subordinate and dominant follicles during the early luteal phase of the bovine estrous cycle using RNAseq identified significant differential miRNA expression (Salilew-Wondim et al. 2014). In granulosa cells isolated from mice following a superovulation protocol, a small number of miRNA in granulosa cells were found to change temporally with respect to administration of an ovulatory dose of the LH mimic, human chorionic gonadotropin (hCG) (Fiedler et al. 2008). Within the highly steroidogenic corpus luteum, small RNAseq was used to identify 15 miRNA that were differentially expressed in pregnant versus non-pregnant cows and these miRNA were predicted to target genes involved in apoptosis and immune signaling pathways (Maalouf et al. 2014). Lastly, several studies have characterized the miRNA associated with extracellular vesicles derived from follicular fluid (Diez-Fraile et al. 2014; Roth et al. 2014; Sohel et al. 2013; Sang et al. 2013; da Silveira et al. 2012). In mares and women, age was shown to influence miRNA expression in extracellular vesicles (da Silveira et al. 2012; Diez-Fraile et al. 2014). Altered miRNA abundance was also seen in the follicular fluid of women with polycystic ovarian syndrome compared to normal controls (Roth et al. 2014; Sang et al. 2013). In another study, differences in miRNA abundance in extracellular vesicles compared to miRNA in vesicle-free fractions of follicular fluid was observed (Sohel et al. 2013). MicroRNA were found in both vesicle and vesicle-free fractions of bovine follicular fluid, and the miRNA showed differential abundance in these two fractions. While this finding is intriguing, the biological function of miRNA within the extracellular vesicles and extracellular vesicle-free miRNA remains to be determined. Furthermore, most studies assessing changes of miRNA abundance in follicular fluid lack experimental controls necessary to determine the parental cell of origin that generated the miRNA. Unfortunately, those studies of the corpus luteum showing differential miRNA expression still suffer from being derived from a very heterogeneous population of cell types (i.e., immune, endothelial, luteal). These studies highlight our current lack of understanding of miRNA expression within ovarian tissues and the need to perform more robust studies aimed at understanding the functional role these small molecules have on ovarian function.

In conclusion, the genetic depletion of critical regulators of miRNA biogenesis and function, and the evidence of differential expression of miRNA within tissues and cells collected at discrete functionally relevant time points, does support a role for these molecules on normal ovarian function.

5 MicroRNA Regulation Within the Antral Follicle

As a major functional unit within the ovary, follicles undergo a progressive transformational process as they grow from primordial to ovulatory follicles. During this developmental process the antral follicle becomes the first tissue to produce significant levels of the major endocrine hormone, estradiol (E2). This estrogen dominates during the follicular phase and is critical for regulation of the hypothalamic/pituitary/gonadal axis, secondary sex characteristics, reproductive tract function, and reproductive behavior. Copious numbers of studies have examined the pathway of E2 production and the role gonadotropins play in support of the endocrine functions of the antral follicle (Knobil and Neill 2014). Indeed, the well-known 'two cell two-gonadotropin theory' describes how compartmentalization of the steroidogenic enzymes and gonadotropin receptors within the two different follicular cells, facilitate thecal cell production of androgens and subsequent granulosa cell aromatization of those androgens to estrogen (Knobil and Neill 2014) (Fig. 6.2). Furthermore, estrogen has negative and positive endocrine effects at the pituitary that are essential for regulation of the LH and FSH secretion that drives folliculogenesis and ultimately the surge of LH that signals the end of



Fig. 6.2 MicroRNA regulation of ovarian endocrine system. Ovarian endocrine regulation is divided into three main temporal periods: the antral follicle, ovulation, and corpus luteum. The cell types involved (theca, granulosa and luteal), the primary endocrine hormones (FSH and LH) and their respective receptors (FSHr and LHr), and the primary steroids (progesterone, P4; androgens, A4; and estrogen, E2) produced by these cells are illustrated. Those miRNA that have been implicated in the direct regulation of these hormone secretion pathways or the regulation of the receptors are shown. Details including the references related to these findings are described in the appropriate sections within this chapter

follicular development and the ovulation of the oocyte. While a substantial number ~26 papers have examined the putative functions of miRNA in mural granulosa cells isolated from antral follicles, most of these have focused on other critical cellular elements involved in folliculogenesis such as proliferation and apoptosis (see Reviews McGinnis et al. 2015; Donadeu et al. 2012). A very limited number of studies have examined the role miRNA play specifically in the regulation of steroidogenesis within the antral follicle, those papers and supporting papers from other areas of biology are described below.

Ovarian endocrine function is dependent on androgen signaling through interactions from the theca cells and granulosa cells (Hu et al. 2004). Androgens can affect ovarian steroidogenesis directly through androgen receptors (AR), or indirectly by the process of aromatization to estrogen. Androgens play a critical role in early follicle development, as demonstrated by mice that lack a functional AR. These mice are less fertile and have a shorter reproductive window than controls (Hu et al. 2004; Lyon and Glenister 1980). Androgen receptors are highly expressed in granulosa cells and theca cells of early stage follicles (Gervasio et al. 2014). AR can enhance the expression of miR-125b, which has anti-apoptotic effects and may regulate follicular atresia by promoting survival of granulosa cells (Sen et al. 2014). In primary mouse granulosa cells, AR binds to the promoter of the miR-125b primary transcript at three androgen response elements, in an androgen dependent fashion, indicating a transcriptional regulation of miR-125b. Interestingly, the transcriptional regulation of miR-125b was dependent upon nuclear and extranuclear actions of androgen, because the transcriptional upregulation of miR-125b was dependent on interactions of AR and paxillin, a scaffolding mediator of kinase cascades. The in vivo effect of miR-125b on ovarian function was demonstrated using bursal injections of miR-125b inhibitors, which significantly increased the expression of pro-apoptotic proteins BAK, BAX, BMF, and p53 in granulosa cells and induced apoptosis. This study demonstrated the in vivo effect of a miRNA on follicular development, in an androgen-dependent manner. MicroRNAs targeting the AR have also been identified systematically in prostate cancer cell lines (Ostling et al. 2011). A gain-of-function screen of 1129 miRNA in a panel of human prostate cancer cell lines was used to quantify changes in AR protein. Identification of 13 miRNA were validated using a luciferase reporter assay containing the AR 3'UTR: (miR-135b, miR-185, miR-297, miR-299-3p, miR-34a, miR-34c, miR-371-3p, miR-421, miR-449a, miR-449b, miR-634, miR-654-5p, and miR-9). Briefly, these assays work by linking the 3'-UTR of the gene of interest (i.e., AR, PGR) to the luciferase gene, followed by transfection of this reporter into cells where the investigator either blocks or enhances the expression of a specific miRNA. The contribution of these AR-modulating miRNA within the ovary remains to be determined.

During folliculogenesis and steroidogenesis, E2 acts together with FSH and is critical in enabling pituitary neuron cells to exhibit fluctuating endocrine patterns allowing for the preovulatory LH surge (Xu et al. 2011). Because of the importance of E2, the regulation of aromatase (CYP19A1) has been extensively studied (reviewed Stocco 2008). In porcine granulosa cells, miR-378 down regulates the expression of CYP19A1 (and therefore estrogen) (Xu et al. 2011). MicroRNA-378 targeted both CYP19A1 and progesterone receptor (PGR), and the down regulation of this miRNA appeared to be necessary for granulosa cell differentiation to the preovulatory stage. These studies were both performed *in vitro*, and it remains unclear how the miRNA is regulated *in vivo*.

Several additional miRNA can target genes involved in E2 production by mouse granulosa cells. MicroRNA-224 can affect E2 production by targeting Smad4, a transcription factor involved in the transforming growth factor-beta1 (TGF- β 1) pathway that can activate expression of a variety of proliferation genes (Yao et al. 2010a). These studies, however, are difficult to interpret because TGF- β did not induce miR-224 expression consistently. The same group identified miR-383 as being down regulated by TGF- β 1. MicroRNA-383 can target RNA binding motif, single stranded interacting protein 1 (RBMS1), thus increasing E2 release (Yin

et al. 2012). Another miRNA associated with E2 synthesis is miR-133b, which directly targets the transcription factor Forkhead L2 (Foxl2), which inhibits transcription of steroidogenic acute regulatory protein (StAR) and CYP19A1, thus leading to an increase in E2 production after FSH stimulation (Dai et al. 2013). Both isoforms of the estrogen receptor (ER α and ER β) are important in fertility, as demonstrated in KO mouse models. Null ER α (α ERKO) mice are infertile, while null ER β (β ERKO) mice are subfertile (Couse et al. 2003). The α ERKO mice contained hemorrhagic and cystic ovaries, and displayed abnormally high levels of serum LH, which resulted in elevation of steroidogenic genes (Cyp17a1, Cyp19, and HSD17b3). ERβ is expressed in granulosa cells, and the βERKO mice had a predominant phenotype in follicles, with evidence of unruptured follicles after superovulation as well as decreased numbers of recovered oocytes (Krege et al. 1998). While the two estrogen receptors are essential to fertility, the contribution of miRNA to their regulation is poorly understood in the ovary. Several studies have identified miRNA that target ER α and ER β in cancer cell lines. In hepatocellular carcinoma cell lines ER α was shown to be directly targeted by miR-18a and miR-22 (Liu et al. 2009), while in breast cancer cell lines miR-206 was shown to affect ER directly (Pandey and Picard 2009; Adams et al. 2007). Fewer studies have analyzed ER β as a target of miRNA, limited to miR-92 in MCF-7 breast cancer cells (Al-Nakhle et al. 2010).

Additionally, endocrine hormones such as FSH that affect antral follicle development and function can also alter the expression of miRNA (Yao et al. 2010b). Specifically, 12 h post-FSH treatment in primary rat granulosa cells, miR-29a and miR-30d were significantly down regulated, but increased in expression after 48 h. However, the specific targets of these miRNA were not identified, and the biologic consequences of the biphasic response at 12 and 48 h were not explored. Importantly, many of the identified miRNA involved in steroid production pathways have been identified through *in vitro* experiments. Few studies have confirmed miRNA effects on steroidogenesis *in vivo*.

6 MicroRNA Regulation Within the Ovulatory Follicle

At the completion of follicular growth, increasing follicular estrogen signals to the hypothalamus and pituitary to initiate a surge of LH secretion that facilitates the release of the oocyte and the conversion or luteinization of the granulosa and thecal cells. During this periovulatory period, P4 becomes highly concentrated in follicular fluid (Emori and Drapkin 2014), and it is critical for the final stages of folliculogenesis and for ovulation (Duffy et al. 2010). Simultaneously, the LH surge induces the transient expression of progesterone receptor (PGR) in the luteinizing granulosa cells of rodents, and in mammals with a true luteal phase, the LH surge induces long-term expression of PGR (Gougeon 2010; Hatzirodos et al. 2014; Iwai et al. 1990; Revelli et al. 1996). In mice the function of the two different isoforms of PGR (A and B) have been evaluated and loss of PGR-A was shown to cause a 70 % decline in the number of oocytes ovulated while in contrast, loss of PGR-B had no effect detectable on ovulation (Conneely et al. 2003; Shao et al. 2003). The transcriptional expression of PGR is clearly upregulated in granulosa cells following

hormonal gonadotropin signaling through FSH and LH (Shimada and Terada 2002; Clemens et al. 1998), but post-transcriptional regulation by miRNA has also been shown to be involved (Toms et al. 2014). In the pig, PGR was inversely correlated with expression of miR-378-3p in developing follicles and *in vitro* studies using isolated granulosa cells showed that miR-378-3p could directly target PGR mRNA as evidenced by 3'-UTR luciferase reporter assays (Toms et al. 2014). The expression of miR-378-3p is lower in porcine large follicles, suggesting that by repressing this miRNA during late folliculogenesis, the repression of genes essential for granulosa cell differentiation (CYP19A1 and PGR) is alleviated. Additionally, in three breast cancer cell lines, miR-513a-5p decreased the luciferase activity in a PGR luciferase reporter assay, indicating that this miRNA also directly targets PGR (Cochrane et al. 2012). Supporting miR-513a-5p as a direct regulator of PGR, this miRNA was also induced by P4 (Cochrane et al. 2012), whether this miRNA is expressed in the ovary and if it may function there has not been assessed.

In another approach to examine whether miRNA can elicit changes in steroidogenesis *in vitro*, primary granulosa cells were harvested from normally cycling women after ovariectomy due to non-metastatic cervical cancer and were transfected with 80 different pre-miRNA constructs (Sirotkin et al. 2009). Markers of proliferation, apoptosis, and secretory activity of P4 were assessed to identify miRNA involved in ovarian function (Sirotkin et al. 2010). Pre-miR-15a treatment enhanced the expression of proliferative and apoptotic signals, and promoted the release of P4 and testosterone, but not E2 (Sirotkin et al. 2014).

In vivo experiments performed in mice identified several miRNA in granulosa cells including miR-21, miR-132 and miR-212 that changed significantly following the LH surge (Fiedler et al. 2008). Inhibition of miR-21 by injection of inhibitors beneath the bursa of mice blocked ovulation (Carletti et al. 2010). Conversely, none of these three miRNA have shown effects on ovarian steroidogenesis or luteinization in vitro or in vivo (Carletti et al. 2010; Fiedler et al. 2008; McGinnis et al. 2015). Lastly, the LH receptor (LHr) itself has been shown to be under regulation by miRNA. Studies performed in whole ovary digestion experiments in rats identified miR-122 as positively correlated with an increase of LHr mRNA binding protein (LRBP) after treatment with hCG, through an indirect and unknown mechanism (Menon et al. 2013). The authors postulate that this mechanism may be through the regulation of sterol regulatory element binding proteins-1 and -2 and the temporal patterns of expression does support this claim, yet direct evidence for this remains to be shown. Another study performed using whole rat ovary homogenates identified miRNA-136-3p as being overexpressed 6 h after hCG treatment and bioinformatic analyses indicated that 3'-UTR of the rat LHR has a miR-136-3p site within it. The biologic activity of miR-136 was then tested in primary rat granulosa cell culture and was shown to directly target the rat LHr (Kitahara et al. 2013). It should be noted that bioinformatic (TargetScan) analysis of the human LHr 3'-UTR does not detect a miR-136-3p site within it. This does not preclude the possibility that a miR-136-3p site is present, but considering that the 3'-UTR of genes have the greatest sequence variability, this would need to be tested empirically. It is thus critical that we do not generalize results across species and that until confirmatory results are produced in vivo the functional roles of miRNA can at best be speculative.

7 MicroRNA Regulation Within the Corpus Luteum

The granulosa and thecal cells following exposure to the LH surge, differentiate (i.e., luteinize) to form the corpus luteum (CL). This luteinization process involves the rapid hypertrophy of the granulosa and thecal steroidogenic cells and rapid proliferation of endothelial cells that allows invasion of capillaries around the previous avascular granulosa cells. These profound cellular and tissue remodeling events enable this newly formed tissue to produce exceptionally high P4 levels that are critical for normal cycles and for establishment and maintenance of pregnancy. To date, the female infertility observed in a mouse Dicer hypomorph provides the best evidence that miRNA play a role in luteal function (Otsuka et al. 2008).

The Dicer hypomorphic female mice were infertile due to CL insufficiency resulting, in part, from impaired growth of new capillaries in the ovary (Otsuka et al. 2008). This phenotype was partially due to the lack of miR-17-5p and let7b, two miRNA that function in angiogenesis by down regulating anti-angiogenic tissue inhibitor of metalloproteinase (TIMP1). The angiogenesis defect was rescued by re-administering the miRNA-17-5p and let7b, however full-term pregnancy was not maintained indicating that the miRNA are not the only factors controlling these complex processes. Recently, global expression analysis of miRNA from CL of sheep, pigs and cows have provided additional evidence for miRNA involvement in luteal function.

In sheep, miRNA involved in angiogenesis and cell survival pathways were highlighted in the transition from pre-ovulatory follicle to CL (McBride et al. 2012). In this study, cells were isolated from pre-ovulatory follicles, early CL, late CL and corpus albicans and subjected to RNAseq PCR then confirmed by Northern blot analysis and qRT-PCR (McBride et al. 2012). Overall, the most abundantly expressed miRNA included miR-21, miR-125b, let-7a and let-7b contributing 40 % of all miRNA in the ovarian tissues examined. Some miRNA were expressed at higher levels in the follicles (miR-199a-3p, miR-125b, miR-145 and miR-31) while others were much higher in the CL (miR-503, miR-21 and miR-142-3p) (McBride et al. 2012). Potential gene targets were determined from miRTarBase and it was found that the majority of these miRNA were involved in the regulation of cell cycle, survival, differentiation and angiogenesis. Increases in these pathways makes sense since the CL consists of multiple cell types that must differentiate, multiply and survive long-term in the case of pregnancy.

Global expression analysis of miRNA from CL of pregnant (day 17) versus non pregnant cows identified 15 miRNA that were differentially expressed (Maalouf et al. 2014). Interestingly, similar to studies in sheep (McBride et al. 2012) miR-21, let-7, and miR-199 were among the highest miRNA expressed in the cow ovaries and confirmed an earlier report from the cow (Hossain et al. 2009). Pathway analysis implied miRNA are especially involved in the regulation of cell survival and apoptosis in the bovine CL, a similar theme to that identified in the sheep (McBride et al. 2012). In the pig, miR-378 expression increased during CL development and decreased during luteolysis indicating a possible function in CL formation and maintenance (Ma et al. 2011). Since miR-378 promotes cell survival and

angiogenesis in other cell types (Lee et al. 2007) these known functions support the proposal that miR-378 may play similar roles in the CL (Ma et al. 2011). However, in a more recent study miR-378-3p was found to be inhibitory to PGR expression in pig granulosa cells (Toms et al. 2015). Since the PGR activity is critical to maintenance of the CL (Maybin and Duncan 2004), further studies are needed to understand these contradictory, possibly species-specific roles for miR-378.

These few studies indicate essential roles for miRNAs in luteal function. However, more studies are needed to identify and fully define these pathways. There is also a critical need for *in vivo* based studies to determine the true effects of miRNA on the physiology of cells and tissues.

8 MicroRNA in Reproductive Diseases

With the importance of miRNA in ovarian physiology described above, differences in miRNA expression might be expected in ovarian diseases. Surprisingly, few studies of miRNA function have been conducted on human patients with ovarian diseases and of those few studies that have been reported, only one disease polycystic ovary syndrome (PCOS) is specifically related to endocrine regulation within the ovary. As one of the most common endocrine-metabolic disorders in women of reproductive age (Imbar and Eisenberg 2014; Norman et al. 2007), PCOS results in menstrual disorders and infertility. Since miRNA were first discovered in follicular fluid (da Silveira et al. 2012), additional studies have compared miRNA contents between PCOS and healthy women (Roth et al. 2014; Sang et al. 2013). Combined genome-wide deep sequencing and TaqMan miRNA arrays then followed-up with cell culture experiments were used to identify miRNA involved in follicle steroidogenesis in PCOS patients (Sang et al. 2013). From their microarray results, they selected miRNA that were postulated to function in steroidogenesis in other tissues. MicroRNA-mimics or inhibitors were generated and transfected into a steroidogenic human granulosa-like tumor cell line (KGN) and cultured to determine the intracellular effects of these miRNA. The results identified five miRNA that regulated E2 in KGN cells: miR-24 decreased E2 secretion while miR-132, miR-320, miR-520c-3p and miR-222 all increased E2 secretion. Three miRNA were found to regulate P4 secretion (miR-24, miR-193b and miR-483-5p) (Sang et al. 2013). Follow-up testing of follicular fluid from PCOS and non-PCOS patients confirmed that there were decreased levels of miR-132 and miR-320 in the PCOS patients (Sang et al. 2013). In spite of the decline in miR-132 and miR-320 and that these two miRNA caused an increase in E2 secretion from KGN cells in vitro, the levels of E2 and P4 were not altered in the follicular fluid of PCOS patients. Therefore, the in vivo effects of these miRNA could not be confirmed *in vitro* and further studies are needed to determine if these miRNA serve a critical function within the follicle.

In another study of PCOS patients, 325 miRNA were identified in human follicular fluid and 29 of them showed expression differences between PCOS and control women (Roth et al. 2014). MiR-32, miR-34c, miR-135a, miR18b and miR-9 were more abundant in PCOS patients and pathway analysis correlated them to insulin

regulation and inflammation. Using a bioinformatics analysis, they identified interleukin 8 (IL8), synaptogamin 1 (SYT1) and insulin receptor substrate 2 (IRS2) as potential targets of these miRNA and as expected for targets of miRNA, all three were decreased in the follicular fluid of PCOS patients (Roth et al. 2014). However, in these two studies, there were no overlapping miRNA that were significantly different between the two groups. This may be because one study used Taqman based miRNA arrays (Roth et al. 2014), whereas the other study performed deep RNAseq (Sang et al. 2013) or it may be due to a different population of controls used in each study. These few studies indicate a potential role for miRNA regulation and their perturbation in PCOS, a disease which profoundly affects ovarian endocrine function. The drive for discovery of miRNA therapeutics for the treatment of human disease (Broderick and Zamore 2011), will hopefully initiate more research into the roles of miRNA in ovarian functions such as folliculogenesis, steroidogenesis, corpora lutea and diseases of the ovary and reproduction in general.

9 Conclusions

The studies highlighted in this review indicate an essential role for miRNA in the regulation of endocrine functions of the ovary. However, many studies assess miRNA expression in ovarian tissue composed of multiple cell types and make it difficult to interpret the results. Controlled experiments that demonstrate a functional and physiologic role of miRNA within the tissues of the ovary need to be performed in order to understand the role in each ovarian cell type (i.e., theca, granulosa and luteal cells). Lastly, advances in the field of exosomes and microvesicles have identified novel considerations for the study of miRNA functions, in physiologic and pathophysiologic conditions. This avenue will also surely bring new insight into the role of miRNA in ovarian endocrine functions.

Competing Interests The authors declare that they have no competing interests. *Funding*: National Institute of Child Health and Development: HD061580 to LKC.

References

- Adams BD, Furneaux H, White BA (2007) The micro-ribonucleic acid (miRNA) miR-206 targets the human estrogen receptor-alpha (ERalpha) and represses ERalpha messenger RNA and protein expression in breast cancer cell lines. Mol Endocrinol 21(5):1132–1147. doi:10.1210/ me.2007-0022
- Ahn HW, Morin RD, Zhao H, Harris RA, Coarfa C, Chen ZJ, Milosavljevic A, Marra MA, Rajkovic A (2010) MicroRNA transcriptome in the newborn mouse ovaries determined by massive parallel sequencing. Mol Hum Reprod 16(7):463–471. doi:10.1093/molehr/gaq017
- Al-Nakhle H, Burns PA, Cummings M, Hanby AM, Hughes TA, Satheesha S, Shaaban AM, Smith L, Speirs V (2010) Estrogen receptor {beta}1 expression is regulated by miR-92 in breast cancer. Cancer Res 70(11):4778–4784. doi:10.1158/0008-5472.CAN-09-4104
- Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136(2):215–233. doi:10.1016/j.cell.2009.01.002

- Berezikov E, Chung WJ, Willis J, Cuppen E, Lai EC (2007) Mammalian mirtron genes. Mol Cell 28(2):328–336. doi:10.1016/j.molcel.2007.09.028
- Bernstein E, Kim SY, Carmell MA, Murchison EP, Alcorn H, Li MZ, Mills AA, Elledge SJ, Anderson KV, Hannon GJ (2003) Dicer is essential for mouse development. Nat Genet 35(3):215–217. doi:10.1038/Ng1253
- Bohnsack MT, Czaplinski K, Gorlich D (2004) Exportin 5 is a RanGTP-dependent dsRNAbinding protein that mediates nuclear export of pre-miRNAs. RNA 10(2):185–191
- Broderick JA, Zamore PD (2011) MicroRNA therapeutics. Gene Ther 18(12):1104–1110. doi:10.1038/gt.2011.50
- Carletti MZ, Christenson LK (2009) MicroRNA in the ovary and female reproductive tract. J Anim Sci 87(14 Suppl):E29–E38. doi:10.2527/jas.2008-1331
- Carletti MZ, Fiedler SD, Christenson LK (2010) MicroRNA 21 blocks apoptosis in mouse periovulatory granulosa cells. Biol Reprod 83(2):286–295. doi:10.1095/biolreprod.109.081448
- Cheloufi S, Dos Santos CO, Chong MM, Hannon GJ (2010) A dicer-independent miRNA biogenesis pathway that requires ago catalysis. Nature 465(7298):584–589. doi:10.1038/nature09092
- Christenson LK (2010) MicroRNA control of ovarian function. Animal Reprod/Col Bras Reprod Anim 7(3):129–133
- Clemens JW, Robker RL, Kraus WL, Katzenellenbogen BS, Richards JS (1998) Hormone induction of progesterone receptor (PR) messenger ribonucleic acid and activation of PR promoter regions in ovarian granulosa cells: evidence for a role of cyclic adenosine 3',5'-monophosphate but not estradiol. Mol Endocrinol 12(8):1201–1214. doi:10.1210/mend.12.8.0157
- Cochrane DR, Jacobsen BM, Connaghan KD, Howe EN, Bain DL, Richer JK (2012) Progestin regulated miRNAs that mediate progesterone receptor action in breast cancer. Mol Cell Endocrinol 355(1):15–24. doi:10.1016/j.mce.2011.12.020
- Conneely OM, Mulac-Jericevic B, Lydon JP (2003) Progesterone-dependent regulation of female reproductive activity by two distinct progesterone receptor isoforms. Steroids 68(10–13):771–778
- Couse JF, Yates MM, Walker VR, Korach KS (2003) Characterization of the hypothalamicpituitary-gonadal axis in estrogen receptor (ER) null mice reveals hypergonadism and endocrine sex reversal in females lacking ERalpha but not ERbeta. Mol Endocrinol 17(6):1039–1053. doi:10.1210/me.2002-0398
- da Silveira JC, Veeramachaneni DN, Winger QA, Carnevale EM, Bouma GJ (2012) Cell-secreted vesicles in equine ovarian follicular fluid contain miRNAs and proteins: a possible new form of cell communication within the ovarian follicle. Biol Reprod 86(3):71. doi:10.1095/biolreprod.111.093252
- Dai A, Sun H, Fang T, Zhang Q, Wu S, Jiang Y, Ding L, Yan G, Hu Y (2013) MicroRNA-133b stimulates ovarian estradiol synthesis by targeting Foxl2. FEBS Lett 587(15):2474–2482. doi:10.1016/j.febslet.2013.06.023
- Diez-Fraile A, Lammens T, Tilleman K, Witkowski W, Verhasselt B, De Sutter P, Benoit Y, Espeel M, D'Herde K (2014) Age-associated differential microRNA levels in human follicular fluid reveal pathways potentially determining fertility and success of in vitro fertilization. Hum Fertil 17(2):90–98. doi:10.3109/14647273.2014.897006
- Donadeu FX, Schauer SN, Sontakke SD (2012) Involvement of miRNAs in ovarian follicular and luteal development. J Endocrinol 215(3):323–334. doi:10.1530/JOE-12-0252
- Duffy DM, McGinnis LK, Vandevoort CA, Christenson LK (2010) Mammalian oocytes are targets for prostaglandin E2 (PGE2) action. Reprod Biol Endocrinol 8:131. doi:10.1186/1477-7827-8-131
- Emori MM, Drapkin R (2014) The hormonal composition of follicular fluid and its implications for ovarian cancer pathogenesis. Reprod Biol Endocrinol 12:60. doi:10.1186/1477-7827-12-60
- Espey LL, Richards JS (2002) Temporal and spatial patterns of ovarian gene transcription following an ovulatory dose of gonadotropin in the rat. Biol Reprod 67(6):1662–1670
- Fiedler SD, Carletti MZ, Hong X, Christenson LK (2008) Hormonal regulation of microRNA expression in periovulatory mouse mural granulosa cells. Biol Reprod 79(6):1030–1037. doi:10.1095/biolreprod.108.069690
- Flemr M, Malik R, Franke V, Nejepinska J, Sedlacek R, Vlahovicek K, Svoboda P (2013) A retrotransposon-driven dicer isoform directs endogenous small interfering RNA production in mouse oocytes. Cell 155(4):807–816. doi:10.1016/j.cell.2013.10.001

- Gervasio CG, Bernuci MP, Silva-de-Sa MF, Rosa ESAC (2014) The role of androgen hormones in early follicular development. ISRN Obstet Gynecol 2014:818010. doi:10.1155/2014/818010
- Gonzalez G, Behringer RR (2009) Dicer is required for female reproductive tract development and fertility in the mouse. Mol Reprod Dev 76(7):678–688. doi:10.1002/mrd.21010
- Gougeon A (2010) Human ovarian follicular development: from activation of resting follicles to preovulatory maturation. Ann Endocrinol 71(3):132–143. doi:10.1016/j.ando.2010.02.021
- Gregory RI, Chendrimada TP, Cooch N, Shiekhattar R (2005) Human RISC couples microRNA biogenesis and posttranscriptional gene silencing. Cell 123(4):631–640. doi:10.1016/j.cell.2005.10.022
- Hatzirodos N, Irving-Rodgers HF, Hummitzsch K, Harland ML, Morris SE, Rodgers RJ (2014) Transcriptome profiling of granulosa cells of bovine ovarian follicles during growth from small to large antral sizes. BMC Genomics 15:24. doi:10.1186/1471-2164-15-24
- Hentze MW, Preiss T (2013) Circular RNAs: splicing's enigma variations. EMBO J 32(7):923– 925. doi:10.1038/emboj.2013.53
- Hock J, Weinmann L, Ender C, Rudel S, Kremmer E, Raabe M, Urlaub H, Meister G (2007) Proteomic and functional analysis of argonaute-containing mRNA-protein complexes in human cells. EMBO Rep 8(11):1052–1060. doi:10.1038/sj.embor.7401088
- Hong X, Luense LJ, McGinnis LK, Nothnick WB, Christenson LK (2008) Dicer1 is essential for female fertility and normal development of the female reproductive system. Endocrinology 149(12):6207–6212. doi:10.1210/en.2008-0294
- Hossain MM, Ghanem N, Hoelker M, Rings F, Phatsara C, Tholen E, Schellander K, Tesfaye D (2009) Identification and characterization of miRNAs expressed in the bovine ovary. BMC Genomics 10:443. doi:10.1186/1471-2164-10-443
- Hossain MM, Sohel MM, Schellander K, Tesfaye D (2012) Characterization and importance of microRNAs in mammalian gonadal functions. Cell Tissue Res 349(3):679–690. doi:10.1007/ s00441-012-1469-6
- Hu YC, Wang PH, Yeh S, Wang RS, Xie C, Xu Q, Zhou X, Chao HT, Tsai MY, Chang C (2004) Subfertility and defective folliculogenesis in female mice lacking androgen receptor. Proc Natl Acad Sci USA 101(31):11209–11214. doi:10.1073/pnas.0404372101
- Huang J, Ju Z, Li Q, Hou Q, Wang C, Li J, Li R, Wang L, Sun T, Hang S, Gao Y, Hou M, Zhong J (2011) Solexa sequencing of novel and differentially expressed microRNAs in testicular and ovarian tissues in Holstein cattle. Int J Biol Sci 7(7):1016–1026
- Imbar T, Eisenberg I (2014) Regulatory role of microRNAs in ovarian function. Fertil Steril 101(6):1524–1530. doi:10.1016/j.fertnstert.2014.04.024
- Iwai T, Nanbu Y, Iwai M, Taii S, Fujii S, Mori T (1990) Immunohistochemical localization of oestrogen receptors and progesterone receptors in the human ovary throughout the menstrual cycle. Virchows Arch A Pathol Anat Histopathol 417(5):369–375
- Kaneda M, Tang F, O'Carroll D, Lao K, Surani MA (2009) Essential role for argonaute2 protein in mouse oogenesis. Epigenetics Chromatin 2(1):9. doi:10.1186/1756-8935-2-9
- Kitahara Y, Nakamura K, Kogure K, Minegishi T (2013) Role of microRNA-136-3p on the expression of luteinizing hormone-human chorionic gonadotropin receptor mRNA in rat ovaries. Biol Reprod 89(5):114. doi:10.1095/biolreprod.113.109207
- Knobil E, Neill JD (2014) The physiology of reproduction, 4th edn. Academic Press, San Diego
- Krege JH, Hodgin JB, Couse JF, Enmark E, Warner M, Mahler JF, Sar M, Korach KS, Gustafsson JA, Smithies O (1998) Generation and reproductive phenotypes of mice lacking estrogen receptor beta. Proc Natl Acad Sci USA 95(26):15677–15682
- Lavoie HA, King SR (2009) Transcriptional regulation of steroidogenic genes: STARD1, CYP11A1 and HSD3B. Exp Biol Med 234(8):880–907. doi:10.3181/0903-MR-97
- Lee DY, Deng Z, Wang CH, Yang BB (2007) MicroRNA-378 promotes cell survival, tumor growth, and angiogenesis by targeting SuFu and Fus-1 expression. Proc Natl Acad Sci USA 104(51):20350–20355. doi:10.1073/pnas.0706901104
- Lei L, Jin SY, Gonzalez G, Behringer RR, Woodruff TK (2010) The regulatory role of Dicer in folliculogenesis in mice. Mol Cell Endocrinol 315(1–2):63–73. doi:10.1016/j.mce.2009.09.021
- Li M, Liu Y, Wang T, Guan J, Luo Z, Chen H, Wang X, Chen L, Ma J, Mu Z, Jiang AA, Zhu L, Lang Q, Zhou X, Wang J, Zeng W, Li N, Li K, Gao X, Li X (2011) Repertoire of porcine microRNAs in adult ovary and testis by deep sequencing. Int J Biol Sci 7(7):1045–1055

- Ling YH, Ren CH, Guo XF, Xu LN, Huang YF, Luo JC, Zhang YH, Zhang XR, Zhang ZJ (2014) Identification and characterization of microRNAs in the ovaries of multiple and uniparous goats (Capra hircus) during follicular phase. BMC Genomics 15:339. doi:10.1186/1471-2164-15-339
- Liu WH, Yeh SH, Lu CC, Yu SL, Chen HY, Lin CY, Chen DS, Chen PJ (2009) MicroRNA-18a prevents estrogen receptor-alpha expression, promoting proliferation of hepatocellular carcinoma cells. Gastroenterology 136(2):683–693. doi:10.1053/j.gastro.2008.10.029
- Luense LJ, Carletti MZ, Christenson LK (2009) Role of Dicer in female fertility. Trends Endocrinol Metab 20(6):265–272. doi:10.1016/j.tem.2009.05.001
- Lyon MF, Glenister PH (1980) Reduced reproductive performance in androgen-resistant Tfm/Tfm female mice. Proc R Soc Lond Ser B, Containing papers of a Biological character Royal Society 208(1170):1–12
- Ma T, Jiang H, Gao Y, Zhao Y, Dai L, Xiong Q, Xu Y, Zhao Z, Zhang J (2011) Microarray analysis of differentially expressed microRNAs in non-regressed and regressed bovine corpus luteum tissue; microRNA-378 may suppress luteal cell apoptosis by targeting the interferon gamma receptor 1 gene. J Appl Genet 52(4):481–486. doi:10.1007/s13353-011-0055-z
- Maalouf SW, Liu WS, Albert I, Pate JL (2014) Regulating life or death: potential role of microRNA in rescue of the corpus luteum. Mol Cell Endocrinol 398(1-2):78–88. doi:10.1016/j. mce.2014.10.005
- Maybin JA, Duncan WC (2004) The human corpus luteum: which cells have progesterone receptors? Reproduction 128(4):423–431. doi:10.1530/rep.1.00051
- McBride D, Carre W, Sontakke SD, Hogg CO, Law A, Donadeu FX, Clinton M (2012) Identification of miRNAs associated with the follicular-luteal transition in the ruminant ovary. Reproduction 144(2):221–233. doi:10.1530/REP-12-0025
- McGinnis LK, Luense LJ, Christenson LK (2015) MicroRNA in ovarian biology and disease. In: Molecular Approaches to Reproductive and Newborn Medicine; Bianchi DW and Norwitz ER (Eds). Cold Spring Harbor Perspectives in Medicine, Cold Spring Harbor Laboratoy Press, Cold Spring Harbor, NY
- Menon B, Sinden J, Franzo-Romain M, Botta RB, Menon KM (2013) Regulation of LH receptor mRNA binding protein by miR-122 in rat ovaries. Endocrinology 154(12):4826–4834. doi:10.1210/en.2013-1619
- Mishima T, Takizawa T, Luo SS, Ishibashi O, Kawahigashi Y, Mizuguchi Y, Ishikawa T, Mori M, Kanda T, Goto T, Takizawa T (2008) MicroRNA (miRNA) cloning analysis reveals sex differences in miRNA expression profiles between adult mouse testis and ovary. Reproduction 136(6):811–822. doi:10.1530/REP-08-0349
- Murchison EP, Stein P, Xuan Z, Pan H, Zhang MQ, Schultz RM, Hannon GJ (2007) Critical roles for Dicer in the female germline. Genes Dev 21(6):682–693. doi:10.1101/gad.1521307
- Nagaraja AK, Andreu-Vieyra C, Franco HL, Ma L, Chen R, Han DY, Zhu H, Agno JE, Gunaratne PH, DeMayo FJ, Matzuk MM (2008) Deletion of Dicer in somatic cells of the female reproductive tract causes sterility. Mol Endocrinol 22(10):2336–2352. doi:10.1210/me.2008-0142
- Norman RJ, Dewailly D, Legro RS, Hickey TE (2007) Polycystic ovary syndrome. Lancet 370(9588):685–697. doi:10.1016/S0140-6736(07)61345-2
- Nothnick WB (2012) The role of micro-RNAs in the female reproductive tract. Reproduction 143(5):559–576. doi:10.1530/REP-11-0240
- Orom UA, Nielsen FC, Lund AH (2008) MicroRNA-10a binds the 5'UTR of ribosomal protein mRNAs and enhances their translation. Mol Cell 30(4):460–471. doi:10.1016/j.molcel.2008.05.001
- Ostling P, Leivonen SK, Aakula A, Kohonen P, Makela R, Hagman Z, Edsjo A, Kangaspeska S, Edgren H, Nicorici D, Bjartell A, Ceder Y, Perala M, Kallioniemi O (2011) Systematic analysis of microRNAs targeting the androgen receptor in prostate cancer cells. Cancer Res 71(5):1956– 1967. doi:10.1158/0008-5472.CAN-10-2421
- Otsuka M, Zheng M, Hayashi M, Lee JD, Yoshino O, Lin S, Han J (2008) Impaired microRNA processing causes corpus luteum insufficiency and infertility in mice. J Clin Invest 118(5):1944–1954. doi:10.1172/JCI33680
- Pandey DP, Picard D (2009) miR-22 inhibits estrogen signaling by directly targeting the estrogen receptor alpha mRNA. Mol Cell Biol 29(13):3783–3790. doi:10.1128/MCB.01875-08

- Pastorelli LM, Wells S, Fray M, Smith A, Hough T, Harfe BD, McManus MT, Smith L, Woolf AS, Cheeseman M, Greenfield A (2009) Genetic analyses reveal a requirement for Dicer1 in the mouse urogenital tract. Mamm Genome 20(3):140–151. doi:10.1007/s00335-008-9169-y
- Place RF, Li LC, Pookot D, Noonan EJ, Dahiya R (2008) MicroRNA-373 induces expression of genes with complementary promoter sequences. Proc Natl Acad Sci USA 105(5):1608–1613. doi:10.1073/pnas.0707594105
- Revelli A, Pacchioni D, Cassoni P, Bussolati G, Massobrio M (1996) In situ hybridization study of messenger RNA for estrogen receptor and immunohistochemical detection of estrogen and progesterone receptors in the human ovary. Gynecol Endocrinol 10(3):177–186
- Richards JS (1994) Hormonal control of gene expression in the ovary. Endocr Rev 15(6):725–751. doi:10.1210/edrv-15-6-725
- Ro S, Song R, Park C, Zheng H, Sanders KM, Yan W (2007) Cloning and expression profiling of small RNAs expressed in the mouse ovary. RNA 13(12):2366–2380. doi:10.1261/rna.754207
- Roth LW, McCallie B, Alvero R, Schoolcraft WB, Minjarez D, Katz-Jaffe MG (2014) Altered microRNA and gene expression in the follicular fluid of women with polycystic ovary syndrome. J Assist Reprod Genet 31(3):355–362. doi:10.1007/s10815-013-0161-4
- Ruby JG, Jan CH, Bartel DP (2007) Intronic microRNA precursors that bypass Drosha processing. Nature 448(7149):83–86. doi:10.1038/nature05983
- Salilew-Wondim D, Ahmad I, Gebremedhn S, Sahadevan S, Hossain MD, Rings F, Hoelker M, Tholen E, Neuhoff C, Looft C, Schellander K, Tesfaye D (2014) The expression pattern of microRNAs in granulosa cells of subordinate and dominant follicles during the early luteal phase of the bovine estrous cycle. PLoS One 9(9), e106795. doi:10.1371/journal.pone.0106795
- Sang Q, Yao Z, Wang H, Feng R, Wang H, Zhao X, Xing Q, Jin L, He L, Wu L, Wang L (2013) Identification of microRNAs in human follicular fluid: characterization of microRNAs that govern steroidogenesis in vitro and are associated with polycystic ovary syndrome in vivo. J Clin Endocrinol Metab 98(7):3068–3079. doi:10.1210/jc.2013-1715
- Sen A, Prizant H, Light A, Biswas A, Hayes E, Lee HJ, Barad D, Gleicher N, Hammes SR (2014) Androgens regulate ovarian follicular development by increasing follicle stimulating hormone receptor and microRNA-125b expression. Proc Natl Acad Sci USA 111(8):3008–3013. doi:10.1073/pnas.1318978111
- Shao R, Markstrom E, Friberg PA, Johansson M, Billig H (2003) Expression of progesterone receptor (PR) A and B isoforms in mouse granulosa cells: stage-dependent PR-mediated regulation of apoptosis and cell proliferation. Biol Reprod 68(3):914–921
- Shimada M, Terada T (2002) FSH and LH induce progesterone production and progesterone receptor synthesis in cumulus cells: a requirement for meiotic resumption in porcine oocytes. Mol Hum Reprod 8(7):612–618
- Sirotkin AV, Kisova G, Brenaut P, Ovcharenko D, Grossmann R, Mlyncek M (2014) Involvement of microRNA Mir15a in control of human ovarian granulosa cell proliferation, apoptosis, steroidogenesis, and response to FSH. MicroRNA 3(1):29–36
- Sirotkin AV, Laukova M, Ovcharenko D, Brenaut P, Mlyncek M (2010) Identification of microR-NAs controlling human ovarian cell proliferation and apoptosis. J Cell Physiol 223(1):49–56. doi:10.1002/jcp.21999
- Sirotkin AV, Ovcharenko D, Grossmann R, Laukova M, Mlyncek M (2009) Identification of microRNAs controlling human ovarian cell steroidogenesis via a genome-scale screen. J Cell Physiol 219(2):415–420. doi:10.1002/jcp.21689
- Sohel MM, Hoelker M, Noferesti SS, Salilew-Wondim D, Tholen E, Looft C, Rings F, Uddin MJ, Spencer TE, Schellander K, Tesfaye D (2013) Exosomal and non-exosomal transport of extracellular microRNAs in follicular fluid: implications for bovine oocyte developmental competence. PLoS One 8(11), e78505. doi:10.1371/journal.pone.0078505
- Stocco C (2008) Aromatase expression in the ovary: hormonal and molecular regulation. Steroids 73(5):473–487. doi:10.1016/j.steroids.2008.01.017
- Suh N, Baehner L, Moltzahn F, Melton C, Shenoy A, Chen J, Blelloch R (2010) MicroRNA function is globally suppressed in mouse oocytes and early embryos. Curr Biol 20(3):271–277. doi:10.1016/j.cub.2009.12.044

- Tang F, Kaneda M, O'Carroll D, Hajkova P, Barton SC, Sun YA, Lee C, Tarakhovsky A, Lao K, Surani MA (2007) Maternal microRNAs are essential for mouse zygotic development. Genes Dev 21(6):644–648. doi:10.1101/gad.418707
- Tay Y, Rinn J, Pandolfi PP (2014) The multilayered complexity of ceRNA crosstalk and competition. Nature 505(7483):344–352. doi:10.1038/nature12986
- Toms D, Xu S, Pan B, Wu D, Li J (2014) Progesterone receptor expression in granulosa cells is suppressed by microRNA-378-3p. Mol Cell Endocrinol. doi:10.1016/j.mce.2014.07.022
- Toms D, Xu S, Pan B, Wu D, Li J (2015) Progesterone receptor expression in granulosa cells is suppressed by microRNA-378-3p. Mol Cell Endocrinol 399:95–102. doi:10.1016/j. mce.2014.07.022
- Tripurani SK, Xiao C, Salem M, Yao J (2010) Cloning and analysis of fetal ovary microRNAs in cattle. Anim Reprod Sci 120(1-4):16–22. doi:10.1016/j.anireprosci.2010.03.001
- Wilczynska A, Bushell M (2015) The complexity of miRNA-mediated repression. Cell Death Differ 22(1):22–33. doi:10.1038/cdd.2014.112
- Xu S, Linher-Melville K, Yang BB, Wu D, Li J (2011) Micro-RNA378 (miR-378) regulates ovarian estradiol production by targeting aromatase. Endocrinology 152(10):3941–3951. doi:10.1210/en.2011-1147
- Yang WJ, Yang DD, Na SQ, Sandusky GE, Zhang Q, Zhao GS (2005) Dicer is required for embryonic angiogenesis during mouse development. J Biol Chem 280(10):9330–9335. doi:10.1074/ jbc.M413394200
- Yao G, Yin M, Lian J, Tian H, Liu L, Li X, Sun F (2010a) MicroRNA-224 is involved in transforming growth factor-beta-mediated mouse granulosa cell proliferation and granulosa cell function by targeting Smad4. Mol Endocrinol 24(3):540–551. doi:10.1210/me.2009-0432
- Yao N, Yang BQ, Liu Y, Tan XY, Lu CL, Yuan XH, Ma X (2010b) Follicle-stimulating hormone regulation of microRNA expression on progesterone production in cultured rat granulosa cells. Endocrine 38(2):158–166. doi:10.1007/s12020-010-9345-1
- Yin M, Lu M, Yao G, Tian H, Lian J, Liu L, Liang M, Wang Y, Sun F (2012) Transactivation of microRNA-383 by steroidogenic factor-1 promotes estradiol release from mouse ovarian granulosa cells by targeting RBMS1. Mol Endocrinol 26(7):1129–1143. doi:10.1210/ me.2011-1341
- Yuan S, Ortogero N, Wu Q, Zheng H, Yan W (2014) Murine follicular development requires oocyte DICER, but not DROSHA. Biol Reprod 91(2):39. doi:10.1095/biolreprod.114.119370

Translational Control in the Germ Line

7

Marco Conti, João Pedro Sousa Martins, Seung Jin Han, and Federica Franciosi

1 Introduction

Given their critical function in transmitting the genetic information across generations, germ cells are unique cells of the body and the only ones with the capacity to generate a complete organism. They use a specialized meiotic cell cycle to generate haploid cells, and, although differentiating into one of the most specialized cell of the body, they maintain some of the traits specific to pluripotent cells. To accomplish this, they often use unusual molecular strategies to control gene expression during their life cycle. Among these unique strategies is the extensive use of posttranscriptional control of mRNA to regulate important transitions throughout lineage specification, lineage expansion, and differentiation. In model organisms, maternal mRNAs accumulated during growth drive development up to the midblastula transition prior to gastrulation and with few exceptions in the absence of significant transcription (Lee et al. 2014). Moreover, the polarity of the female gamete is defined by mRNA deposited at one pole of the oocyte marking the region of the cytoplasm that will be inherited by and will define the germ lineage. The presence of the so called "germ granules" in the cytosol of most germ cells is the physical manifestation of mRNA storage/accumulation found in one form or another in most organisms. Although mammalian embryos have adopted a different strategy and become independent of maternal input earlier during embryo development (Li et al.

Center for Reproductive Sciences, University of California, San Francisco, CA, USA

S.J. Han Department of Biological Sciences, Inje University, Gimhae, Republic of Korea

© Springer International Publishing Switzerland 2016

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_7

M. Conti (🖂) • J.P.S. Martins • F. Franciosi

Eli and Edythe Broad Center of Regeneration Medicine and Stem Cell Research, University of California, San Francisco, CA, USA

Department of Obstetrics and Gynecology and Reproductive Sciences, University of California, San Francisco, CA, USA e-mail: contim@obgyn.ucsf.edu

2010), regulation of translation, in the absence of transcription, is the only form of gene expression that drives oocyte maturation and early embryo development up to the activation of the zygote genome (ZGA) (Clarke 2012). Similarly, translational regulations control critical spermatogenesis steps at the time of transition from mitosis to meiosis, during meiosis itself, as well as during the differentiation of the haploid gamete into a spermatozoon.

The teleological explanation for the extensive use of translational regulations on a background of reduced or absent transcription is matter of debate. It may include the need to rapidly reprogram the genome to totipotency in a short period of time, hence the need to transfer regulation of gene expression to the cytoplasm; it may be due to the need to generate polarity in a large cell such as the oocyte, or to induce rapid changes in protein expression necessary for progression through the meiotic cell cycle. Regardless of the actual reason, translational controls are intimately interwoven with the generation of a totipotent cell and the shift in gene expression control from maternal to embryonic.

Translation of a mRNA is initiated by the assembly of a pre-initiation complex that interacts with the 5' 7-methyl-guanosine cap structure of the mRNA, and involves the eukaryotic initiation factor 4E (eIF4E) complex and the 40S subunit of the ribosome. This is followed by scanning of the 5'UTR, recognition of the translation initiation site, recruitment of the large 60S subunit of the ribosome, and elongation. For a subset of translational regulations, the cap complex interacts with additional complexes assembled at the other end of the mRNA, the 3'UTR. *Cis*acting elements and cognate RNA binding proteins (RBPs) are responsible for the organization of these complexes. The complex assembled on the 3'UTR of a mRNA often controls the length of the poly(A) of a mRNA or polyadenylation, one of the most common mechanism involved in promoting translation. Indeed, changes in poly(A) tail length mediated by the cytoplasmic polyadenylation element binding protein, CPEB, was one of the first to be discovered.

An in depth analysis of the mechanisms of translational initiation are outside the scope of this review and the reader is directed to comprehensive recent review of the field. Due to space limitation, we will also not review the role of non-coding RNAs in translational regulation in germ cells. Here, after a brief summary of the major regulatory circuits controlling translation in germ cells in model organisms and conserved during evolution, we will concentrate on the most relevant mechanisms involved in the development of the male and female gamete in mammals and their function up to the formation of the zygote. The theme that will emerge from this survey is that a set of translational regulators are used over and over again through the life cycle of germ cells from worms to humans. This survey is not meant to be comprehensive but rather provide some basic concepts on the mechanisms controlling translation during germ cell development and the most important *trans*-acting proteins involved.

2 Control of Translation during Gametogenesis in Model Organisms

2.1 Translational Control of the Mitosis-to-Meiosis Transition in *Caenorhabditis elegans*

The gonad of the XX hermaphroditic nematode produces both sperm and oocytes whereas the XO male nematode contains only the spermatogenic lineage (Kimble and Crittenden 2007). The mechanisms of gamete production in the two states are similar and have been a powerful model to discover regulatory circuits involving RBPs. A cluster of Germ Stem Cells (GSCs) at the distal tips of the gonad interacts with somatic cells of the niche required for maintenance of the pluripotency state and to control self-renewal of these adult stem cells (Kimble and Crittenden 2007). Once they move away from this niche, GSCs continue with mitotic divisions until they reach a transition point where they enter into meiosis. Completion of meiosis is followed by differentiation into the haploid gamete. Several critical decisions are made during the germ cells differentiation, including the decision to maintain the proliferative state and the pluripotency state, the decision to transition from the mitotic cell cycle to a meiotic cell cycle, and the commitment to final differentiation to a haploid cell. These transitions depend on translational regulations and the function of a number of RBPs or proteins involved in RNA metabolism. Here we will highlight some of the major concepts on the organization of these circuits, underscoring how the same regulatory cassettes may be utilized during mammalian gametogenesis.

Self-renewal and the mitotic cell cycle in Caernorhabditis elegans GSCs are maintained through Notch signals originating from the somatic niche (Kimble and Simpson 1997); this controls expression of several key regulators including two RBPs with critical role for the GSC fate. Interestingly, Notch signaling is also involved in mammalian gametogenesis, being essential for formation of the ovarian follicle (Vanorny et al. 2014) and contributing to the testis stem cell niche (Garcia et al. 2014). The PUF proteins, Fbf-1 and Fbf-2 (collectively known as Pumilio and FBF proteins or Pumilios in mammals), function as hubs of complex networks within germ cells controlling both self-renewal and the mitosis/meiosis decision (Kimble 2011). These proteins act predominantly as translational repressors of a large number of mRNAs. Among the critical targets are mRNAs coding for components of the meiotic cell cycle machinery, for transcription factors involved in differentiation, as well as RBPs or mRNA modifiers required for entry into meiosis (Kimble 2011). Meiotic entry occurs prematurely in worms double mutant for *fbf-1* and fbf-2. The gonad of these mutants produces only sperm indicating an additional role for these RBPs at later stages during gametogenesis. Puf proteins repress the translation of several genes that are required to enter into meiosis including gld-1, gld-2, and gld-3. Gld1 is a member of the STAR/GSG/quaking-type RBP family and functions as a translational repressor of cyclin E (Marin and Evans 2003), a cyclin required for maintaining the mitotic cell cycle in germ cells. Gld-2 and Gld-3 form a heterodimer with the function of a cytoplasmic poly(A) polymerase and

translational activators. By repressing the expression of these components necessary at different stages of the germ cell differentiation, the Puf protein maintains the germ cell mitotic state. Mechanistically, the Puf proteins repress translation through several mechanisms that include recruitment of the Ccr4-Pop2-NOT deadenylase complex (Goldstrohm et al. 2006), interaction with Ago to stall translation (Friend et al. 2012) or by interacting with other deadenylases (Quenault et al. 2011).

Nos-3 is one of the three *C. elegans* homologs of Drosophila Nanos (Kraemer et al. 1999). In the sperm/oocyte switch of the hermaphrodite germline, Fbf together with Nos-3 represses *gld-1* and *fem-3* mRNAs and promotes oogenesis (Kraemer et al. 1999; Crittenden et al. 2002).

Fog-1 (feminization of the germ line 1) is a *C. elegans* homolog of the cytoplasmic polyadenylation element binding protein (CPEB) which is required for sperm development. Mutations affecting the *fog-1* locus cause a phenotype of sexual reversal with cells that would produce sperm instead differentiate into oocytes (Barton and Kimble 1990). There are additional members of the family of CPEBs with critical function at different stages of gamete development in the nematode. *Cpb-1* mutants do not produce sperm but this RBP functions at a later stage of spermatogenesis compared to Fog-1 (Luitjens et al. 2000). Cpb-3 has been implicated in meiotic progression (Luitjens et al. 2000) functioning in concert with the interacting partner Daz-1. Worms double mutants for *cpb-3* and *daz-1* are sterile (Maruyama et al. 2005). Both functional and physical interactions between members of the Cpb and Fbf proteins have been described. For instance, Fbf-2 and Cpb-1 form a complex that represses translation (Campbell et al. 2012).

Thus, numerous protein/protein interactions and feedbacks contribute to the stability of the control system and its robustness as well as to enforcing irreversibility of the decision switch. Many of these RBPs and some of the feedback arrangements are maintained in the mammalian germ line (see below) (Table 7.1).

Family of RBPs	C. elegans	D. melanogaster	X. laevis	M. musculus	H. sapiens
PUF	Fbf-1	Pum	Pum1	PUM1	PUM1
	Fbf-2		Pum2	PUM2	PUM2
	Puf-8				
Nanos	Nos-3	Nos	Nanos1		
			Nanos2	NANOS2	NANOS2
			Nanos3	NANOS3	NANOS3
CPEB	Fog-1	Orb	Cpeb1	CPEB1	CPEB1
	Cpb-1	Orb2	Cpeb3	CPEB3	CPEB3
	Cpb-3		Cpeb4	CPEB4	CPEB4
DAZ	Daz-1	Bol		BOLL	BOLL
			Dazl	DAZL	DAZ1 DAZLA
DAZAP		Hrb27C	Dazap1	DAZAP1	DAZAP1
			Dazap2	DAZAP2	DAZAP2

Table 7.1 Orthologs of the most relevant RBPs involved in the control of translation in the germline of invertebrate and vertebrate animal models, and in mammals

2.2 Translational Control of the Mitosis to Meiosis Transition in Drosophila

At the tip of the ovariole in the Drosophila germarium, two germ line stem cells (GSC) divide asymmetrically into a daughter cell which is committed to differentiate and one that replenishes the pool of stems cells (Lin and Spradling 1993; Xie and Spradling 2000). The committed cell, termed cystoblast (CS), undergoes four round of mitosis with incomplete cytokinesis as these cells are connected by cytoplasmic bridges, the ring canals. Two cells of the 16 cell cyst are connected by four cytoplasmic bridges whereas the other cells are connected by three or fewer bridges. One of the cells connected with four bridges will become an oocyte and the other cells will differentiate into nurse cells. Once the oocyte is designated, it receives material from the nurse cells via these ring canals.

Like in nematode gametogenesis, the GSC/CS and mitotic/meiotic transition in Drosophila is regulated by mechanism involving predominantly translational control. Like in worms, Pumilio (Pum) and Nanos (Nos) play a central function in repression of transcripts required for differentiation of the gametes (Miller and Olivas 2011).

The bag-of-marbles (Bam) and benign gonial cell neoplasm (Bgcn) proteins function in both males and females to promote gamete differentiation. Bam is expressed at a critical time when the cystoblast replicates and becomes committed to meiosis entry (Chen et al. 2011b; Slaidina and Lehmann 2014). Bam and Bgcn form a complex that binds and antagonizes both Pumilio and eIF4A proteins. An additional protein included in the complex required for inhibition of GSC factors is Mel-P26, a member of the tripartite motif containing TRIM family of proteins (Li et al. 2013). They also regulate the mRNA coding for another RBP, Nanos, via its 3'UTR (Li et al. 2009; Shen et al. 2009; Kim et al. 2010). Therefore, Bam and Bgcn are part of a network of RBPs that control differentiation of GSCs. A mammalian homolog of Bam was identified and termed GM114 (Tang et al. 2008). However, disruption of this locus has no effect on spermatogenesis (Tang et al. 2008). Since GM114 is remotedly related to Bam, it is still possible that another Bam-ortholog is present in the mammalian genome and involved in translational control during gametogenesis.

Similar players are also involved in differentiation of GSC into a gonioblast of the Drosophila testis (Gonczy et al. 1997; Fuller 1998). However, *bam* may also be required for entry into meiosis.

2.3 Translational Control of Egg and Embryo Polarity in Drosophila

A fascinating concept based on observation in model organism is that patterning during development is generated by mRNA localization and restricted translation in distinct subcellular domains of a cell. The *Drosophila* oocyte and embryo is one of the best understood experimental models in terms of development of polarity
generated by mRNA localization/translation. In Drosophila the specification of the anterior fate of the embryo is dependent on the localization of maternal mRNA bicoid (*bcd*) whereas the development of the posterior end depends on the proper localization of *osk* and *nanos* (Kugler and Lasko 2009). The localization and translation of these mRNAs generate the anterior/posterior axis. Here we will highlight some very basic concept regarding these mechanisms whereas exhaustive reviews of this process are available (Kugler and Lasko 2009; Ghosh et al. 2012).

Oskar mRNA localization coupled to its translational control is essential for normal development. *Osk* mRNA is produced by nurse cells and transferred to the oocyte by a complex of proteins including components of the exon junction complex, Staub, and the eIF4E interacting protein Cup and Bruno. Association of the complex with dynein and with microtubules is necessary for the transport. Once in the oocyte, *osk* localizes in a discrete cytoplasmic structure called the Balbiani body. The site of Oskar protein synthesis determines where germ plasm is assembled. Repression of Osk mRNA translation during this journey is dependent on the eIF4E interacting protein Cup in complex with the RBP Bruno. A Cup ortholog is expressed in mouse oocytes (Clast4) but its function in this species is unknown (Villaescusa et al. 2006). Recently, Clast4 was identified in a screen for genes important for oocyte meiotic reentry although its function was not explored further (Pfender et al, 2015).

Gurken protein is a member of the transforming growth factor-a (TGF-A) family, which functions as ligand for the Drosophila EGF receptor (DER or torpedo). Local mRNA accumulation, local secretion of Gurken and localized occupancy of the EGFR on follicular cells is essential for establishing the antero/posterior axis of the oocyte and the embryo. EGF-like growth factors also play a critical role in oocyte maturation and ovulation in mammals (Conti et al. 2006).

There are two CPEB-like proteins expressed in Drosophila termed Orb (ool8 RNA binding) and Orb2; both of them are required for germ cell development. *Orb* mutant flies are sterile (Lantz et al. 1994). Null mutant for *orb* prevents the last replication, and failure to produce oocytes or nurse cells. Later on during differentiation, the Orb protein is localized in the differentiating oocyte indicating an important function in this cell.

Orb undergoes well defined patterns of localization during oocyte development accumulating first in the anterior pole and becoming gradually enriched in the posterior pole of the oocyte. During early embryogenesis, this cytoplasm region will be inherited by primordial germ cells (PGCs). This localization of Orb is necessary for the localization and translation of the *osk* and Gurken mRNAs. The 3'UTR of *osk* mRNA plays a critical role in both localization and translation. While *osk* mRNA is localized to the posterior pole, Gurken mRNA is first deposited to the posterior region of the oocytes and then is translocated together with the nucleus to the anterior dorsal region (Cheung et al. 2011). *Orb* mutations interfere with both localization processes. Gurken acts both early and late in the oocyte assembly process (Cheung et al. 2011). Orb is involved in both phases of Gurken mRNA localization, Orb also acts both early and late in oocyte assembly (Christerson 1994). From the mechanistic standpoint, it has been shown that Orb and

poly-adenosine polymerase (Pap) control the polyadenylation and translation of oskar mRNA confirming their function as translational activators.

2.4 Translational Control of the Meiotic Cell Cycle in Danio Rerio Oocytes

Oocyte maturation in zebrafish is triggered by MIH (17a,20b-Dihydroxy-4-pregnen-3-one) and governed by activation of pre-MPF, which consists of cyclin B and inactive Cdk1 (Brownlie et al. 2003). As with Drosophila, some mRNAs are deposited in zebrafish oocytes in a polarized fashion. Cyclin B mRNA is concentrated in RNA granules along the cytoplasm at the animal pole through an interaction with the RNA binding protein Pumilio1. Formation of RNA granules depends on actin filaments since cytochalasin treatment leads to cyclin B mRNA release from the granules. After stimulation by MIH, the cyclin B mRNA disperses into the cytoplasm, facilitating its translational activation (Kotani et al. 2013). Cyclin B mRNA localization within the oocyte is dependent on sequence elements within the open reading frame (ORF) as well as the 3'UTR (Yasuda et al. 2010).

As in other species, zebrafish cyclin B mRNA is regulated by RBPs. Recently Pumilo and Igf2bp3 (insulin-like growth factor 2 mRNA-binding protein 3) were reported to repress translation by binding the 3'UTR of cyclin B (Takahashi et al. 2014; Kotani et al. 2013). Compared to other model organisms CPEB dependent translation of cyclin B mRNA is considerably less efficient, probably due to different positioning and sequence of the CPEs (Zhang and Sheets 2009), indicating that the optimized level of cyclin B for the oocyte maturation is species specific. Zebrafish oocytes express two classes of CPEBs, the oocyte specific Zorba together with the embryonic type ElrA (O'Connell et al. 2014). Also Zorba mRNA localizes at the animal pole at stage II of oogenesis (GV stage). This localization seems to be independent of microtubules and microfilaments (Bally-Cuif et al. 1998).

2.5 Translational Control of the Meiotic Cell Cycle in *Xenopus laevis* Oocytes

Frog oocytes have been a model used for extensive studies on the role of translational control of germ cells. Cpeb was first described in frog oocytes and much of what is known about the function of this protein comes from studies in Xenopus. In frogs like in most species studied, maternal mRNAs accumulate during oocyte growth but they are not translated into proteins (Clarke 2012). The polyadenylation state of these mRNAs is thought to be central to the repressed state and for translational activation (Richter 2007; Charlesworth et al. 2013). In frog oocytes, a large macromolecular complex is formed involving RBPs (Cpeb1, Cpsf, ePab), an adenylase (Gld2), a deadenylase (Parn), and the scaffold/adaptors Symplekin and Maskin, as well as components of the cap complex (eIF4E, eIF4E-T). This complex is thought to prevent translation by maintaining a short poly(A) tail (Fernandez-Miranda and Mendez 2012; Ivshina et al. 2014). Other complexes have been described as important for repression, including Pumilio proteins (Kim and Richter 2007) or Cup/Orb in Drosophila (Wong and Schedl 2011). Further function of these complexes is discussed below in the context of mammalian oocyte.

3 RBPs and Translational Control in Mammalian Primordial Germ Cells (PGCs)

In the mouse fetal gonad, male PGCs exit mitotic cycle and enter a period of quiescence, whereas female PGCs become committed to enter meiosis. The signal for this decision is extrinsic and derives from somatic cells. It is commonly accepted that retinoic acid (RA) induces female germ cells to enter meiosis and that this signal is suppressed in the male through rapid metabolism of RA (Bowles et al. 2006). A key regulator of meiotic cell cycle entry is stimulated by retinoic acid 8 (Stra8) with probable function of transcription regulator. However, RBPs play a key role in the control of Stra8 expression. Nanos 2 and 3 proteins are essential for PGC development in a dymorphic fashion with Nanos 2 expressed only in male germ cells. Nanos 2 function is to promote the male fate but repress the female fate in germ cells (Saga 2010). Nanos acts as a translational repressor that prevents meiotic entry by suppressing translation of several mRNAs including *Stra8* itself.

The RBP DAZL (deleted in Azoospermia-like) is also considered an important component required for entry into meiosis. It has been proposed that DAZL has a permissive function as a "licensing" factor, priming germ cells to respond to molecular cues and engage germ cells in sex differentiation (Gill et al. 2011) but the molecular mechanisms involved are unclear. Traditionally DAZL is viewed as an activator of translation but it has been recently proposed that it may also function as repressor. Indeed DAZL interacts with Pumilio proteins which are usually repressor of translation (Urano et al. 2005). CPEB1 also plays a role because CPEB1 knock-out prevents PGC entry into meiosis and SYCP3 expression (Tay and Richter 2001). CPEB is necessary for DAZL protein accumulation and it is likely that CPEB1 and DAZL function sequentially and synergistically during female PGC commitment to enter meiosis.

4 Translation Control during Adult Mammalian Spermatogenesis

In the adult mammalian male gonad, the production of gametes is divided into three stages: a proliferation stage, where spermatogonia undergo five rounds of mitotic division (Oatley and Brinster 2008; Hermo et al. 2009), a maturation phase, where diploid spermatocytes undergo two meiotic divisions to yield haploid spermatids (Zickler and Kleckner 1999) and spermiogenesis where haploid spermatids differentiate into spermatozoa (Braun 2001). The progression through each of these

events relies on a tightly controlled network of specific genes regulated at the transcription and post-transcriptional level.

The temporal regulation of the protamine 1 (Prm1) mRNA translation during spermiogenesis is a classic example of translational regulation and uncoupling of transcription and translation. Transcription of this mRNA initiates during the round spermatid stage (Braun et al. 1989) but the mRNA is not translated immediately. The translation of the *Prm1* mRNA is dependent on a highly conserved sequence, the translational repression being mediated by a 17-nucleotide translational control element (TCE) located in the 3'UTR of this mRNA. Mutation of this TCE cis-acting element causes premature synthesis of protamine protein and sterility. The Prm1 mRNA is stored as a cytoplasmic ribonucleoprotein (mRNP) particle in spermatids that are also required for maintaining this repressed state. Translational activation of stored *Prm1* mRNA in elongated spermatids requires the trans-activation responsive RNA-binding protein 2 (TARBP2). Mice mutant for Tarbp2 are defective in proper translational activation of the Prm1 and Prm2 mRNAs and are sterile (Zhong et al., 1999). Tarbp2 is expressed at high levels in post-meiotic spermatids. The cold shock domain Mouse Y-box protein 2 (MSY2) is one of the proteins responsible for repression of Prm1. This is believed to prime transcripts (with a Y-box region within their promoter) to cytoplasmic repression and RNA stability (Yang et al. 2005). In MSY2 knockout models, no spermatozoa are found in the seminiferous tubules, in part due to reduced chromatin condensation (Yang et al. 2005)

Additional RBPs have been described both in the nuclear and cytoplasmic components of cells, some of which shuttle between the nucleus and cytoplasm. Once in the cytoplasm, transcripts are encapsulated in RNPs stabilized by RBP protein–protein and protein–RNA interactions. The effect on mRNA processing by RBPs is further augmented by post-transcriptional modifications of the RBPs (methylation, ubiquitination, phosphorylation, acetylation) that remodel the structure of RNPs and alter their activity (Bettegowda and Wilkinson 2010; Idler and Yan 2012). For example, knockdown of the KH domain-containing protein SAM68 results in infertility associated with germ cell apoptosis and aberrant elongating spermatids. This protein binds polyadenylated transcripts with the AUUAAA motif and upon phosphorylation translocates from the nucleus to the cytoplasm promoting function of signal transduction, alternative splicing and translation upregulation (Lin et al. 1997; Najib et al. 2005; Paronetto et al. 2009; Paronetto and Sette 2010). Here we will focus on the properties and function of four classes of RBPs that are critical for spermatogenesis progression (Fig. 7.1).

4.1 Deleted in Azoospermia Family of RBP

The germ cell specific Deleted in Azoospermia (DAZ) family of proteins is composed of three members; the primate and catarrhine lineage specific protein DAZ, located within the long arm of the Y chromosome (Reijo et al. 1995); the autosomal and family ancestor Boule, with homologs described from flies and worms to mice and humans (Karashima et al. 2000); and the vertebrate specific DAZ-like (DAZL)



Fig. 7.1 Expression profile of RBPs throughout mouse spermatogenesis (**a**) and oogenesis (**b**). Progression through gametogenesis is highly regulated at the translational level, both in the mitotic and meiotic phases. Developmental progression is regulated by complex network of RBPs which control, translation activation, repression and RNA stability. The following references contain details on the expression of genes expressed during spermatogenesis and ovogenesis: Pum2 (Xu et al. 2007), Nanos 2 (Barrios et al. 2010), Nanos 3 (Lolicato et al. 2008), Dazl (Ruggiu et al. 1997; Saunders et al. 2003; Chen et al. 2011a), Boule (Xu et al. 2001). Sam68 (Paronetto et al. 2009), Elavl2 (Chalupnikova et al. 2014), CPEB1 (Tay and Richter 2001; Chen et al. 2011a), Musashi (Gunter and McLaughlin 2011)

(Bielawski and Yang 2001). This family of proteins owes its name to the phenotype of patients were it was first identified. In a genetic screen of azoospermic men, 14 % of cases had deletions within the area containing the DAZ coding region (Reijo et al. 1995). The other members of the DAZ family have also been associated with male infertility as reported in several SNPs studies (Chen et al. 2010).

All members of the DAZ family are required during spermatogenesis. DAZ and Boule are necessary for spermatid development, with Boule-null mice showing a developmental arrest at step 6 of spermatids (VanGompel and Xu 2010). DAZL phenotype can be detected much earlier compared to its orthologues, with a spermatogenic arrest at prophase of meiosis I (Ruggiu et al. 1997; Saunders et al. 2003). This phenotype becomes more penetrant in a pure BL6/C57 background, with apoptosis observed at embryonic day 15.5 of mouse development (Lin and Page 2005). Genetic studies have associated this phenotype with an impairment of a canonical PGC feature, sex differentiation.

The nuclear and cytoplasmic localization of DAZL (Xu et al. 2001), the ability to bind mRNA (Tsui et al. 2000b; Venables et al. 2001), the association with polyribosomes (Tsui et al. 2000b) and the interaction with several known RBPs (Brook et al. 2009) suggest a role in mRNA processing. Several molecular functions have been associated with DAZL, ranging from mRNA repression (Urano et al. 2005; Padmanabhan and Richter 2006) mRNA transport (Lee et al. 2006; Kim et al. 2012), miRNA-mediated repression protection (Takeda et al. 2009), but to date the best characterized role for DAZL is in translation activation (Collier et al. 2005; Reynolds et al. 2005, 2007; Chen et al. 2011a). In vitro studies have demonstrated that DAZ family members all share a common function as translation activators. Tethering assays, where DAZL-RNA binding activity is circumvented, showed that this family upregulates translation of artificially bound luciferase reporters (Collier et al. 2005). In vivo experiments have since demonstrated translation of endogenous transcripts (Reynolds et al. 2005, 2007; Chen et al. 2011a). Insight into the mechanism of DAZ-family translation activation demonstrates that these proteins act at the step of translation initiation. A direct interaction with PABP1 (but none of the other canonical translation initiation factors) was shown to be necessary to promote translation in a reconstitution system. The observation that DAZL was able to stimulate translation of non-adenylated reporter mRNAs indicated that this can work distinctly from the canonical CPEB translation activator (Pique et al. 2008) by promoting PABP recruitment in a poly(A) independent manner (Collier et al. 2005).

DAZL interacts with transcripts with a U-rich 3'UTR (Tsui et al. 2000b). A variety of assays identified the trinucleotide GUU has the minimal binding motif of DAZL (Tsui et al. 2000b; Ruggiu and Cooke 2000; Venables et al. 2001). The crystal structure of DAZL RNA recognition motif (RRM) demonstrated that this presents the highest affinity towards GUU[U/C] sequences (Jenkins et al. 2011). The relatively short sequence of this element greatly challenges the identification of DAZL targets. In vitro evidence for the requirement of multiple DAZL binding sites for a maximal translation activation can in the future help to overcome this challenge (Collier et al. 2005; Chen et al. 2011a). Several studies have identified DAZL targets (Brook et al. 2009). In spermatogenesis, pull down experiments have demonstrated an interaction with messages involved in sperm development (Tpx1, Grsf-1, Trf-2, TssK proteins, etc.) (Jiao et al. 2002; Zeng et al. 2008). Nonetheless, to date the only two bona fide spermatogenic targets for DAZL are Sycp3 and Mvh (Reynolds et al. 2005, 2007).

4.2 Pumilio Family of RBPs

Whereas the role of Puf proteins in model organisms have been extensively investigated, comparable little information is available on the role of Pumilio proteins, Pum1 and Pum2, in spermatogenesis and oogenesis in mammals. A gene trap disruption of *Pum2* was reported to produce a decrease in testis weight and some disruption in spermatogenesis but the exact steps affected were not investigated (Xu et al. 2007). Pum1 knockout mice have decreased fertility and decreased sperm count and 25–30 % reduction in testis weight. It has been proposed that this is due to deregulation of the apoptotic pathway in the Pum1^{-/-} mice during spermatogenesis (Chen et al. 2012). Pum1-dependent repression of Map3k1 is required to maintain P53 inactive. Since Pum1 targets a large number of mRNAs in cell lines and in the testis it is most likely that Pum1 functions go well beyond P53 regulation. In model organism, Pumilios play a critical role in maintaining the germ cell fate and prevent germ cell differentiation. It may be possible that accelerated differentiation of spermatogenic cells in the Pum1 knockout model leads to unbalanced ratio/germ cell supporting cells in the testis, a condition known to trigger apoptosis and spermatogenic failure.

4.3 Deleted in Azoospermia Associated Protein 1 (DAZAP1)

DAZAP1 is a member of the heterogeneous nuclear ribonuclear protein family (hnRNP). This protein was initially identified in a study looking at DAZL interacting proteins (Tsui et al. 2000a). But despite this interaction, both proteins differ in their protein expression pattern and were described to be associated with different subsets of target mRNAs, indicating that both proteins can have different roles (Kurihara et al. 2004).

DAZAP1 is ubiquitously expressed but highly enriched in testes (Dai et al. 2001). Whereas it is predominantly found in the nucleus of somatic cells (Lin and Yen 2006), in testes, DAZAP1 shows a dynamic distribution. In germ cells DAZAP1 is found in the nucleus from mid-pachytene spermatocytes to round spermatids, and relocates to the cytoplasm in elongating spermatids (Vera et al. 2002).

As with other hnRNPs, DAZAP1 binds newly synthesized transcripts and, through processes of splicing, export and translation, controls expression of specific messages (Martinez-Contreras et al. 2007). Recent evidence further elucidated the mechanism employed by DAZAP1 in splicing control based on extracellular cues. Through interaction with hnRNPs governed by 3'UTR *cis*-elements of target transcripts, DAZAP1 promotes splicing of weak exons. Phosphorylation of the C-terminal proline rich domain of DAZAP1 by MEK/Erk pathway alters the ability of DAZAP1 to maintain protein interactions with splicing factors and induces cytoplasmic translocation of DAZAP1 (Choudhury et al. 2014).

The molecular mechanisms of translational modulation of DAZAP1 are still not fully understood. It binds RNA at a consensus sequence AAAUAG and $GU_{[1-3]}AG$ (Yang et al. 2009), and the function is dependent on the protein partners to which it binds. Protein–protein interactions have been described with DAZL, KH-type splicing regulatory protein (KHSRP), hnRNPs, DEAD box polypeptide 20 (DDX20), let 7 and many other RBPs (Yang et al. 2009). The function as a translation activator has been studied in vitro models where DAZAP1 proteins were tethered to luciferase reporters. These show that both mouse and human DAZAP, as observed for *X. laevis*, stimulates translation in a cap-independent manner and it shows preferential translation towards nonadenylated messages (Smith et al. 2011).

Interestingly, DAZAP1 is itself a target for translation control. Although initially present at mid-pachytene spermatocytes, in situ analysis of DAZAP1 detects high levels of this transcript in spermatogonia and early spermatocytes (Vera et al. 2002). Through alternative usage of cleavage and polyadenylation sites, translation of each isoform is regulated by different mechanisms. At day 12 post-partum association with polyribosomes shows that both isoforms are actively translated. Upon reaching puberty the isoform with the longest 3'UTR relocates to the translational inactive mRNP fraction with a concomitant shortening of the poly(A) tail. The smaller isoform is also recruited to the mRNP fraction but this shows no changes in poly(A) length. The germ cell specific DAZL was shown to preferentially bind in vivo to the shorter isoform of DAZAP1, and in vitro assays also described a preferential DAZL promoted translation of a reporter with the 3'UTR of the shorter isoform (Yang and Yen 2013).

4.4 Gonadotropin-Regulated Testicular RNA Helicase (GRTH)

Ddx25 gained its common name, GRTH, from the fact that translation of this protein is dependent on hormonal cues. Three consensus Kozak sequences can be identified in frame with the coding region of GRTH. Whilst the second AUG is used by Leydig cells, germ cells preferentially use the first and third start codons. Although, treatment of rats with hCG, resulted in a shift towards the second AUG by round spermatids, but not spermatocytes (Sheng et al. 2003). This translational control is believed to result from induced paracrine factors that regulate the internal ribosomal entry site mechanisms.

GRTH promotes survival of spermatocytes by regulating apoptotic pathways. In *Grth*-null models, spermatogenesis arrests at step 8 spermatids due to increased apoptosis in spermatocytes at stage XII (Tsai-Morris et al. 2008). Despite not totally understood how GRTH regulates translation, an initial mechanism was directly linked to the function of RNA helicases that control RNA unwinding and nuclear transport (Tsai-Morris et al. 2010). An association with chromatoid bodies and polyribosomes further supported a role of this protein in translation (Tang et al. 1999; Sheng et al. 2006; Tsai-Morris et al. 2004b). Animal models provided the direct link to translation control by providing evidence that decrease levels of GRTH in germ cells resulted in decreased levels of specific proteins (TNPs and tACE) with no differences observed in mRNA levels (Tsai-Morris et al. 2004a). GRTH is present in the nucleus and cytoplasm, and is coupled with nuclear export, dependent on its phosphorylation status. Depletion of GRTH resulting in smaller chromatoid bodies and a deficit in the cytoplasmic ratios of proteins associated with nuclear histone-protamine (Tsai-Morris et al. 2004b; Sheng et al. 2006).

In humans, a mutation on GRTH is associated with non-obstructive azoospermia. This mutation results in a nonphosphorilated form of GRTH, and was proposed to impair RNA-binding and protein–protein interactions due to the location of the substituted residue in a hydrophobic pocket (Tsai-Morris et al. 2007).

5 Regulation of Translation in Mammalian Oocytes

The post-natal ovary is endowed with a pool of quiescent primordial follicles. Upon primordial follicle activation, the oocyte enters a growth phase that is characterized by intense transcriptional activity (Pan et al. 2005). However not all the transcribed mRNAs are immediately translated; a fraction is stored in a translationally-repressed form. As the oocyte approaches its fully grown size, the transcriptional activity ceases (Bouniol-Baly et al. 1999; Liu and Aoki 2002) and the final stages of differentiation and maturation, as well as fertilization and early embryo development occur in absence of transcription. Thus, the meiotic cell cycle progression and genome reprogramming rely on unmasking and translation of stored maternal mRNAs. This dependence continues until the embryo genome becomes activated, an event that occurs with a timing that is species specific (2-cell stage for major embryonic genome activation (EGA) in mice Latham 1999). As mentioned, this unique control of gene expression at the oocyte-to-embryo transition likely confers developmental plasticity by regulating transcript recruitment and translation in the cytoplasm, while the chromatin is free to undergo the reprogramming required for establishing totipotency and supporting embryogenesis.

Here we will present the most relevant mechanisms involved in the regulation of maternal mRNA translation, organizing the discussion into mechanisms controlling the activation, degradation and repression during oocyte development.

5.1 Mechanisms Controlling Repression of Translation and mRNA Degradation

Our current understanding of translational regulation during oocyte growth and maturation is mainly based on findings in the *X. laevis* oocyte model. In the frog system, it has been proposed that a combinatorial code of *cis*-acting elements in the 3'UTR of the messages regulates protein synthesis according to the temporal requirement of meiosis arrest and progression, as well as early embryo development.

One of the best characterized mechanisms of translation regulation in frog oocyte requires the *cis*-acting element CPE and the cognate binding proteins CPEB. By controlling the length of the cytoplasmic poly(A) tail, Cpeb1 represses translation in immature oocytes and, when phosphorylated upon cell cycle reentry, activates translation (McGrew et al. 1989). A mammalian isoform of CPEB, with 80 % identity with the frog Cpeb, was first described in mouse oocytes as a protein bound to the *c-Mos* mRNA (Gebauer and Richter 1996). Early studies of the tissue-type plasminogen activator (tPA) mRNA provided initial evidence that maternal mRNA repression and translation in mammals are regulated similarly to frogs (Huarte et al. 1992; Stutz et al. 1997). For instance, tPA is stored in a silenced form with a short poly(A) tail (30–40 nt) in GV stage oocytes. Upon meiotic re-entry, the poly(A) tail is elongated (to about 200 nt) and tPA becomes translated (Huarte et al. 1992). The role of a CPE *cis*-acting element was soon recognized as an important player in this

regulation. By inserting a CPE or deleting an existing one, the translation is respectively repressed or activated, together with shortening or lengthening of the poly(A) tail (Huarte et al. 1992; Gebauer and Richter 1996).

How CPEB functions in repressing poly(A) elongation and translation of maternal transcripts in mammals is still not fully understood. At least three different models have been proposed in frog oocytes, depending on the composition of the repressor complex, as reviewed in (Villalba et al. 2011). One possibility is that Cpeb simultaneously recruits the poly(A)-ribonuclease Parn and the poly(A)-polymerase Gld-2. The competition between the two enzymatic activities maintains a short poly(A) tail, hence promoting repression (Kim and Richter 2006). Alternatively a repressor protein, identified as Maskin, may bind to Cpeb and eIF4E, blocking the formation of the cap-binding complex on the mRNA, resulting in translational inhibition (Stebbins-Boaz et al. 1999). Similarly, a third model predicts that Cpeb bind to Clast 4 (also known as 4E-transporter (4E-T)), which in turn recruits isoforms of eIF4E, like eIF4E1b, that have low affinity for the cap structure (Minshall et al. 2007).

Rather than being mutually exclusive, these models may be integrated in the control of translational repression of different mRNA species or during different steps of oogenesis. Moreover extra layers of complexity are achieved through multiple regulatory *cis* and *trans*-acting factors. For instance additional regulation can be mediated by Pumilio binding elements (PBEs) that are frequently present in mRNAs that possess CPEs (Ota et al. 2011). As previously discussed in model organisms, Pumilio proteins (PUM1, PUM2) are mainly repressor proteins, but they can also stabilize CPEB binding and promote poly(A) tail elongation in some context (Padmanabhan and Richter 2006; Pique et al. 2008). However their role in mammalian oogenesis has not been elucidated yet. Interestingly, it has been reported that Pum1 protein in zebrafish oocytes is required to localize cyclin B1 to ribonuclear particles and this co-localization was confirmed in mouse oocytes (Kotani et al. 2013).

Another cis-acting element known to repress translation in immature oocytes is the translational control sequence (TCS). The corresponding trans-acting factor has been recently discovered in X. leavis and it is represented by zygote arrest 1 (Zar1) and Zar2. Zar1 and Zar2 bind to Mos and Weel mRNA 3'UTRs via a zinc finger. The TCS-mediated repression of translation is exerted during oocyte growth and initial phases of meiotic resumption until the metaphase I (MI) stage (Charlesworth et al. 2012; Yamamoto et al. 2013). It is not clear if the translational activation of Zar2 target mRNAs occurs following partial degradation of Zar2 and release of the repressive state, or if the control requires more than one *trans*-acting factor, as proposed for CPE. Based on the observation that mice embryos that are null for Zar1 arrest at the 1-cell stage due to failure in EGA, mammalian ZAR family proteins were initially proposed to be transcriptional regulators (Wu et al. 2003). However, a role for ZAR proteins in translational control in mammalian oocytes cannot be excluded. In this view, protein product(s) of ZAR-target mRNAs would be responsible of EGA, and the absence of ZAR would prevent their timely translation with subsequent failure in embryonic transcription and development.

In *X. laevis* oocytes CPE-induced translation is responsible for activating synthesis of RBPs, like the zinc finger protein C3H-4, that are required in the later stages of maturation for deadenylation of transcripts carrying A-U rich elements (AREs) (Belloc and Mendez 2008). In mammals, the embryonic lethal abnormal vision like 2 (ELAVL2) protein regulates ARE-mediated mRNA repression during oocyte growth (Chalupnikova et al. 2014). ELAVL2 decreased in abundance as the oocyte reaches a fully grown, meiotically competent stage of development, which is characterized by the transition from the non-surrounded nucleolus (NSN) chromatin configuration into the transcriptionally quiescent surrounded nucleolus (SN) configuration. ELAVL2 is absent in MII oocytes and zygotes. Mammalian proteins that function analogous to the frog C3H-4 protein to mediate ARE dependent deadenylation and repression during oocyte maturation have not been identified.

A regulatory feedback involving CPE-mediated translation to recruit *decapping mRNA1* and 2 (*Dcp1a* and *Dpc2*) has been recently described in mice (Ma et al. 2013). Decapping, i.e. the removal of the 5' 7-methyl-guanosine cap, exposes mRNAs to exonucleases and represents a critical step in the $5' \rightarrow 3'$ transcript degradation. Both the regulatory and catalytic subunits, DCP1A and DCP2, are synthesized between MI and MII and the inhibition of their accumulation interferes with the proper EGA execution. It is not clear at this point whether and how the DCP1A/DCP2-mediated degradation could be selective. One of the possibilities is that their involvement is the downstream event in a multi-step process that first requires mRNAs to be destabilized and tagged for degradation by sequence specific regulatory factors.

Destabilization can occur through deadenylation (Wu and Brewer 2012) and CDC2A-mediated phosphorylation of the Y box binding protein 2 (YBX2, also known as MSY2), an RBP required for oocyte mRNA stability (Medvedev et al. 2008, 2011). Furthermore, deadenylation does not necessarily target the mRNA for immediate degradation, but it can simply induce a translationally-repressed state. In support of this hypothesis in *X. laevis* oocytes deadenylation can be dissociated from decapping (Gillian-Daniel et al. 1998). In mice oocytes, a subset of messages that are not recruited to the polysome in metaphase II (MII) are degraded, whereas another subset exits the polysome fraction while remaining stable (Chen et al. 2011a). It would be interesting to investigate whether these transcripts are decapped, have a shorter poly(A) tail and/or a YBX2 motif.

Transcript profiling of oocytes and embryos represents one of the most informative approaches to monitor transcript stabilization or degradation. Using a system for oocyte transcript amplification not biased toward polyadenylated mRNA species and microarray approach, Su and co-Authors characterized the extent of transcript degradation during mice oocyte maturation by comparing fully grown—germinal vesicle (GV) stage oocytes and MII stage oocytes (Su et al. 2007). Selective degradation was found during oocyte maturation, targeting transcripts involved in oxidative phosphorylation, pyruvate and citrate metabolism and macromolecule synthesis, while transcripts involved in signal transduction were amongst the more stable. Degradation of the transcripts coding for ribosomal proteins was also a major component in the culling of maternal mRNAs. However, a limitation to this approach is that it cannot provide information on protein synthesis.

5.2 Mechanisms Controlling Activation of Translation during Oocyte Maturation

A survey of the genome-wide pattern of translation was conducted by profiling maternal mRNAs recruited to the polysomes after centrifugation on sucrose cushions or density gradients. The isolation of the polysome fraction allowed for a direct analysis of mRNAs undergoing translation at precise time points during oocyte maturation (Chen et al. 2011a). Using this polysome array approach, stereotypic pattern of maternal mRNA association with the polysome was identified during progression through the meiotic cell cycle. A similar approach has been applied to study the changes in translation at the oocyte to embryo transition, revealing that around 2000 transcripts are differentially translated between MII stage oocytes and 1-cell embryos (Potireddy et al. 2006). Maternal mRNAs fall into three distinct patterns of polysome recruitment during oocyte maturation: Class I, the majority, constitutively recruited to the polysome independently of the meiotic stage; Class II, approximately 20 % of the transcripts, associated to the polysome only at the GV stage and dissociating thereafter; Class III, another 20 %, recruited for translation during the transition to MII oocytes. In addition to cell cycle components, this last class of transcripts was mainly represented by transcription and chromatin remodeling related transcripts, indicating that the machinery used for nuclear reprogramming and transcriptional activation in the zygote is assembled earlier on. This finding explains the observation that somatic nuclear reprogramming is more efficiently supported by the cytoplasm of MII oocytes rather than GV oocytes, even though critical factors removed with the GV of the oocyte may be an additional possibility. Specific motifs for RBPs were also identified by bioinformatics analysis in the 3'UTRs of Class III transcripts, providing mechanistic insights on the regulation of translation at defined times during oocytes maturation (Chen et al. 2011a). In both these genome-wide studies the most abundant motif identified in the 3'UTRs of transcripts activated in MII stage oocytes was the CPE (Potireddy et al. 2006; Chen et al. 2011a).

As described above, CPE plays a major role in driving the activation of translation by promoting the elongation of the poly(A) tail. The CPE consensus sequence (A)UUUUA(A)U is usually located within 100 nt upstream the nuclear polyadenylation signal AAUAAA (Hex) (Pique et al. 2008) which binds the cleavage and polyadenylation specificity factor (CPSF) (Sheets et al. 1994). In *X. laevis* Cpeb is phosphorylated by the kinase Aurora A in response to progesterone (Mendez et al. 2000a). This post-translational modification increases Cpeb affinity for Cpsf (Mendez et al. 2000b) and Gld-2 (Barnard et al. 2004), while decreasing the affinity for Parn (Kim and Richter 2006). Other events required for Cpeb-promoted polyadenylation and activation of translation are the displacement of Maskin (Stebbins-Boaz et al. 1999) and/or 4E-T (Minshall et al. 2007), to allow the recruitment to the 40S ribosomal subunit. Also in mouse oocytes CPEB1 phosphorylation seems to be a crucial player in controlling maternal transcript translation upon meiosis resumption (Chen et al. 2011a) but the exact sequence of events has not been further elucidated and some differences with the frog should be pointed out: (1) *de novo* protein synthesis is essential in frog for meiosis re-entry, while in mice oocytes it is only required for the progression from MI to MII (Hashimoto and Kishimoto 1988); (2) polyadenylation and activation of translation occur at MI in mice, whereas they precede the GV breakdown (GVBD) in frog (McGrew et al. 1989); (3) the kinase responsible for CPEB phosphorylation has not been identified in mammals and is likely not Aurora A (Andresson and Ruderman 1998; Mendez et al. 2000a, b); (4) the homologue of frog Maskin, transforming acidic coiled coil containing protein (TACC3), does not have a domain for interaction with CPEB (Barnard et al. 2005). Given these possible species differences, further studies are needed to elucidate CPEB function in mouse oocytes maturation.

In *Cpeb* null mice, the development of normal adult gametes and gonads are precluded, likely due to a defect in the translation of synaptonemal complex protein mRNAs that prevents the transition from the Pachytene to the Diplotene stage (Tay and Richter 2001; Tay et al. 2003). Consequently loss of function models cannot be used to elucidate the role of CPEB during oocyte maturation. Loss of function experiments have been conducted by morpholino oligonucleotides injection, showing that CPEB1 depletion, but not CPEB3 and 4, decreases the efficiency of progression through meiosis (Chen et al. 2011a). However, the extent of depletion of these proteins from the oocyte after morpholino microinjection could not be determined.

After phosphorylation, CPEB1 is targeted for degradation. The degradation of CPEB1 at the time of metaphase I (MI) and the identification of several other RNA motifs that correlate with recruitment to polysomes indicates that CPEB independent mechanisms also contribute to generate temporal pattern of translation throughout maturation. In frog oocytes, it has been proposed that Cpeb1 induces the translation of Cpeb4, a related RBP but with slightly different properties (Igea and Mendez 2010). The accumulation of this RBP which binds to a CPE element stabilizes CPE-dependent polyadenylation and translation. It is most likely that a similar mechanism operates in mammalian oocytes. As reported above, a first attempt to define the role of CPEB3 and CPEB4 using morpholino oligonucleotide knockdown and extrusion of the polar body as readout produced minimal disruption. However, other RBPs may be involved in this kind of sequential regulation. One of the motifs identified in the 3'UTRs of transcripts actively translated during oocyte maturation is represented by the consensus sequences for the DAZ family of proteins (Daz, Dazl, and Boule). Through its CPE, Dazl transcript is recruited to polysomes and DAZL protein increases fivefold, reaching a maximum in MII. In addition to the CPE, Dazl also contains DAZL consensus sequences. The newly synthesized DAZL therefore increases translation of its own mRNA, establishing a self-reinforcing, positive feedback loop (Chen et al. 2011a). It is most likely that DAZL promotes the translation of several transcripts coding for cell cycle



Fig. 7.2 Proposed model for the regulation of translation during mouse oocyte maturation. (a) Class III mRNAs including *Dazl* mRNA are represented in a repressed translational state at Prophase I of meiosis (GV stage) with CPEB bound to them and a short poly(A) tail. Upon progression through Prometaphase I and CPEB1 phosphorylation, Class III mRNAs, including *Dazl mRNA*, undergo poly(A) elongation and become actively translated. The increase in DAZL protein promotes a positive feedback auto-regulatory loop which amplifies *Dazl* mRNA translation. This positive feedback plays a role in prolonging an activated state of translation may be complementary to other similar regulations, as a CPEB1-dependent translation of CPEB4 in the late stages of oocyte maturation has been reported in frog oocytes (Igea and Mendez 2010). (b) CPEB1 phosphorylation triggers DAZL translation and installs the positive feedback loop that promotes translation of DAZL targets (adapted from Chen et al. 2011)

regulators and chromatin remodelers and its depletion causes disruption of the spindle assembly (Fig. 7.2).

Because Pumilio Binding Elements have been identified between the motifs enriched in transcripts recruited to the polysomes in MII (Chen et al. 2011a), the RBPs Pumilio 1 and 2 may participate to translational activation, possibly by stabilizing the CPEB-mediated polyadenylation, as described in *X. laevis* oocytes (Padmanabhan and Richter 2006).

Early polyadenylation events and activation of translation have been described in *X. laevis* oocytes in a CPE-independent manner. This is the case for instance of the *Mos* mRNA, which is controlled by binding of Musashi to the polyadenylation response element (PRE) on the 3'UTR (Charlesworth et al. 2002). Synthesis of Mos is necessary to induce the activation of mitogen-activated protein kinase (MAPK) that occurs upstream (and is needed for) CPE-directed translation (Charlesworth et al. 2006). The Musashi-CPEB progressive activation is another example of how

the temporal control is exerted on translation to secure the proper progression through meiosis. Musashi is expressed in mammalian cells and, while it is usually associated with translational repression, it acts as an activator of translation in some conditions (MacNicol et al. 2011). Whether Musashi may have a similar function in mammalian oocyte is an open question.

Cell cycle progression and genome reprogramming strictly rely on the coordinated translation of stored maternal mRNAs at the oocyte to zygote transition. Consequently the proper execution of the translational program is a key factor for a successful embryo development. Rather than being completely cell autonomous, meiotic progression and translational activation are modulated by environmental inputs acting through the activation of the follicular EGF network (Chen et al. 2013). These somatic signals activate the PI(3)K-AKT-mTOR pathway in the oocyte soon after GVBD and promote translation of a subset of maternal mRNAs critical for the onset of the oocyte developmental competence (Fig. 7.3).

In this chapter, we have explored the extensive post-transcriptional regulatory factors and mechanisms that control development of the germline, integration of somatic signals, differentiation of gametes, and the earliest stages of embryonic development. Clearly there is a great deal yet to be learned, especially in mammalian reproductive systems.



of the signaling pathway involved in somatic cell control of translation in the oocyte. The scheme represents an oocyte surrounded by its cumulus cells. Amphiregulin and other EGF-like growth factors, released in response to the LH surge, activate EGF receptors expressed on cumulus cells. Through the activation on undefined signaling pathway(s) across the cumulus and oocyte plasma membranes, PI(3)K is activated in the oocyte. The resulting phosphorylation of AKT and mTOR activation leads to an increase in translation of a subset of maternal mRNAs

Acknowledgments The work from the authors' laboratory was supported by Eunice Kennedy Shriver National Institute of Child Health and Human Development/NIH cooperative agreement P50HD055764-06, as part of the Specialized Cooperative Centers Program in Reproduction and Infertility Research and RO1-GM097165 to MC. FF is supported by FP7-PEOPLE-2013-IOF GA 624874 MateRNA. SJH is supported by a grant from Research year of Inje University in 20140022.

References

- Andresson T, Ruderman JV (1998) The kinase Eg2 is a component of the Xenopus oocyte progesterone-activated signaling pathway. EMBO J 17(19):5627–5637
- Bally-Cuif L, Schatz WJ, Ho RK (1998) Characterization of the zebrafish Orb/CPEB-related RNA binding protein and localization of maternal components in the zebrafish oocyte. Mech Dev 77(1):31–47
- Barnard DC, Ryan K, Manley JL, Richter JD (2004) Symplekin and xGLD-2 are required for CPEB-mediated cytoplasmic polyadenylation. Cell 119(5):641–651
- Barnard DC, Cao Q, Richter JD (2005) Differential phosphorylation controls Maskin association with eukaryotic translation initiation factor 4E and localization on the mitotic apparatus. Mol Cell Biol 25(17):7605–7615
- Barrios F, Filipponi D, Pellegrini M, Paronetto MP, Di Siena S, Geremia R, Rossi P, De Felici M, Jannini EA, Dolci S (2010) Opposing effects of retinoic acid and FGF9 on Nanos2 expression and meiotic entry of mouse germ cells. J Cell Sci 123(Pt 6):871–880
- Barton MK, Kimble J (1990) fog-1, a regulatory gene required for specification of spermatogenesis in the germ line of Caenorhabditis elegans. Genetics 125(1):29–39
- Belloc E, Mendez R (2008) A deadenylation negative feedback mechanism governs meiotic metaphase arrest. Nature 452(7190):1017–1021
- Bettegowda A, Wilkinson MF (2010) Transcription and post-transcriptional regulation of spermatogenesis. Philos Trans R Soc Lond B Biol Sci 365(1546):1637–1651
- Bielawski JP, Yang Z (2001) Positive and negative selection in the DAZ gene family. Mol Biol Evol 18(4):523–529
- Bouniol-Baly C, Hamraoui L, Guibert J, Beaujean N, Szollosi MS, Debey P (1999) Differential transcriptional activity associated with chromatin configuration in fully grown mouse germinal vesicle oocytes. Biol Reprod 60(3):580–587
- Bowles J, Knight D, Smith C, Wilhelm D, Richman J, Mamiya S, Yashiro K, Chawengsaksophak K, Wilson MJ, Rossant J, Hamada H, Koopman P (2006) Retinoid signaling determines germ cell fate in mice. Science 312(5773):596–600
- Braun RE (2001) Packaging paternal chromosomes with protamine. Nat Genet 28(1):10–12
- Braun RE, Peschon JJ, Behringer RR, Brinster RL, Palmiter RD (1989) Protamine 3'-untranslated sequences regulate temporal translational control and subcellular localization of growth hormone in spermatids of transgenic mice. Genes Dev 3(6):793–802
- Brook M, Smith JW, Gray NK (2009) The DAZL and PABP families: RNA-binding proteins with interrelated roles in translational control in oocytes. Reproduction 137(4):595–617
- Brownlie A, Hersey C, Oates AC, Paw BH, Falick AM, Witkowska HE, Flint J, Higgs D, Jessen J, Bahary N, Zhu H, Lin S, Zon L (2003) Characterization of embryonic globin genes of the zebrafish. Dev Biol 255(1):48–61
- Campbell ZT, Menichelli E, Friend K, Wu J, Kimble J, Williamson JR, Wickens M (2012) Identification of a conserved interface between PUF and CPEB proteins. J Biol Chem 287(22):18854–18862
- Chalupnikova K, Solc P, Sulimenko V, Sedlacek R, Svoboda P (2014) An oocyte-specific ELAVL2 isoform is a translational repressor ablated from meiotically competent antral oocytes. Cell Cycle 13(7):1187–1200
- Charlesworth A, Ridge JA, King LA, MacNicol MC, MacNicol AM (2002) A novel regulatory element determines the timing of Mos mRNA translation during Xenopus oocyte maturation. EMBO J 21(11):2798–2806

- Charlesworth A, Wilczynska A, Thampi P, Cox LL, MacNicol AM (2006) Musashi regulates the temporal order of mRNA translation during Xenopus oocyte maturation. EMBO J 25(12):2792–2801
- Charlesworth A, Yamamoto TM, Cook JM, Silva KD, Kotter CV, Carter GS, Holt JW, Lavender HF, MacNicol AM, Ying Wang Y, Wilczynska A (2012) Xenopus laevis zygote arrest 2 (zar2) encodes a zinc finger RNA-binding protein that binds to the translational control sequence in the maternal Wee1 mRNA and regulates translation. Dev Biol 369(2):177–190
- Charlesworth A, Meijer HA, de Moor CH (2013) Specificity factors in cytoplasmic polyadenylation. Wiley Interdiscip Rev RNA 4(4):437–461
- Chen P, Ma M, Li L, Zhang S, Su D, Ma Y, Liu Y, Tao D, Lin L, Yang Y (2010) Phenotypic expression of partial AZFc deletions is independent of the variations in DAZL and BOULE in a Han population. J Androl 31(2):163–168
- Chen J, Melton C, Suh N, Oh JS, Horner K, Xie F, Sette C, Blelloch R, Conti M (2011a) Genomewide analysis of translation reveals a critical role for deleted in azoospermia-like (Dazl) at the oocyte-to-zygote transition. Genes Dev 25(7):755–766
- Chen S, Wang S, Xie T (2011b) Restricting self-renewal signals within the stem cell niche: multiple levels of control. Curr Opin Genet Dev 21(6):684–689
- Chen D, Zheng W, Lin A, Uyhazi K, Zhao H, Lin H (2012) Pumilio 1 suppresses multiple activators of p53 to safeguard spermatogenesis. Curr Biol 22(5):420–425
- Chen J, Torcia S, Xie F, Lin CJ, Cakmak H, Franciosi F, Horner K, Onodera C, Song JS, Cedars MI, Ramalho-Santos M, Conti M (2013) Somatic cells regulate maternal mRNA translation and developmental competence of mouse oocytes. Nat Cell Biol 15(12):1415–1423
- Cheung LS, Schupbach T, Shvartsman SY (2011) Pattern formation by receptor tyrosine kinases: analysis of the Gurken gradient in Drosophila oogenesis. Curr Opin Genet Dev 21(6): 719–725
- Choudhury R, Roy SG, Tsai YS, Tripathy A, Graves LM, Wang Z (2014) The splicing activator DAZAP1 integrates splicing control into MEK/Erk-regulated cell proliferation and migration. Nat Commun 5:3078
- Christerson LB, McKearin DM (1994) orb is required for anteroposterior and dorsoventral patterning during Drosophila oogenesis. Genes Dev 8(5):614–628
- Clarke HJ (2012) Post-transcriptional control of gene expression during mouse oogenesis. Results Probl Cell Differ 55:1–21
- Collier B, Gorgoni B, Loveridge C, Cooke HJ, Gray NK (2005) The DAZL family proteins are PABP-binding proteins that regulate translation in germ cells. EMBO J 24(14):2656–2666
- Conti M, Hsieh M, Park JY, Su YQ (2006) Role of the epidermal growth factor network in ovarian follicles. Mol Endocrinol 20(4):715–723
- Crittenden SL, Bernstein DS, Bachorik JL, Thompson BE, Gallegos M, Petcherski AG, Moulder G, Barstead R, Wickens M, Kimble J (2002) A conserved RNA-binding protein controls germline stem cells in Caenorhabditis elegans. Nature 417(6889):660–663
- Dai T, Vera Y, Salido EC, Yen PH (2001) Characterization of the mouse Dazap1 gene encoding an RNA-binding protein that interacts with infertility factors DAZ and DAZL. BMC Genomics 2:6
- Fernandez-Miranda G, Mendez R (2012) The CPEB-family of proteins, translational control in senescence and cancer. Ageing Res Rev 11(4):460–472
- Friend K, Campbell ZT, Cooke A, Kroll-Conner P, Wickens MP, Kimble J (2012) A conserved PUF-Ago-eEF1A complex attenuates translation elongation. Nat Struct Mol Biol 19(2): 176–183
- Fuller MT (1998) Genetic control of cell proliferation and differentiation in Drosophila spermatogenesis. Semin Cell Dev Biol 9(4):433–444
- Garcia TX, Farmaha JK, Kow S, Hofmann MC (2014) RBPJ in mouse Sertoli cells is required for proper regulation of the testis stem cell niche. Development 141(23):4468–4478
- Gebauer F, Richter JD (1996) Mouse cytoplasmic polyadenylylation element binding protein: an evolutionarily conserved protein that interacts with the cytoplasmic polyadenylylation elements of c-mos mRNA. Proc Natl Acad Sci U S A 93(25):14602–14607

- Ghosh S, Marchand V, Gaspar I, Ephrussi A (2012) Control of RNP motility and localization by a splicing-dependent structure in oskar mRNA. Nat Struct Mol Biol 19(4):441–449
- Gill ME, Hu YC, Lin Y, Page DC (2011) Licensing of gametogenesis, dependent on RNA binding protein DAZL, as a gateway to sexual differentiation of fetal germ cells. Proc Natl Acad Sci U S A 108(18):7443–7448
- Gillian-Daniel DL, Gray NK, Astrom J, Barkoff A, Wickens M (1998) Modifications of the 5' cap of mRNAs during Xenopus oocyte maturation: independence from changes in poly(A) length and impact on translation. Mol Cell Biol 18(10):6152–6163
- Goldstrohm AC, Hook BA, Seay DJ, Wickens M (2006) PUF proteins bind Pop2p to regulate messenger RNAs. Nat Struct Mol Biol 13(6):533–539
- Gonczy P, Matunis E, DiNardo S (1997) Bag-of-marbles and benign gonial cell neoplasm act in the germline to restrict proliferation during Drosophila spermatogenesis. Development 124(21):4361–4371
- Gunter KM, McLaughlin EA (2011) Translational control in germ cell development: a role for the RNA-binding proteins Musashi-1 and Musashi-2. IUBMB Life 63(9):678–685
- Hashimoto N, Kishimoto T (1988) Regulation of meiotic metaphase by a cytoplasmic maturationpromoting factor during mouse oocyte maturation. Dev Biol 126(2):242–252
- Hermo L, Pelletier RM, Cyr DG, Smith CE (2009) Surfing the wave, cycle, life history, and genes/ proteins expressed by testicular germ cells. Part 1: background to spermatogenesis, spermatogonia, and spermatocytes. Microsc Res Tech 73(4):241–278
- Huarte J, Stutz A, O'Connell ML, Gubler P, Belin D, Darrow AL, Strickland S, Vassalli JD (1992) Transient translational silencing by reversible mRNA deadenylation. Cell 69(6):1021–1030
- Idler RK, Yan W (2012) Control of messenger RNA fate by RNA-binding proteins: an emphasis on mammalian spermatogenesis. J Androl 33(3):309–337
- Igea A, Mendez R (2010) Meiosis requires a translational positive loop where CPEB1 ensues its replacement by CPEB4. EMBO J 29(13):2182–2193
- Ivshina M, Lasko P, Richter JD (2014) Cytoplasmic polyadenylation element binding proteins in development, health, and disease. Annu Rev Cell Dev Biol 30:393–415
- Jenkins HT, Malkova B, Edwards TA (2011) Kinked beta-strands mediate high-affinity recognition of mRNA targets by the germ-cell regulator DAZL. Proc Natl Acad Sci U S A 108(45):18266–18271
- Jiao X, Trifillis P, Kiledjian M (2002) Identification of target messenger RNA substrates for the murine deleted in azoospermia-like RNA-binding protein. Biol Reprod 66(2):475–485
- Karashima T, Sugimoto A, Yamamoto M (2000) Caenorhabditis elegans homologue of the human azoospermia factor DAZ is required for oogenesis but not for spermatogenesis. Development 127(5):1069–1079
- Kim JH, Richter JD (2006) Opposing polymerase-deadenylase activities regulate cytoplasmic polyadenylation. Mol Cell 24(2):173–183
- Kim JH, Richter JD (2007) RINGO/cdk1 and CPEB mediate poly(A) tail stabilization and translational regulation by ePAB. Genes Dev 21(20):2571–2579
- Kim JY, Lee YC, Kim C (2010) Direct inhibition of Pumilio activity by Bam and Bgcn in Drosophila germ line stem cell differentiation. J Biol Chem 285(7):4741–4746
- Kim B, Cooke HJ, Rhee K (2012) DAZL is essential for stress granule formation implicated in germ cell survival upon heat stress. Development 139(3):568–578
- Kimble J (2011) Molecular regulation of the mitosis/meiosis decision in multicellular organisms. Cold Spring Harb Perspect Biol 3(8):a002683
- Kimble J, Crittenden SL (2007) Controls of germline stem cells, entry into meiosis, and the sperm/ oocyte decision in Caenorhabditis elegans. Annu Rev Cell Dev Biol 23:405–433
- Kimble J, Simpson P (1997) The LIN-12/Notch signaling pathway and its regulation. Annu Rev Cell Dev Biol 13:333–361
- Kotani T, Yasuda K, Ota R, Yamashita M (2013) Cyclin B1 mRNA translation is temporally controlled through formation and disassembly of RNA granules. J Cell Biol 202(7):1041–1055
- Kraemer B, Crittenden S, Gallegos M, Moulder G, Barstead R, Kimble J, Wickens M (1999) NANOS-3 and FBF proteins physically interact to control the sperm-oocyte switch in Caenorhabditis elegans. Curr Biol 9(18):1009–1018

- Kugler JM, Lasko P (2009) Localization, anchoring and translational control of oskar, gurken, bicoid and nanos mRNA during Drosophila oogenesis. Fly 3(1):15–28
- Kurihara Y, Watanabe H, Kawaguchi A, Hori T, Mishiro K, Ono M, Sawada H, Uesugi S (2004) Dynamic changes in intranuclear and subcellular localizations of mouse Prrp/DAZAP1 during spermatogenesis: the necessity of the C-terminal proline-rich region for nuclear import and localization. Arch Histol Cytol 67(4):325–333
- Lantz V, Chang JS, Horabin JI, Bopp D, Schedl P (1994) The Drosophila orb RNA-binding protein is required for the formation of the egg chamber and establishment of polarity. Genes Dev 8(5):598–613
- Latham KE (1999) Mechanisms and control of embryonic genome activation in mammalian embryos. Int Rev Cytol 193:71–124
- Lee KH, Lee S, Kim B, Chang S, Kim SW, Paick JS, Rhee K (2006) Dazl can bind to dynein motor complex and may play a role in transport of specific mRNAs. EMBO J 25(18):4263–4270
- Lee MT, Bonneau AR, Giraldez AJ (2014) Zygotic genome activation during the maternal-tozygotic transition. Annu Rev Cell Dev Biol 30:581–613
- Li L, Zheng P, Dean J (2010) Maternal control of early mouse development. Development 137(6):859–870
- Li Y, Minor NT, Park JK, McKearin DM, Maines JZ (2009) Bam and Bgcn antagonize Nanosdependent germ-line cell maintenance. Proc Natl Acad Sci U S A 106(23):9304–9309
- Li Y, Zhang Q, Carreira-Rosario A, Maines JZ, McKearin DM, Buszczak M (2013) Mei-p26 cooperates with Bam, Bgcn and Sxl to promote early germline development in the Drosophila ovary. PLoS One 8(3), e58301
- Lin Y, Page DC (2005) Dazl deficiency leads to embryonic arrest of germ cell development in XY C57BL/6 mice. Dev Biol 288(2):309–316
- Lin H, Spradling AC (1993) Germline stem cell division and egg chamber development in transplanted Drosophila germaria. Dev Biol 159(1):140–152
- Lin YT, Yen PH (2006) A novel nucleocytoplasmic shuttling sequence of DAZAP1, a testisabundant RNA-binding protein. RNA 12(8):1486–1493
- Lin Q, Taylor SJ, Shalloway D (1997) Specificity and determinants of Sam68 RNA binding. Implications for the biological function of K homology domains. J Biol Chem 272(43):27274–27280
- Liu H, Aoki F (2002) Transcriptional activity associated with meiotic competence in fully grown mouse GV oocytes. Zygote 10(4):327–332
- Lolicato F, Marino R, Paronetto MP, Pellegrini M, Dolci S, Geremia R, Grimaldi P (2008) Potential role of Nanos3 in maintaining the undifferentiated spermatogonia population. Dev Biol 313(2):725–738
- Luitjens C, Gallegos M, Kraemer B, Kimble J, Wickens M (2000) CPEB proteins control two key steps in spermatogenesis in C. elegans. Genes Dev 14(20):2596–2609
- Ma J, Flemr M, Strnad H, Svoboda P, Schultz RM (2013) Maternally recruited DCP1A and DCP2 contribute to messenger RNA degradation during oocyte maturation and genome activation in mouse. Biol Reprod 88(1):11
- MacNicol MC, Cragle CE, MacNicol AM (2011) Context-dependent regulation of Musashimediated mRNA translation and cell cycle regulation. Cell Cycle 10(1):39–44
- Marin VA, Evans TC (2003) Translational repression of a C. elegans Notch mRNA by the STAR/ KH domain protein GLD-1. Development 130(12):2623–2632
- Martinez-Contreras R, Cloutier P, Shkreta L, Fisette JF, Revil T, Chabot B (2007) hnRNP proteins and splicing control. Adv Exp Med Biol 623:123–147
- Maruyama R, Endo S, Sugimoto A, Yamamoto M (2005) Caenorhabditis elegans DAZ-1 is expressed in proliferating germ cells and directs proper nuclear organization and cytoplasmic core formation during oogenesis. Dev Biol 277(1):142–154
- McGrew LL, Dworkin-Rastl E, Dworkin MB, Richter JD (1989) Poly(A) elongation during Xenopus oocyte maturation is required for translational recruitment and is mediated by a short sequence element. Genes Dev 3(6):803–815

- Medvedev S, Yang J, Hecht NB, Schultz RM (2008) CDC2A (CDK1)-mediated phosphorylation of MSY2 triggers maternal mRNA degradation during mouse oocyte maturation. Dev Biol 321(1):205–215
- Medvedev S, Pan H, Schultz RM (2011) Absence of MSY2 in mouse oocytes perturbs oocyte growth and maturation, RNA stability, and the transcriptome. Biol Reprod 85(3):575–583
- Mendez R, Hake LE, Andresson T, Littlepage LE, Ruderman JV, Richter JD (2000a) Phosphorylation of CPE binding factor by Eg2 regulates translation of c-mos mRNA. Nature 404(6775):302–307
- Mendez R, Murthy KG, Ryan K, Manley JL, Richter JD (2000b) Phosphorylation of CPEB by Eg2 mediates the recruitment of CPSF into an active cytoplasmic polyadenylation complex. Mol Cell 6(5):1253–1259
- Miller MA, Olivas WM (2011) Roles of Puf proteins in mRNA degradation and translation. Wiley Interdiscip Rev RNA 2(4):471–492
- Minshall N, Reiter MH, Weil D, Standart N (2007) CPEB interacts with an ovary-specific eIF4E and 4E-T in early Xenopus oocytes. J Biol Chem 282(52):37389–37401
- Najib S, Martin-Romero C, Gonzalez-Yanes C, Sanchez-Margalet V (2005) Role of Sam68 as an adaptor protein in signal transduction. Cell Mol Life Sci 62(1):36–43
- O'Connell ML, Cavallo WC Jr, Firnberg M (2014) The expression of CPEB proteins is sequentially regulated during zebrafish oogenesis and embryogenesis. Mol Reprod Dev 81(4):376–387
- Oatley JM, Brinster RL (2008) Regulation of spermatogonial stem cell self-renewal in mammals. Annu Rev Cell Dev Biol 24:263–286
- Ota R, Kotani T, Yamashita M (2011) Biochemical characterization of Pumilio1 and Pumilio2 in Xenopus oocytes. J Biol Chem 286(4):2853–2863
- Padmanabhan K, Richter JD (2006) Regulated Pumilio-2 binding controls RINGO/Spy mRNA translation and CPEB activation. Genes Dev 20(2):199–209
- Pan H, O'Brien MJ, Wigglesworth K, Eppig JJ, Schultz RM (2005) Transcript profiling during mouse oocyte development and the effect of gonadotropin priming and development in vitro. Dev Biol 286(2):493–506
- Paronetto MP, Sette C (2010) Role of RNA-binding proteins in mammalian spermatogenesis. Int J Androl 33(1):2–12
- Paronetto MP, Messina V, Bianchi E, Barchi M, Vogel G, Moretti C, Palombi F, Stefanini M, Geremia R, Richard S, Sette C (2009) Sam68 regulates translation of target mRNAs in male germ cells, necessary for mouse spermatogenesis. J Cell Biol 185(2):235–249
- Pfender S, Kuznetsov V, Pasternak M, Tischer T, Santhanam B, Schuh M (2015) Live imaging RNAi screen reveals genes essential for meiosis in mammalian oocytes. Nature 524(7564):239–242
- Pique M, Lopez JM, Foissac S, Guigo R, Mendez R (2008) A combinatorial code for CPEmediated translational control. Cell 132(3):434–448
- Potireddy S, Vassena R, Patel BG, Latham KE (2006) Analysis of polysomal mRNA populations of mouse oocytes and zygotes: dynamic changes in maternal mRNA utilization and function. Dev Biol 298(1):155–166
- Quenault T, Lithgow T, Traven A (2011) PUF proteins: repression, activation and mRNA localization. Trends Cell Biol 21(2):104–112
- Reijo R, Lee TY, Salo P, Alagappan R, Brown LG, Rosenberg M, Rozen S, Jaffe T, Straus D, Hovatta O et al (1995) Diverse spermatogenic defects in humans caused by Y chromosome deletions encompassing a novel RNA-binding protein gene. Nat Genet 10(4):383–393
- Reynolds N, Collier B, Maratou K, Bingham V, Speed RM, Taggart M, Semple CA, Gray NK, Cooke HJ (2005) Dazl binds in vivo to specific transcripts and can regulate the pre-meiotic translation of Mvh in germ cells. Hum Mol Genet 14(24):3899–3909
- Reynolds N, Collier B, Bingham V, Gray NK, Cooke HJ (2007) Translation of the synaptonemal complex component Sycp3 is enhanced in vivo by the germ cell specific regulator Dazl. RNA 13(7):974–981
- Richter JD (2007) CPEB: a life in translation. Trends Biochem Sci 32(6):279-285

- Ruggiu M, Cooke HJ (2000) In vivo and in vitro analysis of homodimerisation activity of the mouse Dazl1 protein. Gene 252(1–2):119–126
- Ruggiu M, Speed R, Taggart M, McKay SJ, Kilanowski F, Saunders P, Dorin J, Cooke HJ (1997) The mouse Dazla gene encodes a cytoplasmic protein essential for gametogenesis. Nature 389(6646):73–77
- Saga Y (2010) Function of Nanos2 in the male germ cell lineage in mice. Cell Mol Life Sci 67(22):3815–3822
- Saunders PT, Turner JM, Ruggiu M, Taggart M, Burgoyne PS, Elliott D, Cooke HJ (2003) Absence of mDazl produces a final block on germ cell development at meiosis. Reproduction 126(5):589–597
- Sheets MD, Fox CA, Hunt T, Vande Woude G, Wickens M (1994) The 3'-untranslated regions of c-mos and cyclin mRNAs stimulate translation by regulating cytoplasmic polyadenylation. Genes Dev 8(8):926–938
- Shen R, Xie T (2009) NANOS: a germline stem cell's Guardian Angel. J Mol Cell Biol 2(2):76–77
- Sheng Y, Tsai-Morris CH, Dufau ML (2003) Cell-specific and hormone-regulated expression of gonadotropin-regulated testicular RNA helicase gene (GRTH/Ddx25) resulting from alternative utilization of translation initiation codons in the rat testis. J Biol Chem 278(30):27796–27803
- Sheng Y, Tsai-Morris CH, Gutti R, Maeda Y, Dufau ML (2006) Gonadotropin-regulated testicular RNA helicase (GRTH/Ddx25) is a transport protein involved in gene-specific mRNA export and protein translation during spermatogenesis. J Biol Chem 281(46):35048–35056
- Slaidina M, Lehmann R (2014) Translational control in germline stem cell development. J Cell Biol 207(1):13–21
- Smith RW, Anderson RC, Smith JW, Brook M, Richardson WA, Gray NK (2011) DAZAP1, an RNA-binding protein required for development and spermatogenesis, can regulate mRNA translation. RNA 17(7):1282–1295
- Stebbins-Boaz B, Cao Q, de Moor CH, Mendez R, Richter JD (1999) Maskin is a CPEB-associated factor that transiently interacts with eIF-4E. Mol Cell 4(6):1017–1027
- Stutz A, Huarte J, Gubler P, Conne B, Belin D, Vassalli JD (1997) In vivo antisense oligodeoxynucleotide mapping reveals masked regulatory elements in an mRNA dormant in mouse oocytes. Mol Cell Biol 17(4):1759–1767
- Su YQ, Sugiura K, Woo Y, Wigglesworth K, Kamdar S, Affourtit J, Eppig JJ (2007) Selective degradation of transcripts during meiotic maturation of mouse oocytes. Dev Biol 302(1):104–117
- Takahashi K, Kotani T, Katsu Y, Yamashita M (2014) Possible involvement of insulin-like growth factor 2 mRNA-binding protein 3 in zebrafish oocyte maturation as a novel cyclin B1 mRNA-binding protein that represses the translation in immature oocytes. Biochem Biophys Res Commun 448(1):22–27
- Takeda Y, Mishima Y, Fujiwara T, Sakamoto H, Inoue K (2009) DAZL relieves miRNA-mediated repression of germline mRNAs by controlling poly(A) tail length in zebrafish. PLoS One 4(10), e7513
- Tang PZ, Tsai-Morris CH, Dufau ML (1999) A novel gonadotropin-regulated testicular RNA helicase. A new member of the dead-box family. J Biol Chem 274(53):37932–37940
- Tang H, Ross A, Capel B (2008) Expression and functional analysis of Gm114, a putative mammalian ortholog of Drosophila bam. Dev Biol 318(1):73–81
- Tay J, Richter JD (2001) Germ cell differentiation and synaptonemal complex formation are disrupted in CPEB knockout mice. Dev Cell 1(2):201–213
- Tay J, Hodgman R, Sarkissian M, Richter JD (2003) Regulated CPEB phosphorylation during meiotic progression suggests a mechanism for temporal control of maternal mRNA translation. Genes Dev 17(12):1457–1462
- Tsai-Morris CH, Lei S, Jiang Q, Sheng Y, Dufau ML (2004a) Genomic organization and transcriptional analysis of gonadotropin-regulated testicular RNA helicase—GRTH/DDX25 gene. Gene 331:83–94

- Tsai-Morris CH, Sheng Y, Lee E, Lei KJ, Dufau ML (2004b) Gonadotropin-regulated testicular RNA helicase (GRTH/Ddx25) is essential for spermatid development and completion of spermatogenesis. Proc Natl Acad Sci U S A 101(17):6373–6378
- Tsai-Morris CH, Koh E, Sheng Y, Maeda Y, Gutti R, Namiki M, Dufau ML (2007) Polymorphism of the GRTH/DDX25 gene in normal and infertile Japanese men: a missense mutation associated with loss of GRTH phosphorylation. Mol Hum Reprod 13(12):887–892
- Tsai-Morris CH, Koh E, Dufau ML (2008) Differences in gonadotropin-regulated testicular helicase (GRTH/DDX25) single nucleotide polymorphism between Japanese and Chinese populations. Hum Reprod 23(11):2611–2613
- Tsai-Morris CH, Sheng Y, Gutti R, Li J, Pickel J, Dufau ML (2010) Gonadotropin-regulated testicular RNA helicase (GRTH/DDX25) gene: cell-specific expression and transcriptional regulation by androgen in transgenic mouse testis. J Cell Biochem 109(6):1142–1147
- Tsui S, Dai T, Roettger S, Schempp W, Salido EC, Yen PH (2000a) Identification of two novel proteins that interact with germ-cell-specific RNA-binding proteins DAZ and DAZL1. Genomics 65(3):266–273
- Tsui S, Dai T, Warren ST, Salido EC, Yen PH (2000b) Association of the mouse infertility factor DAZL1 with actively translating polyribosomes. Biol Reprod 62(6):1655–1660
- Urano J, Fox MS, Reijo Pera RA (2005) Interaction of the conserved meiotic regulators, BOULE (BOL) and PUMILIO-2 (PUM2). Mol Reprod Dev 71(3):290–298
- VanGompel MJ, Xu EY (2010) A novel requirement in mammalian spermatid differentiation for the DAZ-family protein Boule. Hum Mol Genet 19(12):2360–2369
- Vanorny DA, Prasasya RD, Chalpe AJ, Kilen SM, Mayo KE (2014) Notch signaling regulates ovarian follicle formation and coordinates follicular growth. Mol Endocrinol 28(4):499–511
- Venables JP, Ruggiu M, Cooke HJ (2001) The RNA-binding specificity of the mouse Dazl protein. Nucleic Acids Res 29(12):2479–2483
- Vera Y, Dai T, Hikim AP, Lue Y, Salido EC, Swerdloff RS, Yen PH (2002) Deleted in azoospermia associated protein 1 shuttles between nucleus and cytoplasm during normal germ cell maturation. J Androl 23(5):622–628
- Villaescusa JC, Allard P, Carminati E, Kontogiannea M, Talarico D, Blasi F, Farookhi R, Verrotti AC (2006) Clast4, the murine homologue of human eIF4E-Transporter, is highly expressed in developing oocytes and post-translationally modified at meiotic maturation. Gene 367:101–109
- Villalba A, Coll O, Gebauer F (2011) Cytoplasmic polyadenylation and translational control. Curr Opin Genet Dev 21(4):452–457
- Wong LC, Schedl P (2011) Cup blocks the precocious activation of the orb autoregulatory loop. PLoS One 6(12), e28261
- Wu X, Brewer G (2012) The regulation of mRNA stability in mammalian cells: 2.0. Gene 500(1):10–21
- Wu X, Viveiros MM, Eppig JJ, Bai Y, Fitzpatrick SL, Matzuk MM (2003) Zygote arrest 1 (Zar1) is a novel maternal-effect gene critical for the oocyte-to-embryo transition. Nat Genet 33(2):187–191
- Xie T, Spradling AC (2000) A niche maintaining germ line stem cells in the Drosophila ovary. Science 290(5490):328–330
- Xu EY, Moore FL, Pera RA (2001) A gene family required for human germ cell development evolved from an ancient meiotic gene conserved in metazoans. Proc Natl Acad Sci U S A 98(13):7414–7419
- Xu EY, Chang R, Salmon NA, Reijo Pera RA (2007) A gene trap mutation of a murine homolog of the Drosophila stem cell factor Pumilio results in smaller testes but does not affect litter size or fertility. Mol Reprod Dev 74(7):912–921
- Yamamoto TM, Cook JM, Kotter CV, Khat T, Silva KD, Ferreyros M, Holt JW, Knight JD, Charlesworth A (2013) Zar1 represses translation in Xenopus oocytes and binds to the TCS in maternal mRNAs with different characteristics than Zar2. Biochim Biophys Acta 1829(10):1034–1046

- Yang CK, Yen P (2013) Differential translation of Dazap1 transcripts during spermatogenesis. PLoS One 8(4), e60873
- Yang J, Medvedev S, Yu J, Tang LC, Agno JE, Matzuk MM, Schultz RM, Hecht NB (2005) Absence of the DNA-/RNA-binding protein MSY2 results in male and female infertility. Proc Natl Acad Sci U S A 102(16):5755–5760
- Yang HT, Peggie M, Cohen P, Rousseau S (2009) DAZAP1 interacts via its RNA-recognition motifs with the C-termini of other RNA-binding proteins. Biochem Biophys Res Commun 380(3):705–709
- Yasuda K, Kotani T, Ota R, Yamashita M (2010) Transgenic zebrafish reveals novel mechanisms of translational control of cyclin B1 mRNA in oocytes. Dev Biol 348(1):76–86
- Zeng M, Deng W, Wang X, Qiu W, Liu Y, Sun H, Tao D, Zhang S, Ma Y (2008) DAZL binds to the transcripts of several Tssk genes in germ cells. BMB Rep 41(4):300–304
- Zhang Y, Sheets MD (2009) Analyses of zebrafish and Xenopus oocyte maturation reveal conserved and diverged features of translational regulation of maternal cyclin B1 mRNA. BMC Dev Biol 9:7
- Zhong J, Peters AH, Lee K, Braun RE (1999) A double-stranded RNA binding protein required for activation of repressed messages in mammalian germ cells. Nat Genet 22(2):171–174
- Zickler D, Kleckner N (1999) Meiotic chromosomes: integrating structure and function. Annu Rev Genet 33:603–754

Post-transcriptional Regulation of VEGF-A

8

Hervé Prats and Christian Touriol

1 Introduction

Vascular Endothelial Growth Factor A (VEGF-A) is a protein of critical importance in embryonic development and during adulthood. Even a slight variation in its normal level can have serious physiological consequences and as a result its expression is stringently regulated at every conceivable stage. Thus, the expression of VEGF-A represents a useful model that demonstrates gene regulation at its most sophisticated.

VEGF-A acts on endothelial cells where it is required for cell survival, growth and migration. It is the main protein involved in the induction of angiogenesis, the formation of new blood vessels, and is therefore essential for adult organ growth and repair. VEGF-A plays an important role in physiological situations that affect blood vessels, such as the menstrual cycle, wound repair, adaptation to hypoxia and, importantly, embryonic development (Carmeliet 2005a; Ferrara 2005). Various pathological conditions also show elevated VEGF-A levels, such as proliferative retinopathies, arthritis, psoriasis and cancer (Ferrara 1999, 2002; Folkman 1995). In cancer, VEGF-A is crucial for tumor development since it stimulates the growth of new blood vessels from nearby capillaries (tumor angiogenesis), which allows tumor cells to acquire oxygen and nutrients and ultimately leads to metastasis.

The importance of keeping tight control of VEGF-A expression has been clearly demonstrated in transgenic mice. Both the deletion of a single VEGF-A allele or the modest overexpression of VEGF-A result in defective vascularization and subsequent embryonic lethality (Carmeliet et al. 1996; Ferrara et al. 1996;

H. Prats • C. Touriol (⊠)

Inserm UMR1037, Centre de Recherches en Cancérologie de Toulouse, ERL5294 CNRS, 2 avenue Hubert Curien, Oncopole entrée C, CS 53717, 31037 Toulouse Cedex 1, France

Université Toulouse III Paul-Sabatier, Toulouse, France e-mail: christian.touriol@inserm.fr

[©] Springer International Publishing Switzerland 2016

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_8

Miquerol et al. 2000). In addition, conditional gain and loss of function experiments in the erythroid lineage have demonstrated that alteration of VEGF-A levels during development significantly affects erythropoiesis in mouse embryos (Drogat et al. 2010). These studies emphasize the critical role of VEGF-A in developmental angiogenesis. In adult mice, tissue-specific overexpression of VEGF-A causes serious problems in processes such as angiogenesis and vascular hyperpermeability (Larcher et al. 1998), angioma formation and organ development (Dor et al. 2002), and can lead to severe proliferative retinopathy and retinal detachment (Ohno-Matsui et al. 2002). These results have been corroborated by conditional transgenic knock-out models of VEGF-A which also show impaired vascular functions (Eremina et al. 2008; Sivaskandarajah et al. 2011).

There are nine different isoforms of VEGF-A which execute their roles through binding to one of two tyrosine kinase receptors, VEGFR1 and VEGFR2 (Carmeliet 2005b; Ferrara et al. 2003). Each isoform is thought to have specific roles depending on its level of expression, and the same isoform can exert different effects at distinct anatomical sites. Expression of the VEGF-A isoforms is regulated at the transcriptional level but it is the post-translational level that provides the greatest display of regulatory diversity, involving mRNA stabilization, alternative splicing and translational regulation mechanisms.

Given that this level of complexity is possible from just one gene is impressive, and shows off the elaborate nature of gene regulation. It is for this reason that VEGF-A is used as a paradigm for the intricate regulation of gene expression, particularly at the post-transcriptional level.

2 VEGF-A Transcriptional Regulation

The human VEGFA gene is approximately 14 kb in length and is located on chromosome 6 at position 6p21.1 (Vincenti et al. 1996). Figure 8.1 shows the structure of the gene, which is composed of eight exons and seven introns (Tischer et al. 1991). In humans, the main VEGFA promoter sequence spans 2.36 kb (Buteau-Lozano et al. 2002) but the promoters for the mouse and rat genes are only 1.2 kb long (Levy et al. 1995; Shima et al. 1996). The human promoter contains several consensus-binding sites for transcriptional regulators such as AP1, AP2 and Sp1, all of which are themselves regulated by growth factors, cytokines, hormones, tumour suppressor genes and oncogenes. A hypoxia response element (HRE) is also present within the 5' flanking region, but interestingly the VEGF-A promoter does not contain a consensus TATA box (reviewed in Pages and Pouyssegur 2005). An alternative promoter has also been found within the human VEGF-A 5'UTR (Akiri et al. 1998), with a transcription start site located 633 nucleotides downstream of the main starting site (Fig. 8.1). Unlike the main promoter this alternative promoter is insensitive to hypoxia, suggesting that there is no cross-regulation with the main promoter region. This therefore offers the potential of regulation under distinct conditions and different spatiotemporal expression patterns (Akiri et al. 1998).



Fig. 8.1 Human *VEGFA* gene structure and exon composition of the isoforms generated by alternative splicing. (*Top*) The human *VEGFA* gene spans 16,272 bp of chromosome 6p21.1 and consists of eight exons and seven introns. In exon 1 *red arrows* show the start of the two promoter sites and *black lines* show the main translation start sites (AUG codons). In exon 8 the two alternative stop codons are indicated in *red. (Bottom)* Exon content of all known VEGF-A isoforms which are named according to the number of amino acids in the human protein. All isoforms contain exons 1–4 and one of two versions of exon 8 (8a or 8b) depending on which terminal splice site is used. Splice site 8a produces the VEGF-Axxx (or xxxa) pro-angiogenic isoforms and 8b produces the VEGF-Axxxb anti-angiogenic isoforms. Exons 6 and 7 encode heparin-binding domains so affect the diffusibility as an inverse relationship with their affinity for the extra cellular matrix

The most widely-studied physiological stimulus causing transcriptional up-regulation of the *VEGFA* gene is hypoxia. Hypoxic conditions initiate angiogenesis via VEGF-A throughout embryogenesis and during cancer progression. It is now wellestablished that the Hypoxia Inducible Factor (HIF) transcriptional activators are key mediators of the hypoxic response and the pathways leading to their activation are well-characterised. Both HIF-1 and HIF-2 bind to the VEGF-A hypoxia response element (HRE) (Forsythe et al. 1996; Blancher et al. 2000).

Since its discovery in 1989, the transcriptional regulation of VEGF-A has been extensively studied and a wide range of factors are known to be at play. However, more recent research has focused on the post-transcriptional regulation of VEGF-A, and this next stage of regulation has proven to provide yet more complexity.

3 VEGF-A Post-transcriptional Regulation

3.1 Alternative Splicing

The splicing process provides an opportunity for regulation of both the final protein structure and the level of protein expression. Different combinations of exons can be retained or removed to create a diverse array of mature mRNAs from the single pre-mRNA. In addition, alternative splicing within the non-coding regions of the mRNA controls the final number of regulatory elements such as translation enhancers or RNA stability domains that can significantly affect protein expression levels.

Through alternative splicing, nine isoforms of VEGF-A have been first found which are named according to the total number of amino acids in the mature proteins: 111, 121, 145, 148, 162, 165, 183, 189 and 206. In addition, some of these isoforms exist in VEGF-Axxxa or VEGF-Axxxb versions (Fig. 8.1). The VEGF-A gene contains eight exons. 1-4 are constitutive, being present in all isoforms, with exon 1 encoding an N-terminal signal of 26 hydrophobic amino acids that is typical of secreted proteins. Exons 5-8 are alternatively spliced to produce the different sized isoforms, with exons 6 and 7 containing basic residues that confer an affinity for heparin-binding. Heparin sulphate proteoglycans are present at the cell surface and within the extracellular matrix, thus the presence or absence of exons 6 and 7 affects the ability of that secreted VEGF-A isoform to diffuse away from the cell. In this way the presence or absence of exons 6 and/or 7 controls the spatial distribution and bioavailability of the VEGF-A isoforms (Vempati et al. 2011). As an example, VEGF-A121 lacks exons 6 and 7 thus does not bind heparin and is free to diffuse away after being released from the cell (Fig. 8.1). In contrast the VEGF-A165 and 189 isoforms are able to bind to heparin sulfate on the cell surface and in the extracellular matrix (Houck et al. 1992). These different heparin binding affinities result in the formation of a VEGF-A gradient which is essential for the process of angiogenesis, with soluble isoforms acting at distal sites to promote vascular recruitment and the extracellular membrane-associated isoforms acting locally to promote the expansion of capillary beds (Grunstein et al. 2000). The fact that an identical isoform can have distinct activities at different anatomical sites suggests that the microenvironment of individual tissues dictates VEGF-A function (Guo et al. 2001).

The C-terminus of VEGF-A is encoded by exon 8 and contains an alternative 3' splice site that gives rise to the VEGF-Axxxa or VEGF-Axxxb versions of the isoform. Even if there is still controversy regarding the existence of the VEGF-Axxxb isoforms (Harris et al. 2012), in vitro and in vivo studies have demonstrated that the

VEGF-Axxx (or -xxxa) isoforms are pro-angiogenic and are up-regulated in tumours whereas the VEGF-Axxxb isoforms are anti-angiogenic and are downregulated in tumours (Bates et al. 2002; Woolard et al. 2004; Pritchard-Jones et al. 2007). To mediate their anti-angiogenic properties, the VEGF-Axxxb isoforms are thought to bind to the VEGF-A receptors but impair their downstream signaling (Woolard et al. 2004; Cebe Suarez et al. 2006), since when the VEGF-Axxxb and -xxx isoforms were co-expressed they acted in a dominant negative way and only partial receptor agonist activity was detected. This also suggests that the inhibitory function of the xxxb isoforms is mediated through competitive binding (Kawamura et al. 2008; Woolard et al. 2009). Further evidence for this has been found using recombinant human VEGF-A165b which possesses similar affinity towards the anti-Vascular Endothelial Growth Factor antibody (bevacizumab) as that of VEGF-A165, supporting the idea of inhibition through competitive binding. Interestingly, this also suggests that the balance between the expression of the anti-angiogenic and pro-angiogenic isoforms could regulate tumour growth and in the same way the antiangiogenic xxxb isoforms could affect the sensitivity of tumours to bevacizumab through competitive binding (Varey et al. 2008). It was also demonstrated that the balance between the expression of the anti and pro-angiogenic isoforms could regulate follicle development (McFee et al. 2012), spermatogonial stem cell homeostasis in vivo (Caires et al. 2012) but also plays a critical role in the regulation of glomerular permeability (Oltean et al. 2012).

Recombinant human VEGF-A165b shows anti-angiogenic activity in eye hypoxiadriven angiogenesis (Konopatskaya et al. 2006). In addition, recombinant human VEGF-A165b possesses a similar affinity towards the anti-VEGF antibody (bevacizumab) as that of VEGF-A165, supporting the idea of inhibition through competitive binding. Interestingly, this also suggests that the balance between the expression of the anti-angiogenic and pro-angiogenic isoforms could regulate tumour growth and in the same way the anti-angiogenic xxxb isoforms could affect the sensitivity of tumours to bevacizumab through competitive binding (Varey et al. 2008).

The distal splice site used to generate the VEGF-Axxxb isoforms is 66 nucleotides further along the gene from the proximal splice site (Figs. 8.1 and 8.2) and results in a Ser-Leu-Thr-Arg-Lys-Asp C-terminal (Bates et al. 2002). This is very different from the Cys-Asp-Lys-Pro-Arg-Arg C-terminal generated by the proximal splice site in VEGF-Axxx isoforms and undoubtedly results in a distinct tertiary structure of the VEGF-Axxxb isoforms since they contain an acidic residue (Asp) in place of the Cys-160 present in the VEGF-Axxx isoforms, which forms a disulphide bond with the Cys-146 from exon 7. This was clearly proven for VEGF-A165 when the highly-charged C-terminal Pro-Arg-Arg tail of the VEGF-A165 isoform was replaced by the neutral Arg-Lys-Asp of the VEGF-A165b isoform and resulted in a profound alteration of the structure-function relationship of VEGF-A (Cui et al. 2004). VEGF-Axxxb isoforms have been identified for VEGF-A121, VEGF-A183, VEGF-A145, VEGF-165 (Perrin et al. 2005) and VEGF-A189 (Miller-Kasprzak and Jagodzinski 2008). VEGF-A165b was the first xxxb isoform identified and it is the most widely-studied (Bates et al. 2002). Its over-expression inhibits the growth of prostate carcinoma, Ewing's sarcoma and renal cell carcinoma in xenografted mouse



Fig. 8.2 Human VEGF-A mRNAs and their major regulatory elements. (*Top*) Schematic of VEGF-A mRNA transcribed from the two alternative promoters. Regulatory elements in the untranslated regions (UTR) and the coding region are indicated; positions are numbered according to the mRNA sequence of the 5' end of VEGF-A189. In the 5'UTR, two AUG start codons and three non-canonical CUG start codons are shown. IRES-B: internal ribosome entry site B, which controls transcription initiation from the CUG start codon. The IRES-A controls transcription initiation from the AUG start codons. uORF: upstream open reading frame, located in the IRES-A. In the 3'UTR, the two alternative polyadenylation sites are shown (polyA signal). *AREs* AU rich elements, *HSR* hypoxia stability region. The area around the riboswitch element has been expanded to show the binding sites for hnRNPL and the GAIT translation complex, which compete with each other for binding to this element in the regulation of oxidative stress. (*Bottom*) Initiation of translation at the CUG start codon produces the L-VEGF-A isoforms. These can be cleaved during the maturation process to yield an intracellular N-VEGF-A product and the mature secreted VEGF-A isoform that resembles the VEGF-A isoforms translated from the AUG start codon

tumor models (Rennel et al. 2008) and inhibits tumor cell-mediated migration and the proliferation of endothelial cells (Rennel et al. 2008). Interestingly, mammary alveolar development during lactation is also inhibited by VEGF-A165b (Qiu et al. 2008) and endogenous VEGF-A165b contributes to survival of trophoblasts exposed to lower oxygen tensions through an autocrine pathway (Bills et al. 2014).

Although many VEGF-A isoforms have now been identified, most VEGF-Aproducing cells are thought to preferentially express VEGF-A121, VEGF-A165 and VEGF-A189. These are known as VEGF-A120, 164 and 188 in the mouse since there is one less amino acid in each mouse VEGF-A isoform. To clearly demonstrate the crucial role of alternative splicing in the regulation of VEGF-A activity, and to investigate the specific roles of the individual isoforms, transgenic mice have been generated that express only a single VEGF-A isoform. Mouse embryos expressing only VEGF-A120 (*VEGFA120/120*) show impaired post-natal cardiac angiogenesis, resulting in severe myocardial ischemia, early post-natal death and impaired lung vascular development (Mattot et al. 2002). Half of the embryos die in the perinatal period due to congenital birth defects, with the other half perishing within 2 weeks after birth, in part due to myocardial ischemia (Carmeliet et al. 1999). Similar studies with mice expressing only VEGF-A188 (*VEGFA188/188*) also show significant effects, with half of all embryos dying between embryonic stage E9.5 and E13.5 (Stalmans et al. 2002). Interestingly, mice expressing only VEGF-A164 (*VEGFA164/164*) are healthy (Stalmans et al. 2002).

Together these transgenic mouse studies suggest that different alternatively spliced isoforms are required at different stages in normal development. Other findings support this since it has been shown that different isoforms are expressed in distinct spatio-temporal patterns both during embryonic development and in adult tissues (Ng et al. 2001). However, it is also known that there is some functional redundancy between isoforms, for example during the initial formation of arch arteries (Stalmans et al. 2003).

The importance of alternative splicing of VEGF-A pre-mRNA in terms of VEGF-A regulation is now well-established. Despite this, very little is known about the mechanisms regulating alternative splicing and thus how the cell controls the levels of the different VEGF-A isoforms. Exonic splicing enhancer/silencer sequences are sequences of DNA that promote/reduce the recognition and use of splice sites. An exonic silencer sequence (ESS) has been proposed to be present in exon 6 of VEGF-A, but the regulatory proteins interacting with this sequence remain unknown (Wang et al. 2009). In the alternative splicing process, Serine/Arginine rich proteins (SR proteins) also play a key role in identifying splice sites. Under hypoxic conditions the expression and phosphorylation of the SR proteins ASF/ SF2, SRp20 and SRp40 were shown to correlate with increased VEGF-A expression and a shift towards expression of the VEGF-A121 isoform in the endometrial cancer cell line RL95 (Elias and Dias 2008). Increased expression of VEGF-A121 over the longer isoforms is known to be mediated by two splicing factors of the U2AF65 protein family, namely CAPERalpha and CAPERbeta (Dowhan et al. 2005). SR proteins have also been implicated in the alternative splicing of exon 8, with the ASF/SF2 and SRp40 proteins favoring the use of the pro-angiogenic proximal splice-site, and the SRp55 protein promoting the use of the anti-angiogenic distal splice-site (Nowak et al. 2008). However, other factors are also known to play a role in the pro-/anti-angiogenic decision: the E2F1 transcription factor has been shown to favor the expression of anti-angiogenic VEGF-Axxxb isoforms through a mechanism involving the expression of the splicing factor SC35 (Merdzhanova et al. 2010), and now a link between the Wilms' tumour suppressor gene (WT1) and the regulation of VEGF-A pro-/anti-angiogenic alternative splicing has been established (Amin et al. 2011). WT1 binds to the promoter region of SRPK1 (Serine/ Arginine-rich protein-specific kinase 1), a kinase that regulates the activity of SR proteins. This decreases the expression of the kinase, resulting in decreased AFS/ SF2 phosphorylation and nuclear localization, and an increase in expression of antiangiogenic VEGF-A165b. Conversely, mutation of the WT1 tumor suppressor gene

results in ASF/SF2 hyperphosphorylation and expression of pro-angiogenic VEGF-A isoforms (Amin et al. 2011). These data suggest that tumors lacking functional WT1 or containing enhanced SRPK1 expression may control angiogenesis through the regulation of alternative splicing of VEGF-A.

There is no doubt that our understanding of the mechanisms regulating alternative splicing is still in its infancy. A recent study proved that the alternative splicing field still has many more doors to open when it described a novel VEGF-A isoform in the lung tissue of a legally aborted female fetus. This alternatively spliced isoform possessed a 20 bases insertion in the third intron, inducing a frame shift mutation that introduced a stop codon in the middle of the fourth exon (Zhou et al. 2012). The mechanism involved in this splicing regulation and whether this alteration impact VEGF-A protein structure and/or function is still unknown but this discovery highlights the discrepancies in our understanding of this important process.

Reinforcing this biological control, a VEGF-A stop-codon readthrough mechanism has been recently discovered that leads to a 22 amino-acids extension generating new antiangiogenic isoforms (Eswarappa et al. 2014).

3.2 mRNA Stability

In cells different mRNAs have very different stabilities. The more stable an mRNA is the longer its lifetime and hence the more protein is produced from it. Short mRNA half-lives allow cells to alter protein synthesis rapidly when necessary (for example in response to environmental changes such as nutrient levels, cytokines, hormones or stress) and in this way the control of mRNA stability is an important regulator of protein expression (Mitchell and Tollervey 2000; Shim and Karin 2002).

VEGF-A mRNA is highly unstable under normal oxygen and nutrient conditions with a half-life of 15-40 min in vitro (Dibbens et al. 1999; Ikeda et al. 1995; Levy et al. 1996; Shima et al. 1995). Hypoxic conditions induce VEGF-A expression by increasing mRNA stability through a regulatory process that is independent of HIF1A-induced transcription (Ryan et al. 2000). Like most mRNAs with a short half-life, VEGF-A mRNA contains many AU-rich elements (AREs) which are well-known cis-sequences generally composed of different nonameric (AUUUA) or pentameric (AUUUAUUUA) consensus or U-rich sequences. They increase the rate of degradation of the mRNA and so by blocking or promoting access to these sites the cell can control the stability of the mRNA (Barreau et al. 2005; Gingerich et al. 2004). The VEGF-A AREs are clustered in the 3'UTR section of VEGF-A mRNA (Fig. 8.2). In humans, the stability of VEGF-A mRNA under hypoxic conditions is mediated through the 3'UTR (Claffey et al. 1998; Goldberg-Cohen et al. 2002; Levy et al. 1998), therefore research on the control of VEGF-A expression in response to hypoxia has focused on ARE-binding proteins. Interestingly, it has been shown that degradation of mouse VEGF-A mRNA under normoxic conditions requires the independent action of destabilizing elements from across the mRNA (in the 3'UTR, the coding region and the 5'UTR). In contrast to human VEGF-A mRNA, under hypoxic conditions mouse VEGF-A mRNA requires all three sections and these

coordinate to stabilize the mRNA (Dibbens et al. 1999). In humans a number of RNA-binding proteins have been shown to interact with the 3'-UTR elements of VEGF-A mRNA and increase its stability including: the CSD/PTB complex (Cold Shock Domain/Polypyrimidine Tract Binding Protein) (Coles et al. 2004); MDM2, a protein translocated from the nucleus to the cytoplasm under hypoxic conditions (Zhou et al. 2011); hnRNPL, which associates with hnRNP complexes to promote mRNA formation, processing and packaging (Shih and Claffey 1999); the double strand RNA binding protein DRBP76/NF90, which under hypoxic conditions facilitates VEGF-A expression by promoting mRNA loading onto polysomes and translation (Vumbaca et al. 2008); HSP70, which was found to bind and stabilize VEGF-A mRNA through a mechanism independent of its chaperone function (Kishor et al. 2012); and HuR (Levy 1998). HuR is a member of the ELAV protein family, which also includes the Hel-N1, HuC and HuD proteins (Fan and Steitz 1998; King 2000; Levine et al. 1993; Peng et al. 1998). ELAV proteins bind to AREs and are thought to regulate the stability of mRNA from splicing through to translation by generating multimeric "ribonucleosomes" due to their nucleation and cooperative binding properties. For example, HuR bound to the 3'UTR has been shown to interact directly with PAIP2 (Poly(A)-Binding Interacting Protein 2) that is bound to a distinct region of the mRNA, and it has been suggested that interaction between these proteins can influence the overall RNA structure and render it inaccessible to endonucleases (Onesto et al. 2004, 2006). Recently it was also proposed that ELAV/Hu proteins can displace miRNAs (which would otherwise silence the mRNA; see Sect. 3.4.3) and thus suppress their destabilizing properties (Simone and Keene 2013). This seems to be the case for mouse VEGF-A mRNA since the HuR and miR-200b binding sites overlap and it was shown that the miR-200binduced suppression of VEGF-A expression was competitively antagonized by HuR at the 3'UTR (Chang et al. 2013). Finally, HuR has also been implicated in effective VEGF-A mRNA export from the nucleus and loading onto active polysomes under hypoxia. Along with hnRNPL and hnRNPA1, the extra-nuclear shuttling of these mRNA-binding proteins has also been shown to regulate VEGF-A mRNA stability (Vumbaca et al. 2008).

The 3'UTR ARE of VEGF-A mRNA can also be the target of proteins that destabilize mRNA in various mammalian cell types, for example AUF1 and tristetraprolin (TTP) (Bernstein and Ross 1989; DeMaria and Brewer 1996; Gorgoni and Gray 2004; Stoecklin et al. 2003; Zhang et al. 2002). AUF1 promotes the assembly of other factors necessary for recruiting the mRNA degradation machinery, such as translation initiation factor eIF4G, heat-shock cognate protein hsc70, lactate dehydrogenase and the poly(A)-binding protein. Interestingly, although the poly(A)binding protein is a stabilizing factor for polyadenylated mRNAs, it has destabilizing effects on VEGF-A mRNA (Gorgoni and Gray 2004; Ma et al. 2006). It was recently shown that AUF1-mediated regulation of VEGF-A expression in mouse macrophagelike RAW-264.7 cells is mediated through its C-terminus domain that contains a region with three arginine-glycine-glycine (RGG) motifs (Fellows et al. 2012). AUF1 itself can be regulated by many signaling pathways and therefore is subjected post-translational modifications numerous such as phosphorylation, to

glycosylation, methylation and ubiquitination (for a review, see Gratacos and Brewer 2010). TIS11/Tristetraprolin (TTP) is another ARE-binding protein that regulates the destabilization of VEGF-A mRNA through a signaling pathway involving casein kinase 2 and p38 MAPK (Lee et al. 2011). These examples show the breadth of cellular signaling pathways involved in the regulation of VEGF-A mRNA stability and the depth of their control by the cell.

3.3 Alternative Polyadenylation

Polyadenylation is the addition of a poly(A) tail to the 3'UTR of an mRNA and is important for mRNA stability, nuclear export and translation. Before the poly(A) tail is added, the 3'end of the mRNA is cleaved at specific sites, thus if an mRNA contains more than one polyadenylation site this can produce alternative transcripts, as for alternative splicing.

Although the 3'UTR is known to play a pivotal role in the post-transcriptional control of VEGF-A expression (Bartel 2009), our knowledge of the use and regulation of poly(A) signals is very limited. VEGF-A cDNA sequencing has revealed the presence of two major polyadenylation sites: a consensus AAUAAA site and a noncanonical AUUAAA site, located 399 and 1902 nucleotides after the stop codon respectively (Fig. 8.2) (Claffey et al. 1998; Dibbens et al. 2001). The resulting mRNAs from both sites have been characterized (Leung et al. 1989; Keck et al. 1989), but only one study has investigated the use of VEGF-A alternative polyadenylation (Dibbens et al. 2001). Using a mouse model it reported that the majority of VEGF-A transcripts are processed at the distal polyadenylation site (resulting in mRNAs containing the longer form of the 3'UTR), and that the same site is used under both hypoxic and normoxic growth conditions (Dibbens et al. 2001). Given the large section of the regulatory 3'UTR that is missing in transcripts using the proximal polyadenylation site, it is possible that the resulting VEGF-A isoforms play distinct roles. Therefore, much more work is needed to ascertain the importance of this method of VEGF-A regulation.

3.4 Translational Regulation

3.4.1 Internal Ribosome Entry Site (IRES) Elements and Alternative Initiation at Non-AUG Codons

In eukaryotic cells translation usually depends on the presence of an m7G cap at the 5' terminus of the mRNA. The cap is required for assembly of the initiation complex, and allows translation to begin at the start codon further downstream. Traditionally the start codon is an AUG coding for methionine; however, alternative start codons are now known to exist, such as CUG which codes for leucine and whose initiation pathways are thought to be independent of those of the AUG codon (Starck et al. 2012). In addition, translation of a number of cellular mRNAs can be initiated from the middle of an mRNA sequence through a cap-independent

mechanism. These mRNAs contain an internal ribosome entry site (IRES) within their 5' UTR. IRESs are structural elements that are specifically required to maintain and/or activate the expression of specific proteins during cell stress situations when cap-dependent translation is compromised (Spriggs et al. 2008). They are thought to have an extensive predicted secondary structure and have been reported mostly in mRNAs containing long 5'UTRs with a high GC content. To date more than 70 viruses and more than hundred eukaryotic mRNAs are known to contain at least one IRES, including VEGF-A (Mokrejs et al. 2010).

VEGF-A mRNAs have a 1038-nucleotide-long GC-rich 5'UTR upstream of the classical AUG start codon which contains three in-frame alternative CUG start codons and two IRESs (Fig. 8.2) (Akiri et al. 1998; Huez et al. 1998; Miller et al. 1998). The first IRES to be identified, IRES-A, lies within the 300 nucleotides upstream from the AUG start codon and controls the initiation of translation at this point (Fig. 8.2). IRES-B is located closer to the 5'cap, 16 nucleotides upstream of the first CUG codon and controls initiation from the two alternative CUG start codons. This IRES-mediated control of AUG versus CUG has been demonstrated both in vitro (Akiri et al. 1998; Huez et al. 1998) and in vivo (Bornes et al. 2007). In vivo, VEGF-A IRESs are thought to be particularly important for responding to local environment stresses where cap-dependent translation is inhibited, such as hypoxia, and indeed it has been shown that IRES have a low activity in embryos and adult tissues but allow efficient translation at early time points in ischemic muscle (Bornes et al. 2007). Recent studies have now focused on identifying the regulatory factors involved in IRES-mediated translation. One study used a high-throughput screening approach that combined siRNA treatment with transfection of a VEGF-A IRES reporter mRNA, to identify and validate MAPK3 kinase as a novel positive regulator (Casanova et al. 2012). The DEAD-box RNA helicase 6 (DDX6) has also been identified as an IRES trans-acting factor. Using MCF-7 cell line extracts it was shown that under normoxic conditions recombinant DDX6 inhibits VEGF-A IRESmediated translation whereas hypoxic conditions caused a decrease in DDX6 levels that led to the induction of VEGF-A expression (de Vries et al. 2013). VEGF-A IRES can also be controlled from within the mRNA and in this way they have been shown to play a role in dictating the use of the alternative start codons, AUG and CUG. For example, mRNA encoding the alternatively spliced VEGF-A165 and -189 sequences can be efficiently expressed through initiation events using both start codons. The choice of initiation site is determined through modulation of IRES-A activity by the alternatively spliced coding sequences (Bornes et al. 2004). It is thought that different splice variants contain different elements that, through long-range interactions within the VEGF-A mRNA, could regulate IRES-A activity to promote/inhibit the use of the AUG site, however the exact molecular mechanism of this control is still unknown.

Initiation of translation from the CUG codon is equally efficient regardless of which splice variant is expressed (Bornes et al. 2004). The first CUG start codon is located 539 nucleotides upstream of the coding sequence (Huez et al. 2001; Meiron et al. 2001; Tee and Jaffe 2001; Touriol et al. 2003). Initiation of translation at this CUG codon leads to the production of longer VEGF-A isoforms (L-VEGF-A)

containing an additional 180 amino acids (Fig. 8.2). During maturation, L-VEGF-A can be cleaved at the peptide signal sequence to generate both a shorter secreted VEGF-A isoform and an intracellular portion termed N-VEGF-A. Interestingly, the VEGF-A121 isoform is known to be exclusively expressed through initiation at the CUG codon and thus is a maturation product of L-VEGF-A121. Following L-VEGF-A cleavage, the N-VEGF-A fragment is released and translocates to the nucleus (hence the "N" prefix). It is a 23-kDa NH2-specific peptide containing 206 amino acids which are highly conserved among mammals, suggesting that N-VEGF-A has an important function, the details of which are yet to be identified (Fig. 8.2) (Huez et al. 2001; Tee and Jaffe 2001; Rosenbaum-Dekel et al. 2005). The role and purpose of L-VEGF-A is also not well-understood although it has been suggested that it may serve as a reservoir for the generation of shorter isoforms since an increase in the longer intracellular L-VEGF-A165 and L-VEGF-A189 protein isoforms has been linked to a reduction in shorter secreted isoforms (Chiarini et al. 2006). Despite our lack of understanding of the role of L-VEGF-A, its importance is demonstrated by a single nucleotide polymorphism in the human VEGF-A gene (-634 C-G) that causes IRES-B dysfunction and leads to a 17 % reduction in initiation at the CUG codon and thus a decrease in L-VEGF-A expression (Lambrechts et al. 2003). This polymorphism has been associated with a number of serious conditions, such as an increased risk of motor neuron degeneration in amyotrophic lateral sclerosis (ALS) (Lambrechts et al. 2003), the development of diabetic macular edema that correlates with macular retinal thickness in type 2 diabetes (Awata et al. 2005), an increased aggressiveness of breast cancer (Jin et al. 2005), and an increased risk of gastric (Guan et al. 2009) and prostate cancer (Sfar et al. 2006). These studies not only provide evidence of the crucial role of IRES function and the significance of the L-VEGF-A isoforms but also illustrate how these distinct regulatory elements are interdependent such that a single mutation in one element has such a serious downstream effect on many others.

3.4.2 RNA G-Quadruplex Structure

G-quadruplexes are guanine-rich regions of the mRNA that can fold up to form secondary structures organized in stacks of planar layers of guanine tetrad (or quartet) units. It is emerging that these elements have regulatory functions in different steps of RNA metabolism, including mRNA translation (Millevoi et al. 2012). G-quadruplexes are found in coding and non-coding mRNA. The majority of those that are located in 5'UTRs play an inhibitory role in cap-dependent translation (for example in Zic-1, ESR1, NRAS or MT3-MMP mRNA), probably through the recruitment of stabilizing proteins to prevent ribosome scanning (Bugaut and Balasubramanian 2012).

VEGF-A mRNA contains a 17 nucleotide-long element within the IRES-A site (nucleotides 774-790) that can fold into a two-G-quartet quadruplex structure. It is thought that this is important for IRES-A function since IRES-A activity was blocked by mutations that disrupted the intramolecular G-quadruplex structure (Morris et al. 2010). This finding is quite surprising since most naturally-occurring G-quadruplexes with the ability to regulate translation are at least three-G-quartet

quadruplex structures while artificially engineered two-G-quartet quadruplexes like that found in VEGF-A have shown only a moderate stability (Bugaut and Balasubramanian 2012). Thus, it is important to establish whether the VEGF-A G-quadruplex contributes to IRES-A activation under stress conditions such as hypoxia, endoplasmic reticulum stress or ischemia, since this is when the IRESs are active and required for the initiation of translation. If this structure affects IRES-A activation under these conditions then G-quadruplex formation will be confirmed as yet another mechanism of VEGF-A regulation.

3.4.3 Upstream Open Reading Frame (uORF)

Upstream open reading frames (uORFs) are generally short sequences located within the 5'UTR of an mRNA. They can regulate protein translation under normal and stress conditions, usually by inhibiting expression of the main transcript (Calvo et al. 2009; Spriggs et al. 2010). Generally, uORFs act as a constitutive barrier to the scanning ribosome and thereby reduce ribosome access to the main AUG codon. uORFs are found in the 5'UTR of proto-oncogenes, cytokines and many other eukaryote mRNAs, including human 5-HT3A (5-hydroxytryptamine receptor 3A), TPO, BACE1 [β -site APP (Amyloid Precursor Protein) cleavage enzyme 1] and the huntingtin protein (the protein associated with Huntington's disease) (Chatterjee and Pal 2009).

The VEGF-A 5'UTR has a unique, short uORF that is highly-conserved between species and begins at AUG 852, 186 nucleotides upstream of the main AUG start codon (Fig. 8.2). It is located within the IRES-A sequence and is translated through a cap-independent mechanism. It has been suggested that the VEGF-A uORF acts as a cis-regulatory element involved in the shut-off of the CUG codon in the control of VEGF-A121 expression since mutation of AUG 852 (Fig. 8.2) increased translation of the VEGF-A 121 isoform but had no effect on the expression of the VEGF-A165 or -189 isoforms (Bastide et al. 2008). VEGF-A121 translation occurs exclusively downstream of the CUG start codon, thus an active uORF could inhibit its translation and in that way contribute towards controlling the expression of the different VEGF-A isoforms. As a potentially constitutive inhibitor, the VEGF-A uORF may be regulated by specific sequences, trans-acting factors or conditions that have not yet been identified due to the difficulties in predicting the effect of uORF alterations. Despite this, the VEGF-A uORF clearly affects its expression and is another factor contributing to its fine-tuning.

3.4.4 miRNA-Mediated Regulation of VEGF-A

MicroRNAs (miRNAs) consist of a small (20–25 nucleotides) sequence of noncoding RNA that control the translation of genes through RNA silencing. They are generated from local hairpin structures by two RNA endonucleases, Drosha and Dicer (Kim et al. 2009). The importance of miRNA in VEGF-A regulation was first suggested when researchers mutated the dicer gene in mice. This led to retarded development, defective angiogenesis and the death of embryos between days 12.5 and 14.5 of gestation. This phenotype correlated with over-expression of VEGF-A, therefore it was concluded that the defect in angiogenesis was most likely due to a


Fig. 8.3 miRNA-targeting sites in the human VEGF-A mRNA 3'UTR. The 3' untranslated region of human VEGF-A mRNA is shown with the two alternative polyadenylation (PA) sites. The position of each miRNA target site is illustrated and numbered according to the 5'end of the VEGF-A 189 mRNA sequence

deficiency in VEGF-A mRNA processing (Yang et al. 2005). Since then many miR-NAs have been identified that target VEGF-A mRNA, all of which bind to the 3'UTR region. Figure 8.3 summarizes the different miRNA binding sites found. Close to the end of the main ORF is a binding sequence for miR-125a, which has been shown to inhibit the proliferation and metastasis of hepatocellular carcinoma by targeting VEGF-A (Bi et al. 2012). This is followed by a sequence for miR-205, whose expression in glioma cell lines increases VEGF-A expression (Yue et al. 2012). Further along is a binding site for a number of miRNAs, including miR-20a and miR-20b. These were some of the first VEGF-A-binding miRNAs to be identified and, like miR-15b and miR-16, they were found to co-regulate other angiogenic factors in addition to VEGF-A regulation (Hua et al. 2006; Lei et al. 2009). miR-93b also binds to this site; it was shown to be down-regulated under hyperglycemic conditions which correlated with increased VEGF-A expression (Long et al. 2010). The miR-16 responsive element is further along, located 270 nucleotides downstream of the translation stop codon (Karaa et al. 2009), and affects angiogenesis in multiple myeloma (Sun et al. 2013). Interestingly, miR-195, which was demonstrated to suppress angiogenesis and metastasis of hepatocellular carcinoma (Wang et al. 2013), overlaps the miR-16 binding site, suggesting a mutually exclusive interaction of these 2 miRNAs. Close to the first poly(A) site lies a CA Rich Element (CARE) which is targeted by at least four miRNAs, miR-297, miR-299, miR-567, and miR-609. Their binding can induce a robust inhibition of protein synthesis under normoxic conditions (Jafarifar et al. 2011). miR-145 and miR-199a-5P then binds to an overlapping sequence further downstream. Through its posttranscriptional regulation of VEGF-A in endometrial mesenchymal stem cells miR-199a5P contributes to the pathogenesis of endometriosis (Hsu et al. 2014) while miR-145 has been shown to inhibit tumor angiogenesis, invasion and growth (Zou et al. 2012; Fan et al. 2012). Following this are sites for miR-126 and miR-200b binding; their respective down-regulation in lung cancer cells and in the diabetic

retina cause increased VEGF-A expression (Liu et al. 2009; McArthur et al. 2011). By targeting a sequence located just downstream of the miR-200b binding site, miR-377 regulates mesenchymal stem cell-induced angiogenesis in ischemic hearts (Wen et al. 2014). Finally, the miR-361-5p, miR-29a, miR-29b and miR-190-binding sites are close to the second poly(A) site. MicroRNA miR-361-5p level is inversely correlated with VEGF-A expression in human cutaneous squamous cell carcinoma (Kanitz et al. 2012), and miR-29a, miR-29b and miR-190 were demonstrated to repress metastasis and angiogenesis (Chen et al. 2014; Melo and Kalluri 2013; Hao et al. 2014).

At least two of the miRNA binding sites are known to control VEGF-A expression through competitive binding with regulatory proteins. The RNA binding protein HuR, which plays a role in mRNA stability (see Sect. 3.2), can also bind to the same site as miR-200b. Their competitive binding has been shown to control VEGF-A expression in bone marrow-derived macrophages (Chang et al. 2013) in a mechanism that is evolutionarily conserved, at least between mouse and zebrafish. At a separate location, the hnRNPL protein can bind to the CARE sequence of the 3'UTR when it is delocalized from the nucleus to the cytoplasm under hypoxic conditions. This presumably renders this element inaccessible to the RNA-Induced Silencing Complex with its incorporated miRNAs (the miRISC), which target the same element (Jafarifar et al. 2011). This mechanism also demonstrates the key role of hnRNPL during hypoxia since it is also able to promote VEGF-A mRNA stability (see Sect. 3.2) and plays a role in the VEGF-A mRNA riboswitch (see below).

Interestingly, a miRNA has also been reported to affect the activity of the VEGF-A IRESs. MiR-16 can specifically negatively regulate IRES-B, thereby controlling the expression of the diffusible VEGF-A121 isoform. In contrast, IRES-A is insensitive to miR-16 inhibition (Karaa et al. 2009). This is the first study detailing miRNA-mediated control of the translation of specific VEGF-A isoforms, but it is unlikely to be the last.

3.4.5 Riboswitch

A riboswitch is an element usually found in the untranslated region of mRNA which binds to small molecules or proteins and alters their folding pattern to induce a change in the translation of the mRNA. Binding of the effector molecule thereby allows the mRNA to directly control its own expression in response to changes in the cellular environment. Riboswitches are most commonly known to exist in bacteria, fungi and plants but some have been identified in eukaryotes. One such example is found in the 3'UTR of VEGF-A mRNA (Fig. 8.2). This atypical riboswitch is metabolite-independent and undergoes a binary conformational change that is controlled by differential protein binding in response to environmental signals. Within the riboswitch element are binding sites for the interferon- γ -activated inhibitor of translation (GAIT) complex and the hnRNPL protein, the proximity of which suggests that interactions between these proteins and the riboswitch element are mutually exclusive (Fig. 8.2). In myeloid cells under hypoxic conditions, hnRNPL accumulates within the cell and promotes a conformational switch to promote translation, thereby increasing VEGF-A levels. Under different environmental conditions the stimulus-dependent proteasomal degradation of hnRNPL allows GAIT to bind and repress translation by maintaining the mRNA in its alternative conformation (Ray and Fox 2007; Ray et al. 2009). Within the 3'UTR, the VEGF-A riboswitch sequence encompasses the first polyadenylation site, therefore it is possible that RNA conformational changes could influence the availability of this polyadenylation site and consequently affect the presence of other regulatory elements within the 3'UTR such as AU rich elements and miR-binding sites. The novelty and significance of these studies highlight the continuing expansion of our knowledge of the scope of eukaryotic ribo-regulation.

4 Discussion and Conclusion

The critical roles of VEGF-A during embryogenesis and in the adult are emphasized by the extraordinary levels to which evolution has gone to maintain control of its expression. From one gene, the presence of two promoters, two polyadenylation signals and 14 alternative splicing options mean that 56 potential VEGF-A mRNAs could be expressed. As well as the wide spectrum of post-transcriptional mechanisms described in this chapter, VEGF-A mRNA contains one of the largest varieties of regulatory elements of all known mRNAs. In addition, there are proteins which are involved in multiple regulatory processes (e.g. HuR, hnRNPL) and there is cross-talk between the processes (e.g. differential IRES sensitivity to miRNA inhibition). Moreover, the use of alternative initiation codons enables the expression of at least 29 putative pre-proteins (14 AUG-initiated forms, 14 CUGinitiated "L-VEGF-A" forms and the amino terminal extension N-VEGF-A), each of which exhibit distinct biological functions. It is for all these reasons that VEGF-A is a paradigm for the intricate regulation of gene expression at the posttranscriptional level.

VEGF-A is certainly not alone in its complex regulation. Alternative splicing has been shown to be used by more than 90 % of genes (Pan et al. 2008) and a recent study showed that around 79 % of transcripts have multiple polyadenylation sites (Hoque et al. 2013). On the other hand, translational control is less common and to the best of our knowledge there are only around 100 human mRNAs initiating translation from alternative start codons and only approximately 115 eukaryotic cellular mRNAs reported to contain IRESs, although there is controversy regarding the validity of some suggested IRESs (Gilbert 2010; Shatsky et al. 2010). As new approaches and technologies emerge, it is highly likely that the use of translational mechanisms to regulate gene expression will be found to be more widespread (Ingolia et al. 2009, 2011, 2012). Thus in the future there may be many genes with a level of regulation as complex as that of VEGF-A. There is also undoubtedly much more to learn about the post-transcriptional regulation of VEGF-A itself, but even in its incomplete state it is an impressive example of the depth of our understanding of the different post-transcriptional regulatory processes and provides us with an opportunity to appreciate the incredible complexity that underlies the day-to-day cellular control of just one of our 25,000 genes.

References

- Akiri G, Nahari D, Finkelstein Y, Le SY, Elroy-Stein O, Levi BZ (1998) Regulation of vascular endothelial growth factor (VEGF) expression is mediated by internal initiation of translation and alternative initiation of transcription. Oncogene 17(2):227–236
- Amin EM, Oltean S, Hua J, Gammons MV, Hamdollah-Zadeh M, Welsh GI, Cheung MK, Ni L, Kase S, Rennel ES, Symonds KE, Nowak DG, Royer-Pokora B, Saleem MA, Hagiwara M, Schumacher VA, Harper SJ, Hinton DR, Bates DO, Ladomery MR (2011) WT1 mutants reveal SRPK1 to be a downstream angiogenesis target by altering VEGF splicing. Cancer Cell 20(6):768–780
- Awata T, Kurihara S, Takata N, Neda T, Iizuka H, Ohkubo T, Osaki M, Watanabe M, Nakashima Y, Inukai K, Inoue I, Kawasaki I, Mori K, Yoneya S, Katayama S (2005) Functional VEGF C-634G polymorphism is associated with development of diabetic macular edema and correlated with macular retinal thickness in type 2 diabetes. Biochem Biophys Res Commun 333(3):679–685
- Barreau C, Paillard L, Osborne HB (2005) AU-rich elements and associated factors: are there unifying principles? Nucleic Acids Res 33(22):7138–7150
- Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136(2):215-233
- Bastide A, Karaa Z, Bornes S, Hieblot C, Lacazette E, Prats H, Touriol C (2008) An upstream open reading frame within an IRES controls expression of a specific VEGF-A isoform. Nucleic Acids Res 36(7):2434–2445
- Bates DO, Cui TG, Doughty JM, Winkler M, Sugiono M, Shields JD, Peat D, Gillatt D, Harper SJ (2002) VEGF165b, an inhibitory splice variant of vascular endothelial growth factor, is down-regulated in renal cell carcinoma. Cancer Res 62(14):4123–4131
- Bernstein P, Ross J (1989) Poly(A), poly(A) binding protein and the regulation of mRNA stability. Trends Biochem Sci 14(9):373–377
- Bi Q, Tang S, Xia L, Du R, Fan R, Gao L, Jin J, Liang S, Chen Z, Xu G, Nie Y, Wu K, Liu J, Shi Y, Ding J, Fan D (2012) Ectopic expression of MiR-125a inhibits the proliferation and metastasis of hepatocellular carcinoma by targeting MMP11 and VEGF. PLoS One 7(6), e40169
- Bills VL, Hamdollah-Zadeh M, Soothill PW, Harper SJ, Bates DO (2014) The role of VEGF-A165b in trophoblast survival. BMC Pregnancy Childbirth 14:278
- Blancher C, Moore JW, Talks KL, Houlbrook S, Harris AL (2000) Relationship of hypoxiainducible factor (HIF)-1alpha and HIF-2alpha expression to vascular endothelial growth factor induction and hypoxia survival in human breast cancer cell lines. Cancer Res 60(24):7106–7113
- Bornes S, Boulard M, Hieblot C, Zanibellato C, Iacovoni JS, Prats H, Touriol C (2004) Control of the vascular endothelial growth factor internal ribosome entry site (IRES) activity and translation initiation by alternatively spliced coding sequences. J Biol Chem 279(18):18717–18726
- Bornes S, Prado-Lourenco L, Bastide A, Zanibellato C, Iacovoni JS, Lacazette E, Prats AC, Touriol C, Prats H (2007) Translational induction of VEGF internal ribosome entry site elements during the early response to ischemic stress. Circ Res 100(3):305–308
- Bugaut A, Balasubramanian S (2012) 5'-UTR RNA G-quadruplexes: translation regulation and targeting. Nucleic Acids Res 40(11):4727–4741
- Buteau-Lozano H, Ancelin M, Lardeux B, Milanini J, Perrot-Applanat M (2002) Transcriptional regulation of vascular endothelial growth factor by estradiol and tamoxifen in breast cancer cells: a complex interplay between estrogen receptors alpha and beta. Cancer Res 62(17): 4977–4984
- Caires KC, de Avila JM, Cupp AS, McLean DJ (2012) VEGFA family isoforms regulate spermatogonial stem cell homeostasis in vivo. Endocrinology 153(2):887–900
- Calvo SE, Pagliarini DJ, Mootha VK (2009) Upstream open reading frames cause widespread reduction of protein expression and are polymorphic among humans. Proc Natl Acad Sci U S A 106(18):7507–7512
- Carmeliet P (2005a) Angiogenesis in life, disease and medicine. Nature 438(7070):932-936

- Carmeliet P (2005b) VEGF as a key mediator of angiogenesis in cancer. Oncology 69(Suppl 3): 4-10
- Carmeliet P, Ferreira V, Breier G, Pollefeyt S, Kieckens L, Gertsenstein M, Fahrig M, Vandenhoeck A, Harpal K, Eberhardt C, Declercq C, Pawling J, Moons L, Collen D, Risau W, Nagy A (1996) Abnormal blood vessel development and lethality in embryos lacking a single VEGF allele. Nature 380(6573):435–439
- Carmeliet P, Ng YS, Nuyens D, Theilmeier G, Brusselmans K, Cornelissen I, Ehler E, Kakkar VV, Stalmans I, Mattot V, Perriard JC, Dewerchin M, Flameng W, Nagy A, Lupu F, Moons L, Collen D, D'Amore PA, Shima DT (1999) Impaired myocardial angiogenesis and ischemic cardiomyopathy in mice lacking the vascular endothelial growth factor isoforms VEGF164 and VEGF188. Nat Med 5(5):495–502
- Casanova CM, Sehr P, Putzker K, Hentze MW, Neumann B, Duncan KE, Thoma C (2012) Automated high-throughput RNAi screening in human cells combined with reporter mRNA transfection to identify novel regulators of translation. PLoS One 7(9), e45943
- Cebe Suarez S, Pieren M, Cariolato L, Arn S, Hoffmann U, Bogucki A, Manlius C, Wood J, Ballmer-Hofer K (2006) A VEGF-A splice variant defective for heparan sulfate and neuropilin-1 binding shows attenuated signaling through VEGFR-2. Cell Mol Life Sci 63(17):2067–2077
- Chang SH, Lu YC, Li X, Hsieh WY, Xiong Y, Ghosh M, Evans T, Elemento O, Hla T (2013) Antagonistic function of the RNA-binding protein HuR and miR-200b in post-transcriptional regulation of vascular endothelial growth factor-a expression and angiogenesis. J Biol Chem 288(7):4908–4921
- Chatterjee S, Pal JK (2009) Role of 5'- and 3'-untranslated regions of mRNAs in human diseases. Biol Cell 101(5):251–262
- Chen L, Xiao H, Wang ZH, Huang Y, Liu ZP, Ren H, Song H (2014) miR-29a suppresses growth and invasion of gastric cancer cells in vitro by targeting VEGF-A. BMB Rep 47(1):39–44
- Chiarini A, Dal Pra I, Pacchiana R, Menapace L, Zumiani G, Zanoni M, Armato U (2006) Comano's (Trentino) thermal water interferes with the expression and secretion of vascular endothelial growth factor-A protein isoforms by cultured human psoriatic keratinocytes: a potential mechanism of its anti-psoriatic action. Int J Mol Med 18(1):17–25
- Claffey KP, Shih SC, Mullen A, Dziennis S, Cusick JL, Abrams KR, Lee SW, Detmar M (1998) Identification of a human VPF/VEGF 3' untranslated region mediating hypoxia-induced mRNA stability. Mol Biol Cell 9(2):469–481
- Coles LS, Bartley MA, Bert A, Hunter J, Polyak S, Diamond P, Vadas MA, Goodall GJ (2004) A multi-protein complex containing cold shock domain (Y-box) and polypyrimidine tract binding proteins forms on the vascular endothelial growth factor mRNA. Potential role in mRNA stabilization. Eur J Biochem 271(3):648–660
- Cui TG, Foster RR, Saleem M, Mathieson PW, Gillatt DA, Bates DO, Harper SJ (2004) Differentiated human podocytes endogenously express an inhibitory isoform of vascular endothelial growth factor (VEGF165b) mRNA and protein. Am J Physiol Renal Physiol 286(4):F767–F773
- de Vries S, Naarmann-de Vries IS, Urlaub H, Lue H, Bernhagen J, Ostareck DH, Ostareck-Lederer A (2013) Identification of DDX6 as a cellular modulator of VEGF expression under hypoxia. J Biol 288(8):5815–5827
- DeMaria CT, Brewer G (1996) AUF1 binding affinity to A+U-rich elements correlates with rapid mRNA degradation. J Biol Chem 271(21):12179–12184
- Dibbens JA, Miller DL, Damert A, Risau W, Vadas MA, Goodall GJ (1999) Hypoxic regulation of vascular endothelial growth factor mRNA stability requires the cooperation of multiple RNA elements. Mol Biol Cell 10(4):907–919
- Dibbens JA, Polyak SW, Damert A, Risau W, Vadas MA, Goodall GJ (2001) Nucleotide sequence of the mouse VEGF 3'UTR and quantitative analysis of sites of polyadenylation. Biochim Biophys Acta 1518(1-2):57–62
- Dor Y, Djonov V, Abramovitch R, Itin A, Fishman GI, Carmeliet P, Goelman G, Keshet E (2002) Conditional switching of VEGF provides new insights into adult neovascularization and pro-angiogenic therapy. EMBO J 21(8):1939–1947

- Dowhan DH, Hong EP, Auboeuf D, Dennis AP, Wilson MM, Berget SM, O'Malley BW (2005) Steroid hormone receptor coactivation and alternative RNA splicing by U2AF65-related proteins CAPERalpha and CAPERbeta. Mol Cell 17(3):429–439
- Drogat B, Kalucka J, Gutierrez L, Hammad H, Goossens S, Farhang Ghahremani M, Bartunkova S, Haigh K, Deswarte K, Nyabi O, Naessens M, Ferrara N, Klingmuller U, Lambrecht BN, Nagy A, Philipsen S, Haigh JJ (2010) VEGF regulates embryonic erythroid development through Gata1 modulation. Blood 116(12):2141–2151
- Elias AP, Dias S (2008) Microenvironment changes (in pH) affect VEGF alternative splicing. Cancer Microenviron 1(1):131–139
- Eremina V, Jefferson JA, Kowalewska J, Hochster H, Haas M, Weisstuch J, Richardson C, Kopp JB, Kabir MG, Backx PH, Gerber HP, Ferrara N, Barisoni L, Alpers CE, Quaggin SE (2008) VEGF inhibition and renal thrombotic microangiopathy. N Engl J Med 358(11):1129–1136
- Eswarappa SM, Potdar AA, Koch WJ, Fan Y, Vasu K, Lindner D, Willard B, Graham LM, DiCorleto PE, Fox PL (2014) Programmed translational readthrough generates antiangiogenic VEGFAx. Cell 19;157(7):1605–18
- Fan XC, Steitz JA (1998) Overexpression of HuR, a nuclear-cytoplasmic shuttling protein, increases the in vivo stability of ARE-containing mRNAs. EMBO J 17(12):3448–3460
- Fan L, Wu Q, Xing X, Wei Y, Shao Z (2012) MicroRNA-145 targets vascular endothelial growth factor and inhibits invasion and metastasis of osteosarcoma cells. Acta Biochim Biophys Sin (Shanghai) 44(5):407–414
- Fellows A, Griffin ME, Petrella BL, Zhong L, Parvin-Nejad FP, Fava R, Morganelli P, Robey RB, Nichols RC (2012) AUF1/hnRNP D represses expression of VEGF in macrophages. Mol Biol Cell 23(8):1414–1422
- Ferrara N (1999) Role of vascular endothelial growth factor in the regulation of angiogenesis. Kidney Int 56(3):794–814
- Ferrara N (2002) VEGF and the quest for tumour angiogenesis factors. Nat Rev Cancer 2(10):795–803
- Ferrara N (2005) The role of VEGF in the regulation of physiological and pathological angiogenesis. EXS 94:209–231
- Ferrara N, Carver-Moore K, Chen H, Dowd M, Lu L, O'Shea KS, Powell-Braxton L, Hillan KJ, Moore MW (1996) Heterozygous embryonic lethality induced by targeted inactivation of the VEGF gene. Nature 380(6573):439–442
- Ferrara N, Gerber HP, LeCouter J (2003) The biology of VEGF and its receptors. Nat Med 9(6):669-676
- Folkman J (1995) Angiogenesis in cancer, vascular, rheumatoid and other disease. Nat Med 1(1):27-31
- Forsythe JA, Jiang BH, Iyer NV, Agani F, Leung SW, Koos RD, Semenza GL (1996) Activation of vascular endothelial growth factor gene transcription by hypoxia-inducible factor 1. Mol Cell Biol 16(9):4604–4613
- Gilbert WV (2010) Alternative ways to think about cellular internal ribosome entry. J Biol Chem 285(38):29033–29038
- Gingerich TJ, Feige JJ, LaMarre J (2004) AU-rich elements and the control of gene expression through regulated mRNA stability. Anim Health Res Rev 5(1):49–63
- Goldberg-Cohen I, Furneauxb H, Levy AP (2002) A 40-bp RNA element that mediates stabilization of vascular endothelial growth factor mRNA by HuR. J Biol Chem 277(16):13635–13640
- Gorgoni B, Gray NK (2004) The roles of cytoplasmic poly(A)-binding proteins in regulating gene expression: a developmental perspective. Brief Funct Genomic Proteomic 3(2):125–141
- Gratacos FM, Brewer G (2010) The role of AUF1 in regulated mRNA decay. Wiley Interdiscip Rev RNA 1(3):457–473
- Grunstein J, Masbad JJ, Hickey R, Giordano F, Johnson RS (2000) Isoforms of vascular endothelial growth factor act in a coordinate fashion to recruit and expand tumor vasculature. Mol Cell Biol 20(19):7282–7291
- Guan X, Zhao H, Niu J, Tang D, Ajani JA, Wei Q (2009) The VEGF -634G>C promoter polymorphism is associated with risk of gastric cancer. BMC Gastroenterol 9:77

- Guo P, Xu L, Pan S, Brekken RA, Yang ST, Whitaker GB, Nagane M, Thorpe PE, Rosenbaum JS, Su Huang HJ, Cavenee WK, Cheng SY (2001) Vascular endothelial growth factor isoforms display distinct activities in promoting tumor angiogenesis at different anatomic sites. Cancer Res 61(23):8569–8577
- Hao Y, Yang J, Yin S, Zhang H, Fan Y, Sun C, Gu J, Xi JJ (2014) The synergistic regulation of VEGF-mediated angiogenesis through miR-190 and target genes. RNA 20(8):1328–1336
- Harris S, Craze M, Newton J, Fisher M, Shima DT, Tozer GM, Kanthou C (2012) Do antiangiogenic VEGF (VEGFxxxb) isoforms exist? A cautionary tale. PLoS One 7(5), e35231
- Hoque M, Ji Z, Zheng D, Luo W, Li W, You B, Park JY, Yehia G, Tian B (2013) Analysis of alternative cleavage and polyadenylation by 3' region extraction and deep sequencing. Nat Methods 10(2):133–139
- Houck KA, Leung DW, Rowland AM, Winer J, Ferrara N (1992) Dual regulation of vascular endothelial growth factor bioavailability by genetic and proteolytic mechanisms. J Biol Chem 267(36):26031–26037
- Hsu CY, Hsieh TH, Tsai CF, Tsai HP, Chen HS, Chang Y, Chuang HY, Lee JN, Hsu YL, Tsai EM (2014) miRNA-199a-5p regulates VEGFA in endometrial mesenchymal stem cells and contributes to the pathogenesis of endometriosis. J Pathol 232(3):330–343
- Hua Z, Lv Q, Ye W, Wong CK, Cai G, Gu D, Ji Y, Zhao C, Wang J, Yang BB, Zhang Y (2006) MiRNA-directed regulation of VEGF and other angiogenic factors under hypoxia. PLoS One 1, e116
- Huez I, Creancier L, Audigier S, Gensac MC, Prats AC, Prats H (1998) Two independent internal ribosome entry sites are involved in translation initiation of vascular endothelial growth factor mRNA. Mol Cell Biol 18(11):6178–6190
- Huez I, Bornes S, Bresson D, Creancier L, Prats H (2001) New vascular endothelial growth factor isoform generated by internal ribosome entry site-driven CUG translation initiation. Mol Endocrinol 15(12):2197–2210
- Ikeda E, Achen MG, Breier G, Risau W (1995) Hypoxia-induced transcriptional activation and increased mRNA stability of vascular endothelial growth factor in C6 glioma cells. J Biol Chem 270(34):19761–19766
- Ingolia NT, Ghaemmaghami S, Newman JR, Weissman JS (2009) Genome-wide analysis in vivo of translation with nucleotide resolution using ribosome profiling. Science 324(5924):218–223
- Ingolia NT, Lareau LF, Weissman JS (2011) Ribosome profiling of mouse embryonic stem cells reveals the complexity and dynamics of mammalian proteomes. Cell 147(4):789–802
- Ingolia NT, Brar GA, Rouskin S, McGeachy AM, Weissman JS (2012) The ribosome profiling strategy for monitoring translation in vivo by deep sequencing of ribosome-protected mRNA fragments. Nat Protoc 7(8):1534–1550
- Jafarifar F, Yao P, Eswarappa SM, Fox PL (2011) Repression of VEGFA by CA-rich elementbinding microRNAs is modulated by hnRNP L. EMBO J 30(7):1324–1334
- Jin Q, Hemminki K, Enquist K, Lenner P, Grzybowska E, Klaes R, Henriksson R, Chen B, Pamula J, Pekala W, Zientek H, Rogozinska-Szczepka J, Utracka-Hutka B, Hallmans G, Forsti A (2005) Vascular endothelial growth factor polymorphisms in relation to breast cancer development and prognosis. Clin Cancer Res 11(10):3647–3653
- Kanitz A, Imig J, Dziunycz PJ, Primorac A, Galgano A, Hofbauer GF, Gerber AP, Detmar M (2012) The expression levels of MicroRNA-361-5p and its target VEGFA are inversely correlated in human cutaneous squamous cell carcinoma. PLoS One 7(11), e49568
- Karaa ZS, Iacovoni JS, Bastide A, Lacazette E, Touriol C, Prats H (2009) The VEGF IRESes are differentially susceptible to translation inhibition by miR-16. RNA 15(2):249–254
- Kawamura H, Li X, Harper SJ, Bates DO, Claesson-Welsh L (2008) Vascular endothelial growth factor (VEGF)-A165b is a weak in vitro agonist for VEGF receptor-2 due to lack of coreceptor binding and deficient regulation of kinase activity. Cancer Res 68(12):4683–4692
- Keck PJ, Hauser SD, Krivi G, Sanzo K, Warren T, Feder J, Connolly DT (1989) Vascular permeability factor, an endothelial cell mitogen related to PDGF. Science 246(4935):1309–1312
- Kim GJ, Georg I, Scherthan H, Merkenschlager M, Guillou F, Scherer G, Barrionuevo F (2009) Dicer is required for Sertoli cell function and survival. Int J Dev Biol 54(5):867–875

- King PH (2000) RNA-binding analyses of HuC and HuD with the VEGF and c-myc 3'-untranslated regions using a novel ELISA-based assay. Nucleic Acids Res 28(7), E20
- Kishor A, Tandukar B, Ly YV, Toth EA, Suarez Y, Brewer G, Wilson GM (2012) Hsp70 is a novel posttranscriptional regulator of gene expression that binds and stabilizes selected mRNAs containing AU-rich elements. Mol Cell Biol 33(1):71–84
- Konopatskaya O, Churchill AJ, Harper SJ, Bates DO, Gardiner TA (2006) VEGF165b, an endogenous C-terminal splice variant of VEGF, inhibits retinal neovascularization in mice. Mol Vis 12:626–632
- Lambrechts D, Storkebaum E, Morimoto M, Del-Favero J, Desmet F, Marklund SL, Wyns S, Thijs V, Andersson J, van Marion I, Al-Chalabi A, Bornes S, Musson R, Hansen V, Beckman L, Adolfsson R, Pall HS, Prats H, Vermeire S, Rutgeerts P, Katayama S, Awata T, Leigh N, Lang-Lazdunski L, Dewerchin M, Shaw C, Moons L, Vlietinck R, Morrison KE, Robberecht W, Van Broeckhoven C, Collen D, Andersen PM, Carmeliet P (2003) VEGF is a modifier of amyo-trophic lateral sclerosis in mice and humans and protects motoneurons against ischemic death. Nat Genet 34(4):383–394
- Larcher F, Murillas R, Bolontrade M, Conti CJ, Jorcano JL (1998) VEGF/VPF overexpression in skin of transgenic mice induces angiogenesis, vascular hyperpermeability and accelerated tumor development. Oncogene 17(3):303–311
- Lee WH, Lee HH, Vo MT, Kim HJ, Ko MS, Im YC, Min YJ, Lee BJ, Cho WJ, Park JW (2011) Casein kinase 2 regulates the mRNA-destabilizing activity of tristetraprolin. J Biol Chem 286(24):21577–21587
- Lei Z, Li B, Yang Z, Fang H, Zhang GM, Feng ZH, Huang B (2009) Regulation of HIF-1alpha and VEGF by miR-20b tunes tumor cells to adapt to the alteration of oxygen concentration. PLoS One 4(10), e7629
- Leung DW, Cachianes G, Kuang WJ, Goeddel DV, Ferrara N (1989) Vascular endothelial growth factor is a secreted angiogenic mitogen. Science 246(4935):1306–1309
- Levine TD, Gao F, King PH, Andrews LG, Keene JD (1993) Hel-N1: an autoimmune RNA-binding protein with specificity for 3' uridylate-rich untranslated regions of growth factor mRNAs. Mol Cell Biol 13(6):3494–3504
- Levy AP (1998) Hypoxic regulation of VEGF mRNA stability by RNA-binding proteins. Trends Cardiovasc Med 8(6):246–250
- Levy AP, Levy NS, Wegner S, Goldberg MA (1995) Transcriptional regulation of the rat vascular endothelial growth factor gene by hypoxia. J Biol Chem 270(22):13333–13340
- Levy AP, Levy NS, Goldberg MA (1996) Post-transcriptional regulation of vascular endothelial growth factor by hypoxia. J Biol Chem 271(5):2746–2753
- Levy NS, Chung S, Furneaux H, Levy AP (1998) Hypoxic stabilization of vascular endothelial growth factor mRNA by the RNA-binding protein HuR. J Biol Chem 273(11):6417–6423
- Liu B, Peng XC, Zheng XL, Wang J, Qin YW (2009) MiR-126 restoration down-regulate VEGF and inhibit the growth of lung cancer cell lines in vitro and in vivo. Lung Cancer 66(2):169–175
- Long J, Wang Y, Wang W, Chang BH, Danesh FR (2010) Identification of microRNA-93 as a novel regulator of vascular endothelial growth factor in hyperglycemic conditions. J Biol Chem 285(30):23457–23465
- Ma S, Musa T, Bag J (2006) Reduced stability of mitogen-activated protein kinase kinase-2 mRNA and phosphorylation of poly(A)-binding protein (PABP) in cells overexpressing PABP. J Biol Chem 281(6):3145–3156
- Mattot V, Moons L, Lupu F, Chernavvsky D, Gomez RA, Collen D, Carmeliet P (2002) Loss of the VEGF(164) and VEGF(188) isoforms impairs postnatal glomerular angiogenesis and renal arteriogenesis in mice. J Am Soc Nephrol 13(6):1548–1560
- McArthur K, Feng B, Wu Y, Chen S, Chakrabarti S (2011) MicroRNA-200b regulates vascular endothelial growth factor-mediated alterations in diabetic retinopathy. Diabetes 60(4): 1314–1323
- McFee RM, Rozell TG, Cupp AS (2012) The balance of proangiogenic and antiangiogenic VEGFA isoforms regulate follicle development. Cell Tissue Res 349(3):635–647

- Meiron M, Anunu R, Scheinman EJ, Hashmueli S, Levi BZ (2001) New isoforms of VEGF are translated from alternative initiation CUG codons located in its 5'UTR. Biochem Biophys Res Commun 282(4):1053–1060
- Melo SA, Kalluri R (2013) miR-29b moulds the tumour microenvironment to repress metastasis. Nat Cell Biol 15(2):139–140
- Merdzhanova G, Gout S, Keramidas M, Edmond V, Coll JL, Brambilla C, Brambilla E, Gazzeri S, Eymin B (2010) The transcription factor E2F1 and the SR protein SC35 control the ratio of pro-angiogenic versus antiangiogenic isoforms of vascular endothelial growth factor-A to inhibit neovascularization in vivo. Oncogene 29(39):5392–5403
- Miller DL, Dibbens JA, Damert A, Risau W, Vadas MA, Goodall GJ (1998) The vascular endothelial growth factor mRNA contains an internal ribosome entry site. FEBS Lett 434(3):417–420
- Miller-Kasprzak E, Jagodzinski PP (2008) 5-Aza-2'-deoxycytidine increases the expression of anti-angiogenic vascular endothelial growth factor 189b variant in human lung microvascular endothelial cells. Biomed Pharmacother 62(3):158–163
- Millevoi S, Moine H, Vagner S (2012) G-quadruplexes in RNA biology. Wiley Interdiscip Rev RNA 3(4):495–507
- Miquerol L, Langille BL, Nagy A (2000) Embryonic development is disrupted by modest increases in vascular endothelial growth factor gene expression. Development 127(18):3941–3946
- Mitchell P, Tollervey D (2000) mRNA stability in eukaryotes. Curr Opin Genet Dev 10(2):193–198
- Mokrejs M, Masek T, Vopalensky V, Hlubucek P, Delbos P, Pospisek M (2010) IRESite--a tool for the examination of viral and cellular internal ribosome entry sites. Nucleic Acids Res 38(Database issue):D131–D136
- Morris MJ, Negishi Y, Pazsint C, Schonhoft JD, Basu S (2010) An RNA G-quadruplex is essential for cap-independent translation initiation in human VEGF IRES. J Am Chem Soc 132(50):17831–17839
- Ng YS, Rohan R, Sunday ME, Demello DE, D'Amore PA (2001) Differential expression of VEGF isoforms in mouse during development and in the adult. Dev Dyn 220(2):112–121
- Nowak DG, Woolard J, Amin EM, Konopatskaya O, Saleem MA, Churchill AJ, Ladomery MR, Harper SJ, Bates DO (2008) Expression of pro- and anti-angiogenic isoforms of VEGF is differentially regulated by splicing and growth factors. J Cell Sci 121(Pt 20):3487–3495
- Ohno-Matsui K, Hirose A, Yamamoto S, Saikia J, Okamoto N, Gehlbach P, Duh EJ, Hackett S, Chang M, Bok D, Zack DJ, Campochiaro PA (2002) Inducible expression of vascular endothelial growth factor in adult mice causes severe proliferative retinopathy and retinal detachment. Am J Pathol 160(2):711–719
- Oltean S, Neal CR, Mavrou A, Patel P, Ahad T, Alsop C, Lee T, Sison K, Qiu Y, Harper SJ, Bates DO, Salmon AH (2012) VEGF165b overexpression restores normal glomerular water permeability in VEGF164-overexpressing adult mice. Am J Physiol Renal Physiol 303(7):F1026–F1036
- Onesto C, Berra E, Grepin R, Pages G (2004) Poly(A)-binding protein-interacting protein 2, a strong regulator of vascular endothelial growth factor mRNA. J Biol Chem 279(33):34217–34226
- Onesto C, Hannoun-Levi JM, Chamorey E, Formento JL, Ramaioli A, Pages G (2006) Vascular endothelial growth factor-A and Poly(A) binding protein-interacting protein 2 expression in human head and neck carcinomas: correlation and prognostic significance. Br J Cancer 94(10):1516–1523
- Pages G, Pouyssegur J (2005) Transcriptional regulation of the vascular endothelial growth factor gene--a concert of activating factors. Cardiovasc Res 65(3):564–573
- Pan Q, Shai O, Lee LJ, Frey BJ, Blencowe BJ (2008) Deep surveying of alternative splicing complexity in the human transcriptome by high-throughput sequencing. Nat Genet 40(12):1413–1415
- Peng SS, Chen CY, Xu N, Shyu AB (1998) RNA stabilization by the AU-rich element binding protein, HuR, an ELAV protein. EMBO J 17(12):3461–3470
- Perrin RM, Konopatskaya O, Qiu Y, Harper S, Bates DO, Churchill AJ (2005) Diabetic retinopathy is associated with a switch in splicing from anti- to pro-angiogenic isoforms of vascular endothelial growth factor. Diabetologia 48(11):2422–2427

- Pritchard-Jones RO, Dunn DB, Qiu Y, Varey AH, Orlando A, Rigby H, Harper SJ, Bates DO (2007) Expression of VEGF(xxx)b, the inhibitory isoforms of VEGF, in malignant melanoma. Br J Cancer 97(2):223–230
- Qiu Y, Bevan H, Weeraperuma S, Wratting D, Murphy D, Neal CR, Bates DO, Harper SJ (2008) Mammary alveolar development during lactation is inhibited by the endogenous antiangiogenic growth factor isoform, VEGF165b. FASEB J 22(4):1104–1112
- Ray PS, Fox PL (2007) A post-transcriptional pathway represses monocyte VEGF-A expression and angiogenic activity. EMBO J 26(14):3360–3372
- Ray PS, Jia J, Yao P, Majumder M, Hatzoglou M, Fox PL (2009) A stress-responsive RNA switch regulates VEGFA expression. Nature 457(7231):915–919
- Rennel E, Waine E, Guan H, Schuler Y, Leenders W, Woolard J, Sugiono M, Gillatt D, Kleinerman E, Bates D, Harper S (2008) The endogenous anti-angiogenic VEGF isoform, VEGF165b inhibits human tumour growth in mice. Br J Cancer 98(7):1250–1257
- Rosenbaum-Dekel Y, Fuchs A, Yakirevich E, Azriel A, Mazareb S, Resnick MB, Levi BZ (2005) Nuclear localization of long-VEGF is associated with hypoxia and tumor angiogenesis. Biochem Biophys Res Commun 332(1):271–278
- Ryan HE, Poloni M, McNulty W, Elson D, Gassmann M, Arbeit JM, Johnson RS (2000) Hypoxiainducible factor-1alpha is a positive factor in solid tumor growth. Cancer Res 60(15):4010–4015
- Sfar S, Hassen E, Saad H, Mosbah F, Chouchane L (2006) Association of VEGF genetic polymorphisms with prostate carcinoma risk and clinical outcome. Cytokine 35(1-2):21–28
- Shatsky IN, Dmitriev SE, Terenin IM, Andreev DE (2010) Cap- and IRES-independent scanning mechanism of translation initiation as an alternative to the concept of cellular IRESs. Mol Cells 30(4):285–293
- Shih SC, Claffey KP (1999) Regulation of human vascular endothelial growth factor mRNA stability in hypoxia by heterogeneous nuclear ribonucleoprotein L. J Biol Chem 274(3):1359–1365
- Shim J, Karin M (2002) The control of mRNA stability in response to extracellular stimuli. Mol Cells 14(3):323–331
- Shima DT, Deutsch U, D'Amore PA (1995) Hypoxic induction of vascular endothelial growth factor (VEGF) in human epithelial cells is mediated by increases in mRNA stability. FEBS Lett 370(3):203–208
- Shima DT, Kuroki M, Deutsch U, Ng YS, Adamis AP, D'Amore PA (1996) The mouse gene for vascular endothelial growth factor. Genomic structure, definition of the transcriptional unit, and characterization of transcriptional and post-transcriptional regulatory sequences. J Biol Chem 271(7):3877–3883
- Simone LE, Keene JD (2013) Mechanisms coordinating ELAV/Hu mRNA regulons. Curr Opin Genet Dev 23(1):35–43
- Sivaskandarajah GA, Jeansson M, Maezawa Y, Eremina V, Baelde HJ, Quaggin SE (2011) Vegfa protects the glomerular microvasculature in diabetes. Diabetes 61(11):2958–2966
- Spriggs KA, Stoneley M, Bushell M, Willis AE (2008) Re-programming of translation following cell stress allows IRES-mediated translation to predominate. Biol Cell 100(1):27–38
- Spriggs KA, Bushell M, Willis AE (2010) Translational regulation of gene expression during conditions of cell stress. Mol Cell 40(2):228–237
- Stalmans I, Ng YS, Rohan R, Fruttiger M, Bouche A, Yuce A, Fujisawa H, Hermans B, Shani M, Jansen S, Hicklin D, Anderson DJ, Gardiner T, Hammes HP, Moons L, Dewerchin M, Collen D, Carmeliet P, D'Amore PA (2002) Arteriolar and venular patterning in retinas of mice selectively expressing VEGF isoforms. J Clin Invest 109(3):327–336
- Stalmans I, Lambrechts D, De Smet F, Jansen S, Wang J, Maity S, Kneer P, von der Ohe M, Swillen A, Maes C, Gewillig M, Molin DG, Hellings P, Boetel T, Haardt M, Compernolle V, Dewerchin M, Plaisance S, Vlietinck R, Emanuel B, Gittenberger-de Groot AC, Scambler P, Morrow B, Driscol DA, Moons L, Esguerra CV, Carmeliet G, Behn-Krappa A, Devriendt K, Collen D, Conway SJ, Carmeliet P (2003) VEGF: a modifier of the del22q11 (DiGeorge) syndrome? Nat Med 9(2):173–182
- Starck SR, Jiang V, Pavon-Eternod M, Prasad S, McCarthy B, Pan T, Shastri N (2012) LeucinetRNA initiates at CUG start codons for protein synthesis and presentation by MHC class I. Science 336(6089):1719–1723

- Stoecklin G, Gross B, Ming XF, Moroni C (2003) A novel mechanism of tumor suppression by destabilizing AU-rich growth factor mRNA. Oncogene 22(23):3554–3561
- Sun CY, She XM, Qin Y, Chu ZB, Chen L, Ai LS, Zhang L, Hu Y (2013) miR-15a and miR-16 affect the angiogenesis of multiple myeloma by targeting VEGF. Carcinogenesis 34(2): 426–435
- Tee MK, Jaffe RB (2001) A precursor form of vascular endothelial growth factor arises by initiation from an upstream in-frame CUG codon. Biochem J 359(Pt 1):219–226
- Tischer E, Mitchell R, Hartman T, Silva M, Gospodarowicz D, Fiddes JC, Abraham JA (1991) The human gene for vascular endothelial growth factor. Multiple protein forms are encoded through alternative exon splicing. J Biol Chem 266(18):11947–11954
- Touriol C, Bornes S, Bonnal S, Audigier S, Prats H, Prats AC, Vagner S (2003) Generation of protein isoform diversity by alternative initiation of translation at non-AUG codons. Biol Cell 95(3-4):169–178
- Varey AH, Rennel ES, Qiu Y, Bevan HS, Perrin RM, Raffy S, Dixon AR, Paraskeva C, Zaccheo O, Hassan AB, Harper SJ, Bates DO (2008) VEGF 165 b, an antiangiogenic VEGF-A isoform, binds and inhibits bevacizumab treatment in experimental colorectal carcinoma: balance of pro- and antiangiogenic VEGF-A isoforms has implications for therapy. Br J Cancer 98(8):1366–1379
- Vempati P, Popel AS, Mac Gabhann F (2011) Formation of VEGF isoform-specific spatial distributions governing angiogenesis: computational analysis. BMC Syst Biol 5:59
- Vincenti V, Cassano C, Rocchi M, Persico G (1996) Assignment of the vascular endothelial growth factor gene to human chromosome 6p21.3. Circulation 93(8):1493–1495
- Vumbaca F, Phoenix KN, Rodriguez-Pinto D, Han DK, Claffey KP (2008) Double-stranded RNA-binding protein regulates vascular endothelial growth factor mRNA stability, translation, and breast cancer angiogenesis. Mol Cell Biol 28(2):772–783
- Wang R, Crystal RG, Hackett NR (2009) Identification of an exonic splicing silencer in exon 6A of the human VEGF gene. BMC Mol Biol 10:103
- Wang R, Zhao N, Li S, Fang JH, Chen MX, Yang J, Jia WH, Yuan Y, Zhuang SM (2013) MicroRNA-195 suppresses angiogenesis and metastasis of hepatocellular carcinoma by inhibiting the expression of VEGF, VAV2, and CDC42. Hepatology 58(2):642–653
- Wen Z, Huang W, Feng Y, Cai W, Wang Y, Wang X, Liang J, Wani M, Chen J, Zhu P, Chen JM, Millard RW, Fan GC (2014) MicroRNA-377 regulates mesenchymal stem cell-induced angiogenesis in ischemic hearts by targeting VEGF. PLoS One 9(9), e104666
- Woolard J, Wang WY, Bevan HS, Qiu Y, Morbidelli L, Pritchard-Jones RO, Cui TG, Sugiono M, Waine E, Perrin R, Foster R, Digby-Bell J, Shields JD, Whittles CE, Mushens RE, Gillatt DA, Ziche M, Harper SJ, Bates DO (2004) VEGF165b, an inhibitory vascular endothelial growth factor splice variant: mechanism of action, in vivo effect on angiogenesis and endogenous protein expression. Cancer Res 64(21):7822–7835
- Woolard J, Bevan HS, Harper SJ, Bates DO (2009) Molecular diversity of VEGF-A as a regulator of its biological activity. Microcirculation 16(7):572–592
- Yang WJ, Yang DD, Na S, Sandusky GE, Zhang Q, Zhao G (2005) Dicer is required for embryonic angiogenesis during mouse development. J Biol Chem 280(10):9330–9335
- Yue X, Wang P, Xu J, Zhu Y, Sun G, Pang Q, Tao R (2012) MicroRNA-205 functions as a tumor suppressor in human glioblastoma cells by targeting VEGF-A. Oncol Rep 27(4):1200–1206
- Zhang T, Kruys V, Huez G, Gueydan C (2002) AU-rich element-mediated translational control: complexity and multiple activities of trans-activating factors. Biochem Soc Trans 30(Pt 6): 952–958
- Zhou S, Gu L, He J, Zhang H, Zhou M (2011) MDM2 regulates vascular endothelial growth factor mRNA stabilization in hypoxia. Mol Cell Biol 31(24):4928–4937
- Zhou Z, Ye H, Cui K, Chen X, Liu Y (2012) A probable novel splicing isoform of human vascular endothelial growth factor. Nan Fang Yi Ke Da Xue Xue Bao 32(6):755–759
- Zou C, Xu Q, Mao F, Li D, Bian C, Liu LZ, Jiang Y, Chen X, Qi Y, Zhang X, Wang X, Sun Q, Kung HF, Lin MC, Dress A, Wardle F, Jiang BH, Lai L (2012) MiR-145 inhibits tumor angiogenesis and growth by N-RAS and VEGF. Cell Cycle 11(11):2137–2145

Post-transcriptional Regulation of Prostaglandin Biosynthesis

9

Fernando F. Blanco, Noémie Legrand*, Cyril Sobolewski*, and Dan A. Dixon

1 General Aspects of Prostanoid Synthesis

Prostaglandins represent a class of potent bioactive lipid mediators derived from arachidonic acid (AA), a 20-carbon tetraenoic fatty acid (C20:4 ω 6). Initially isolated from prostate, semen and seminal vesicles (Horton and Thompson 1964; Samuelsson 1963), the biosynthesis of prostanoids involves several enzymes including phospholipases, cyclooxygenases and prostaglandins synthases. The clinical relevance of prostaglandins are apparent in their regulation of homeostatic functions including reproductive physiology (Salleh 2014), whereas alteration of prostaglandin homeostasis is often associated with pathologies associated with inflammatory syndromes and cancer (Ricciotti and FitzGerald 2011; Wang and Dubois 2010).

F.F. Blanco

Department of Cellular Physiology and Metabolism, Geneva Medical Faculty, Geneva, Switzerland

D.A. Dixon (⊠)
Department of Cancer Biology, University of Kansas Medical Center, 3901 Rainbow Blvd., Kansas City, KS 66610, USA
e-mail: ddixon3@kumc.edu

© Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_9

^{*}Equal contribution.

Department of Cancer Biology, University of Kansas Medical Center, 3901 Rainbow Blvd., Kansas City, KS 66610, USA

Department of Pharmacology and Experimental Therapeutics, Thomas Jefferson University, Philadelphia, PA, USA

N. Legrand Department of Microbiology and Molecular Medicine, Geneva Medical Faculty, Geneva, Switzerland

C. Sobolewski Department of Cancer Biology, University of Kansas Medical Center, 3901 Rainbow Blvd., Kansas City, KS 66610, USA

1.1 Phospholipases

Phospholipases represent a family of enzymes catalyzing the hydrolysis of phospholipids, especially glycerophospholipids (e.g., phosphatidylcholine), which contain glycerol, saturated fatty acid at sn-1 position and unsaturated fatty acid in the sn-2 position. Hydrolysis at the sn-2 position by members of the phospholipase A_2 (PLA₂) family, of which the Ca⁺²-dependent cytosolic PLA₂ (cPLA₂) plays a dominant role, enables AA release from the cell membrane (Park et al. 2006; Simmons et al. 2004; Betz and Hansch 1984; Pniewska and Pawliczak 2013). Several isoforms of PLA₂ can be distinguished depending on their subcellular localization, such the secreted sPLA₂, cytosolic PLA₂ (cPLA₂), and lipoprotein-associated PLA₂ (lp-PLA₂) (Murakami and Kudo 2002; Hiraoka et al. 2005). The expression of PLA₂ is induced by pro-inflammatory cytokines through the activity of several transcription factors such as NF κ B, PPAR, and C/EBP (Pfeilschifter et al. 1993; Touqui and Alaoui-El-Azher 2001). Additionally, Phospholipase C (PLC) can influence AA levels through metabolism of diacylglycerol (Rebecchi and Pentyala 2000; Tang et al. 2006; Pniewska and Pawliczak 2013).

1.2 Cyclooxygenases

Following its release, arachidonic acid is converted to PGH_2 by cyclooxygenase (COX) enzymes that are located at the luminal side of the endoplasmic reticulum and nuclear membrane (Chandrasekharan and Simmons 2004). COXs serve as bi-functional enzymes converting AA into prostaglandin G_2 (PGG₂) through cyclooxygenase activity, followed by synthesis of prostaglandin H_2 (PGH₂) via a peroxidase activity (Smith et al. 2000; Simmons et al. 2004; Chandrasekharan and Simmons 2004).

Two main isoforms of COXs have been identified. Cyclooxygenase-1 (COX-1) is constitutively expressed in most cell types and modulates several biological processes necessary for maintaining physiological homeostasis (Smith et al. 2000; Simmons et al. 2004; Chandrasekharan and Simmons 2004). Cyclooxygenase-2 (COX-2) represents the inducible isoform of COX, whose expression is induced by various growth factors as well as pro-inflammatory cytokines such as IL-1 β , IL-6, and TNF α (Ramsay et al. 2003). The promoter of COX-2 contains an NF κ B response element as well as other cytokine-dependent response elements (Ramsay et al. 2003; Chandrasekharan and Simmons 2004). COX-2 protein displays 60 % homology with COX-1 (Hinz and Brune 2002), however possesses a C-terminal "pocket", which is preferentially targeted by COX-2 specific inhibitors (Kurumbail et al. 1996). While most noted for its role in various patho-physiological functions, such as inflammation and cancer (Ricciotti and FitzGerald 2011; Wang and Dubois 2010; Ristimaki 2004), COX-2 plays a prominent role in the various reproduction stages ranging from ovulation to implantation, to decidualization and subsequent delivery (Salleh 2014; Chan 2004). Within the vascular, constitutive low-level expression of COX-2 contributes to the continuous generation of vasoprotective PGI_2 in endothelial cells (characterized by a low hydroperoxide tone) constantly exposed to mechanical forces resulting from steady physiological blood flow (Funk and FitzGerald 2007; McAdam et al. 1999; Inoue et al. 2002).

1.3 Prostaglandin Synthases and Receptors

Prostanoids (prostaglandins and thromboxanes) are synthesized from PGH₂ by various prostaglandin synthases. Following their synthesis, prostaglandins can act in an autocrine and/or paracrine manner through binding of specific membrane or nuclear receptors (Narumiya and FitzGerald 2001). Listed below are the major physiological prostanoids, along with a description of their respective synthases and receptors. Figure 9.1 illustrates the synthesis and signaling of the predominant prostaglandin PGE₂.

1.3.1 PGE₂

PGE₂ is the most abundant prostaglandin associated with pathologies involving inflammation, pain sensitivity and fever, and neoplasia (Park et al. 2006; Kawahara et al. 2014; Wang and Dubois 2010). PGE₂ also promotes gastric mucosa protection, renal hemodynamics, and stimulates ovulation and myometrium contraction during parturition (Takeuchi 2010; Stouffer et al. 2007; Yount and Lassiter 2013). Three PGE₂ synthases (PGES), cytosolic cPGES and two membrane-associated PGES, mPGES-1 and the microsomal mPGES-2, produce PGE₂ by using endoperoxide PGH₂ as a substrate (Regan 2003). mPGES-2 and cPGES are constitutively expressed, whereas mPGES-1 is induced by pro-inflammatory cytokines (Kudo and Murakami 2005; Ricciotti and FitzGerald 2011; de Oliveira et al. 2008). cPGES interacts with COX-1 and preferentially converts COX-1-dependent PGH₂, whereas mPGES-1 is the major enzyme involved in PGE₂ production from COX-2 derived PGH₂ (Regan 2003; Kudo and Murakami 2005). mPGES-2 is produced as a golgi membrane-associated protein and is associated with both COX-1 and COX-2 (Kudo and Murakami 2005).

PGE₂ signaling is mediated *via* the four membrane receptors EP1, EP2, EP3, and EP4 encoded by PTGER1, PTGER2, PTGER3 and PTGER4 genes, respectively (Funk 2001; Hull et al. 2004; Breyer et al. 2001; Sugimoto and Narumiya 2007). EP receptors are G protein-coupled receptors primarily localized to the cell surface, although evidence exists indicating nuclear membrane localization (Bhattacharya et al. 1998; Breyer and Breyer 2000; Funk 2001). EP1 is a G α_q -coupled receptor coupled phospholipase C/inositol triphosphate signaling and free Ca²⁺ mobilization (Breyer et al. 1996). EP2 and EP4 are G α_s -coupled receptors that stimulate adenylate cyclase to produce cAMP, which in turn activates kinases such as protein kinase A (Hull et al. 2004; Regan 2003). EP3 receptor consists of multiple splice variants and has been shown to couple to both G α_i and G α_s proteins, leading to reduced and increased cAMP levels, respectively (Namba et al. 1993; Kotani et al. 1997; Hatae et al. 2002).



Fig. 9.1 Overview of the PGE₂ synthesis pathway. Arachidonic acid is liberated from phospholipids by phospholipase A_2 and phospholipase C activity. Free arachidonic acid can then be converted into the intermediate PGH₂ by either isoform of the cyclooxygenase enzymes, COX-1 and COX-2. PGE synthases cPGES, mPGES-1, and mPGES-2, convert PGH₂ into PGE₂ that signals through the G protein-coupled receptors EP1, EP2, EP3, and EP4 in both an autocrine and paracrine manner

The action of PGE_2 in a tissue-specific physiology predominantly depends on the cell-specific EP receptor expression and levels of the four receptor subtypes (Breyer et al. 2001). EP3 and EP4 receptors are the most widely expressed subtypes, observed in all tissues examined (Sugimoto and Narumiya 2007). Based on observations using pharmacological antagonists and genetic knockout studies of EP receptors, it is of considerable interest to further explore their post-transcriptional regulation in order to control their expression levels (Kawamori et al. 2001; Yang et al. 2006; Keith et al. 2006).

1.3.2 PGI₂

PGI₂ (prostacyclin) is produced by PGI₂ synthase (PGIS) primarily in vascular endothelial cells and smooth muscle cells for maintenance of cardiovascular homeostasis (Dorris and Peebles 2012; Ricciotti and FitzGerald 2011). The biological effects of PGI₂ are mediated through the $G\alpha_s$ -coupled IP receptor, whose expression has been found in platelets, heart, aorta, lung, kidney, and liver (Dorris and Peebles 2012). Similar to EP2 and EP4 receptors, the IP receptor promotes activation of adenvlate cyclase and PKA (Dorris and Peebles 2012). PGI_2 is also involved in physiological functions such as gastric mucosa protection, smooth muscle relaxation, and vasodilatation (Williams et al. 1994; Harada et al. 1999; Tanaka et al. 2004). Moreover, PGI₂ is a potent inhibitor of platelet aggregation, leukocyte adhesion, and vascular smooth muscle cells proliferation (Dorris and Peebles 2012). In cancer cells, PGI_2 stimulates angiogenesis, through IP receptor expressed at the surface of tumor endothe lial cells (Osawa et al. 2012). It has also been shown that PGI_2 can signal through the nuclear receptor PPAR δ (peroxisome proliferator-activated nuclear receptor) and PGI₂-induced PPAR δ activation can regulate arterial thrombus formation, blastocyst development, and limits inflammation (Kang et al. 2011; Barbieri et al. 2012; Chen et al. 2009; Dorris and Peebles 2012).

1.3.3 PGD₂

 PGD_2 is mainly synthesized in the CNS, mast cells, dendritic cells, macrophages, and Th2 lymphocytes (Ricciotti and FitzGerald 2011). PGD₂ is implicated in platelet aggregation, smooth muscle contraction, broncho-constriction, immune cells chemotaxis, and regulation of sleep and wake cycles (Urade and Hayaishi 2011; Kanaoka and Urade 2003). PGD₂ is synthesized by two enzymes PTGDS and HPGDS. PTGDS is mainly expressed in brain tissues, male genital organs and in the heart (Urade and Havaishi 2000), where HPGDS is hematopoietic in origin (Kanaoka and Urade 2003). PGD₂ plays a pro-inflammatory role through binding to the DP receptors DP1 and DP2 encoded by PTGDR1 and PTGDR2 genes, respectively. DP1 is a $G\alpha_s$ coupled receptor ubiquitously expressed and the activation of this receptor leads the activation of adenylate cyclase and PKA (Narumiya et al. 1999). This pathway is implicated in sex determination via activation and nuclear translocation of SOX-9 (Malki et al. 2005). DP2 is a $G\alpha_i$ -coupled receptor, which enhances intracellular calcium and decreases cAMP leading to the migration of Th2 cells, eosinophils, and basophils (Hirai et al. 2001). PGD₂ released by mast cells is also implicated in asthma (Matsuoka et al. 2000) and possess tumor suppressor activities (Wu et al. 2012).

1.3.4 PGF₂α

 $PGF_{2\alpha}$ is implicated in reproduction by stimulating ovulation, luteolysis, contraction of uterine smooth muscle, and initiation of parturition (Yount and Lassiter 2013; Sugimoto et al. 1997; Saito et al. 2003; De Rensis et al. 2012). $PGF_{2\alpha}$ plays also a

role in renal salt transport, myocardial dysfunction, brain damage, and mechanical allodynia (Breyer and Breyer 2001; Kunori et al. 2009; Takayama et al. 2005; Saleem et al. 2009). PGF_{2α} is generated by PGF synthase and mediates its effects through FP receptors. Two variants of FP receptors have been described, FP_a and FP_b, and both of them are associated with a G α_q protein, which activates phospholipase C and leads to an increase of intracellular Ca⁺² and PKC activation (Ito et al. 1994; Bos et al. 2004). PGF_{2α} is produced in human myometrium and ovary (Matsumoto et al. 1997; Sugimoto et al. 1997) and interestingly, its expression is stimulated by oxytocin in endometrial cells (Asselin et al. 1997).

1.3.5 TXA₂

TXA₂ (thromboxane) produced by TXA₂ synthase is a crucial mediator of platelet adhesion and aggregation, smooth muscle constriction, and activation of endothelial inflammatory response that promotes vasoconstriction (Bos et al. 2004; Ricciotti and FitzGerald 2011). TXA₂ is preferentially expressed in platelets but can also be expressed in macrophages (Bos et al. 2004; Ricciotti and FitzGerald 2011). Secreted TXA₂ mediates its effects through the binding to a membrane Gaq-coupled receptor named "TP receptor", which is mainly expressed in platelets. Once activated, this Gaq protein activates phospholipase C, which in turn promotes Ca⁺² release and PKC activation, leading to platelet aggregation. The production of TXA₂ is implicated in several cardiovascular patho-physiological processes including myocardial infarction, atherosclerosis, thrombosis, and stroke (Bos et al. 2004; Ricciotti and FitzGerald 2011).

1.4 Prostaglandin Transporter and 15-Hydroxyprostaglandin Dehydrogenase

The release of prostaglandins can be mediated by simple diffusion across the cell membrane, but recent findings have suggested that specific carriers are necessary for their transport. The prostaglandin transporter (PGT) is a broadly expressed transporter, which possess the ability to uptake various prostaglandins (i.e. PGE_2 , $PGF_{2\alpha}$ and PGD_2) from the extracellular milieu (Kanai et al. 1995; Schuster 2002). Moreover, this uptake is a prerequisite for the inactivation of PGE_2 and $PGF_{2\alpha}$ in the cytosol, where 15-hydroxyprostaglandin dehydrogenase (15-PGDH) converts these prostaglandins into their stable 13,14-dihydro-15-keto-PGE₂ (PGEM) and 13,14-dihydro-15-keto-PGF_{2\alpha} forms (Holla et al. 2008; Wang and Dubois 2010). Non-enzymatic metabolism is attributed to turnover of other prostaglandins (Wang and DuBois 2007). Interestingly, it has been reported that 15-PGDH is reduced in human colorectal and gastric cancer, thus increasing the amount of PGE_2 in the tumor microenvironment and favors immune escape, cancer cell proliferation, and survival (Backlund et al. 2008; Yan et al. 2004).

2 Post-transcriptional Regulation of Prostaglandin Synthesis: Mechanisms of Messenger RNA Decay

Messenger RNA turnover is a highly regulated cellular process that can occur in a rapid, acute manner in response to intracellular and extracellular signals (Garneau et al. 2007). The necessity of post-transcriptional regulation is apparent as 40-50 %of global gene expression changes in response to cellular signals occur at the level of mRNA stability (Balagopal et al. 2012; Fan et al. 2002; Cheadle et al. 2005). Following transcription, eukaryotic mRNAs undergo splicing, RNA editing, and the addition of the 5' 7-methylguanosine cap and 3' poly(A) tail (Lutz and Cornett 2013). The 5' cap and 3' poly(A) tail, along with other transcript-specific, *cis*-acting regulatory elements, associate with RNA-binding proteins in order to promote mRNA nuclear export and translation initiation (Day and Tuite 1998; Garneau et al. 2007). Cytoplasmic mRNA decay initiates with the shortening of the poly(A) tail by a complex of enzymes known as mRNA deadenylases, with mammalian cells utilizing three major deadenylation complexes, the Ccr4/Caf1/Not (Caf1) complex, the poly A-specific ribonuclease (PARN) complex, and the Pan2/Pan3 complex (Sanduja et al. 2012; Chen and Shyu 2011; Fabian et al. 2013; Funakoshi et al. 2007; Goldstrohm and Wickens 2008). At this point, degradation initiating at the 5' end involves removal of the 5'-7-methyl guanosine cap by the decapping complex Dcp1/ Dcp2, leaving the mRNA body susceptible to degradation by the 5'-3' exonuclease Xrn1 (Garneau et al. 2007; Schoenberg and Maquat 2012; Balagopal et al. 2012). Alternatively, the mRNA can be degraded by 3'-to-5' exonucleolytic degradation through a complex of exonucleases known as the exosome (Garneau et al. 2007; Balagopal et al. 2012). Many mRNAs targeted for degradation are localized to processing (P)-bodies, which are small cytoplasmic foci that contain components of the 3'-to-5' and 5'-to-3' decay machinery along with factors involved in microRNA (miRNA) silencing, nonsense-mediated decay (NMD), and translational silencing (Eulalio et al. 2007; Kulkarni et al. 2010; Zheng et al. 2011; Blanco et al. 2014). Recent evidence demonstrating COX-2 mRNA localization occurring at P-bodies in a signal-dependent manner illustrates the functional significance these discrete cytoplasmic RNA granules have upon prostaglandin biosynthesis (Blanco et al. 2014). The pathways illustrating general cytoplasmic mRNA decay are shown in Fig. 9.2. This section will focus on the functional contribution of regulatory elements within the mRNA 3' untranslated region (3'UTR) as well as the role of RNA-binding proteins, miRNAs, and P-bodies upon in prostaglandin biosynthesis.

2.1 Post-transcriptional Regulation and AU-Rich 3'UTR Elements

The mRNA 3'UTR is a critical region of the transcript that mediates the interaction with RNA-binding proteins and miRNAs. Considerable evidence exists to support the notion that highly conserved 3'UTR elements play a critical role in post-transcriptional regulation of prostaglandin biosynthesis, with a majority of studies



Fig. 9.2 Cytoplasmic mRNA decay pathways. A mRNA transcript undergoes nuclear splicing, and addition of a 5' 7-methylguanosine cap (m^7G) and 3' poly(A) tail, is exported to the cytoplasm. Degradation initiated through poly(A) shortening by the deadenylase complex is the rate-limiting step of mRNA decay. Once partially deadenylated to 10–15 adenosines, the mRNA transcript can be sequestered to P-bodies wherein mRNA decapping and subsequent Xrn-1-mediated decay occurs. Alternatively, mRNA transcripts may be degraded from the 3' end by a complex of exonucleases that comprise the exosome

pertaining to regulation of COX-2 expression (Raz et al. 1989; Bailey and Verma 1990; Dixon 2003; Moore et al. 2011).

A commonly observed, well-characterized 3'UTR feature observed in many inflammatory cytokines, growth factors, and proto-oncogenes is the AU-rich element, or ARE (Barreau et al. 2005; Beisang and Bohjanen 2012). The presence of one or several AREs in the mRNA 3'UTR characterizes the short-lived lifespan of many immediate-early response genes and the importance of this particular RNA element is evident, since estimates ranging from 5 to 8 % and 16 % of coding

genes contain a 3'UTR ARE sequence (Bakheet et al. 2006; Gruber et al. 2010). The functional ARE is characterized by the 3'UTR consensus sequence AUUUA and is most often composed of multiple AUUUA motif copies. AREs are organized into several classes and clusters on the basis of the number and context of the AUUUA pentamer. For example, various immediate-early response genes (e.g. proto-oncogenes) have scattered repeats of the AUUUA motif, whereas mRNAs encoding inflammatory mediators and cytokines (e.g. COX-2) that have multiple repeats of the AUUUA pentamer clustered together (Beisang and Bohjanen 2012; Bakheet et al. 2006). The use of search engines such as AREsite (http://rna.tbi.univie.ac.at/cgi-bin/AREsite.cgi) and ARED (ARE Database; http://brp.kfshrc.edu.sa/ ARED/) have provided investigators online tools to identify the presence of AREs in eukaryotic mRNAs (Bakheet et al. 2006; Gruber et al. 2010), and the presence of this 3'UTR element was observed in a majority of PGE₂ pathway genes (Moore et al. 2011). The role of ARE-mediated post-transcriptional regulation in prostaglandin biosynthesis has been best described for COX-2 and cPLA₂ (Dixon 2004; Young and Dixon 2010; Tay et al. 1994). The functional ARE of COX-2 is comprised of an evolutionarily-conserved 116 nucleotide stretch containing six AUUUA clusters located proximal to the stop codon (Dixon et al. 2000). The COX-2 ARE mediates important interaction with ARE RNA-binding proteins that can either target the mRNA for rapid decay or enhance the half-life of the transcript, depending on the cellular context (Sengupta et al. 2003; Young et al. 2009). Similar to COX-2, AREs present in the cPLA₂ exert an mRNA-destabilizing effect and impact overall PGE_2 synthesis (Tay et al. 1994). A further refinement of our understanding of these post-transcriptional mechanisms may provide a novel therapeutic window for the development of innovative NSAIDs targeting aberrant ARE function for the treatment of pain, inflammation, and cancer.

2.2 ARE RNA-Binding Proteins

Through its presence, the ARE serves to target mRNAs for rapid degradation and/or translational suppression within the cytoplasm (Barreau et al. 2005; Beisang and Bohjanen 2012). AREs mediate this through association with RNA-binding proteins and miRNAs that bind with high affinity (Beisang and Bohjanen 2012; Stumpo et al. 2010; Young et al. 2012; Jing et al. 2005; Garneau et al. 2007). There are >20 known ARE-binding proteins with a majority being associated with promoting mRNA stabilization, mRNA decay, or controlling translation by directing ARE-containing mRNAs to P-bodies and stress granules (Beisang and Bohjanen 2012; Garneau et al. 2007). Through these mechanisms, ARE-binding proteins exhibit pleiotropic effects on gene expression, since a single ARE-binding protein can bind to multiple mRNAs and binding can occur among different classes of AREs (Lopez de Silanes et al. 2007).

Considerable evidence exists to support the integral role of *trans*-acting AREbinding proteins in prostaglandin biosynthesis that allow for fine control of gene expression in response to various cellular cues and signaling within the



Fig. 9.3 ARE RNA-binding proteins mediating post-transcriptional regulation of COX-2. COX-2 expression is tightly regulated at the post-transcriptional level by RNA-binding proteins that promote mRNA stability (HuR, CUGBP2, and RBM3), mRNA decay (TTP), and translational inhibition (TIA-1 and CUGBP2) through their binding of the COX-2 AU-rich element (ARE)

microenvironment. Various cytoplasmic proteins have been detected to bind AREs and a majority of work has focused on COX-2 ARE-binding proteins with 16 different RNA-binding proteins identified to bind the COX-2 3'UTR (Young and Dixon 2010). This section will describe the most well characterized ARE RNA-binding proteins that regulate COX-2 expression and associated prostaglandin biosynthesis (Fig. 9.3).

2.2.1 HuR

The HuR protein (ELAVL1, Hu antigen R) is a ubiquitously expressed member of the ELAV (Embryonic-Lethal Abnormal Vision in *Drosophila*) family of RNAbinding proteins that consists of HuR and primarily neuronal-specific HuB, HuC, and HuD (Brennan and Steitz 2001; Hinman and Lou 2008). It is primarily nuclear localized (>90 %), where it assists in pre-mRNA splicing and cytoplasmic export of mRNA transcripts, followed by rapid nuclear re-localization (Brennan and Steitz 2001; Fan and Steitz 1998; Srikantan and Gorospe 2011). HuR is comprised of two tandemly arrayed RNA-recognition motifs (RRM), followed by a hinge region and a third RRM. The hinge region contains an HNS domain, which mediates nucleo-cytoplasmic shuttling of the HuR protein (Fan and Steitz 1998; Keene 1999).

The ability of HuR to function as an ARE-stability factor and enhance mRNA half-life appears to be linked to its subcellular localization (Brennan and Steitz 2001; Srikantan and Gorospe 2012). In response to cellular stress (e.g., inflammation, hypoxia, DNA damage), HuR translocates to the cytoplasm and enhances the stability of inflammation- and other survival-associated transcripts. Similarly, HuR overexpression observed in a variety of tumor types results in cytoplasmic localization and subsequent ARE-mRNA stabilization (Wang et al. 2013a; Abdelmohsen and Gorospe 2010; Srikantan and Gorospe 2012). Although the precise molecular underpinnings that regulate HuR nucleo-cytoplasmic localization are not entirely

known, a variety of cellular signals known to activate MAPK pathways (ERK and p38 kinases), PI-3-kinase pathway, and Wnt signaling pathway, have been shown to trigger HuR cytoplasmic localization and influence ARE-mRNA stabilization (Winzen et al. 1999; Yang et al. 2004; Tran et al. 2003; Ming et al. 2001; Briata et al. 2003).

HuR has been shown to bind and post-transcriptionally regulate numerous AREcontaining transcripts associated with cancer traits and inflammation (Wang et al. 2013a; Abdelmohsen and Gorospe 2010). Based on its ability to bind the COX-2 ARE region consisting of a highly conserved cluster of six AUUUA elements located near the stop codon, HuR has been identified as a trans-acting factor involved in regulating COX-2 expression (Dixon et al. 2001). The enhanced stabilization of COX-2 mRNA observed in colon cancer cells and inflamed myeloid leukocytes is, in part, due to elevated cytoplasmic levels of HuR (Dixon et al. 2001, 2006; Young et al. 2009). Furthermore, several studies indicate that HuR overexpression and cytoplasmic localization is a marker for elevated COX-2 that is correlated with advanced tumor stage and poor clinical outcome (Dixon 2003; Abdelmohsen and Gorospe 2010; Wang et al. 2013a). PLA₂ and mPGES-1 have also been identified as a direct targets of HuR, with inflammatory signaling enhancing the association of cPLA_{2 α} mRNA with cytosolic HuR (Liao et al. 2011; Lopez de Silanes et al. 2004). These findings suggest HuR to play a distinct regulatory role by integrating expression of factors involved in prostaglandin synthesis through mRNA stabilization.

It is generally accepted that HuR stabilizes mRNAs by competing or displacing destabilizing factors from the ARE (Benjamin and Moroni 2007). Another mechanism indicates that HuR can inhibit mRNA association with the decay machinery or possibly protect the poly(A) tail from degradation (Linker et al. 2005; Lal et al. 2004; Ma et al. 1997). Current evidence now indicates that HuR's ability to influence stabilization of bound mRNAs can be mediated through HuR's interplay with miRNAs that associate with the same transcript (Srikantan et al. 2012), and recent findings demonstrate that HuR promotes COX-2 mRNA stabilization by outcompeting miRNAs that share the same ARE binding sites in the COX-2 3'UTR (Young et al. 2012). In line with these findings, HuR has been shown to rescue mRNA transcripts held translationally silenced by miRNAs in P-bodies to promote their translation (Bhattacharyya et al. 2006; Srikantan et al. 2012).

High-throughput based biochemical screens have identified small-molecule compounds with the ability to disrupt HuR/ARE interactions (Meisner and Filipowicz 2010; Meisner et al. 2007; Chae et al. 2009; D'Agostino et al. 2013). Work conducted by Meisner and coworkers had identified and characterized three compounds (okicenone, dehydromutacin, and MS-444) as specific, low-molecular-weight HuR inhibitors (Meisner et al. 2007). These compounds are polyketides purified from plant and microbial extracts and the latter, MS-444, has been extensively studied with regard to its mechanism of action and ability to inhibit HuR. Mechanistically, MS-444 inhibits HuR homodimerization and blocks its cytoplasmic export. The impact of this results in loss of HuR-dependent mRNA stabilization and disruption of HuR/miRNA interaction (Young et al. 2012; Meisner et al. 2007).

2.2.2 TTP

Tristetraprolin (TTP, ZFP36, TIS11, NUP475) is a member of a small family of Cys3His zinc finger proteins. This family is comprised of TTP, ZFP36L1 (BRF-1), and ZFP36L2 (BRF-2), all which have been shown to play a critical role in regulated mRNA decay (Sanduja et al. 2011; Blackshear 2002; Ciais et al. 2013). TTP is one of the best-characterized post-transcriptional regulators, whereby it's binding of AREs promotes rapid mRNA decay (Brooks and Blackshear 2013; Carballo et al. 1998). The binding of TTP to AREs targets the transcript for rapid degradation through the recruitment of mRNA deadenylases, translational repressors, and mRNA decapping proteins onto the mRNA transcript (Chen et al. 2001; Mukherjee et al. 2002; Fenger-Gron et al. 2005; Lykke-Andersen and Wagner 2005; Franks and Lykke-Andersen 2007; Hau et al. 2007). At this point, TTP promotes the nucleation of P-bodies by delivering its cargo mRNA for subsequent degradation (Kedersha et al. 2005; Blanco et al. 2014; Franks and Lykke-Andersen 2007) (Fig. 9.4a).



Fig. 9.4 P-bodies are sites of mRNA storage or decay. (a) Schematic of a P-body. Once an AREcontaining mRNA is targeted for decay and partially deadenylated, TTP recruits mRNA degradation machinery facilitating mRNA decapping (Dcp1a/Dcp2/Hedls/Lsm1-7) and further deadenylation. RNA degradation occurs in a $5' \rightarrow 3'$ direction by the exonuclease Xrn-1. TTP also facilitates miRNA-mediated mRNA degradation through association with Ago proteins. (b) Example of P-bodies present in intestinal epithelial cells. P-bodies were detected by immunofluorescence staining for the P-body marker DCP1a (*green signal*). Cellular nuclei are visualized by DAPI (*blue signal*). Scale bar 10 μ m

TTP has also been implicated as a novel mediator of miRNA-dependent posttranscriptional regulation through its ability to associate with the argonaute (Ago) protein family members and promote decay of ARE-containing mRNAs localized to P-bodies (Eulalio et al. 2007; Jing et al. 2005). Alternatively, TTP can promote exosome-mediated mRNA degradation through its ability to interact with the exosome and recruit it to the AU-rich mRNAs (Chen et al. 2001).

TTP plays a critical role in normal physiology as TTP knockout mice develop early on a severe syndrome of growth retardation, cachexia, arthritis, inflammation and autoimmunity (Taylor et al. 1996). Based on this severe inflammatory phenotype, inflammatory mediators such as TNF- α , GM-CSF and COX-2 have been identified as ARE-containing targets of TTP (Lai et al. 1999; Carballo et al. 2000; Brooks and Blackshear 2013; Sawaoka et al. 2003; Young et al. 2009). TTP preferentially binds the nonameric sequence motif, UUAUUUAUU, the core destabilizing element of many ARE-containing mRNAs (Blackshear et al. 2003; Brewer et al. 2004). As nonameric sequences are the second most abundant sequence motif present in genes involved in the PGE₂ pathway (Moore et al. 2011), TTP's role in the regulation of prostaglandin biosynthesis may extend beyond COX-2 regulation.

TTP is an inducible, immediate-early response gene whose expression can be induced by mitogenic stimuli such as insulin, growth factors (e.g., TGF- β), and proinflammatory signals (Blanco et al. 2014; Ogawa et al. 2003). However, a consistent loss of TTP expression occurs in a variety of human cancers such as breast, colon, cervix, prostate, and lung (Sanduja et al. 2009, 2012; Brennan et al. 2009; Young et al. 2009), and suppressed TTP expression can serve as a negative prognostic breast cancer indicator (Brennan et al. 2009). These findings indicate that the presence of TTP in normal tissues serves a protective role by controlling expression of various pro-inflammatory and prostaglandin synthesis mediators, while loss of TTP expression in tumors contributes to aberrant overexpression of these transcripts.

2.2.3 RBM3

RBM3 (RNA-binding motif protein 3) is a member of a family of glycine-rich RNA-binding proteins (Derry et al. 1995). RBM3 is comprised of a single RRM and has been shown to regulate COX-2 expression through ARE binding (Cok and Morrison 2001; Sureban et al. 2008). In addition to COX-2, RBM3 has been shown to bind and stabilize the IL-8 and VEGF mRNAs and prevents mitotic catastrophe in colon cancer cells (Sureban et al. 2008). Together with the findings that RBM3 is significantly upregulated in colorectal tumors and overexpression promotes oncogenic transformation in fibroblasts (Sureban et al. 2008), identify a pro-oncogenic role of this ARE-binding protein.

Previously, yeast two-hybrid screens identified the ability of RBM3 to interact with HuR (Anant et al. 2010). Although the significance of this interaction remains to be determined, RBM3, like HuR, is a nucleo-cytoplasmic shuttling protein whose cytoplasmic export is triggered by stresses such as hypothermia, hypoxia, serum deprivation (Wellmann et al. 2010). Through its ability to promote the translation of otherwise unstable mRNA transcripts, RBM3 is being recognized as a contributing factor promoting enhanced prostaglandin levels in pathogenic states.

2.2.4 CUGBP2

A member of the CUGBP-ETR-3-like factors family, CUG triplet repeat-binding protein 2 (CUGBP2) is a ubiquitously expressed RNA-binding protein comprised of two N-terminal RRMs and one C-terminal RRM (Mukhopadhyay et al. 2003; Murmu et al. 2004).

Like other members of this family of proteins, CUGBP2 regulates alternative splicing, RNA editing, and mRNA translation (Anant et al. 2001). Importantly, CUGBP2 displays high affinity for the COX-2 ARE and has been shown to regulate its expression in response to ionizing radiation in colon cancer cells and growth factor stimulation of colonic stromal cells (Mukhopadhyay et al. 2003; Walker et al. 2010). Studies in vascular smooth muscle cells demonstrated that the association between CUGBP2 and the COX-2 ARE occurs in response to platelet-derived growth factor (PDGF)-dependent phosphorylation of the CUGBP2 protein. To this extent, the kinases c-Src and c-Abl directly phosphorylate CUGBP2 thus mediating its ability to inhibit ribosomal loading of the COX-2 mRNA (Xu et al. 2007). In a model of cardiac hypertrophy, CUGBP2 was identified to control pro-inflammatory stimulus involving COX-2/PGE₂ and promote subcellular mRNA trafficking specific cytoplasmic stress granules to maintain homeostasis in cardiac cells (Moraes et al. 2013).

CUGBP2 exhibits a complex biology such that it promotes COX-2 mRNA stabilization coupled with translational inhibition. Although CUGBP2 and HuR share similar affinities for the same binding sites in the COX-2 ARE, CUGBP2 effectively outcompetes HuR and promotes translational silencing of COX-2 (Murmu et al. 2004). This dynamic, seemingly antagonistic role of CUGBP2 indicates a possible role for this protein in the early stages of inflammation-induced tumorigenesis by attenuating PGE₂ synthesis through repression of COX-2 expression.

2.2.5 TIA-1

T cell intracellular antigen 1 (TIA-1) is an RNA-binding protein originally identified in activated T lymphocytes and is comprised of three RRMs. TIA-1 displays high affinity for short repeats of uridylate rich regions in non-coding regions (e.g., 3'UTR) of many pro-inflammatory cytokines (Tian et al. 1991; Lopez de Silanes et al. 2005). TIA-1 acts as a translational repressor, and under conditions of stress, it sequesters its target mRNA transcripts to cytoplasmic RNA granules known as stress granules (SGs). There, mRNAs are held translationally silenced and are sorted to re-enter translation or proceed to ARE-mediated mRNA decay in P-bodies (Kedersha et al. 2005; Anderson and Kedersha 2008). Along with it recognized role in translational suppression, TIA-1 is also implicated in alternative splicing regulation of various pre-mRNAs through binding adjacent to exon/intron boundaries (Wang et al. 2010b, 2014).

TIA-1 has been shown to bind the COX-2 ARE and regulate its expression through translational inhibition without altering COX-2 mRNA turnover. However, TIA-1-mediated regulation of COX-2 is deficient in colon cancer cells, thus contributing to increased polysome association with the COX-2 mRNA (Dixon et al. 2003). This regulation of COX-2 expression was also observed in TIA-1-deficient

fibroblasts that produce significantly more COX-2 protein and PGE₂ than wild-type fibroblasts. The physiological role of TIA-1 was best described in studies with the TIA-1 knockout mice. Similar to TTP deficient mice, TIA-1 null mice exhibited aberrant expression of pro-inflammatory cytokines (e.g., COX-2), thus leading to the development of arthritis (Phillips et al. 2004). These findings indicate that loss of TIA-1-mediated regulation of COX-2 contributes to increase COX-2 protein expression and PGE₂ synthesis in cancer and inflammation.

2.3 MicroRNAs

MicroRNAs (miRNAs) represent a class of endogenous small non-coding RNAs involved in the post-transcriptional regulation of broad biological functions (Frankel and Lund 2012; Ambros 2004; Graves and Zeng 2012). MiRNAs are \approx 22 nucleotides and regulate the expression of complementary target mRNA(s) primarily through 3'UTR binding by influencing translation and by causing degradation of target mRNAs (Fabian et al. 2010) and findings have implicated miRNA-mediated mRNA decay to be the predominant mechanism (Guo et al. 2010). MiRNAs are transcribed by RNA polymerase II to produce a longer primary precursor primiRNA (Graves and Zeng 2012). This pri-miRNA is maturated by the Drosha endonuclease microprocessor complex to release $\approx 60-70$ nucleotides long stem-loop intermediate pre-miRNAs. Alternatively, the pre-miRNA can be generated in a Drosha-independent manner, through intronic pri-miRNAs, which are maturated by the spliceosome into pre-miRNAs (Bartel 2004). The pre-miRNA is then exported in the cytosol through by exportin 5 and processed by Dicer to release a \approx 22 nucleotides mature miRNA (guide strand)/miRNA* (passenger strand) duplex (Bartel 2004). According to the current canonical model, the guide strand is associated with argonaute (Ago) proteins and incorporated in the RNA-Inducing Silencing Complex (RISC) where the miRNA binds its mRNA target(s), while the miR* is degraded (Yates et al. 2013; Bartel 2004). Current findings indicate that the miR* can also play an important role in biological processes by targeting different mRNAs than their respective guide strands (e.g. miR-21* in kidney fibrosis or miR-199* in PGs biosynthesis) (Rayner et al. 2011). Consequently, the role of the passenger strand needs to be considered with caution, especially in studies using knockout mice, where the guide and the passenger strands are both deleted.

miRNAs regulate more than 60 % of coding mRNAs and their biological effects are organized in a complex network (Friedman et al. 2009). This level of complexity occurs through the pleiotropic actions of miRNAs targeting numerous mRNAs and it is likely that the phenotype arising alteration of one particular miRNA is the consequence of a concerted action on several targets (Ambros 2004; Bartel 2004). A further degree of complexity is provided by other non-coding RNAs such as competitive endogenous RNAs (ceRNA including pseudogenes, circular RNAs or long non-coding RNAs), which are potent regulators of miRNAs by acting as a "sink" for miRNAs (Tay et al. 2014). Other post-transcriptional regulators need also to be taken into account in the miRNAs network, such as RNA-binding proteins

including ARE-binding proteins, which can promote or inhibit miRNAs function (Adams et al. 2014; Young et al. 2012; Srikantan et al. 2012).

miRNAs have been associated with a variety of pathologies including metabolic diseases, cardiovascular disease, and cancer (Farazi et al. 2013). Alterations in expression or activity of miRNAs regulating the expression of key enzymes involved in prostaglandin homeostasis are likely to contribute to the onset and progression of these pathologies. Numerous miRNAs have been predicted by different computational methods to bind directly to the 3'UTR of key enzymes involved in prostaglandin biosynthesis (e.g., TargetScan, miRanda, microRNA.org) (Moore et al. 2011). Several miRNAs targeting phospholipase A2, phospholipase C, COX-1, COX-2, as well as prostaglandin synthases have been identified, with a majority of efforts on investigating COX-2 regulation (Moore et al. 2011; Young and Dixon 2010). Furthermore, miRNAs can affect the expression of these enzymes through indirect mechanisms, for instance targeting transcription factors controlling the expression of the expression of the enzymes implicated in PG synthesis (e.g. NFkB) (Ma et al. 2011). Herein, we will primarily discuss functionally validated miRNAs regulating the expression of the different enzymes involved in PGs synthesis (Table 9.1).

2.3.1 miR-16-1 and miR-15a

The first reports implicating a role for miRNAs in cancer progression examined miR-16-1/15a cluster and its function in chronic lymphocytic leukemia (CLL) due to a commonly deleted genomic region in CLL accompanied with loss of miR-16-1 and miR-15a expression (Calin et al. 2002; Dohner et al. 2000). The expression of miR-16 is also reduced in other cancers, including colorectal, liver, prostate, lung, and gliomas (Young et al. 2012; Ge et al. 2014; Yang et al. 2014; Bottoni et al. 2005; Lu et al. 2005; Bonci et al. 2008; Navarro et al. 2011). Various mechanisms are responsible for the downregulation of miR-15a and miR-16 in cancers. MiR-16 is encoded within an intronic region of DLEU2 (deleted in lymphocytic leukemia) on chromosome 13q14, a region which is frequently deleted in CLL (Pekarsky and Croce 2014). Other evidence indicates downregulation of miR-16 can be also mediated by epigenetic events such as HDACs overexpression (Sampath et al. 2012). MiR-16-1/15a serves in a tumor suppressor capacity by regulating expression of anti-apoptotic factors such as Mcl-1 and Bcl-2 (Yang et al. 2014; Liu et al. 2014; Sanchez-Beato et al. 2003; Cimmino et al. 2005). MiR-16 in particular has additionally been shown to play a role in cell cycle maintenance through regulation of several cell cycle regulatory genes (Liu et al. 2008).

The initial connection between miR-16 and COX-2 was demonstrated in the work by Jing et al. that identified the ability of miR-16 to promote degradation of ARE-containing transcripts as miR-16-1 contains sequence homology to AU-rich elements (Jing et al. 2005). Other studies have validated miR-16-1 targeting of the COX-2 3'UTR in work examining COX-2 3'UTR-mediated post-transcriptional regulation by miR-16-1 in response to diabetic stimuli in leukocytes (Shanmugam et al. 2008). A similar link between miR-16 and COX-2 was observed in hepatoma and colorectal cancer cells (Agra Andrieu et al. 2012; Young et al. 2012). While these findings underscore the importance of miR-16 in PG synthesis, RNA-binding proteins also regulate miR-16 function. MiR-16-induced mRNA decay requires the

Gene	Validated miRNAs ^a (References)
cPLA ₂ -α (PLA2G4A)	N.D.
cPLA ₂ -β (PLA2G4B)	miR-338 (Montenegro et al. 2009)
Phospholipase C (PLCy1)	miR-200b, miR-200c, miR-429 (Uhlmann et al. 2010) miR-218-2 (Guan et al. 2013)
COX-1 (PTGS1)	N.D.
COX-2 (PTGS2)	miR-16-1 (Shanmugam et al. 2008; Young et al. 2012; Agra Andrieu et al. 2012) miR-26b (Ji et al. 2010; Li et al. 2013) miR-101a (Chakrabarty et al. 2007; Daikoku et al. 2008; Strillacci et al. 2009; Tanaka et al. 2009; He et al. 2012; Wang et al. 2010a) miR-137 (Chen et al. 2012) miR-143 (Song et al. 2011; Pham et al. 2013; Wu et al. 2013) miR-146a (Sato et al. 2010; Cornett and Lutz 2014) miR-199a-5p (Chakrabarty et al. 2007; Daikoku et al. 2008; Akhtar and Haqqi 2012) miR-199a-3p/miR-214 (Williams et al. 2012) miR-558 (Park et al. 2013; Strillacci et al. 2009)
mPGES-1 (PTGES)	N.D.
mPGES-2 (PTGES2)	miR-146a (Matysiak et al. 2013)
cPGES (PTGES3)	N.D.
EP1 Receptor (PTGER1) EP2 Receptor (PTGER2) EP3 Receptor (PTGER3) EP4 Receptor (PTGER4)	N.D. N.D. miR-101 (Chandramouli et al. 2012)
Prostacyclin synthase (PTGIS)	N.D.
IP Receptor (PTGIR)	N.D.
PGD ₂ synthase (PTGDS) PGD ₂ synthase (HPGDS)	N.D. N.D.
DP Receptor (PTGDR) DP Receptor (PTGDR2)	N.D. N.D.
PGF ₂₀ Synthase (PGFS)	N.D.
FP Receptor (PTGFR)	N.D.
TXA ₂ Synthase (TBXAS1)	N.D.
TXA ₂ Receptor (TBXA2R)	miR-765 (Nossent et al. 2011)
Prostaglandin Transporter (SLCO2A1, PGT)	N.D.
15-Hydroxyprostaglandin Dehydrogenase (HPGD, 15-PGDH)	miR-21 (Lu et al. 2014)

 Table 9.1
 Validated microRNAs targeting genes involved in prostaglandin synthesis

^aBased on strong evidence criteria using MiRTarBase database (Hsu et al. 2014) or functionally validated in literature. *N.D.* not detected

mRNA-destabilizing factor TTP to promote decay of ARE-mRNAs (Jing et al. 2005). Other proteins able to stabilize COX-2 mRNA, such as HuR, can interfere with miR-16 binding of the COX-2 ARE by binding miR-16 (Young et al. 2012). Similarly, hnRNPK (heterogeneous nuclear ribonucleoprotein K) leads to a

decreased binding of miR-16 to COX-2 mRNA (Shanmugam et al. 2008). Interestingly, studies have shown miR-16 also directly regulates HuR expression. In breast cancer, miR-16-1 has been implicated to repress translation of HuR and loss of miR-16-1 was correlated with HuR overexpression (Xu et al. 2010). Consistent with this, miR-16 has been shown to decrease COX-2 expression in hepatocellular carcinoma by two distinct mechanisms. First, through direct binding of the COX-2 3'UTR and impeding translation and second, miR-16 downregulates HuR expression (Agra Andrieu et al. 2012).

2.3.2 miR-200 Family

The miR-200 family encompasses five miRNAs, including miR-200a, miR-200b, miR-200d, miR-141 and miR-429, which are encoded by two different polycistronic pri-miRNA transcripts (Moore et al. 2011). MiR-200a, b and miR-429 are located on Chr-1, while miR-200c and miR-141 are on Chr-12 (Gregory et al. 2008). Interestingly, the chromosomal regions encoding miR-200 family members are frequently deleted in cancers, leading to a downregulation of these miRNA (Mongroo and Rustgi 2010). MiR-200 family is also downregulated in non-cancerous diseases such as liver fibrosis (Murakami et al. 2011). The link between miR-200 family and PGs synthesis was highlighted in breast cancer cells, where overexpression of miR-200b, miR-200c and miR-429 promoted downregulation of PLC γ 1 mRNA expression, and an increase of PLC γ 1 mRNA was observed with a miR-200bc/429 inhibitor in MCF7 cells (Uhlmann et al. 2010). Moreover, the increase of miR-200bc/429 in breast cancer cells reduces cell viability and induces apoptosis (Uhlmann et al. 2010).

2.3.3 miR-137

MiR-137 loss has been linked with several diseases such as schizophrenia (Schmidt et al. 2013) and cancers including lung, ovarian, and neuroblastoma (Li et al. 2014; Guo et al. 2013; Althoff et al. 2013). Based on predicted binding sites for miR-137 in the COX-2 3'UTR, expression of miR-137 in glioma cells promoted COX-2 downregulation and inhibition of cell proliferation and invasion. These observations are coupled with an inverse correlation between the expression of miR-137 and COX-2 in glioblastoma cells, suggesting a potential role of miR-137 as a tumor suppressor in glioblastoma (Chen et al. 2012).

2.3.4 miR-21

MiR-21 is the only miRNA known to date to be upregulated in all human cancers (Pan et al. 2011; Volinia et al. 2006). A large amount of experimental data has demonstrated that overexpression of miR-21 impacts various aspects of tumorigenesis including tumor cell proliferation, evasion of apoptosis, and invasion and metastasis (Pan et al. 2011). In addition, miR-21 is implicated in metabolic disorders such as NAFLD (Non-Alcoholic Fatty Liver Disease), where its ability to downregulate PTEN affects insulin sensitivity (Vinciguerra et al. 2009; Ling et al. 2012). MiR-21 is an important regulator of prostaglandin synthesis by directly targeting 15-hydroxyprostaglandin dehydrogenase (15-PGDH), which converts PGE₂ into its inactive form, 15-keto-PGE₂ (Lu et al. 2014). In cholangiocarcinoma, where elevated expression of miR-21 is correlated with enhanced PGE₂ levels and loss of 15-PGDH (Lu et al. 2014). It has been found that nicotine stimulation upregulates the expression of miR-21 in gastric cancer cells and this effect was in an NF κ B-dependent manner (Shin et al. 2011), indicating that ability of miR-21 to affect PGs synthesis can be also influenced by indirect effects. On these same lines, a connection between PTEN loss and the increase of cPLA₂ has been found in the prostate cancer cells (Vignarajan et al. 2014). Thus, the ability of miR-21 to downregulate PTEN may indirectly affect cPLA2 expression and impact PGs biosynthesis.

2.3.5 miR-101

The genomic locus encoding miR-101 is located on Chr 1p31.3 and loss of miR-101 has been observed in various tumor types including in 67 % of metastatic cancer cells, suggesting its involvement in tumor cell invasion (Varambally et al. 2008; Su et al. 2009; Liang et al. 2014; Hiroki et al. 2010; Zhang et al. 2011; Strillacci et al. 2009). In endometrial cancer, miR-101 is involved in regulating expression of oncogenes such as Mcl-1 and Fos (Konno et al. 2014). MiR-101 has been shown to regulate COX-2 expression in a number of models. Using a toxin-induced liver injury, it has been shown that miR-101 regulates COX-2 expression and prostaglandin synthesis (Yoshioka et al. 2011). Another study has also shown the downregulation of miR-101 and a concomitant upregulation of COX-2 in cervical cancer tissue from patients, indicating miR-101 loss as a biomarker of COX-2 elevation (Lin et al. 2014). Supporting these findings, it has been found that the expression of miR-101 is low in HeLa cells and the restoration of miR-101 expression downregulates COX-2 expression with decreased cell proliferation and migration and an induction of apoptosis (Huang et al. 2013). Similar results were observed in colorectal and gastric cancer cells where COX-2 downregulation occurred following miR-101 overexpression, where miR-101 was observed to interact with COX-2 3'UTR and reduce protein expression (Hao et al. 2011; Wang et al. 2010a; Strillacci et al. 2009). miR-101 also plays an important role in embryo implantation by targeting directly COX-2. It has been shown that miR-101 is induced in mouse uterus during embryo implantation, while the expression of COX-2 is reduced (Chakrabarty et al. 2007). Given the importance of COX-2-derived prostaglandins during embryo implantation (Kennedy et al. 2007; Lim et al. 1999), the ability of miR-101 to downregulate COX-2 expression may influence the implantation process.

Beside its ability to downregulate COX-2, it has been suggested that miR-101 downregulates also EP4 receptor expression. An inverse correlation between the expression of miR-101 and EP4 has been observed in colorectal cancer and the overexpression of miR-101 in colon cancer cells causes a reduction of EP4 expression and impairs cell proliferation and migration (Chandramouli et al. 2012). These findings show that miR-101 is a potent inhibitor of PGE₂ signaling by impacting PGE₂ synthesis and its transduction pathway.

2.3.6 miR-146a

This miRNA is induced by NF κ B and is an important regulator of innate immune response by targeting directly mRNAs in the TNF α and IL-6 signaling pathways (Taganov et al. 2006). More recently, miR-146a was found to be an important regulator of PG synthesis. The link was shown in fibroblasts from chronic obstructive pulmonary disease patients (COPD) where increases COX-2/PGE₂ expression were observed (Sato et al. 2010). This effect was partially associated with reduced levels of miR-146a and inhibition of COX-2 mRNA degradation, and overexpression of miR146a prevented IL-1 β or TNF α -induced COX-2 expression and PGE₂ synthesis (Sato et al. 2010). Interestingly, the expression of miR-146a is positively regulated by NF κ B, thus suggesting that the induction of this miRNA may represent part of a negative feedback loop aiming at lowering PG synthesis (Ghose et al. 2011; Taganov et al. 2006).

Recent findings have shown that miR-146a can also affect other factors involved in PG synthesis. MiR-146a inhibits an immunoregulatory function of bone marrow stem cells (BMSCs) by targeting directly mPGES-2 mRNA and limiting PGE₂ synthesis (Matysiak et al. 2013). In this study, neuronal differentiated BMSC (nBMSC) express a higher level of miR-146a compared to non-differentiated BMSCs and inhibition of miR-146a with a selective antagomiR in nBMSC leads to an increase of PGE₂ synthesis (Matysiak et al. 2013). Alteration of miR-146a expression has been observed in a variety of pathologies such as lung cancer (Cornett and Lutz 2014), where the expression of miR-146a is lost. Paradoxically, herpes simplex virus-1 can induce the expression of miR-146a in neuronal cells, which leads to an induction of inflammation and particularly an induction of PLA2 and COX-2 (Hill et al. 2009). These findings suggest that different mechanisms may occur, depending on the disease etiology and the cellular context.

2.3.7 miR-26b

MiR-26b is encoded within the intronic region of genes encoding for proteins of carboxy-terminal domain RNA polymerase II polypeptide A small phosphatase (CTDSP) family (Zhu et al. 2012). The expression of this miRNA is reduced in different neoplasias such as hepatocellular, nasopharyngeal, lung, and breast cancers (Shen et al. 2014; Ji et al. 2010; Gao et al. 2011; Li et al. 2013). MiR-26b is associated with obesity and its expression is downregulated by free fatty acids, glucose, or glucocorticoids in adipocytes, indicating a link between this miRNA and insulin resistance (Xu et al. 2014). It has been shown that miR-26b regulates COX-2 expression in desferrioxamine-treated nasopharyngeal cancer cells and overexpression of miR-26b leads to a reduction in cell proliferation associated with the COX-2 downregulation (Ji et al. 2010). In breast cancer, the expression of miR-26b contributes to decreased COX-2 mRNA and protein levels and inhibits cell proliferation (Li et al. 2013).

2.3.8 miR-199/miR-199*

Currently, few studies have documented the function of miR-199 in physiological processes but its overexpression has been associated with liver fibrosis (Murakami et al. 2011). miR-199 appears to play an important function during pregnancy given

that its downregulation is observed in laboring myometrium while COX-2 expression is increased (Williams et al. 2012). More strikingly, overexpression of miR-199-3p/ miR-214 using miRNA mimics in myometrial cells reduced COX-2 protein expression (Williams et al. 2012). The passenger strand of miR-199, miR-199* is also a potent inhibitor of COX-2 expression. In human osteoarthritis chondrocytes, it has been shown that miR-199* inhibits IL-1β-induced COX-2 expression (Akhtar and Haqqi 2012). Interestingly, IL-1β-induced p38 MAPK activation is inversely correlated with the level of miR-199* and an induction of miR-199* was observed in chondrocytes treated with a p38 inhibitor (Akhtar and Haqqi 2012). COX-2-derived prostaglandins play an important role during embryo implantation (Kennedy et al. 2007; Lim et al. 1999). The role of miRNA in this process has been observed with the expression of miR-101 and miR-199* increased in the murine uterus during embryo implantation, while the expression of COX-2 is reduced (Chakrabarty et al. 2007). Consistent with these findings, the expression of miR-199a* are reduced in a model of endometrial cancer induced by PTEN loss, while the expression of COX-2 is induced (Daikoku et al. 2008).

2.3.9 miR-143/145

The miR-143/145 cluster is located on Chr 5q32 and current findings suggest that they originate from the same primary miRNA (Moore et al. 2011). Downregulation of miR-143/145 is observed in several disorders including colorectal, prostate, esophageal, bladder, osteosarcoma cancers, and hematologic malignancies such as B cell lymphoma or chronic lymphocytic leukemia (Luo et al. 2011; Kojima et al. 2014; Wu et al. 2011; Yoshino et al. 2013; Hu et al. 2012; Fabbri et al. 2008). The suppressive effect of miR-143 on COX-2 expression has been described in gastric, bladder, and pancreatic cancer cells through targeting the COX-2 3'UTR (Wu et al. 2013; Song et al. 2011; Pham et al. 2013). In addition, miR-143-3p and miR-143-5p decrease cell viability and proliferation and induce apoptosis in gastric cancer cells (Wu et al. 2013). MiR-143 expression is diminished in bladder cancer cells and its overexpression leads to COX-2 downregulation and suppression of metastasis (Song et al. 2011).

2.3.10 miR-542-3p

Very few studies have been performed on this miRNAs but current findings suggest that miR-542 acts as a tumor suppressor, through its ability to downregulate the prosurvival oncogene survivin (Althoff et al. 2014). More recently, it has been shown that miR-542-3p regulates PG synthesis by directly targeting the COX-2 3'UTR (Moore et al. 2012). Interestingly, the 3'UTR region targeted by miR-542-3p contains a common single nucleotide polymorphism (SNP) in the COX-2 gene at position 8473 in exon 10 (T8473C; rs5275) that is associated with increased risk and/or NSAID responsiveness in a number of cancers where COX-2 over-expression is a contributing factor. Mir-542-3p was identified to bind transcripts of the 8473T allele and promote mRNA decay, whereas the cancer-associated variant 8473C allele interfered with miR-542-3p binding. Colon cancer cells and tissue displayed COX-2 expression levels that were dependent on T8473C allele dosage and allelic-specific expression of COX-2 was observed to be a contributing factor promoting COX-2 overexpression (Moore et al. 2012). These findings provide a novel molecular explanation underlying cancer susceptibility associated with COX-2 T8473C SNP and suggest that other SNPs might influence miRNA activity involved in PG synthesis.

2.3.11 miR-338

A relationship between miR-338 loss and several cancers including gastric, colorectal, and neuroblastoma has been found (Peng et al. 2014; Chen et al. 2013; Sun et al. 2014). MiR-338 plays an important regulatory role in PG synthesis by targeting directly PLA2G4B (Phospholipase A2 Group VI B), a member of the cPLA₂ family in human chorioamniotic membrane (Montenegro et al. 2009). The downregulation of miR-388 at term is associated with an induction of PLA2G4B and inhibition of miR-338 leads to an induction of PLA2G4B in decidual cells (Montenegro et al. 2009).

2.3.12 Targeting miRNAs as a Therapeutic Approach to Restore Prostaglandin Homeostasis

Altered prostaglandin homeostasis in various pathologies has driven various therapeutic strategies aimed at targeting the activity of the key enzymes involved in this process. In this context, several pharmacologic inhibitors specifically targeting the activity of COX-2 have been developed (e.g. celecoxib, rofecoxib). However, this therapeutic approach revealed also several off-target and severe side effects, such as an increased risk of myocardial infarction, gastric ulceritis or hepatic toxicity that limit clinical feasibility (Tegeder et al. 2001; Tan et al. 2007; Steffel et al. 2006; Dogne et al. 2006). More recently, it has been suggested that targeting the expression of COX-2, rather than its activity would be a better approach with less side effects (Cerella et al. 2010). An attractive alternative strategy may reside in modulating the expression or activity of pathogenic miRNAs involved in promoting altered prostaglandin homeostasis. Moreover, the ability of some miRNAs to target multiple players in the PG pathway would allow an efficient inhibition of PGs synthesis. Recent findings have shown that chemically modified synthetic nucleotides mimicking or inhibiting endogenous miRNAs can be used for therapeutic purpose (van Rooij et al. 2012; Stenvang et al. 2012; Thorsen et al. 2012; Garzon et al. 2010; Deiters 2010). Antisense modified oligonucleotides (AMOs) (2'-O-methoxyethyl phosphorothioate modified antisense oligonucleotide; 2'-Fluoro modified antisense oligonucleotide), antagomiRs (3'cholesterol-conjugated, 2'-O-Me oligonucleotides having terminal phosphorothioate), and locked nucleic acid (LNA) modifications display increased affinity for their targets and are stable in serum allowing for in vivo delivery (Stenvang et al. 2012). Delivery of synthetic oligonucleotides has shown in vivo feasibility through intravenous or intraperitoneal administration. Miravirsen, a locked nucleic acid-modified DNA phosphorothioate antisense oligonucleotide targeting miR-122, was the first miRNA-targeted drug to enter clinical trials in 2008 for the treatment of HCV by intravenous injection (Hu et al. 2012). Other studies ability have demonstrated the of new formulated

2'-O-(2-methoxyethyl) modified antisense oligonucleotide (2'-MOE ASO) combined with a permeation enhancer (sodium caprate) for oral administration in humans (Tillman et al. 2008). In addition, special formulations have been made in order to improve the delivery of oligonucleotides such as liposomal or oleic-based nanoparticles formulation (Trajkovski et al. 2011; Wang et al. 2013b).

Several molecules from natural origins and particularly dietary polyphenols have the ability to modulate the expression of miRNAs involved in various pathologies. These compounds, such as curcumin and coffee polyphenols, can be found in many fruits and vegetables and display beneficial effects in various disorders *via* oral delivery (Prasad et al. 2014; Murase et al. 2011). Some of these molecules can modulate the expression of specific miRNAs involved in PGs biosynthesis. For instance, epigallocatechin gallate (EGCG), from green tea upregulates miR-16 in hepatocellular carcinoma cells (Tsang and Kwok 2010).

2.4 Processing Bodies and Stress Granules

Processing bodies (P-bodies) and stress granules (SG) are RNA granules that play an integral role in mRNA turnover and translational regulation. Although P-bodies were originally identified as cytoplasmic sites of mRNA degradation, evidence indicates that it can also serve as sites of short-term mRNA storage (Eulalio et al. 2007; Garneau et al. 2007). Similarly, SGs serve as interim sites of mRNA storage in cells subjected to severe stress. Both P-bodies and SGs are non-membrane bound granula whose formation is dynamic and reversible. Immunofluorescence analysis demonstrates that, in response to stress, P-bodies and SGs exist in close proximity and partial co-localize. This aspect supports the hypothesis that transcripts traffic between P-bodies and SGs, and that irreversible stresses may result in shuttling of mRNA transcripts from SGs to P-bodies for subsequent decay (Kedersha et al. 2005; Anderson and Kedersha 2008).

Since their discovery, many studies have catalogued the core components that comprise these RNA granules. P-bodies harbor enzymes responsible for mRNA decapping (e.g., DCP1a, DCP2, HEDLS), mRNA deadenylation (CAF-1, CCR4) and degradation (XRN-1), miRNA decay machinery (e.g., AGO2), and components of nonsense-mediated mRNA decay (NMD) (Eulalio et al. 2007; Kulkarni et al. 2010; Zheng et al. 2011) (Fig. 9.4). SGs are comprised of largely stalled translational pre-initiation complexes, small ribosomal subunits (e.g., eIF3, eIF4E, eIF4G) and their core components (Kedersha and Anderson 2007). In addition to these core components, several ARE-binding proteins have been shown to transiently localize to P-bodies and SGs, allowing for delivering of their mRNA target for translational suppression and/or decay (Blanco et al. 2014; Franks and Lykke-Andersen 2007; Gilks et al. 2004). Recently it has been shown that ARE-binding proteins play an important role in the control of P-body formation. Induction of TTP expression by the growth-inhibitory cytokine TGF- β results in enhanced P-body formation, resulting in TTP-driven sequestration of the COX-2 mRNA transcript to P-bodies

(Blanco et al. 2014). This ability of TTP to promote P-body formation was also observed in vivo, where analysis of colonic epithelium in TTP knockout mice showed limited numbers of P-bodies per cell as compared to wild-type littermates. Similarly, fibroblasts derived from these mice indicate that absence of TTP expression dramatically impairs P-body formation (Blanco et al. 2014). In line with these findings, loss of TTP expression in epithelial tumors correlates with low cellular number of P-bodies. These findings provide mechanistic insights into how TTP promotes the decay of COX-2 mRNA and controls prostaglandin formation in normal intestinal epithelium.

Similar to P-bodies, ARE-binding proteins can deliver mRNA transcripts to SGs. Although many stressors have been shown to induce SG formation (nutrient deprivation, oxidative stress, heat-shock), and oxidative stress appears to enhance TTP localization to SGs (Kedersha and Anderson 2007). This indicates the ability of TTP may serve as a shuttling protein mediating a functional interplay between SGs and P-bodies. The translational repressor TIA-1 has been shown to localize to SGs. Similarly, HuR and CUGBP2 have been observed in SGs when cells are subjected to heat-shock and other stresses (Gilks et al. 2004; Kedersha and Anderson 2007). The presence of ARE-binding proteins may be indicative of a protective cellular mechanism against stress in an attempt to orchestrate post-stress cellular repair without the high-energy cost of de novo transcription.

3 Conclusions

The biological effects of prostaglandins are diverse in normal physiology and pathophysiological conditions. It is well established that increased PGE₂ synthesis and signaling is a contributing factor in inflammation and cancer. Substantial evidence has demonstrated the causal role of unregulated COX-2 expression in many chronic diseases and cancer, along with evidence indicating the benefit of inhibiting COX-2 activity. Due to the unwanted side effects associated with long-term COX-2 inhibition, potential alternatives may reside in pharmacological targeting of other key factors (i.e. mPGES and EP receptors) or using natural product-based inhibitors. The work reviewed here demonstrates targeting gene expression by influencing the activity of post-transcriptional regulators to be feasible approach and gaining momentum as our understanding of these ARE-binding proteins and miRNAs increases. MiRNAs have been linked to a variety of human diseases and are promising therapeutic targets due to their pleiotropic effects on a large number of genes. Developing therapeutic strategies that will substitute or restore miRNA expression controlling genes in prostaglandin biosynthesis may prove to be more comprehensive than targeting individual genes. Further identification and characterization of RNA-bindings proteins that interact with 3'UTR AU-rich elements present in genes discussed holds great promise for therapeutic targeting that can selectively inhibit RNA-binding protein function. Ultimately, these novel approaches will allow the nascent cellular post-transcriptional machinery to counteract the pathogenic effects of chronic mRNA stabilization.

Acknowledgments This work was supported by the National Institutes of Health R01 CA134609 (D.A.D.), P30 CA168524 (D.A.D.), and T32 GM008562 20 (F.F.B.). We thank P. Patrignani and C. Lutz for helpful suggestions. We apologize to our colleagues for not being able to reference all primary work due to space limitations.

References

- Abdelmohsen K, Gorospe M (2010) Posttranscriptional regulation of cancer traits by HuR. Wiley Interdiscip Rev RNA 1:214–229
- Adams BD, Kasinski AL, Slack FJ (2014) Aberrant regulation and function of microRNAs in cancer. Curr Biol 24(16):R762–R776
- Agra Andrieu N, Motino O, Mayoral R, Llorente Izquierdo C, Fernandez-Alvarez A, Bosca L, Casado M, Martin-Sanz P (2012) Cyclooxygenase-2 is a target of microRNA-16 in human hepatoma cells. PLoS One 7(11):e50935
- Akhtar N, Haqqi TM (2012) MicroRNA-199a* regulates the expression of cyclooxygenase-2 in human chondrocytes. Ann Rheum Dis 71(6):1073–1080
- Althoff K, Beckers A, Odersky A, Mestdagh P, Koster J, Bray IM, Bryan K, Vandesompele J, Speleman F, Stallings RL, Schramm A, Eggert A, Sprussel A, Schulte JH (2013) MiR-137 functions as a tumor suppressor in neuroblastoma by downregulating KDM1A. Int J Cancer 133(5):1064–1073
- Althoff K, Lindner S, Odersky A, Mestdagh P, Beckers A, Karczewski S, Molenaar JJ, Bohrer A, Knauer S, Speleman F, Epple M, Kozlova D, Yoon S, Baek K, Vandesompele J, Eggert A, Schramm A, Schulte JH (2014) miR-542-3p exerts tumor suppressive functions in neuroblastoma by downregulating survivin. Int J Cancer. doi:10.1002/ijc.29091
- Ambros V (2004) The functions of animal microRNAs. Nature 431(7006):350-355
- Anant S, Henderson JO, Mukhopadhyay D, Navaratnam N, Kennedy S, Min J, Davidson NO (2001) Novel role for RNA-binding protein CUGBP2 in mammalian RNA editing. CUGBP2 modulates C to U editing of apolipoprotein B mRNA by interacting with apobec-1 and ACF, the apobec-1 complementation factor. J Biol Chem 276(50):47338–47351
- Anant S, Houchen CW, Pawar V, Ramalingam S (2010) Role of RNA-binding proteins in colorectal carcinogenesis. Curr Colorectal Cancer Rep 6(2):68–73
- Anderson P, Kedersha N (2008) Stress granules: the Tao of RNA triage. Trends Biochem Sci 33(3):141–150
- Asselin E, Drolet P, Fortier MA (1997) Cellular mechanisms involved during oxytocin-induced prostaglandin F2alpha production in endometrial epithelial cells in vitro: role of cyclooxygenase-2. Endocrinology 138(11):4798–4805
- Backlund MG, Mann JR, Holla VR, Shi Q, Daikoku T, Dey SK, DuBois RN (2008) Repression of 15-hydroxyprostaglandin dehydrogenase involves histone deacetylase 2 and snail in colorectal cancer. Cancer Res 68(22):9331–9337
- Bailey JM, Verma M (1990) Identification of a highly conserved 3' UTR in the translationally regulated mRNA for prostaglandin synthase. Prostaglandins 40(6):585–590
- Bakheet T, Williams BR, Khabar KS (2006) ARED 3.0: the large and diverse AU-rich transcriptome. Nucleic Acids Res 34:D111–D114
- Balagopal V, Fluch L, Nissan T (2012) Ways and means of eukaryotic mRNA decay. Biochim Biophys Acta 1819(6):593–603
- Barbieri SS, Amadio P, Gianellini S, Tarantino E, Zacchi E, Veglia F, Howe LR, Weksler BB, Mussoni L, Tremoli E (2012) Cyclooxygenase-2-derived prostacyclin regulates arterial thrombus formation by suppressing tissue factor in a sirtuin-1-dependent-manner. Circulation 126(11):1373–1384
- Barreau C, Paillard L, Osborne HB (2005) AU-rich elements and associated factors: are there unifying principles? Nucleic Acids Res 33(22):7138–7150
- Bartel DP (2004) MicroRNAs: genomics, biogenesis, mechanism, and function. Cell 116(2):281-297
- Beisang D, Bohjanen PR (2012) Perspectives on the ARE as it turns 25 years old. Wiley Interdiscip Rev RNA 3(5):719–731
- Benjamin D, Moroni C (2007) mRNA stability and cancer: an emerging link? Expert Opin Biol Ther 7(10):1515–1529
- Betz M, Hansch GM (1984) Release of arachidonic acid: a new function of the late complement components. Immunobiology 166(4–5):473–483
- Bhattacharya M, Peri KG, Almazan G, Ribeiro-da-Silva A, Shichi H, Durocher Y, Abramovitz M, Hou X, Varma DR, Chemtob S (1998) Nuclear localization of prostaglandin E2 receptors. Proc Natl Acad Sci U S A 95(26):15792–15797
- Bhattacharyya SN, Habermacher R, Martine U, Closs EI, Filipowicz W (2006) Relief of microRNA-mediated translational repression in human cells subjected to stress. Cell 125(6):1111–1124
- Blackshear PJ (2002) Tristetraprolin and other CCCH tandem zinc-finger proteins in the regulation of mRNA turnover. Biochem Soc Trans 30:945–952
- Blackshear PJ, Lai WS, Kennington EA, Brewer G, Wilson GM, Guan X, Zhou P (2003) Characteristics of the interaction of a synthetic human tristetraprolin tandem zinc finger peptide with AU-rich element-containing RNA substrates. J Biol Chem 278(22):19947–19955
- Blanco FF, Sanduja S, Deane NG, Blackshear PJ, Dixon DA (2014) Transforming growth factor beta regulates P-body formation through induction of the mRNA decay factor tristetraprolin. Mol Cell Biol 34(2):180–195
- Brooks SA, Blackshear PJ (2013) Tristetraprolin (TTP): Interactions with mRNA and proteins, and current thoughts on mechanisms of action. Biochimica et biophysica acta 1829 (6-7):666–679.
- Bonci D, Coppola V, Musumeci M, Addario A, Giuffrida R, Memeo L, D'Urso L, Pagliuca A, Biffoni M, Labbaye C, Bartucci M, Muto G, Peschle C, De Maria R (2008) The miR-15amiR-16-1 cluster controls prostate cancer by targeting multiple oncogenic activities. Nat Med 14(11):1271–1277
- Bos CL, Richel DJ, Ritsema T, Peppelenbosch MP, Versteeg HH (2004) Prostanoids and prostanoid receptors in signal transduction. Int J Biochem Cell Biol 36(7):1187–1205
- Bottoni A, Piccin D, Tagliati F, Luchin A, Zatelli MC, degli Uberti EC (2005) miR-15a and miR-16-1 down-regulation in pituitary adenomas. J Cell Physiol 204(1):280–285
- Brennan CM, Steitz JA (2001) HuR and mRNA stability. Cell Mol Life Sci 58:266-277
- Brennan SE, Kuwano Y, Alkharouf N, Blackshear PJ, Gorospe M, Wilson GM (2009) The mRNAdestabilizing protein tristetraprolin is suppressed in many cancers, altering tumorigenic phenotypes and patient prognosis. Cancer Res 69(12):5168–5176
- Brewer BY, Malicka J, Blackshear PJ, Wilson GM (2004) RNA sequence elements required for high affinity binding by the zinc finger domain of tristetraprolin: conformational changes coupled to the bipartite nature of Au-rich MRNA-destabilizing motifs. J Biol Chem 279(27):27870–27877
- Breyer MD, Breyer RM (2000) Prostaglandin E receptors and the kidney. Am J Physiol Renal Physiol 279(1):F12–F23
- Breyer MD, Breyer RM (2001) G protein-coupled prostanoid receptors and the kidney. Annu Rev Physiol 63:579–605
- Breyer MD, Jacobson HR, Breyer RM (1996) Functional and molecular aspects of renal prostaglandin receptors. J Am Soc Nephrol 7(1):8–17
- Breyer RM, Bagdassarian CK, Myers SA, Breyer MD (2001) Prostanoid receptors: subtypes and signaling. Annu Rev Pharmacol Toxicol 41:661–690
- Briata P, Ilengo C, Corte G, Moroni C, Rosenfeld MG, Chen CY, Gherzi R (2003) The Wnt/betacatenin→Pitx2 pathway controls the turnover of Pitx2 and other unstable mRNAs. Mol Cell 12(5):1201–1211
- Calin GA, Dumitru CD, Shimizu M, Bichi R, Zupo S, Noch E, Aldler H, Rattan S, Keating M, Rai K, Rassenti L, Kipps T, Negrini M, Bullrich F, Croce CM (2002) Frequent deletions and down-regulation of micro- RNA genes miR15 and miR16 at 13q14 in chronic lymphocytic leukemia. Proc Natl Acad Sci U S A 99(24):15524–15529

- Carballo E, Lai WS, Blackshear PJ (1998) Feedback inhibition of macrophage tumor necrosis factor-α production by tristetraprolin. Science 281(5379):1001–1005
- Carballo E, Lai WS, Blackshear PJ (2000) Evidence that tristetraprolin is a physiological regulator of granulocyte-macrophage colony-stimulating factor messenger RNA deadenylation and stability. Blood 95(6):1891–1899
- Cerella C, Sobolewski C, Dicato M, Diederich M (2010) Targeting COX-2 expression by natural compounds: a promising alternative strategy to synthetic COX-2 inhibitors for cancer chemoprevention and therapy. Biochem Pharmacol 80(12):1801–1815
- Chae MJ, Sung HY, Kim EH, Lee M, Kwak H, Chae CH, Kim S, Park WY (2009) Chemical inhibitors destabilize HuR binding to the AU-rich element of TNF-alpha mRNA. Exp Mol Med 41(11):824–831
- Chakrabarty A, Tranguch S, Daikoku T, Jensen K, Furneaux H, Dey SK (2007) MicroRNA regulation of cyclooxygenase-2 during embryo implantation. Proc Natl Acad Sci U S A 104(38):15144–15149
- Chan VS (2004) A mechanistic perspective on the specificity and extent of COX-2 inhibition in pregnancy. Drug Saf 27(7):421–426
- Chandramouli A, Onyeagucha BC, Mercado-Pimentel ME, Stankova L, Shahin NA, LaFleur BJ, Heimark RL, Bhattacharyya AK, Nelson MA (2012) MicroRNA-101 (miR-101) posttranscriptionally regulates the expression of EP4 receptor in colon cancers. Cancer Biol Ther 13(3):175–183
- Chandrasekharan NV, Simmons DL (2004) The cyclooxygenases. Genome Biol 5(9):241
- Cheadle C, Fan J, Cho-Chung YS, Werner T, Ray J, Do L, Gorospe M, Becker KG (2005) Control of gene expression during T cell activation: alternate regulation of mRNA transcription and mRNA stability. BMC Genomics 6(1):75
- Chen CY, Shyu AB (2011) Mechanisms of deadenylation-dependent decay. Wiley Interdiscip Rev RNA 2(2):167–183
- Chen CY, Gherzi R, Ong SE, Chan EL, Raijmakers R, Pruijn GJ, Stoecklin G, Moroni C, Mann M, Karin M (2001) AU binding proteins recruit the exosome to degrade ARE-containing mRNAs. Cell 107(4):451–464
- Chen HH, Chen TW, Lin H (2009) Prostacyclin-induced peroxisome proliferator-activated receptor-alpha translocation attenuates NF-kappaB and TNF-alpha activation after renal ischemia-reperfusion injury. Am J Physiol Renal Physiol 297(4):F1109–F1118
- Chen L, Wang X, Wang H, Li Y, Yan W, Han L, Zhang K, Zhang J, Wang Y, Feng Y, Pu P, Jiang T, Kang C, Jiang C (2012) miR-137 is frequently down-regulated in glioblastoma and is a negative regulator of Cox-2. Eur J Cancer 48(16):3104–3111
- Chen X, Pan M, Han L, Lu H, Hao X, Dong Q (2013) miR-338-3p suppresses neuroblastoma proliferation, invasion and migration through targeting PREX2a. FEBS Lett 587(22):3729–3737
- Ciais D, Cherradi N, Feige JJ (2013) Multiple functions of tristetraprolin/TIS11 RNA-binding proteins in the regulation of mRNA biogenesis and degradation. Cell Mol Life Sci 70(12):2031–2044
- Cimmino A, Calin GA, Fabbri M, Iorio MV, Ferracin M, Shimizu M, Wojcik SE, Aqeilan RI, Zupo S, Dono M, Rassenti L, Alder H, Volinia S, Liu CG, Kipps TJ, Negrini M, Croce CM (2005) miR-15 and miR-16 induce apoptosis by targeting BCL2. Proc Natl Acad Sci U S A 102(39):13944–13949
- Cok SJ, Morrison AR (2001) The 3'-untranslated region of murine cyclooxygenase-2 contains multiple regulatory elements that alter message stability and translational efficiency. J Biol Chem 276(25):23179–23185
- Cornett AL, Lutz CS (2014) Regulation of COX-2 expression by miR-146a in lung cancer cells. RNA 20(9):1419–1430
- D'Agostino VG, Adami V, Provenzani A (2013) A novel high throughput biochemical assay to evaluate the HuR protein-RNA complex formation. PLoS One 8(8):e72426–e72434
- Daikoku T, Hirota Y, Tranguch S, Joshi AR, DeMayo FJ, Lydon JP, Ellenson LH, Dey SK (2008) Conditional loss of uterine Pten unfailingly and rapidly induces endometrial cancer in mice. Cancer Res 68(14):5619–5627

- Day DA, Tuite MF (1998) Post-transcriptional gene regulatory mechanisms in eukaryotes: an overview. J Endocrinol 157(3):361–371
- de Oliveira AC, Candelario-Jalil E, Bhatia HS, Lieb K, Hull M, Fiebich BL (2008) Regulation of prostaglandin E2 synthase expression in activated primary rat microglia: evidence for uncoupled regulation of mPGES-1 and COX-2. Glia 56(8):844–855
- De Rensis F, Saleri R, Tummaruk P, Techakumphu M, Kirkwood RN (2012) Prostaglandin F2alpha and control of reproduction in female swine: a review. Theriogenology 77(1):1–11
- Deiters A (2010) Small molecule modifiers of the microRNA and RNA interference pathway. AAPS J 12(1):51–60
- Derry JM, Kerns JA, Francke U (1995) RBM3, a novel human gene in Xp11.23 with a putative RNA-binding domain. Hum Mol Genet 4(12):2307–2311
- Dixon DA (2003) Regulation of COX-2 expression in human cancer. Prog Exp Tumor Res 37:52–71
- Dixon DA (2004) Dysregulated post-transcriptional control of COX-2 gene expression in cancer. Curr Pharm Des 10(6):635–646
- Dixon DA, Kaplan CD, McIntyre TM, Zimmerman GA, Prescott SM (2000) Post-transcriptional control of cyclooxygenase-2 gene expression. The role of the 3'-untranslated region. J Biol Chem 275(16):11750–11757
- Dixon DA, Tolley ND, King PH, Nabors LB, McIntyre TM, Zimmerman GA, Prescott SM (2001) Altered expression of the mRNA stability factor HuR promotes cyclooxygenase-2 expression in colon cancer cells. J Clin Invest 108(11):1657–1665
- Dixon DA, Balch GC, Kedersha N, Anderson P, Zimmerman GA, Beauchamp RD, Prescott SM (2003) Regulation of cyclooxygenase-2 expression by the translational silencer TIA-1. J Exp Med 198(3):475–481
- Dixon DA, Tolley ND, Bemis-Standoli K, Martinez ML, Weyrich AS, Morrow JD, Prescott SM, Zimmerman GA (2006) Expression of COX-2 in platelet-monocyte interactions occurs via combinatorial regulation involving adhesion and cytokine signaling. J Clin Invest 116(10):2727–2738
- Dogne JM, Hanson J, Supuran C, Pratico D (2006) Coxibs and cardiovascular side-effects: from light to shadow. Curr Pharm Des 12(8):971–975
- Dohner H, Stilgenbauer S, Benner A, Leupolt E, Krober A, Bullinger L, Dohner K, Bentz M, Lichter P (2000) Genomic aberrations and survival in chronic lymphocytic leukemia. N Engl J Med 343(26):1910–1916
- Dorris SL, Peebles RS Jr (2012) PGI2 as a regulator of inflammatory diseases. Mediators Inflamm 2012:926968
- Eulalio A, Behm-Ansmant I, Izaurralde E (2007) P bodies: at the crossroads of post-transcriptional pathways. Nat Rev Mol Cell Biol 8(1):9–22
- Fabbri M, Garzon R, Andreeff M, Kantarjian HM, Garcia-Manero G, Calin GA (2008) MicroRNAs and noncoding RNAs in hematological malignancies: molecular, clinical and therapeutic implications. Leukemia 22(6):1095–1105
- Fabian MR, Sonenberg N, Filipowicz W (2010) Regulation of mRNA translation and stability by microRNAs. Annu Rev Biochem 79:351–379
- Fabian MR, Frank F, Rouya C, Siddiqui N, Lai WS, Karetnikov A, Blackshear PJ, Nagar B, Sonenberg N (2013) Structural basis for the recruitment of the human CCR4-NOT deadenylase complex by tristetraprolin. Nat Struct Mol Biol 20(6):735–739
- Fan XC, Steitz JA (1998) HNS, a nuclear-cytoplasmic shuttling sequence in HuR. Proc Natl Acad Sci U S A 95(26):15293–15298
- Fan J, Yang X, Wang W, Wood WH III, Becker KG, Gorospe M (2002) Global analysis of stress-regulated mRNA turnover by using cDNA arrays. Proc Natl Acad Sci U S A 99(16):10611–10616
- Farazi TA, Hoell JI, Morozov P, Tuschl T (2013) MicroRNAs in human cancer. Adv Exp Med Biol 774:1–20
- Fenger-Gron M, Fillman C, Norrild B, Lykke-Andersen J (2005) Multiple processing body factors and the ARE binding protein TTP activate mRNA decapping. Mol Cell 20(6):905–915

Frankel LB, Lund AH (2012) MicroRNA regulation of autophagy. Carcinogenesis 33(11):2018–2025

- Franks TM, Lykke-Andersen J (2007) TTP and BRF proteins nucleate processing body formation to silence mRNAs with AU-rich elements. Genes Dev 21(6):719–735
- Friedman RC, Farh KK, Burge CB, Bartel DP (2009) Most mammalian mRNAs are conserved targets of microRNAs. Genome Res 19(1):92–105
- Funakoshi Y, Doi Y, Hosoda N, Uchida N, Osawa M, Shimada I, Tsujimoto M, Suzuki T, Katada T, Hoshino S (2007) Mechanism of mRNA deadenylation: evidence for a molecular interplay between translation termination factor eRF3 and mRNA deadenylases. Genes Dev 21(23):3135–3148
- Funk CD (2001) Prostaglandins and leukotrienes: advances in eicosanoid biology. Science 294(5548):1871–1875
- Funk CD, FitzGerald GA (2007) COX-2 inhibitors and cardiovascular risk. J Cardiovasc Pharmacol 50(5):470–479
- Gao W, Shen H, Liu L, Xu J, Shu Y (2011) MiR-21 overexpression in human primary squamous cell lung carcinoma is associated with poor patient prognosis. J Cancer Res Clin Oncol 137(4):557–566
- Garneau NL, Wilusz J, Wilusz CJ (2007) The highways and byways of mRNA decay. Nat Rev Mol Cell Biol 8(2):113–126
- Garzon R, Marcucci G, Croce CM (2010) Targeting microRNAs in cancer: rationale, strategies and challenges. Nat Rev Drug Discov 9(10):775–789
- Ge W, Yu DC, Li QG, Chen X, Zhang CY, Ding YT (2014) Expression of serum miR-16, let-7f, and miR-21 in patients with hepatocellular carcinoma and their clinical significances. Clin Lab 60(3):427–434
- Ghose J, Sinha M, Das E, Jana NR, Bhattacharyya NP (2011) Regulation of miR-146a by RelA/ NFkB and p53 in STHdh(Q111)/Hdh(Q111) cells, a cell model of Huntington's disease. PLoS One 6(8):e23837
- Gilks N, Kedersha N, Ayodele M, Shen L, Stoecklin G, Dember LM, Anderson P (2004) Stress granule assembly is mediated by prion-like aggregation of TIA-1. Mol Biol Cell 15(12):5383–5398
- Goldstrohm AC, Wickens M (2008) Multifunctional deadenylase complexes diversify mRNA control. Nat Rev Mol Cell Biol 9(4):337–344
- Graves P, Zeng Y (2012) Biogenesis of mammalian microRNAs: a global view. Genomics Proteomics Bioinformatics 10(5):239–245
- Gregory PA, Bert AG, Paterson EL, Barry SC, Tsykin A, Farshid G, Vadas MA, Khew-Goodall Y, Goodall GJ (2008) The miR-200 family and miR-205 regulate epithelial to mesenchymal transition by targeting ZEB1 and SIP1. Nat Cell Biol 10(5):593–601
- Gruber AR, Fallmann J, Kratochvill F, Kovarik P, Hofacker IL (2010) AREsite: a database for the comprehensive investigation of AU-rich elements. Nucleic Acids Res 39(Database issue):D66–D69
- Guan H, Wei G, Wu J, Fang D, Liao Z, Xiao H, Li M, Li Y (2013) Down-regulation of miR-218-2 and its host gene SLIT3 cooperate to promote invasion and progression of thyroid cancer. J Clin Endocrinol Metab 98(8):E1334–E1344
- Guo H, Ingolia NT, Weissman JS, Bartel DP (2010) Mammalian microRNAs predominantly act to decrease target mRNA levels. Nature 466(7308):835–840
- Guo J, Xia B, Meng F, Lou G (2013) miR-137 suppresses cell growth in ovarian cancer by targeting AEG-1. Biochem Biophys Res Commun 441(2):357–363
- Hao Y, Gu X, Zhao Y, Greene S, Sha W, Smoot DT, Califano J, Wu TC, Pang X (2011) Enforced expression of miR-101 inhibits prostate cancer cell growth by modulating the COX-2 pathway in vivo. Cancer Prev Res (Phila) 4(7):1073–1083
- Harada N, Okajima K, Murakami K, Isobe H, Liu W (1999) Gastric prostacyclin (PGI2) prevents stress-induced gastric mucosal injury in rats primarily by inhibiting leukocyte activation. Prostaglandins Other Lipid Mediat 57(5–6):291–303
- Hatae N, Sugimoto Y, Ichikawa A (2002) Prostaglandin receptors: advances in the study of EP3 receptor signaling. J Biochem 131(6):781–784

- Hau HH, Walsh RJ, Ogilvie RL, Williams DA, Reilly CS, Bohjanen PR (2007) Tristetraprolin recruits functional mRNA decay complexes to ARE sequences. J Cell Biochem 100(6):1477–1492
- He XP, Shao Y, Li XL, Xu W, Chen GS, Sun HH, Xu HC, Xu X, Tang D, Zheng XF, Xue YP, Huang GC, Sun WH (2012) Downregulation of miR-101 in gastric cancer correlates with cyclooxygenase-2 overexpression and tumor growth. FEBS J 279(22):4201–4212
- Hill JM, Zhao Y, Clement C, Neumann DM, Lukiw WJ (2009) HSV-1 infection of human brain cells induces miRNA-146a and Alzheimer-type inflammatory signaling. Neuroreport 20(16):1500–1505
- Hinman MN, Lou H (2008) Diverse molecular functions of Hu proteins. Cell Mol Life Sci 65(20):3168–3181
- Hinz B, Brune K (2002) Cyclooxygenase-2--10 years later. J Pharmacol Exp Ther 300(2):367–375
- Hirai H, Tanaka K, Yoshie O, Ogawa K, Kenmotsu K, Takamori Y, Ichimasa M, Sugamura K, Nakamura M, Takano S, Nagata K (2001) Prostaglandin D2 selectively induces chemotaxis in T helper type 2 cells, eosinophils, and basophils via seven-transmembrane receptor CRTH2. J Exp Med 193(2):255–261
- Hiraoka M, Abe A, Shayman JA (2005) Structure and function of lysosomal phospholipase A2: identification of the catalytic triad and the role of cysteine residues. J Lipid Res 46(11):2441–2447
- Hiroki E, Akahira J, Suzuki F, Nagase S, Ito K, Suzuki T, Sasano H, Yaegashi N (2010) Changes in microRNA expression levels correlate with clinicopathological features and prognoses in endometrial serous adenocarcinomas. Cancer Sci 101(1):241–249
- Holla VR, Backlund MG, Yang P, Newman RA, DuBois RN (2008) Regulation of prostaglandin transporters in colorectal neoplasia. Cancer Prev Res (Phila) 1(2):93–99
- Horton EW, Thompson CJ (1964) Thin-layer chromatography and bioassay of prostaglandins in extracts of semen and tissues of the male reproductive tract. Br J Pharmacol Chemother 22:183–188
- Hsu SD, Tseng YT, Shrestha S, Lin YL, Khaleel A, Chou CH, Chu CF, Huang HY, Lin CM, Ho SY, Jian TY, Lin FM, Chang TH, Weng SL, Liao KW, Liao IE, Liu CC, Huang HD (2014) miRTarBase update 2014: an information resource for experimentally validated miRNA-target interactions. Nucleic Acids Res 42(Database issue):D78–D85
- Hu H, Zhang Y, Cai XH, Huang JF, Cai L (2012) Changes in microRNA expression in the MG-63 osteosarcoma cell line compared with osteoblasts. Oncol Lett 4(5):1037–1042
- Huang F, Lin C, Shi YH, Kuerban G (2013) MicroRNA-101 inhibits cell proliferation, invasion, and promotes apoptosis by regulating cyclooxygenase-2 in Hela cervical carcinoma cells. Asian Pac J Cancer Prev 14(10):5915–5920
- Hull MA, Ko SC, Hawcroft G (2004) Prostaglandin EP receptors: targets for treatment and prevention of colorectal cancer? Mol Cancer Ther 3(8):1031–1039
- Inoue H, Taba Y, Miwa Y, Yokota C, Miyagi M, Sasaguri T (2002) Transcriptional and posttranscriptional regulation of cyclooxygenase-2 expression by fluid shear stress in vascular endothelial cells. Arterioscler Thromb Vasc Biol 22(9):1415–1420
- Ito S, Sakamoto K, Mochizuki-Oda N, Ezashi T, Miwa K, Okuda-Ashitaka E, Shevchenko VI, Kiso Y, Hayaishi O (1994) Prostaglandin F2 alpha receptor is coupled to Gq in cDNAtransfected Chinese hamster ovary cells. Biochem Biophys Res Commun 200(2):756–762
- Ji Y, He Y, Liu L, Zhong X (2010) MiRNA-26b regulates the expression of cyclooxygenase-2 in desferrioxamine-treated CNE cells. FEBS Lett 584(5):961–967
- Jing Q, Huang S, Guth S, Zarubin T, Motoyama A, Chen J, Di Padova F, Lin SC, Gram H, Han J (2005) Involvement of microRNA in AU-rich element-mediated mRNA instability. Cell 120(5):623–634
- Kanai N, Lu R, Satriano JA, Bao Y, Wolkoff AW, Schuster VL (1995) Identification and characterization of a prostaglandin transporter. Science 268(5212):866–869
- Kanaoka Y, Urade Y (2003) Hematopoietic prostaglandin D synthase. Prostaglandins Leukot Essent Fatty Acids 69(2–3):163–167

- Kang HJ, Hwang SJ, Yoon JA, Jun JH, Lim HJ, Yoon TK, Song H (2011) Activation of peroxisome proliferators-activated receptor delta (PPARdelta) promotes blastocyst hatching in mice. Mol Hum Reprod 17(10):653–660
- Kawahara K, Hohjoh H, Inazumi T, Tsuchiya S, Sugimoto Y (2014) Prostaglandin E-induced inflammation: relevance of prostaglandin E receptors. Biochim Biophys Acta. doi:10.1016/j. bbalip.2014.07.008
- Kawamori T, Uchiya N, Nakatsugi S, Watanabe K, Ohuchida S, Yamamoto H, Maruyama T, Kondo K, Sugimura T, Wakabayashi K (2001) Chemopreventive effects of ONO-8711, a selective prostaglandin E receptor EP(1) antagonist, on breast cancer development. Carcinogenesis 22(12):2001–2004
- Kedersha N, Anderson P (2007) Mammalian stress granules and processing bodies. Methods Enzymol 431:61–81
- Kedersha N, Stoecklin G, Ayodele M, Yacono P, Lykke-Andersen J, Fritzler MJ, Scheuner D, Kaufman RJ, Golan DE, Anderson P (2005) Stress granules and processing bodies are dynamically linked sites of mRNP remodeling. J Cell Biol 169(6):871–884
- Keene J (1999) Why is Hu where? Shuttling of early-response messenger RNA subsets. Proc Natl Acad Sci U S A 96(1):5–7
- Keith RL, Geraci MW, Nana-Sinkam SP, Breyer RM, Hudish TM, Meyer AM, Malkinson AM, Dwyer-Nield LD (2006) Prostaglandin E2 receptor subtype 2 (EP2) null mice are protected against murine lung tumorigenesis. Anticancer Res 26(4B):2857–2861
- Kennedy TG, Gillio-Meina C, Phang SH (2007) Prostaglandins and the initiation of blastocyst implantation and decidualization. Reproduction 134(5):635–643
- Kojima S, Enokida H, Yoshino H, Itesako T, Chiyomaru T, Kinoshita T, Fuse M, Nishikawa R, Goto Y, Naya Y, Nakagawa M, Seki N (2014) The tumor-suppressive microRNA-143/145 cluster inhibits cell migration and invasion by targeting GOLM1 in prostate cancer. J Hum Genet 59(2):78–87
- Konno Y, Dong P, Xiong Y, Suzuki F, Lu J, Cai M, Watari H, Mitamura T, Hosaka M, Hanley SJ, Kudo M, Sakuragi N (2014) MicroRNA-101 targets EZH2, MCL-1 and FOS to suppress proliferation, invasion and stem cell-like phenotype of aggressive endometrial cancer cells. Oncotarget 5(15):6049–6062
- Kotani M, Tanaka I, Ogawa Y, Usui T, Tamura N, Mori K, Narumiya S, Yoshimi T, Nakao K (1997) Structural organization of the human prostaglandin EP3 receptor subtype gene (PTGER3). Genomics 40(3):425–434
- Kudo I, Murakami M (2005) Prostaglandin E synthase, a terminal enzyme for prostaglandin E2 biosynthesis. J Biochem Mol Biol 38(6):633–638
- Kulkarni M, Ozgur S, Stoecklin G (2010) On track with P-bodies. Biochem Soc Trans 38(Pt 1): 242–251
- Kunori S, Matsumura S, Mabuchi T, Tatsumi S, Sugimoto Y, Minami T, Ito S (2009) Involvement of prostaglandin F 2 alpha receptor in ATP-induced mechanical allodynia. Neuroscience 163(1):362–371
- Kurumbail RG, Stevens AM, Gierse JK, McDonald JJ, Stegeman RA, Pak JY, Gildehaus D, Miyashiro JM, Penning TD, Seibert K, Isakson PC, Stallings WC (1996) Structural basis for selective inhibition of cyclooxygenase-2 by anti-inflammatory agents. Nature 384(6610): 644–648
- Lai WS, Carballo E, Strum JR, Kennington EA, Phillips RS, Blackshear PJ (1999) Evidence that tristetraprolin binds to AU-rich elements and promotes the deadenylation and destabilization of tumor necrosis factor alpha mRNA. Mol Cell Biol 19(6):4311–4323
- Lal A, Mazan-Mamczarz K, Kawai T, Yang X, Martindale JL, Gorospe M (2004) Concurrent versus individual binding of HuR and AUF1 to common labile target mRNAs. EMBO J 23(15):3092–3102
- Li J, Kong X, Zhang J, Luo Q, Li X, Fang L (2013) MiRNA-26b inhibits proliferation by targeting PTGS2 in breast cancer. Cancer Cell Int 13(1):7
- Li P, Ma L, Zhang Y, Ji F, Jin F (2014) MicroRNA-137 down-regulates KIT and inhibits small cell lung cancer cell proliferation. Biomed Pharmacother 68(1):7–12

- Liang X, Liu Y, Zeng L, Yu C, Hu Z, Zhou Q, Yang Z (2014) miR-101 inhibits the G1-to-S phase transition of cervical cancer cells by targeting Fos. Int J Gynecol Cancer 24(7):1165–1172
- Liao WL, Wang WC, Chang WC, Tseng JT (2011) The RNA-binding protein HuR stabilizes cytosolic phospholipase A2alpha mRNA under interleukin-1beta treatment in non-small cell lung cancer A549 Cells. J Biol Chem 286(41):35499–35508
- Lim H, Gupta RA, Ma WG, Paria BC, Moller DE, Morrow JD, DuBois RN, Trzaskos JM, Dey SK (1999) Cyclo-oxygenase-2-derived prostacyclin mediates embryo implantation in the mouse via PPARdelta. Genes Dev 13(12):1561–1574
- Lin C, Huang F, Zhang YJ, Tuokan T, Kuerban G (2014) Roles of MiR-101 and its target gene Cox-2 in early diagnosis of cervical cancer in Uygur women. Asian Pac J Cancer Prev 15(1):45–48
- Ling HY, Hu B, Hu XB, Zhong J, Feng SD, Qin L, Liu G, Wen GB, Liao DF (2012) MiRNA-21 reverses high glucose and high insulin induced insulin resistance in 3T3-L1 adipocytes through targeting phosphatase and tensin homologue. Exp Clin Endocrinol Diabetes 120(9):553–559
- Linker K, Pautz A, Fechir M, Hubrich T, Greeve J, Kleinert H (2005) Involvement of KSRP in the post-transcriptional regulation of human iNOS expression-complex interplay of KSRP with TTP and HuR. Nucleic Acids Res 33(15):4813–4827
- Liu Q, Fu H, Sun F, Zhang H, Tie Y, Zhu J, Xing R, Sun Z, Zheng X (2008) miR-16 family induces cell cycle arrest by regulating multiple cell cycle genes. Nucleic Acids Res 36(16):5391–5404
- Liu J, Chen G, Feng L, Zhang W, Pelicano H, Wang F, Ogasawara MA, Lu W, Amin HM, Croce CM, Keating MJ, Huang P (2014) Loss of p53 and altered miR15-a/16-1 MCL-1 pathway in CLL: insights from TCL1-Tg:p53(-/-) mouse model and primary human leukemia cells. Leukemia 28(1):118–128
- Lopez de Silanes I, Zhan M, Lal A, Yang X, Gorospe M (2004) Identification of a target RNA motif for RNA-binding protein HuR. Proc Natl Acad Sci U S A 101(9):2987–2992
- Lopez de Silanes I, Galban S, Martindale JL, Yang X, Mazan-Mamczarz K, Indig FE, Falco G, Zhan M, Gorospe M (2005) Identification and functional outcome of mRNAs associated with RNA-binding protein TIA-1. Mol Cell Biol 25(21):9520–9531
- Lopez de Silanes I, Quesada MP, Esteller M (2007) Aberrant regulation of messenger RNA 3'-untranslated region in human cancer. Cell Oncol 29(1):1–17
- Lu J, Getz G, Miska EA, Alvarez-Saavedra E, Lamb J, Peck D, Sweet-Cordero A, Ebert BL, Mak RH, Ferrando AA, Downing JR, Jacks T, Horvitz HR, Golub TR (2005) MicroRNA expression profiles classify human cancers. Nature 435(7043):834–838
- Lu L, Byrnes K, Han C, Wang Y, Wu T (2014) miR-21 targets 15-PGDH and promotes cholangiocarcinoma growth. Mol Cancer Res 12(6):890–900
- Luo X, Burwinkel B, Tao S, Brenner H (2011) MicroRNA signatures: novel biomarker for colorectal cancer? Cancer Epidemiol Biomarkers Prev 20(7):1272–1286
- Lutz CS, Cornett AL (2013) Regulation of genes in the arachidonic acid metabolic pathway by RNA processing and RNA-mediated mechanisms. Wiley Interdiscip Rev RNA 4(5):593–605
- Lykke-Andersen J, Wagner E (2005) Recruitment and activation of mRNA decay enzymes by two ARE-mediated decay activation domains in the proteins TTP and BRF-1. Genes Dev 19(3):351–361
- Ma W-J, Chung S, Furneaux H (1997) The ELAV-like proteins bind to AU-rich elements and to the poly(A) tail of mRNA. Nucleic Acids Res 25:3564–3569
- Ma X, Becker Buscaglia LE, Barker JR, Li Y (2011) MicroRNAs in NF-kappaB signaling. J Mol Cell Biol 3(3):159–166
- Malki S, Nef S, Notarnicola C, Thevenet L, Gasca S, Mejean C, Berta P, Poulat F, Boizet-Bonhoure B (2005) Prostaglandin D2 induces nuclear import of the sex-determining factor SOX9 via its cAMP-PKA phosphorylation. EMBO J 24(10):1798–1809
- Matsumoto T, Sagawa N, Yoshida M, Mori T, Tanaka I, Mukoyama M, Kotani M, Nakao K (1997) The prostaglandin E2 and F2 alpha receptor genes are expressed in human myometrium and are down-regulated during pregnancy. Biochem Biophys Res Commun 238(3):838–841
- Matsuoka T, Hirata M, Tanaka H, Takahashi Y, Murata T, Kabashima K, Sugimoto Y, Kobayashi T, Ushikubi F, Aze Y, Eguchi N, Urade Y, Yoshida N, Kimura K, Mizoguchi A, Honda Y,

Nagai H, Narumiya S (2000) Prostaglandin D2 as a mediator of allergic asthma. Science 287(5460):2013–2017

- Matysiak M, Fortak-Michalska M, Szymanska B, Orlowski W, Jurewicz A, Selmaj K (2013) MicroRNA-146a negatively regulates the immunoregulatory activity of bone marrow stem cells by targeting prostaglandin E2 synthase-2. J Immunol 190(10):5102–5109
- McAdam BF, Catella-Lawson F, Mardini IA, Kapoor S, Lawson JA, FitzGerald GA (1999) Systemic biosynthesis of prostacyclin by cyclooxygenase (COX)-2: the human pharmacology of a selective inhibitor of COX-2. Proc Natl Acad Sci U S A 96(1):272–277
- Meisner NC, Filipowicz W (2010) Properties of the regulatory RNA-binding protein HuR and its role in controlling miRNA repression. Adv Exp Med Biol 700:106–123
- Meisner NC, Hintersteiner M, Mueller K, Bauer R, Seifert JM, Naegeli HU, Ottl J, Oberer L, Guenat C, Moss S, Harrer N, Woisetschlaeger M, Buehler C, Uhl V, Auer M (2007) Identification and mechanistic characterization of low-molecular-weight inhibitors for HuR. Nat Chem Biol 3(8):508–515
- Ming XF, Stoecklin G, Lu M, Looser R, Moroni C (2001) Parallel and independent regulation of interleukin-3 mRNA turnover by phosphatidylinositol 3-kinase and p38 mitogen-activated protein kinase. Mol Cell Biol 21(17):5778–5789
- Mongroo PS, Rustgi AK (2010) The role of the miR-200 family in epithelial-mesenchymal transition. Cancer Biol Ther 10(3):219–222
- Montenegro D, Romero R, Kim SS, Tarca AL, Draghici S, Kusanovic JP, Kim JS, Lee DC, Erez O, Gotsch F, Hassan SS, Kim CJ (2009) Expression patterns of microRNAs in the chorioamniotic membranes: a role for microRNAs in human pregnancy and parturition. J Pathol 217(1):113–121
- Moore AE, Young LE, Dixon DA (2011) MicroRNA and AU-rich element regulation of prostaglandin synthesis. Cancer Metastasis Rev 30(3–4):419–435
- Moore AE, Young LE, Dixon DA (2012) A common single-nucleotide polymorphism in cyclooxygenase-2 disrupts microRNA-mediated regulation. Oncogene 31(12):1592–1598
- Moraes KC, Monteiro CJ, Pacheco-Soares C (2013) A novel function for CUGBP2 in controlling the pro-inflammatory stimulus in H9c2 cells: subcellular trafficking of messenger molecules. Cell Biol Int 37(10):1129–1138
- Mukherjee D, Gao M, O'Connor JP, Raijmakers R, Pruijn G, Lutz CS, Wilusz J (2002) The mammalian exosome mediates the efficient degradation of mRNAs that contain AU-rich elements. EMBO J 21(1–2):165–174
- Mukhopadhyay D, Houchen CW, Kennedy S, Dieckgraefe BK, Anant S (2003) Coupled mRNA stabilization and translational silencing of cyclooxygenase-2 by a novel RNA binding protein, CUGBP2. Mol Cell 11(1):113–126
- Murakami M, Kudo I (2002) Phospholipase A2. J Biochem 131(3):285-292
- Murakami Y, Toyoda H, Tanaka M, Kuroda M, Harada Y, Matsuda F, Tajima A, Kosaka N, Ochiya T, Shimotohno K (2011) The progression of liver fibrosis is related with overexpression of the miR-199 and 200 families. PLoS One 6(1):e16081
- Murase T, Misawa K, Minegishi Y, Aoki M, Ominami H, Suzuki Y, Shibuya Y, Hase T (2011) Coffee polyphenols suppress diet-induced body fat accumulation by downregulating SREBP-1c and related molecules in C57BL/6J mice. Am J Physiol Endocrinol Metab 300(1):E122–E133
- Murmu N, Jung J, Mukhopadhyay D, Houchen CW, Riehl TE, Stenson WF, Morrison AR, Arumugam T, Dieckgraefe BK, Anant S (2004) Dynamic antagonism between RNA-binding protein CUGBP2 and cyclooxygenase-2-mediated prostaglandin E2 in radiation damage. Proc Natl Acad Sci U S A 101(38):13873–13878
- Namba T, Sugimoto Y, Negishi M, Irie A, Ushikubi F, Kakizuka A, Ito S, Ichikawa A, Narumiya S (1993) Alternative splicing of C-terminal tail of prostaglandin E receptor subtype EP3 determines G-protein specificity. Nature 365(6442):166–170
- Narumiya S, FitzGerald GA (2001) Genetic and pharmacological analysis of prostanoid receptor function. J Clin Invest 108(1):25–30
- Narumiya S, Sugimoto Y, Ushikubi F (1999) Prostanoid receptors: structures, properties, and functions. Physiol Rev 79(4):1193–1226

- Navarro A, Diaz T, Gallardo E, Vinolas N, Marrades RM, Gel B, Campayo M, Quera A, Bandres E, Garcia-Foncillas J, Ramirez J, Monzo M (2011) Prognostic implications of miR-16 expression levels in resected non-small-cell lung cancer. J Surg Oncol 103(5):411–415
- Nossent AY, Hansen JL, Doggen C, Quax PH, Sheikh SP, Rosendaal FR (2011) SNPs in microRNA binding sites in 3'-UTRs of RAAS genes influence arterial blood pressure and risk of myocardial infarction. Am J Hypertens 24(9):999–1006
- Ogawa K, Chen F, Kim YJ, Chen Y (2003) Transcriptional regulation of tristetraprolin by transforming growth factor-beta in human T cells. J Biol Chem 278(32):30373–30381
- Osawa T, Ohga N, Hida Y, Kitayama K, Akiyama K, Onodera Y, Fujie M, Shinohara N, Shindoh M, Nonomura K, Hida K (2012) Prostacyclin receptor in tumor endothelial cells promotes angiogenesis in an autocrine manner. Cancer Sci 103(6):1038–1044
- Pan X, Wang ZX, Wang R (2011) MicroRNA-21: a novel therapeutic target in human cancer. Cancer Biol Ther 10(12):1224–1232
- Park JY, Pillinger MH, Abramson SB (2006) Prostaglandin E2 synthesis and secretion: the role of PGE2 synthases. Clin Immunol 119(3):229–240
- Park SJ, Cheon EJ, Kim HA (2013) MicroRNA-558 regulates the expression of cyclooxygenase-2 and IL-1beta-induced catabolic effects in human articular chondrocytes. Osteoarthritis Cartilage 21(7):981–989
- Pekarsky Y, Croce CM (2014) Role of miR-15/16 in CLL. Cell Death Differ. doi:10.1038/ cdd.2014.87
- Peng Y, Liu YM, Li LC, Wang LL, Wu XL (2014) MicroRNA-338 inhibits growth, invasion and metastasis of gastric cancer by targeting NRP1 expression. PLoS One 9(4):e94422
- Pfeilschifter J, Schalkwijk C, Briner VA, van den Bosch H (1993) Cytokine-stimulated secretion of group II phospholipase A2 by rat mesangial cells. Its contribution to arachidonic acid release and prostaglandin synthesis by cultured rat glomerular cells. J Clin Invest 92(5):2516–2523
- Pham H, Rodriguez CE, Donald GW, Hertzer KM, Jung XS, Chang HH, Moro A, Reber HA, Hines OJ, Eibl G (2013) miR-143 decreases COX-2 mRNA stability and expression in pancreatic cancer cells. Biochem Biophys Res Commun 439(1):6–11
- Phillips K, Kedersha N, Shen L, Blackshear PJ, Anderson P (2004) Arthritis suppressor genes TIA-1 and TTP dampen the expression of tumor necrosis factor alpha, cyclooxygenase 2, and inflammatory arthritis. Proc Natl Acad Sci U S A 101(7):2011–2016
- Pniewska E, Pawliczak R (2013) The involvement of phospholipases A2 in asthma and chronic obstructive pulmonary disease. Mediators Inflamm 2013:793505
- Prasad S, Gupta SC, Tyagi AK, Aggarwal BB (2014) Curcumin, a component of golden spice: from bedside to bench and back. Biotechnol Adv 32(6):1053–1064
- Ramsay RG, Ciznadija D, Vanevski M, Mantamadiotis T (2003) Transcriptional regulation of cyclo-oxygenase expression: three pillars of control. Int J Immunopathol Pharmacol 16(2 Suppl):59–67
- Rayner KJ, Esau CC, Hussain FN, McDaniel AL, Marshall SM, van Gils JM, Ray TD, Sheedy FJ, Goedeke L, Liu X, Khatsenko OG, Kaimal V, Lees CJ, Fernandez-Hernando C, Fisher EA, Temel RE, Moore KJ (2011) Inhibition of miR-33a/b in non-human primates raises plasma HDL and lowers VLDL triglycerides. Nature 478(7369):404–407
- Raz A, Wyche A, Needleman P (1989) Temporal and pharmacological division of fibroblast cyclooxygenase expression into transcriptional and translational phases. Proc Natl Acad Sci U S A 86(5):1657–1661
- Rebecchi MJ, Pentyala SN (2000) Structure, function, and control of phosphoinositide-specific phospholipase C. Physiol Rev 80(4):1291–1335
- Regan JW (2003) EP2 and EP4 prostanoid receptor signaling. Life Sci 74(2-3):143-153
- Ricciotti E, FitzGerald GA (2011) Prostaglandins and inflammation. Arterioscler Thromb Vasc Biol 31(5):986–1000
- Ristimaki A (2004) Cyclooxygenase 2: from inflammation to carcinogenesis. Novartis Found Symp 256:215–221
- Saito O, Guan Y, Qi Z, Davis LS, Komhoff M, Sugimoto Y, Narumiya S, Breyer RM, Breyer MD (2003) Expression of the prostaglandin F receptor (FP) gene along the mouse genitourinary tract. Am J Physiol Renal Physiol 284(6):F1164–F1170

Saleem S, Ahmad AS, Maruyama T, Narumiya S, Dore S (2009) PGF(2alpha) FP receptor contributes to brain damage following transient focal brain ischemia. Neurotox Res 15(1):62–70

Salleh N (2014) Diverse roles of prostaglandins in blastocyst implantation. Sci World J 2014:968141

- Sampath D, Liu C, Vasan K, Sulda M, Puduvalli VK, Wierda WG, Keating MJ (2012) Histone deacetylases mediate the silencing of miR-15a, miR-16, and miR-29b in chronic lymphocytic leukemia. Blood 119(5):1162–1172
- Samuelsson B (1963) Isolation and identification of prostaglandins from human seminal plasma. 18. Prostaglandins and related factors. J Biol Chem 238:3229–3234
- Sanchez-Beato M, Sanchez-Aguilera A, Piris MA (2003) Cell cycle deregulation in B-cell lymphomas. Blood 101(4):1220–1235
- Sanduja S, Kaza V, Dixon DA (2009) The mRNA decay factor tristetraprolin (TTP) induces senescence in human papillomavirus-transformed cervical cancer cells by targeting E6-AP ubiquitin ligase. Aging 1(9):803–817
- Sanduja S, Blanco FF, Dixon DA (2011) The roles of TTP and BRF proteins in regulated mRNA decay. Wiley Interdiscip Rev RNA 2(1):42–57
- Sanduja S, Blanco FF, Young LE, Kaza V, Dixon DA (2012) The role of tristetraprolin in cancer and inflammation. Front Biosci 17:174–188
- Sato T, Liu X, Nelson A, Nakanishi M, Kanaji N, Wang X, Kim M, Li Y, Sun J, Michalski J, Patil A, Basma H, Holz O, Magnussen H, Rennard SI (2010) Reduced miR-146a increases prostaglandin E(2)in chronic obstructive pulmonary disease fibroblasts. Am J Respir Crit Care Med 182(8):1020–1029
- Sawaoka H, Dixon DA, Oates JA, Boutaud O (2003) Tristetrapolin binds to the 3' untranslated region of cyclooxygenase-2 mRNA: a polyadenylation variant in a cancer cell line lacks the binding site. J Biol Chem 278(16):13928–13935
- Schmidt L, Ceglarek U, Kortz L, Hoop M, Kirkby KC, Thiery J, Himmerich H (2013) Mechanisms of involvement of eicosanoids and their precursors in the pathophysiology and treatment of schizophrenia. Med Chem 9(6):763–773
- Schoenberg DR, Maquat LE (2012) Regulation of cytoplasmic mRNA decay. Nat Rev Genet 13(4):246–259
- Schuster VL (2002) Prostaglandin transport. Prostaglandins Other Lipid Mediat 68-69:633-647
- Sengupta S, Jang BC, Wu MT, Paik JH, Furneaux H, Hla T (2003) The RNA-binding protein HuR regulates the expression of cyclooxygenase-2. J Biol Chem 278(27):25227–25233
- Shanmugam N, Reddy MA, Natarajan R (2008) Distinct roles of heterogeneous nuclear ribonuclear protein K and microRNA-16 in cyclooxygenase-2 RNA stability induced by S100b, a ligand of the receptor for advanced glycation end products. J Biol Chem 283(52): 36221–36233
- Shen G, Lin Y, Yang X, Zhang J, Xu Z, Jia H (2014) MicroRNA-26b inhibits epithelialmesenchymal transition in hepatocellular carcinoma by targeting USP9X. BMC Cancer 14:393
- Shin VY, Jin H, Ng EK, Cheng AS, Chong WW, Wong CY, Leung WK, Sung JJ, Chu KM (2011) NF-kappaB targets miR-16 and miR-21 in gastric cancer: involvement of prostaglandin E receptors. Carcinogenesis 32(2):240–245
- Simmons DL, Botting RM, Hla T (2004) Cyclooxygenase isozymes: the biology of prostaglandin synthesis and inhibition. Pharmacol Rev 56(3):387–437
- Smith WL, DeWitt DL, Garavito RM (2000) Cyclooxygenases: structural, cellular, and molecular biology. Annu Rev Biochem 69:145–182
- Song T, Zhang X, Wang C, Wu Y, Dong J, Gao J, Cai W, Hong B (2011) Expression of miR-143 reduces growth and migration of human bladder carcinoma cells by targeting cyclooxygenase-2. Asian Pac J Cancer Prev 12(4):929–933
- Srikantan S, Gorospe M (2011) UneCLIPsing HuR nuclear function. Mol Cell 43(3):319-321
- Srikantan S, Gorospe M (2012) HuR function in disease. Front Biosci (Landmark Ed) 17:189–205
- Srikantan S, Tominaga K, Gorospe M (2012) Functional interplay between RNA-binding protein HuR and microRNAs. Curr Protein Pept Sci 13(4):372–379
- Steffel J, Luscher TF, Ruschitzka F, Tanner FC (2006) Cyclooxygenase-2 inhibition and coagulation. J Cardiovasc Pharmacol 47(Suppl 1):S15–S20

- Stenvang J, Petri A, Lindow M, Obad S, Kauppinen S (2012) Inhibition of microRNA function by antimiR oligonucleotides. Silence 3(1):1
- Stouffer RL, Xu F, Duffy DM (2007) Molecular control of ovulation and luteinization in the primate follicle. Front Biosci 12:297–307
- Strillacci A, Griffoni C, Sansone P, Paterini P, Piazzi G, Lazzarini G, Spisni E, Pantaleo MA, Biasco G, Tomasi V (2009) MiR-101 downregulation is involved in cyclooxygenase-2 overexpression in human colon cancer cells. Exp Cell Res 315(8):1439–1447
- Stumpo DJ, Lai WS, Blackshear PJ (2010) Inflammation: cytokines and RNA-based regulation. Wiley Interdiscip Rev RNA 1(1):60–80
- Su H, Yang JR, Xu T, Huang J, Xu L, Yuan Y, Zhuang SM (2009) MicroRNA-101, down-regulated in hepatocellular carcinoma, promotes apoptosis and suppresses tumorigenicity. Cancer Res 69(3):1135–1142
- Sugimoto Y, Narumiya S (2007) Prostaglandin E receptors. J Biol Chem 282(16):11613-11617
- Sugimoto Y, Yamasaki A, Segi E, Tsuboi K, Aze Y, Nishimura T, Oida H, Yoshida N, Tanaka T, Katsuyama M, Hasumoto K, Murata T, Hirata M, Ushikubi F, Negishi M, Ichikawa A, Narumiya S (1997) Failure of parturition in mice lacking the prostaglandin F receptor. Science 277(5326):681–683
- Sun K, Su G, Deng H, Dong J, Lei S, Li G (2014) Relationship between miRNA-338-3p expression and progression and prognosis of human colorectal carcinoma. Chin Med J 127(10): 1884–1890
- Sureban SM, Ramalingam S, Natarajan G, May R, Subramaniam D, Bishnupuri KS, Morrison AR, Dieckgraefe BK, Brackett DJ, Postier RG, Houchen CW, Anant S (2008) Translation regulatory factor RBM3 is a proto-oncogene that prevents mitotic catastrophe. Oncogene 27(33):4544–4556
- Taganov KD, Boldin MP, Chang KJ, Baltimore D (2006) NF-kappaB-dependent induction of microRNA miR-146, an inhibitor targeted to signaling proteins of innate immune responses. Proc Natl Acad Sci U S A 103(33):12481–12486
- Takayama K, Yuhki K, Ono K, Fujino T, Hara A, Yamada T, Kuriyama S, Karibe H, Okada Y, Takahata O, Taniguchi T, Iijima T, Iwasaki H, Narumiya S, Ushikubi F (2005) Thromboxane A2 and prostaglandin F2alpha mediate inflammatory tachycardia. Nat Med 11(5):562–566
- Takeuchi K (2010) Prostaglandin EP receptors and their roles in mucosal protection and ulcer healing in the gastrointestinal tract. Adv Clin Chem 51:121–144
- Tan HH, Ong WM, Lai SH, Chow WC (2007) Nimesulide-induced hepatotoxicity and fatal hepatic failure. Singapore Med J 48(6):582–585
- Tanaka Y, Yamaki F, Koike K, Toro L (2004) New insights into the intracellular mechanisms by which PGI2 analogues elicit vascular relaxation: cyclic AMP-independent, Gs-protein mediatedactivation of MaxiK channel. Curr Med Chem Cardiovasc Hematol Agents 2(3):257–265
- Tanaka T, Haneda S, Imakawa K, Sakai S, Nagaoka K (2009) A microRNA, miR-101a, controls mammary gland development by regulating cyclooxygenase-2 expression. Differentiation 77(2):181–187
- Tang X, Edwards EM, Holmes BB, Falck JR, Campbell WB (2006) Role of phospholipase C and diacylglyceride lipase pathway in arachidonic acid release and acetylcholine-induced vascular relaxation in rabbit aorta. Am J Physiol Heart Circ Physiol 290(1):H37–H45
- Tay A, Maxwell P, Li ZG, Goldberg H, Skorecki K (1994) Cytosolic phospholipase A2 gene expression in rat mesangial cells is regulated post-transcriptionally. Biochem J 304(Pt 2): 417–422
- Tay Y, Rinn J, Pandolfi PP (2014) The multilayered complexity of ceRNA crosstalk and competition. Nature 505(7483):344–352
- Taylor GA, Carballo E, Lee DM, Lai WS, Thompson MJ, Patel DD, Schenkman DI, Gilkeson GS, Broxmeyer HE, Haynes BF, Blackshear PJ (1996) A pathogenetic role for TNF alpha in the syndrome of cachexia, arthritis, and autoimmunity resulting from tristetraprolin (TTP) deficiency. Immunity 4(5):445–454
- Tegeder I, Pfeilschifter J, Geisslinger G (2001) Cyclooxygenase-independent actions of cyclooxygenase inhibitors. FASEB J 15(12):2057–2072

- Thorsen SB, Obad S, Jensen NF, Stenvang J, Kauppinen S (2012) The therapeutic potential of microRNAs in cancer. Cancer J 18(3):275–284
- Tian Q, Streuli M, Saito H, Schlossman SF, Anderson P (1991) A polyadenylate binding protein localized to the granules of cytolytic lymphocytes induces DNA fragmentation in target cells. Cell 67(3):629–639
- Tillman LG, Geary RS, Hardee GE (2008) Oral delivery of antisense oligonucleotides in man. J Pharm Sci 97(1):225–236
- Touqui L, Alaoui-El-Azher M (2001) Mammalian secreted phospholipases A2 and their pathophysiological significance in inflammatory diseases. Curr Mol Med 1(6):739–754
- Trajkovski M, Hausser J, Soutschek J, Bhat B, Akin A, Zavolan M, Heim MH, Stoffel M (2011) MicroRNAs 103 and 107 regulate insulin sensitivity. Nature 474(7353):649–653
- Tran H, Maurer F, Nagamine Y (2003) Stabilization of urokinase and urokinase receptor mRNAs by HuR is linked to its cytoplasmic accumulation induced by activated mitogen-activated protein kinase-activated protein kinase 2. Mol Cell Biol 23(20):7177–7188
- Tsang WP, Kwok TT (2010) Epigallocatechin gallate up-regulation of miR-16 and induction of apoptosis in human cancer cells. J Nutr Biochem 21(2):140–146
- Uhlmann S, Zhang JD, Schwager A, Mannsperger H, Riazalhosseini Y, Burmester S, Ward A, Korf U, Wiemann S, Sahin O (2010) miR-200bc/429 cluster targets PLCgamma and differentially regulates proliferation and EGF-driven invasion than miR-200a/141 in breast cancer. Oncogene 29(30):4297–4306
- Urade Y, Hayaishi O (2000) Biochemical, structural, genetic, physiological, and pathophysiological features of lipocalin-type prostaglandin D synthase. Biochim Biophys Acta 1482(1–2):259–271
- Urade Y, Hayaishi O (2011) Prostaglandin D2 and sleep/wake regulation. Sleep Med Rev 15(6):411-418
- van Rooij E, Purcell AL, Levin AA (2012) Developing microRNA therapeutics. Circ Res 110(3):496–507
- Varambally S, Cao Q, Mani RS, Shankar S, Wang X, Ateeq B, Laxman B, Cao X, Jing X, Ramnarayanan K, Brenner JC, Yu J, Kim JH, Han B, Tan P, Kumar-Sinha C, Lonigro RJ, Palanisamy N, Maher CA, Chinnaiyan AM (2008) Genomic loss of microRNA-101 leads to overexpression of histone methyltransferase EZH2 in cancer. Science 322(5908):1695–1699
- Vignarajan S, Xie C, Yao M, Sun Y, Simanainen U, Sved P, Liu T, Dong Q (2014) Loss of PTEN stabilizes the lipid modifying enzyme cytosolic phospholipase A(2)alpha via AKT in prostate cancer cells. Oncotarget 5(15):6289–6299
- Vinciguerra M, Sgroi A, Veyrat-Durebex C, Rubbia-Brandt L, Buhler LH, Foti M (2009) Unsaturated fatty acids inhibit the expression of tumor suppressor phosphatase and tensin homolog (PTEN) via microRNA-21 up-regulation in hepatocytes. Hepatology 49(4):1176–1184
- Volinia S, Calin GA, Liu CG, Ambs S, Cimmino A, Petrocca F, Visone R, Iorio M, Roldo C, Ferracin M, Prueitt RL, Yanaihara N, Lanza G, Scarpa A, Vecchione A, Negrini M, Harris CC, Croce CM (2006) A microRNA expression signature of human solid tumors defines cancer gene targets. Proc Natl Acad Sci U S A 103(7):2257–2261
- Walker MR, Brown SL, Riehl TE, Stenson WF, Stappenbeck TS (2010) Growth factor regulation of prostaglandin-endoperoxide synthase 2 (Ptgs2) expression in colonic mesenchymal stem cells. J Biol Chem 285(7):5026–5039
- Wang D, DuBois RN (2007) Measurement of eicosanoids in cancer tissues. Methods Enzymol 433:27–50
- Wang D, Dubois RN (2010) Eicosanoids and cancer. Nat Rev Cancer 10(3):181-193
- Wang HJ, Ruan HJ, He XJ, Ma YY, Jiang XT, Xia YJ, Ye ZY, Tao HQ (2010a) MicroRNA-101 is down-regulated in gastric cancer and involved in cell migration and invasion. Eur J Cancer 46(12):2295–2303
- Wang Z, Kayikci M, Briese M, Zarnack K, Luscombe NM, Rot G, Zupan B, Curk T, Ule J (2010b) iCLIP predicts the dual splicing effects of TIA-RNA interactions. PLoS Biol 8(10):e1000530
- Wang J, Guo Y, Chu H, Guan Y, Bi J, Wang B (2013a) Multiple functions of the RNA-binding protein HuR in cancer progression, treatment responses and prognosis. Int J Mol Sci 14(5):10015–10041

- Wang X, Yu B, Ren W, Mo X, Zhou C, He H, Jia H, Wang L, Jacob ST, Lee RJ, Ghoshal K, Lee LJ (2013b) Enhanced hepatic delivery of siRNA and microRNA using oleic acid based lipid nanoparticle formulations. J Control Release 172(3):690–698
- Wang I, Hennig J, Jagtap PK, Sonntag M, Valcarcel J, Sattler M (2014) Structure, dynamics and RNA binding of the multi-domain splicing factor TIA-1. Nucleic Acids Res 42(9):5949–5966
- Wellmann S, Truss M, Bruder E, Tornillo L, Zelmer A, Seeger K, Buhrer C (2010) The RNAbinding protein RBM3 is required for cell proliferation and protects against serum deprivationinduced cell death. Pediatr Res 67(1):35–41
- Williams SP, Dorn GW II, Rapoport RM (1994) Prostaglandin I2 mediates contraction and relaxation of vascular smooth muscle. Am J Physiol 267(2 Pt 2):H796–H803
- Williams KC, Renthal NE, Gerard RD, Mendelson CR (2012) The microRNA (miR)-199a/214 cluster mediates opposing effects of progesterone and estrogen on uterine contractility during pregnancy and labor. Mol Endocrinol 26(11):1857–1867
- Winzen R, Kracht M, Ritter B, Wilhelm A, Chen CY, Shyu AB, Muller M, Gaestel M, Resch K, Holtmann H (1999) The p38 MAP kinase pathway signals for cytokine-induced mRNA stabilization via MAP kinase-activated protein kinase 2 and an AU-rich region-targeted mechanism. EMBO J 18(18):4969–4980
- Wu BL, Xu LY, Du ZP, Liao LD, Zhang HF, Huang Q, Fang GQ, Li EM (2011) MiRNA profile in esophageal squamous cell carcinoma: downregulation of miR-143 and miR-145. World J Gastroenterol 17(1):79–88
- Wu CC, Shyu RY, Wang CH, Tsai TC, Wang LK, Chen ML, Jiang SY, Tsai FM (2012) Involvement of the prostaglandin D2 signal pathway in retinoid-inducible gene 1 (RIG1)-mediated suppression of cell invasion in testis cancer cells. Biochim Biophys Acta 1823(12):2227–2236
- Wu XL, Cheng B, Li PY, Huang HJ, Zhao Q, Dan ZL, Tian DA, Zhang P (2013) MicroRNA-143 suppresses gastric cancer cell growth and induces apoptosis by targeting COX-2. World J Gastroenterol 19(43):7758–7765
- Xu K, Kitchen CM, Shu HK, Murphy TJ (2007) Platelet-derived growth factor-induced stabilization of cyclooxygenase 2 mRNA in rat smooth muscle cells requires the c-Src family of protein-tyrosine kinases. J Biol Chem 282(45):32699–32709
- Xu F, Zhang X, Lei Y, Liu X, Liu Z, Tong T, Wang W (2010) Loss of repression of HuR translation by miR-16 may be responsible for the elevation of HuR in human breast carcinoma. J Cell Biochem 111(3):727–734
- Xu G, Shi C, Ji C, Song G, Chen L, Yang L, Zhao Y, Guo X (2014) Expression of microRNA-26b, an obesity-related microRNA, is regulated by free fatty acids, glucose, dexamethasone and growth hormone in human adipocytes. Mol Med Rep 10(1):223–228
- Yan M, Rerko RM, Platzer P, Dawson D, Willis J, Tong M, Lawrence E, Lutterbaugh J, Lu S, Willson JK, Luo G, Hensold J, Tai HH, Wilson K, Markowitz SD (2004) 15-Hydroxyprostaglandin dehydrogenase, a COX-2 oncogene antagonist, is a TGF-beta-induced suppressor of human gastrointestinal cancers. Proc Natl Acad Sci U S A 101(50):17468–17473
- Yang X, Wang W, Fan J, Lal A, Yang D, Cheng H, Gorospe M (2004) Prostaglandin A2-mediated stabilization of p21 mRNA through an ERK-dependent pathway requiring the RNA-binding protein HuR. J Biol Chem 279(47):49298–49306
- Yang L, Huang Y, Porta R, Yanagisawa K, Gonzalez A, Segi E, Johnson DH, Narumiya S, Carbone DP (2006) Host and direct antitumor effects and profound reduction in tumor metastasis with selective EP4 receptor antagonism. Cancer Res 66(19):9665–9672
- Yang TQ, Lu XJ, Wu TF, Ding DD, Zhao ZH, Chen GL, Xie XS, Li B, Wei YX, Guo LC, Zhang Y, Huang YL, Zhou YX, Du ZW (2014) MicroRNA-16 inhibits glioma cell growth and invasion through suppression of BCL2 and the nuclear factor-kappaB1/MMP9 signaling pathway. Cancer Sci 105(3):265–271
- Yates LA, Norbury CJ, Gilbert RJ (2013) The long and short of microRNA. Cell 153(3):516-519
- Yoshino H, Seki N, Itesako T, Chiyomaru T, Nakagawa M, Enokida H (2013) Aberrant expression of microRNAs in bladder cancer. Nat Rev Urol 10(7):396–404
- Yoshioka W, Higashiyama W, Tohyama C (2011) Involvement of microRNAs in dioxin-induced liver damage in the mouse. Toxicol Sci 122(2):457–465

- Young LE, Dixon DA (2010) Posttranscriptional regulation of cyclooxygenase 2 expression in colorectal cancer. Curr Colorectal Cancer Rep 6(2):60–67
- Young LE, Sanduja S, Bemis-Standoli K, Pena EA, Price RL, Dixon DA (2009) The mRNA binding proteins HuR and tristetraprolin regulate cyclooxygenase 2 expression during colon carcinogenesis. Gastroenterology 136(5):1669–1679
- Young LE, Moore AE, Sokol L, Meisner-Kober N, Dixon DA (2012) The mRNA stability factor HuR inhibits microRNA-16 targeting of COX-2. Mol Cancer Res 10(1):167–180
- Yount SM, Lassiter N (2013) The pharmacology of prostaglandins for induction of labor. J Midwifery Womens Health 58(2):133–144, quiz 238–139
- Zhang JG, Guo JF, Liu DL, Liu Q, Wang JJ (2011) MicroRNA-101 exerts tumor-suppressive functions in non-small cell lung cancer through directly targeting enhancer of Zeste Homolog 2. J Thorac Oncol 6(4):671–678
- Zheng D, Chen CY, Shyu AB (2011) Unraveling regulation and new components of human P-bodies through a protein interaction framework and experimental validation. RNA 17(9):1619–1634
- Zhu Y, Lu Y, Zhang Q, Liu JJ, Li TJ, Yang JR, Zeng C, Zhuang SM (2012) MicroRNA-26a/b and their host genes cooperate to inhibit the G1/S transition by activating the pRb protein. Nucleic Acids Res 40(10):4615–4625

Translational and Post-translational Control of Leptin Production by Fat Cells

10

Konstantin V. Kandror

1 Introduction

Leptin, a 16 kDa product of the *ob* gene, is synthesized predominantly in adipocytes and targets central nervous system regulating food intake, energy expenditure, and several other important physiological functions of mammalian organisms (Dalamaga et al. 2013; Ahima and Flier 2000; Friedman 2009). Discovery of leptin 20 years ago (Zhang et al. 1994) opened a new era in obesity research. Originally, it has been thought that leptin works as a "lipostat", as food intake and accumulation of energy stores in fat cells result in an increase in leptin production thus leading to inhibition of appetite and elevation of energy expenditure. Conversely, when fat stores decline, adipocytes reduce leptin production, and food intake is increased. This simple or better, over-simplified model has nonetheless triggered a great interest to leptin as a potential anti-obesity medication. However, in spite of early impressive results obtained in rodents and several promising studies in humans (Friedman 2009) it appears that human obesity is, for the most part, accompanied by elevated circulating leptin levels and is often resistant to exogenous leptin. At the same time, other human studies have convincingly demonstrated that weight loss results in a decrease in plasma leptin concentrations, and that low leptin levels predispose human patients to weight gain. These and other experiments have suggested that although humans may be resistant to increased leptin levels, a fall in plasma leptin stimulates food intake, so that leptin's main role in humans may be to protect fat stores in order to improve survival when food is scarce (Unger 2004). In agreement with this idea, it was shown that in humans, feeding increases and starvation decreases leptin levels which may explain the high failure rate of dieting (Ahima and Flier 2000). Indeed,

K.V. Kandror (🖂)

Boston University School of Medicine, Silvio Conte Building, K120D, 72 E. Concord Street, Boston, MA 02118, USA e-mail: kkandror@bu.edu

[©] Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms*

in Endocrine Regulation, DOI 10.1007/978-3-319-25124-0_10

administration of low doses of leptin may help to lose weight without the anguish and distress that usually accompany this process (Rosenbaum et al. 2005).

Regardless of how leptin exerts its biological activity, it is clear that nutrient uptake/status is directly coupled to leptin production in adipocytes. The mechanism(s) of this phenomenon are only beginning to emerge. Since this understudied and inadequately discussed regulatory connection is central to all proposed mechanisms of leptin action, it will be the main focus of our review. The problem has at least two aspects: a short term and a long term connection. First, circulating leptin levels increase within hours after feeding and decrease shortly after food deprivation. Although in humans, this effect may not be as fast and robust as in rodents, all mammals studied thus far still demonstrate a direct link between food intake and circulating leptin (Ahima and Flier 2000). Second, leptin production depends on the adipocyte size, since larger cells contain and secret more leptin than smaller cells (Hamilton et al. 1995). Since adipocyte size is defined by the volume of the central lipid droplet that is not likely to change significantly in response to each meal, this phenomenon shows a long lasting correlation between the amount of stored energy and leptin expression. At the physiological level, this translates into elevated leptin levels in obesity (Ahima and Flier 2000; Friedman 2009). At present, it is unknown whether or not the acute regulation of leptin expression by feeding has the same molecular basis as the long-term regulation by fat accumulation and/or adipocyte volume. We suggest that post-transcriptional mechanisms may be responsible for the regulation of leptin production by nutrients.

2 An Acute Connection between Food Intake and Leptin Production

2.1 Expression of Leptin in Adipocytes is Regulated at Multiple Levels

As circulating leptin and insulin levels increase after feeding and decrease after food deprivation (Ahima and Flier 2000; Levy et al. 1997; Frederich et al. 1995), the predominant hypothesis in the field has been that leptin production is controlled by insulin (Ahima and Flier 2000). In particular, it has been proposed that insulin activates leptin expression at the level of transcription (Becker et al. 1995; Saladin et al. 1995; MacDougald et al. 1995) via an as yet unclear mechanism(s). This idea, however, has not been confirmed by experiments in vitro. Indeed, the effect of insulin on leptin production in isolated rat adipocytes is resistant to the inhibitor of transcription actinomycin D (Bradley and Cheatham 1999; Roh et al. 2003). In addition, regulation of leptin production by insulin and nutrients is sustained in isolated adipocytes and samples of fat tissue (Bradley and Cheatham 1999; Roh et al. 2003; Lee and Fried 2006; Lee et al. 2007a; Barr et al. 1997; Levy and Stevens 2001; Mueller et al. 1998) where neither insulin (Bradley and Cheatham 1999; Roh et al. 2003; Lee et al. 2007b) nor nutrient status (Lee and Fried 2006; Lee et al. 2007a) affect the levels of leptin mRNA. This suggests that insulin and nutrients control the expression of the ob gene not (or not exclusively) at the level of transcription.

2.2 Regulation of Leptin Production at the Level of Translation

Alternatively, it has been suggested that leptin expression is regulated at the level of translation via mTORC1. This is an attractive hypothesis, as the activity of mTORC1 depends not only on insulin levels but also, on nutrient and energy availability (Dibble and Manning 2013; Sengupta et al. 2010) providing an additional physiological dimension to the regulation of leptin production.

A brief description of regulation and signaling of mTOR is warranted here. mTOR is an evolutionarily conserved serine/threonine protein kinase which is distributed between the two functionally distinct protein complexes, TORC1 and less studied TORC2. The former is composed of mTOR, raptor, mLST8, PRAS40, and deptor (Sengupta et al. 2010; Foster and Fingar 2010; Howell and Manning 2011) and is regulated by the TSC protein complex. One of its subunits, TSC2, is phosphorylated by a variety of hormone- and nutrient-dependent protein kinases, such as Akt, ERK and AMP-activated protein kinase (AMPK) (Huang and Manning 2008). These phosphorylation events control the ability of TSC2 to serve as a GTPase activating protein for the low molecular weight GTPase Rheb that directly binds to and activates mTORC1 (Foster and Fingar 2010; Huang and Manning 2008; Inoki et al. 2003). Phosphorylation of TSC2 by Akt and ERK increases, while phosphorylation of TSC2 by AMPK decreases levels of GTP-bound Rheb thus leading to up- and down-regulation of mTORC1 activity. In addition, mTORC1 is activated by amino acids (in particular, by branched chain amino acids, such as leucine) via recently described mechanisms (Efeyan et al. 2012). Although additional pathways of mTORC1 regulation exist, it is clear that insulin and nutrients activate while lack of energy rapidly inhibits mTORC1. Therefore, this complex has emerged as a central sensor of energy- and hormone-dependent metabolic regulation in metazoans that could potentially control leptin expression.

Early reports have demonstrated that leptin expression in adipocytes is not only regulated by insulin but also depends on the intracellular energy metabolism (Levy and Stevens 2001; Mueller et al. 1998). Leptin expression correlates very well with the level of intracellular ATP (Levy et al. 2000), and is rapidly up-regulated by leucine (Roh et al. 2003) indicating the involvement of mTORC1 in its regulation. In parallel, it has been shown that the activity of mTORC1 in adipocytes in vitro and in vivo is increased by insulin and by nutrient signals (in particular, by leucine) in the meal (Lynch et al. 2002, 2006). We and others have found that insulin- and leucine-induced increase in leptin production in isolated primary adipocytes and in cultured 3T3-L1 adipocytes is blocked by the mTORC1 inhibitor, rapamycin (Bradley and Cheatham 1999; Roh et al. 2003; Lee et al. 2007a). Finally, stable expression of the constitutively active form of Rheb increases leptin expression in 3T3-L1 adipocytes (Chakrabarti et al. 2008). Activation of mTORC1 by stable over-expression of the dominant-negative form of AMP-activated protein kinase has a similar effect (Chakrabarti et al. 2008). Thus, it is very likely that insulin and nutrients activate leptin expression in adipocytes via the mTORC1-mediated pathway. Quantitatively, plasma leptin levels in rats rise 2.5-3 fold within 3 h after food intake (Levy et al. 1997). Activation of mTORC1 in rat adipocytes by leucine administration (Roh et al. 2003) or by over-expression of constitutively active Rheb

or dominant-negative form of AMPK (Chakrabarti et al. 2008) causes a comparable activation of leptin biosynthesis. This suggests that regulation of leptin expression at the level of translation via mTORC1 plays a predominant role in the post-prandial control of leptin levels irrespective of potential changes in transcription of the *ob* gene.

How exactly mTORC1 regulates leptin translation is not clear. mTORC1 exerts its biological effects on protein biosynthesis via phosphorylation of the protein kinase S6K1 and translational repressor 4E-BP which leads to activation of the former and inhibition of the latter. Importantly, phosphorylation of S6K1 and 4E-BP may represent two independent pathways of translational control in the cell (Sonenberg and Hinnebusch 2009). In particular, phosphorylation of 4E-BP is required for the initiation of translation of mRNAs with double stranded regions in the 5'-UTR. The effect of S6K1 on translation is more complicated and may or may not require the presence of a 5'-terminal oligopyrimidine tract in target mRNAs.

Leptin mRNA lacks a 5'-terminus oligopyrimidine tract, but contains predicted hairpins in its 61-nt long (in case of humans) 5'-UTR. However, addition of leptin 5'-UTR to the reporter message does not inhibit, but rather stimulates its basal level of translation and, in addition, renders this mRNA resistant to regulation by mTORC1 (Lee et al. 2007a; Chakrabarti et al. 2008). This suggests that mTORC1 controls translation of leptin mRNA via a novel mechanism that may be somehow related to multiple structural elements localized in its long (ca. 3 kb) 3'-UTR (Zhang et al. 1994). Fried's laboratory has explored the role of leptin's 3'-UTR in translation and found that it decreases the efficiency of translation of luciferase mRNA under basal conditions, whereas insulin administration stimulates its translation (Lee et al. 2007a).

Importantly, specific sequences in the 3'-UTR are known not only to regulate the efficiency of mRNA translation but also, to define its intracellular localization (Kloc et al. 2002). In primary adipocytes, ca. two thirds of leptin mRNA is associated with heavy membrane fraction (primarily, endoplasmic reticulum) and only one third is found in the cytosol in a form of free 80S mRNP (Roh et al. 2003). Since this ratio is not significantly changed in response to activation of mTORC1, it is likely that mTORC1 specifically activates expression of leptin messages that are compartmentalized on the endoplasmic reticulum (i.e. the site where secreted proteins are synthesized). Thus, it is possible that the long 3'-UTR of leptin mRNA contains elements that are responsible for its compartmentalization on the endoplasmic reticulum and for the regulation of its translation there. However, the detailed structure-functional analysis of this potentially highly important part of the leptin message has not yet been carried out.

2.3 Regulation of Leptin Production at the Level of Secretion

Studies both in vivo and in vitro have demonstrated that secretion of leptin from adipose cells has two components: constitutive and regulated. In other words, adipocytes continuously release leptin into the medium, but this process may be acutely and substantially stimulated by insulin without any marked changes in the constitutive secretory pathway. The acute effect of insulin on leptin secretion precedes major changes in its biosynthesis and is preserved, at least partially, in the presence of cycloheximide (Bradley and Cheatham 1999; Lee and Fried 2006; Barr et al. 1997; Roh et al. 2000; Zeigerer et al. 2008). This suggests that fat cells may possess a regulatable pool of pre-synthesized leptin that can be discharged by insulin. In agreement with this idea, it has been shown that the major pool of the intracellular leptin in adipocytes is localized in small membrane vesicles (Barr et al. 1997; Roh et al. 2000, 2001; Bornstein et al. 2000; Ye et al. 2010) that are rapidly depleted by insulin and may thus represent a regulated secretory compartment. However, the storage capacity of this compartment is limited. It has been estimated that the size of the leptin storage pool in rat adipocytes is roughly equivalent to 1 h of constitutive secretion (Lee and Fried 2006; Roh et al. 2000), although this number may be underestimated as adipocytes tend to loose leptin in the process of cell isolation and, especially, during collagenase treatment (Barr et al. 1997). It is possible that regulation of leptin production at the level of translation and at the level of secretion have different biological functions: the former is likely to couple leptin production to food intake as described in the previous section while the latter may be responsible for the rapid pulsatile oscillations in circulating leptin levels that take place in vivo (Licinio et al. 1997, 1998; Sinha et al. 1996).

In general, an important functional feature of the adipocyte is that, unlike highly specialized endocrine and exocrine cells, it secretes a large variety of the biologically active protein products that include dozens of hormones, cytokines, acute-phase reactants, and enzymes (Adamczak and Wiecek 2013). The ability of fat cells to produce and to secrete adipokines in an orderly fashion is based on the unique structure of the secretory machinery that is clearly different from ones present in "classical" secretory and "constitutive" cells. Typically, secreted proteins are transported by specialized vesicular carriers through either the constitutive secretory pathway that exists in all cell types or through the regulated secretory pathways that are thought to be restricted to neuronal, exocrine and endocrine cells only (Burgess and Kelly 1987; Miller and Moore 1990). Secretion from adipose cells, however, does not fit this model. Although adipose cells continuously secrete their protein products, this process does not meet the definition of constitutive secretion. Firstly, fat cells accumulate substantial intracellular amounts of adipokines while typical constitutively secreted proteins are normally not stored inside the cell in any significant quantities (Miller and Moore 1990). Secondly, the release of such adipokines as leptin, adiponectin (Bogan and Lodish 1999; Scherer et al. 1995; Xie et al. 2008), adipsin (Kitagawa et al. 1989; Millar et al. 2000), and resistin (Ye et al. 2010) may acutely be stimulated by insulin and, possibly, by other secretagogues. However, leptin is the only adipokine for which the existence of a separate "storage compartment" has been demonstrated so far. Although the cell biological nature of leptin-containing secretory vesicles in adipocytes remains largely unexplored, they are different from the peptide-containing secretory granules present in neuronal, endocrine, and exocrine cells in buoyant density, sedimentational behavior, and kinetics of secretion (Roh et al. 2000). In this regard, it has been long known that

"constitutive secretory" cells may, in fact, possess regulated secretory pathways as well (Saucan and Palade 1994; Chavez et al. 1996). However, such pathways have not been systematically studied.

Secretion of another major adipocyte product, lipoprotein lipase or LPL, from fat cells follows its own non-canonical pathway. It has been demonstrated that adipocytes accumulate large amounts of active LPL (Enerback and Gimble 1993) suggesting that this enzyme is not efficiently targeted to the constitutive secretory pathway but is stored in intracellular compartment(s) that represents/includes the endoplasmic reticulum (ER). It has also been reported that insulin and serum may acutely stimulate the release of LPL activity from adipose cells (Fried and DiGirolamo 1986; Pradines-Figueres et al. 1988). However, this effect of insulin is completely blocked by cycloheximide (Eckel et al. 1984). Thus, insulin may or may not have a direct effect on secretion of LPL; rather, insulin may acutely increase its biosynthesis and/or activity.

The major intracellular pool of leptin does not overlap with that of LPL (Roh et al. 2001), adiponectin (Xie et al. 2008), and resistin (Ye et al. 2010) suggesting that leptin is localized in its own specialized storage vesicles. On the contrary, most of LPL and adiponectin reside in the ER (Roh et al. 2001; Qiang et al. 2007; Wang et al. 2007; Karki et al. 2011). Thus, the secretory process in adipose cells may be regulated at different levels: at the exit from the ER, as may be the case for LPL and adiponectin (Qiang et al. 2007; Wang et al. 2007) and at the level of the downstream vesicular storage compartment (leptin). Constitutively secreted proteins should be able to pass freely through all checkpoints in the secretory pathway. The exit of LPL and adiponectin from the ER may be controlled by different mechanisms. LPL requires a specialized chaperone, lipase maturation factor 1 (Peterfy 2012) and the adaptor protein Sel1L (Sha et al. 2014), whereas the release of adiponectin from the ER takes place in a thiol-mediated fashion and depends on the interplay between the two ER chaperones, ERp44 and Ero1-L α (Qiang et al. 2007; Wang et al. 2007). Also, exit from the ER does not seem to be the only regulatable step in the intracellular trafficking pathway of adiponectin as this protein bypasses the endosomal compartment as well (Xie et al. 2008). Secretion of another adipokine, adipsin, also seems to follow the endosome-mediated pathway (Millar et al. 2000). Unfortunately, the molecular details of this process are not known.

In addition to secretion, adipocytes possess other routes of regulated protein traffic to the plasma membrane. In particular, these cells acutely translocate intracellular Glut1- and Glut4-containing membrane vesicles to the cell surface in response to insulin stimulation (Bogan 2012; Kandror and Pilch 1996). However, the integral membrane protein Glut4 and free soluble adipokines, such as leptin, adiponectin, adipsin, resistin, and LPL, are localized in different vesicular carriers (Barr et al. 1997; Roh et al. 2000, 2001; Ye et al. 2010; Bogan and Lodish 1999; Millar et al. 2000) strongly suggesting that the "Glut4 pathway" is different from secretory pathway(s) in adipocytes.

In summary, we are still far from the complete understanding of the cell biology of adipocyte secretion, although there is little doubt that the mechanistic dissection of this process represents an urgent biomedical problem. Eventually, we should learn how to facilitate the release of adipokines of our choice and to inhibit secretion of unwanted proteins.

2.4 Regulation of Leptin Production at the Level of Degradation

Lee and Fried have found that up to 50 % of newly synthesized leptin molecules is rapidly degraded in lysosomes instead of being secreted (Lee and Fried 2006). This phenomenon is not unique for leptin. In adipocytes, a large fraction of de novo synthesized secreted proteins undergoes immediate lysosomal degradation. For example, up to 80 % of newly synthesized LPL molecules (Appel and Fried 1992; Olivecrona et al. 1987; Semb and Olivecrona 1987; Vannier and Ailhaud 1989) and a non-specified but significant part of newly synthesized adiponectin (Karki et al. 2011) entry the degradative pathway instead of being secreted. It is yet hard to tell why the cell decides to operate in such a way; it is clear, however, that "futile cycles" of biosynthesis and degradation are often seen in metabolism. One possibility is that such a mechanism provides a faster response to rapid changes in metabolic conditions. According to this idea, the cell may prefer to re-route secreted proteins from degradation to secretion and vice versa rather than to induce/suppress protein expression that takes a much longer time. Other factors, however, may also play a role. For example, it is feasible that the secretory capacity of the adipocyte is limited so that intracellular adipokines compete for being released outside the cell. In any case, interfering with lysosomal targeting/degradation may be a promising way to increase levels of circulating leptin and to improve metabolism in obese and insulin resistant patients.

3 A Long-Term Connection between Obesity and Leptin Secretion

The problem of leptin production has another important aspect. In addition to the short-term connection with food intake, circulating leptin levels are known to be steadily elevated in obesity (Ahima and Flier 2000; Friedman 2009). At the cellular level, larger adipocytes contain and secrete more leptin than smaller cells (Hamilton et al. 1995; Russell et al. 2001). Since the adipocyte size is defined primarily by the volume of the central lipid droplet, this phenomenon may show a cell-autonomous connection between the amount of stored energy (i.e. obesity at the molecular level) and leptin expression. The correlation between the size of the adipocyte and the level of leptin production was recognized for a long time but its mechanism remained obscure. Clearly, a single act of food intake cannot possibly change the size of the adipocyte in a significant fashion, so there should be another explanation for this phenomenon. We suggest that a simultaneous increase in the

adipocyte size and leptin production may represent two independent consequences of the same stimulus.

Recent work has revealed that fat storage in adipocytes is also controlled by mTORC1. In particular, Polak et al. have found that adipose-specific raptor null mice have less adipose tissue and are protected against diet-induced obesity (Polak et al. 2008). The authors attributed this effect to elevated energy expenditure due to the mitochondrial uncoupling. However, this model contradicts data from several research groups who have reported that the mTORC1 pathway positively regulates mitochondrial respiration (Ramanathan and Schreiber 2009; Schieke et al. 2006; Cunningham et al. 2007; Bentzinger et al. 2008). Furthermore, knock down of raptor in 3T3-L1 adipocytes does not uncouple mitochondria in vitro (Polak et al. 2008), suggesting that the latter phenomenon may not be directly linked to mTORC1 inhibition, but rather, reflect some secondary physiological effects the molecular mechanisms of which have yet to be elucidated.

It has also been shown that inhibition of mTORC1 signaling suppresses early adipogenesis (Polak et al. 2008; Kim and Chen 2004; Carnevalli et al. 2010; Zhang et al. 2009; Yeh et al. 1995) and/or lipogenesis (Lamming and Sabatini 2013; Ricoult and Manning 2013; Porstmann et al. 2008; Li et al. 2010; Chakrabarti et al. 2010; Laplante and Sabatini 2009) which may represent an alternative connection between the mTORC1 activity and fat storage. Simultaneously with stimulation of lipogenesis, mTORC1 inhibits lipolysis in adipocytes likely via the transcriptional regulation of the rate-limiting lipolytic enzyme, ATGL (Chakrabarti et al. 2010, 2013). The latter effect of mTORC1 may be especially significant since mTORC1-mediated inhibition of lipolysis contributes to overall triglyceride accumulation in adipocytes at least ten times more than stimulation of de novo lipogenesis (Chakrabarti et al. 2010).

Thus, the same mTORC1-mediated signaling pathway is responsible not only for the regulation of leptin expression at the level of translation but also, controls triglyceride storage in adipocytes via several different mechanisms (Fig. 10.1). Importantly, chronic over-nutrition and obesity lead to continuous activation of mTORC1 (Chakrabarti et al. 2013; Khamzina et al. 2005; Wang et al. 2009; Um et al. 2004) which should promote triglyceride storage (thus increasing the size of adipocytes) on one hand and stimulate leptin production on the other. In other words, the size of the adipocyte may serve as an indicator of the cumulative mTORC1 activity that may explain the correlation between the adjpocyte size and the levels of leptin production. Normally, this situation should create a feedback regulatory loop that terminates eating and limits the expansion of fat stores. In humans, years of elevated leptin may lead to the classical desensitization response which may explain resistance to leptin in human obesity (Friedman 2009). However, more studies are needed to determine the molecular nature of leptin resistance and to discover safe ways to overcome it.



Fig. 10.1 mTORC1 couples leptin secretion to the adipocyte size. Insulin and nutrients activate leptin biosynthesis via the mTORC1-mediated pathway that also promotes storage of triglycerides and increases the size of the cell. De novo synthesized leptin molecules are partitioned between constitutive and regulated secretion and immediate degradation

References

- Adamczak M, Wiecek A (2013) The adipose tissue as an endocrine organ. Semin Nephrol 33:2-13
- Ahima RS, Flier JS (2000) Leptin. Annu Rev Physiol 62:413-437
- Appel B, Fried SK (1992) Effects of insulin and dexamethasone on lipoprotein lipase in human adipose tissue. Am J Physiol 262:E695–E699
- Barr VA, Malide D, Zarnowski MJ, Taylor SI, Cushman SW (1997) Insulin stimulates both leptin secretion and production by rat white adipose tissue. Endocrinology 138:4463–4472
- Becker DJ, Ongemba LN, Brichard V, Henquin JC, Brichard SM (1995) Diet- and diabetesinduced changes of ob gene expression in rat adipose tissue. FEBS Lett 371:324–328
- Bentzinger CF, Romanino K, Cloetta D, Lin S, Mascarenhas JB, Oliveri F, Xia J, Casanova E, Costa CF, Brink M, Zorzato F, Hall MN, Ruegg MA (2008) Skeletal muscle-specific ablation of raptor, but not of rictor, causes metabolic changes and results in muscle dystrophy. Cell Metab 8:411–424
- Bogan JS (2012) Regulation of glucose transporter translocation in health and diabetes. Annu Rev Biochem 81:507–532
- Bogan JS, Lodish HF (1999) Two compartments for insulin-stimulated exocytosis in 3T3-L1 adipocytes defined by endogenous ACRP30 and GLUT4. J Cell Biol 146:609–620
- Bornstein SR, Abu-Asab M, Glasow A, Path G, Hauner H, Tsokos M, Chrousos GP, Scherbaum WA (2000) Immunohistochemical and ultrastructural localization of leptin and leptin receptor in human white adipose tissue and differentiating human adipose cells in primary culture. Diabetes 49:532–538
- Bradley RL, Cheatham B (1999) Regulation of ob gene expression and leptin secretion by insulin and dexamethasone in rat adipocytes. Diabetes 48:272–278

- Burgess TL, Kelly RB (1987) Constitutive and regulated secretion of proteins. Annu Rev Cell Biol 3:243–293
- Carnevalli LS, Masuda K, Frigerio F, Le Bacquer O, Um SH, Gandin V, Topisirovic I, Sonenberg N, Thomas G, Kozma SC (2010) S6K1 plays a critical role in early adipocyte differentiation. Dev Cell 18:763–774
- Chakrabarti P, Anno T, Manning BD, Luo Z, Kandror KV (2008) The mammalian target of rapamycin complex 1 regulates leptin biosynthesis in adipocytes at the level of translation: the role of the 5'-untranslated region in the expression of leptin messenger ribonucleic acid. Mol Endocrinol 22:2260–2267
- Chakrabarti P, English T, Shi J, Smas CM, Kandror KV (2010) The mTOR complex 1 suppresses lipolysis, stimulates lipogenesis and promotes fat storage. Diabetes 59:775–781
- Chakrabarti P, Kim JY, Singh M, Shin YK, Kim J, Kumbrink J, Wu Y, Lee MJ, Kirsch KH, Fried SK, Kandror KV (2013) Insulin inhibits lipolysis in adipocytes via the evolutionary conserved mTORC1-Egr1-ATGL-mediated pathway. Mol Cell Biol 33:3659–3666
- Chavez RA, Miller SG, Moore HP (1996) A biosynthetic regulated secretory pathway in constitutive secretory cells. J Cell Biol 133:1177–1191
- Cunningham JT, Rodgers JT, Arlow DH, Vazquez F, Mootha VK, Puigserver P (2007) mTOR controls mitochondrial oxidative function through a YY1-PGC-1alpha transcriptional complex. Nature 450:736–740
- Dalamaga M, Chou SH, Shields K, Papageorgiou P, Polyzos SA, Mantzoros CS (2013) Leptin at the intersection of neuroendocrinology and metabolism: current evidence and therapeutic perspectives. Cell Metab 18:29–42
- Dibble CC, Manning BD (2013) Signal integration by mTORC1 coordinates nutrient input with biosynthetic output. Nat Cell Biol 15:555–564
- Eckel RH, Prasad JE, Kern PA, Marshall S (1984) Insulin regulation of lipoprotein lipase in cultured isolated rat adipocytes. Endocrinology 114:1665–1671
- Efeyan A, Zoncu R, Sabatini DM (2012) Amino acids and mTORC1: from lysosomes to disease. Trends Mol Med 18:524–533
- Enerback S, Gimble JM (1993) Lipoprotein lipase gene expression: physiological regulators at the transcriptional and post-transcriptional level. Biochim Biophys Acta 1169:107–125
- Foster KG, Fingar DC (2010) Mammalian target of rapamycin (mTOR): conducting the cellular signaling symphony. J Biol Chem 285:14071–14077
- Frederich RC, Lollmann B, Hamann A, Napolitano-Rosen A, Kahn BB, Lowell BB, Flier JS (1995) Expression of ob mRNA and its encoded protein in rodents. Impact of nutrition and obesity. J Clin Invest 96:1658–1663
- Fried SK, DiGirolamo M (1986) Lipoprotein lipase secretion from isolated rat fat cells of different size. Life Sci 39:2111–2119
- Friedman JM (2009) Leptin at 14 y of age: an ongoing story. Am J Clin Nutr 89:973S-979S
- Hamilton BS, Paglia D, Kwan AY, Deitel M (1995) Increased obese mRNA expression in omental fat cells from massively obese humans. Nat Med 1:953–956
- Howell JJ, Manning BD (2011) mTOR couples cellular nutrient sensing to organismal metabolic homeostasis. Trends Endocrinol Metab 22:94–102
- Huang J, Manning BD (2008) The TSC1-TSC2 complex: a molecular switchboard controlling cell growth. Biochem J 412:179–190
- Inoki K, Zhu T, Guan KL (2003) TSC2 mediates cellular energy response to control cell growth and survival. Cell 115:577–590
- Kandror KV, Pilch PF (1996) Compartmentalization of protein traffic in insulin-sensitive cells. Am J Physiol 271:E1–E14
- Karki S, Chakrabarti P, Huang G, Wang H, Farmer SR, Kandror KV (2011) The multi-level action of fatty acids on adiponectin production by fat cells. PLoS One 6, e28146
- Khamzina L, Veilleux A, Bergeron S, Marette A (2005) Increased activation of the mammalian target of rapamycin pathway in liver and skeletal muscle of obese rats: possible involvement in obesity-linked insulin resistance. Endocrinology 146:1473–1481
- Kim JE, Chen J (2004) Regulation of peroxisome proliferator-activated receptor-gamma activity by mammalian target of rapamycin and amino acids in adipogenesis. Diabetes 53:2748–2756

- Kitagawa K, Rosen BS, Spiegelman BM, Lienhard GE, Tanner LI (1989) Insulin stimulates the acute release of adipsin from 3T3-L1 adipocytes. Biochim Biophys Acta 1014:83–89
- Kloc M, Zearfoss NR, Etkin LD (2002) Mechanisms of subcellular mRNA localization. Cell 108:533–544
- Lamming DW, Sabatini DM (2013) A central role for mTOR in lipid homeostasis. Cell Metab 18:465–469
- Laplante M, Sabatini DM (2009) An emerging role of mTOR in lipid biosynthesis. Curr Biol 19:R1046–R1052
- Lee MJ, Fried SK (2006) Multilevel regulation of leptin storage, turnover, and secretion by feeding and insulin in rat adipose tissue. J Lipid Res 47:1984–1993
- Lee MJ, Yang RZ, Gong DW, Fried SK (2007a) Feeding and insulin increase leptin translation. Importance of the leptin mRNA untranslated regions. J Biol Chem 282:72–80
- Lee MJ, Wang Y, Ricci MR, Sullivan S, Russell CD, Fried SK (2007b) Acute and chronic regulation of leptin synthesis, storage, and secretion by insulin and dexamethasone in human adipose tissue. Am J Physiol Endocrinol Metab 292:E858–E864
- Levy JR, Stevens W (2001) The effects of insulin, glucose, and pyruvate on the kinetics of leptin secretion. Endocrinology 142:3558–3562
- Levy JR, LeGall-Salmon E, Santos M, Pandak WM, Stevens W (1997) Effect of enteral versus parenteral nutrition on leptin gene expression and release into the circulation. Biochem Biophys Res Commun 237:98–102
- Levy JR, Gyarmati J, Lesko JM, Adler RA, Stevens W (2000) Dual regulation of leptin secretion: intracellular energy and calcium dependence of regulated pathway. Am J Physiol Endocrinol Metab 278:E892–E901
- Li S, Brown MS, Goldstein JL (2010) Bifurcation of insulin signaling pathway in rat liver: mTORC1 required for stimulation of lipogenesis, but not inhibition of gluconeogenesis. Proc Natl Acad Sci U S A 107:3441–3446
- Licinio J, Mantzoros C, Negrao AB, Cizza G, Wong M-L, Bongiorno PB, Chrousos GP, Karp B, Allen C, Flier JS, Gold PW (1997) Human leptin levels are pulsatile and inversely related to pituitary-adrenal function. Nat Med 3:575–579
- Licinio J, Negrao AB, Mantzoros C, Kaklamani V, Wong M-L, Bongiorno PB, Mulla A, Cearnal L, Veldhuis JD, Flier JS, McCann SM, Gold PW (1998) Synchronicity of frequently sampled, 24-h concentrations of circulating leptin, luteinizing hormone, and estradiol in healthy women. Proc Natl Acad Sci U S A 95:2541–2546
- Lynch CJ, Patson BJ, Anthony J, Vaval A, Jefferson LS, Vary TC (2002) Leucine is a direct-acting nutrient signal that regulates protein synthesis in adipose tissue. Am J Physiol Endocrinol Metab 283:E503–E513
- Lynch CJ, Gern B, Lloyd C, Hutson SM, Eicher R, Vary TC (2006) Leucine in food mediates some of the postprandial rise in plasma leptin concentrations. Am J Physiol Endocrinol Metab 291:E621–E630
- MacDougald OA, Hwang C-S, Fan H, Lane MD (1995) Regulated expression of the obese gene product (leptin) in white adipose tissue and 3T3-L1 adipocytes. Proc Natl Acad Sci U S A 92: 9034–9037
- Millar CA, Meerloo T, Martin S, Hickson GRX, Shimwell NJ, Wakelam MJO, James DE, Gould GW (2000) Adipsin and the glucose transporter Glut4 traffic to the cell surface via independent pathways in adipocytes. Traffic 1:141–151
- Miller SG, Moore H-P (1990) Regulated secretion. Curr Opin Cell Biol 2:642-647
- Mueller WM, Gregoire FM, Stanhope KL, Mobbs CV, Mizuno TM, Warden CH, Stern JS, Havel PJ (1998) Evidence that glucose metabolism regulates leptin secretion from cultured rat adipocytes. Endocrinology 139:551–558
- Olivecrona T, Chernick SS, Bengtsson-Olivecrona G, Garrison M, Scow RO (1987) Synthesis and secretion of lipoprotein lipase in 3T3-L1 adipocytes. Demonstration of inactive forms of lipase in cells. J Biol Chem 262:10748–10759
- Peterfy M (2012) Lipase maturation factor 1: a lipase chaperone involved in lipid metabolism. Biochim Biophys Acta 1821:790–794

- Polak P, Cybulski N, Feige JN, Auwerx J, Ruegg MA, Hall MN (2008) Adipose-specific knockout of raptor results in lean mice with enhanced mitochondrial respiration. Cell Metab 8:399–410
- Porstmann T, Santos CR, Griffiths B, Cully M, Wu M, Leevers S, Griffiths JR, Chung YL, Schulze A (2008) SREBP activity is regulated by mTORC1 and contributes to Akt-dependent cell growth. Cell Metab 8:224–236
- Pradines-Figueres A, Vannier C, Ailhaud G (1988) Short-term stimulation by insulin of lipoprotein lipase secretion in adipose cells. Biochem Biophys Res Commun 154:982–990
- Qiang L, Wang H, Farmer SR (2007) Adiponectin secretion is regulated by SIRT1 and the endoplasmic reticulum oxidoreductase Ero1-L alpha. Mol Cell Biol 27:4698–4707
- Ramanathan A, Schreiber SL (2009) Direct control of mitochondrial function by mTOR. Proc Natl Acad Sci U S A 106:22229–22232
- Ricoult SJ, Manning BD (2013) The multifaceted role of mTORC1 in the control of lipid metabolism. EMBO Rep 14:242–251
- Roh C, Thoidis G, Farmer SR, Kandror KV (2000) Identification and characterization of leptincontaining intracellular compartment in rat adipose cells. Am J Physiol 279:E893–E899
- Roh C, Roduit R, Thorens B, Fried S, Kandror KV (2001) Lipoprotein lipase and leptin are accumulated in different secretory compartments in rat adipocytes. J Biol Chem 276: 35990–35994
- Roh C, Han J, Tzatsos A, Kandror KV (2003) Nutrient-sensing mTOR-mediated pathway regulates leptin production in isolated rat adipocytes. Am J Physiol Endocrinol Metab 284:E322–E330
- Rosenbaum M, Goldsmith R, Bloomfield D, Magnano A, Weimer L, Heymsfield S, Gallagher D, Mayer L, Murphy E, Leibel RL (2005) Low-dose leptin reverses skeletal muscle, autonomic, and neuroendocrine adaptations to maintenance of reduced weight. J Clin Invest 115:3579–3586
- Russell CD, Ricci MR, Brolin RE, Magill E, Fried SK (2001) Regulation of the leptin content of obese human adipose tissue. Am J Physiol Endocrinol Metab 280:E399–E404
- Saladin R, De Vos P, Guerre-Millo M, Leturque A, Girard J, Staels B, Auwerx J (1995) Transient increase in obese gene expression after food intake or insulin administration. Nature 377:527–529
- Saucan L, Palade GE (1994) Membrane and secretory proteins are transported from the golgi complex to the sinusoidal plasmalemma of hepatocytes by distinct vesicular carriers. J Cell Biol 125:733–741
- Scherer PE, Williams S, Fogliano M, Baldini G, Lodish HF (1995) A novel serum protein similar to C1q, produced exclusively in adipocytes. J Biol Chem 270:26746–26749
- Schieke SM, Phillips D, McCoy JP Jr, Aponte AM, Shen RF, Balaban RS, Finkel T (2006) The mammalian target of rapamycin (mTOR) pathway regulates mitochondrial oxygen consumption and oxidative capacity. J Biol Chem 281:27643–27652
- Semb H, Olivecrona T (1987) Mechanisms for turnover of lipoprotein lipase in guinea pig adipocytes. Biochim Biophys Acta 921:104–115
- Sengupta S, Peterson TR, Sabatini DM (2010) Regulation of the mTOR complex 1 pathway by nutrients, growth factors, and stress. Mol Cell 40:310–322
- Sha H, Sun S, Francisco AB, Ehrhardt N, Xue Z, Liu L, Lawrence P, Mattijssen F, Guber RD, Panhwar MS, Brenna JT, Shi H, Xue B, Kersten S, Bensadoun A, Peterfy M, Long Q, Qi L (2014) The ER-associated degradation adaptor protein Sel1L regulates LPL secretion and lipid metabolism. Cell Metab 20:458–470
- Sinha MK, Sturis JS, Ohannesian J, Magosin S, Stephens T, Heiman ML, Polonsky KS, Caro JF (1996) Ultradian oscillations of leptin secretion in humans. Biochem Biophys Res Commun 228:733–738
- Sonenberg N, Hinnebusch AG (2009) Regulation of translation initiation in eukaryotes: mechanisms and biological targets. Cell 136:731–745
- Um SH, Frigerio F, Watanabe M, Picard F, Joaquin M, Sticker M, Fumagalli S, Allegrini PR, Kozma SC, Auwerx J, Thomas G (2004) Absence of S6K1 protects against age- and diet-induced obesity while enhancing insulin sensitivity. Nature 431:200–205
- Unger RH (2004) The hyperleptinemia of obesity-regulator of caloric surpluses. Cell 117:145-146
- Vannier C, Ailhaud G (1989) Biosynthesis of lipoprotein lipase in cultured mouse adipocytes. II. Processing, subunit assembly, and intracellular transport. J Biol Chem 264:13206–13216

- Wang ZV, Schraw TD, Kim JY, Khan T, Rajala MW, Follenzi A, Scherer PE (2007) Secretion of the adipocyte-specific secretory protein adiponectin critically depends on thiol-mediated protein retention. Mol Cell Biol 27:3716–3731
- Wang CY, Kim HH, Hiroi Y, Sawada N, Salomone S, Benjamin LE, Walsh K, Moskowitz MA, Liao JK (2009) Obesity increases vascular senescence and susceptibility to ischemic injury through chronic activation of Akt and mTOR. Sci Signal 2:ra11
- Xie L, O'Reilly CP, Chapes SK, Mora S (2008) Adiponectin and leptin are secreted through distinct trafficking pathways in adipocytes. Biochim Biophys Acta 1782:99–108
- Ye F, Than A, Zhao Y, Goh KH, Chen P (2010) Vesicular storage, vesicle trafficking, and secretion of leptin and resistin: the similarities, differences, and interplays. J Endocrinol 206:27–36
- Yeh WC, Bierer BE, McKnight SL (1995) Rapamycin inhibits clonal expansion and adipogenic differentiation of 3T3-L1 cells. Proc Natl Acad Sci U S A 92:11086–11090
- Zeigerer A, Rodeheffer MS, McGraw TE, Friedman JM (2008) Insulin regulates leptin secretion from 3T3-L1 adipocytes by a PI 3 kinase independent mechanism. Exp Cell Res 314:2249–2256
- Zhang Y, Proenca R, Maffei M, Barone M, Leopold L, Friedman JM (1994) Positional cloning of the mouse obese gene and its human homologue. Nature 372:425–432
- Zhang HH, Huang J, Duvel K, Boback B, Wu S, Squillace RM, Wu CL, Manning BD (2009) Insulin stimulates adipogenesis through the Akt-TSC2-mTORC1 pathway. PLoS One 4, e6189

Post-transcriptional Regulation of Parathyroid Hormone Gene Expression in Health and Disease

11

Tally Naveh-Many

1 Introduction

Parathyroid hormone (PTH) regulates serum calcium and phosphate levels and bone strength. Changes in serum calcium are sensed by the seven *trans* membrane G-protein coupled calcium sensing receptor (CaR) on the parathyroid cell membrane. The parathyroid is unique in that the trigger for PTH secretion is a low extra-cellular calcium rather than high calcium as for other hormones. Small decreases in serum calcium and more prolonged increases in serum phosphate stimulate the parathyroid to secrete PTH which then acts on its target organs the kidney and bone to correct serum calcium and phosphate levels (Fig. 11.1) (Silver and Naveh-Many 2009). Dietary induced hypocalcemia or uremia lead to secondary hyperparathyroidism (SHP) with increased serum PTH, PTH mRNA levels and parathyroid cell proliferation (Moallem et al. 1998; Naveh-Many et al. 1995). SHP is a common disorder in patients with CKD and is characterized by excessive serum PTH levels, parathyroid hyperplasia and an imbalance in calcium and phosphorus metabolism. SHP develops early in the course of CKD and becomes more prominent as kidney function declines. SHP of CKD is associated with an increased morbidity and mortality (Silver et al. 2002).

Parathyroid cells have few secretory granules as compared to other endocrine cells and therefore PTH production is regulated largely at the levels of PTH gene expression and parathyroid cell proliferation (Habener et al. 1984). The changes in PTH gene expression by calcium phosphate and CKD are due to post-transcriptional mechanisms affecting PTH mRNA stability. The parathyroid also responds to changes in serum 1,25(OH)₂ vitamin D (1,25D) which decreases PTH levels. PTH

K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_11

T. Naveh-Many, PhD (🖂)

Minerva Center for Calcium and Bone Metabolism, Department of Nephrology, Hadassah Hospital, Hadassah Hebrew University Medical Center, POB 12000, Jerusalem 91120, Israel e-mail: tally@cc.huji.ac.il

[©] Springer International Publishing Switzerland 2016



Fig. 11.1 Calcium, phosphate, PTH, FGF23, 1,25D and FGF23 interactions. There are endocrinological feedback loops that govern mineral homeostasis. FGF23, fibroblast growth factor 23; PTH, parathyroid hormone, P, serum phosphate (Silver and Naveh-Many 2009)

in turn, increases the renal synthesis of 1,25D. 1,25D then increases blood calcium largely by increasing the efficiency of intestinal calcium absorption (Fig. 11.1). The increased serum calcium would shut down PTH secretion by activating the parathyroid calcium sensing receptor (CaR) (Brown et al. 1993). In contrast to the post-transcriptional regulation of PTH gene expression by changes in serum calcium and phosphate levels and CKD, 1,25D decreases PTH gene transcription in vitro and in vivo (Silver et al. 1985, 1986; Russell et al. 1986). The 1,25D receptor (VDR) is expressed in the rat parathyroid, confirming that the parathyroid is indeed a target organ for vitamin D (Naveh-Many et al. 1990). Several groups have identified DNA sequences in the 5'-flanking region of the PTH gene that may mediate the negative regulation of PTH gene transcription by 1,25D (Demay et al. 1992; Fujiki et al. 2005; Murayama et al. 2004). The action of 1,25D to decrease serum PTH is used therapeutically in the management of CKD patients. They are given 1,25D or its prodrug 1α (OH)-vitamin D₃ to treat or prevent the secondary hyperparathyroidism (2HPT) of CKD. Fibroblast growth factor 23 (FGF23) is a bone-derived phosphaturic hormone that acts on the kidney to increase phosphate excretion and suppress biosynthesis of 1,25D (Fig. 11.1). FGF23 signals through the fibroblast growth factor receptor 1 (FGFR1) bound by the transmembrane protein Klotho (Kurosu et al. 2006). We have identified the parathyroid as a target organ for FGF23 (Ben Dov et al. 2007a). Recombinant FGF23 decreases PTH gene expression and secretion in vivo and in vitro by activation of the MAP kinase pathway (Fig. 11.1). In CKD there are very high levels of serum FGF23 together with increased serum PTH levels, due to down regulation of the kloth-FGFR1 FGF23 receptor complex in the parathyroid, leading to resistance of the parathyroid to FGF23 (Galitzer et al. 2010; Komaba et al. 2010). In this chapter we discuss the molecular mechanisms of the posttranscriptional regulation of PTH gene expression by calcium, phosphate and CKD.

2 Post-transcriptional Regulation of PTH Gene Expression by Calcium, Phosphate and CKD

Hypocalcemia, hypophosphatemia and CKD regulate PTH mRNA stability by posttranscriptional mechanisms. Dietary induced hypocalcemia and experimental CKD markedly increase PTH secretion, mRNA levels and after prolonged stimulation, parathyroid cell proliferation (Moallem et al. 1998; Naveh-Many et al. 1995). In the rat, hypocalcemia leads to a >10-fold increase in PTH mRNA levels and this increase is post-transcriptional affecting mRNA stability (Moallem et al. 1998). Serum phosphate also has a direct effect on PTH secretion, PTH mRNA levels and parathyroid cell proliferation (Kilav et al. 1995; Naveh-Many et al. 1995). Careful in vivo studies showed that the regulation of PTH gene expression by dietary induced hypophosphatemia is independent of changes in serum calcium and 1,25D (Fig. 11.1) (Kilav et al. 1995). In vitro studies, when tissue architecture was maintained, confirmed the direct effect of phosphate on the parathyroid (Almaden et al. 1996, 1998; Nielsen et al. 1996; Rodriguez et al. 1996; Slatopolsky et al. 1996). In vivo studies showed that phosphorus depletion leads to a dramatic decrease in rat PTH mRNA levels and that this effect is post-transcriptional as is the effect of hypocalcemia to increase PTH mRNA levels (Kilav et al. 1995; Moallem et al. 1998). There is a ~60-fold difference in PTH mRNA levels between hypocalcemic and hypophosphatemic rats and these dietary models were used as tools to define the mechanism of the post-transcriptional regulation of PTH gene expression (Moallem et al. 1998; Nechama et al. 2008).

Secondary hyperparathyroidism is a common disorder in patients with CKD and in experimental models where there are increases in PTH gene expression, secretion and parathyroid hyperplasia (Naveh-Many and Silver 1990; Silver et al. 2002). In CKD patients with 2HPT, calcimimetics and oral phosphate binders are effective drugs used to control the high serum PTH levels (Block et al. 2004; D'Haese et al. 2003; Joy and Finn 2003; Moe et al. 2005). In a rat model of CKD induced by an adenine high phosphorus diet, PTH mRNA levels were increased already after 7 days of the diet and more so at 21 days (Levi et al. 2006). The addition of the calcimimetic R568 or an oral phosphate binder, lanthanum carbonate (La), decreased PTH mRNA and serum PTH levels in the CKD rats (Ben Dov et al. 2007b). The effects of CKD and the calcimimetic or the phosphorus binder were posttranscriptional as were those of changes in serum calcium and phosphate (Moallem et al. 1998; Nechama et al. 2009a; Yalcindag et al. 1999).

Therefore, PTH gene expression is regulated post-transcriptionally by serum calcium, phosphate, CKD and its management by calcimimetics and oral phosphorus binders. The changes in PTH mRNA stability are mediated by protein-PTH mRNA interactions that determine the susceptibility of PTH mRNA to the degradation machinery (Kilav et al. 1995; Levi et al. 2006; Moallem et al. 1998; Nechama et al.

2008). The balanced interaction of stabilizing and destabilizing proteins with the PTH mRNA determines PTH mRNA stability and levels, serum PTH and the resultant response of the parathyroid to calcium, phosphate and CKD (Nechama et al. 2008, 2009c).

3 The PTH mRNA *cis*-Acting Protein Binding Element

For many mRNAs, post-transcriptional regulation involves critical *cis*-acting elements, mostly in the untranslated regions (UTR) that are targets for *trans*-acting proteins regulating mRNA stability and translation (Barreau et al. 2006). Adenine and Uridine-rich elements (ARE) are a well-defined family of *cis*-acting elements critical for the expression of many unstable mRNAs that code for cytokines, transcription factors, proto-oncogenes and other mRNAs (Fig. 11.2) (Brewer 2002). Three classes of AREs have been identified, two of which contain several scattered or overlapping copies of the pentanucleotide AUUUA, while class III AREs lack this motif but contain A and U rich sequences and possibly other unknown determinants. A number of ARE binding proteins (ARBPs) have been identified. K-homology splicing regulator protein (KSRP) is an examples for decay promoting factors (Barreau et al. 2006; Gherzi et al. 2004). KSRP interacts with the large multiprotein 3'-5' exoribonuclease complex, the exosome, and recruits it to target ARE containing mRNAs thereby promoting their rapid degradation (Chou et al. 2006; Linker et al. 2005). Other ARE binding proteins, such as the ELAV protein family members (mainly HuR), are stabilizing factors and AU rich binding factor 1 (AUF-1) promotes either decay or stabilization, depending on the mRNA and cell type (Fig. 11.2) (Wilusz and Wilusz 2004).

PTH mRNAs are typical eukaryotic mRNAs that contain a 7-methylguanosine cap at the 5' terminus and a poly adenylic nucleic acid (poly A) stretch at the 3' terminus (Naveh-Many 2005; Kemper 1986). PTH mRNA consists of three exons



Fig. 11.2 Regulation of mRNA fate. mRNAa are comprised of the 5' methyl cap (5⁷G), open reading frame (ORF), 5' and 3'-untranslated regions (UTRs) and poly A tail. Following transcription, Adenylate/uridylate (AU) rich binding proteins (ARBPs) bind to the mRNA and determine mRNA stability, localization and translation. AU-rich elements (AREs) within the mRNA 3'-UTR act to stabilize or destabilize the mRNA. Stabilized RNA undergoes translation in ribosomes, whereas destabilized RNA undergoes deadenylation, decapping, and degradation by the exo- or endo-nucleases. microRNAs (miRNAs) as part of the RNA-induced silencing complex (RISC), bind to target mRNAs at their 3'-UTR and target them for translational inhibition or degradation



Fig. 11.3 Model for the regulation of PTH mRNA stability by changes in calcium and phosphate levels and CKD. a. Schematic representation of the PTH mRNA including the 5'-UTR (red), coding region (orange) the 3'-UTR (gray) and the 26 nucleotide cis acting AU rich element (ARE) (green). The nucleotide sequence of the element in different species is shown. Nucleotides that differ from the rat sequence are in **bold**. In hypocalcemic rats and in rats with experimental chronic renal failure (CKD) there is increased PTH mRNA stability and levels and this is associated with decreased binding of the destabilizing protein KSRP and increased binding of the protective proteins, AUF1 (AU rich binding factor) and Unr (Up-stream of N-ras), to the PTH mRNA 3'-UTR ARE. In control and more so in rats fed a low Pi diet, or treated with a calcimimetic (R568) or a phosphate binder (La) there is increased interaction of PTH mRNA with KSRP and decreased interaction with AUF1. KSRP recruits the exosome to PTH mRNA leading to PTH mRNA decay. The peptidyl prolyl cis/trans isomerase Pin1 is upstream of KSRP and leads to KSRP dephosphorylation and activation thus increasing KSRP-PTH mRNA interaction and mRNA degradation. In hypocalcemia or renal failure, Pin1 is inactive and KSRP is phosphorylated and hence less potent to bind PTH mRNA. AUF1 then binds the PTH mRNA 3'-UTR ARE with a greater affinity leading to increased PTH mRNA stability. Reproduced from Nechama et al. (2009c) and Silver and Naveh-Many (2009). b. Model for ARE-directed PTH mRNA decay. Basal PTH levels are obtained when the mRNA destabilizing ARBPs, such as KSRP and the stabilizing proteins such as AUF1, provide a balance of mRNA degradation and stabilization. In hypocalcemia or CKD, Pin1 is less active, resulting in KSRP phosphorylation and decreased binding to PTH mRNA, shifting the balance to mRNA stabilization

coding for the 5'-untranslated region (5'-UTR) (exon I), the prepro region (exon II) and the structural PTH hormone together with the 3'-UTR (exon III) (Fig. 11.3a). The PTH mRNA 3'-UTR in all species is rich in A and U nucleotides (Bell et al. 2005b).

Protein-PTH mRNA binding experiments demonstrated specific interaction of rat and human parathyroid extracts with transcripts for the rat and human PTH mRNA 3'-UTR terminal region. This binding was regulated by calcium or phosphorus depletion and correlated with PTH mRNA levels and stability in vivo (Moallem et al. 1998). A 26 nucleotide ARE at the 3' end of the PTH mRNA 3'-UTR is the

minimal protein binding region and is highly conserved in the PTH mRNA 3'-UTRs of rat, mouse, man, dog and cat (Fig. 11.3a). The conservation of sequences within a region that does not code for protein (UTR) suggests that the binding element is a functional unit that has been evolutionarily conserved (Bell et al. 2005a; Kilav et al. 2001). This PTH mRNA 3'-UTR element is a *cis*-acting type III ARE that determines PTH mRNA stability and its regulation by changes in serum calcium, phosphate and CKD (see below) (Kilav et al. 2001; Nechama et al. 2008).

There is no parathyroid cell line, therefore a cell-free mRNA in vitro degradation assay (IVDA) was utilized to identify the factors involved in PTH mRNA decay. In the IVDA, extracts from parathyroids were incubated with in vitro transcribed RNA probes for the PTH mRNA to estimate mRNA decay mediated by the parathyroid extracts. IVDA has been shown to reproduce differences in mRNA stability that occur in vivo and specifically the differences in PTH mRNA stability induced by calcium, phosphate and CKD (Fritz et al. 2000; Moallem et al. 1998; Nechama et al. 2008). Parathyroid extracts from hypocalcemic and CKD rats stabilized and extracts from hypophosphatemic rats destabilized transcripts for the PTH mRNA compared to parathyroid extracts from control rat, correlating with steady state PTH mRNA levels in vivo. These effects were dependent upon the terminal 60 nucleotides of the PTH mRNA 3'-UTR that contain the ARE (Moallem et al. 1998; Nechama et al. 2009a; Yalcindag et al. 1999). A 63 nucleotide transcript containing the conserved 26 nucleotide ARE and flanking regions was both necessary and sufficient to regulate PTH mRNA stability and to confer responsiveness of reporter mRNAs to changes in calcium and phosphate (Kilav et al. 2001). Structural analysis showed that the PTH mRNA 3'-UTR and in particular the ARE is dominated by significant open regions with little folded base pairing (Kilav et al. 2004).

4 The PTH mRNA trans Acting Stabilizing Proteins

Two PTH mRNA binding and stabilizing proteins have been identified by PTH RNA affinity chromatography. These trans acting proteins are AUF1 and Up-stream of N-ras (Unr) (Dinur et al. 2006; Sela-Brown et al. 2000). AUF1 consists of four isoforms (p37, p40, p42, and p45) that are generated by alternative pre-mRNA splicing of AUF1 mRNA (Wagner et al. 1998). These proteins bind the PTH mRNA 3'-UTR and are part of the PTH mRNA-parathyroid protein binding complex. Addition of recombinant AUF1 isoforms to parathyroid extracts from phosphorus depleted rats prevented the rapid degradation of PTH transcripts in IVDAs (Sela-Brown et al. 2000). Interestingly, calcium and phosphorus depletion as well as kidney failure lead to post-translational modifications of the PTH mRNA stabilizing protein AUF1, with no change in AUF1 protein levels (Bell et al. 2005c; Levi et al. 2006). These modifications may lead to the differences in AUF1-PTH mRNA binding affinity and PTH mRNA stability. Over-expression of AUF1 in human embryonic kidney (HEK) 293 cells co-transfected with expression plasmids for the PTH gene or a chimeric GH gene containing the PTH mRNA 3'-UTR 63 nucleotide cis-acting element, stabilized PTH mRNA and chimeric reporter mRNA, but not a PTH mRNA lacking the cis element nor a reporter mRNA containing a truncated PTH element (Bell et al. 2005a; Dinur et al. 2006). Unr over-expression had a similar effect. Knock-down of all four AUF1 isoforms or Unr by siRNA led to the opposite effect, decreasing PTH mRNA levels (Bell et al. 2005c; Dinur et al. 2006). AUF1 also stabilized a reporter mRNA containing the bovine PTH mRNA protein-binding element that is different from the one characterized in the rat (Bell et al. 2005a). These studies identified AUF1 and Unr as PTH mRNA stabilizing proteins (Fig. 11.3). However, the half-life of mRNAs is determined by the coordinate association of both stabilizing and destabilizing factors with the specific mRNA in the cytoplasm (Fig. 11.3b).

5 The PTH mRNA *trans* Acting Destabilizing Protein KSRP

The decay promoting protein KSRP was identified as a PTH mRNA *trans* acting destabilizing protein. KSRP is a RNA-binding protein implicated in a variety of cellular processes, including transcription, alternative pre-mRNA splicing, and editing as well as mRNA localization and stability (Gherzi et al. 2004; Lellek et al. 2000; Min et al. 1997; Snee et al. 2002). KSRP binds to ARE containing mRNA and promotes rapid mRNA decay of several inherently labile mRNAs, recruiting the 3' to 5' exoribonucleolitic complex exosome to the RNAs (Gherzi et al. 2004). The central part of the KSRP contains 4 adjacent K homology (KH) domains that are required for AREs recognition and interaction with the mRNA degradation machinery to promote decay of target mRNAs (Gherzi et al. 2004).

KSRP binds the PTH mRNA 3'-UTR ARE both in vivo in the parathyroid glands and in vitro in transfected cells (Nechama et al. 2008). This binding is decreased in glands from calcium-depleted or experimental CKD rats in which PTH mRNA is more stable compared to parathyroid glands from control and phosphorus-depleted rats in which PTH mRNA is less stable. KSRP also interacts with the PTH mRNA 3'-UTR ARE in vitro in transfected cells. The KH domains 3-4 of KSRP were sufficient for this association, as reported for other mRNAs (Gherzi et al. 2004; Nechama et al. 2008). KSRP over-expression and knock down experiments showed that KSRP decreased co-transfected PTH mRNA steady-state levels through the PTH mRNA ARE. Overexpression of KSRP specifically decreased both rat and human PTH mRNA levels in cotransfected HEK293 cells and accelerated PTH mRNA decay. This effect of KSRP was dependent on the PTH mRNA ARE. Conversely, KSRP knock-down increased both PTH mRNA stability and PTH mRNA steady-state levels. Moreover, PTH mRNA decay was dependent on the KSRP-recruited exosome in parathyroid extracts (Nechama et al. 2008). By its interaction with PTH mRNA ARE, KSRP would recruit the exosome to degrade PTH mRNA. These findings suggested that KSRP-PTH mRNA interactions control PTH mRNA t_{1/2} by recruitment of a degradation complex to PTH mRNA. AUF1 binding to the same region in the PTH mRNA 3'-UTR competes with the binding of KSRP and thus protect PTH mRNA from degradation (Fig. 11.3).

Of interest, KSRP interacts with the endoribonuclease polysomal ribonuclease 1 (PMR1) (Chernokalskaya et al. 1998; Nechama et al. 2009b). PMR1 facilitates PTH mRNA degradation, adding an additional layer of complexity to the regulation

of PTH mRNA stability. PMR1 mediated decrease in PTH mRNA levels involves the PTH mRNA 3'-UTR ARE, KSRP and the exosome. KSRP recruits a degradation complex, comprising both exo- and endo-ribonucleases to PTH mRNA, thus controlling its mRNA half-life (Nechama et al. 2009b).

6 The Balanced Interactions of Stabilizing AUF1 and Destabilizing KSRP Determine PTH mRNA Stability and Levels

The pattern of interactions of KSRP and AUF1 with PTH mRNA suggest that these proteins have opposing roles in the regulation of PTH gene expression in vivo (Fig. 11.3). KSRP and AUF1 protein-PTH mRNA interactions in the parathyroid were studied using a RNA immunoprecipitation (RIP) assay which provides a snap-shot of protein-mRNA interactions at a specific time point in vivo. In this assay the parathyroid glands were cross-linked, AUF1 or KSRP containing complexes immunoprecipitated and the amount of PTH mRNA associated with each of the proteins determined by qRT-PCR analysis. KSRP-PTH mRNA interaction was decreased in glands from calcium depleted or CKD rats, where PTH mRNA is more stable, and increased in parathyroids from phosphorus depleted rats, where PTH mRNA is less stable. In contrast, AUF1-PTH mRNA interactions were increased by hypocalcemia and CKD and decreased in the phosphorus depleted rat parathyroids (Nechama et al. 2008). Both proteins bind to the same cis ARE in the PTH mRNA 3'-UTR (Fig. 11.3a). Consequently, the differential interactions of KSRP and AUF1 suggest that these proteins compete for their binding to the PTH mRNA 3'-UTR element, having antagonistic effects on PTH mRNA stability (Fig. 11.3) (Naveh-Many and Nechama 2007). Indeed, in vitro, over-expression of the PTH mRNA stabilizing protein AUF1 isoform p45 blocked KSRP-PTH mRNA binding and partially prevented the KSRP mediated decrease in PTH mRNA levels (Nechama et al. 2008). Therefore, the balanced interaction of these proteins with the PTH mRNA 3'-UTR determines basal PTH mRNA levels and the regulation of PTH mRNA levels (Fig. 11.3b).

Calcimimetics and oral phosphorus binders that are widely used to treat secondary hyperparathyroidism decrease PTH mRNA levels post-transcriptionally in CKD rats (Nechama et al. 2009a). KSRP-PTH mRNA interaction was increased by both the calcimimetic R568 and the oral phosphorus binder La which decreased PTH mRNA levels. IVDAs showed that PTH mRNA is destabilized by parathyroid extracts from CKD rats treated with R568 or La compared to parathyroid extracts from untreated CKD rats. This destabilizing effect of R568 and La was dependent upon KSRP and the PTH mRNA 3'-UTR. Therefore, the calcimimetic R568 and correction of serum phosphate by La determine PTH mRNA stability through KSRP-mediated PTH mRNA decay, thereby decreasing PTH expression (Nechama et al. 2009a). Changes in binding of the *trans* acting factors to the PTH mRNA therefore determine PTH gene expression in CKD and after management of the 2HP of CKD by both calcimimetics or oral phosphorus binders as well as after changes in calcium and phosphate (Fig. 11.3) (Nechama et al. 2009a).

6.1 The Peptidyl-Prolyl Isomerase Pin1 Determines PTH mRNA Levels and Stability in Secondary Hyperparathyroidism

The above results indicate that KSRP and AUF1 directly or indirectly respond to changes in serum calcium and phosphate concentrations and CKD by altering their association with PTH mRNA leading to differences in PTH mRNA stability and levels (Nechama et al. 2008, 2009a). These changes could be a result of post-translational modifications of these ARE binding proteins, affecting their binding affinity to the PTH mRNA. Indeed, as stated above, AUF1 is post-translationally modified in the parathyroids of 2HPT rats and this is at least in part due to differences in protein phosphorylation (Bell et al. 2005c; Levi et al. 2006). KSRP is also a phospho-protein with two identified phosphorylation sites at serine 193 (S193) and threonine residue 692 (T692). Phosphorylation at these sites prevents KSRP association with the ribonuclease degradation complex exosome (S193) or compromises KSRP binding to ARE-containing target mRNAs (T692) and hence their decay (Gherzi et al. 2006; Ruggiero et al. 2007). Therefore, the differential interaction of KSRP and AUF1 with PTH mRNA after changes in serum calcium and phosphate may involve KSRP and AUF1 post-translational modifications.

The peptidyl prolyl cis/trans isomerase Pin1 specifically binds phosphorylated serine/threonine-proline protein motifs and catalyzes the cis/trans isomerization of the peptide bonds thereby changing the biological activity, phosphorylation and turn-over of its target proteins (Wulf et al. 2002; Zhou et al. 2000). Pin1-catalysed conformational regulation has a profound impact on many key proteins involved in various cell functions (Winkler et al. 2000; Lu et al. 1999). Pin1 was shown to regulate the turnover of ARE containing mRNAs, mainly cytokine mRNAs, through the interaction and isomerization of ARE binding proteins. Pin1 interacts with AUF1 and thereby stabilizes both GM-CSF and TGF^β mRNAs (Shen et al. 2005, 2008). These observations led us to speculate that Pin1 may be involved in PTH gene expression through AUF1 and/or KSRP interaction and isomerization. Indeed, Pin1 is a PTH mRNA destabilizing protein in vivo and in vitro (Nechama et al. 2009c). The regulation of PTH mRNA stability by Pin1 was mediated by the PTH mRNA 3'-UTR ARE and by the mRNA destabilizing protein KSRP. We showed for the first time that KSRP is a Pin1 target protein. Pin1 interacts with phosphorylated KSRP at S181, leading to KSRP dephosphorylation and activation. Importantly, Pin1 enzymatic activity was decreased in parathyroid extracts from rats with 2HPT due to either a calcium depleted diet or CKD. Pharmacological inhibition of Pin1 increased PTH mRNA levels and stability and decreased KSRP-PTH mRNA interaction in the parathyroid. This decreased interaction would increase PTH gene expression after Pin1 inhibition. Furthermore, Pin1-/- mice display increased serum PTH and PTH mRNA levels. Therefore, Pin1 determines basal PTH expression in vivo and in vitro and decreased Pin1 activity correlates with increased PTH mRNA levels in rats with 2HPT. These results demonstrate that Pin1 is a key mediator of PTH mRNA stability and indicate a role for Pin1 in the pathogenesis of the 2HPT of CKD (Nechama et al. 2009c). Our data suggest that phosphorylated KSRP at S181 is inactive. Upon interaction with Pin1, *cis-trans* isomerization of
the proline bond in KSRP leads to conformational change, exposing the phosphorylated S181 residue and possibly additional phosphorylation sites. This leads to KSRP dephosphorylation by a still unidentified phosphatase. KSRP then interacts with PTH mRNA and enhances its decay. A low calcium diet and CKD lead to decreased Pin1 isomerase activity in the parathyroids of these rats. This decreased Pin1 activity would prevent KSRP dephosphorylation, resulting in decreased KSRP-PTH mRNA interaction, inhibition of PTH mRNA degradation, and increased PTH mRNA levels (Fig. 11.3a). The trigger for the reduced Pin1 activity in the parathyroid glands of 2HPT rats is not known. Post-translational modifications of Pin1 protein itself may play a role in this regulation. Pin1 is post-translationally modified by phosphorylation that affects its ability to interact with target proteins and its activity. PKA mediated phosphorylation at Ser residue 16 affects the interaction of Pin1 with its target proteins. Phosphorylation at Ser residue 71 by the protein kinase DAPK1, inhibits Pin1 isomerization activity (Lu et al. 2002; Lee et al. 2011). Future studies may identify the factors that decrease Pin1 activity in the hyper-functioning parathyroid glands of 2HPT. Another main challenge will be to unravel the cell signaling cascade and the particular kinases responsible for KSRP phosphorylation at S181 that determine PTH mRNA-KSRP interactions and PTH mRNA decay.

7 Parathyroid miRNAs Are Essential for the Response of the Parathyroid to Hypocalcemia and Uremia

miRNAs are abundant, in part cell type-specific short non-coding RNA molecules that affect gene expression by sequence-specific translation repression and/or mRNA degradation. miRNAs have crucial roles in developmental processes and disease states. miRNAs are transcribed as long pri-miRNA transcripts that undergo two processing reaction steps (Gurtan and Sharp 2013). Drosha cleavage releases a hairpin precursor miRNA (pre-miRNA) of ~70 nucleotides that is exported to the cytoplasm and further processed by Dicer. Dicer, along with AGO and TRBP is a component of the RISC (RNA-induced silencing complex) loading complex which is guided by the functional strand of the mature (~22 nt) miRNA to target mRNAs (Fig. 11.4) (Bartel 2009). Lack of functional Dicer blocks pre-miRNA processing of to their mature form, thus perturbing any regulatory circuit in which miRNAs are involved (Bernstein et al. 2003). Total body inactivation of Dicer1 in both mice and zebrafish results in early embryonic lethality indicating that miRNAs are crucial for normal development (Giraldez et al. 2005; Wienholds et al. 2003). Conditional knock-out of Dicer1 in various organs or cell lineages has shown that Dicer is required for normal function of many cell types or organs, suggesting an essential role of Dicer-dependent miRNAs in normal physiology.

There is no information about miRNA function in parathyroid physiology. miRNA profiling was recently reported from parathyroid glands of patients with the rare parathyroid carcinoma with almost no information data on SHP (Corbetta et al. 2010; Rahbari et al. 2011; Vaira et al. 2012). Three miRNAs (miR-26b, miR-30b and miR-126) were significantly dysregulated between parathyroid carcinoma and parathyroid adenoma with down-regulation of miR-126 being the most significant



translation repression and/or mRNA degradation

Fig. 11.4 miRNA maturation. Simplified presentation of miRNA maturation. Primary miRNA (primiRNA) is transcribed and cleaved by Drosha in the nucleus. The resulting precursor miRNA (pre-miRNA) exits to the cytoplasm and is processed by Dicer to form the mature miRNA. The miRNA enters the RNA-induced silencing complex (RISC) and leads to sequence-specific translational suppression or degradation of target mRNAs. Conditional tissue specific knock-out of *Dicer* in a specific organ is a useful approach to study the function of Dicer-dependent miRNAs in vivo

(Rahbari et al. 2011). Multiple Endocrine Neoplasia type 1 (MEN1) syndrome which is characterized by tumors of the parathyroids, neuroendocrine cells, cells of the gastro-entero-pancreatic tract and of the anterior pituitary is due to mutations in the *MEN1* gene. *MEN1* tumorigenesis may be under the control of a negative feedback loop between miR-24 and menin protein expression (Luzi et al. 2012).

To study the significance of miRNAs in parathyroid physiology and the development of SHP we have generated parathyroid specific *Dicer1* knock-out (PT-*Dicer^{-/-}*) mice that do not express miRNAs specifically in their parathyroids (Fig. 11.4). The mice developed normally and are fertile. Remarkably, the PT-*Dicer^{-/-}* mice did not increase serum PTH after acute hypocalcemia. In addition, in vitro in parathyroid organ culture, PTH secretion was markedly impaired in parathyroids from the PT-*Dicer^{-/-}* mice when the glands were incubated in a calcium depleted medium. Similarly, PT-*Dicer^{-/-}* mice failed to increase serum PTH, PTH mRNA levels and parathyroid cell proliferation after the stimuli of dietary induced prolonged hypocalcemia. Moreover, adenine induced uremia led only to a moderate increase in serum PTH in the PT*Dicer^{-/-}* mice compared to controls. In contrast, the PT-*Dicer*^{-/-} mice responded normally to activation of the parathyroid calcium receptor (CaR) by both hypercalcemia and a calcimimetic. These findings show that miRNAs are essential for the response of the parathyroid to both acute and chronic hypocalcemia and uremia which are the major stimuli for PTH secretion. Therefore, miRNAs are crucial to activation of parathyroid gland function at the levels of PTH secretion, gene expression and parathyroid cell proliferation, providing a further level of the post-transcriptional regulation of PTH gene expression.

8 Conclusions

The parathyroid is regulated at the levels of PTH secretion, gene expression and parathyroid cell proliferation (Fig. 11.5). Dietary induced hypocalcemia, hypophosphatemia and CKD determine PTH gene expression post-transcriptionally by the interaction of RNA binding proteins to the PTH mRNA 3'-UTR ARE. Pin1 enzymatic activity affects these protein-PTH mRNA interactions and consequently PTH mRNA decay. Changes in Pin1 enzymatic activity alter KSRP phosphorylation status and KSRP-AUF1-PTH mRNA interactions in the parathyroid cell and hence PTH mRNA stability and levels (Figs. 11.3 and 11.5). In addition, miRNAs are essential for activation of parathyroid gland function at the levels of PTH secretion, gene expression and parathyroid cell proliferation after acute and chronic hypocalcaemia and CKD (Fig. 11.5).



Fig. 11.5 Regulation of PTH gene expression, secretion and parathyroid cell proliferation. The parathyroid cell synthesizes and secretes PTH unless it is restrained by the parathyroid calcium receptor (CaR) which senses extracellular serum calcium levels. A low serum calcium leads to increased PTH mRNA stability and levels, PTH secretion, and parathyroid cell proliferation. A high serum phosphate and chronic kidney disease (CKD) lead to similar changes in all these parameters. PTH mRNA stability is regulated by the balanced interactions of the protective

References

- Almaden Y, Canalejo A, Hernandez A, Ballesteros E, Garcia-Navarro S, Torres A, Rodriguez M (1996) Direct effect of phosphorus on parathyroid hormone secretion from whole rat parathyroid glands in vitro. J Bone Miner Res 11:970–976
- Almaden Y, Hernandez A, Torregrosa V, Canalejo A, Sabate L, Fernandez CL, Campistol JM, Torres A, Rodriguez M (1998) High phosphate level directly stimulates parathyroid hormone secretion and synthesis by human parathyroid tissue in vitro. J Am Soc Nephrol 9(10):1845–1852
- Barreau C, Paillard L, Osborne HB (2006) AU-rich elements and associated factors: are there unifying principles? Nucleic Acids Res 33(22):7138–7150
- Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136(2):215-233
- Bell O, Gaberman E, Kilav R, Levi R, Cox KB, Molkentin JD, Silver J, Naveh-Many T (2005a) The protein phosphatase calcineurin determines basal parathyroid hormone gene expression. Mol Endocrinol 19:516–526
- Bell O, Silver J, Naveh-Many T (2005b) Identification and characterization of *cis*-acting elements in the human and bovine parathyroid hormone mRNA 3'-untranslated region. J Bone Miner Res 20:858–866
- Bell O, Silver J, Naveh-Many T (2005c) Parathyroid hormone, from gene to protein. In: Naveh-Many T (ed) Molecular biology of the parathyroid, 1st edn. Landes Bioscience and Kluwer Academic/Plenum Publishers, New York, pp 8–28
- Ben Dov IZ, Galitzer H, Lavi-Moshayoff V, Goetz R, Kuro-o M, Mohammadi M, Sirkis R, Naveh-Many T, Silver J (2007a) The parathyroid is a target organ for FGF23 in rats. J Clin Invest 117(12):4003–4008
- Ben Dov IZ, Pappo O, Sklair-Levy M, Galitzer H, Ilan Y, Naveh-Many T, Silver J (2007b) Lanthanum carbonate decreases PTH gene expression with no hepatotoxicity in uraemic rats. Nephrol Dial Transplant 22(2):362–368
- Bernstein E, Kim SY, Carmell MA, Murchison EP, Alcorn H, Li MZ, Mills AA, Elledge SJ, Anderson KV, Hannon GJ (2003) Dicer is essential for mouse development. Nat Genet 35(3):215–217
- Block GA, Martin KJ, de Francisco AL, Turner SA, Avram MM, Suranyi MG, Hercz G, Cunningham J, Abu-Alfa AK, Messa P, Coyne DW, Locatelli F, Cohen RM, Evenepoel P, Moe SM, Fournier A, Braun J, McCary LC, Zani VJ, Olson KA, Drueke TB, Goodman WG (2004) Cinacalcet for secondary hyperparathyroidism in patients receiving hemodialysis. N Engl J Med 350(15):1516–1525
- Brewer G (2002) Messenger RNA decay during aging and development. Ageing Res Rev 1(4):607-625
- Brown EM, Gamba G, Riccardi D, Lombardi M, Butters R, Kifor O, Sun A, Hediger MA, Lytton J, Hebert J (1993) Cloning and characterization of an extracellular Ca²⁺-sensing receptor from bovine parathyroid. Nature 366:575–580
- Chernokalskaya E, Dubell AN, Cunningham KS, Hanson MN, Dompenciel RE, Schoenberg DR (1998) A polysomal ribonuclease involved in the destabilization of albumin mRNA is a novel member of the peroxidase gene family. RNA 4(12):1537–1548

Fig. 11.5 (continued) parathyroid *trans* acting factors AUF1 and Unr and the decay promoting protein KSRP, that bind to a defined *cis* element in the PTH mRNA 3'-UTR. These interactions are regulated by the activity of the peptidyl prolyl *cis/trans* isomerase Pin1. In addition, miRNAs are essential for activation of parathyroid gland function at the levels of PTH secretion, gene expression and parathyroid cell proliferation after both acute and chronic hypocalcaemia and CKD. 1,25D (1,25(OH)₂D) decreases PTH gene transcription and parathyroid cell proliferation. Fibroblast growth factor 23 (FGF23) is a bone derived hormone that decreases PTH gene expression and secretion by acting through its Kloth FGFR1 (fibroblast growth factor 1) receptor complex. Reproduced from Silver (2001)

- Chou CF, Mulky A, Maitra S, Lin WJ, Gherzi R, Kappes J, Chen CY (2006) Tethering KSRP, a decay-promoting AU-rich element-binding protein, to mRNAs elicits mRNA decay. Mol Cell Biol 26(10):3695–3706
- Corbetta S, Vaira V, Guarnieri V, Scillitani A, Eller-Vainicher C, Ferrero S, Vicentini L, Chiodini I, Bisceglia M, Beck-Peccoz P, Bosari S, Spada A (2010) Differential expression of microRNAs in human parathyroid carcinomas compared with normal parathyroid tissue. Endocr Relat Cancer 17(1):135–146
- Demay MB, Kiernan MS, DeLuca HF, Kronenberg HM (1992) Sequences in the human parathyroid hormone gene that bind the l,25-dihydroxyvitamin D-3 receptor and mediate transcriptional repression in response to l,25-dihydroxyvitamin D-3. Proc Natl Acad Sci U S A 89:8097–8101
- D'Haese PC, Spasovski GB, Sikole A, Hutchison A, Freemont TJ, Sulkova S, Swanepoel C, Pejanovic S, Djukanovic L, Balducci A, Coen G, Sulowicz W, Ferreira A, Torres A, Curic S, Popovic M, Dimkovic N, De Broe ME (2003) A multicenter study on the effects of lanthanum carbonate (Fosrenol) and calcium carbonate on renal bone disease in dialysis patients. Kidney Int 63(Suppl 85):S73–S78
- Dinur M, Kilav R, Sela-Brown A, Jacquemin-Sablon H, Naveh-Many T (2006) In vitro evidence that upstream of N-ras participates in the regulation of parathyroid hormone messenger ribonucleic acid stability. Mol Endocrinol 20(7):1652–1660
- Fritz DT, Ford LP, Wilusz J (2000) An in vitro assay to study regulated mRNA stability. Sci STKE 2000(61):L1
- Fujiki R, Kim MS, Sasaki Y, Yoshimura K, Kitagawa H, Kato S (2005) Ligand-induced transrepression by VDR through association of WSTF with acetylated histones. EMBO J 24(22):3881–3894
- Galitzer H, Ben Dov IZ, Silver J, Naveh-Many T (2010) Parathyroid cell resistance to fibroblast growth factor 23 in secondary hyperparathyroidism of chronic kidney disease. Kidney Int 77(3):211–218
- Gherzi R, Lee KY, Briata P, Wegmuller D, Moroni C, Karin M, Chen CY (2004) A KH domain RNA binding protein, KSRP, promotes ARE-directed mRNA turnover by recruiting the degradation machinery. Mol Cell 14(5):571–583
- Gherzi R, Trabucchi M, Ponassi M, Ruggiero T, Corte G, Moroni C, Chen CY, Khabar KS, Andersen JS, Briata P (2006) The RNA-binding protein KSRP promotes decay of beta-catenin mRNA and is inactivated by PI3K-AKT signaling. PLoS Biol 5(1):e5
- Giraldez AJ, Cinalli RM, Glasner ME, Enright AJ, Thomson JM, Baskerville S, Hammond SM, Bartel DP, Schier AF (2005) MicroRNAs regulate brain morphogenesis in zebrafish. Science 308(5723):833–838
- Gurtan AM, Sharp PA (2013) The role of miRNAs in regulating gene expression networks. J Mol Biol 425(19):3582–3600
- Habener JF, Rosenblatt M, Potts JTJ (1984) Parathyroid hormone: biochemical aspects of biosynthesis, secretion, action, and metabolism. Physiol Rev 64:985–1053
- Joy MS, Finn WF (2003) Randomized, double-blind, placebo-controlled, dose-titration, phase III study assessing the efficacy and tolerability of lanthanum carbonate: a new phosphate binder for the treatment of hyperphosphatemia. Am J Kidney Dis 42(1):96–107
- Kemper B (1986) Molecular biology of parathyroid hormone. CRC Crit Rev Biochem 19:353–379
- Kilav R, Silver J, Naveh-Many T (1995) Parathyroid hormone gene expression in hypophosphatemic rats. J Clin Invest 96:327–333
- Kilav R, Silver J, Naveh-Many T (2001) A conserved cis-acting element in the parathyroid hormone 3'-untranslated region is sufficient for regulation of RNA stability by calcium and phosphate. J Biol Chem 276:8727–8733
- Kilav R, Bell O, Le SY, Silver J, Naveh-Many T (2004) The parathyroid hormone mRNA 3'-untranslated region AU-rich element is an unstructured functional element. J Biol Chem 279(3):2109–2116

- Komaba H, Goto S, Fujii H, Hamada Y, Kobayashi A, Shibuya K, Tominaga Y, Otsuki N, Nibu KI, Nakagawa K, Tsugawa N, Okano T, Kitazawa R, Fukagawa M (2010) Depressed expression of Klotho and FGF receptor 1 in hyperplastic parathyroid glands from uremic patients. Kidney Int 77:232–238
- Kurosu H, Ogawa Y, Miyoshi M, Yamamoto M, Nandi A, Rosenblatt KP, Baum MG, Schiavi S, Hu MC, Moe OW, Kuro-o M (2006) Regulation of fibroblast growth factor-23 signaling by Klotho. J Biol Chem 281(10):6120–6123
- Lee TH, Chen CH, Suizu F, Huang PY, Schiene-Fischer C, Daum S, Zhang YJ, Goate A, Chen RH, Zhou XZ, Lu KP (2011) Death-associated protein kinase 1 phosphorylates Pin1 and inhibits its prolyl isomerase activity and cellular function. Mol Cell 42(2):147–159
- Lellek H, Kirsten R, Diehl I, Apostel F, Buck F, Greeve J (2000) Purification and molecular cloning of a novel essential component of the apolipoprotein B mRNA editing enzyme-complex. J Biol Chem 275(26):19848–19856
- Levi R, Ben Dov IZ, Lavi-Moshayoff V, Dinur M, Martin D, Naveh-Many T, Silver J (2006) Increased parathyroid hormone gene expression in secondary hyperparathyroidism of experimental uremia is reversed by calcimimetics: correlation with posttranslational modification of the trans acting factor AUF1. J Am Soc Nephrol 17(1):107–112
- Linker K, Pautz A, Fechir M, Hubrich T, Greeve J, Kleinert H (2005) Involvement of KSRP in the post-transcriptional regulation of human iNOS expression-complex interplay of KSRP with TTP and HuR. Nucleic Acids Res 33(15):4813–4827
- Lu PJ, Wulf G, Zhou XZ, Davies P, Lu KP (1999) The prolyl isomerase Pin1 restores the function of Alzheimer-associated phosphorylated tau protein. Nature 399(6738):784–788
- Lu PJ, Zhou XZ, Liou YC, Noel JP, Lu KP (2002) Critical role of WW domain phosphorylation in regulating phosphoserine binding activity and Pin1 function. J Biol Chem 277(4):2381–2384
- Luzi E, Marini F, Giusti F, Galli G, Cavalli L, Brandi ML (2012) The negative feedback-loop between the oncomir Mir-24-1 and menin modulates the Men1 tumorigenesis by mimicking the "Knudson's Second Hit". PLoS One 7(6):e39767
- Min H, Turck CW, Nikolic JM, Black DL (1997) A new regulatory protein, KSRP, mediates exon inclusion through an intronic splicing enhancer. Genes Dev 11(8):1023–1036
- Moallem E, Silver J, Kilav R, Naveh-Many T (1998) RNA protein binding and post-transcriptional regulation of PTH gene expression by calcium and phosphate. J Biol Chem 273:5253–5259
- Moe SM, Cunningham J, Bommer J, Adler S, Rosansky SJ, Urena-Torres P, Albizem MB, Guo MD, Zani VJ, Goodman WG, Sprague SM (2005) Long-term treatment of secondary hyperparathyroidism with the calcimimetic cinacalcet HCl. Nephrol Dial Transplant 20(10): 2186–2193
- Murayama A, Kim MS, Yanagisawa J, Takeyama K, Kato S (2004) Transrepression by a liganded nuclear receptor via a bHLH activator through co-regulator switching. EMBO J 23(7): 1598–1608
- Naveh-Many T (2005) Molecular biology of the parathyroid, 1st edn. Landes Bioscience/Eurekah. com and Kluwer Academic Press, Georgetown, Texas and New York
- Naveh-Many T, Nechama M (2007) Regulation of parathyroid hormone mRNA stability by calcium, phosphate and uremia. Curr Opin Nephrol Hypertens 16(4):305–310
- Naveh-Many T, Silver J (1990) Regulation of parathyroid hormone gene expression by hypocalcemia, hypercalcemia, and vitamin D in the rat. J Clin Invest 86:1313–1319
- Naveh-Many T, Marx R, Keshet E, Pike JW, Silver J (1990) Regulation of 1,25-dihydroxyvitamin D3 receptor gene expression by 1,25-dihydroxyvitamin D3 in the parathyroid in vivo. J Clin Invest 86:1968–1975
- Naveh-Many T, Rahamimov R, Livni N, Silver J (1995) Parathyroid cell proliferation in normal and chronic renal failure rats: the effects of calcium, phosphate and vitamin D. J Clin Invest 96:1786–1793
- Nechama M, Ben Dov IZ, Briata P, Gherzi R, Naveh-Many T (2008) The mRNA decay promoting factor K-homology splicing regulator protein post-transcriptionally determines parathyroid hormone mRNA levels. FASEB J 22:3458–3468

- Nechama M, Ben Dov IZ, Silver J, Naveh-Many T (2009a) Regulation of PTH mRNA stability by the calcimimetic R568 and the phosphorus binder lanthanum carbonate in CKD. AJP Renal Physiol 296(4):F795–F800
- Nechama M, Peng Y, Bell O, Briata P, Gherzi R, Schoenberg DR, Naveh-Many T (2009b) KSRP-PMR1-exosome association determines parathyroid hormone mRNA levels and stability in transfected cells. BMC Cell Biol 10:70–81
- Nechama M, Uchida T, Yosef-Levi IM, Silver J, Naveh-Many T (2009c) The peptidyl-prolyl isomerase Pin1 determines parathyroid hormone mRNA levels and stability in rat models of secondary hyperparathyroidism. J Clin Invest 119(10):3102–3114
- Nielsen PK, Feldt-Rasmusen U, Olgaard K (1996) A direct effect of phosphate on PTH release from bovine parathyroid tissue slices but not from dispersed parathyroid cells. Nephrol Dial Transplant 11:1762–1768
- Rahbari R, Holloway AK, He M, Khanafshar E, Clark OH, Kebebew E (2011) Identification of differentially expressed MicroRNA in parathyroid tumors. Ann Surg Oncol 18(4):1158–1165
- Rodriguez M, Almaden Y, Hernandez A, Torres A (1996) Effect of phosphate on the parathyroid gland: direct and indirect? Curr Opin Nephrol Hypertens 5(4):321–328
- Ruggiero T, Trabucchi M, Ponassi M, Corte G, Chen CY, al Haj L, Khabar KS, Briata P, Gherzi R (2007) Identification of a set of KSRP target transcripts upregulated by PI3K-AKT signaling. BMC Mol Biol 8:28
- Russell J, Lettieri D, Sherwood LM (1986) Suppression by 1,25(OH)2D3 of transcription of the pre-proparathyroid hormone gene. Endocrinology 119:2864–2866
- Sela-Brown A, Silver J, Brewer G, Naveh-Many T (2000) Identification of AUF1 as a parathyroid hormone mRNA 3'-untranslated region binding protein that determines parathyroid hormone mRNA stability. J Biol Chem 275(10):7424–7429
- Shen ZJ, Esnault S, Malter JS (2005) The peptidyl-prolyl isomerase Pin1 regulates the stability of granulocyte-macrophage colony-stimulating factor mRNA in activated eosinophils. Nat Immunol 6(12):1280–1287
- Shen ZJ, Esnault S, Rosenthal LA, Szakaly RJ, Sorkness RL, Westmark PR, Sandor M, Malter JS (2008) Pin1 regulates TGF-beta1 production by activated human and murine eosinophils and contributes to allergic lung fibrosis. J Clin Invest 118(2):479–490
- Silver J (2001) Cycling with the parathyroid. J Clin Invest 107(9):1079-1080
- Silver J, Naveh-Many T (2009) Phosphate and the parathyroid. Kidney Int 75(9):898-905
- Silver J, Russell J, Sherwood LM (1985) Regulation by vitamin D metabolites of messenger ribonucleic acid for preproparathyroid hormone in isolated bovine parathyroid cells. Proc Natl Acad Sci U S A 82:4270–4273
- Silver J, Naveh-Many T, Mayer H, Schmelzer HJ, Popovtzer MM (1986) Regulation by vitamin D metabolites of parathyroid hormone gene transcription in vivo in the rat. J Clin Invest 78: 1296–1301
- Silver J, Kilav R, Naveh-Many T (2002) Mechanisms of secondary hyperparathyroidism. Am J Physiol Renal Physiol 283(3):F367–F376
- Slatopolsky E, Finch J, Denda M, Ritter C, Zhong A, Dusso A, MacDonald P, Brown AJ (1996) Phosphate restriction prevents parathyroid cell growth in uremic rats. High phosphate directly stimulates PTH secretion in vitro. J Clin Invest 97:2534–2540
- Snee M, Kidd GJ, Munro TP, Smith R (2002) RNA trafficking and stabilization elements associate with multiple brain proteins. J Cell Sci 115(Pt 23):4661–4669
- Vaira V, Elli F, Forno I, Guarnieri V, Verdelli C, Ferrero S, Scillitani A, Vicentini L, Cetani F, Mantovani G, Spada A, Bosari S, Corbetta S (2012) The microRNA cluster C19MC is deregulated in parathyroid tumours. J Mol Endocrinol 49(2):115–124
- Wagner BJ, DeMaria CT, Sun Y, Wilson GM, Brewer G (1998) Structure and genomic organization of the human AUF1 gene: alternative pre-mRNA splicing generates four protein isoforms. Genomics 48(2):195–202
- Wienholds E, Koudijs MJ, van Eeden FJM, Cuppen E, Plasterk RHA (2003) The microRNAproducing enzyme Dicer1 is essential for zebrafish development. Nat Genet 35(3):217–218

- Wilusz CJ, Wilusz J (2004) Bringing the role of mRNA decay in the control of gene expression into focus. Trends Genet 20(10):491–497
- Winkler KE, Swenson KI, Kornbluth S, Means AR (2000) Requirement of the prolyl isomerase Pin1 for the replication checkpoint. Science 287(5458):1644–1647
- Wulf GM, Liou YC, Ryo A, Lee SW, Lu KP (2002) Role of Pin1 in the regulation of p53 stability and p21 transactivation, and cell cycle checkpoints in response to DNA damage. J Biol Chem 277(50):47976–47979
- Yalcindag C, Silver J, Naveh-Many T (1999) Mechanism of increased parathyroid hormone mRNA in experimental uremia: roles of protein RNA binding and RNA degradation. J Am Soc Nephrol 10(12):2562–2568
- Zhou XZ, Kops O, Werner A, Lu PJ, Shen M, Stoller G, Kullertz G, Stark M, Fischer G, Lu KP (2000) Pin1-dependent prolyl isomerization regulates dephosphorylation of Cdc25C and tau proteins. Mol Cell 6(4):873–883

Post-transcriptional and Post-translational Regulation of Steroidogenesis

Wen-Jun Shen, Zhigang Hu, Jie Hu, Fredric B. Kraemer, and Salman Azhar

Abbreviations

ACAT1	Acyl-coenzyme A:cholesterol acyltransferase 1
ACTH	Adrenocorticotropic hormone
AII	Angiotensin II
ANC	Adenine nucleotide transporter

W.-J. Shen

Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, 3801 Miranda Avenue, Palo Alto, CA 94304, USA

Division of Endocrinology, Stanford University, Stanford, CA 94305, USA

Z. Hu

Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, 3801 Miranda Avenue, Palo Alto, CA 94304, USA

Division of Endocrinology, Stanford University, Stanford, CA 94305, USA

Jiangsu Key Laboratory for Molecular and Medical Biotechnology and College of Life Science, Nanjing Normal University, Nanjing 210046, China

J. Hu

Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, 3801 Miranda Avenue, Palo Alto, CA 94304, USA

F.B. Kraemer Division of Endocrinology, Stanford University, Stanford, CA 94305, USA

S. Azhar (🖂)

Geriatric Research, Education and Clinical Center (GRECC-182B), Veterans Affairs Palo Alto Health Care System, 3801 Miranda Avenue, Palo Alto, CA 94304, USA

Division of Gastroenterology and Hepatology, Stanford University, Stanford, CA 94305, USA e-mail: salman.azhar@va.gov

© Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_12

AALO	Allopregnanolone
CEH	Neutral cholesteryl ester hydrolase
CE	Cholesteryl ester
CEs	Cholesteryl esters
ECD	Extracellular domain
ER	Endoplasmic reticulum
FRET	Quantitative fluorescence resonance energy transfer
FSH	Follicle-stimulating hormone
hCG	Human chorionic gonadotropin
HDL	High-density lipoprotein
HSL	Hormone-sensitive lipase
IMM	Inner mitochondrial membrane
LDL	Low-density lipoprotein
LH	Luteinizing hormone
OMM	Outer mitochondrial membrane
NSF	N-ethylmaleimide-sensitive factor
CYP11A1	P450c11A
CYP11B1	P450c11
CYP11B2	Aldosterone synthase
CYP17	P450c17
CYP21A2	P450c21
PBR	Peripheral-type benzodiazepine receptor
PDZ	PSD-95, DglA, ZO-1
PKA	cAMP-dependent protein kinase
SR-BI	Scavenger receptor Class B, type I
StAR	Steroidogenic acute regulatory protein
TSPO	Translocator protein

1 Introduction

Steroid hormones play important roles in virtually every aspect of cellular metabolism, including the regulation of carbohydrate, lipid and protein metabolism and immune function (glucocorticoids), as well as salt and water balance and blood pressure regulation (mineralocorticoids). They are also critically involved in the maintenance of secondary sex characteristics, reproductive functions and muscle and bone growth (testosterone, progestins and estrogens) (Hu et al. 2010). The common precursor of steroid hormone biosynthesis is cholesterol, and steroids are synthesized most prominently in the steroidogenic cells of the adrenal gland, gonads and placenta (Payne and Hales 2004; LaVoie and King 2009; Miller and Bose 2011; Strauss et al. 1996). Brain also synthesizes steroids, which are commonly referred to as "neurosteroids". However, circulating steroid hormones (progesterone, deoxycorticosterone, testosterone, estradiol), and not cholesterol, serve as precursors for neurosteroids, which are produced locally in the hippocampus and other brain regions (Mellon and Griffin 2002; Reddy 2010; Giatti et al. 2012).

1.1 Overview of Steroidogenesis

In steroidogenic cells of the adrenal gland, ovary and testis, the overall rate of steroid hormone production is controlled by tropic (peptide) hormones. Therefore, the type of steroid hormone that can be synthesized by a particular cell type is dictated by its complement of peptide hormone receptor, its response to peptide hormone stimulation and its genetically expressed complement of steroidogenic enzymes. Thus, adrenocorticotropic hormone (ACTH) stimulates cortisol/corticosterone in adrenocortical fasciculata-reticularis cells; angiotensin II (AII) and potassium regulate aldosterone synthesis in adrenal glomerulosa cells; follicle-stimulating hormone (FSH) controls the progesterone and estrogen synthesis in ovarian granulosa cells, whereas luteinizing hormone (LH) regulates progesterone synthesis in luteinized ovarian granulosa-luteal cells, androgen production in ovarian theca-interstitial cells and testosterone synthesis in testicular Leydig cells. The adrenal gland is also responsible for the synthesis of adrenal androgens (McKenna et al. 1997; Miller 2002). Tropic hormones (LH, FSH or ACTH) induce adrenocortical and gonadal steroidogenesis by binding to their respective G protein-coupled receptors, leading to activation of adenylate cyclase, which generates cAMP and activates cAMP-dependent protein kinase (PKA) (Marsh 1976; Simpson and Waternan 1983; Sanorn et al. 1980; Strauss et al. 1988). Stimulation of the cAMP-PKA signaling cascade exerts both acute and chronic effects on the regulation of steroid hormone production. Angiotensin (AII) stimulation of aldosterone biosynthesis in adrenal glomerulosa cells is primarily mediated by the protein kinase C signaling cascade, whereas potassium stimulation of aldosterone production also involves Ca2+-calmodulin-dependent kinase (Spat and Hunyady 2004).

The trophic hormone (ACTH, LH, FSH) regulation of adrenal and gonadal steroidogenesis is subject to both acute (Miller and Bose 2011; Pon et al. 1986; Pon and Orme-Johnson 1988; Epstein and Orme-Johnson 1991a; Stocco and Clark 1996) and chronic regulation (LaVoie and King 2009; Miller and Bose 2011; Miller 1988; Simpson and Waterman 1988; Payne et al. 1992; Simpson et al. 1992). Acute steroid synthesis that occurs over minutes in response to trophic hormone stimulation is controlled at the level of cholesterol delivery to the inner mitochondrial membrane (IMM) for the first enzymatic step in the pathway, the conversion of cholesterol to pregnenolone by the P450scc. This rate limiting step, i.e., cholesterol transfer from the outer mitochondrial membrane (OMM) to the IMM, is dependent upon the trophic hormone stimulated rapid increase in transcription, as well as post-transcriptional modification, of the steroidogenic acute regulatory protein (StAR) (Hu et al. 2010; Miller and Bose 2011; Stocco 2001; Manna et al. 2009; Rone et al. 2009). Chronic stimulation (hours to days) occurs through the induction of P450scc gene transcription leading to increased P450scc and consequent increased steroidogenic capacity (Miller and Bose 2011; Miller 1988; Simpson and Waterman 1988; Payne et al. 1992; Simpson et al. 1992).

1.2 Steroid Biosynthetic Pathways

The process of steroid hormone synthesis, or steroidogenesis, represents a complex multistep and multi-enzyme process by which precursor cholesterol is converted to pregnenolone and subsequently metabolized into other biologically active steroids in a tissue specific manner (Hu et al. 2010; Payne and Hales 2004; LaVoie and King 2009; Miller and Bose 2011). This process can be broadly divided into five major steps: (1) acquisition of cholesterol from exogenous (lipoproteins) and endogenous (de novo synthesis) sources for storage in the form of cholesterol esters (CEs) in lipid droplets, (2) mobilization of cholesterol from lipid droplet stored CEs, (3) transport of cholesterol to and from the outer mitochondrial membrane (OMM) to the inner mitochondrial membrane (IMM), where cytochrome P450 side chain cleavage enzyme (P450scc, encoded by CYP11A1) is localized, (4) P450scc catalyzed cleavage of a 6-carbon unit from the cholesterol side chain producing pregnenolone, the common precursor for the synthesis of all of the other steroid hormones, and (5) efflux of pregnenolone from the mitochondria to the endoplasmic reticulum (ER), where it is converted by ER enzymes into intermediate precursors, which further shuttle between mitochondria and ER for the tissue specific production of progestins, estrogens, androgens, glucocorticoids or mineralocorticoids (LaVoie and King 2009; Miller 2008). In rodents, the majority of cholesterol needed for steroidogenesis in adrenal and ovary (and testicular Leydig cells under certain conditions) is obtained via SR-BI mediated selective delivery of HDL-cholesterol (Fig. 12.1), whereas in humans the bulk of the cholesterol needed for steroid synthesis is supplied by the LDL-receptor/endocytic pathway.

The major synthetic pathways for steroid hormones in the adrenal gland and gonads are shown in Fig. 12.2. Although the final steroid product differs in a tissue-specific manner, the first committed reaction in the biosynthetic pathway is the same, i.e., the conversion of cholesterol to pregnenolone by the cytochrome P450scc enzyme (CYP11A1). P450scc is an enzyme complex consisting of a flavoprotein (NADH-adrenodoxin reductase), a ferredox (adrenodoxin) and a cytochrome P450 localized on an inner mitochondrial membrane (Miller 1988, 2008). P450scc catalyzes three distinct reactions: 20α -hydroxylation, 22-hydroxylation and scission of 20, 22 carbon–carbon bonds, thus converting cholesterol to pregnenolone (Miller 1988, 2008). Next, the pregnenolone is converted to the final hormone product by sequential steps along the pathway depending on the enzymes that are present in that tissue (Fig. 12.2). Thus, adrenal zona fasciculata-reticularis and zona glomerulosa cells produce cortisol (corticosterone in rodents) and aldosterone, respectively. Likewise, ovarian theca cells secrete androgens, while granulosa cells synthesize progesterone and estradiol. Testicular Leydig cells primarily synthesize and secrete testosterone.

1.3 Post-transcriptional/Post-translational Regulation of Steroidogenesis

While much is known about the transcriptional regulation of steroidogenesis, relatively little is known about the post-transcriptional and post-translational regulation of this process. Below we summarize what is currently known about the post-transcriptional/post-translational modulation of steroidogenesis:



Fig. 12.1 Overview of steroidogenesis. Tropic hormones bind to their respective G proteincoupled receptors leading to activation of the cAMP-PKA signaling cascade which in turn induces acute and/or chronicle effects on the regulation of steroidogenesis. Upon hormone stimulation, cholesterol acquired from endocytic or selective uptake pathways or from de novo synthesis is mobilized from lipid droplets and transported to the inner membrane of mitochondria through the collective actions of HSL, StAR and their associated proteins. At the IMM, cytochrome P450 side chain cleavage enzyme catalyzes the cleavage of a 6-carbon unit from cholesterol to produce pregnenolone, which is effluxed from mitochondria to the endoplasmic reticulum and used as the precursor for the synthesis of all of the other steroid hormones

1.3.1 Phosphorylation/Dephosphorylation-Dependent Regulation of Steroidogenesis

Reversible phosphorylation of proteins, which can result in a conformational change in the structure in many enzymes and receptors, and in turn switches enzymes and receptors "on" and "off", is an important regulatory mechanism that occurs in both prokaryotic and eukaryotic organisms (Krebs and Fischer 1964; Fischer and Krebs 1966; Burnett and Kennedy 1954; Bahler et al. 1990). Phosphorylation commonly occurs on serine, threonine, tyrosine and histidine residues in eukaryotic proteins (Ciesla et al. 2011; Hanks et al. 1988; Cohen 2000). Several key transcription factors and enzymes that contribute to the regulation of steroidogenesis, including steroidogenic transcription factors, SF-1, DAX-1, CREB and TORC (Babu et al. 2000; Desclozeaux et al. 2002; Sands and Palmer 2008; Takemori and Okamoto 2008), StAR (Arakane et al. 1997), HSL (Kraemer and Shen 2002) and downstream enzymes such as CYP11A1 and CYP17A1 (Miller 2008; Vilgrain et al. 1984), have been shown to be regulated via phosphorylation/dephosphorylation mechanisms.

(1a) Phosphorylation of StAR:

Orme-Johnson and colleagues first identified StAR in ACTH treated primary rat adrenocortical cell suspensions (Pon et al. 1986; Krueger and Orme-Johnson 1983;



Fig. 12.2 The major synthetic pathways for steroid hormones

Epstein and Orme-Johnson 1991b). Using a two-dimensional gel electrophoresis of ³⁵S-radiolabeled proteins, a hormone-induced phosphoprotein was identified by radioautography, termed i_b and later as pp30. The synthesis of this protein was shown to occur with similar kinetics and dose response as corticosterone production after trophic hormone stimulation (Mellon and Griffin 2002; Stocco 2001; Manna et al. 2009). Subsequently, these seminal findings were further confirmed in the steroidogenic cells of the ovary and testis (Epstein and Orme-Johnson 1991a; Pon and Orme-Johnson 1998). Stocco and colleagues cloned this protein and renamed it steroidogenic acute regulatory protein, StAR (Clark et al. 1994). A number of in vitro studies have established that cAMP-dependent phosphorylation of murine StAR at serine residue 194 (Ser¹⁹⁴; Ser¹⁹⁵ in human StAR) is required for its maximal biological activity (Arakane et al. 1997; Fleury et al. 2004). Very recently, investigation into the functional importance of Ser¹⁹⁴ phosphorylation in vivo using transgenic mice expressing either wild-type (WT) StAR or StAR mutation S194A to rescue StAR deficient (knockout) mice has been reported (Sasaki et al. 2014). The results showed that, despite protein expression comparable to or higher than amounts seen with control animals or rescued with WT StAR, mutant S194A StAR did not rescue the neonatal lethality and only partially rescued the sex reversal in male mice observed uniformly in StAR KO mice. Like the StAR KO mice, the adrenal cortex and testicular Leydig cells contained abundant lipid droplets when tissue sections were stained with Oil red O for lipids. Moreover, adrenal StAR from S194A rescued animals lacked an acidic species, which was detectable in response to ACTH treatment in animals rescued with WT StAR. These results are consistent with defective StAR phosphorylation. On the other hand, the WT-StAR transgene consistently restored viability and steroidogenic function in StAR-deficient (-/- mice) (Sasaki et al. 2008). These findings further establish that phosphorylation of Ser¹⁹⁴ is essential for normal functioning of StAR protein in the adrenal cortex, testes and possibly ovaries of mice.

(1b) Phosphorylation of HSL:

Hormone sensitive lipase (HSL) was shown to be the neutral cholesterol esterase for the hormone-induced mobilization of stored CEs to supply precursor cholesterol for steroidogenesis (Kraemer and Shen 2002; Kraemer et al. 2002, 2004). Investigation into the functional significance of HSL in steroid production using HSL-/- mice showed that the cholesteryl ester content was substantially elevated in adrenals of HSL-/- mice, and basal corticosterone production was reduced approximately 50 %. The maximum corticosterone production induced by dibutyryl cAMP and lipoproteins was approximately 75-85 % lower in adrenal cells from HSL-/mice compared with control. There is no intrinsic defect in the conversion of cholesterol into steroids in HSL-/- mice. Dibutyryl cAMP-stimulated conversion of high-density lipoprotein CEs into corticosterone was reduced 97 % in HSL-/mice (Kraemer et al. 2004). The cholesteryl esterase activity of HSL is regulated by reversible phosphorylation/dephosphorylation (Yeaman 1990). A number of publications have reported PKA catalyzed phosphorylation of HSL at Ser⁵⁶³, Ser⁶⁵⁹ and Ser⁶⁶⁰ in rat HSL, resulting in increases in HSL hydrolytic activity (Yeaman 1990; Shen et al. 1998; Anthonsen et al. 1998). In addition, ERKs can phosphorylate adipocyte HSL on Ser⁶⁰⁰ and stimulate its activity (Greenberg et al. 2001). In contrast to activation of activity seen with PKA or ERK phosphorylation, other kinases, such as glycogen synthase kinase-4, Ca++/calmodulin-dependent protein kinase II, and AMP-activated protein kinase, phosphorylate HSL at a secondary basal site Ser⁵⁶⁵ in rat HSL (Yeaman 1990). Phosphorylation at Ser⁵⁶⁵ interferes with the phosphorylation of Ser⁵⁶³ by PKA (Yeaman 1990). HSL activity can be inactivated by protein phosphatases as well. The most active phosphatases against Ser⁵⁶³ are phosphatase 2A and 2C, while Ser⁵⁶⁵ is predominately dephosphorylated by phosphatase 2A (Wood et al. 1993).

(1c) Phosphorylation of Key Steroidogenic Transcription Factors and Enzymes:

SF-1, DAX-1, CREB and TORC have all been shown to be important transcription factors involved in steroidogenic pathways. SF1 can be phosphorylated by Erk2 at Ser²⁰³ in response to multiple components of the MAPK pathway and results in upregulated transcriptional activity (Babu et al. 2000; Desclozeaux et al. 2002). CREB can be phosphorylated by PKA as well as salt induced kinase in response to cellular stimuli and results in transcriptional activation of multiple targets involved in steroidogenesis (Sands and Palmer 2008; Takemori and Okamoto 2008). A few of the downstream enzymes, such as CYP11A1 and CYP17A1 (Miller 2008; Vilgrain et al. 1984), have been shown to be regulated via phosphorylation/dephosphorylation mechanisms.

1.3.2 Modulation of Steroidogenesis through Protein–Protein Interactions

In living organisms, cellular functions are often conveyed, at both cellular and systemic levels, by a large number of protein complexes through specific interactions between protein partners. Protein complex assembly can result in the formation of homo-oligomeric or hetero-oligomeric complexes. During steroidogenesis, many proteins, including SR-BI, StAR, HSL, are involved in formation of unique protein complexes for their functions.

(2a) SR-BI: Dimerization and interaction with PDZ domain containing proteins

In response to hormone stimulation, SR-BI, the HDL receptor responsible for selective CE uptake, changes to oligomeric form to facilitate CE uptake (for simplicity, we use the term dimerization here to include the multiple forms of the SR-BI protein; i.e., dimers and higher order oligomers). In one of the earliest direct demonstrations of protein-protein interactions involving SR-BI, it was shown to exist as homodimers in microvilli-enriched adrenal plasma membranes from 17α -ethinyl estradiol $(17\alpha - E_2)$ primed rats (Azhar et al. 2002). Subsequent studies demonstrated that SR-BI exists in dimeric and high order oligomeric forms in all steroidogenic and non-steroidogenic cells and tissue which are active in 'selective' uptake of HDL-CEs (Reaven et al. 2001, 2006). Direct functional evidence for SR-BI dimerization came from the observation that SR-BI exists primarily in a monomeric form with some dimer formation in normal rat adrenal tissue. ACTH stimulation increased the dimerization of SR-BI in this tissue along with increased selective CE uptake, and dexamethasone-induced loss of ACTH led dramatically to the loss of SR-BI, SR-BI dimers and selective HDL-CE uptake. These results, coupled with striking architectural changes of the microvillar compartment at the adrenocortical cell surface, suggest that SR-BI dimers may, in a very basic way, be associated with SR-BI sites of action and function.

Additional functional studies further revealed a strong correlation between the levels of SR-BI dimers and increased selective HDL-CE uptake in cells and tissues. Moreover, co-immunoprecipitation studies of epitope-tagged SR-BIs (SR-BI-cMyc and SR-BI-V5) confirmed that SR-BI can exist as homodimers (Reaven et al. 2006). The use of cross-linking agents provided additional evidence that SR-BI forms dimers in native steroidogenic cell lines (endogenous), as well as in a heterologous insect cell expression system (Reaven et al. 2001). Also, analysis of cellular extracts from SR-BI transfected HEK-293 cells or ACTH-treated Y1-BS1 cells by size-exclusion chromatography and sucrose density centrifugation demonstrated that a significant portion of SR-BI exists in dimeric and oligomeric forms.

Morphological analysis by immunoelectron microscopy provided independent confirmation of SR-BI homodimerization. More specifically, when double tagged-SR-BI proteins (SR-BI-cMyc and SR-BI-V5) were co-expressed in HEK-293 cells and the different proteins were subsequently immunostained and identified with two differently stained gold particles, there was mixing and clustering of gold particles suggesting (1) that the proteins travel to the same cell location, and (2) that many of the gold particles are in exceedingly close physical contact, i.e., within the distance accepted for protein dimers by fluorescent resonance energy transfer (FRET) technique. Similar results were obtained when Y1-BS1 mouse adrenocortical cells were transfected with V5 and/or cMyc tagged-SR-BI proteins. Interestingly, SR-BI transfected Y1-BS1 demonstrated major architectural changes along with the formation of double membranes in flower like arrangements. Gold-labeled secondary antibodies against V5 or cMyc antibody localized SR-BI to these sites, and revealed substantial dimer formation of this protein–shown by close contact between gold particles (Reaven et al. 2004, 2006).

Further investigations concentrated on the contribution of the cysteine residues in the extracellular domain (ECD) of SR-BI either independently or in cooperation with the *C*-terminal domain on SR-BI dimerization. SR-BI contains a total of eight cysteine (C) residues (C21, C251, C280, C321, C323, C334, C384, and C470) and six of them are located in the ECD. Mutagenesis studies showed that C280, C321, C323 and C334 residues in the ECD are necessary for preserving normal SR-BI (HDL) binding activity, selective CE uptake, and/or cell surface expression. Interestingly, mutation of any of these four cysteine residues to serine resulted in a robust induction of SR-BI dimer formation, but they were rendered non-functional because these residues are most likely essential for optimal HDL binding and, hence, selective CE uptake (Hu et al. 2011).

(2b) SR-BI interaction with accessory proteins

Emerging evidence now indicates that accessory proteins are also required for the proper cellular expression of SR-BI and SR-BI-mediated HDL-CE transport and other functions (Ikemoto et al. 2000; Silver 2002; Kocher et al. 2003; Komori et al. 2008; Robichaud et al. 2008; Yuhanna et al. 2008; Zhu et al. 2008; Fenske et al. 2009; Eyre et al. 2010; Kocher and Krieger 2009). For example, it has been shown that PDZK1/NHERF3 regulates hepatic SR-BI stability and steady state protein levels (Hu et al. 2013a). Interestingly, PDZK1/NHERF3 is neither expressed nor essential for SR-BI abundance or its cellular localization in steroidogenic cells of the adrenal gland, ovary and testis (Kocher et al. 2003). Recently, our laboratory made an important discovery that two other NHERF family members, NHERF1 and NHERF2, negatively regulate the expression and function of SR-BI in steroidogenic cells of adrenal and gonads, as well as the liver (Hu et al. 2013a). More specifically, we showed that NHERF1 and NHERF2 mRNA levels decrease in response to cAMP stimulation, whereas mRNA levels of SR-BI are upregulated. Co-immunoprecipitation, colocalization, bimolecular fluorescence complementation, and mutational analysis all indicated that NHERF1 and NHERF2 form complexes with SR-BI protein and, as a result, inhibit SR-BI-mediated selective CE transport and steroidogenesis. The PDZ1 or PDZ2 domain of NHERF1, the PDZ2 domain of NHERF2, or the MERM domains of NHERF1/2 and an intact COOH-terminal PDZ recognition motif (EAKL) in SR-BI is needed for the interaction. Both NHERF1 and NHERF2 also inhibit the de novo synthesis of SR-BI (Hu et al. 2013a). In contrast, no effect of NHERF4 was noted on selective HDL-CE uptake or steroidogenesis. Collectively, these data establish NHERF1 and NHERF2 as SR-BI protein binding partners that play a negative role in the regulation of SR-BI expression, selective CE transport, and steroidogenesis (Hu et al. 2013a).

(2c) StAR-TSPO interaction

StAR facilitates the rate limiting step in steroidogenesis, i.e., delivery of substrate cholesterol from the outer to the inner mitochondrial membrane where P450scc resides for the production of pregnenolone. Although the exact mechanism of action of StAR protein in mediating cholesterol transfer across the mitochondrial membranes is not known, limited evidence has previously suggested that StAR might work in concert with several other proteins including peripheral benzodiazepine receptor (PBR)/18-kDa transporter protein (TSPO), voltage-dependent anion channel 1 (VDAC1), phosphate carrier protein, cAMP-dependent protein kinase 1 α (PKA-RI α) and TSPO-associated acyl-coenzyme A binding domain containing 3 (ABCD3) protein by forming a protein complex on the OMM (Stocco 2001; Rone et al. 2009; Miller 2007; Bose et al. 2008). However, two recent reports have provided evidence that PBR/TSPO is not involved in the steroidogenic process (Tu et al. 2014; Morohaku et al. 2014).

(2d) StAR-HSL interaction and HSL oligomerization

A significant amount of cholesterol transferred by StAR to mitochondria for steroidogenesis is hydrolyzed through HSL from CEs stored in steroidogenic cells. Using in vitro glutathione S-transferase pull-down experiments, we have demonstrated a direct interaction between HSL and StAR via the N-terminal as well as a central region of StAR (Shen et al. 2003). In addition, the 37-kDa StAR was co-immunoprecipitated with HSL from adrenals of animals treated with ACTH. Co-expression of HSL and StAR in Chinese hamster ovary cells resulted in higher cholesteryl ester hydrolytic activity of HSL. Transient overexpression of HSL in Y1 adrenocortical cells increased mitochondrial cholesterol content under conditions in which StAR was induced. It is proposed that the interaction of HSL with StAR in cytosol increases the hydrolytic activity of HSL and that together HSL and StAR facilitate cholesterol movement from lipid droplets to mitochondria for steroidogenesis.

In a separate study, using sucrose gradient centrifugation and in vivo and in vitro protein–protein interactions, we have demonstrated that HSL exists as a functional dimer composed of homologous subunits. Dimeric HSL displayed approximately 40-fold greater activity against cholesteryl ester substrate when compared with monomeric HSL without any differences in affinity for the substrate. Truncations of

HSL identified the importance of the N-terminal 300 amino acids, as well as other regions, in participating in the oligomerization of HSL (Shen et al. 2000).

(2c) StAR-14-3-3y interactions

Recently, the adaptor/scaffold protein 14-3-3 γ was shown to interact with StAR by binding to Ser¹⁹⁴ of StAR and thus retain StAR in the cytosol and delay maximum steroidogenesis in a pattern opposite to 14-3-3 γ homodimerization (Aghazadeh et al. 2012, 2014; Liu et al. 2006; Papadopoulos et al. 2007). Under basal conditions, StAR resides in the cytosol and 14-3-3 γ exists as homodimers; upon hormone stimulation, phosphorylation and acetylation of 14-3-3 γ on residue Ser⁵⁸ and Lys⁴⁹ increase its binding with StAR, which delays maximum steroidogenesis. After 2 h, this interaction falls apart, allowing induction of StAR activity and steroid production to proceed at a maximal rate.

1.3.3 Modulation of Steroidogenesis by miRNAs

Recently, we and others have shown that SR-BI is post-transcriptionally regulated by microRNAs (Hu et al. 2012, 2013b). In addition, miRNAs have been identified that regulate LH/hCG receptor and steroidogenic enzymes in a tissue specific manner (Menon et al. 2013).

MicroRNAs (miRNAs) comprise a novel class of endogenous non-proteincoding single-stranded small RNAs approximately 22-25 nucleotides long that have emerged as key post-transcriptional regulators of gene expression (Ambros 2004; Bartel 2004, 2009; Kim 2005; Fabian et al. 2010). They are transcribed in the nucleus by RNA polymerase II or RNA polymerase III into primary transcripts (pri-miRNAs) (Schwarz et al. 2003; Du and Zamore 2005, 2007), which are generally capped, polyadenylated and contain a hairpin stem of 33 bp, a terminal loop and two single-stranded flanking regions (Cai et al. 2004). These pri-miRNAs are then processed sequentially in the nucleus and cytoplasm by a complex of RNase III-endonucleases Drosha and Dicer to generate pre-miRNAs, and mature miRNAs, respectively (Siomi and Siomi 2010; Finnegan and Pasquinelli 2013). miRNAs cause post-transcriptional repression of protein synthesis by pairing with partially complementary seed sites in the 3'-untranslated regions (UTRs) of target mRNAs, leading to either deadenylation and subsequent mRNA degradation and/or translational inhibition (Bartel 2004, 2009; Fabian et al. 2010; Bushati and Cohen 2007; Eulalio et al. 2008; Filipowicz et al. 2008; Ghildiyal and Zamore 2009). Importantly, a single miRNA can regulate expression of hundreds of target genes (Krishnan et al. 2013; Venkataraman et al. 2013), whereas the expression of a single gene can be regulated by multiple miRNAs (Hu et al. 2012; Gillen et al. 2011). Since the their original discovery in 1993, miRNAs has been shown to be key posttranscriptional regulators of gene expression in metazoan animals, plants, protozoa and viruses (Bushati and Cohen 2007). In mammals miRNAs are suggested to control the activity of more than 60 % of all protein coding genes (Friedman et al. 2009), and to participate in diverse cellular processes including development, cellcycle control, metabolism, stem-cell differentiation, inflammation and immunity, oncogenesis, and diseases (Bushati and Cohen 2007; Barter 2009; Ambros 2004;

Fabian et al. 2010; O'Connell et al. 2010; Yi and Fuchs 2011; Sayed and Abdellatif 2011; O'Connell et al. 2012; Abe and Bonini 2013; Di Leva et al. 2013; Fernández-Hernando et al. 2013; Flowers et al. 2013; Hata 2013; Szabo and Bala 2013; Rottiers and Naar 2012). Accumulating evidence now suggests that miRNAs also participate in the regulation of steroidogenesis (Menon et al. 2013; Yao et al. 2010; Romero et al. 2008; Robertson et al. 2013; Schmitz et al. 2011; Velazquez-Fernandez et al. 2014; Sirotkin et al. 2009; Yin et al. 2012; Yin et al. 2014; Dai et al. 2013; Xu et al. 2011; Kitahara et al. 2013).

Here we review the expression of miRNA in different steroidogenic tissues and their potential involvement in the regulation of steroid biosynthesis (Table 12.1).

miRNA	Tissue	Process(es)/target(s) Ref.				
(1) Adrenal						
miRNA-132, miRNA-212	Adrenal	Aldosterone secretion	Hu et al. (2013b)			
miRNA-21	Adrenal	Aldosterone secretion	Romero et al. (2008)			
	Adrenocortical cells					
miR-24	Adrenal	Cortisol and aldosterone production	Robertson et al. (2013)			
	Adrenal cortical cells	CYP11B1 and CYP11B2				
Human studies with	pathophysiological impl	ications				
miR-675, miR-139-3p	Adrenal	Discriminating ACCs from ACAs	Schmitz et al. (2011)			
miR-335						
miR-21, miR-10b	Adrenal	Diagnose for ACC development, progression	Chabre et al. (2013), Chen et al. (2013), Wang et al. (2013b)			
miR-1395p, miR-let-7f						
(2) Ovary						
miR-21, miR-23a, miR-145, miR-503						
miR-224, miR-383, miR-378, miR-132, and miR-212	Equine follicle development					
miR-122, miR-136-3p	Ovary	LRBP, LHR	Menon et al. (2013), Kitahara et al. (2013)			
miR-455, miR-125a	Ovary	SR-BI	Hu et al. (2012)			
miR-133b	Ovary, granulosa cells	Foxl2, changes StAR and CYP19A level	Dai et al. (2013)			

Table 12.1 Summary of reported roles of miRNA during steroidogenesis

(continued)

Tissue	Process(es)/target(s)	Ref.				
Ovary	LHCGR	Troppmann et al. (2014)				
Ovary, follicle development						
Ovary, granulosa cells	Activin receptor 1B, Smad 2	Yan et al. (2012), Yang et al. (2013)				
Ovary, granulosa cells	Activin receptor IIA	Zhang et al. (2013)				
Ovary, granulosa cells	TGF-b, Smad4	Yao et al. (2010)				
Ovary, granulosa cells	Targets E2F, SF1, suppress granulosa cell proliferation	Yin et al. (2012, 2014)				
Ovary, granulosa cells	RBMS1, c-Myc					
pathophysiological impl	ications					
Ovary, follicular fluid	Regulates estradiol concentration	Sang et al. (2013)				
	Regulates progesterone concentration					
Follicular fluid	Down-regulated in follicular fluid of polycystic ovary patients					
Follicular fluid	Up-regulated in POC					
(3) Testis						
Testicular Leydig cells, MLTC-1		Hu et al. (2013b)				
Mouse testis	In response to oxidative stress	Fatemi et al. (2014)				
Human studies with pathophysiological implications						
Human serum	Plasma biomarker for TGCT	Syring et al. (2014)				
	Tissue Ovary Ovary, follicle development Ovary, granulosa cells Ovary, granulosa cells Ovary, granulosa cells Ovary, granulosa cells Ovary, granulosa cells pathophysiological impl Ovary, follicular fluid Follicular fluid Follicular fluid Testicular Leydig cells, MLTC-1 Mouse testis pathophysiological impl Human serum	TissueProcess(es)/target(s)OvaryLHCGROvary, follicle development-Ovary, granulosa cellsActivin receptor 1B, Smad 2Ovary, granulosa cellsActivin receptor IIAOvary, granulosa cellsTGF-b, Smad4Ovary, granulosa cellsTargets E2F, SF1, suppress granulosa cell proliferationOvary, granulosa cellsRBMS1, c-Mycpathophysiological impitotionsOvary, follicular fluid concentrationOvary, follicular fluidRegulates estradiol concentrationFollicular fluidDown-regulated in follicular fluid of polycystic ovary patientsFollicular fluidUp-regulated in POCFollicular fluidIn response to oxidative stresspathophysiological impitotionsIn response to oxidative stressFollicular fluidIn response to oxidative stressPathophysiological impitotionsIn response to oxidative stress				

Table 12.1 (continued)

2 MiRNA Regulation of LH/hCG and SR-BI Receptors

Mevalonate kinase (Mvk) has been identified as a novel luteinizing hormone receptor (LHR) mRNA binding protein (LRBP), which binds to the coding region of LHR mRNA and causes translational suppression, leading to accelerated degradation of ovarian LHR mRNA in response to treatment with trophic hormones (Menon et al. 2013; Nair et al. 2002). Recently, it was demonstrated that miR-136-3p

and miR-122 contribute to LH/hCG-induced down-regulation of ovarian LHR mRNA (Menon et al. 2013; Kitahara et al. 2013). In addition, miR-513a-3p was shown to target the LH/hCG receptor and control the level of LHCGR expression by an inversely regulated mechanism at the post-transcriptional level (Troppmann et al. 2014).

In a search for miRNAs regulating the expression of SR-BI, we recently provided evidence that miRNAs, such as miR-125a and miRA-455, target SR-BI mRNA, thus post-transcriptionally and negatively regulate SR-BI-mediated selective delivery of HDL-cholesterol in steroidogenic cells and consequently inhibit SR-BI-mediated and HDL-supported steroidogenesis (Hu et al. 2012). In addition, miR-185, miR-96, and miR-223 have been implicated in the negative regulation of hepatic SR-BI (Wang et al. 2013a).

3 miRNA Expression in Steroidogenic Cells of Adrenal Gland, Ovary and Testis

(3a) Adrenal

In an effort to examine the role of miRNAs in steroidogenesis, the expression profiling of miRNAs in rat adrenals in response to in vivo treatment of animals with hormones, ACTH, 17α -ethinyl estradiol (17α -E2) or dexamethasone (Hu et al. 2013b), was examined. Forty five out of the 72 expressed rat miRNAs showed significant changes in the adrenal in response to ACTH treatment. Among these, 27 mature or precursor miRNAs were up-regulated and 18 mature or precursor miR-NAs were down-regulated in ACTH-exposed versus control adrenals. Among the various adrenal miRNAs whose expression was altered in response to treatment of rats with any of these three hormones, the most robust effect was on the expression of miR-132 and miR-212 (3-4 fold induction) in response to ACTH treatment (Hu et al. 2013b). The precursor for miRNA-212 was also up-regulated. Real-time PCR (qRT-PCR) confirmed that miRNA-212, miRNA-183, miRNA-182, miRNA-132 and miRNA-96 were up-regulated by ACTH, and miRNA-466b, miRNA-214, miRNA-503 and miRNA-27a were down-regulated by ACTH. MiR-132, together with miR-212, comprises the evolutionary conserved miR-132/212 family, encoded from the same intron of a small non-coding gene that is located on chromosome 11 in mice, chromosome 10 in rats and chromosome 17 in humans. Mature miRNA-132 and miRNA-212 share the same seed sequence (Wanet et al. 2012; Remenyi et al. 2013). In cardiac fibroblasts, miRNA-132/212 can modulate the angiotensin II signaling pathway (Eskildsen et al. 2014). In adrenal, angiotensin II is an important modulator of adrenal zona glomerulosa cell function, including aldosterone production and cell proliferation. The hormone-mediated increases in miR-132 and miR-212 levels suggest the possibility that they might mediate some of the trophic hormone regulation of steroidogenesis. Likewise, the level of miR-21 has also been shown to be specifically up-regulated by angiotensin II, and its expression levels were correlated with increased aldosterone secretion and proliferation in adrenocortical cells (Romero et al. 2008). In a screen using non-diseased human adrenal and aldosterone-producing adenoma samples, miR-24 was shown to be differentially expressed. Further analysis showed that miR-24 was able to modulate CYP11B1 and CYP11B2 expression, as well as cortisol and aldosterone production in human adrenal cortex (Robertson et al. 2013).

Apart from their involvement in a variety of biological processes, microRNAs are gaining recognition as potential pathogenic biomarkers for a number of metabolic and genetic diseases (Iorio and Croce 2012; Maegdefessel 2014). To study the clinical implications of miRNA expression in human adrenocortical carcinomas, an analysis of normal adrenal tissue, adrenocortical adenomas (ACAs), adrenocortical carcinomas (ACCs) and metastases showed that ACCs exhibited significantly lower levels of miR-139-3p, miR-675 and miR-335, and miRNA expression, suggesting that profiling miR-675 and miR-335 helps in discriminating ACCs from ACAs (Schmitz et al. 2011). In a different study, more than 40 miRNAs were shown to be differentially expressed in hyperfunctioning adrenocortical adenomas (Velazquez-Fernandez et al. 2014). Mir-21, mir-10b, mir-139-5p and mir-LET-7f have been shown to be involved in ACC development, progression, and aggressiveness (Chabre et al. 2013; Chen et al. 2013; Wang et al. 2013b).

(3b) Ovary

A study designed to identify miRNAs affecting the release of the major sex steroids progesterone, androgen and estrogen by human ovarian cells showed that 36 out of 80 tested miRNA constructs resulted in inhibition of progesterone release in granulosa cells, and ten miRNAs promoted progesterone release. Fifty-seven miRNAs tested inhibited testosterone release, and only one miRNA enhanced testosterone output. Fifty-one miRNAs suppressed estradiol release, while none of the miRNAs tested stimulated it (Sirotkin et al. 2009).

Many studies have focused on ovarian granulosa cells and have shown that multiple miRNAs, including miR-21, miR-23a, miR-145, miR-503, miR-224, miR-383, miR-378, miR-132, and miR-212, are involved in follicle proliferation and granulosa cell function (Yin et al. 2012, 2014; Yao et al. 2010; Schauer et al. 2013). miR-224 was shown to target Smad4 and to regulate the TGF- β 1 signaling pathway (Yao et al. 2010). Overexpression of miR-224 enhanced TGF- β 1 induced granulosa proliferation, whereas inhibition of endogenous miR-224 attenuated TGF- β 1 induced proliferation of granulosa cells. Both miR-224 and TGF- β 1 can function to increase CYP19A1 mRNA levels and promote estradiol release from granulosa cells. miR-133b was shown to target the 3' UTR of Fox12 and inhibited the transcriptional repression activity of FOX12 against StAR and CYP19A1, thereby regulating estradiol production in granulosa cells (Dai et al. 2013). The expression levels of both miR-132 and miR-212 were up-regulated in primary rat ovarian granulosa cells by the second messenger of the trophic hormone, cAMP

(Hu et al. 2013b). miR-383 has been shown to have multiple targets in its role in regulating steroidogenesis. On the one hand, miR-383, which is itself trans-activated by SF-1, affects the stability of RBMS1, suppresses the level of c-Myc and regulates estradiol release from granulosa cells (Yin et al. 2012). On the other hand, miR-383 has also been shown to increase the expression of miR-320 and to suppress granulosa cell proliferation. The targets of miR-320 were shown to be E2F1 and SF-1, which result in suppression of follicle development and steroid production (Yin et al. 2014). In a study searching for regulators of estradiol production, micro-RNA378 (miR-378) was shown to be spatiotemporally expressed in porcine granulosa cells, and to regulate ovarian estradiol production by targeting aromatase (Xu et al. 2011).

With respect to clinical relevance, evidence has been presented that the expression levels of miR-320 and miR-383 are up-regulated in the follicular fluid of patients with polycystic ovarian syndrome. Additional analysis of human follicular fluid led to the identification of miR-132, miR-320, miR-520c-3p, miR-24 and miR-222, which have been implicated in the regulation of estradiol levels, whereas miR-24, miR-193b and miR-483-5p have been suggested to regulate progesterone production. Finally, miR-132 and miR-320 are expressed at significantly lower levels in the follicular fluid of patients with polycystic ovarian syndrome (Sang et al. 2013).

(3c) Testis

To date, most of the miRNA studies in testis have been directed towards their potential involvement in the spermatogenic process and other testicular functions rather than steroidogenesis. Many miRNAs have also been identified that specifically target germ-cell specific genes (for more information, please consult some recent reviews McIver et al. 2012; Papaioannou and Nef 2010; Sree et al. 2014; Kotaja 2014).

With respect to testicular Leydig cells, most of the studies have been focused on documenting changes in the expression profile of miRNAs using Leydig tumor cell lines. In a screen profiling the expression of miRNAs, the expression levels of both miR-132 and miR-212 were upregulated in mouse testicular Leydig tumor cells, MLTC-1, by the second messenger cAMP (Hu et al. 2013b). Another study reported changes in the expression of miRNA-34a, miRNA-181b and miRNA-1221 in mouse testis in response to hydroperoxide (TBHP)-induced excessive oxidative stress in vivo (Fatemi et al. 2014). These miRNAs have been implicated in the regulation of various cellular signaling pathways associated with inflammation, antioxidant responses and spermatogenesis.

A recent clinical study demonstrated that miR-367-3p, miR-371a-3p, miR-372-3p, miR-373-3 could potentially serve as biomarkers for the diagnosis of testicular germ cell tumor (TGCT) (Syring et al. 2014). In particular, miR-371a-3p showed a better correlation with TGCT than either AFP or hCG, and could potentially be employed as a biomarker in clinical management of TGCT, especially monitoring surveillance therapy and residual disease after chemotherapy.

4 Conclusions

During the past six decades, the understanding of the complex process of steroidogenesis has progressed rapidly in the steroid genic cells of the adrenal gland, ovary and testis. At the cellular level, this represents a multistep and multienzyme process. In the past two decades or so, with the introduction of state-of-the-art molecular, cellular and biochemical techniques, considerable progress has been made in understanding the various steps involved in the acquisition of cholesterol substrate, both from external and internal sources, for its utilization by mitochondrial CYP11A1 (P450scc) for pregnenolone production, and subsequent enzymatic conversion of pregnenolone into tissue specific steroids. Moreover, events connected with the trophic hormone stimulation of steroidogenesis both at the acute and chronic levels have been reasonably defined. In addition, ample information is currently available about the transcriptional regulation of key steroidogenic enzymes, as well as StAR protein, which plays a pivotal role in the transport of cholesterol to mitochondria for steroid production. During the past few years, it is becoming increasingly clear that the steroidogenic process in general and certain steroidogenic proteins in particular are also subject to post-transcriptional and post-translational regulation. Broadly, these post-transcriptional and post-translational regulations include processes such as phosphorylation/dephosphorylation and protein-protein interactions, as well as the involvement of specific miRNAs. A further mechanistic understanding of these events in the near future should greatly aid in delineating the underlying mechanisms involved in cellular cholesterol trafficking, cholesterol transport to mitochondria and steroid production.

Acknowledgments This work was supported by National Institutes of Health, NHLBI, Grant 2R01HL33881 and by the Office of Research and Development, Medical Service, Department of Veterans Affairs.

The authors are thankful to Mr. Kris Morrow, Medical Media, VA Palo Alto Health Care System for his assistance in the preparation of graphic illustration.

References

- Abe M, Bonini N (2013) MicroRNAs and neurodegeneration: role and impact. Trends Cell Biol 23(1):30–36
- Aghazadeh Y, Rone MB, Blonder J et al (2012) Hormone-induced 14-3-3gamma adaptor protein regulates steroidogenic acute regulatory protein activity and steroid biosynthesis in ma-10 Leydig cells. J Biol Chem 287(19):15380–15394
- Aghazadeh Y, Ye X, Blonder J et al (2014) Protein modifications regulate the role of 14-3-3gamma adaptor protein in camp-induced steroidogenesis in ma-10 Leydig cells. J Biol Chem 289(38):26542–26553
- Ambros V (2004) The functions of animal microRNAs. Nature 431(7006):350-355
- Anthonsen MW, Rönnstrandt L, Wernstedt C et al (1998) Identification of novel phosphorylation sites in hormone-sensitive lipase that are phosphorylated in response to isoproterenol and govern activation properties in vitro. J Biol Chem 273:215–221
- Arakane F, King SR, Du Y et al (1997) Phosphorylation of steroidogenic acute regulatory protein (star) modulates its steroidogenic activity. J Biol Chem 272(51):32656–32662

- Azhar S, Nomoto A, Reaven E (2002) Hormonal regulation of adrenal microvillar channel formation. J Lipid Res 43:861–871
- Babu P, Baves D, Shah S et al (2000) Role of phosphorylation, gene dosage and dax-1in sf-1mediated steroidogenesis. Endocr Res 26:985–994
- Bahler M, Benfenati F, Valtorta F et al (1990) The synapsins and the regulation of synaptic function. Bioessays 12(6):259–263
- Bartel DP (2004) microRNAs: genomics, biogenesis, mechanism, and function. Cell 116(2):281-297
- Bartel DP (2009) MicroRNAs: target recognition and regulatory functions. Cell 136(2):215-233
- Barter D (2009) MicroRNAs: target recognition and regulatory function. Cell 136:215-233
- Bose H, Whittal R, Ran Y et al (2008) Star-like activity and molten globule behavior of stard6, a male germ-line protein. Biochemistry 47(8):2277–2288
- Burnett G, Kennedy E (1954) The enzymatic phosphorylation of proteins. J Biol Chem 211:969–980
- Bushati N, Cohen SM (2007) Microrna functions. Annu Rev Cell Dev Biol 23:175-205
- Cai X, Hagedorn CH, Cullen BR (2004) Human microRNAs are processed from capped, polyadenylated transcripts that can also function as mRNAS. RNA 10(12):1957–1966
- Chabre O, Libe R, Assie G et al (2013) Serum miR-483-5p and miR-195 are predictive of recurrence risk in adrenocortical cancer patients. Endocr Relat Cancer 20(4):579–594
- Chen W, Cai F, Zhang B et al (2013) The level of circulating miRNA-10b and miRNA-373 in detecting lymph node metastasis of breast cancer: potential biomarkers. Tumour Biol 34(1):455–462
- Ciesla J, Fraczyk T, Rode W (2011) Phosphorylation of basic amino acid residues in proteins: important but easily missed. Acta Biochim Pol 58:137–147
- Clark B, Wells J, King S et al (1994) The purification, cloning, and expression of a novel luteinizing hormone-induced mitochondrial protein in ma-10 mouse Leydig tumor cells. Characterization of the steroidogenic acute regulatory protein (star). J Biol Chem 269:28314–28322
- Cohen P (2000) The regulation of protein function by multisite phosphorylation—a 25 year update. Trends Biochem Sci 25(12):596–601
- Dai A, Sun H, Fang T et al (2013) microRNA-133b stimulates ovarian estradiol synthesis by targeting FOX12. FEBS Lett 587(15):2474–2482
- Desclozeaux M, Kylova I, Horn F et al (2002) Phosphorylation and intramolecular stabilization of ligand binding domain in the steroidogenic factor 1. Mol Cell Biol 22:7193–7203
- Di Leva G, Garofalo M, Croce C (2013) MicroRNAs in cancer. Annu Rev Pathol Mech Dis 9:287–314
- Du T, Zamore PD (2005) Microprimer: the biogenesis and function of microRNA. Development 132(21):4645–4652
- Du T, Zamore PD (2007) Beginning to understand microRNA function. Cell Res 17(8):661-663
- Epstein LF, Orme-Johnson NR (1991a) Acute action of luteinizing hormone on mouse Leydig cells: accumulation of mitochondrial phosphoproteins and stimulation of testosterone synthesis. Mol Cell Endocrinol 81(1–3):113–126
- Epstein L, Orme-Johnson N (1991b) Regulation of steroid hormone biosynthesis: identification of precursors of a phosphoprotein targeted to the mitochondrion in stimulated rat adrenal cortex cells. J Biol Chem 266:19739–19745
- Eskildsen TV, Schneider M, Sandberg MB et al (2014) The microRNA-132/212 family fine-tunes multiple targets in angiotensin ii signalling in cardiac fibroblasts. J Renin Angiotensin Aldosterone Syst
- Eulalio A, Huntzinger E, Izaurralde E (2008) Getting to the root of miRNA-mediated gene silencing. Cell 132(1):9–14
- Eyre N, Drummer H, Beard M (2010) The SR-BI partner PDZK1 facilitates hepatitis C virus entry. PLoS Pathog 6:e1001130
- Fabian MR, Sonenberg N, Filipowicz W (2010) Regulation of mRNA translation and stability by microRNAs. Annu Rev Biochem 79:351–379

- Fatemi N, Sanati MH, Shamsara M et al (2014) TBHP-induced oxidative stress alters microRNAs expression in mouse testis. J Assist Reprod Genet
- Fenske S, Yesilaltay A, Pal R et al (2009) Normal hepatic cell surface localization of the high density lipoprotein receptor, scavenger receptor class B, type I, depends on all four PDZ domains of PDZK1. J Biol Chem 284:5797–5806
- Fernández-Hernando C, Ramírez C, Goedeke L et al (2013) MicroRNAs in metabolic disease. Arterioscler Thromb Vasc Biol 33:178–185
- Filipowicz W, Bhattacharyya SN, Sonenberg N (2008) Mechanisms of post-transcriptional regulation by microRNAs: are the answers in sight? Nat Rev Genet 9(2):102–114
- Finnegan EF, Pasquinelli AE (2013) MicroRNA biogenesis: regulating the regulators. Crit Rev Biochem Mol Biol 48(1):51–68
- Fischer EH, Krebs EG (1966) Relationship of structure to function of muscle phosphorylase. Fed Proc 25:1511–1520
- Fleury A, Mathieu AP, Ducharme L et al (2004) Phosphorylation and function of the hamster adrenal steroidogenic acute regulatory protein (star). J Steroid Biochem Mol Biol 91(4–5):259–271
- Flowers E, Froelicher E, Aouizerat B (2013) MicroRNA regulation of lipid metabolism. Metabolism 62(1):12–20
- Friedman RC, Farh KK-H, Burge CB et al (2009) Most mammalian mRNAs are conserved targets of microRNAs. Genome Res 19(1):92–9105
- Ghildiyal M, Zamore PD (2009) Small silencing RNAs: an expanding universe. Nat Rev Genet 10(2):94–108
- Giatti S, Boraso M, Melcangi RC et al (2012) Neuroactive steroids, their metabolites, and neuroinflammation. J Mol Endocrinol 49:R125–R134
- Gillen AE, Gosalia N, Leir SH et al (2011) MicroRNA regulation of expression of the cystic fibrosis transmembrane conductance regulator gene. Biochem J 438(1):25–32
- Greenberg AS, Shen WJ, Muliro K et al (2001) Stimulation of lipolysis and hormone-sensitive lipase via the extracellular signal-regulated kinase pathway. J Biol Chem 276(48):45456–45461
- Hanks SK, Quinn AM, Hunter T (1988) The protein kinase family: conserved features and deduced phylogeny of the catalytic domains. Science 241(4861):42–52
- Hata A (2013) Functions of microRNAs in cardiovascular biology and disease. Annu Rev Physiol 75:69–93
- Hu J, Zhang Z, Shen W et al (2010) Cellular cholesterol delivery, intracellular processing and utilization for biosynthesis of steroid hormones. Nutr Metab (Lond) 7:47
- Hu J, Zhang Z, Shen W et al (2011) Differential roles of cysteine residues in the cellular trafficking, dimerization, and function of the high-density lipoprotein receptor, SR-BI. Biochemistry 50(50):10860–10875
- Hu Z, Shen W, Kraemer F et al (2012) microRNAs 125a and 455 repress lipoprotein-supported steroidogenesis by targeting scavenger receptor class B type I in steroidogenic cells. Mol Cell Biol 32(24):5035–5045
- Hu Z, Hu J, Zhang Z et al (2013a) Regulation of expression and function of scavenger receptor class B, type I (SR-BI) by na+/h+ exchanger regulatory factors (nherfs). J Biol Chem 288(16):11416–11435
- Hu Z, Shen W, Cortez Y et al (2013b) Hormonal regulation of microRNA expression in steroid producing cells of the ovary, testis and adrenal gland. PLoS One 8(10), e78040
- Ikemoto M, Arai H, Feng D et al (2000) Identification of a PDZ-domain-containing protein that interacts with the scavenger receptor class B type I. Proc Natl Acad Sci U S A 97:6538–6543
- Iorio MV, Croce CM (2012) MicroRNA dysregulation in cancer: diagnostics, monitoring and therapeutics. A comprehensive review. EMBO Mol Med 4(3):143–159
- Kim VN (2005) MicroRNA biogenesis: coordinated cropping and dicing. Nat Rev Mol Cell Biol 6(5):376–385
- Kitahara Y, Nakamura K, Koguru K et al (2013) Role of microRNA-136-3p on the expression of luteinizing hormone-human chorionic gonadotropin receptor mRNA in rat ovaries. Biol Reprod 89:1–10

- Kocher O, Krieger M (2009) Role of the adaptor protein PDZK1 in controlling the HDL receptor SR-BI. Curr Opin Lipidol 20(3):236–241
- Kocher O, Yesilaltay A, Cirovic C et al (2003) Targeted disruption of the PDZK1 gene in mice causes tissue-specific depletion of the high density lipoprotein receptor scavenger receptor class B type I and altered lipoprotein metabolism. J Biol Chem 278:52820–52825
- Komori H, Arai H, Kashima T et al (2008) Coexpression of CLA-1 and human PDZK1 in murine liver modulates HDL cholesterol metabolism. Arterioscler Thromb Vasc Biol 28:1298–1303
- Kotaja N (2014) MicroRNAs and spermatogenesis. Fertil Steril 101(6):1552-1562
- Kraemer F, Shen W (2002) Hormone-sensitive lipase: control of intracellular tri-(di)acylglycerol and cholesteryl ester hydrolysis. J Lipid Res 43:1585–1594
- Kraemer F, Shen W, Natu V et al (2002) Adrenal neutral cholesteryl hydrolase: identification, subcellular distribution and sex differences. Endocrinology 143:801–806
- Kraemer FB, Shen WJ, Harada K et al (2004) Hormone-sensitive lipase is required for highdensity lipoprotein cholesteryl ester-supported adrenal steroidogenesis. Mol Endocrinol 18(3):549–557
- Krebs EG, Fischer EH (1964) Phosphorylase and related enzymes of glycogen metabolism. Vitam Horm 22:399–410
- Krishnan K, Steptoe AL, Martin HC et al (2013) Microrna-182-5p targets a network of genes involved in DNA repair. RNA 19(2):230–242
- Krueger R, Orme-Johnson N (1983) Acute adrenocorticotropic hormone stimulation of adrenal corticosteroidogenesis. Discovery of a rapidly induced protein. J Biol Chem 258:10159–10167
- LaVoie H, King S (2009) Transcriptional regulation of steroidogenic genes: Stard1, cyp11a1 and hsd3b. Exp Biol Med (Maywood) 234:880–907
- Liu J, Rone MB, Papadopoulos V (2006) Protein–protein interactions mediate mitochondrial cholesterol transport and steroid biosynthesis. J Biol Chem 281(50):38879–38893
- Maegdefessel L (2014) The emerging role of microRNAs in cardiovascular disease. J Intern Med
- Manna P, Dyson M, Stocco D (2009) Regulation of the steroidogenic acute regulatory protein gene expression: present and future perspective. Mol Hum Reprod 15:321–333
- Marsh JM (1976) The role of cyclic AMP in gonadal steroidogenesis. Biol Reprod 14:30-53
- McIver SC, Roman SD, Nixon B et al (2012) MiRNA and mammalian male germ cells. Hum Reprod Update 18(1):44–59
- McKenna T, Fearon U, Clarke D et al (1997) A critical review of the origin and control of adrenal androgens. Baillieres Clin Obstet Gynecol 11:229–248
- Mellon SH, Griffin D (2002) Neurosteroids: biochemistry and clinical significance. Trends Endocrinol Metab 13:35–43
- Menon B, Sinden J, Franzo-Romain M et al (2013) Regulation of LH receptor mRNA binding protein by miR-122 in rat ovaries. Endocrinology 154:4826–4834
- Miller WL (1988) Molecular biology of steroid hormone synthesis. Endocr Rev 9(3):295-318
- Miller W (2002) Androgen biosynthesis from cholesterol to dhea. Mol Cell Endocrinol 198:7-14
- Miller WL (2007) Star search—what we know about how the steroidogenic acute regulatory protein mediates mitochondrial cholesterol import. Mol Endocrinol 21(3):589–601

Miller W (2008) Steroidogenic enzymes. Endocr Dev 13:1-18

- Miller WL, Bose HS (2011) Early steps in steroidogenesis: intracellular cholesterol trafficking thematic review series: genetics of human lipid diseases. J Lipid Res 52(12):2111–2135
- Morohaku K, Pelton SH, Daugherty DJ et al (2014) Translocator protein/peripheral benzodiazepine receptor is not required for steroid hormone biosynthesis. Endocrinology 155(1):89–97
- Nair A, Kash J, Peegel H et al (2002) Post-translational regulation of luteinizing hormone receptor mRNA in the ovary by a novel mRNA-binding protein. J Biol Chem 277:21468–21473
- O'Connell R, Rao D, Chaudhuri A et al (2010) Physiological and pathological roles for microR-NAs in the immature system. Nat Rev Immunol 10:111–122
- O'Connell R, Rao D, Baltimore D (2012) MicroRNA regulation of inflammatory responses. Annu Rev Immunol 30:295–312
- Papadopoulos V, Liu J, Culty M (2007) Is there a mitochondrial signaling complex facilitating cholesterol import? Mol Cell Endocrinol 265–266:59–64

- Papaioannou MD, Nef S (2010) MicroRNAs in the testis: building up male fertility. J Androl 31(1):26–33
- Payne A, Hales D (2004) Overview of steroidogenic enzymes in the pathway from cholesterol to active steroid hormones. Endocr Rev 25:947–970
- Payne AH, Youngblood GL, Sha L et al (1992) Hormonal regulation of steroidogenic enzyme gene expression in Leydig cells. J Steroid Biochem Mol Biol 43(8):895–906
- Pon LA, Orme-Johnson NR (1988) Acute stimulation of corpus luteum cells by gonadotrophin or adenosine 3',5'-monophosphate causes accumulation of a phosphoprotein concurrent with acceleration of steroid synthesis. Endocrinology 123(4):1942–1948
- Pon L, Orme-Johnson N (1998) Acute stimulation of corpus luteum cells by gonadotropin or adenosine 3',5'-monophosphate causes accumulation of a phosphoprotein concurrent with acceleration of steroid synthesis. Endocrinology 123:1942–1948
- Pon LA, Hartigan JA, Orme-Johnson NR (1986) Acute acth regulation of adrenal corticosteroid biosynthesis. Rapid accumulation of a phosphoprotein. J Biol Chem 261(28):13309–13316
- Reaven E, Leers-Sucheta S, Nomoto A et al (2001) Expression of scavenger receptor class B type 1 (SR-BI) promotes microvillar channel formation and selective cholesteryl ester transport in a heterologous reconstituted system. Proc Natl Acad Sci U S A 98:1613
- Reaven E, Cortez Y, Leers-Sucheta S et al (2004) Dimerization of the scavenger receptor class B type I: formation, function, and localization in diverse cells and tissues. J Lipid Res 45(3):513–528
- Reaven E, Nomoto A, Cortez Y et al (2006) Consequences of over-expression of rat scavenger receptor, SR-BI, in an adrenal cell model. Nutr Metab (Lond) 3:43
- Reddy DS (2010) Neurosteroids: endogenous role in the human brain and therapeutic potential. Prog Brain Res 186:113–137
- Remenyi J, van den Bosch MWM, Palygin O et al (2013) miR-132/212 knockout mice reveal roles for these miRNAs in regulating cortical synaptic transmission and plasticity. PLoS One 8, e62509
- Robertson S, MacKenzie S, Alvarez-Madrazo S et al (2013) Microrna-24 is a novel regulator of aldosterone and cortisol production in the human adrenal cortex. Hypertension 62(3):572–578
- Robichaud J, Francis G, Vance D (2008) A role for hepatic scavenger receptor class B, type I in decreasing high density lipoprotein levels in mice that lack phosphatidylethanolamine n-methyltransferase. J Biol Chem 283:35496–35506
- Romero DG, Plonczynski MW, Carvajal CA et al (2008) Microribonucleic acid-21 increases aldosterone secretion and proliferation in h295r human adrenocortical cells. Endocrinology 149(5):2477–2483
- Rone MB, Fan J, Papadopoulos V (2009) Cholesterol transport in steroid biosynthesis: role of protein–protein interactions and implications in disease states. Biochim Biophys Acta 1791(7):646–658
- Rottiers V, Naar AM (2012) MicroRNAs in metabolism and metabolic disorders. Nat Rev Mol Cell Biol 13(4):239–250
- Sands W, Palmer T (2008) Regulation of gene transcription in response to camp elevation. Cell Signal 20:460–466
- Sang Q, Yao Z, Wang H et al (2013) Identification of microRNAs in human follicular fluid: characterization of microRNAs that govern steroidogenesis in vitro and are associated with polycystic ovary syndrome in vivo. J Clin Endocrinol Metab 98(7):3068–3079
- Sanorn B, Heindel J, Robinson G (1980) The role of cyclic nucleotides in the reproductive processes. Ann Rev Physiol 42:37–57
- Sasaki G, Ishii T, Jeyasuria P et al (2008) Complex role of the mitochondrial targeting signal in the function of steroidogenic acute regulatory protein revealed by bacterial artificial chromosome transgenesis in vivo. Mol Endocrinol 22(4):951–964
- Sasaki G, Zubair M, Ishii T et al (2014) The contribution of serine 194 phosphorylation to steroidogenic acute regulatory protein function. Mol Endocrinol 28(7):1088–1096
- Sayed D, Abdellatif M (2011) MicroRNAs in development and disease. Physiol Rev 91(3):827-887
- Schauer SN, Sontakke SD, Watson ED et al (2013) Involvement of miRNAs in equine follicle development. Reproduction 146(3):273–282

- Schmitz KJ, Helwig J, Bertram S et al (2011) Differential expression of microRNA-675, microRNA-139-3p and microRNA-335 in benign and malignant adrenocortical tumours. J Clin Pathol 64(6):529–535
- Schwarz DS, Hutvagner G, Du T et al (2003) Asymmetry in the assembly of the RNAi enzyme complex. Cell 115(2):199–208
- Shen WJ, Patel S, Natu V et al (1998) Mutational analysis of structural features of rat hormonesensitive lipase. Biochemistry 37(25):8973–8979
- Shen WJ, Patel S, Hong R et al (2000) Hormone-sensitive lipase functions as an oligomer. Biochemistry 39(9):2392–2398
- Shen WJ, Patel S, Natu V et al (2003) Interaction of hormone-sensitive lipase with steroidogenic acute regulatory protein: facilitation of cholesterol transfer in adrenal. J Biol Chem 278(44): 43870–43876
- Silver D (2002) A carboxyl-terminal PDZ-interacting domain of scavenger receptor B, type I is essential for cell surface expression in liver. J Biol Chem 277:34042–34047
- Simpson ER, Waterman MR (1988) Regulation of the synthesis of steroidogenic enzymes in adrenal cortical cells by acth. Annu Rev Physiol 50:427–440
- Simpson E, Waternan M (1983) Regulation by acth of steroid hormone biosynthesis in the adrenal cortex. Can J Biochem Cell Biol 61:692–707
- Simpson E, Lauber M, Demeter M et al (1992) Regulation of expression of the genes encoding steroidogenic enzymes in the ovary. J Steroid Biochem Mol Biol 41(3–8):409–413
- Siomi H, Siomi MC (2010) Posttranscriptional regulation of microRNA biogenesis in animals. Mol Cell 38(3):323–332
- Sirotkin AV, Ovcharenko D, Grossmann R et al (2009) Identification of microRNAs controlling human ovarian cell steroidogenesis via a genome-scale screen. J Cell Physiol 219(2):415–420
- Spat A, Hunyady L (2004) Control of aldosterone secretion: a model for convergence in cellular signaling pathways. Physiol Rev 84:489–539
- Sree S, Radhakrishnan K, Indu S et al (2014) Dramatic changes in 67 microRNAs during initiation of first wave of spermatogenesis in mus musculus testis: global regulatory insights generated by microRNA-mrna network analysis. Biol Reprod
- Stocco DM (2001) Star protein and the regulation of steroid hormone biosynthesis. Annu Rev Physiol 63(1):193–213
- Stocco DM, Clark BJ (1996) Regulation of the acute production of steroids in steroidogenic cells. Endocr Rev 17(3):221–244
- Strauss J, Golos T, Silavin S et al (1988) Involvement of cyclic amp in the functions of granulosa and luteal cells: regulation of steroidogenesis. Prog Clin Biol Res 267:177–200
- Strauss JF III, Martinezm F, Kiriakidu M (1996) Placental steroid hormone synthesis: unique features and unanswered questions. Biol Reprod 54:303–311
- Syring I, Bartels J, Holdenrieder S et al (2014) Circulating serum microRNA (miR-367-3p, miR-371a-3p, miR-372-3p, miR-373-3p) as biomarkers for patients with testicular germ cell cancers. J Urol
- Szabo G, Bala S (2013) MicroRNAs in liver disease. Nat Rev Gastroenterol Hepatol 10(9):542-552
- Takemori H, Okamoto M (2008) Regulation of CREB-mediated gene expression by salt inducible kinase. J Steroid Biochem Mol Biol 108:287–291
- Troppmann B, Kossack N, Nordhoff V et al (2014) MicroRNA miR-513a-3p acts as a co-regulator of luteinizing hormone/chorionic gonadotropin receptor gene expression in human granulosa cells. Mol Cell Endocrinol 390(1–2):65–72
- Tu LN, Morohaku K, Manna PR et al (2014) Peripheral benzodiazepine receptor/translocator protein global knockout mice are viable with no effects on steroid hormone biosynthesis. J Biol Chem
- Velazquez-Fernandez D, Caramuta S, Ozata DM et al (2014) Microrna expression patterns associated with hyperfunctioning and non-hyperfunctioning phenotypes in adrenocortical adenomas. Eur J Endocrinol 170(4):583–591
- Venkataraman S, Birks DK, Balakrishnan I et al (2013) MicroRNA 218 acts as a tumor suppressor by targeting multiple cancer phenotype-associated genes in medulloblastoma. J Biol Chem 288(3):1918–1928

- Vilgrain I, Defraye G, Ghambaz E (1984) Adrenocortical cytochrome p-450 responsible for cholesterol side chain cleavage (p-450scc) is phosphorylated by the calcium activated, phospholipidsensitive protein kinase (protein kinase c). Biochem Biophys Res Commun 125:554–561
- Wanet A, Tacheny A, Arnould T et al (2012) miR-212/132 expression and functions: within and beyond the neuronal compartment. Nucleic Acids Res 40(11):4742–4753
- Wang L, Jia X-J, Jiang H-J et al (2013a) MicroRNAs 185, 96, and 223 repress selective highdensity lipoprotein cholesterol uptake through posttranscriptional inhibition. Mol Cell Biol 33:1956–1964
- Wang YY, Ye ZY, Zhao ZS et al (2013b) Clinicopathologic significance of miR-10b expression in gastric carcinoma. Hum Pathol 44(7):1278–1285
- Wood S, Emmison N, Borthwick A et al (1993) The protein phosphatases responsible for dephosphorylation of hormone-sensitive lipase in isolated rat adipocytes. Biochem J 295:531–535
- Xu S, Linher-Melville K, Yang B et al (2011) Micro-RNA378 (miR-378) regulates ovarian estradiol production by targeting aromatase. Endocrinology 152(10):3941–3951
- Yan G, Zhang L, Fang T et al (2012) MicroRNA-145 suppresses mouse granulosa cell proliferation by targeting activin receptor iB. FEBS Lett 586(19):3263–3270
- Yang S, Wang S, Luo A et al (2013) Expression patterns and regulatory functions of microRNAs during the initiation of primordial follicle development in the neonatal mouse ovary. Biol Reprod 89(5):126
- Yao G, Yin M, Lian J et al (2010) Microrna-224 is involved in transforming growth factorbeta-mediated mouse granulosa cell proliferation and granulosa cell function by targeting Smad4. Mol Endocrinol 24(3):540–551
- Yeaman S (1990) Hormone-sensitive lipase–a multipurpose enzyme in lipid metabolism. Biochim Biophys Acta 1052:128–132
- Yi R, Fuchs E (2011) MicroRNAs and their roles in mammalian stem cells. J Cell Sci 124(Pt 11):1775–1783
- Yin M, Lu M, Yao G et al (2012) Transactivation of microRNA-383 by steroidogenic factor-1 promotes estradiol release from mouse ovarian granulosa cells by targeting RBMS1. Mol Endocrinol 26(7):1129–1143
- Yin M, Wang X, Yao G et al (2014) Transactivation of microRNA-320 by microRNA-383 regulates granulosa cell functions by targeting E2F1 and SF-1 proteins. J Biol Chem 289(26):18239–18257
- Yuhanna I, Shaul P, Mineo C (2008) The scavenger receptor class B type I adaptor protein PDZK1 maintains endothelial monolayer integrity. Circ Res 102(480–487)
- Zhang Q, Sun H, Jiang Y et al (2013) MicroRNA-181a suppresses mouse granulosa cell proliferation by targeting activin receptor IIA. PLoS One 8(3), e59667
- Zhu W, Saddar S, Seetharam D et al (2008) The scavenger receptor class B type I adaptor protein PDZK1 maintains endothelial monolayer integrity. Circ Res 102:480–487

Post-transcriptional Regulation of Glucocorticoid Function

13

Faoud T. Ishmael and Cristiana Stellato

1 Post-transcriptional Gene Regulation in Inflammation as Target of GC Anti-inflammatory Action

1.1 Mechanisms of PTR in Inflammation: Rationale for Therapeutic Anti-inflammatory Targeting

Post-transcriptional gene regulation (PTR) is a critical control mechanism of the inflammatory response integrated with transcriptional control of gene expression. By implementing changes in mRNA turnover and translation rates, PTR mechanisms adapt the amplitude and timing of protein expression to endogenous or environmental changes (Stoecklin and Anderson 2006a; Hollams et al. 2002). Transcriptional and post-transcriptional regulation are coordinately mediated by common signaling pathways—chiefly MAP kinases (Winzen et al. 1999; Frevel et al. 2003; Gaestel 2006; Gao et al. 2013). Rapid mRNA transcript degradation of cytokines, chemokines, enzymes and other mediators contribute to successful

F.T. Ishmael

C. Stellato, MD, PhD (⊠) Department of Medicine and Surgery, University of Salerno, Via Salvador Allende, Baronissi, SA 84081, Italy

Department of Medicine, Division of Pulmonary, Allergy, and Critical Care Medicine, Section of Allergy and Immunology, The Pennsylvania State University Milton S. Hershey Medical Center, 500 University Drive, Hershey, PA 17033, USA

Department of Medicine, Division of Allergy and Clinical Immunology, Johns Hopkins University, Baltimore, MD, USA e-mail: cstellato@unisa.it

cessation of an acute inflammatory reaction; conversely, aberrant stabilization of the targeted transcripts can support overexpression of inflammatory genes during chronic inflammation. Stress- and inflammation-driven signals can also rapidly adapt protein translation rates to changing extracellular environment (Stoecklin and Anderson 2006a; Hollams et al. 2002). Genome-wide studies indicate that up to 50 % of genes induced during a stress response are mainly regulated posttranscriptionally (Fan et al. 2002). RNA-binding proteins (RBP), microRNA (miRNA) and other classes of small noncoding RNAs (sncRNA) constitute the heterogeneous group of regulatory factors conveying PTR through binding to conserved sequences mainly present in the untranslated regions (UTR) of their mRNA targets, with which they form ribonucleoprotein (RNP) complexes ultimately conveying PTR action. The activation of regulatory factors and their assembly in multimeric RNP complexes is a dynamic process susceptible of regulation by inflammatory signaling (Winzen et al. 1999; Frevel et al. 2003; Gaestel 2006; Gao et al. 2013) and therefore, potentially amenable to therapeutic targeting. Small variations in mRNA half-life, in the range of two- to fourfold changes, can rapidly lead to over a 1000-fold difference in mRNA levels (Ross 1995). With additional regulation at the level of protein translation, the rate-limiting control provided by PTR is critical in inflammatory and immune responses, where changes in concentration and length of expression for dangerous and protective genes are in dynamic balance and become critical in determining either successful resolution of acute inflammation or chronic overexpression of the adaptive immune response (Shyu and Wilkinson 2000; Anderson et al. 2004; Baltimore et al. 2008).

The key importance of PTR as regulatory hub in inflammation was clearly illustrated early on by two independent studies in mouse models whose design was complementary, as one component of the RNP complex regulating TNFa mRNA expression was disrupted in each study: in the work by Kontoyannis et al. by mutation of the adenylate-urydilate-rich elements (ARE) present in the mRNA 3'UTR regulating the transcript's turnover and translation rates (Kontoyiannis et al. 1999) and, in the study by Taylor et al., by ablation of tristetraprolin (TTP), the ARE-binding protein necessary for rapid degradation of TNFa and many other ARE-bearing inflammatory transcripts (Taylor et al. 1996). In both models, circulating TNF α levels were aberrantly increased without changes in transcription rates in the TNFα-producing macrophages. Strikingly, both models displayed strong inflammatory phenotypes that were largely overlapping: in mice carrying the deletion in the TNF α ARE, high circulating levels of TNF- α associated with early onset of inflammatory cell infiltration within the joints and the bowel, closely resembling the clinical features of human rheumatoid arthritis and Crohn's disease, respectively (Kontoyiannis et al. 1999). Along the same lines, mice lacking TTP displayed early onset of cachexia, severe inflammatory arthritis, autoimmune dysfunction and myeloid hyperplasia through the overexpression of TNFa and GM-CSF (Taylor et al. 1996). Thus, disruption of the RNP complex regulating post-transcriptionally TNFα recapitulated major features of human inflammation and identified conclusively the pathogenic role that PTR can play when aberrantly regulated. Along the same lines, mice lacking the ARE-RBP heterogeneous nuclear ribonuclear protein D (*Hnrnpd*), also named AUF-1, a regulator of inflammatory cytokine mRNA stability, displayed another important inflammatory phenotype. The animals developed a chronic pruritic dermatitis with eczematous lesions closely resembling atopic dermatitis, with increased IgE levels, hypereosinophilia and a Th2-skewed immune profile (Sadri and Schneider 2009). Also in this model, deregulation of TNF α was partially accountable for the skin manifestations, together with CCL27 chemokine-driven mechanisms. This model further points at the essential participation of specific RBPs in skewed immune responses, which should be more closely scrutinized for therapeutic purposes.

Studies on the miRNA component of PTR have also clearly indicated, in many in vitro and animal models, their role in shaping the immune response and their participation in pathophysiological mechanisms of those chronic inflammatory and autoimmune diseases that are treated with GCs (Chen et al. 2013; Foster et al. 2013; Ivanov and Anderson 2013; Lu and Rothenberg 2013). Taken together, these data clearly indicate the importance of PTR mechanisms in controlling the inflammatory response and thus lend themselves to be studied as potentially key determinants of GC anti-inflammatory action.

1.2 Mechanisms of GC Action: A Brief Overview

Systemic or topical GCs are the first-line therapy in immune-mediated and inflammatory diseases such as persistent eosinophilic asthma, moderate to severe COPD, rheumatoid arthritis, systemic lupus erythematosus (Stellato 2007). Their anti-inflammatory action is generated through multiple mechanisms mediating coordinate changes in gene expression. While a thorough review of the molecular mechanisms of GC action is beyond the scope of this chapter and is provided by recent reviews (Barnes and Adcock 2009; Oakley and Cidlowski 2013), a brief overview with a focus on the pathways discussed herein is presented in Fig. 13.1. Binding of GC activates the glucocorticoid receptor (GR), which disassociates from a multimeric complex with chaperone proteins and translocates in the nucleus. Ligand-activated GR homodimerizes in the nucleus and acts as a transcription factor, controlling gene expression at transcriptional level (green boxes) by either DNA-dependent mechanisms, mediated by binding to GC responsive elements (GRE)—or in rare instances to negative GRE (nGRE)—or by DNA-independent mechanisms. In the latter case, GR engages in protein:protein interactions with subunits of NF-κB, AP1 and other transcription factors and cofactors, producing either transrepression of inflammatory genes, or cooperation in the induction of innate host defense (HD) genes such as Toll-like Receptor (TLR)-2 and others, together with genes carrying non-immune and metabolic functions. GC-induced genes relevant to anti-inflammatory action also act as transcriptional inhibitors (as in the case of Glucocorticoid-induced leucine



Fig. 13.1 Mechanisms of GC action discussed in this review. GCs exert their anti-inflammatory action through multiple mechanisms mediating coordinate changes in gene expression (see text). Regulation of inflammatory responses by GC is the result of integrated transcriptional and post-transcriptional pathways. The genes described so far to mediate post-transcriptional control involved in GC action modulate (*yellow boxes*) inflammatory signaling impacting mRNA turnover and translation, in part by the function of MKP-1, or regulate components that participate directly to the RNP complex conveying changes in mRNA turnover and translation, such as the RNA-binding protein TTP, several miRNAs and their processing enzymes

zipper: GILZ), or mediate inhibitory GC effects downstream of transcription. The GC-induced—or more generally, the GC-regulated genes—so far identified as affecting mRNA turnover and translation rates are involved in controlling inflammatory signaling or regulate directly the RNP complex, such as RBPs, miRNAs and their processing enzymes.

1.3 Inflammation-Related Genes Targeted by GC through PTR Effects

Alteration of mRNA turnover or translation has been documented for a large number genes involved in inflammation and immunity, such as TNF- α , IL-1 α , IL-6, IL-8, interferon- β , granulocyte macrophage colony-stimulating factor (GM-CSF), cyclooxygenase (COX)-2, IL-2, IL-3, inducible nitric oxide synthase (iNOS), vascular endothelial growth factor (VEGF) and many chemokine genes

	mRNA	mRNA region (and		
Gene	stability	element) involved	RBP	References
COX-2 ^a	Decreased	3'UTR (ARE)	Unknown	Lasa et al. (2001)
Cyclin D3	Decreased	3'UTR	Unknown	Garcia-Gras et al. (2000)
CCL2	Decreased	3'UTR, 5'UTR (G/C rich region)	TTP, GR	Ishmael et al. (2008, 2010)
CCL7	Decreased	3'UTR, 5'UTR (G/C rich region)	TTP, GR	Ishmael et al. (2008, 2010) ^b
CXCL1	Decreased	3'UTR	TTP	Ishmael et al. (2008)
Fibronectin	Increased	Intronic (unspecified)	Unknown	Ehretsmann et al. (1995)
GM-CSF	Decreased	Unknown	Unknown	Tobler et al. (1992)
IFNβ	Decreased	ARE	Unknown	Peppel et al. (1991)
IL-1	Decreased	Unknown	Unknown	Amano et al. (1993)
IL-4Rα	Decreased	Unknown	Unknown	Mozo et al. (1998)
IL-6	Decreased	3'UTR	TTP	Tobler et al. (1992); Ishmael et al. (2008)
CXCL8	Decreased	Unknown	Unknown	Tobler et al. (1992)
iNOS	Decreased	Unknown	Unknown	Korhonen et al. (2002)
LIF	Decreased	Unknown	Unknown	Grosset et al. (1999)
SP-1	Increased	3'UTR, 7.6S region	Unknown	Huang et al. (2012)
ΤΝFα	Decreased	ARE	TTP	Smoak and Cidlowski (2006)
VEGF	Decreased	Unknown	Unknown	Gille et al. (2001)

 Table 13.1
 Genes regulated by glucocorticoids through alteration of mRNA stability: mRNA binding sequences and RBP found to be involved in GC action

Modified and extended from Stellato (2004)

^a*Abbreviations*: *COX* Cyclooxygenase, *GM-CSF* Granulocyte-Monocyte Colony Stimulator Factor, *IFN* Interferon, *IL* Interleukin, *IL-4Rα* IL-4 Receptor α, *iNOS* inducible Nitric Oxide Synthase, *LIF* Leukemia Inhibitory Factor, *VEGF* Vascular Endothelial Growth Factor ^bComplete list of TTP-dependent and GR-dependent GC sensitive genes, respectively, in these

references

such as CXCL1, CCL2, CCL7, CCL11, CCL13, and many others (Amano et al. 1993; Tobler et al. 1992; Shaw and Kamen 1986; Peppel et al. 1991; Carballo et al. 2000; Sawaoka et al. 2003; Stoecklin et al. 2003; Ogilvie et al. 2005; Fechir et al. 2005; Fan et al. 2011; Hamilton et al. 2010)—a comprehensive list is in recent reviews (Anderson 2010). The majority of these genes have been demonstrated to be GC-sensitive in many experimental and therapeutic settings and, at the same time, it is known that GC sensitivity can vary in different cell types and can be hampered by inflammatory conditions, leading to GC resistance or at least loss of full therapeutic control. However, the list of GC-regulated genes for which the existence of a PTR component is demonstrated is much shorter (see Table 13.1), though studies utilizing genome-wide approaches are accelerating the pace of the identification of PTR-dependence of GC-mediated gene regulation (Ishmael et al. 2008, 2010). Knowledge of the participation of aberrant PTR response as mechanism of GC resistance is even more ill-defined.
2 Post-transcriptional Determinants of GC Anti-inflammatory Action

Mechanistic knowledge on how immune responses rely on PTR for coordinate expression of cytokines, chemokines and other inflammation-related factors is lending a strong rationale to investigate GCs' ability to target PTR mechanisms as part of their anti-inflammatory activity. Following are studies that have started unraveling the specific PTR determinants—the signaling mediators, specific regulatory functions, RNA binding sequences, RBPs and miRNA functions, etc.—involved and necessary for GC action. Further studies are necessary to fully characterize GC-driven PTR components and to develop experimental systems that could establish and measure the overall impact of PTR in GC's anti-inflammatory therapeutic action.

2.1 GC Effect on Signaling Pathways Regulating PTR Functions

According to transcript-, stimulus- and cell-specific-mechanisms, multiple signaling pathways can regulate-through different post-translational modifications-the trafficking, binding affinity, compartmentalization and other functional aspects of RNA-binding proteins and miRNAs. Members of the mitogen-activated protein kinases (MAPK) family, such as the stress-activated protein kinase/c-Jun N-terminal kinase (SAPK/JNK), p38/SAPK2 and extracellular regulated kinase (ERK) are essential-though obviously not exclusive-regulators of PTR, in full integration with their upstream control on gene transcription (Kracht and Saklatvala 2002; Gaestel 2006). Post-transcriptional activities of MAP kinases have been increasingly well-characterized in mechanistic terms and recognized as necessary for inflammation-driven overexpression of a large number of inflammatory targets (Kracht and Saklatvala 2002; Dean et al. 2004; Anderson 2008; Gaestel 2013). At the same time, inhibition of MAPK signaling constitutes a major and powerful component of GC anti-inflammatory action, better characterized with regard to the MAPK signaling control of transcriptional mechanism (Kracht and Saklatvala 2002; Clark et al. 2003; Clark and Lasa 2003). Although MAPK-dependent transcript stabilization by inflammatory stimulation has been demonstrated for many transcripts that are described as susceptible to GC inhibition through acceleration of mRNA decay-among these IL-1, VEGF, COX-2, TNFa, IL-6 and IL-8 and many chemokines (Amano et al. 1993; Gille et al. 2001; Lasa et al. 2001; Swantek et al. 1997; Tobler et al. 1992; Ishmael et al. 2008; Stellato 2004), relatively few studies to date have *directly* investigated the PTR regulatory sequences and binding partners targeted by GC as part of the inhibitory effect on MAPK-dependent gene regulation, and attempt to measure the overall anti-inflammatory impact of this regulatory mode. Such experimental evidence would be relevant for identifying specific pathways potentially targetable for a selective anti-inflammatory therapeutic intervention. The most relevant studies on GC effect to this end have focused, in several experimental models, on p38-regulated PTR showing that treatment with specific p38 inhibitors in activated cells antagonized kinase-induced mRNA stabilization (or increased translation) and promoted acceleration of mRNA decay or decreased translation, effects that were found to be mediated by the ARE-bearing 3'-UTRs of these genes (Kracht and Saklatvala 2002; Zhang et al. 2002; Dean et al. 2004). One of the first studies to characterize the involvement of p38 MAPKmediated PTR by GC action used COX-2 mRNA turnover as model (Lasa et al. 2000, 2001). The decay of a β -globin reporter including the ARE-bearing 3'-UTR of COX-2 mRNA was transiently transfected in HeLA cells and examined upon stimulation by a constitutively active form of MAPK kinase-6 (MKK-6), an upstream activator of p38, which increased the reporter's mRNA stability. This effect was inhibited both by cell pretreatment with dexamethasone (Lasa et al. 2001) and p38 antagonists (Lasa et al. 2000). As in multiple systems (Clark and Lasa 2003), GC cell treatment inhibited stimulus-induced p38 phosphorylation, supporting the conclusion that GC effect on the reporter mRNA turnover occurred through the inhibition of p38 MAPK activity, in turn responsible for COX-2 mRNA stabilization (Lasa et al. 2001).

The effect of GC on MAPK-driven PTR control-that mediated by p38 in particular-was later found to be significantly linked to GC-induced transcriptional activation of the MAPK phosphatase 1 gene (MKP-1), which antagonized the p38-mediated stabilization of COX2 mRNA (Lasa et al. 2002). GC-driven MKP-1 induction was confirmed in several cell types including macrophages and airway epithelium (Imasato et al. 2002). Earlier data showed that GC effects on mRNA stability of COX-2 and other inflammatory transcripts did require de novo gene expression (Newton et al. 1998), supporting the hypothesis that MKP-1 contributed to the post-transcriptional mechanisms of GC action observed for COX-2 regulation. More recent studies using either transgenic or KO models of MKP-1 gene have shown its non-redundant role as negative regulator of inflammation and as important factor of GC anti-inflammatory mechanism, as suggested by the significant impairment of GC response in macrophages of MKP-1-/- mice (Abraham and Clark 2006; Liu et al. 2007). Recently, overexpression or ablation of MKP-1 in LPSstimulated rat macrophages was found to regulate levels of IL-6, IL-10 and TNF α mRNA through changes in mRNA stability in conjunction with regulation of nucleocytoplasmic shuttling of AUF-1, the ARE-binding RNP promoting the mRNA decay of those cytokines (Yu et al. 2011) and, as discussed earlier, shown in KO model to be involved in control of TNF- α , chemokine and Th2-driven inflammation (Sadri and Schneider 2009). Studies on GC effects in similar systems will be necessary to identify the specific molecular species downstream of MAPK through which GC exert PTR control of inflammatory signaling-and how much of it is dependent on MKP-1—in order to dampen the inflammatory process.

The PTR effects by GC documented to be ARE-dependent and mediated by MAP kinases indicate that such mechanism can be cell-specific (Lasa et al. 2001, 2002) and, possibly, stimulus- and transcript-specific. As not all ARE-containing genes are p38-dependent (Tebo et al. 2003; Frevel et al. 2003), the signaling

pathways conveying ARE-dependent, GC-driven PTR requires further scrutiny. It is clear, in fact, that additional mechanisms of control by GC on PTR signaling may very well exist outside the p38-MKP1 axis. For example, dexamethasone inhibited cytokine-induced IL-8 in human airway epithelial cell lines by acceleration of mRNA decay, in parallel with induction of MKP-1 expression; however the effect on IL-8 mRNA stability was not recapitulated by overexpression of MKP-1 nor influenced in opposite way by its silencing, suggesting that the post-transcriptional effect of GC on IL-8 expression can be conveyed by molecular species downstream of MKP-1 (Dauletbaev et al. 2011). In particular, the effect of GC on MAPK-activated protein kinases (MAPKAP) 2 and 3, which are master regulators of mRNA decay and translation through phosphorylation of TTP and other major RBPs deserves further experimental testing (Moens et al. 2013; Gaestel 2013).

2.2 Sequence Elements for Regulations and RNA-Binding Proteins Conveying GC Action

Genes required for a specific biological process—a specific pool of transcription factors, cytokines, chemokines, enzymes required for a TLR-mediated activation, for example—can be regulated post-transcriptionally in a coordinate fashion via a specific RBP that binds to common regulatory sequences present in those mRNAs, as a post-transcriptional regulon (Anderson 2010; Keene 2007). Therefore, control of expression or function of the RNP components by GC may convey a large, yet rather specific impact on gene networks involved in specific pathophysiological processes, by acting on a regulatory step well downstream of transcription. In keeping with the immunomodulatory function of GC action, it can be expected however that components of post-transcriptional gene control may change in sensitivity to GC according to tissue- and cell-specific issues, and according to the type of inflammatory event and the ensuing signaling. Characterization of GC action on the components of the RNP complex—the RNA regulatory sequences and their binding partners—is a growing area of investigation aimed at identifying more specific anti-inflammatory strategies.

GC and Sequence Elements for Regulations The interaction of regulatory factors—RNA-binding proteins, microRNAs and potentially other sncRNAs—with sets of relatively specific sequences present on the targeted mRNA is at the core of post-transcriptional gene regulation (Bakheet et al. 2006). The targeted sequences— heterogeneous in sequence, secondary structure, position and specific functions— are more frequently represented in the 5' and 3' untranslated regions (UTR) of the mature cytoplasmic mRNA, and have been collectively denominated USER (Untranslated Sequence Elements for Regulations) (Wilusz and Wilusz 2004; Keene 2007). The majority of inflammation-driven, MAPK-dependent PTR as well as many of the described post-transcriptional effects by GC occur through association of regulatory factors to adenylate/uridylate-rich elements (AREs) within the 3'-UTR of mRNAs. This evolutionarily conserved, heterogeneous group of sequences have

been the first to be described as central in PTR events (Shaw and Kamen 1986) and in immune-mediated PTR processes (Chen and Shyu 1995; Anderson 2010), as indicated by the fact that immune-related genes are highly enriched in AREs (see http:// rc.kfshrc.edu.sa/ared, http://rna.tbi.univie.ac.at/AREsite) (Bakheet et al. 2006; Gruber et al. 2011). In one of the first demonstrations of ARE-dependent GC effect, macrophages from mice expressing the TNF- α gene carrying the deletion of the ARE (Δ ARE) overexpressed TNF- α protein due to decreased decay rate of TNF- α mRNA as well as to loss of translational inhibition (Kontoyiannis et al. 1999). In these cells, LPS-induced increase in TNF- α became unresponsive to the inhibitory effect that a p38 inhibitor displayed on WT cells, showing dependence of p38/JNKmediated activation of TNF-a translation. Conversely, cell treatment with dexamethasone of \triangle ARE macrophages retained in part its ability to inhibit TNF- α secretion, suggesting that the ARE-dependent inhibition of TNF- α is an important—but not exclusively p38/JNK-mediated- PTR mechanism of inhibition of TNF- α by GC. There are currently no data on GC regulation on ARE-BP acting as repressors of TNF-α translation, such as T-cell intracellular antigen (TIA) and TIA-related protein (TIAR) (Zhang et al. 2002). As further experimental proof of ARE-dependent GC action, the dexamethasone-induced acceleration of an mRNA reporter expressed by a construct containing COX-2 3'UTR was suppressed by deletion of a small ARE cluster within the 3'-UTR, which bears a large number of AREs (Lasa et al. 2001). Dependence from 3'UTR of GC action on mRNA stability, though not specifically addressing ARE-dependence, has been described for several inflammatory genes (Stellato 2004) (see Table 13.1). No data are so far available on the participation to GC action of other classes of 3'UTR USER, such as the GU-rich elements (Halees et al. 2011) and is so far unknown whether endogenous GC would support PTR mechanisms keeping inflammatory transcripts highly unstable at baseline conditions, as documented through binding of the RBP Roquin to the constitutive decay element (CDE) present as well in the 3'UTR for TNF α mRNA and other transcripts (Leppek et al. 2013). Other sequences outside the 3'UTR have been found to mediate the effect of GC on mRNA turnover and/or translation. In particular, GC-mediated PTR has been recently linked to sequences present in the 5'UTR of CCL2 mRNA and other transcripts. Initial data generated in rat smooth muscle cells (SMC) indicated that the increase in decay rate of CCL2/MCP-1 mRNA induced by cell incubation with GC was not dependent on the ARE-bearing 3'-UTR but was rather mediated by a unique sequence in the 5'-UTR (Poon et al. 1999). Recently, GC-induced increased decay of CCL2 mRNA in human airway epithelial cells has been ascribed to a Guanidine/Cytidine (GC)-rich motif identified in the 5'UTR region in the CCL2 mRNA (Ishmael et al. 2010). This sequence mediated the transcript association of a complex containing the GC receptor (GR), which operated in this experimental setting as a bona-fide RNA-binding protein (see next paragraph).

An important non-immune gene induced by GC in alveolar type 2 epithelial cells encodes the surfactant protein B (SP-B), necessary for appropriate function of the surfactant fluid lining the alveolar space in the lungs. Together with transcriptional control, GC increase SP-B mRNA stabilization by a 3'UTR-dependent mechanism requiring a 126-nt-long segment denominated 7.6S region (Huang et al. 2008).

Mutagenesis studies indicated that within this region, a 30-long element predicted to form a stem-loop structure was sufficient for GC-mediated mRNA stabilization (Huang et al. 2012).

Considering the number of regulatory sequence elements within 5'-UTR and 3'-UTR of eukaryotic mRNAs besides ARE (see http://utrdb.ba.itb.cnr.it/) (Grillo et al. 2010), the interactions that could occur among the regulatory factors assembling over multiple USERs present in the same transcript, together with specific changes brought to the RNP complexes by stimulus-dependent signaling, it is evident that much more remains to be discovered on the identity and characteristic of additional USER supporting GC action and their binding partners.

RBPs as Targets/Mediators of GC Function RNA-binding proteins are among the most abundant and well-conserved eukaryotic genes (Gerstberger et al. 2014). They regulate several aspects of the mRNA metabolism, such as its maturation, nucleocytoplasmic transport, subcellular localization, rate of decay, and translation (Dreyfuss et al. 2002; Anderson 2008, 2010) and their key regulatory roles in immune responses is now well established (Ivanov and Anderson 2013; Yuan and Muljo 2013). Target-bound RBPs implement their functions by forming dynamic multimeric complexes with miRNAs, other RBPs and RNA-degrading enzymatic proteins not directly interacting with the mRNA (Jiang and Coller 2012; Ciafre and Galardi 2013; Srikantan et al. 2012; Steitz and Vasudevan 2009), often co-localized in cytoplasmic foci of mRNA metabolism (P bodies and stress granules) (Anderson and Kedersha 2009; Bhattacharyya et al. 2006; Jing et al. 2005). Numerous AREbinding proteins (ARE-BPs) have been functionally characterized as regulatory factors of mRNA decay and translation of inflammatory and immune genes [for review see (Dreyfuss et al. 2002; Stoecklin and Anderson 2006a; Anderson 2008, 2010)], but very few have been investigated with regard to their potential function in GC action. While numerous proteins-among them TIA-1, TIAR, FXR1P, CUGBP2can exert translational repression, changes in mRNA stability leading to prolonged half-life and increased expression of a vast number of inflammatory genes have been ascribed mostly to HuR, the ubiquitous member of the Hu family of RBPs (Brennan and Steitz 2001; Meisner and Filipowicz 2011; Simone and Keene 2013) and to YB1, described for a more limited number of transcripts, among which IL-2 mRNA (Chen et al. 2000; Lyabin et al. 2014). In contrast, numerous RBPs have been characterized as involved in destabilizing ARE-bearing inflammatory mRNAs, such as AUF1, KSRP, RHAU, and members of the TTP/BRF family of proteins (for review see Stoecklin and Anderson 2006b). Despite their documented role as regulatory factors for established GC-sensitive genes-cytokines, chemokines and enzymes involved in inflammation-as well as their strong functional dependence from MAPK signaling pathways (Sugiura et al. 2011) that are as well GC-sensitive, the biology of RNA-binding proteins in human inflammatory diseases is presently ill-defined and knowledge on the mechanisms of control by GC, or by antiinflammatory intervention in general, on their expression and function is lagging behind in translational research settings. Following are the main observations of studies that have so far addressed directly the role of RBPs in GC-mediated PTR.

Tristetraprolin: A Mediator of Post-transcriptional GC Action Tristetraprolin (TTP) is the product of the ZFP-36 gene (also known as TIS11, Nup475, and GOS24) and member of a family of CCCH zinc finger proteins that include TTP, Butyrate-response factor (BRF)-1 and BRF-2 (Carrick et al. 2004). These proteins mediate deadenylation-dependent mRNA decay (Raineri et al. 2004) through binding of zinc finger domains to adjacent UUAU/UUAU half-sites, which trigger complex interactions with components of the exosome, the decapping/Xrn1 complex, and with the RISC complex (Worthington et al. 2002; Blackshear et al. 2003a; Brewer et al. 2004; Stoecklin and Anderson 2006b). Tristetraprolin is induced as an immediate early response gene by inflammatory mediators, phorbol esters, LPS and growth factors in a number of immune and structural cell types, including T cells, macrophages and fibroblasts where it displays a predominantly cytoplasmic localization (Blackshear 2002). TTP is also expressed in lung, liver, and intestine tissues (Cao 2004; Lu and Schneider 2004). As previously mentioned, the importance of TTP as endogenous rate-limiting factor of the inflammatory response has been convincingly demonstrated in TTP^{-/-} mice, which develop severe inflammatory arthritis, autoimmune dysfunction and myeloid hyperplasia through the deregulated expression of TNF- α and GM-CSF (Taylor et al. 1996). Tumor necrosis factor α is both a major target and regulator of TTP, as TNF- α induces TTP synthesis, which in turn leads to destabilization of TNF-α mRNA (Carballo et al. 1998). TTP also modulates mRNA decay rates of GM-CSF, COX-2, INOS, IL-2, IL-3, IL-10, and several chemokines, such as CCL2, CCL7, CXCL1, CXCL8 and others (Lai et al. 2006; Carballo et al. 2000; Sawaoka et al. 2003; Stoecklin et al. 2003; Ogilvie et al. 2005; Fechir et al. 2005; Ishmael et al. 2008). Additional transcripts whose decay is regulated by TTP have been identified in a recent genome-wide study of mouse embryonic fibroblasts (MEFs) isolated from TTP-knockout (TTP-/-) mice (Lai et al. 2006), and in mouse macrophages in which TTP expression was silenced (Jalonen et al. 2006). Importantly, phosphorylation of TTP by p38-MAPK and the downstream MAPKAP, MK2 (Carballo et al. 2001; Mahtani et al. 2001) is an important mechanism of post-transcriptional control during inflammation, as it leads to functional inactivation of TTP due to an inhibitory complex with the adaptor protein 14-3-3 (Chrestensen et al. 2004; Johnson et al. 2002; Stoecklin et al. 2004), though this could be dependent by contextual stimuli (Rigby et al. 2005). Furthermore, TTP has a key role as MK2/3-regulated RBP in limiting TNFa translation (reviewed in Gaestel 2006, 2013). Overall, the function, regulation and targeted genes of TTP make this molecule an ideal mediator of GC anti-inflammatory activity.

Indeed, expression of TTP and BRF-1 has been reported to be induced by GC in primary human airway epithelial cells and epithelial cell lines (Pawliczak et al. 2005; Ishmael et al. 2007). Induction of TTP by GC has been shown in vivo by treatment with dexamethasone in adrenalectomized rats, which caused a three- to fourfold increase of TTP mRNA expression in the lung, liver and thymus (Smoak and Cidlowski 2006). Based on these results, the mechanism of TTP induction by GC was further defined using the airway human epithelial cell line A549. An important characteristic of TTP mRNA, which was detectable up to 8 h following GC

stimulation. This contrasts with the rapid but transient induction by $TNF\alpha$, where TTP mRNA peaked after 2 h and returned to baseline within 4 h from stimulation. At the protein level, both TTP and TNFa sustained in A549 cells a prolonged expression of TTP up to 24 h post-stimulation. Concordance of the action on TTP expression by TNFa and GC may appear counterintuitive, yet it can be reconciled viewing TTP as a molecule involved in homeostatic limitation of an acute inflammation, a process which 'begins with the end in mind' (Anderson 2010), with induction of TTP being advantageous as a negative feedback mechanism in both settings. From a mechanistic standpoint, GC-induced TTP expression was found to be transcription-and GR binding-dependent, as it was abolished by cell pretreatment with the transcriptional inhibitor actinomycin D and by GR antagonist RU486, respectively; paired with experiments showing no change in TTP mRNA stability, run-on experiments confirmed that TTP induction by GC was driven in A549 cells by increased transcription, though a bona fide GRE was not identifiable from the published hTTP sequence. Chromatin immunoprecipitation (ChIP) experiments in GC-treated cells showed enrichment of GR recruitment in two regions of the TTP gene: a 5' flanking portion of the promoter containing binding sites for STAT, Smad, and NF-KB-all transcription factors known to engage protein-protein interaction with GR-and a portion of the 3' flanking region which contains a half-GRE sequence. Silencing of TTP expression in A549 cells significantly impaired inhibition of its major target, TNF- α by GC, demonstrating a functional role of GC-induced TTP. This process was dependent on the TNF- α 3'UTR, as GC treatment selectively decreased, in transfected A549 cells, the expression of a luciferase reporter carrying the TNF- α 3'UTR. These results indicate for the first time that GC inhibition of TNF- α is mediated by 3'UTR-mediated post-transcriptional mechanisms critically dependent by induction of TTP. However, additional tissue- and/or stimulus-specific factors regulating GC influence on TTP expression are also suggested by a study reporting that LPS-induced TTP expression is instead inhibited by GC in mouse macrophages (Jalonen et al. 2005).

The work by Ishmael et al. (2008) extended GC sensitivity of TTP expression also to human primary bronchial epithelial cells and examined the role of TTP in GC anti-inflammatory action on a genome-wide scale. Mouse embryonic fibroblasts (MEFs) cell lines developed from TTP KO mice, in comparison to MEFs of wild-type (WT) littermates were treated the topical GC budesonide or diluent control for 3 h and then treated in the absence or presence of 10 ng/ml TNF- α for 1 h. The number of genes regulated in WT cells was largely reduced in TTP-/- cells, with the greatest loss of response (97 %) accounted for genes that in WT were down-regulated by GCs: only 5 of the 145 GC-responsive genes in WT were still comparably repressed in TTP-/- cells (Fig. 13.2a). Interestingly, also GC-induced genes were diminished in TTP-/- cells by 89 % compared to WT cells, suggesting a vast network of indirect effects. TNF- α -dependent responses were also globally affected in the absence of GC treatment, pointing again to the role of TTP in limiting the amplitude of acute inflammation either in response to triggering stimuli or as part of the endogenous anti-inflammatory mechanisms. Of particular relevance, the chemokines CCL2, CCL7, CXCL1, CXCL5 and CXCL7 were among the genes whose inhibitory response to GC was significantly lost in TTP^{-/-} MEF. Such targeting reinforces the importance of TTP-mediated function in the anti-inflammatory mechanism of GC action: chemokines are in fact key determinants of recruitment and activation of inflammatory cells within the inflamed mucosa and as such, they are main targets of GC therapy in chronic inflammatory diseases such as rhinosinusitis, asthma and COPD (Stellato 2007; Stellato and Schleimer 2000). Genome ontology analysis confirmed that regulated genes were functionally enriched in immunity and signal transduction, processes that are chiefly targeted by GCs for their anti-inflammatory action (Fig. 13.2b). Involvement of TTP also on genes important for GC control of metabolism and cell development indicate that TTP is likely to mediate other homeostatic and nonimmune functions driven by GCs. Loss of GC response was validated in the same study for selected targets in WT MEFS, as well as in primary human epithelial cells (unpublished data, C. Stellato), following transient TTP silencing, validating the general hypothesis beyond the TTP^{-/-} mouse model.

Probing the mechanism of TTP function in GC response revealed a high level of complexity involving direct, ARE-mediated effects as well as indirect effects. For a group of 11 transcripts selected among those that lost their GC sensitivity in TTP-/cells, those bearing TTP- compatible sequences such as AUUUA pentamers or UUAUUUAUU nonamers (Stoecklin et al. 2008)-CCL2, CCL7, CXCL1, CXCL5 as well as IL-6 and MMP9-showed association with TTP, documented through selective immunoprecipitation of ribonucleoprotein complexes (RNP-IP) using anti-TTP ab and subsequent real-time PCR; the transcripts showing no association with TTP by RNP-IP, despite profound changes in GC sensitivity in TTP-/- cells-IL1RL1, CXCL7, CCL5, Serpin3n, Egr1-presented instead only one or no AREs in the 3'-UTRs, with the exception of EGR-1 mRNA, which bears two UAUUUAU heptamers in an A/U-rich setting (Fig. 13.2c). However, among the transcripts found to be associated to TTP, GC-dependent changes in mRNA decay rate in TTP-/- cells-assessed by actinomycin D treatment-were heterogeneous, challenging the assumption that TTP would mediate GC action exclusively by accelerating ARE-mediated mRNA decay (Fig. 13.2d). For CCL7 and CCL2 mRNA, the acceleration of mRNA decay induced by GC in WT MEFs was no longer present in TTP-/- cells, in line with the more straightforward hypothesis. On the contrary, IL-6 mRNA decay was not modified by either GC treatment or by lack of TTP, despite significant mRNA enrichment in the RNP-IP assay and the presence of a 3'-UTR extremely rich in AREs. CXCL5 mRNA became unstable in TTP-/- cells compared to WT cells already in the absence of GC, indicating that TTP supported in this case a relative stabilization of CXCL5 basal mRNA turnover rate and that the inhibitory effect of GC seen in TTP-/- cells was likely a consequence to the loss of this function, rather than the GC action in itself. Further remarking the complexity of TTP biology-and possibly also the limitation of the Actinomycin D-based experimental system used in the study—these data implicate that besides ARE-mediated mRNA decay, TTP could form RNPs through protein interaction with other RBPs (Anderson 2008) and/or by establishing cooperative or antagonistic function with miRNAs (Jing et al. 2005; van Kouwenhove et al. 2011)



Fig.13.2 GC-mediated gene regulation is significantly altered in TTP ^{-/-} murine fibroblasts. (a) Differential regulation of gene expression in WT and TTP ^{-/-}
MEFs. Left, Heat maps of the array data sets from WT or TTP ^{-r} cells and right, changes in GC response for selected genes (expressed as Z ratio) in WT vs
TTP cells upon exposure to budesonide (Bud) versus diluent-treated cells. (b) Genome ontology analysis by Gene Set Matrix Analysis (GSMA) showing loss
of GC response for genes involved in asthma pathogenesis in TTP ^{-/-} MEFs. The Gene Set Matrix Analysis (GSMA) algorithm allows the identification of
coordinated changes in gene expression. Experimental conditions are examined against "gene lists" pooled according to a biologically relevant grouping system
instead of against each gene present on the array. (c) Association of TTP with GC-regulated mRNA targets identified by immunoprecipitation of endogenous
mRNA-TTP complexes. Upper right, Representative real-time PCR amplification plot of fluorescence intensity over background (ΔRn) against PCR cycle (C_r)
showing enrichment of CXCL1, but not of 18S RNA transcripts, in lysates of GC-treated MEFs subjected to IP with an anti-TTP Ab (filled symbols) vs the IP
with the isotype-matched Ab (open symbols). Bargraph, enrichment of immunoprecipitated transcripts by the anti-TTP relative to IP with isotype Ab control
(fold values indicated). Of the 11 transcripts tested, 6 (grev bars) contain multiple AREs. (d) Differential effect of TTP loss in GC-induced CCL7 and IL-6
mRNA decay. Results are shown as the percent of mRNA at time 0 in actinomycin-treated cells. The bar graph insets indicate the half-life (t ₁₀₂ , in h) in each
condition. Modified from Ishmael et al. (2008), reprinted with permission

(see following dedicated paragraph). Prototypic of this context-driven, rather than strictly ARE-dependent, function is how TTP promote stabilization of the ARE-rich INOS mRNA by interacting with the ARE-bound KH-type splicing regulatory protein (KSRP). Under proinflammatory stimulation, TTP association with KSRP would remove this mRNA decay-inducing factor and the associated exosome and allow binding for the RBP HuR, resulting in INOS mRNA stabilization (Fechir et al. 2005). Many of the transcripts found to be up-regulated upon GCs treatment in TTP^{-/-} cells could be similarly susceptible of TTP regulation through critical changes within the regulatory RNP complexes. In addition, GC could modulate TTP function also through changes in TTP phosphorylation, thus affecting functional aspects (change its localization, function or affinity for its binding site) rather than solely its synthesis.

TTP-mediated effect of GC can indirectly affect a larger gene pool when targeting a regulatory protein: an example from the study by Ishmael is the TTPdependence of GC inhibition of the transcription factor Egr-1, which would also affect downstream expression of Egr-1-dependent proinflammatory genes (Cho et al. 2006; Ingram et al. 2006) underscoring how regulation of mRNA turnover can indirectly affect transcriptional control of multiple genes.

The effect and mechanisms of GC action on TTP-mediated translational control of gene expression, whose importance is documented for TNF α , remains to be explored as for that exerted on other RBPs known to regulate translation—such as TIA, TIAR and others (Gaestel 2013).

Glucocorticoid-Receptor (GR)-Mediated Chemokine Decay: A Novel Mechanism of GC-Mediated PTR The glucocorticoid receptor (GR) is a steroid hormone receptor belonging to the superfamily of nuclear receptors (NR), which exert gene regulatory functions as transcription factors (Gronemeyer et al. 2004). Transcription factors, such as the GR, and RBPs share important functional characteristics, such as sequence-specific interaction with nucleic acids, activation-dependent nucleocytoplasmic shuttling, and the ability to affect multiple targets through binding to conserved nucleic acid sequences. Taking this similarity even further, some protein factors can bind to both DNA and RNA-as in the case of the zinc-finger protein NF-90 that regulates IL-2 transcription through binding to the NFAT site (Shi et al. 2007) as well as mRNA turnover and translation of a vast group of transcripts through AU-rich-specific interactions (Kim et al. 2007). Other two zinc-finger proteins, Yin-Yang (YY)1 and TTP, as well as nucleolin participate in both nuclear and cytoplasmic regulatory events by ribonucleoprotein- and protein-protein interactions either with DNA and transcription factors or with RNA, miRNA and RBPs in the cytoplasm (Chen et al. 2000; Singh et al. 2004; Mongelard and Bouvet 2007; Liang et al. 2009; Stoecklin et al. 2008).

Whether the GR may have a similar ability was hypothesized based on the growing data on PTR contribution to GC action (Stellato 2004). The specific role of GR in mRNA stability was tested for the first time in a model of rat aortic smooth muscle cells (SMC) (Dhawan et al. 2007), based on earlier observation by the same group of the ability of the synthetic GC Dexamethasone (Dex) to accelerate mRNA decay for CCL2/MCP-1 mRNA (Poon et al. 1999). Experiments using RNA electromobility



Fig. 13.3 The Glucocorticoid Receptor (GR) associates with MCP-1/CCL2 mRNA and promotes its degradation. Rat aortic endothelial cells were treated with diluent control (C)/dex (D) for 3 h and cytoplasmic lysates was incubated with full length, in vitro transcribed CCL2/MCP-1 mRNA for 30 min \pm GR Ab for EMSA essay. (a) Incubation with control cell lysates (C) shows a probe shift indicating formation of an RNP complex; CCL2 probe is degraded in dex-treated cells (D), an effect prevented specifically and concentration-dependently by lysate preincubation with an anti-GR ab. (b) Direct binding of GR with in vitro transcribed CCL2/MCP-1 mRNA was demonstrated by complex formation following probe incubation with recombinant GR; the interaction was specific with respect to other nuclear receptors. Modified from Dhawan et al. (2007), reprinted with permission

shift assay (R-EMSA) showed an upward mobility shift of the complex formed by cytoplasmic extracts of Dex-treated SMC after a short incubation (5 min) with radiolabeled full length CCL2/MCP1 mRNA compared to control cytoplasmic extracts, suggesting that additional proteins bind to the transcript under Dex control. Using a cell-free mRNA stability assay, in which degradation of a labeled synthetic CCL2/MCP-1 transcript was detected over time following incubation with cytoplasmic SMC lysates, the authors showed that incubation for 30 min of the Dextreated SMC extracts with the CCL2/MCP-1 probe led to its complete degradation, while the extracts from control SMC did not change its detection; strikingly, the probe degradation was specifically inhibited by preincubation of Dex-treated SMC extracts with multiple anti-GR antibodies (Fig. 13.3a). Direct GR-mRNA interaction was confirmed by R-EMSA using recombinant human GR and an in vitro transcribed, human CCL2/MCP-1 mRNA probe (Fig. 13.3b). This interaction was hormone receptor—and mRNA—specific, as neither incubation with other NRs or usage of three alternative transcripts reproduced the results obtained with GR-CCL2 coupling. It is also noteworthy that the shift observed with extracts from Dex-treated cells was higher than that obtained by rhGR alone, suggesting that in intact cells, GR is likely part of a multimeric RNP complex; this was further suggested by the ability of recombinant GR to inhibit transcript degradation by Dex-treated extracts and by enrichment of MCP-1 mRNA detected by immunoprecipitation of RNP complexes (RNP-IP) with an anti-GR specific antibody. These data indicated a previously unknown cytoplasmic function for the GR as well as the existence of novel regulatory sequences, in addition to the conserved ARE, as determinants of CCL2 mRNA stability.

The study by Ishmael et al. (2010) aimed to study whether this novel mechanism was in place in a human cellular model of GC response and tested the hypothesis that in functioning with an RBP-like function, GR-mediated control of mRNA decay would have a more global role in GC-mediated PTR action. First, R-EMSA experiments conducted with cytoplasmic extracts from human primary airway epithelial cells and the epithelial cell line BEAS-2B indicated that biotin-labeled CCL2 and CCL7 mRNA, but not CCL5, associated with GR upon cell treatment with the synthetic GC budesonide and were degraded by GC treatment, an outcome prevented by GR neutralization by a specific anti-GR ab in epithelial cell lysates. The association was validated by RNP-IP (Helou et al. 2008) in which cytoplasmic lysates from control and GC-treated epithelial cells were incubated with anti-GR antibody-or with an IgG isotype control to measure background binding-to immunoprecipitate endogenous GR-mRNA complexes, and identified bound CCL2 and CCL7 by PCR in GC-treated samples, similarly to what the same group reported for chemokine transcripts interaction with TTP (Ishmael et al. 2008). Biotin pulldown experiments probed for the binding site of GR using truncated transcripts containing the 5'UTR, coding region (CR) and 3'UTR sections (Fig. 13.4a). In the study of GC-induced decay of rat CCL2 mRNA, transcript stability was not dependent from the ARE or other sequences in the 3'UTR but relied instead on the presence of three stem loops in the transcript 5'-UTR (Poon et al. 1999). As in the rat CCL2 system, association of GR was observed predominantly with the 5'UTR construct, with minimal signal with the 3'UTR construct. The binding site was further mapped through cross-linking of purified recombinant GR with truncated 5'UTR regions (indicating actual nucleic acid-protein interaction) and ultimately localized to a 15-mer fragment localized between nucleotide 44 and 60 (Fig. 13.4b).

As for transcription factors, RBPs interact with nucleic acids over specific cis-regulatory sequences to implement common regulatory function over multiple genes. Therefore, validation of GR function as an RBP would entail the ability to associate with multiple transcripts through a specific sequence and regulate their decay/translation rate. This paradigm was tested in the human airway epithelial cell line BEAS-2B employing RIP-ChIP analysis in which transcripts bound to the endogenous GR were identified by gene arrays after specific IP (Tenenbaum et al. 2002; Keene et al. 2006). Statistically significant mRNA enrichment in GR-IP arrays versus IgG control-IP arrays was found for approximately 500 transcripts, supporting the hypothesis of a broader RBP function for the GR besides interaction with CCL2 and CCL7 mRNAs. The common sequence motif shared by the transcripts associated with the GR, searched using a specific algorithm for RNA sequences (Lopez de Silanes et al. 2004; Kim et al. 2007; Kuwano et al. 2010), consisted in a Guanidine/ Cytidine (GC)-rich sequence (78 % of the signature motif), spanning 27–31 nucleo-tides and forming two contiguous loops on a short stem (Fig. 13.4c), that was proven



Fig. 13.4 The GR binds to CCL2 mRNA and to multiple targets in human epithelial cells via a G/C-rich Motif present in the 5' UTR. (**a**) *Top*: schematic of CCL2 mRNA, with nucleotide numbers corresponding to untranslated regions (UTR) and coding region (CR). *Below*, detection of GR by Western blot by biotin pulldown assay from unstimulated BEAS-2B cell lysate, using biotinylated CCL2 full-length RNA (FL), the 5' and 3' UTRs, or the CR. The RBP HuR is detected as positive control for association with CCL2 3'UTR. (**b**). UV cross-linking of purified GR protein to 15-nt fragments of the CCL2 5'UTR, followed by detection by Western blot of the covalently-linked GR via mobility shift of the RNA-bound GR protein (GR-RNA). (**c**) *Left*, graphic logo representing the probability matrix of the GR motif, showing the relative nucleotide frequency for each position within the motif sequence. *Right, se*condary structure of the GR motif comprising the nucleotides with highest frequency for each position within the motif shown in A. (**d**) Biotin pulldown assay showing association of recombinant GR from unstimulated BEAS-2B cell lysates with the GR motif shown in (**c**) (*sequence shown below*), compared to GR association with the fulllength 5'UTR of CCL2, CCL7 and CCL5 mRNA (the latter as negative control). Modified from Ishmael et al. (2010); Panganiban et al. (2014), reprinted with permission

to bind to recombinant GR via biotin pulldown assay. The entire UniGene database was queried based on this motif and 7889 transcripts displayed a compatible sequence in their 5'UTR (45 % of the entire database) (Fig. 13.4d).

Collectively, the data from these three studies convincingly indicate that the GR can mediate GC action beyond its core function in transcriptional control, and that it may participate in cytoplasmic regulatory mechanisms of gene expression via association with mRNA and protein complexes mediating PTR. Unquestionably, much remains to be tested to understand the molecular basis of this cytoplasmic, PTR-related function of GR and ultimately, its relevance in the global anti-inflammatory action of GC. On the structural level, it will be important to identify the GR domain interfacing with the GC-rich motif and the stoichiometry of this binding, as the GR lacks a conserved RNA-binding domain (RBD) that for other transcription factors/RBPs, like NF-90 and nucleolin, mediates association with mRNA (Ginisty et al.

1999; Shi et al. 2005). Furthermore, it will be necessary to define potential interactions with other RBPs and miRNAs that could mediate GR participation to RNPs, in addition to that mediated by RNA-bound mechanisms, or change its affinity for RNA binding. From a functional standpoint, it remains to be tested if acceleration of mRNA decay occurs for most of the putative GR motif-bearing targets identified, whether it is the only RNA-associated action of GR and whether other GR motifs, or RNP interactions, can mediate alternative cytoplasmic functions of the GR. It can be postulated that GR association with mRNAs can be modified by stimulus-induced changes in the composition of the RNP complex, and/or by action upon the association, such as P bodies and stress granules, where these functions are critically regulated during inflammation (Anderson and Kedersha 2009).

2.3 GC Effects on miRNA Expression, Metabolism and Function

MicroRNAs (miRNAs), central components of post-transcriptional gene regulation, are now recognized to play important roles in mediating and modulating the effects of GCs. MiRNAs are effectors of essential biological processes such as cell activation, development, differentiation, and apoptosis. As such, miRNAs may participate in the effects of GCs to regulate inflammatory pathways which could convey, enhance, or even inhibit GC effects.

MiRNAs are small, non-coding RNAs (20–25 nucleotides long) which are transcribed in a Pol II-dependent manner from primary transcripts or introns of genes. A number of transcription factors have been identified as stimulators of miRNA synthesis, including inflammatory transcription factors such as AP-1 and NF-kB as well as the glucocorticoid receptor. Primary miRNA transcripts (pri-miRNAs) are trimmed by a complex of proteins that includes the nuclear RNAse III Drosha and the RBP DGCR8/Pasha into a pre-miRNA hairpin which includes the miRNA sequence and a complementary guide stand. RNA-binding proteins such as KSRP, TTP, and Lin28 can bind to pri-miRNAs to enhance or retard processing to premiRNAs. Pre-miRNA hairpins are transported into the cytoplasm where they are cleaved by the RNAse III enzyme DICER into the mature miRNA and incorporated into the multiprotein RNA-induced silencing complex (RISC). The RISC binds to transcripts, usually in the 3'UTR, and promotes degradation and/or translation inhibition. GCs have been shown to regulate the expression and processing of miRNAs at multiple points in the miRNA biosynthetic pathway, as discussed below.

Effect of GC on miRNA Expression Our understanding of how miRNAs are regulated and how they exert their effects on cellular pathways has been greatly accelerated by high throughput technologies such as deep sequencing, microarrays, and qPCR analyses. Though these means, it has been revealed that GCs alter miRNA expression in a wide variety of cells and tissues (Table 13.2). The targeted miRNAs may convey anti-inflammatory, apoptotic, and metabolic effects of glucocorticoids and serve as potential therapeutic targets in multiple disease processes. The pro-apoptotic effect of GCs on lymphocytes constitutes a large component of its

•	•		
miRNA(s)	Effect	Mechanism	Cell type/disease process
Inflammation			
Let7a, miR-146a, miR-155	Anti-inflammatory effect	miRNA down-regulation	Leukocytes, lupus mouse model (Chafin et al. 2013)
miR-155	Anti-inflammatory effect	miRNA down-regulation via NFkB inhibition	Mouse macrophages (Zheng et al. 2012)
miRs-1246, -663a, -1275, -92b-5p, -24-30, -34a-e-5p, Let7c	Altered expression after inhaled GCs	Up- and down-regulation of miRNAs	Airway epithelium of asthmatics (Solberg et al. 2012)
miR-98	T-cell inhibition	Up-regulation of miR-98	Human T-cells (Davis et al. 2013)
miR-29b-c	Apoptosis	Down-regulation of miRNAs	Human plasmacytoid dendritic cells, lupus (Hong et al. 2013)
miRs-223, -16,-15b	Apoptosis	Up-regulation of miRNA	Human lymphocytes, acute lymphocytic leukemia (Rainer et al. 2009)
miR-17-92 cluster	Apoptosis, development	miRNA down-regulation by inhibition of Dicer, Drosha, DGCR8	Human lymphocytes (Smith et al. 2010), acute lymphocytic leukemia (Harada et al. 2012) chondrocytes (Xing et al. 2014)
Metabolic			
miR-29a	Differentiation	Transcriptional down-regulation of miRNA	Human osteoclasts (Ko et al. 2013)
miR-27b	Fat accumulation	GR-dependent transcriptional up-regulation of miRNA	Adipocytes (Kong et al. 2014)
miR-1	Skeletal muscle atrophy	GR-dependent transcriptional up-regulation of miRNA	Mouse muscle cells (Kukreti et al. 2013)
miRs-1, -147,-322,-351, -503*, -708	Skeletal muscle atrophy	Up- and down-regulation of miRNAs	Mouse muscle cells (Shen et al. 2013)

anti-inflammatory actions and is a central target in lymphoid malignancies. Deep sequencing and miRNA microarray analysis revealed that the miR-17-92 cluster of miRNAs was repressed in apoptotic rat lymphocytes following treatment with dexamethasone (Smith et al. 2010). These miRNAs were confirmed to be repressed by GCs in human lymphocytes by qPCR analysis, and shown to be necessary for cell survival in acute lymphocytic leukemia cell lines (Harada et al. 2012).

One of the first studies examining the effect of GC therapy on an inflammatory disease-driven miRNA signature was implemented in eosinophilic esophagitis (EoE), a chronic eosinophilic inflammatory disease of the esophagus with a strong Th2-driven pathophysiology (Lu et al. 2012). The miRNA expression profile obtained from esophageal biopsies in patients with EoE showed specific changes from those obtained from healthy control subjects and patients with chronic, noneosinophilic esophagitis; importantly, 27 of the 32 differentially expressed miRNAs were normalized in patients with EoE that were responding to glucocorticoid therapy (Fig. 13.5a). This study also identified miR-675 as a marker of response to GC therapy, as this miRNA resulted to be upregulated only in GC-responsive patients compared with the



Fig. 13.5 GC treatment in patients with eosinophilic esophagitis (EoE) reveals a GC-sensitive miRNA signature and identifies miR-675 as biomarker of response to GC treatment. (a) Heat map showing the expression level of 32 differentially expressed miRNAs in patients with EoE compared with patients with chronic esophagitis and patients with EoE in remission after GC therapy. Genes are upregulated (*red*) and downregulated (*blue*) in comparison with healthy control subjects. (b) *Above*, heat map showing changes in expression of miR-675, the only miRNA differentially regulated in patients with EoE who responded to glucocorticoid therapy, compared with that seen in healthy control subjects, patients with chronic esophagitis, and patients with EoE. *Below*, normalized expression of miR-675 detected by RT-PCR in healthy control subjects, patients with EoE, fluticasone propionate-responsive patients, and fluticasone propionate-nonresponsive patients. Modified from Lu et al. (2012), reprinted with permission

other subsets of patients and controls, but its levels were not increased in biopsies of patients with poor clinical and histological response to GC (Fig. 13.5b).

In bronchial asthma, where GC treatment is the therapeutic mainstay, expression of nine miRNAs was found to be altered in the airway of patients after GC treatment (Solberg et al. 2012). The miRNAs upregulated by GCs include Let7, which has been shown to have anti-inflammatory properties in the airway (Kumar et al. 2011). In this report, miRNA profile was also investigated in IL-13-treated airway epithelial cells from healthy donors. Of 262 regulated miRNAs, members of the miR-34/449 family (miR-34c-5p, miR-34c-5p, miR-449a, and miR-449b-5p) were significantly suppressed. Interestingly, in vitro treatment with GC of IL-13-stimulated cells did not correct this suppression. Considering that IL-13-driven changes are pathogenic in asthma, it can be hypothesized that the lack of GC sensitivity of miR-34/449 family in this setting may play a role in some of the severe asthma phenotypes showing poor response to GC therapy. Lack of regulation by GC treatment, however, may also have a different implication: for example, in keeping with the concept that activation of innate immune pathways are spared by GC regulation (Schleimer 2004), TLR4-mediated changes in miRNA profile in the lungs of mice following exposure to LPS was not affected by GC treatment (Moschos et al. 2007). The extent of GC action in regulating miRNA expression and mechanisms of action is now being realized, and is clear that GCs may regulate both the transcription and the subsequent processing of miRNAs.

GC-Mediated Transcriptional Regulation of miRNA Expression Numerous miR-NAs have been shown to contain the GRE in their promoters, and transcriptional regulation of these miRNAs can occur via GRE-mediated action (Ko et al. 2013; Kong et al. 2014; Kukreti et al. 2013; Guan et al. 2011). Some miRNAs, such as Let7a, have anti-inflammatory functions and their induction may convey some of the anti-inflammatory GC effects. However, it is also becoming evident that many of the miRNAs transcriptionally regulated by GC may play roles in GR-dependent metabolic effects, which constitute the major limiting factor of GC therapeutic applications. GC-induction of miR-27b in adipocytes and miR-1 in skeletal muscle are mediated by GR-GRE interactions, and results in fat accumulation and skeletal muscle atrophy in the respective tissue. As weight gain and muscle atrophy are main adverse effects of GCs which limit their systemic use in patients, these pathways have important therapeutic implications.

The transcription factors NF κ B and AP-1 are required for transcription of many primary miRNA transcripts involved in inflammation (Cheng et al. 2013; Fujita et al. 2008; Gatto et al. 2008; Indraccolo et al. 2014; Kumar et al. 2014; Madhyastha et al. 2014; Taganov et al. 2006; Yin et al. 2008; Zheng et al. 2014; Zhou et al. 2014) that can be stimulated by infection, toll-like receptor activation, or cytokine signaling (Seddiki et al. 2014). Inflammatory gene transrepression through ligand-activated GR interaction with NF κ B or AP-1 is a major mechanism of GR action, documented also for GC inhibition of important proinflammatory miRNAs such as miR-155 (Zheng et al. 2012). MiR-155 has been identified to play a central role in immunity and inflammation, and controls differentiation of T-cells into Th1, Th2, and Th17 cells (Seddiki et al. 2014).

GC-Mediated Regulation in the Processing of miRNAs Microarray analysis and deep sequencing performed in dexamethasone-treated primary rat thymocytes, a model of



treated primary thymocytes, induced (red) or downregulated (green) versus control. (b) MicroRNA processing enzymes are repressed during GC-induced

Fig. 13.6 (continued) apoptosis of primary thymocytes. Western blot of microRNA processing enzymes Drosha, DGCR8/Pasha, and Dicer in thymocytes unstimulated or treated with dexamethasone for 6 or 12 h. (c) Stable overexpression in Jurkat cells of individual mature ha-miR-17-92 polycistron members (upper panel), found in the study to be downregulated by Dex treatment, blunts glucocorticoid- induced apoptosis. Middle panel, flow cytometric analysis of propidium iodide (PI) uptake in empty vector or hsa-miR-17-92 stable overexpressors showing blunted apoptosis in response to 10 and 100 nM dexamethasone reatment. Lower panel, similar results were obtained in primary thymocytes from transgenic hsa-miR-17-92 mice. Shown is apoptosis monitored by flow cytometric analysis of annexin-FTTC/propidium iodide staining. Modified from Smith et al. (2010), reprinted with permission GC-induced lymphocyte apoptosis, indicated that 79 % of affected miRNAs were *down-regulated* (Fig. 13.6a), which raised the question of whether GCs could regulate proteins involved in miRNA processing (Smith et al. 2010). The repression of these miRNAs resulted to be primarily linked to a GC-dependent inhibition of components of the miRNA microprocessor complex, the RNAse Drosha and the RBP DGCR8/Pasha, and of DICER expression (Fig. 13.6b). Reduction of these proteins results in the accumulation of unprocessed pri-miRNAs in the nucleus and a decrease in mature cytoplasmic miRNAs. Restoring exogenously the expression of the targeted regulatory proteins relieved the block in miRNA processing. Furthermore, overexpression of the ability of GCs to induce apoptosis in mouse thymocytes and in Jurkat T-cells (Fig. 13.6c). These results suggest that GC-mediated down-regulation of miRNA-processing complexes could be a mechanism to induce lymphocyte apoptosis, by reducing the levels of mature miRNAs required to maintain cell survival.

Other RBPs also modulate the processing of miRNAs and may be regulated by GCs (Trabucchi et al. 2009; Bhattacharyya et al. 2013). Pri-miRNA sequences may contain AU-rich elements which can be bound by RBPs in the nucleus (Trabucchi et al. 2009). For instance, the primary transcript of miR-155 contains eight class I AREs. The RBPs TTP and KSRP recognize AREs and can regulate the processing of pri-miRNAs (Bhattacharyya et al. 2013; Trabucchi et al. 2009). KSRP enhances the conversion of pri-miR-155 to pre-miR-155 in airway cells, while TTP inhibits this process (Bhattacharyya et al. 2013; Trabucchi et al. 2009). In the airway, miR-155 promotes the production of IL-8 (Bhattacharyya et al. 2013), an important cytokine in airway inflammation which is repressed by GCs (Nakagome et al. 2012). As TTP is a regulatory target of GCs, it could play a major role in the effects of GCs on miRNA processing.

In some cases, TTP may promote miRNA biogenesis to enhance mature miRNA formation. TTP has been shown to inhibit expression of the RBP Lin28 by destabilizing its transcript (Kim et al. 2012; Lee et al. 2013). As Lin28 inhibits Let7 biogenesis in multiple cancer cells by interfering with the pri- to pre-miRNA processing, TTP is capable of removing this repression (Kim et al. 2012; Lee et al. 2013). In the case of Let7a, GCs exert its effect both transcriptionally and post-transcriptionally. Let7a contains a GRE in the promoter which results in GR-dependent transcriptional up-regulation (Guan et al. 2011), and the TTP-dependent effects on Lin28 promotes maturation of pri-Let7a to pre-Let7a (Kim et al. 2012).

The role of miRNAs in the actions of GCs is beginning to be unraveled. As miR-NAs play crucial roles in biological processes, the actions of these molecules could convey many of the effects of GCs. Deregulation of these pathways could produce a lack of clinical GC-response, and GC-resistance is emerging as a major issue in diseases such as asthma, autoimmunity, and cancer. As such, these pathways may represent mechanisms of disease pathogenesis and serve as potential target for generation of novel therapeutics.

3 Clinical Implications of PTR in GC Function

The immunomodulatory and anti-inflammatory actions of glucocorticoid hormones are used by the potent synthetic GCs used in therapy for a large spectrum of malignancies, autoimmune conditions, allergic and inflammatory diseases. However, their chronic systemic use is compounded by severe systemic side effects and by the occurrence, in a subset of patients, of incomplete clinical efficacy even with high-dose regimens, leading to a GC-dependent status, or to full GC resistance (Adcock and Barnes 2008). The search for anti-inflammatory therapies carrying the efficacy and potency of this class of compounds, but devoid of their significant side effects on metabolism and growth is an important but so far elusive therapeutic goal. Increasing understanding of how PTR pathways coordinately regulate gene expression and growing appreciation of their involvement in the pathophysiology of GC-targeted diseases is indicating that therapeutic interventions exploiting PTR mechanisms and effector molecules can carry significant therapeutic effects. Understanding the PTR mechanisms and mediators conveying GC action, therefore, may provide valuable insight to this endeavor. In support of this approach, overexpression of TTP in a rat model of experimental LPS-mediated periodontitis prevented the local bone loss induced by LPS, significantly decreased inflammatory cell and osteoclast recruitment to the inflammatory site, and decreased levels of the TTP targets TNFa, IL-6 and COX2 compared to the control animals (Patil et al. 2008). This study highlights the potential value of TTPmediated action—and at large, of targeting PTR mechanisms—as a self-standing anti-inflammatory strategy. Closer to clinical setting, development of small molecule inhibitors of signaling pathways regulating RBP function, as in the case of p38 and MK2/3 (Gaestel 2013), or generation of miRNA-based technology to attack specific pathologic processes (Krutzfeldt et al. 2005; Lu and Rothenberg 2013; Mattes et al. 2008) are examples of how GC-driven/GC-sensitive PTR mechanisms, once identified, can independently provide additional or alternative strategies for controlling overexpressed or aberrant immune responses. In addition to studies on the mechanisms of PTR-mediated GC action, it is pressing to better identify altered PTR regulation in clinical settings where GCs are administered and uncover mutations that could affect regulatory binding sequences, or levels and functionality of RBPs and miRNAs, which could impair PTR-mediated GC response. To this end, multiple single nucleotide polymorphisms have been identified, for example, in the human TTP gene (Blackshear et al. 2003b), some of which could be associated with rheumatoid arthritis and autoimmune diseases (Carrick et al. 2006).

4 Summary and Conclusions

The studies discussed in this review indicate that the impact of post-transcriptional regulation in the mechanism of anti-inflammatory action of GCs is larger than previously appreciated (summarized in Fig. 13.7), and yet ill-defined. They provide



Fig. 13.7 Mechanisms of Post-transcriptional control of inflammatory response by GC, as discussed in this chapter. Heterogeneous sequence elements for regulation (indicated as yellow boxes, ARE and G/C rich elements discussed in the text) present in the coding or-more frequently-in untranslated regions (UTR) mediate binding of cognate regulatory factors such as RBPs and miRNA. Inflammatory stimuli trigger kinase-dependent binding, association and activation of factors inducing mRNA stabilization and/or increased protein translation (in red boxes and arrows). In balance with this action, binding of multimeric regulatory complexes functionally supporting mRNA decay and suppressing, or lowering, translation rates are able to limit the amplitude and the duration of the inflammatory response—and regulate as well appropriate baseline levels in homeostatic conditions. These mechanisms are regulated by GC at multiple levels (in *blue boxes* and arrows), supporting—either directly or indirectly—processes inducing a net decrease in mRNA stability and/or translation rates as means of downregulating overexpressed inflammatory genes. GC-induced MAPK phosphatase (MKP-1) indirectly lead to a PTR-mediated anti-inflammatory outcome by dephosphorylation of MAPK members, which chiefly support increased mRNA stability and translation of inflammatory genes. A more direct influence by GC on PTR is through the induction of TTP, an ARE-binding protein conveying accelerated mRNA decay and translational suppression of immune-related genes. Influence of GC on PTR relies heavily on changes in miRNA expression, through transcriptional regulation and regulation of the miRNA processing enzymes. Lastly, upon GC stimulation the GR has been described in cellular models to associate with mRNAs in the cytoplasm and mediate acceleration of mRNA decay through specific interactions with a G/C-rich sequence present in the 5'UTR

compelling arguments towards testing a paradigm of anti-inflammatory approach based on targeting of post-transcriptional mechanisms. Regulatory factors of PTR dynamically assembling in specific, context-driven RNP complexes are increasingly scrutinized as therapeutically targetable. Considering that these molecular species coordinately regulate large subsets of functionally related transcripts through common regulatory elements, a better understanding of GC ability of intervening at the structural and functional interface between RBPs, miRNAs and their targeted transcripts may uncover anti-inflammatory and immunoregulatory outcomes potentially not subjected to the limitations of GC systemic side effects and incomplete therapeutic response due to GC resistance.

References

- Abraham SM, Clark AR (2006) Dual-specificity phosphatase 1: a critical regulator of innate immune responses. Biochem Soc Trans 34:1018–1023
- Adcock IM, Barnes PJ (2008) Molecular mechanisms of corticosteroid resistance. Chest 134:394–401
- Amano Y, Lee SW, Allison AC (1993) Inhibition by glucocorticoids of the formation of interleukin-1 alpha, interleukin-1 beta, and interleukin-6: mediation by decreased mRNA stability. Mol Pharmacol 43:176–182
- Anderson P (2008) Post-transcriptional control of cytokine production. Nat Immunol 9:353-359
- Anderson P (2010) Post-transcriptional regulons coordinate the initiation and resolution of inflammation. Nat Rev Immunol 10:24–35
- Anderson P, Kedersha N (2009) RNA granules: post-transcriptional and epigenetic modulators of gene expression. Nat Rev Mol Cell Biol 10:430–436
- Anderson P, Phillips K, Stoecklin G, Kedersha N (2004) Post-transcriptional regulation of proinflammatory proteins. J Leukoc Biol 76:42–47. doi:10.1189/jlb.1103536
- Bakheet T, Williams BR, Khabar KS (2006) ARED 3.0: the large and diverse AU-rich transcriptome. Nucleic Acids Res 34:D111–D114
- Baltimore D, Boldin MP, O'Connell RM, Rao DS, Taganov KD (2008) MicroRNAs: new regulators of immune cell development and function. Nat Immunol 9:839–845
- Barnes PJ, Adcock IM (2009) Glucocorticoid resistance in inflammatory diseases. Lancet 373:1905–1917
- Bhattacharyya SN, Habermacher R, Martine U, Closs EI, Filipowicz W (2006) Relief of microRNA-mediated translational repression in human cells subjected to stress. Cell 125: 1111–1124
- Bhattacharyya S, Kumar P, Tsuchiya M, Bhattacharyya A, Biswas R (2013) Regulation of miR-155 biogenesis in cystic fibrosis lung epithelial cells: antagonistic role of two mRNAdestabilizing proteins, KSRP and TTP. Biochem Biophys Res Commun 433:484–488
- Blackshear PJ (2002) Tristetraprolin and other CCCH tandem zinc-finger proteins in the regulation of mRNA turnover. Biochem Soc Trans 30:945–952
- Blackshear PJ, Lai WS, Kennington EA, Brewer G, Wilson GM, Guan X, Zhou P (2003a) Characteristics of the interaction of a synthetic human tristetraprolin tandem zinc finger peptide with AU-rich element-containing RNA substrates. J Biol Chem 278:19947–19955
- Blackshear PJ, Phillips RS, Vazquez-Matias J, Mohrenweiser H (2003b) Polymorphisms in the genes encoding members of the tristetraprolin family of human tandem CCCH zinc finger proteins. Prog Nucleic Acid Res Mol Biol 75:43–68
- Brennan CM, Steitz JA (2001) HuR and mRNA stability. Cell Mol Life Sci 58:266-277
- Brewer BY, Malicka J, Blackshear PJ, Wilson GM (2004) RNA sequence elements required for high affinity binding by the zinc finger domain of tristetraprolin: conformational changes coupled to the bipartite nature of AU-RICH mRNA-destabilizing motifs. J Biol Chem 279:27870–27877
- Cao H (2004) Expression, purification, and biochemical characterization of the antiinflammatory tristetraprolin: a zinc-dependent mRNA binding protein affected by posttranslational modifications. Biochemistry 43:13724–13738

- Carballo E, Lai WS, Blackshear PJ (1998) Feedback inhibition of macrophage tumor necrosis factor-alpha production by tristetraprolin. Science 281:1001–1005
- Carballo E, Lai WS, Blackshear PJ (2000) Evidence that tristetraprolin is a physiological regulator of granulocyte-macrophage colony-stimulating factor messenger RNA deadenylation and stability. Blood 95:1891–1899
- Carballo E, Cao H, Lai WS, Kennington EA, Campbell D, Blackshear PJ (2001) Decreased sensitivity of tristetraprolin-deficient cells to p38 inhibitors suggests the involvement of tristetraprolin in the p38 signaling pathway. J Biol Chem 276:42580–42587
- Carrick DM, Lai WS, Blackshear PJ (2004) The tandem CCCH zinc finger protein tristetraprolin and its relevance to cytokine mRNA turnover and arthritis. Arthritis Res Ther 6:248–264
- Carrick DM, Chulada P, Donn R, Fabris M, Mcnicholl J, Whitworth W, Blackshear PJ (2006) Genetic variations in ZFP36 and their possible relationship to autoimmune diseases. J Autoimmun 26:182–196
- Chafin CB, Regna NL, Hammond SE, Reilly CM (2013) Cellular and urinary microRNA alterations in NZB/W mice with hydroxychloroquine or prednisone treatment. Int Immunopharmacol 17:894–906
- Chen CY, Shyu AB (1995) AU-rich elements: characterization and importance in mRNA degradation. Trends Biochem Sci 20:465–470
- Chen CY, Gherzi R, Andersen JS, Gaietta G, Jurchott K, Royer HD, Mann M, Karin M (2000) Nucleolin and YB-1 are required for JNK-mediated interleukin-2 mRNA stabilization during T-cell activation. Genes Dev 14:1236–1248
- Chen C-Z, Schaffert S, Fragoso R, Loh C (2013) Regulation of immune responses and tolerance: the microRNA perspective. Immunol Rev 253:112–128
- Cheng C, Li W, Zhang Z, Yoshimura S, Hao Q, Zhang C, Wang Z (2013) MicroRNA-144 is regulated by activator protein-1 (AP-1) and decreases expression of Alzheimer disease-related a disintegrin and metalloprotease 10 (ADAM10). J Biol Chem 288:13748–13761
- Cho SJ, Kang MJ, Homer RJ, Kang HR, Zhang X, Lee PJ, Elias JA, Lee CG (2006) Role of early growth response-1 (Egr-1) in interleukin-13-induced inflammation and remodeling. J Biol Chem 281:8161–8168. doi:10.1074/jbc.M506770200
- Chrestensen CA, Schroeder MJ, Shabanowitz J, Hunt DF, Pelo JW, Worthington MT, Sturgill TW (2004) MAPKAP kinase 2 phosphorylates tristetraprolin on in vivo sites including Ser178, a site required for 14-3-3 binding. J Biol Chem 279:10176–10184
- Ciafre SA, Galardi S (2013) microRNAs and RNA-binding proteins: a complex network of interactions and reciprocal regulations in cancer. RNA Biol 10:935–942
- Clark AR, Lasa M (2003) Crosstalk between glucocorticoids and mitogen-activated protein kinase signalling pathways. Curr Opin Pharmacol 3:404–411
- Clark AR, Dean JL, Saklatvala J (2003) Post-transcriptional regulation of gene expression by mitogen-activated protein kinase p38. FEBS Lett 546:37–44
- Dauletbaev N, Eklove D, Mawji N, Iskandar M, DI Marco S, Gallouzi IE, Lands LC (2011) Downregulation of cytokine-induced interleukin-8 requires inhibition of p38 mitogen-activated protein kinase (MAPK) via MAPK phosphatase 1-dependent and -independent mechanisms. J Biol Chem 286:15998–16007
- Davis TE, Kis-Toth K, Szanto A, Tsokos GC (2013) Glucocorticoids suppress T cell function by up-regulating microRNA-98. Arthritis Rheum 65:1882–1890
- Dean JL, Sully G, Clark AR, Saklatvala J (2004) The involvement of AU-rich element-binding proteins in p38 mitogen-activated protein kinase pathway-mediated mRNA stabilisation. Cell Signal 16:1113–1121
- Dhawan L, Liu B, Blaxall BC, Taubman MB (2007) A novel role for the glucocorticoid receptor in the regulation of monocyte chemoattractant protein-1 mRNA stability. J Biol Chem 282:10146–10152. doi:10.1074/jbc.M605925200
- Dreyfuss G, Kim VN, Kataoka N (2002) Messenger-RNA-binding proteins and the messages they carry. Nat Rev Mol Cell Biol 3:195–205
- Ehretsmann CP, Chandler LA, Bourgeois S (1995) A nuclear post-transcriptional mechanism mediates the induction of fibronectin by glucocorticoids. Mol Cell Endocrinol 110:185–194

- Fan J, Yang X, Wang W, Wood WH 3rd, Becker KG, Gorospe M (2002) Global analysis of stressregulated mRNA turnover by using cDNA arrays. Proc Natl Acad Sci U S A 99:10611–10616
- Fan J, Ishmael FT, Fang X, Myers A, Cheadle C, Huang SK, Atasoy U, Gorospe M, Stellato C (2011) Chemokine transcripts as targets of the RNA-binding protein HuR in human airway epithelium. J Immunol 186:2482–2494
- Fechir M, Linker K, Pautz A, Hubrich T, Forstermann U, Rodriguez-Pascual F, Kleinert H (2005) Tristetraprolin regulates the expression of the human inducible nitric-oxide synthase gene. Mol Pharmacol 67:2148–2161
- Foster PS, Plank M, Collison A, Tay HL, Kaiko GE, Li J, Johnston SL, Hansbro PM, Kumar RK, Yang M, Mattes J (2013) The emerging role of microRNAs in regulating immune and inflammatory responses in the lung. Immunol Rev 253:198–215
- Frevel MA, Bakheet T, Silva AM, Hissong JG, Khabar KS, Williams BR (2003) p38 Mitogenactivated protein kinase-dependent and -independent signaling of mRNA stability of AU-rich element-containing transcripts. Mol Cell Biol 23:425–436
- Fujita S, Ito T, Mizutani T, Minoguchi S, Yamamichi N, Sakurai K, Iba H (2008) miR-21 Gene expression triggered by AP-1 is sustained through a double-negative feedback mechanism. J Mol Biol 378:492–504
- Gaestel M (2006) MAPKAP kinases -MKs- two's company, three's a crowd. Nat Rev Mol Cell Biol 7:120–130
- Gaestel M (2013) What goes up must come down: molecular basis of MAPKAP kinase 2/3-dependent regulation of the inflammatory response and its inhibition. Biol Chem 394:1301–1315
- Gao J, Wagnon JL, Protacio RM, Glazko GV, Beggs M, Raj V, Davidson MK, Wahls WP (2013) A stress-activated, p38 mitogen-activated protein kinase-ATF/CREB pathway regulates posttranscriptional, sequence-dependent decay of target RNAs. Mol Cell Biol 33:3026–3035
- Garcia-Gras EA, Chi P, Thompson EA (2000) Glucocorticoid-mediated destabilization of cyclin D3 mRNA involves RNA-protein interactions in the 3'-untranslated region of the mRNA. J Biol Chem 275:22001–22008
- Gatto G, Rossi A, Rossi D, Kroening S, Bonatti S, Mallardo M (2008) Epstein-Barr virus latent membrane protein 1 trans-activates miR-155 transcription through the NF-kappaB pathway. Nucleic Acids Res 36:6608–6619
- Gerstberger S, Hafner M, Tuschl T (2014) A census of human RNA-binding proteins. Nat Rev Genet 15:829–845
- Gille J, Reisinger K, Westphal-Varghese B, Kaufmann R (2001) Decreased mRNA stability as a mechanism of glucocorticoid-mediated inhibition of vascular endothelial growth factor gene expression by cultured keratinocytes. J Invest Dermatol 117:1581–1587
- Ginisty H, Sicard H, Roger B, Bouvet P (1999) Structure and functions of nucleolin. J Cell Sci 112(Pt 6):761–772
- Grillo G, Turi A, Licciulli F, Mignone F, Liuni S, Banfi S, Gennarino VA, Horner DS, Pavesi G, Picardi E, Pesole G (2010) UTRdb and UTRsite (RELEASE 2010): a collection of sequences and regulatory motifs of the untranslated regions of eukaryotic mRNAs. Nucleic Acids Res 38:D75–D80
- Gronemeyer H, Gustafsson JA, Laudet V (2004) Principles for modulation of the nuclear receptor superfamily. Nat Rev Drug Discov 3:950–964
- Grosset C, Taupin JL, Lemercier C, Moreau JF, Reiffers J, Ripoche J (1999) Leukaemia inhibitory factor expression is inhibited by glucocorticoids through post-transcriptional mechanisms. Cytokine 11:29–36
- Gruber AR, Fallmann J, Kratochvill F, Kovarik P, Hofacker IL (2011) AREsite: a database for the comprehensive investigation of AU-rich elements. Nucleic Acids Res 39:D66–D69
- Guan H, Zhang P, Liu C, Zhang J, Huang Z, Chen W, Chen Z, Ni N, Liu Q, Jiang A (2011) Characterization and functional analysis of the human microRNA let-7a2 promoter in lung cancer A549 cell lines. Mol Biol Rep 38:5327–5334
- Halees AS, Hitti E, Al-Saif M, Mahmoud L, Vlasova-St Louis IA, Beisang DJ, Bohjanen PR, Khabar K (2011) Global assessment of GU-rich regulatory content and function in the human transcriptome. RNA Biol 8:681–691

- Hamilton T, Novotny M, Pavicic PJ Jr, Herjan T, Hartupee J, Sun D, Zhao C, Datta S (2010) Diversity in post-transcriptional control of neutrophil chemoattractant cytokine gene expression. Cytokine 52:116–122
- Harada M, Pokrovskaja-Tamm K, Soderhall S, Heyman M, Grander D, Corcoran M (2012) Involvement of miR17 pathway in glucocorticoid-induced cell death in pediatric acute lymphoblastic leukemia. Leuk Lymphoma 53:2041–2050
- Helou EF, Simonson J, Arora AS (2008) 3-yr-follow-up of topical corticosteroid treatment for eosinophilic esophagitis in adults. Am J Gastroenterol 103:2194–2199
- Hollams EM, Giles KM, Thomson AM, Leedman PJ (2002) MRNA stability and the control of gene expression: implications for human disease. Neurochem Res 27:957–980
- Hong Y, Wu J, Zhao J, Wang H, Liu Y, Chen T, Kan X, Tao Q, Shen X, Yan K, Zhai Z (2013) miR-29b and miR-29c are involved in Toll-like receptor control of glucocorticoid-induced apoptosis in human plasmacytoid dendritic cells. PLoS One 8:e69926
- Huang HW, Bi W, Jenkins GN, Alcorn JL (2008) Glucocorticoid regulation of human pulmonary surfactant protein-B mRNA stability involves the 3'-untranslated region. Am J Respir Cell Mol Biol 38:473–482
- Huang HW, Payne DE, Bi W, Pan S, Bruce SR, Alcorn JL (2012) Sequences of a hairpin structure in the 3'-untranslated region mediate regulation of human pulmonary surfactant protein B mRNA stability. Am J Physiol Lung Cell Mol Physiol 302:L1107–L1117
- Imasato A, Desbois-Mouthon C, Han J, Kai H, Cato AC, Akira S, Li JD (2002) Inhibition of p38 MAPK by glucocorticoids via induction of MAPK phosphatase-1 enhances nontypeable Haemophilus influenzae-induced expression of toll-like receptor 2. J Biol Chem 277: 47444–47450
- Indraccolo S, Amadori A, Ferretti E, Gulino A, Vacca A, Screpanti I, Ma J, Liu J, Wang Z, Gu X, Fan Y, Zhang W, Xu L, Zhang J, Cai D (2014) NF-kappaB-dependent microRNA-425 upregulation promotes gastric cancer cell growth by targeting PTEN upon IL-1beta induction. Leukemia 13:40
- Ingram JL, Antao-Menezes A, Mangum JB, Lyght O, Lee PJ, Elias JA, Bonner JC (2006) Opposing actions of Stat1 and Stat6 on IL-13-induced up-regulation of early growth response-1 and platelet-derived growth factor ligands in pulmonary fibroblasts. J Immunol 177:4141–4148
- Ishmael FT, Fang X, Heller NH, Fan J, Atasoy U, Cheadle C, Stellato C (2007) Role of the RNAbinding protein tristetraprolin (TTP) in glucocorticoid (GC)-mediated gene regulation. J Allergy Clin Immunol 119:S339
- Ishmael FT, Fang X, Galdiero MR, Atasoy U, Rigby WF, Gorospe M, Cheadle C, Stellato C (2008) Role of the RNA-binding protein tristetraprolin in glucocorticoid-mediated gene regulation. J Immunol 180:8342–8353
- Ishmael FT, Fang X, Houser KR, Pearce K, Abdelmohsen K, Zhan M, Gorospe M, Stellato C (2010) The human glucocorticoid receptor as an RNA-binding protein: global analysis of glucocorticoid receptor-associated transcripts and identification of a target RNA motif. J Immunol
- Ivanov P, Anderson P (2013) Post-transcriptional regulatory networks in immunity. Immunol Rev 253:253–272
- Jalonen U, Lahti A, Korhonen R, Kankaanranta H, Moilanen E (2005) Inhibition of tristetraprolin expression by dexamethasone in activated macrophages. Biochem Pharmacol 69:733–740
- Jalonen U, Nieminen R, Vuolteenaho K, Kankaanranta H, Moilanen E (2006) Down-regulation of tristetraprolin expression results in enhanced IL-12 and MIP-2 production and reduced MIP-3alpha synthesis in activated macrophages. Mediat Inflamm 2006:40691
- Jiang P, Coller H (2012) Functional interactions between microRNAs and RNA binding proteins. Microrna 1:70–79
- Jing Q, Huang S, Guth S, Zarubin T, Motoyama A, Chen J, DI Padova F, Lin S-C, Gram H, Han J (2005) Involvement of MicroRNA in AU-rich element-mediated mRNA instability. Cell 120:623–634
- Johnson BA, Stehn JR, Yaffe MB, Blackwell TK (2002) Cytoplasmic localization of tristetraprolin involves 14-3-3-dependent and -independent mechanisms. J Biol Chem 277:18029–18036

- Keene JD (2007) RNA regulons: coordination of post-transcriptional events. Nat Rev Genet 8:533–543
- Keene JD, Komisarow JM, Friedersdorf MB (2006) RIP-chip: the isolation and identification of mRNAs, microRNAs and protein components of ribonucleoprotein complexes from cell extracts. Nat Protoc 1:302–307
- Kim HS, Kuwano Y, Zhan M, Pullmann R Jr, Mazan-Mamczarz K, Li H, Kedersha N, Anderson P, Wilce MC, Gorospe M, Wilce JA (2007) Elucidation of a C-rich signature motif in target mRNAs of RNA-binding protein TIAR. Mol Cell Biol 27:6806–6817
- Kim CW, Vo MT, Kim HK, Lee HH, Yoon NA, Lee BJ, Min YJ, Joo WD, Cha HJ, Park JW, Cho WJ (2012) Ectopic over-expression of tristetraprolin in human cancer cells promotes biogenesis of let-7 by down-regulation of Lin28. Nucleic Acids Res 40:3856–3869
- Ko JY, Chuang PC, Chen MW, Ke HC, Wu SL, Chang YH, Chen YS, Wang FS (2013) MicroRNA-29a ameliorates glucocorticoid-induced suppression of osteoblast differentiation by regulating beta-catenin acetylation. Bone 57:468–475
- Kong X, Yu J, Bi J, Qi H, Di W, Wu L, Wang L, Zha J, Lv S, Zhang F, Li Y, Hu F, Liu F, Zhou H, Liu J, Ding G (2014) Glucocorticoids transcriptionally regulate miR-27b expression promoting body fat accumulation via suppressing the browning of white adipose tissue. Diabetes 64:393–404
- Kontoyiannis D, Pasparakis M, Pizarro TT, Cominelli F, Kollias G (1999) Impaired on/off regulation of TNF biosynthesis in mice lacking TNF AU-rich elements: implications for joint and gut-associated immunopathologies. Immunity 10:387–398
- Korhonen R, Lahti A, Hamalainen M, Kankaanranta H, Moilanen E (2002) Dexamethasone inhibits inducible nitric-oxide synthase expression and nitric oxide production by destabilizing mRNA in lipopolysaccharide-treated macrophages. Mol Pharmacol 62:698–704
- Kracht M, Saklatvala J (2002) Transcriptional and post-transcriptional control of gene expression in inflammation. Cytokine 20:91–106
- Krutzfeldt J, Rajewsky N, Braich R, Rajeev KG, Tuschl T, Manoharan M, Stoffel M (2005) Silencing of microRNAs in vivo with 'antagomirs'. Nature 438:685–689
- Kukreti H, Amuthavalli K, Harikumar A, Sathiyamoorthy S, Feng PZ, Anantharaj R, Tan SL, Lokireddy S, Bonala S, Sriram S, McFarlane C, Kambadur R, Sharma M (2013) Musclespecific microRNA1 (miR1) targets heat shock protein 70 (HSP70) during dexamethasonemediated atrophy. J Biol Chem 288:6663–6678
- Kumar M, Ahmad T, Sharma A, Mabalirajan U, Kulshreshtha A, Agrawal A, Ghosh B (2011) Let-7 microRNA-mediated regulation of IL-13 and allergic airway inflammation. J Allergy Clin Immunol 128:1077–1085.e1-10
- Kumar V, Palermo R, Talora C, Campese AF, Checquolo S, Bellavia D, Tottone L, Testa G, Miele E (2014) Notch and NF-kB signaling pathways regulate miR-223/FBXW7 axis in T-cell acute lymphoblastic leukemia. Leukemia 28:2324–2335
- Kuwano Y, Pullmann R Jr, Marasa BS, Abdelmohsen K, Lee EK, Yang X, Martindale JL, Zhan M, Gorospe M (2010) NF90 selectively represses the translation of target mRNAs bearing an AU-rich signature motif. Nucleic Acids Res 38:225–238
- Lai WS, Parker JS, Grissom SF, Stumpo DJ, Blackshear PJ (2006) Novel mRNA targets for tristetraprolin (TTP) identified by global analysis of stabilized transcripts in TTP-deficient fibroblasts. Mol Cell Biol 26:9196–9208
- Lasa M, Mahtani KR, Finch A, Brewer G, Saklatvala J, Clark AR (2000) Regulation of cyclooxygenase 2 mRNA stability by the mitogen-activated protein kinase p38 signaling cascade. Mol Cell Biol 20:4265–4274
- Lasa M, Brook M, Saklatvala J, Clark AR (2001) Dexamethasone destabilizes cyclooxygenase 2 mRNA by inhibiting mitogen-activated protein kinase p38. Mol Cell Biol 21:771–780
- Lasa M, Abraham SM, Boucheron C, Saklatvala J, Clark AR (2002) Dexamethasone causes sustained expression of mitogen-activated protein kinase (MAPK) phosphatase 1 and phosphatasemediated inhibition of MAPK p38. Mol Cell Biol 22:7802–7811
- Lee JY, Kim HJ, Yoon NA, Lee WH, Min YJ, Ko BK, Lee BJ, Lee A, Cha HJ, Cho WJ, Park JW (2013) Tumor suppressor p53 plays a key role in induction of both tristetraprolin and let-7 in human cancer cells. Nucleic Acids Res 41:5614–5625

- Leppek K, Schott J, Reitter S, Poetz F, Hammond MC, Stoecklin G (2013) Roquin promotes constitutive mRNA decay via a conserved class of stem-loop recognition motifs. Cell 153:869–881
- Liang J, Lei T, Song Y, Yanes N, Qi Y, Fu M (2009) RNA-destabilizing factor tristetraprolin negatively regulates NF-kappaB signaling. J Biol Chem 284:29383–29390
- Liu Y, Shepherd EG, Nelin LD (2007) MAPK phosphatases—regulating the immune response. Nat Rev Immunol 7:202–212
- Lopez De Silanes I, Zhan M, Lal A, Yang X, Gorospe M (2004) Identification of a target RNA motif for RNA-binding protein HuR. Proc Natl Acad Sci U S A 101:2987–2992
- Lu TX, Rothenberg ME (2013) Diagnostic, functional, and therapeutic roles of microRNA in allergic diseases. J Allergy Clin Immunol 132:3–13, quiz 14
- Lu JY, Schneider RJ (2004) Tissue distribution of AU-rich mRNA-binding proteins involved in regulation of mRNA decay. J Biol Chem 279:12974–12979
- Lu TX, Sherrill JD, Wen T, Plassard AJ, Besse JA, Abonia JP, Franciosi JP, Putnam PE, Eby M, Martin LJ, Aronow BJ, Rothenberg ME (2012) MicroRNA signature in patients with eosinophilic esophagitis, reversibility with glucocorticoids, and assessment as disease biomarkers. J Allergy Clin Immunol 129:1064–1075.e9
- Lyabin DN, Eliseeva IA, Ovchinnikov LP (2014) YB-1 protein: functions and regulation. Wiley Interdiscip Rev RNA 5:95–110
- Madhyastha R, Madhyastha H, Pengjam Y, Nakajima Y, Omura S, Maruyama M (2014) NFkappaB activation is essential for miR-21 induction by TGFbeta1 in high glucose conditions. Biochem Biophys Res Commun 451:615–621
- Mahtani KR, Brook M, Dean JL, Sully G, Saklatvala J, Clark AR (2001) Mitogen-activated protein kinase p38 controls the expression and posttranslational modification of tristetraprolin, a regulator of tumor necrosis factor alpha mRNA stability. Mol Cell Biol 21:6461–6469
- Mattes J, Collison A, Foster PS (2008) Emerging role of microRNAs in disease pathogenesis and strategies for therapeutic modulation. Curr Opin Mol Ther 10:150–157
- Meisner NC, Filipowicz W (2011) Properties of the regulatory RNA-binding protein HuR and its role in controlling miRNA repression. Adv Exp Med Biol 700:106–123
- Moens U, Kostenko S, Sveinbjornsson B (2013) The role of mitogen-activated protein kinaseactivated protein kinases (MAPKAPKs) in inflammation. Genes 4:101–133
- Mongelard F, Bouvet P (2007) Nucleolin: a multiFACeTed protein. Trends Cell Biol 17:80-86
- Moschos SA, Williams AE, Perry MM, Birrell MA, Belvisi MG, Lindsay MA (2007) Expression profiling in vivo demonstrates rapid changes in lung microRNA levels following lipopolysaccharide-induced inflammation but not in the anti-inflammatory action of glucocorticoids. BMC Genomics 8:240
- Mozo L, Gayo A, Suarez A, Rivas D, Zamorano J, Gutierrez C (1998) Glucocorticoids inhibit IL-4 and mitogen-induced IL-4R alpha chain expression by different posttranscriptional mechanisms. J Allergy Clin Immunol 102:968–976
- Nakagome K, Matsushita S, Nagata M (2012) Neutrophilic inflammation in severe asthma. Int Arch Allergy Immunol 158(Suppl 1):96–102
- Newton R, Seybold J, Kuitert LM, Bergmann M, Barnes PJ (1998) Repression of cyclooxygenase-2 and prostaglandin E2 release by dexamethasone occurs by transcriptional and posttranscriptional mechanisms involving loss of polyadenylated mRNA. J Biol Chem 273:32312–32321
- Oakley RH, Cidlowski JA (2013) The biology of the glucocorticoid receptor: new signaling mechanisms in health and disease. J Allergy Clin Immunol 132:1033–1044
- Ogilvie RL, Abelson M, Hau HH, Vlasova I, Blackshear PJ, Bohjanen PR (2005) Tristetraprolin down-regulates IL-2 gene expression through AU-rich element-mediated mRNA decay. J Immunol 174:953–961
- Panganiban RP, Vonakis BM, Ishmael FT, Stellato C (2014) Coordinated post-transcriptional regulation of the chemokine system: messages from CCL2. J Interferon Cytokine Res 34:255–266

- Patil CS, Liu M, Zhao W, Coatney DD, Li F, Vantubergen EA, D'silva NJ, Kirkwood KL (2008) Targeting mRNA stability arrests inflammatory bone loss. Mol Ther 16:1657–1664
- Pawliczak R, Logun C, Madara P, Barb J, Suffredini AF, Munson PJ, Danner RL, Shelhamer JH (2005) Influence of IFN-gamma on gene expression in normal human bronchial epithelial cells: modulation of IFN-gamma effects by dexamethasone. Physiol Genomics 23:28–45
- Peppel K, Vinci JM, Baglioni C (1991) The AU-rich sequences in the 3' untranslated region mediate the increased turnover of interferon mRNA induced by glucocorticoids. J Exp Med 173:349–355
- Poon M, Liu B, Taubman MB (1999) Identification of a novel dexamethasone-sensitive RNAdestabilizing region on rat monocyte chemoattractant protein 1 mRNA. Mol Cell Biol 19:6471–6478
- Rainer J, Ploner C, Jesacher S, Ploner A, Eduardoff M, Mansha M, Wasim M, Panzer-Grumayer R, Trajanoski Z, Niederegger H, Kofler R (2009) Glucocorticoid-regulated microRNAs and mirtrons in acute lymphoblastic leukemia. Leukemia 23:746–752
- Raineri I, Wegmueller D, Gross B, Certa U, Moroni C (2004) Roles of AUF1 isoforms, HuR and BRF1 in ARE-dependent mRNA turnover studied by RNA interference. Nucleic Acids Res 32:1279–1288
- Rigby WF, Roy K, Collins J, Rigby S, Connolly JE, Bloch DB, Brooks SA (2005) Structure/function analysis of tristetraprolin (TTP): p38 stress-activated protein kinase and lipopolysaccharide stimulation do not alter TTP function. J Immunol 174:7883–7893
- Ross J (1995) mRNA stability in mammalian cells. Microbiol Rev 59:423-450
- Sadri N, Schneider RJ (2009) Auf1/Hnrnpd-deficient mice develop pruritic inflammatory skin disease. J Invest Dermatol 129:657–670
- Sawaoka H, Dixon DA, Oates JA, Boutaud O (2003) Tristetraprolin binds to the 3'-untranslated region of cyclooxygenase-2 mRNA. A polyadenylation variant in a cancer cell line lacks the binding site. J Biol Chem 278:13928–13935
- Schleimer RP (2004) Glucocorticoids suppress inflammation but spare innate immune responses in airway epithelium. Proc Am Thorac Soc 1:222–230
- Seddiki N, Brezar V, Ruffin N, Levy Y, Swaminathan S (2014) Role of miR-155 in the regulation of lymphocyte immune function and disease. Immunology 142:32–38
- Shaw G, Kamen R (1986) A conserved AU sequence from the 3' untranslated region of GM-CSF mRNA mediates selective mRNA degradation. Cell 46:659–667
- Shen H, Liu T, Fu L, Zhao S, Fan B, Cao J, Li X (2013) Identification of microRNAs involved in dexamethasone-induced muscle atrophy. Mol Cell Biochem 381:105–113
- Shi L, Zhao G, Qiu D, Godfrey WR, Vogel H, Rando TA, Hu H, Kao PN (2005) NF90 regulates cell cycle exit and terminal myogenic differentiation by direct binding to the 3'-untranslated region of MyoD and p21WAF1/CIP1 mRNAs. J Biol Chem 280:18981–18989
- Shi L, Godfrey WR, Lin J, Zhao G, Kao PN (2007) NF90 regulates inducible IL-2 gene expression in T cells. J Exp Med 204:971–977
- Shyu AB, Wilkinson MF (2000) The double lives of shuttling mRNA binding proteins. Cell 102:135–138
- Simone LE, Keene JD (2013) Mechanisms coordinating ELAV/Hu mRNA regulons. Curr Opin Genet Dev 23:35–43
- Singh K, Laughlin J, Kosinski PA, Covey LR (2004) Nucleolin is a second component of the CD154 mRNA stability complex that regulates mRNA turnover in activated T cells. J Immunol 173:976–985
- Smith LK, Shah RR, Cidlowski JA (2010) Glucocorticoids modulate microRNA expression and processing during lymphocyte apoptosis. J Biol Chem 285:36698–36708
- Smoak K, Cidlowski JA (2006) Glucocorticoids regulate tristetraprolin synthesis and posttranscriptionally regulate tumor necrosis factor alpha inflammatory signaling. Mol Cell Biol 26:9126–9135
- Solberg OD, Ostrin EJ, Love MI, Peng JC, Bhakta NR, Hou L, Nguyen C, Solon M, Barczak AJ, Zlock LT, Blagev DP, Finkbeiner WE, Ansel KM, Arron JR, Erle DJ, Woodruff PG (2012)

Airway epithelial miRNA expression is altered in asthma. Am J Respir Crit Care Med 186:965–974

- Srikantan S, Tominaga K, Gorospe M (2012) Functional interplay between RNA-binding protein HuR and microRNAs. Curr Protein Pept Sci 13:372–379
- Steitz JA, Vasudevan S (2009) miRNPs: versatile regulators of gene expression in vertebrate cells. Biochem Soc Trans 37:931–935
- Stellato C (2004) Post-transcriptional and nongenomic effects of glucocorticoids. Proc Am Thorac Soc 1:255–263
- Stellato C (2007) Glucocorticoid actions on airway epithelial responses in immunity: functional outcomes and molecular targets. J Allergy Clin Immunol 120:1247–1263, quiz 1264–5
- Stellato C, Schleimer RP (2000) Regulation of chemokines by glucocorticoids. In: Rothenberg ME (ed) Chemokines in allergic disease. Marcel Dekker, New York
- Stoecklin G, Anderson P (2006a) Posttranscriptional mechanisms regulating the inflammatory response. In: Alt FW, Austen KF, Honjo T, Melchers F, Uhr JW, Unanue ER (eds) Advances in immunology, vol 89. Academic, San Diego, CA
- Stoecklin G, Anderson P (2006b) Posttranscriptional mechanisms regulating the inflammatory response. Adv Immunol 89:1–37
- Stoecklin G, Gross B, Ming XF, Moroni C (2003) A novel mechanism of tumor suppression by destabilizing AU-rich growth factor mRNA. Oncogene 22:3554–3561
- Stoecklin G, Stubbs T, Kedersha N, Wax S, Rigby WF, Blackwell TK, Anderson P (2004) MK2induced tristetraprolin:14-3-3 complexes prevent stress granule association and ARE-mRNA decay. EMBO J 23:1313–1324
- Stoecklin G, Tenenbaum SA, Mayo T, Chittur SV, George AD, Baroni TE, Blackshear PJ, Anderson P (2008) Genome-wide analysis identifies interleukin-10 mRNA as target of tristetraprolin. J Biol Chem 283:11689–11699, M709657200
- Sugiura R, Satoh R, Ishiwata S, Umeda N, Kita A (2011) Role of RNA-binding proteins in MAPK signal transduction pathway. J Signal Transduct 2011:109746
- Swantek JL, Cobb MH, Geppert TD (1997) Jun N-terminal kinase/stress-activated protein kinase (JNK/SAPK) is required for lipopolysaccharide stimulation of tumor necrosis factor alpha (TNF-alpha) translation: glucocorticoids inhibit TNF-alpha translation by blocking JNK/ SAPK. Mol Cell Biol 17:6274–6282
- Taganov KD, Boldin MP, Chang KJ, Baltimore D (2006) NF-kappaB-dependent induction of microRNA miR-146, an inhibitor targeted to signaling proteins of innate immune responses. Proc Natl Acad Sci U S A 103:12481–12486
- Taylor GA, Carballo E, Lee DM, Lai WS, Thompson MJ, Patel DD, Schenkman DI, Gilkeson GS, Broxmeyer HE, Haynes BF, Blackshear PJ (1996) A pathogenetic role for TNF alpha in the syndrome of cachexia, arthritis, and autoimmunity resulting from tristetraprolin (TTP) deficiency. Immunity 4:445–454
- Tebo J, Der S, Frevel M, Khabar KS, Williams BR, Hamilton TA (2003) Heterogeneity in control of mRNA stability by AU-rich elements. J Biol Chem 278:12085–12093
- Tenenbaum SA, Lager PJ, Carson CC, Keene JD (2002) Ribonomics: identifying mRNA subsets in mRNP complexes using antibodies to RNA-binding proteins and genomic arrays. Methods 26:191–198
- Tobler A, Meier R, Seitz M, Dewald B, Baggiolini M, Fey MF (1992) Glucocorticoids downregulate gene expression of GM-CSF, NAP-1/IL-8, and IL-6, but not of M-CSF in human fibroblasts. Blood 79:45–51
- Trabucchi M, Briata P, Garcia-Mayoral M, Haase AD, Filipowicz W, Ramos A, Gherzi R, Rosenfeld MG (2009) The RNA-binding protein KSRP promotes the biogenesis of a subset of microRNAs. Nature 459:1010–1014
- VAN Kouwenhove M, Kedde M, Agami R (2011) MicroRNA regulation by RNA-binding proteins and its implications for cancer. Nat Rev Cancer 11:644–656
- Wilusz CJ, Wilusz J (2004) Bringing the role of mRNA decay in the control of gene expression into focus. Trends Genet 20:491–497

- Winzen R, Kracht M, Ritter B, Wilhelm A, Chen CY, Shyu AB, Muller M, Gaestel M, Resch K, Holtmann H (1999) The p38 MAP kinase pathway signals for cytokine-induced mRNA stabilization via MAP kinase-activated protein kinase 2 and an AU-rich region-targeted mechanism. EMBO J 18:4969–4980
- Worthington MT, Pelo JW, Sachedina MA, Applegate JL, Arseneau KO, Pizarro TT (2002) RNA binding properties of the AU-rich element-binding recombinant Nup475/TIS11/tristetraprolin protein. J Biol Chem 277:48558–48564
- Xing W, Hao L, Yang X, Li F, Huo H (2014) Glucocorticoids induce apoptosis by inhibiting microRNA cluster miR1792 expression in chondrocytic cells. Mol Med Rep 10:881–886
- Yin Q, McBride J, Fewell C, Lacey M, Wang X, Lin Z, Cameron J, Flemington EK (2008) MicroRNA-155 is an Epstein-Barr virus-induced gene that modulates Epstein-Barr virusregulated gene expression pathways. J Virol 82:5295–5306
- Yu H, Sun Y, Haycraft C, Palanisamy V, Kirkwood KL (2011) MKP-1 regulates cytokine mRNA stability through selectively modulation subcellular translocation of AUF1. Cytokine 56: 245–255
- Yuan J, Muljo SA (2013) Exploring the RNA world in hematopoietic cells through the lens of RNA-binding proteins. Immunol Rev 253:290–303
- Zhang T, Kruys V, Huez G, Gueydan C (2002) AU-rich element-mediated translational control: complexity and multiple activities of trans-activating factors. Biochem Soc Trans 30:952–958
- Zheng Y, Xiong S, Jiang P, Liu R, Liu X, Qian J, Zheng X, Chu Y (2012) Glucocorticoids inhibit lipopolysaccharide-mediated inflammatory response by downregulating microRNA-155: a novel anti-inflammation mechanism. Free Radic Biol Med 52:1307–1317
- Zheng H, Li W, Wang Y, Xie T, Cai Y, Wang Z, Jiang B (2014) miR-23a inhibits E-cadherin expression and is regulated by AP-1 and NFAT4 complex during Fas-induced EMT in gastrointestinal cancer. Carcinogenesis 35:173–183
- Zhou F, Wang W, Xing Y, Wang T, Xu X, Wang J (2014) NF-kappaB target microRNAs and their target genes in TNFalpha-stimulated HeLa cells. Biochim Biophys Acta 1839:344–354

Post-transcriptional Regulation by Brain-Derived Neurotrophic Factor in the Nervous System

Alexandra M. Amen, Daniel L. Pham, and Mollie K. Meffert

1 Introduction

Brain-derived neurotrophic factor (BDNF) is a secreted protein that belongs to the neurotrophin family of growth factors and influences the growth and survival of neurons in both the central and peripheral nervous systems. BDNF signaling is crucial in early development where it plays significant roles in neuronal progenitor proliferation and differentiation, cell survival, the outgrowth of neuronal processes, and both the formation and pruning of synaptic connections. Mice homozygous for loss of BDNF (BDNF knockout) often die post-natally, and surviving mice show significant nervous system dysfunction (Jones et al. 1994; Ernfors et al. 1995; Patterson et al. 1996). Additionally, throughout adulthood, secretion of BDNF in response to neuronal activity functions to promote synaptic strengthening and enhance learning and memory, as well as to provide general trophic support for neuronal health and survival. These critical roles mean that abnormal BDNF signaling is closely associated with a range of mental health disorders, including autism spectrum disorder, depression, schizophrenia, bipolar disorder, and anxiety disorders, as well as neurodegenerative diseases, including Alzheimer's Disease and Parkinson's Disease (McAllister et al. 1999; Muglia et al. 2002; Tyler et al. 2002; Zuccato and Cattaneo 2009; Santos et al. 2010; Nagahara and Tuszynski 2011; Andero and Ressler 2012).

A.M. Amen • D.L. Pham

M.K. Meffert (🖂)

Solomon H. Snyder Department of Neuroscience, The Johns Hopkins University School of Medicine, Baltimore, MD 21205, USA

Solomon H. Snyder Department of Neuroscience, The Johns Hopkins University School of Medicine, Baltimore, MD 21205, USA

Department of Biological Chemistry, The Johns Hopkins University School of Medicine, Baltimore, MD 21205, USA e-mail: mkm@jhmi.edu

[©] Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0_14

BDNF expression is highest in the brain and is present in all brain regions, with the highest BDNF mRNA levels generally observed in the hippocampus and cerebral cortex (Wetmore et al. 1990; Conner et al. 1997). Despite its name, BDNF is also expressed and functional in a range of tissues outside the brain, including the peripheral nervous system, retina, skeletal muscle, kidneys, bone, and ovaries (Johnson et al. 1986; Jones et al. 1994; Ernfors et al. 1990; Griesbeck et al. 1995; Huber et al. 1996). In each of these tissues BDNF has been reported to play trophic roles in cell growth, survival, or repair (Fig. 14.1), but this chapter will focus particularly on BDNF in the central nervous system (CNS) where knowledge of downstream signaling events is the most comprehensive. BDNF has been well-characterized as a critical regulator of learning and memory in the hippocampus, and these studies have played a prominent role in elucidating the post-transcriptional mechanisms of BDNF action reviewed in this chapter.

The subcellular localization and release sites of BDNF in CNS neurons has been investigated in multiple reports, but studies have often been hampered by the low levels of endogenous BDNF message and protein in vivo and in vitro, making detection difficult. Bdnf mRNA has been reported in both the neuronal soma and dendrites depending on the length of its 3' UTR. However, recent research using deep-sequencing and high-resolution in-situ hybridization approaches indicates that all forms of bdnf mRNA appear to be restricted to the soma and proximal dendrites (Will et al. 2013). Similarly, BDNF has been observed in the synaptic compartment and was thought to be released from both axonal and dendritic sites, while recent research indicates that endogenous BDNF is primarily stored in pre-synaptic vesicles (Andreska et al. 2014). Conflicting data regarding the subcellular location and release sites of BDNF protein and message may in part arise from discrepancies between the behavior of endogenous BDNF and the exogenous BDNF expression constructs sometimes employed. Additionally, post-endoplasmic reticulum trafficking of secretory cargo, presumably including BDNF, has been shown to be subject to both developmental and acute spatial regulation in dendrites, where synaptic activity acts to restrict long-range transport of cargo (Hanus et al. 2014).

BDNF interacts with the high affinity tropomyosin-related kinase B (TrkB) tyrosine kinase receptors to activate multiple intracellular signaling cascades, including the MAPK, mTOR, and PLC γ pathways. Signaling downstream of TrkB receptors mediates the trophic effects of BDNF on neurons and synaptic connections, and blocking the BDNF/TrkB interaction results in significant impairments in synaptic plasticity and learning (Takei et al. 2001; Tanaka et al. 2008; Lai et al. 2012). Enduring effects of BDNF on growth and plasticity require changes in gene expression, and BDNF has been shown to regulate both transcription and translation. The precise complement of proteins present in neurons and at synapses fundamentally shapes neuronal growth and activity, ultimately determining synaptic response. Through the activation of select transcription factors such as serum response factor (SRF), nuclear factor kappa B (NF- κ B), and CREB (cAMP response element binding protein, Finkbeiner et al. 1997; Kalita et al. 2006; Riccio et al. 2006; Kajiya et al. 2009), BDNF is able to specifically upregulate trophic targets at the level of transcription. However, recent high-throughput approaches have shown that the



Fig. 14.1 BDNF signaling impacts biological processes throughout the human body. BDNF binds to the TrkB receptor, resulting in receptor dimerization and autophosphorylation. Binding to the TrkB receptor activates multiple downstream pathways, including PLCy, PI3K/AKt, and Mapk/Erk. Through PLCy, BDNF increases intracellular calcium concentration and activates calcium-binding molecules such as CaMKIIa, leading to CREB activation and regulation of gene transcription. BDNF-dependent Erk activation also phosphorylates CREB. Both Erk and mTOR regulate translation by binding to the cellular translational machinery. Through these pathways, BDNF regulates cell function, growth, and survival. In the brain, BDNF increases dendritic arborization (Horch and Katz 2002; Jaworski et al. 2005; Cheung et al. 2007; Takemoto-Kimura et al. 2007; Tanaka et al. 2008; Je et al. 2009; Lazo et al. 2013) and spine growth and number (Tyler and Pozzo-Miller 2001, 2003; Alonso et al. 2004) in neurons (I). Synaptic strengthening through long-term potentiation has been shown to require BDNF (2) (Korte et al. 1995; Kang et al. 1997; Rex et al. 2007). BDNF is also required for the formation and maintenance of memory (3) (Tyler et al. 2002; Heldt et al. 2007; Pardon 2010; Bekinschtein et al. 2007). Mechanisms underlying the effects of BDNF in the brain have been an active area of study, and critical roles for post-transcriptional regulation by BDNF have been demonstrated in the biological processes illustrated in boxes 1-3. BDNFdependent modulation of dopaminergic and serotinergic networks regulates mood, anxiety, and depression (4) [see reviews by Martinowich and Lu (2007) and Nikulina et al. (2014)]. BDNF regulates energy intake and body weight through its signaling in numerous hypothalamic nuclei [see review by Rios (2013)], and modulates stress response through the hypothalamic-pituitary axis (Givalois et al. 2001; Givalois et al. 2004; Jeanneteau et al. 2012 (5). Outside of the central nervous system, BDNF expression is low under basal conditions in mature bone cells and is upregulated in osteoblasts responding to bone fracture (6) (Kilian et al. 2014). BDNF also is known to promote the development and survival of B cells (7) in the immune system (Schuhmann et al. 2005; Fauchais et al. 2008). BDNF expression in the ovary has been shown to regulate follicular development and oocyte maturation (8) (Paredes et al. 2004; Spears et al. 2003; Jensen and Johnson 2001). In the vasculature, BDNF plays a role in the maintenance of blood vasculature and pericyte function (9) (Donovan et al. 2000; Anastasia et al. 2014). As novel functions of BDNF are discovered in multiple biological systems, our knowledge of the post-transcriptional mechanisms contributing to BDNF action is expected to continue to expand

cellular transcriptomes appear to correlate only moderately with corresponding proteomes (Tian et al. 2004; Schwanhausser et al. 2011; Ideker et al. 2013), revealing that the levels of many proteins are predominantly controlled post-transcriptionally. These studies collectively highlight the importance of understanding mechanisms of post-transcriptional regulation of gene expression. Given the dramatic role that BDNF plays in neuronal health, survival, and plasticity, understanding how BDNF signaling can generate trophic programs of gene expression at a post-transcriptional level has the potential to reveal critical regulatory points in both normal and abnormal brain function, and will be the central theme of this chapter.

2 BDNF Regulation of Global Protein Synthesis

Activity-dependent alterations in neuronal protein synthesis have long been known to be required for the endurance of synaptic changes and memory consolidation (Pfeiffer and Huber 2006; Gal-Ben-Ari et al. 2012). For example, secretion of BDNF during neuronal activity regulates translation, which allows for modulation of the neuronal proteome and accounts in part for the important role of BDNF in synaptic plasticity and cognition (Klann and Denver 2004; Soule et al. 2006; Santos et al. 2010). BDNF signaling through its TrkB receptor increases global cellular protein synthesis as measured by metabolic label incorporation in rodent cortical or hippocampal cultured neurons (Takei et al. 2001, 2009; Huang et al. 2012). The role of BDNF in upregulating total translation has also been demonstrated in studies using brain slices from transgenic mice engineered to either lack or overexpress the gene encoding BDNF. Brain slices from BDNF-deficient mice exhibit a significant decrease in basal translation, whereas basal translation is increased in brain slices from transgenic mice that overexpress BDNF (Takei et al. 2009).

The three major steps of eukaryotic translation involve translation initiation, elongation, and termination, and BDNF has been shown to enhance total cellular protein synthesis by influencing molecular components of both the initiation and elongation phases. In eukaryotic cells, translation is believed to be primarily regulated at the initiation phase, which begins when eukaryotic initiation factor 4E (eIF4E) binds to the 7-methyl-guanosine residue that constitutes the 5' cap of the mRNA. eIF4E binding to the mRNA allows recruitment of additional initiation factors, ultimately leading to ribosomal association with the eIF-mRNA complex. In the mammalian brain, eIF4E itself is regulated by eIF4E-binding protein 2 (4EBP2), which binds to and negatively regulates eIF4E (Banko et al. 2005). BDNF signaling through TrkB results in phosphorylation of 4EBP by the PI3K/mTOR pathway, which causes 4EBP to dissociate from eIF4E, leading to increased eIF4E activity and generally enhancing mRNA translation. Additionally, eIF4E itself may be phosphorylated by BDNF-mediated activation of the MAPK pathway, which appears to lead to stabilization of eIF4E and, again, enhanced total translation (Gingras et al. 1999, 2004; Takei et al. 2001; Kanhema et al. 2006). BDNF also decreases phosphorylation of eukaryotic initiation factor 2a (eIF2a) and increases phosphorylation of its guanine exchange factor eIF2B, both of which have been shown to enhance binding of tRNA to the ribosome (Takei et al. 2001).
319

BDNF acts on the elongation step of translation primarily by regulating the activity of eukaryotic elongation factor 2 (eEF2). eEF2 is a GTP-binding protein that is responsible for translocation of the tRNA from the A to P site in the ribosome, and is therefore a crucial regulator of the rate of elongation. Following treatment with BDNF, the translation elongation rate in neurons has been reported to be twice that of baseline, as measured by ribosomal transit time (Inamura et al. 2005). Phosphorylation of eEF2 is inactivating and thought to arrest translation, although translation of some mRNAs may be selectively upregulated when eEF2 is phosphorylated (Scheetz et al. 2000). BDNF appears to increase total levels of eEF2 protein (Takei et al. 2009), while decreasing phospho-eEF2 in an mTOR-dependent manner (Inamura et al. 2005; Takei et al. 2009). In contrast, infusion of BDNF into the dentate gyrus of adult rats lead to a rapid, transient increase in eEF2 phosphorylation in whole cell homogenates but not synaptodendrosomes (a subcellular fraction containing a portion of the dendritic shaft and associated spines; Kanhema et al. 2006). Such results suggest that while BDNF-mediated control of translation initiation may be widespread, control via translation elongation could be more compartment or transcript-specific. Collectively, these alterations in both initiation and elongation contribute to the modest BNDF-mediated upregulation of protein synthesis observed by global measures such as radiolabel incorporation.

3 BDNF Regulates the Local Protein Composition in Dendrites and Synapses

Neurons are structurally complex cells with elaborate dendritic and axonal processes that extend significant distances from the cell soma. Both passive and active mechanisms can mediate transport of proteins and mRNAs to distal processes. However, following exposure to an appropriate incoming stimulus, the levels of selective proteins are also specifically and rapidly upregulated at post-synaptic sites, as a component of synaptic plasticity. The finding of both mRNAs and polyribosomes in dendrites and dendritic spines, the major site of excitatory synaptic input, provided the initial suggestion of a role for local translation in dendrites and synapses (Steward and Levy 1982; Steward and Schuman 2001; Ostroff et al. 2002). Activitydependent local translation is now well-established and is known to be required for multiple forms of plasticity (Sutton and Schuman 2006; Wang et al. 2009; Holt and Schuman 2013). A number of studies have been aimed at compiling a profile of the synaptic transcriptome, with over 2500 transcripts reported to be localized to the synapse via deep sequencing (Cajigas et al. 2012). As might be expected, a number of mRNAs that code for proteins involved in such processes as neuronal signaling, synaptic plasticity, and synapse organization are preferentially enriched in synaptic and dendritic compartments (Poon et al. 2006; Zhang et al. 2006; Cajigas et al. 2012). The mRNAs for some important plasticity-related proteins, such as CamKII α , were reported to be present, although not enriched, in the dendritic cytoplasm, suggesting the possibility that preferential local translation of these mRNAs could contribute to the abundance of the proteins at synapses (Poon et al. 2006; Zhong et al. 2006).

Mechanisms through which BDNF increases global protein synthesis, as previously discussed, can also function to increase protein synthesis in dendrites and synapses (Takei et al. 2004; Kanhema et al. 2006). For example, BDNF leads to phosphorylation of 4EPB1 in both the dendrite and synapse in an mTOR–dependent manner (Takei et al. 2004; Kanhema et al. 2006). Additionally, in isolated synapses, BDNF causes the inhibitory protein CYFIP1 to dissociate from eIF4E, leading to enhanced translation (Napoli et al. 2008). Similarly, phosphorylation of p70S6K via BDNF activation of mTOR results in phosphorylation of ribosomal S6 protein, increasing translation activation at the synapse (Takei et al. 2004). In other cell types, it is thought that the association of ribosomal subunits, translation factors, and mRNAs with F-actin functions to hold the molecular players in an ordered state and facilitate translation. BDNF has been reported to cause the translocation of eIF4E into spines where it associates with cytoskeleton-bound RNA granules, an effect that is blocked by inhibiting F-actin polymerization (Smart et al. 2003).

Not surprisingly, neuronal studies on local protein regulation by BDNF have often focused on the upregulation by BDNF of proteins that are known to play important roles in synaptic function and plasticity. Increased translation of multiple specific mRNAs such as glutamate receptor subunits, Homer2, and CamKIIa is observed at synapses in response to BDNF (Schratt et al. 2004; Caldeira et al. 2007; Guire et al. 2008; Fortin et al. 2012). In a study performed by Aakalu et al. (2001), the authors constructed a GFP reporter containing the 5' and 3' UTRs of CamKIIa, to reveal that stimulation with BDNF led to translational "hot spots" in dendrites that may correlate with active synapses. BDNF has also been shown to increase the levels of certain mRNAs in dendrites (Ying et al. 2002; Rao et al. 2006; Messaoudi et al. 2007), supporting the function of BDNF as a regulator of local gene expression. It is worth noting that BDNF has also been implicated in modulating synaptic protein content through post-translational modifications. For example, BDNFinduced phosphorylation of both NDMA and AMPA receptor subunits, leads to increased surface trafficking and activity of these receptors (Suen et al. 1997; Lin et al. 1998; Caldeira et al. 2007). Evidence also suggests that BDNF may posttranslationally regulate synaptic protein stability through the ubiquitin proteasome pathway (Jia et al. 2008), and that neuronal activity leads to increased proteasomes in the synapse, which can function to remodel synaptic protein content (Bingol and Schuman 2006).

4 Effect of BDNF on Translational Specificity

4.1 BDNF Increases Translation of a Select Group of mRNAs

Long-term changes in synaptic and circuit function are known to require changes in gene expression. Enduring strengthening or weakening of synapses each necessitate organized alterations in programs of gene expression in order to ensure that the appropriate proteins are coordinately regulated to enhance or reduce synaptic responses. BDNF effectively promotes synapse growth and plasticity by coordinating a response that specifically and selectively increases plasticity-related proteins. Despite the ability of BDNF to affect general translational machinery and total cellular translation, the modest increase in global translation that is observed in response to BDNF has actually been attributed to a robust effect on a relatively small number of specific transcripts. Using a candidate-based approach, several initial studies demonstrated that BDNF can increase the translation of specific plasticity-related proteins—such as CamKII α , Arc, and glutamate receptor subunits—in both dissociated neuronal culture and synaptic preparations (Aakalu et al. 2001; Yin et al. 2002; Kelleher et al. 2004; Takei et al. 2004; Kanhema et al. 2006; Jourdi et al. 2009). In contrast, BDNF has been observed to decrease translation of certain mRNAs, such as potassium channels and co-transporters (Rivera et al. 2002; Raab-Graham et al. 2006). Radiolabelled synapses stimulated with BDNF and subjected to 2D electrophoresis also revealed a robust increase in a very specific set of proteins, while some proteins were decreased and the majority were unchanged (Yin et al. 2002).

The development of high-throughput techniques enabled an appreciation of the truly impressive extent to which BDNF mediates target specificity. In a comprehensive study from the Greenberg lab (Schratt et al. 2004), the authors used polysome profiling to show that only roughly 4 % of transcripts present in neurons were increased by BDNF stimulation, and that these increases were sensitive to mTOR signaling. Additionally, multidimensional protein identification technology (MudPIT) demonstrated that a brief, 30 min BDNF stimulation of isolated synapses was sufficient to specifically increase proteins involved in synaptic vesicle formation and trafficking, translation, and synaptic components (Liao et al. 2007). In line with previous studies, the translation from highly selective pools of mRNAs were specifically increased as well as decreased by BDNF, while most mRNAs remained unchanged (Liao et al. 2007). Similarly, gel based proteome profiling revealed that a long-term (12 h) BDNF stimulation could increase levels of a number of proteins involved in cellular metabolism and proliferation, while most protein levels were decreased or unchanged (Manadas et al. 2009). Interestingly, the authors reported that BDNF was capable of affecting both the mRNA and protein levels of its targets, but that these changes were not always correlated, again implicating posttranscriptional roles for BDNF in regulating *de novo* protein synthesis or protein stability (Manadas et al. 2009).

4.2 Mechanisms Enabling BDNF-Mediated Translational Specificity

The ability of BDNF to rapidly and significantly increase the translation of a restricted subset of mRNAs is in accordance with its role as a positive modulator of growth and excitatory synaptic function, but how is BDNF able to confer such a remarkable degree of specificity to translational processes?

4.2.1 RNA-Binding Proteins

One well-established method of achieving translational specificity is through RNA-binding proteins, which play important roles in post-transcriptional regulation of mRNAs by enhancing or repressing the translation of certain mRNAs. Several different RNA-binding proteins have been observed both to be present at synapses, and to regulate the translation of major targets of BDNF (Huang et al. 2002; Napoli et al. 2008; Wu et al. 1998; Bhakar et al. 2012). For example, the RNA-binding protein CPEB (cytoplasmic polyadenylation element-binding protein) regulates mRNA translation by mediating cytoplasmic polyadenylation. CPEB binds to a cytoplasmic polyadenylation element (CPE) present in the 3' UTR of multiple BDNF target mRNAs, such as CamKIIα (Huang et al. 2002; Wu et al. 1998). CPEB also associates with an inhibitory eIF4E binding protein, Maskin, which when present in the CPEB/Maskin/eIF4E complex specifically prevents translation of bound CPE-containing mRNAs. Activity-dependent phosphorylation of CPEB causes increased cytoplasmic polyadenylation, which disrupts the Maskin/eIF4E interaction, leading to increased translation of CPE-containing mRNAs (Richter and Sonenberg 2005). In this regulatory pathway, CPEB phosphorylation in response to synaptic activity has been shown to result in an accumulation of CamKII α at the synapse (Huang et al. 2002). Although CPEB regulation has not been directly studied in relation to BDNF-mediated post-transcriptional gene target specificity, the fact that BDNF initiates several major phosphorylation cascades and increases translation of many CPE-containing mRNAs suggests that BDNF could potentially regulate cytoplasmic polyadenylation in the dendrite and synapse.

As discussed, BDNF regulates translation both locally and globally through mTOR-dependent phosphorylation of 4EBP2, leading to enhanced eIF4E-dependent initiation. Though this pathway might initially be expected to non-selectively increase translation of mRNAs, recent research has shown that 4EBP2 knockout does not significantly alter the neuronal polysome profile, but instead leads to increased translation of specific mRNAs such as neuroligins (Gkogkas et al. 2013). In addition, several studies have demonstrated a specific requirement for 4EBP2-dependent modulation of eIF4E activity in LTP and memory. These results indicate an important role for such molecules in translation regulation and might suggest selectivity for proteins involved in synaptic plasticity (Banko et al. 2005; Costa-Mattioli et al. 2009). Ultimately these investigations imply that 4EBP2-mediated regulation of cap-dependent initiation could contribute to translational specificity for a subset of transcripts in response to BDNF rather than mediating a general increase in translation.

BDNF signaling has also been linked to another RNA-binding protein, the Fragile X Mental Retardation Protein (FMRP), an RNA-binding protein that is enriched at CNS synapses and could participate in conferring specificity to BDNF-mediated translation. Loss of FMRP results in human Fragile X Syndrome, which is an autism spectrum disorder and the most common monogenic inheritable form of intellectual disability. FMRP is generally thought to function as a repressor of protein synthesis, but participates in the regulation of translation through multiple incompletely characterized mechanisms (Bhakar et al. 2012). Loss of FMRP leads to exaggerated

translation of a variety of synaptic proteins including many major targets of BDNF, such as CamKIIα, Arc, Map1B, and APP (Napoli et al. 2008; Bhakar et al. 2012). FMRP function and expression has been directly linked to BDNF signaling in multiple studies (Castren et al. 2002; Napoli et al. 2008; Wang et al. 2012b; De Rubeis et al. 2013). BDNF decreased FMR1 mRNA levels in cultured hippocampal neurons, and both FMR1 mRNA and FMRP protein levels were found to be decreased relative to controls in transgenic mice overexpressing the BDNF receptor TrKB (Castren et al. 2002). The phosphorylation status of FMRP may play a role in its inhibition of translation, and dephosphorylation of FMRP has been suggested to enhance translation of FMRP target mRNAs (Ceman et al. 2003; Narayanan et al. 2007). One study directly examining the relationship of BDNF to FMRP phosphorylation showed that BDNF caused calcineurin-mediated dephosphorylation of FMRP, leading to increased translation of FMRP targeted mRNAs (Wang et al. 2012b). Similarly, PP2A, another phosphatase that is reported to dephosphorylate FMRP (Niere et al. 2012), is itself increased by BDNF (Takei et al. 2009). Research aimed at understanding the mechanism by which FMRP regulates translation has suggested that FMRP may work in part by binding to the protein CYPIP1 to harbor FMRP mRNA targets in the FMRP-CYPIP1 complex. CYPIP1 itself also binds to and sequesters eIF4E, preventing translation of these targets (Napoli et al. 2008; De Rubeis et al. 2013). BDNF was shown to decrease co-immunoprecipitation of CYPIP1 and eIF4E, suggesting that eIF4E is released and able to initiate translation of FMRP target mRNAs (Napoli et al. 2008; De Rubeis et al. 2013).

4.2.2 microRNAs

Although the discovery of microRNA-mediated regulation of protein synthesis is relatively recent (Lee et al. 1993), in the past 20 years many ground-breaking studies have demonstrated the importance of microRNA (miRNA) function in all cell types. miRNAs are short (19-25 nucleotide), non-coding RNA molecules that are endogenously expressed. They are able to recognize and bind partiallycomplementary sites in their target mRNAs, causing translational suppression and/ or degradation of these targets. miRNA function in the brain has been shown to be crucial for normal neuronal development and plasticity in both mammalian and non-mammalian species (Krichevsky et al. 2003; Sempere et al. 2004; Giraldez et al. 2005; Smalheiser and Lugli 2009) Given that a single miRNA can regulate expression of an entire suite of proteins, miRNAs are attractive candidates for coordinating complex responses such as neuronal development, plasticity and synaptic remodeling at a post-transcriptional level. There is strong evidence for the presence of both miRNAs and miRNA processing machinery at the synapse (Schratt et al. 2006; Ashraf et al. 2006; Wayman et al. 2008; Huang et al. 2012; Lugli et al. 2012), further supporting a potential role for miRNAs in synaptic plasticity. The miRNA processing enzyme Dicer has been found to be present and active in synaptosomes, and there is additional evidence for enrichment of both precursor (pre-miRNAs) and mature miRNAs at synapses (Lugli et al. 2005, 2008). Interestingly, enrichment of mature miRNAs has been correlated with the presence of the corresponding pre-miRNA, again suggesting that miRNA processing might happen locally at the

synapse (Lugli et al. 2008). Though it is still not known how miRNAs could be directed or trafficked to a particular synapse, it has been suggested that synaptic pre-miRNAs might be enriched for pre-miRNA structures with shorter stems and more bulges, offering a potential structural basis for their transport as opposed to immediate processing (Smalheiser 2008).

Initial studies addressing the potential for miRNA regulation by BNDF examined the brain-specific miRNA miR-134, which is associated with learning and memory in rodents and negatively regulates spine size by inhibiting Limk1 mRNA in the synapse (Schratt et al. 2004; Gao et al. 2010). Limk1 undergoes increased translation in response to BDNF, and the authors found that BDNF prevents miR-134-mediated repression of Limk1 through an unknown mechanism (Schratt et al. 2004, 2006). BDNF was shown to decrease levels of miR-9, relieving suppression of MAP1b and leading to increased axon branching (Dajas-Bailador et al. 2012). In addition to negative regulation of miRNAs, BDNF was also reported to upregulate the transcription of several miRNAs. For example, miR-134 transcription was enhanced by BDNF, promoting inhibition of the RNA-binding protein Pumilio2 (Pum2). This caused an increase in dendritic arborization of young cultured neurons (Fiore et al. 2009; Vessey et al. 2010). Additionally, BDNF is thought to increase transcription of miR-132, a regulator of dendrite outgrowth and morphology, in a MAPK-dependent manner (Vo et al. 2005; Wayman et al. 2008; Kawashima et al. 2010). Interestingly, inhibition of miR-132 was shown to reduce BDNF-mediated upregulation of several synaptic glutamate receptor proteins such as NR2A, NR2B, and GluA1 (Kawashima et al. 2010).

In addition to these reports of BDNF effects on individual miRNAs, recent work from our laboratory has shown that BDNF can act upstream in processing steps to coordinately regulate the biogenesis of multiple miRNAs. Precursor miRNAs undergo several well-established processing steps to produce the functional mature miRNA. The canonical miRNA biogenesis pathway starts with transcription of a larger precursor miRNA (called a pri-miRNA) in the nucleus, where it is processed into a pre-miRNA stem-loop structure (of ~70 nucleotides) by the enzyme Drosha. This pre-miRNA hairpin is then exported into the cytoplasm, where it undergoes a second processing step by the RNaseIII enzyme Dicer and its binding partner Tar-RNA Binding Protein (TRBP) into a mature miRNA duplex. Regulation of either of these processing steps has the potential to dramatically affect the levels of functional mature miRNAs, and the translation of their corresponding mRNA targets. We demonstrated that BDNF achieves gene target specificity in protein synthesis by regulating the miRNA biogenesis pathway (Huang et al. 2012). BDNF stimulation of cultured hippocampal neurons elevates both Dicer and TRBP proteins in a rapid, transcription-independent manner, which enhances pre-miRNA processing and is associated with a concurrent increase in many mature miRNAs. This finding might initially appear somewhat counterintuitive, given that an increase in mature miR-NAs could suggest a decrease rather than the modest increase in total translation produced by BDNF. However, we also observed that BDNF selectively decreases the processing of a specific class of miRNAs that are inhibited by the RNA-binding protein Lin28. The Let-7 family of miRNAs are a highly abundant group of these

Lin28-targeted miRNAs, which repress a range of pro-growth mRNAs that are crucial for synaptic function, such as CamKIIα and GluA1 (Huang et al. 2012). The Let-7 class of miRNAs has received widespread attention due to their role in development and cancer. Because they inhibit translation of a range of important pro-growth proteins, the Let-7 miRNAs are virtually absent from stem cells, and are substantially upregulated during developmental progression (Moss et al. 1997; Loohuis et al. 2012). Additionally, Let-7 miRNAs function as tumor suppressor genes and are downregulated in roughly 15 % of cancers (Wang et al. 2012a). The Let-7 miRNAs are one of the most highly expressed classes of miRNAs in mature cell types, and high-throughput data suggests that Let-7 miRNAs make up at least 50 % of miRNAs present in mature neurons (Juhila et al. 2011; Shinohara et al. 2011). By reducing Let-7 levels and relieving Let-7 miRNA-mediated repression, BDNF is able to produce enhanced translation of a restricted subset of targeted mRNAs which presumably underlies the modest global increase in translation caused by BDNF.

Regulation of Let-7 miRNA biogenesis by Lin28 has been well documented (Moss et al. 1997; Heo et al. 2008, 2009; Nam et al. 2011). Lin28a and its mammalian paralog Lin28b are RNA-binding proteins and pro-growth pluripotency factors that are highly expressed in stem cells and a range of cancer cells and are downregulated during development, mirroring the increase in Let-7 miRNAs (Moss et al. 1997; Loohuis et al. 2012). This opposing expression pattern occurs through negative feedback between Lin28 and Let-7 miRNAs. Lin28 proteins recognize and bind a 'GGAG' motif in the terminal loop of Let-7 miRNA precursors, which causes degradation or inhibits processing of the precursor Let-7 miRNAs into mature miRNAs (Heo et al. 2008, 2009; Nam et al. 2011). Thus, Lin28 proteins are able to prevent Let-7 miRNA maturation prior to Dicer/TRBP processing. Lin28 mRNAs are, in turn, subject to repression by Let-7 miRNAs which can create feed-forward derepression when Lin28 proteins are elevated. Though Lin28 is generally considered to be absent from differentiated cell types such as neurons, we found that Lin28a protein was substantially elevated by BDNF in a rapid, transcription-independent manner, and that this accounted for the specific decrease in mature Let-7 miRNA expression following BDNF stimulation (Huang et al. 2012). Collectively, BDNF positively regulates general miRNA biogenesis by upregulating Dicer and TRBP proteins, and specifically negatively regulates Let-7 miRNA biogenesis in particular by increasing Lin28a protein. In this way, BDNF increases miRNA-mediated repression of many mRNAs while selectively relieving translational repression of mRNAs that are targets of Let-7 miRNAs. Increased miRNA biogenesis by Dicer/TRBP induction could account for the transcripts that have been shown to undergo decreased translation in response to BDNF (Rivera et al. 2002; Raab-Graham et al. 2006), and indeed knockdown of Dicer protein in cultured neurons could prevent BDNFmediated decrease in such targets (Huang et al. 2012). In contrast, BDNF leads to increased translation of specific pro-growth mRNAs that harbor Let-7 miRNA binding sites by increasing Lin28a protein levels to inhibit Let-7 miRNA processing. Knockdown of Lin28a was validated to prevent a BDNF-mediated increase in multiple important Let-7 containing targets, such as CamKIIa, that are known to contribute to the synaptic and cognitive effects of BDNF (Huang et al. 2012). Further, synthetic insertion of a Let-7 miRNA binding site in an mRNA 3'UTR, was shown to be sufficient to create a BDNF-upregulated target gene.

5 Physiological Effects of Post-transcriptional Regulation by BDNF

5.1 Neuronal Morphology

Alterations in neuronal structure, such as dendritic outgrowth and spine maturation, are physiological components of enduring plasticity that have been shown to require BDNF-dependent changes in protein synthesis (Jaworski et al. 2005; Tanaka et al. 2008). In a translation-dependent manner, BDNF signals through TrkB receptors to enhance dendritic complexity and branching in both hippocampal and cortical neurons (Horch and Katz 2002; Jaworski et al. 2005; Cheung et al. 2007; Takemoto-Kimura et al. 2007; Tanaka et al. 2008; Je et al. 2009; Lazo et al. 2013). BDNF released from one neuron can enhance dendritic arborization of a closely neighboring neuron by binding to its TrkB receptors, and pyramidal neurons lacking TrkB receptors have significantly reduced pyramidal dendritic arbors (Horch and Katz 2002; Xu et al. 2000) This ability of BDNF to increase dendritic arborization is clearly linked to its regulation of translation. For example, Jaworski et al. (2005) demonstrated that enhanced dendritic complexity in response to BDNF expression was dependent on the mTOR signaling pathway, and could be blocked by inhibiting translation initiation. Studies from our laboratory have shown that the gene target specificity of BDNF-mediated translation is also crucial for its promotion of dendritic arborization. In particular, enhanced dendritic growth requires BDNF-induced reductions in Let-7 miRNA levels mediated by the RNA-binding protein Lin28a, presumably due to the many plasticity and growth-related mRNAs that are targeted by Let-7 miRNAs. In hippocampal neurons expressing a Lin28-resistant precursor of Let-7 miRNA, BDNF is no longer able to enhance dendritic arborization (Huang et al. 2012).

BDNF is also a well-characterized regulator of spine dynamics, and can induce changes in dendritic spine size and number. In cultured hippocampal neurons, BDNF has been shown to robustly increase spine density on the apical dendrite of pyramidal neurons in a MAPK/ERK-dependent manner (Tyler and Pozzo-Miller 2001, 2003; Alonso et al. 2004). Research has also suggested a differential effect of acute (one time) vs gradual (low-level repeated) BDNF stimulation on spine morphology. Specifically, acute BDNF stimulation of cultured hippocampal neurons produced transient TrkB activation and resulted in spine head enlargement, indicative of a more mature spine with a larger post-synaptic density. Alternatively, gradual BDNF stimulation led to long-term TrkB elevation and spine neck elongation, suggesting spine destabilization and remodeling. Both types of BDNF stimulation caused an increase in overall spine density (Ji et al. 2010).

Again, the effect of BDNF on spine plasticity is dependent on its regulation of protein-synthesis. In general, activity-induced upregulation of dedritic spines can be

blocked by treating cultured neurons with the drug anisomycin, an inhibitor of protein synthesis (Fifková et al. 1982; Srivastava et al. 2012). More directly, long-term increases in hippocampal spine density mediated by BDNF have been shown to require new translation utilizing a system that allows inducible cell-autonomous inactivation of protein synthesis (Je et al. 2009). BDNF signaling through TrkB was also demonstrated to be necessary and sufficient for LTP-mediated spine head enlargement, and this effect was entirely blocked by anisomycin (Tanaka et al. 2008). Collectively, this research indicates that post-transcriptional regulation of gene expression by BDNF can induce structural changes in both synaptic and dendritic morphology that participate in plastic responses.

5.2 LTP and Memory

Synaptic plasticity is widely considered the cellular correlate of learning and memory, and involves the selective strengthening and weakening of synapses in response to incoming stimuli. The strengthening of specific synapses occurs through a phenomenon known as long-term potentiation (LTP), and is the most widely studied and best understood form of synaptic plasticity. LTP involves sequential phases that include both early-LTP (E-LTP), which lasts 1–2 h and is transcription and translation independent, and late-LTP (L-LTP) which is of longer duration and is known to require novel transcription and translation (Krug et al. 1984; Stanton and Sarvey 1984; Costa-Mattioli et al. 2009). Interestingly, L-LTP can be blocked locally by restricted dendritic administration of protein synthesis inhibitors to prevent local translation (Bradshaw et al. 2003).

BDNF is stored near synaptic sites and is released during neuronal activity, including paradigms of LTP induction (Hartmann et al. 2001; Aicardi et al. 2004; Matsuda et al. 2009). LTP requires the expression and activity of BDNF (Korte et al. 1995; Kang et al. 1997; Rex et al. 2007). Exogenous administration of BDNF has been reported to induce LTP in multiple brain regions including the hippocampus (Kang and Schuman 1996; Tanaka et al. 2008), though the sufficiency of BDNF for this effect may depend upon experimental context. In line with the ability of BDNF to control translation both locally and cell-wide, BDNF has been shown to regulate LTP in a global but also synapse-specific manner (Kang and Schuman 1996; Tanaka et al. 2008). Generally it is thought that initial phosphorylation cascades initiated by BDNF-TrkB signaling lead to changes associated with E-LTP, and that subsequent BDNF-induced protein synthesis is required for the enduring late phase of LTP (Kang and Schuman 1996; Messaoudi et al. 2007; Tanaka et al. 2008).

Consistent with the endurance of LTP requiring new protein synthesis, the persistence of long-term memory (LTM) also relies upon a consolidation period that is protein-synthesis dependent (Bailey et al. 2004; Lamprecht and LeDoux 2004). Consolidation of LTM leads to memories that can persist for days, weeks, or years. The hippocampus is thought to be a major locus of memory consolidation, and in-vivo hippocampal injection of protein synthesis inhibitors prevents hippocampal-dependent memory formation (Tronel et al. 2005; Power et al. 2006).

The requirement for BDNF in tests of hippocampal-dependent learning and memory is also well established (Tyler et al. 2002; Heldt et al. 2007; Pardon 2010). While few studies have directly attempted to link BDNF function with novel translation in hippocampal-based learning, recent research has shown that LTM in the hippocampus requires multiple cycles of translation that occur following learning paradigms, and that BDNF is elevated concurrent with the initiation of these cycles (Bekinschtein et al. 2007). Additionally, inhibition of BDNF prevented LTM formation in the same timecourse as intrahippocampal anisomycin injection (Bekinschtein et al. 2007). These findings are consistent overall with a role for BDNF in translation-dependent changes of neuronal and synaptic function that are associated with learning and memory consolidation.

5.3 Post-transcriptional Regulation by BDNF in Autism Spectrum Disorder

Multiple cognitive disorders have been associated with dysregulated protein synthesis (for a review, see Buffington et al. 2014), and several have also been strongly linked to dysregulated BDNF signaling. In particular, there is abundant evidence for aberrant BDNF signaling and downstream protein synthesis in autism spectrum disorder (ASD) arising from diverse genetic causes. ASD has a global prevalence of about 1 % and is characterized primarily through impairments in social interaction and communication. ASD is largely a genetic disorder, however it can arise as a result of monogenic or polygenic causes, and frequently involves *de novo* mutations and gene copy number variants. Abnormalities in activity-dependent neuronal signaling and synapse development and function are thought to be shared features of ASD arising from known monogenic causes, as well as other syndromic disorders with features of autism (for a review, see Ebert and Greenberg 2013).

Fragile X Syndrome (FXS) is the most common monogenic cause of autism, and occurs as a result of mutation in the *FMR1* gene promoter leading to loss of fragile X mental retardation protein (FMRP), a negative regulator of protein synthesis. Using *FMR1* knockout mice as a model of the human syndrome, researchers have observed that exogenous BDNF application is able to rescue significant hippocampal LTP deficits in FXS, and that genetic reduction of BDNF alongside FMRP knockout further impairs cognitive deficits (Lauterborn et al. 2007; Uutela et al. 2012). As was previously discussed, FMRP regulates translation of a number of mRNAs that are also regulated by BDNF, such as Arc and CamKIIα (Napoli et al. 2008) as well as *bdnf* and *TrkB* mRNA themselves (Louhivuori et al. 2011). BDNF signaling through its TrkB receptor has also been shown to decrease *FMR1* mRNA as well as FMRP protein (Castren et al. 2002). Together, this research suggests that BDNF expression, function, and the subsequent post-transcriptional regulation of FMRP target mRNAs is abnormal in FXS.

Tuberous sclerosis (TSC) is another autism spectrum disorder with a known genetic cause, resulting from loss-of-function mutations in the *TSC1* or *TSC2* genes.

329

The TSC1–TSC2 protein complex functions to inhibit mTOR signaling via Rheb activity. BDNF binding to TrkB leads to Akt or Erk-mediated phosphorylation of TSC1-TSC2, causing its inactivation and allowing for appropriate mTOR function (Han and Sahin 2011). Loss of TSC1 or TSC2 results in misregulation of mTOR and abnormal protein synthesis, which can be reversed through inhibition of mTOR (Ehninger et al. 2008). Similarly, multiple cases of syndromic autism have been linked to mutations in phosphatase and tensin homolog (PTEN), which also results in overactivation of the PI3K/Akt/mTOR pathway and is decreased in response to BDNF (Briz et al. 2013). Given that Akt, Erk, and mTOR are all major targets of BDNF signaling, it is likely that BDNF-mediated control of protein synthesis could be impaired in neurons with deficiency in major regulators of this pathway. Mutations affecting the methyl-CpG-binding protein 2 (MeCP2) transcriptional regulator results in an additional monogenic ASD, termed Rett Syndrome. MeCP2 is known to bind methylated CpG sites in the BDNF gene (Klose et al. 2005), and MeCP2's function as either an activator or a repressor has been shown lead to complex misregulation of BDNF expression and evoked secretion linked to Rett Syndrome phenotypes (Chen et al. 2003; Martinowich et al. 2003; Chang et al. 2006; Li et al. 2012; Ebert and Greenberg 2013). Future studies will be aimed at determining whether regulating important targets of BDNF signaling could ameliorate neuronal abnormalities observed in diverse forms of ASD.

6 Concluding Remarks

BDNF has been well-established as an important regulator of both global and local protein synthesis in neurons. The equally well-established and consistent roles of BDNF in promoting pro-growth responses, whether studied at the level of molecules, synapses, or cognition, highlight the significance of the remarkable pro-growth gene target specificity that BDNF confers to translation. Physiological effects of BDNF such as enhanced spine density, synaptic function, and cognitive performance necessitate that BDNF enhances translation not of all mRNAs, but instead of a subset of mRNAs that support growth and plasticity. Recent research has begun to reveal the mechanisms allowing BDNF-mediated specificity in protein synthesis, but links between translational specificity in response to BDNF and the physiological and behavioral requirements of BDNF are not yet well studied. A better understanding of the post-transcriptional mechanisms allowing BDNF to selectively generate trophic responses has the potential to reveal novel control points in both normal physiology and disease states. Future research should also aim to help us determine whether the mechanisms behind BDNF-mediated post-transcriptional regulation can be generalized across brain regions and tissue types. Overall, further elucidation of BDNF signaling pathways and mechanistic understanding of how they impact structural and functional effects of BDNF will be vital to understanding of the role of BDNF in brain function, and for potential development of therapeutics aimed at treating the wide variety of diseases and disorders associated with BDNF misregulation.

References

- Aakalu G, Smith WB, Nguyen N, Jiang C, Schuman EM (2001) Dynamic visualization of local protein synthesis in hippocampal neurons. Neuron 30:489–502
- Aicardi G, Argilli E, Cappello S, Santi S, Riccio M, Thoenen H, Canossa M (2004) Induction of long-term potentiation and depression is reflected by corresponding changes in secretion of endogenous brain-derived neurotrophic factor. Proc Natl Acad Sci U S A 101:15788–15792
- Alonso M, Medina JH, Pozzo-Miller L (2004) ERK1/2 activation is necessary for BDNF to increase dendritic spine density in hippocampal CA1 pyramidal neurons. Learn Mem 11:172–178
- Anastasia A, Deinhardt K, Wang S, Martin L, Nichol D, Irmady K, Trinh J, Parada L, Rafii S, Hempstead BL, Kermani P (2014) Trkb signaling in pericytes is required for cardiac microvessel stabilization. PLoS One 9:87406
- Andero R, Ressler KJ (2012) Fear extinction and BDNF: translating animal models of PTSD to the clinic. Genes Brain Behav 11:503–512
- Andreska T, Aufmkolk S, Sauer M, Blum R (2014) High abundance of BDNF within glutamatergic presynapses of cultured hippocampal neurons. FNINS 8:107
- Ashraf SI, McLoon AL, Sclarsic SM, Kunes S (2006) Synaptic protein synthesis associated with memory is regulated by the RISC pathway in Drosophila. Cell 124:191–205
- Bailey CH, Kandel ER, Si K (2004) The persistence of long-term memory: a molecular approach to self-sustaining changes in learning-induced synaptic growth. Neuron 44:49–57
- Banko JL, Poulin F, Hou L, DeMaria CT, Sonenberg N, Klann E (2005) The translation repressor 4E–BP2 is critical for eIF4F complex formation, synaptic plasticity, and memory in the hippocampus. J Neurosci 25:9581–9590
- Bekinschtein P, Cammarota M, Igaz LM, Bevilaqua LR, Izquierdo I, Medina JH (2007) Persistence of long-term memory storage requires a late protein synthesis and BDNF-dependent phase in the hippocampus. Neuron 53:261–277
- Bhakar AL, Dolen G, Bear MF (2012) The pathophysiology of fragile X (and what it teaches us about synapses). Annu Rev Neurosci 35:417–443
- Bingol B, Schuman EM (2006) Activity-dependent dynamics and sequestration of proteasomes in dendritic spines. Nature 441:1144–1148
- Bradshaw KD, Emptage NJ, Bliss TV (2003) A role for dendritic protein synthesis in hippocampal late LTP. Eur J Neurosci 18:3150–3152
- Briz V, Hsu YT, Li Y, Lee E, Bi X, Baudry M (2013) Calpain-2-mediated PTEN degradation contributes to BDNF-induced stimulation of dendritic protein synthesis. J Neurosci 33:4317–4328
- Buffington SA, Huang W, Costa-Mattioli M (2014) Translational control in synaptic plasticity and cognitive dysfunction. Annu Rev Neurosci 37:17–38
- Cajigas IJ, Tushev G, Will TJ, tom Dieck S, Fuerst N, Schuman EM (2012) The local transcriptome in the synaptic neuropil revealed by deep sequencing and high-resolution imaging. Neuron 74:453–466
- Caldeira MV, Melo CV, Pereira DB, Carvalho R, Correia SS, Backos DS, Carvalho AL, Esteban JA, Duarte CB (2007) Brain-derived neurotrophic factor regulates the expression and synaptic delivery of alpha-amino-3- hydroxy-5-methyl-4-isoxazole propionic acid receptor subunits in hippocampal neurons. J Biol Chem 282:12619–12628
- Castren M, Lampinen KE, Miettinen R, Koponen E, Sipola I, Bakker CE, Oostra BA, Castren E (2002) BDNF regulates the expression of fragile X mental retardation protein mRNA in the hippocampus. Neurobiol Dis 11:221–229
- Ceman S, O'Donnell WT, Reed M, Patton S, Pohl J, Warren ST (2003) Phosphorylation influences the translation state of FMRP-associated polyribosomes. Human Mol Gen 12:3295–3305
- Chang Q, Khare G, Dani V, Nelson S, Jaenisch R (2006) The disease progression of *Mecp2* mutant mice is affected by the level of BDNF expression. Neuron 49:341–348
- Chen WG, Chang Q, Lin Y, Meissner A, West AE, Griffith EC et al (2003) Derepression of BDNF transcription involves calcium-dependent phosphorylation of MeCP2. Science 302:885–889

- Cheung ZH, Chin WH, Chen Y, Ng YP, Ip NY (2007) Cdk5 is involved in BDNF-stimulated dendritic growth in hippocampal neurons. PLoS Biol 5:e63
- Conner JM, Lauterborn JC, Yan Q, Gall CM, Varon S (1997). Distribution of brain-derived neurotrophic factor (BDNF) protein and mRNA in the normal adult rat CNS: evidence for anterograde axonal transport. J Neurosci 17: 2295–2313
- Costa-Mattioli M, Sossin WS, Klann E, Sonenberg N (2009) Translational control of long-lasting synaptic plasticity and memory. Neuron 61:10–26
- Dajas-Bailador F, Bonev B, Garcez P, Stanley P, Guillemot F, Papalopulu N (2012) microRNA-9 regulates axon extension and branching by targeting Map1b in mouse cortical neurons. Nat Neurosci 15:697–699
- De Rubeis S, Pascuito E, Li KW, Fernandez E, Di Marino D, Buzzi A, Ostroff LE, Klann E, Zwartkruis FJ, Komiyama NH, Grant SG, Poujol C, Choquet D, Achsel T, Posthuma D, Smit AB, Bagni C (2013) CYPIP1 coordinates mRNA translation and cytoskeleton remodeling to ensure proper dendritic spine formation. Neuron 79:1169–1182
- Donovan MJ, Lin MI, Wiegn P, Ringstedt T, Kraemer R, Hahn R, Wang S, Ibanez CF, Rafii S, Hempstead BL (2000) Brain derived neurotrophic factor is an endothelial cell survival factor required for intramyocardial vessel stabilization. Development 127:4531–40
- Ebert DH, Greenberg ME (2013) Activity-dependent neuronal signalling and autism spectrum disorder. Nature 493:327–337
- Ehninger D, Han S, Shilyansky C, Zhou Y, Li W, Kwiatkowski DJ et al (2008) Reversal of learning deficits in a Tsc2+/- mouse model of tuberous sclerosis. Nature Med 14:843–848
- Ernfors P, Wetmore C, Olson L, Persson H (1990) Identification of cells in rat brain and peripheral tissues expressing mRNA for members of the nerve growth factor family. Neuron 5:511–526
- Ernfors P, Kucera J, Lee KF, Loring J, Jaenisch R (1995) Studies on the physiological role of brainderived neurotrophic factor and neurotrophin-3 in knockout mice. Int J Dev Biol 39:799–807
- Fauchais AL, Lalloue F, Lise MC, Boumediene A, Preud'homme JL, Vidal E, Jauberteau MO (2008) Role of endogenous brain-derived neurotrophic factor and sortilin in b cell survival. J Immunol 181:3027–38
- Fifková E, Anderson CL, Young SJ, Van Harreveld A (1982) Effect of anisomycin on stimulationinduced changes in dendritic spines of the dentate granule cells. J Neurocyt 11:183–210
- Finkbeiner S, Tavazoie SF, Maloratsky A, Jacobs KM, Harris KM, Greenberg ME (1997) CREB: a major mediator of neuronal neurotrophin responses. Neuron 19:1031–1047
- Fiore R, Khudayberdiev S, Christensen M, Siegel G, Flavell SW, Kim TK, Greenberg ME, Schratt G (2009) Mef2-mediated transcription of the miR379-410 cluster regulates activity-dependent dendritogenesis by fine-tuning Pumilio2 protein levels. EMBO J 28:697–710
- Fortin DA, Srivastava T, Dwarakanath D, Pierre P, Nygaard S, Derkach VA, Soderling TR (2012) Brain-derived neurotrophic factor activation of CaM-kinase kinase via transient receptor potential canonical channels induces the translation and synaptic incorporation of GluA1containing calcium-permeable AMPA receptors. J Neurosci 32:8127–8137
- Gal-Ben-Ari S, Kenney JW, Ounalia-Saad H, Taha E, David O, Levitan D, Gildish I, Panja D, Pai B, Wibrand K, Simpson TI, Proud CG, Bramham CR, Armstrong JD, Rosenblum K (2012) Consolidation and translation regulation. Learn Mem 19:410–422
- Gao J, Wang WY, Mao YW, Graff J, Guan JS, Pan L, Mak G, Kim D, Su SC, Tsai LH (2010) A novel pathway regulates memory and plasticity via SIRT1 and miR-134. Nature 466:1105–1109
- Gingras AC, Raught B, Sonenberg N (1999) eIF4 initiation factors: effectors of mRNA recruitment to ribosomes and regulators of translation. Annu Rev Biochem 68:913–963
- Gingras AC, Raught B, Sonenberg N (2004) mTOR signaling to translation. Curr Top Microbiol Immunol 279:169–197
- Giraldez AJ, Cinalli RM, Glasner ME, Enright AJ, Thomson JM, Baskerville S, Hammond SM, Bartel DP, Schler AF (2005) microRNAs regulate brain morphogenesis in zebrafish. Science 308:833–838
- Givalois L, Marmigère F, Rage F, Ixart G, Arancibia S, Tapia-Arancibia L (2001) Immobilization stress rapidly and differentially modulates BDNF and TrkB mRNA expression in the pituitary gland of adult male rats. Neuroendocrinology 74:148–159

- Givalois L, Naert G, Rage F, Ixart G, Arancibia S, Tapia-Arancibia L (2004) A single brain-derived neurotrophic factor injection modifies hypothalamo–pituitary–adrenocortical axis activity in adult male rats. Mol Cell Neurosci 27:280–295
- Gkogkas CG, Khoutorsky A, Ran I, Rampakakis E, Nevarko T, Weatherill DB et al (2013) Autismrelated deficits via dysregulated eIF4E-dependent translational control. Nature 493:371–377
- Griesbeck O, Parsadanian AS, Sendtner M, Thoenen H (1995) Expression of neurotrophins in skeletal muscle: quantitative comparison and significance for motoneuron survival and maintenance of function. J Neurosci Res 42:21–33
- Guire ES, Oh MC, Soderling TR, Derkach VA (2008) Recruitment of calcium-permeable AMPA receptors during synaptic potentiation is regulated by CaM-kinase I. J Neurosci 28:6000–6009
- Han JM, Sahin M (2011) TSC1/TSC2 signaling in the CNS. FEBS Lett 58:973–980
- Hanus C, Kochen L, tom Dieck S, Racine V, Sibarita JB, Schuman EM, Ehlers MD (2014) Synaptic control of secretory trafficking in dendrites. Cell Rep 7:1771–1778
- Heldt SA, Stanek L, Chhatwal JP, Ressler KJ (2007) Hippocampus-specific deletion of BDNF in adult mice impairs spatial memory and extinction of aversive memories. Mol Psychiatr 12:656–670
- Heo I, Joo C, Cho J, Ha M, Han J, Kim VN (2008) Lin28 mediates the terminal uridylation of let-7 precursor microRNA. Mol Cell 32:276–284
- Heo I, Joo C, Kim YK, Ha M, Yoon MJ, Cho J, Yeom KH, Han J, Kim VN (2009) TUT4 in concert with Lin28 suppresses microRNA biogenesis through pre-microRNA uridylation. Cell 138:696–708
- Holt CE, Schuman EM (2013) The central dogma decentralized: new perspectives on RNA function and local translation in neurons. Neuron 80:648–657
- Horch HW, Katz LC (2002) BDNF release from single cells elicits local dendritic growth in nearby neurons. Nat Neurosci 5:1177–1184
- Huang YS, Jung MY, Sarkissian M, Richter JD (2002) N-methyl-D-aspartate receptor signaling results in Aurora kinase-catalyzed CPEB phosphorylation and alpha CaMKII mRNA polyadenylation at synapses. EMBO J 21:2139–2148
- Huang YW, Ruiz CR, Eyler EC, Lin K, Meffert MK (2012) Dual regulation of miRNA biogenesis generates target specificity in neurotrophin-induced protein synthesis. Cell 148:933–946
- Huber LJ, Hempstead B, Donovan MJ (1996) Neurotrophin and neurotrophin receptors in human fetal kidney. Dev Biol 179:369–381
- Ideker T, Thorsson V, Ranish JA, Christmas R, Buhler J, Eng JK, Bumgarner R, Goodlett DR, Aebersold R, Hood L (2013) Integrated genomic and proteomic analyses of a systematically perturbed metabolic network. Science 292:929–934
- Inamura N, Nawa H, Takei N (2005) Enhancement of translation elongation in neurons by brainderived neurotrophic factor: implications for mammalian target of rapamycin signaling. J Neurochem 95:1438–1445
- Jaworski J, Spangler S, Seeburg DP, Hoogenraad CC, Sheng M (2005) Control of dendritic arborization by the phosphoinositide-30-kinase-Akt-mammalian target of rapamycin pathway. J Neurosci 25:11300–11312
- Je HS, Lu Y, Yang F, Nagappan G, Zhou J, Jiang Z et al (2009) Chemically inducible inactivation of protein synthesis in genetically targeted neurons. J Neurosci 29:6761–6766
- Jeanneteau FD, Lambert WM, Ismaili N, Bath KG, Lee FS, Garabedian MJ, Chao MV (2012) BDNF and glucocorticoids regulate corticotrophin-releasing hormone (CRH) homeostasis in the hypothalamus. Proc Natl Acad Sci U S A 109:1305–1310
- Jensen T, Johnson AL (2001) Expression and function of brain-derived neurotrophin factor and its receptor, TrkB, in ovarian follicles from the domestic hen. J Exp Biol 204:2087–95
- Ji Y, Lu Y, Yang F, Shen W, Tang TTT, Feng L, Duan S, Lu B (2010) Acute and gradual increases in BDNF concentration elicit distinct signaling and functions in neurons. Nat Neurosci 13:302–309
- Jia JM, Chen Q, Zhou Y, Miao S, Zheng J, Zhang C, Xiong ZQ (2008) Brain-derived neurotrophic factor-tropomyosin-related kinase B signaling contributes to activity-dependent changes in synaptic proteins. J Biol Chem 283:21242–21250

- Johnson JE, Barde YA, Schwab M, Thoenen H (1986) Brain-derived neurotrophic factor supports the survival of cultured rat retinal ganglion cells. J Neurosci 6:3031–3038
- Jones KR, Fariñas I, Backus C, Reichardt LF (1994) Targeted disruption of the BDNF gene perturbs brain and sensory neuron development but not motor neuron development. Cell 76:989–999
- Jourdi H, Hsu YT, Zhou M, Qin Q, Bi X, Baudry M (2009) Positive AMPA receptor modulation rapidly stimulates BDNF release and increases dendritic mRNA translation. J Neurosci 29:8688–8697
- Juhila J, Sipila T, Icay K, Nicorici D, Ellonen P, Kallio A, Korpelainen E, Greco D, Hovatta I (2011) MicroRNA expression profiling reveals miRNA families regulating specific biological pathways in mouse frontal cortex and hippocampus. PLoS One 6:e21495
- Kajiya M, Shiba H, Fujita T, Takeda K, Uchida Y, Kawaguchi H et al (2009) Brain-derived neurotrophic factor protects cementoblasts from serum starvation-induced cell death. J Cell Physiol 221:696–706
- Kalita K, Kharebava G, Zheng JJ, Hetman M (2006) Role of megakaryoblastic acute leukemia-1 in ERK1/2-dependent stimulation of serum response factor-driven transcription by BDNF or increased synaptic activity. J Neurosci 26:10020–10032
- Kang H, Schuman EM (1996) A requirement for local protein synthesis in neurotrophin-induced hippocampal synaptic plasticity. Science 273:1402–1406
- Kang H, Welcher AA, Shelton D, Schuman EM (1997) Neurotrophins and time: different roles for TrkB signaling in hippocampal long-term potentiation. Neuron 19:653–664
- Kanhema T, Dagestad G, Panja D, Tiron A, Messaoudi E, Havik B, Ying SW, Nairn AC, Sonenberg N, Bramham CR (2006) Dual regulation of translation initiation and peptide chain elongation during BDNF-induced LTP in vivo: evidence for compartment-specific translation control. J Neurochem 99:1328–1337
- Kawashima H, Numakawa T, Kumamaru E, Adachi N, Mizuno H, Ninomiya M, Kunugi H, Hashido K (2010) Glucocorticoid attenuates brain-derived neurotrophic factor-dependent upregulation of glutamate receptors via the suppression of microRNA-132 expression. Neuroscience 165:1301–1311
- Kelleher RJ, Govindarajan A, Jung HY, Kang H, Tonegawa S (2004) Translational control by MAPK signaling in long-term synaptic plasticity and memory. Cell 116:467–479
- Kilian O, Hartmann S, Dongowski N, Karnati S, Baumgart-Vogt E, Hartel FV, Noll T, Schnettler R, Lips KS (2014) BDNF and its TrkB receptor in human fracture healing. Ann Anat 196:286–295
- Klann E, Denver TE (2004) Biochemical mechanisms for translational regulation in synaptic plasticity. Nat Rev Neurosci 5:931–942
- Klose RJ, Sarraf SA, Schmiedeberg L, McDermott SM, Stancheva I, Bird AP (2005). DNA binding selectivity of MeCP2 due to a requirement for A/T sequences adjeacent to methyl-CpG. Mol Cell 19:667–678
- Korte M, Carroll P, Wolf E, Brem G, Thoenen H, Bonhoeffer T (1995) Hippocampal long-term potentiation is impaired in mice lacking brain- derived neurotrophic factor. Proc Natl Acad Sci U S A 92:8856–8860
- Krichevsky AM, King KS, Donahue CP, Khrapko K, Kosik KS (2003) A microRNA array reveals extensive regulation of microRNAs during brain development. RNA 9:1274–1281
- Krug M, Lössner B, Ott T (1984) Anisomycin blocks the late phase of long-term potentiation in the dentate gyrus of freely moving rats. Brain Res Bull 13:39–42
- Lai KO, Wong AS, Cheung MC, Xu P, Liang Z, Lok KC, Xie H, Palko ME, Yung WH, Tessarollo L, Cheung ZH, Ip NY (2012) TrkB phosphorylation by Cdk5 is required for activity-dependent structural plasticity and spatial memory. Nat Neurosci 15:1506–1515
- Lamprecht R, LeDoux J (2004) Structural plasticity and memory. Nat Rev Neurosci 5:45-54
- Lauterborn JC, Rex CS, Kramár E, Chen LY, Pandyarajan V, Lynch G, Gall CM (2007) Brain-derived neurotrophic factor rescues synaptic plasticity in a mouse model of fragile X syndrome. J Neurosci 27:10685–10694

- Lazo OM, Gonzalez A, Ascaño M, Kuruvilla R, Couve A, Bronfman FC (2013) BDNF regulates Rab11-mediated recycling endosome dynamics to induce dendritic branching. J Neurosci 33:6112–6122
- Lee RC, Feinbaum RL, Ambros V (1993) The C. elegans heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14. Cell 75:843–854
- Li W, Calfa G, Larimore J, Pozzo-Miller L (2012) Activity-dependent BDNF release and TRPC signaling is impaired in hippocampal neurons of Mecp2 mutant mice. Proc Natl Acad Sci U S A 109:17087–17092
- Liao L, Pilotte J, Xu T, Wong CC, Edelman GM, Vanderklish P, Yates JR (2007) BDNF induces widespread changes in synaptic protein content and upregulates components of the translation machinery: an analysis using high throughput proteomics. J Proteome Res 6:1059–1071
- Lin SY, Wu K, Levine ES, Mount HT, Suen PC, Black IB (1998) BDNF acutely increases tyrosine phosphorylation of the NMDA receptor subunit 2B in cortical and hippocampal postsynaptic densities. Mol Brain Res 55:20–27
- Loohuis NO, Kos A, Marens GJ, Van Bokhoven H, Kasri NN, Aschrafi A (2012) MicroRNA networks direct neuronal development and plasticity. Cell Mol Life Sci 69:89–102
- Louhivuori V, Vicario A, Uutela M, Rantamäki T, Louhivuori LM, Castrén E et al (2011) BDNF and TrkB in neuronal differentiation of *Fmr1*-knockout mouse. Neurobiol Dis 41:469–480
- Lugli G, Larson J, Martone ME, Jones Y, Smalheiser NR (2005) Dicer and eIF2c are enriched at postsynaptic densities in adult mouse brain and are modified by neuronal activity in a calpaindependent manner. J Neurochem 94:896–905
- Lugli G, Torvik VI, Larson J, Smalheiser NR (2008) Expression of microRNAs and their precursors in synaptic fractions of adult mouse forebrain. J Neurochem 106:650–661
- Lugli G, Larson J, Demars MP, Smalheiser NR (2012) Primary microRNA precursor transcripts are localized at post-synaptic densities in adult mouse forebrain. J Neurochem 123:459–466
- Manadas B, Santos AR, Szabadfi K, Gomes JR, Garbis SD, Fountoulakis M, Duarte CB (2009) BDNF induced changes in the expression of the translation machinery in hippocampal neurons: protein levels and dendritic mRNA. J Proteome Res 8:4536–4552
- Martinowich K, Lu B (2007) Interaction between BDNF and serotonin: role in mood disorders. Neuropsychopharmacology 33:73–83
- Martinowich K, Hattori D, Wu H, Fouse S, He F, Hu Y et al (2003) DNA methylation-related chromatin remodeling in activity-dependent BDNF gene regulation. Science 302:890–893
- Matsuda N, Lu H, Fukata Y, Noritake J, Gao H, Mukherjee S, Nemoto T, Fukata M, Poo MM (2009) Differential activity-dependent secretion of brainderived neurotrophic factor from axon and dendrite. J Neurosci 29:14185–14198
- McAllister AK, Katz LC, Lo DC (1999) Neurotrophins and synaptic plasticity. Annu Rev Neurosci 22:295–318
- Messaoudi E, Kanhema T, Soule J, Tiron A, Dagyte G, da Silva B, Bramham CR (2007) Sustained Arc/Arg3.1 synthesis controls long-term potentiation consolidation through regulation of local actin polymerization in the dentate gyrus in vivo. J Neurosci 27:10445–10455
- Moss EG, Lee RC, Ambros V (1997) The cold shock domain protein LIN-28 controls developmental timing in C. elegans and is regulated by the lin-4 RNA. Cell 88:637–646
- Muglia P, Vicente AM, Verga M, King N, Macciardi F, Kennedy JL (2002) Association between the BDNF gene and schizophrenia. Mol Psychiatr 8:146–147
- Nagahara AH, Tuszynski MH (2011) Potential therapeutic uses of BDNF in neurological and psychiatric disorders. Nat Rev Drug Discov 10:209–219
- Nam Y, Chen C, Gregory RI, Chou JJ, Sliz P (2011) Molecular basis for interaction of let-7 microRNAs with Lin28. Cell 147:1080–1091
- Napoli I, Mercaldo V, Boyl PP, Eleuteri B, Zalfa F, De Rubeis S, Di Marino D, Mohr E, Massimi M, Falconi M, Witke W, Costa-Mattioli M, Sonenberg N, Achsel T, Bagni C (2008) The fragile X syndrome protein represses activity-dependent translation through CYFIP1, a new 4E-BP. Cell 134:1042–1054
- Narayanan U, Nalavadi V, Nakamoto M, Pallas DC, Ceman S, Bassell GJ, Warren ST (2007) FMRP phosphorylation reveals an immediate-early signaling pathway triggered by group I mGluR and mediated by PP2A. J Neurosci 27:14349–14357

- Nikulina EM, Johnston CE, Wang J, Hammer RP (2014) Neurotrophins in the ventral tegmental area: Role in social stress, mood disorders and drug abuse. Neuroscience 282C:122–138
- Niere F, Wiklerson JR, Huber KM (2012) Evidence for a fragile X mental retardation proteinmediated translational switch in metabotropic glumatate receptor-triggered Arc translation and long-term depression. J Neurosci 32:5924–5936
- Ostroff LE, Fiala JC, Allwardt B, Harris KM (2002) Polyribosomes redistribute from dendritic shafts into spines with enlarged synapses during LTP in developing rat hippocampal slices. Neuron 35:535–545
- Pardon MC (2010) Role of neurotrophic factors in behavioral processes: implications for the treatment of psychiatric and neurodegenerative disorders. Vitam Horm 82:185–200
- Paredes A, Romero C, Dissen GA, DeChiara TM, Reichardt L, Cornea A, Ojeda SR, Xu B (2004) TrkB receptors are required for follicular growth and oocyte survival in the mammalian ovary. Dev Biol 267:430–449
- Patterson SL, Abel T, Deuel TA, Martin KC, Rose JC, Kandel ER (1996) Recombinant BDNF rescues deficits in basal synaptic transmission and hippocampal LTP in BDNF knockout mice. Neuron 16:1137–1145
- Pfeiffer BE, Huber KM (2006) Current advances in local protein synthesis and synaptic plasticity. J Neurosci 26:7147–7150
- Poon MM, Choi SH, Jamieson CAM, Geschwind DH, Martin KC (2006) Identification of process-localized mRNAs from cultured rodent hippocampal neurons. J Neurosci 26:13390–13399
- Power AE, Berlau DJ, McGaugh JL, Steward O (2006) Anisomycin infused into the hippocampus fails to block "reconsolidation" but impairs extinction: the role of re-exposure duration. Learn Mem 13:27–34
- Raab-Graham KF, Haddick PC, Jan YN, Jan LY (2006) Activity and mTOR-dependent suppression of Kv1.1 channel mRNA translation in dendrites. Science 314:144–148
- Rao VR, Pintchovski SA, Chin J, Peebles CL, Mitra S, Finkbeiner S (2006) AMPA receptors regulate transcription of the plasticity-related immediate-early gene Arc. Nat Neurosci 9:887–895
- Rex CS, Lin CY, Kramár EA, Chen LY, Gall CM, Lynch G (2007) Brain-derived neurotrophic factor promotes long-term potentiation-related cytoskeletal changes in adult hippocampus. J Neurosci 27:3017–3029
- Riccio A, Alvania RS, Lonze BE, Ramanan N, Kim T, Huang Y, Dawson TM, Snyder SH, Ginty DD (2006) A nitric oxide signaling pathway controls CREB-mediated gene expression in neurons. Mol Cell 21:283–294
- Richter JD, Sonenberg N (2005) Regulation of cap-dependent translation by eIF4E inhibitory proteins. Nature 433:477–480
- Rios M (2013) BDNF and the central control of feeding: accidental bystander or essential player? Trends Neurosci 36:83–90
- Rivera C, Li H, Thomas-Crusells J, Lahtinen H, Viitanen T, Nanobashvili A, Kokaia Z, Airaksinen MS, Voipio J, Kaila K, Saarma M (2002) BDNF-induced TrkB activation down-regulates the K⁺ Cl⁻ cotransporter KCC2 and impairs neuronal Cl-extrusion. J Cell Biol 159:747–752
- Santos AR, Comprido D, Duarte CB (2010) Regulation of local translation at the synapse by BDNF. Prog Neurobiol 4:505–516
- Scheetz AJ, Nairn AC, Constantine-Paton M (2000) NMDA receptor-mediated control of protein synthesis at developing synapses. Nat Neurosci 3:211–216
- Schratt GM, Nigh EA, Chen WG, Hu L, Greenberg ME (2004) BDNF regulates the translation of a select group of mRNAs by a mammalian target of rapamycin-phosphatidylinositol 3-kinasedependent pathway during neuronal development. J Neurosci 24:7366–7377
- Schratt GM, Tuebing F, Nigh EA, Kane CG, Sabatini ME, Kiebler M, Greenberg ME (2006) A brain-specific microRNA regulates dendritic spine development. Nature 439:283–289
- Schuhmann B, Dietrich A, Sel S, Hahn C, Klingenspor M, Lommatzsch M, Gudermann T, Braun A, Renz H, Nockher WA (2005) A role for brain-derived neurotrophic factor in B cell development. J Neuroimmunol 163:15–23

- Schwanhausser B, Busse D, Li N, Dittmar G, Schunchhardt J, Wolf J, Chen W, Selbach M (2011) Global quantification of mammalian gene expression control. Nature 473:337–327
- Sempere LF, Freemantle S, Pitha-Rowe I, Moss E, Dmitrovsky E, Ambros V (2004) Expression profiling of mammalian microRNAs uncovers a subset of brain-expressed microRNAs with possible roles in murine and human neuronal differentiation. Genom Biol 5:R13
- Shinohara Y, Yahagi K, Kawano M, Nishiyori H, Kawazu C, Suzuki N, Manabe R, Hirase H (2011) miRNA profiling of bilateral rat hippocampus CA3 by deep sequencing. Biochem Biophys Res Commun 409:293–298
- Smalheiser NR (2008) Regulation of mammalian microRNA processing and function by cellular signaling and subcellular localization. BBA-Gene Reg Mech 1779:678–681
- Smalheiser NR, Lugli G (2009) microRNA regulation of synaptic plasticity. Neuromol Med 11:133–140
- Smart FM, Edelman GM, Vanderklish PW (2003) BDNF induces translocation of initiation factor 4E to mRNA granules: evidence for a role of synaptic microfilaments and integrins. Proc Natl Acad Sci U S A 100:14403–14408
- Spears N, Molinek MD, Robinson LL, Fulton N, Cameron H, Shimoda K, Telfer EE, Anderson RA, Price DJ (2003) The role of neurotrophin receptors in female germ-cell survival in mouse and human. Development 130:5481–5491
- Soule J, Messaoudi E, Bramham CR (2006) Brain-derived neurotrophic factor and control of synaptic consolidation in the adult brain. Biochem Soc Trans 34:600–604
- Srivastava T, Fortin DA, Nygaard S, Kaech S, Sonenberg N, Edelman AM, Soderling TR (2012) Regulation of neuronal mRNA translation by CaM-kinase I phosphorylation of eIF4GII. J Neurosci 32:5620–5630
- Stanton PK, Sarvey JM (1984) Blockade of long-term potentiation in rat hippocampal CA1 region by inhibitors of protein synthesis. J Neurosci 4:3080–3088
- Steward O, Levy WB (1982) Preferential localization of polyribosomes under the base of dendritic spines in granule cells of the dentate gyrus. J Neurosci 2:284–291
- Steward O, Schuman EM (2001) Protein synthesis at synaptic sites on dendrites. Annu Rev Neurosci 24:299–325
- Suen PC, Wu K, Levine ES, Mount HT, Xu JL, Lin SY, Black IB (1997) Brain-derived neurotrophic factor rapidly enhances phosphorylation of the postsynaptic N-methyl-D-aspartate receptor subunit 1. Proc Natl Acad Sci U S A 94:8191–8195
- Sutton MA, Schuman EM (2006) Dendritic protein synthesis, synaptic plasticity, and memory. Cell 127:49–58
- Takei N, Kawamura M, Hara K, Yonezawa K, Nawa H (2001) Brain-derived neurotrophic factor enhances neuronal translation by activation multiple initiation processes: comparison with the effects of insulin. J Biol Chem 276:42818–42825
- Takei N, Inamura N, Kawamura M, Namba H, Hara K, Yonezawa K, Nawa H (2004) Brainderived neurotrophic factor induces mammalian target of rapamycin-dependent local activation of translation machinery and protein synthesis in neuronal dendrites. J Neurosci 24:9760–9760
- Takei N, Kawamura M, Ishizuka Y, Kakiya N, Inamura N, Namba H, Nawa H (2009) Brainderived neurotrophic factor enhances the basal rate of protein synthesis by increasing active eukaryotic elongation factor 2 levels and promoting translation elongation in cortical neurons. J Biol Chem 284:26340–26348
- Takemoto-Kimura S, Ageta-Ishihara N, Nonaka M, Adachi-Morishima A, Mano T, Okamura M et al (2007) Regulation of dendritogenesis via a lipid-raft-associated Ca^{2+/}calmodulindependent protein kinase CLICK-III/CaMKIγ. Neuron 54:755–770
- Tanaka JI, Horiike Y, Matsuzaki M, Miyazaki T, Ellis-Davies GCR, Kasai H (2008) Protein synthesis and neurotrophin-dependent structural plasticity of single dendritic spines. Science 319:1683–1687
- Tian Q, Stepaniants SB, Mao M, Wong L, Feetham MC, Doyle MJ, Yi EC, Dai H, Thorsson V, Eng J, Goodlett D, Berger JP, Gunter B, Linseley PS, Soughton RB, Aebersold R, Collins SJ,

Hanlon WA, Hood LE (2004) Integrated genomic and proteomic analyses of gene expression in mammalian cells. Mol Cell Proteomics 3:960–969

- Tronel S, Milekic MH, Alberini CM (2005) Linking new information to a reactivated memory requires consolidation and not reconsolidation mechanisms. PLoS Biol 3:e293
- Tyler WJ, Pozzo-Miller LD (2001) BDNF enhances quantal neurotransmitter release and increases the number of docked vesicles at the active zones of hippocampal excitatory synapses. J Neurosci 21:4249–4258
- Tyler WJ, Pozzo-Miller L (2003) Miniature synaptic transmission and BDNF modulate dendritic spine growth and form in rat CA1 neurones. J Physiol 553:497–509
- Tyler WJ, Alonso M, Bramham CR, Pozzo-Miller LD (2002) From acquisition to consolidation: on the role of brain-derived neurotrophic factor signaling in hippocampal-dependent learning. Learn Mem 9:224–237
- Uutela M, Lindholm J, Louhivuori V, Wei H, Louhivuori LM, Pertovaara A et al (2012) Reduction of BDNF expression in Fmr1 knockout mice worsens cognitive deficits but improves hyperactivity and sensorimotor deficits. Genes Brain Behav 11:513–523
- Vessey JP, Schoderboeck L, Gingl E, Luzi E, Riefler J, Di Leva F, Karra D, Thomas S, Kiebler MA, Macchi P (2010) Mammalian Pumilio 2 regulates dendrite morphogenesis and synaptic function. Proc Natl Acad Sci U S A 107:3222–3227
- Vo N, Klein ME, Varlamova O, Keller DM, Yamamoto T, Goodman RH, Impey S (2005) A cAMPresponse element binding protein-induced microRNA regulates neuronal morphogenesis. Proc Natl Acad Sci U S A 102:16426–16431
- Wang DO, Kim SM, Zhao Y, Hwang H, Miura SK, Sossin WS, Martin KC (2009) Synapse-and stimulus–specific local translation during longterm neuronal plasticity. Science 324: 1536–1540
- Wang QZ, Lv YH, Gong YH, Li ZF, Xu W, Diao Y, Xu R (2012a) Double-stranded let-7 mimics, potential candidates for cancer gene therapy. J Physiol Biochem 68:107–119
- Wang W, Zhu JZ, Chang KT, Min KT (2012b) DSCR1 interacts with FMRP and is required for spine morphogenesis and local protein synthesis. J EMBO 31:3655–3666
- Wayman GA, Davare M, Ando H, Fortin D, Varlamova O, Cheng HY, Marks D, Obrietan K, Soderling TR, Goodman RH, Impey S (2008) An activity-regulated microRNA controls dendritic plasticity by down-regulating p250GAP. Proc Natl Acad Sci U S A 105:9093–9098
- Wetmore C, Ernfors P, Persson H, Olson L (1990) Localization of brain-derived neurotrophic factor mRNA to neurons in the brain by *in* situ hybridization. Exp Neurol 109:141–152
- Will TJ, Tushev G, Kochen L, Nassim-Assir B, Cajigas IJ, Schuman EM (2013) Deep sequencing and high-resolution imaging reveal compartment-specific localization of Bdnf mRNA in hippocampal neurons. Sci Signal 6:rs16
- Wu L, Wells D, Tay J, Mendis D, Abbott MA, Barnitt A, Quinlan E, Heynen A, Fallon JR, Richter JD (1998) CPEB-mediated cytoplasmic polyadenylation and the regulation of experiencedependent translation of alpha-CaMKII mRNA at synapses. Neuron 21:1129–1139
- Xu B, Zang K, Ruff NL, Zhang YA, McConnell SK, Stryker MP, Reichardt LF (2000) Cortical degeneration in the absence of neurotrophin signaling: dendritic retraction and neuronal loss after removal of the receptor TrkB. Neuron 26:233–245
- Yin Y, Edelman GM, Vanderklish PW (2002) The brain-derived neurotrophic factor enhances synthesis of Arc in synaptoneurosomes. Proc Natl Acad Sci U S A 99:2368–2373
- Ying SW, Futter M, Rosenblum K, Webber MJ, Hunt SP, Bliss TV, Bramham CR (2002) Brainderived neurotrophic factor induces long-term potentiation in intact adult hippocampus: requirement for ERK activation coupled to CREB and upregulation of Arc synthesis. J Neurosci 22:1532–1540
- Zhong J, Zhang T, Bloch LM (2006) Dendritic mRNAs encode diversified functionalities in hippocampal pyramidal neurons. BMC Neurosci 17:17
- Zuccato C, Cattaneo E (2009) Brain-derived neurotrophic factor in neurodegenerative disease. Nat Rev Neurol 5:311–322



Correction to: Post-transcriptional Regulation of Cytokine Signaling During Inflammatory Responses

Irina Vlasova-St. Louis and Paul R. Bohjanen

Correction to: Chapter 3 in: K. M. J. Menon, A. C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation,* https://doi.org/10.1007/978-3-319-25124-0_3

The original online version of this chapter was inadvertently published with part of last name of the author "Irina Vlasova-St. Louis" included along with the first name. The correct first and last name of this author is given below:

First name: Irina Last name: Vlasova-St. Louis

The updated online version of this chapter can be found at https://doi.org/10.1007/978-3-319-25124-0_3

Index

A

- Activation/repression, 2, 10, 40, 56, 77, 98, 101, 112, 130, 133, 159, 185, 194, 223, 236, 244, 255, 263, 278, 279, 316, 324
- Alternative splicing, 25, 26, 38–40, 43, 45, 48, 94, 97, 137, 158–160, 162–164, 166, 172, 194
- ARE binding proteins (AREBP), 14, 18, 238, 243
- AUF1, 58, 93, 97, 165, 239–244, 246–247, 286
- AU-rich element (ARE), 10, 14, 15, 18, 56–59, 61, 63–66, 93–95, 144, 164, 165, 188–196, 203, 204, 238–243, 246, 278, 281, 283, 285, 286, 289, 294, 304

B

Brain-derived neurotrophic factor (BDNF), 315–329

С

Calcium, 185, 235–243, 245–247, 317 Circular RNAs, 20, 195 Combinatorial control, 142 Competing Endogenous RNAs (ceRNAs), 26 Competitive binding, 58, 161, 171 CUGBP2, 190, 194, 204, 286 Cytoplasmic polyadenylation, 130, 132, 322 Cytoplasmic Poly-adenylation element binding protein (CPEB), 9–11, 130, 132, 134–136, 139, 142, 143, 146, 147, 322

D

DAZ, Boule, 146 Deadenylases, 12–14

- Deadenylation, 10–13, 14, 19, 23, 58, 144, 187, 192, 203, 238, 263, 287
- Decapping, 10, 12, 13, 15, 16, 18, 24, 81, 82, 112, 144, 187, 188, 192, 203, 238, 287
- Developmental regulation, 47
- Dicer, 17, 18, 21, 22, 110-113, 120, 169, 195,
- 244, 245, 263, 297, 300–301, 323–325
- Differential mRNA stability, 38, 114, 115, 298
- Drosha, 17, 18, 21, 110–113, 169, 195, 244, 245, 263, 296, 297, 300–302, 324

E

eIF4E binding protein (4EBP), 318, 322 eIF5A, 82, 84 Endogenous siRNAs, 22 Endoribonuclease, 17, 241 Estrogen, 95, 96 Estrogen receptor mRNA Eukaryotic Elongation Factor 2 (eEF2), 6, 319 Exoribonuclease, 15, 16, 238 Exosome, 13

F

Fragile X mental retardation protein (FMRP), 43, 322, 328

G

Gene regulation, 19, 20, 91, 102, 157, 158, 277, 281, 282, 284, 291, 296 Germ line granules, 109

- Global translation regulation, 321
- Glucocorticoid, 26, 93, 97, 101, 279, 292, 293, 296, 298, 300–301, 303
- G protein coupled receptor (GPCR), 71, 72
- GU-rich elements (GREs), 18, 56, 57, 59–65, 285

© Springer International Publishing Switzerland 2016 K.M.J. Menon, A.C. Goldstrohm (eds.), *Post-transcriptional Mechanisms in Endocrine Regulation*, DOI 10.1007/978-3-319-25124-0

Н

- HuD RNA binding Protein, 43, 165
- HuR, 18, 42, 43, 58, 93, 96, 98, 165, 171, 172, 190–191, 193, 194, 197, 204, 238, 286, 292, 295
- Hypoxia, 157–159, 161, 162, 164, 167, 169, 171, 190, 193

I

- Insulin, 8, 18, 20, 37–45, 47, 48, 71, 121, 135, 193, 198, 200, 222–227, 229
- Insulin-like growth factor (IGFs), 37, 38, 45, 48
- Internal ribosome entry site (IRES), 11, 162, 166–169, 171, 172

K

- Kozak sequence, 5, 7, 92, 141
- KSRP, 58, 238, 239, 241–244, 246–247, 286, 292, 296, 302

L

LH receptor, 71, 72, 74–79, 81, 83–85, 119 LHR mRNA binding protein (LRBP), 74–85, 119, 264, 266 Ligand-induced downregulation, 74, 76, 83–85 Local translation, 319, 327 Long term potentiation (LTP), 322, 327–328

M

Maskin, 10, 11, 135, 143, 145, 322 Memory formation, 327 Mevalonate kinase, 78, 79, 83, 264 MicroRNA (miRNAs), 2, 7, 17, 20-24, 26, 38-42, 45-46, 48, 83-85, 92, 95, 98-102, 110, 114, 116, 118-122, 165, 169, 171, 187, 189, 191, 195-198, 201-204, 238, 244-247, 263-265, 267-269, 278, 280, 282, 284, 286, 289, 296-303, 305, 323-326 biogenesis, 21 synthesis, processing, 113, 296, 300, 302, 304, 323–325 mediated repression, 26 sponges, 110 miR-122, 83-85, 119, 202, 264, 266, 267 mRNA decay, 10, 12–20, 23, 24, 27, 38, 56-59, 64, 65, 77, 83, 97, 187-190, 192, 194-196, 201, 203, 239-242, 244, 246, 282-284, 286, 287, 289, 291, 292, 294, 296, 304

- mRNA degradation, 1, 7, 10, 20, 24, 27, 47, 56, 60, 61, 74, 77, 91, 93, 165, 192, 193, 200, 203, 239, 241, 244, 263
- mRNA localization, 133-135, 187, 241
- mRNA protein complex (mRNP), 57, 82, 83, 137, 141, 224
- mRNA stability, 1, 18, 24, 26, 38, 39, 43, 47, 48, 56, 86, 91, 92, 95, 96, 100, 102, 110, 144, 164, 166, 171, 187, 190, 235, 237–243, 246–247, 279, 281, 283–286, 288, 292, 304
- mRNA stabilization, 92, 158, 189–191, 194, 204, 239, 283, 285, 292, 304 Musashi, 138, 147

Ν

Nanos, 132, 133, 136, 138 Non-coding RNA, 19–27

0

Ovary, 71, 73, 74, 80, 84, 86, 109, 112, 114, 115, 117–122, 142, 186, 255, 256, 258, 261, 262, 264–269, 317

Р

Parathyroid hormone, 17, 236 Processing bodies (P-bodies), 187, 189, 191-194, 203-204 Phosphate, 21, 79, 235-240, 242, 243, 246-247, 262 Poly-adenosine binding protein (PABP), 5, 7, 8, 10, 12-14, 23, 24, 44, 139 Polyadenosine ribonuclease (PARN), 10, 14, 19, 187 Polysomes, 17, 81, 145-147, 165 Post-transcriptional regulation, 38, 39, 91, 101.109 Prostanoid synthesis, 181-186 PTB, 43, 44, 165 Pumilio and Fem-3 binding factor proteins (PUFs), 139

R

- Regulation of elongation, 6-7, 11
- Ribosomal protein S6 Kinase (S6K1), 224
- RNA binding proteins, 1, 4, 7, 9, 14, 15, 16, 18, 19, 26, 38, 74, 78, 79, 94, 95, 130, 246
- RNA decay, 12, 18, 82
- RNA induced silencing complex (RISC), 16, 22–24, 38, 95, 110–112, 195, 238, 244, 245, 287, 296

RNA localization, 43 RNA processing, 1

S

S6K kinase, 224
SREBP, 83
Steroid hormones, 254
Stress, 2, 7, 8, 17, 18, 26, 97, 162, 164, 167, 169, 189, 190, 194, 203–204, 267, 268, 278, 282, 286, 296, 317
Stress granules (SG), 189, 194, 203–204, 286, 296
Synapses, 316, 319–323, 327, 329
Synaptic plasticity, 316, 318, 319, 322, 323, 327

Т

- Target of Rapamycin (TOR), 8 TORC1, 223
 - TORC2, 223
- T cell intracellular antigen 1 (TIA-1), 43, 58, 190, 194–195, 204
- 5' Terminal oligopyrimidine (TOP) mRNAs, 7
- Terminal uridyl transferases (TUTase), 21, 46
- Transcription, 137, 259, 292
- Translational activation, 10, 135, 137, 143, 147, 148
- Translational control, 2, 8, 20, 26, 130, 133–135, 137, 141, 143, 172, 224, 292

- Translational suppression, 38, 80, 81, 83, 84, 189, 194, 203, 245, 264, 304, 323 Translation initiation and elongation, 24, 318 TTP, 10, 14, 15, 18, 58, 165, 190, 192–193,
- 195, 197, 203, 204, 278, 280, 281, 284, 286–289, 291, 292, 294, 296, 302–304

U

Untranslated regions, (UTRs), 4–7, 9–11, 18, 23, 24, 26, 39, 40, 42–48, 56–59, 73, 92–102, 111, 112, 117, 119, 130, 133–135, 137, 139–143, 145–147, 158, 162, 164–171, 187–191, 194–196, 198, 199, 201, 204, 224, 238–243, 246–247, 263, 267, 278, 281, 283–286, 288, 289, 294–296, 304, 316, 320, 322, 326 UNR, 240, 241 Upstream microORFs, 169

V

Vascular endothelial growth factor (VEGF), 12, 58, 65, 86, 157–172, 193, 280–282 Vitamin D, 235

Y

Yeast two hybrid screen, 193